2.5 GEOLOGY, SEISMOLOGY, AND GEOTECHNICAL ENGINEERING

2.5.1 Basic Geologic and Seismic Information

The Beaver Valley Power Station - Unit 2 (BVPS-2) site is located on the south bank of the Ohio River in the borough of Shippingport, Pennsylvania, approximately 25 miles northwest of Pittsburgh. As shown on Figure 2.5.1-1, the site lies near the center of the Appalachian Plateau physiographic province. The plant is located upon a terrace of alluvial gravels about 100 feet thick, deposited by higher stages of the Ohio River during the Pleistocene Epoch. The surrounding topography consists of steep-sided, flat-topped hills separated by narrow stream valleys.

Two flat-lying bedrock formations of Pennsylvanian age outcrop in the site vicinity and form the adjacent hills, which rise to el 1,200 feet, as shown on Figure 2.5.1-2. The rocks are unmetamorphosed and show no evidence of major geologic deformation or tectonic activity. Regionally, the site lies near the center of a basin structure of Permian age (Figure 2.5.1-3). The Upper Freeport coal seam lies at el 900 feet, approximately 150 feet above the plant elevation and has been exploited to some extent in the site area. Several thinner coal seams are believed to exist beneath the site between 0 and 200 feet depth, but are not of sufficient extent or quality for commercial development at this time.

Massive sandstones were utilized locally in the past as dimension blocks for use in nearby bridge abutments, retaining walls, and building construction. None of this activity is carried on presently in the site area.

Minor oil and gas production has been realized within 4 miles of the site, drawing mainly from the Pocono Group of Early Mississippian age. There has been neither mining activity nor hydrocarbon extraction beneath the site which could cause ground settlement or collapse, nor is any anticipated.

The regional geologic history of the site region is discussed in Section 2.5.1.1.4, and reveals that the site area has not been subjected to severe diastrophic events at any time since the Precambrian. The area has remained tectonically stable since the Allegheny orogeny, approximately 250 million years ago, with the exception of epeirogenic uplifts, downwarping, and rebound, due to glacial loading.

The site is located in an area of very low seismicity within the Appalachian Plateau tectonic province as shown on Figure 2.5.1-5.

The nearest earthquake of epicentral Intensity V Modified Mercalli (MM), or greater, took place June 27, 1906, at Fairport, Ohio (near Cleveland), 80 miles northwest of the site. Only one earthquake has been reported within 50 miles of the site, reported at Sharon, Pennsylvania, on August 17, 1873, approximately 40 miles north of the site. Limited details have resulted in an estimated intensity of

III-IV (MM). The largest earthquake in the site region is the intensity VII-VIII (MM) event which occurred March 8, 1937, near Anna, Ohio, approximately 200 miles southwest of the site.

Results of geologic mapping in the site area indicate there is no hazard of surface faulting at or near the site. Additionally, no hazard due to ground subsidence is present which could affect the site.

In general, the founding material at the plant site is glacial outwash, overlain locally by thin deposits of silt, sand, and clay deposited during higher stages of the Ohio River. The founding materials for the plant structures are discussed in detail in Section 2.5.4.

Exploratory borings have shown that the bedrock beneath the site is a hard, black shale, and gray sandstone, probably belonging to the Allegheny Formation of Pennsylvanian age. It is estimated to be 350 feet thick in the plant area and shows no indication of extensive weathering, solution cavities, or other deleterious characteristics.

The work described herein was performed by Stone & Webster Engineering Corporation (SWEC), Boston, Massachusetts, with the following exceptions:

- 1. Raymond International, Hackensack, New Jersey, performed subsurface test borings, standard penetration tests, and soil, and rock sampling in the site area under the direction of SWEC.
- 2. Weston Geophysical Engineers, Inc., Westboro, Massachusetts, performed in situ seismic velocity measurements, including seismic refraction profiles and cross-hole, down-hole, and up-hole seismic tests for determining shear moduli of the overburden and bedrock characteristics. They also prepared the original seismicity analysis.
- 3. Pennsylvania Drilling Company, Inc., Pittsburgh, Pennsylvania, performed subsurface test borings, standard penetration tests, soil and rock sampling, and piezometer installations under the direction of SWEC.
- 4. Eger Drilling Company, Inc., Bridgeville, Pennsylvania performed subsurface test borings, standard penetration tests, soil sampling and piezometer installations under the direction of SWEC.

2.5.1.1 Regional Geology

2.5.1.1.1 Regional Physiography

The BVPS-2 site lies on the south bank of the Ohio River within the Appalachian Plateau physiographic province (Figure 2.5.1-1) (Fenneman 1938). This province is characterized by relatively undeformed

Paleozoic sediments which have been gently tilted and extensively dissected, producing the appearance of a rejuvenated peneplain. In Beaver County, the flat hill tops lie at approximately el 1,200 feet above narrow valleys cut 200 to 400 feet below the top of the hills.

Alluvial deposits of varying thickness are found at various elevations on the valley walls, and consist mainly of interbedded sands and gravels. These deposits frequently occur as terraces, and are the result of higher stages of rivers carrying large amounts of glacial outwash during the Pleistocene. The limit of glacial ice has been mapped as being approximately 20 miles north of the site (Shepps et al 1959).

Geologically, the province is a broad, gentle basin whose youngest rocks are the Dunkard Group of Early Permian age. The province is bounded on the west by the Central Lowland province, the boundary being an escarpment of Pennylvanian rocks 1,000 feet high in Tennessee and Kentucky, and a lower escarpment of Mississippian rocks in central southern Ohio. The boundary is somewhat indistinct in central Ohio, but generally follows the limit of glacial till plains. To the northwest and north of the site area, the boundary again follows an escarpment, formed by Silurian and Devonian sandstones, limestones, and Ordovician shales, to their contact with the Precambrian rocks of the Adirondacks (Fenneman 1938). The eastern boundary with the Valley and Ridge province occurs 105 miles east of the site and is marked by an abrupt topographic escarpment as much as 1,500 feet high, called the Allegheny Front. The Valley and Ridge province is characterized by a series of narrow, parallel ridges and valleys; the ridges being limbs of folds composed of resistant rock, and the valleys being the crests and troughs.

A portion of the Central Lowland physiographic province also lies within the site region, the boundary occurring 85 miles to the west. This province is characterized by having a Precambrian basement, or craton, with a veneer of nearly horizontal sedimentary rock of varying thicknesses. The structure is generally controlled by several basins and domes formed by Paleozoic epeirogenic activity. This has produced a diversity of geomorphic features, a large part of which has been extensively modified by Pleistocene glaciation (Eardley 1962).

2.5.1.1.2 Regional Stratigraphy

The distribution of the major geologic units within the site region is shown on Figure 2.5.1-3. A generalized cross section and regional stratigraphic column are given on Figures 2.5.1-4 and 2.5.1-6, respectively. The detailed stratigraphy of the Appalachian basin is thoroughly discussed by Colton (1970) and is summarized subsequently.

2.5.1.1.2.1 Precambrian

Rocks of Precambrian age are exposed in the site region, approximately 180 miles to the southeast in the Blue Ridge anticlinorium. These are a sequence of metasediments and metavolcanics which unconformably overlie what is commonly called Precambrian basement. The basement complex consists of schists, gneisses, and a wide variety of intrusives. The nearly peneplained basement surface has been determined to slope gently to the southeast under the region. Little is known about the stratigraphy of the Precambrian beneath the Appalachian basin.

2.5.1.1.2.2 Cambrian

Lower Cambrian deposition was concentrated in the southeast part of the Appalachian basin and resulted in a thick clastic wedge sequence, which included the sequence of rocks from the Loudoun to Waynesboro Formations. Equivalent rocks were not deposited on the west limb of the basin. Deposition on the western edge began in the Middle Cambrian with the Mt. Simon, Rome, and Conasauga Formations. The clastic sequence was followed in Middle and Late Cambrian time by basinwide deposition of carbonate rocks. These include the Elbrook Dolomite, Conococheague Limestone, the Lower Ordovician Beekmantown and Middle Ordovician Chambersburg, and Trenton Group rocks.

2.5.1.1.2.3 Ordovician

The carbonate sequence gives way abruptly during the Middle Ordovician to marine clastic deposits, beginning with the Martinsburg Shale. A thick clastic wedge about 8,000 feet thick is believed to have apexed in western Virginia and North Carolina during this time (Eardley 1962). Ordovician clastic sedimentation ended with the deposition of the Oswego Sandstone and Queenston Formation. Evidence exists for a diastrophic event affecting the Upper Ordovician along the east side of the basin, with some units missing, and an angular unconformity being developed (Colton 1970).

2.5.1.1.2.4 Silurian

The Upper Ordovician clastic sequence is overlain by a thin sequence of predominantly clastic rocks, mainly of Early Silurian age, that extends throughout most of the Appalachian basin. It is thickest in the northeast part, and thins to the west, southwest, and north, being totally absent in eastern New York. These rocks include some of the major ridge forming rocks of the Valley and Ridge, and include the Tuscarora Sandstone and Clinton Group.

A carbonate sequence began in the Middle Silurian, and continued into Late Silurian, Early Devonian, and even into the Middle Devonian in some areas. Like the underlying sequences, it is wedge-shaped, being thickest on the east and thinning out to the west and south. The sequence begins with the Lockport Dolomite, and includes the Salina and Bass Islands Group of Late Silurian age, deposited in evaporitic basin conditions, and the Helderberg Group. This sequence is overlain in most parts of the basin by the Oriskany Sandstone.

2.5.1.1.2.5 Devonian

Early Devonian carbonate deposition gave way to predominantly clastic sedimentation in the Middle and Late Devonian throughout most of the basin. The contact is conformable in most places and begins with the Needmore Shale. The sequence includes the Olentangy, Chemung, Catskill, and Bedford formations in the site region and ends before deposition of the Berea Sandstone or Pocono Group.

2.5.1.1.2.6 Mississippian

The Mississippian sequence conformably overlies the Devonian clastic sequence in most areas. The sequence is basically wedge-shaped with modifications due to erosion in eastern Ohio-western Pennsylvania, occurring in the Late Mississippian or Early Pennsylvanian time. The sequence includes the Pocono Group, the Greenbrier Group, and the Mauch Chunk Formation.

2.5.1.1.2.7 Pennsylvanian

Pemsylvanian strata are commonly disconformable on the Mississippian and are distinctly clastic. They include thick sequences of alternating beds of sandstone, shale, and siltstone with lesser amounts of coal and limestone. The Pottsville, Allegheny, Conemaugh, and Monongahela Formations were deposited at this time, and include the great coal-bearing formations of the basin. It has been determined that the Pennsylvanian sequence was originally much thicker, and more extensive, than at present, and nearly 75 percent of the sequence has been eroded since the Paleozoic (Colton 1970).

2.5.1.1.2.8 Permian

Overlying the Monongahela Formation in an oval area in West Virginia, southeast Ohio, and southwest Pennsylvania, is the Dunkard Group, of probable Lower Permian age. The Dunkard Group continued the variable deposition sequence started in the Pennsylvanian. The Allegheny orogeny, occurring progressively from the southeast during the Pennsylvanian and Permian, ended the extensive depositional cycle of the Paleozoic and exposed the sediments of the basin to a long period of erosion which is continuing today.

2.5.1.1.2.9 Triassic

The only rocks found in the site region younger than the Dunkard Group are located approximately 165 miles east of the site in the Gettysburg basin. They belong to the Newark Group and are believed to be Late Triassic or perhaps Early Jurassic in age. They are mainly continental sandstones, arkoses, conglomerates, and fanglomerates deposited in long, narrow, fault-bound basins. Numerous mafic sills and dikes are found associated with these deposits, and are believed to be the result of continental rifting during the Late Triassic and Early Jurassic.

2.5.1.1.2.10 Pleistocene

Unconformably overlying the Paleozoic rocks of northern Pennsylvania and Ohio are the unconsolidated deposits of several episodes of Pleistocene glaciation. Although glacial ice never advanced as far as the site area, the effects of its proximity are evident by the presence of high-level gravel terraces along the Ohio River and its major tributaries. These deposits provide the substrata on which the great majority of the cultural and industrial centers are founded.

Erosion since the retreat of the ice has considerably modified and removed much of the original deposits, but three terraces remain in the site area. The plant is situated on the uppermost terrace and is more fully described in Section 2.5.1.2.

2.5.1.1.3 Regional Structural Geology

The site lies on the west limb of the Appalachian sedimentological basin near the axis of the Appalachian coal basin (Rodgers 1970) or the axis of the Pittsburgh-Huntington basin (Wagner et al 1970). This area is just east of the Central Stable Region of North America and west of the Blue Ridge and Piedmont provinces as indicated on Figure 2.5.1-7.

All Carboniferous rocks in this area dip gently (less than 5 degrees) toward the basin axis, a line through Pittsburgh to Huntington, West Virginia. The basin contains the youngest sedimentary rocks in this part of the country, the Dunkard Group of Lower Permian age, exposed in the axial area 26 miles south of the site. The underlying Middle and Lower Paleozoic strata continue to thicken eastward, so that the axis of deposition is displaced somewhat east of Pittsburgh. The Precambrian surface also continues to dip southeastward under the entire Appalachian basin (Rodgers 1970). East of Pittsburgh, the dip of the Carboniferous strata reverses, and the units are deformed into broad gentle folds. This trend continues eastward into the Allegheny Mountains, until the Allegheny Front is reached at the outermost Carboniferous outcrop. Three anticlines form prominent ridges before the Allegheny Front is reached. These are the Chestnut Ridge, Laurel Hill, and Accident Mountain anticlines. Most of the structures found in the site region are the result of the Allegheny orogeny, which culminated in Late Permian time. No diastrophic events have occurred in the site region since the Early Jurassic.

2.5.1.1.3.1 Structures of the Appalachian Plateau

<u>Foldinq</u>

Major folds within 200 miles of the site are discussed in terms of their history of development, geologic setting, and effects on the geology of the site. The major structural features mentioned herein are shown on Figure 2.5.1-7.

Folds within the site region of the Appalachian Plateau are well displayed southeast of the site between Pittsburgh and the Allegheny

Front. Their wavelengths range from 6 to 12 miles and their structural reliefs vary from 500 to 2,500 feet (Rodgers 1970; Cardwell et al 1968). The three largest of these are the Chestnut Ridge, Laurel Hill, and Accident Mountain anticlines which bring uppermost Devonian rocks to the surface. They are 65, 75, and 85 miles from the site, respectively. Farther east, 150 miles from the site, the higher Deer Park anticline brings to the surface a larger section of Upper Devonian shale. Dips are usually less than 10 degrees and no regularity of plunge of the folds is apparent. Faulting at the surface is rare, but oil and gas drilling has revealed several major faults at depth, mostly in the Devonian section. Evidence indicates that these may be sole thrusts for westward movement of the overlying Plateau rocks (Gwinn 1964). The trend of the folds closely parallels the trend of the major folds in the Valley and Ridge of central Pennsylvania, swinging from northnortheast near Pittsburgh, to east-northeast in north-central Pennsylvania. Two of the Plateau folds can be traced southwestward into West Virginia, where they steepen significantly and increase in amplitude. The Deer Park anticline and the Briery Mountain anticline, the continuation of the Accident Mountain anticline, converge in West Virginia, 100 miles south of the site, and become the Elkins Valley anticline, whose west flank has a structural relief of 9,000 feet and is locally overturned. Thrust faults have been suggested beneath the anticline from well data (Rodgers 1970). All of the anticlines show westward offset across the west-northwest trending Morgantown-Sang Run and Fairmount-Rowlesburg lines. The nature of these features is not clearly understood at this time but may be related to reactivated basement fracture zones along which strike slip movement has occurred (Rodgers 1970; Cardwell et al 1968).

Of significantly different trend from these folds is the Burning Springs anticline in west-central West Virginia, about 100 miles southwest of the site. It trends nearly north-south across the center of the coal basin with a structural relief of 1,650 feet. The limbs dip very steeply, and the fold structure terminates rather abruptly at either end. Its existence has been interpreted to be due to several repetitions of the Devonian section along imbricate thrust surfaces, possibly facilitated by the presence of Salina Group salt beds (Rodgers 1970).

The folds of the Plateau are so parallel to those in the adjacent Valley and Ridge, that no one doubts their formation at the same time and by the same forces. The difference in complexity and degree of deformation between the two areas indicates that the stress levels were considerably lower in the Plateau, or the rocks responded to the force differently because of an anisotropic property of the rocks. Thin-skinned tectonics, with movement occurring along zones of salt or weak shales, seems to be the best explanation for the origin of structures found in the Appalachian Plateau of the site region (Rodgers 1963, 1970; Gwinn 1964).

Recent seismic reflection profiling in the southern Appalachians appears to confirm large scale decollement movement of rocks in the Appalachian Plateau and Valley and Ridge. Movement was generally to the northwest and occurred mainly during the Allegheny orogeny (Cook et al 1979, 1980).

Other folds of note exist in the Appalachian Plateau section of eastern Ohio. The Parkersburg-Lorain syncline is the westernmost fold of the western Appalachian basin, and can be traced from Parkersburg, West Virginia, to Lorain County on Lake Erie. The syncline is a structural trough trending N10W and is nearly 5 miles wide in the Marietta region, approximately 80 miles west of the site (Lemborn 1951). The Cambridge arch is the anticlinal counterpart of the Parkersburg-Lorain syncline, and parallels it to the east. It can be traced from the Ohio River in Washington County, Ohio, northwestward into Summit County. The structure has a relief of 450 feet in Washington County, but becomes less well defined northward (Lamborn 1951). Both of these folds are known to affect the Devonian shale sequence above the Onondaga Formation. The folds are underlain by pinchouts of bedded salt, and their location may be due to movements along this zone during the Allegheny orogeny.

<u>Faults</u>

The Clarendon-Linden fault zone is a major tectonic feature in the Appalachian Plateau and Central Stable Region of western New York. Ιt trends north-south for over 60 miles from near Lake Ontario to northern Allegheny County. The Clarendon-Linden fault is postulated to be not a single fault but instead, a zone consisting of several parallel basement faults which become surface flexures (Van Tyne 1976). Most of the movement is believed to be confined to those formations below the Silurian deposits. Movement is believed to have been downthrown to the east, reversing later to become downthrown to the west. A significant amount of seismic activity has taken place in the area of Attica, New York, in close proximity to the Clarendon-Linden structures (Sbar and Sykes 1977; Pomeroy et al in press). Recent low-level seismic activity has been correlated with highpressure fluid injection operations in brine fields which are developed in the area, and it is believed to be relieving stress along the fault system. For this reason, the fault must be considered an active feature. The south end of the fault zone is about 160 miles northeast of the BVPS-2 site. The seismicity of the Clarendon-Linden fault zone is discussed in Section 2.5.2. The Intensity VIII (MM) event in 1929 near Attica is the largest event to occur in western New York and has been correlated with the Clarendon-Linden structure.

Recent work in south-central Pennsylvania has resulted in a proposed fault zone near latitude 40°N called the Transylvania fault (Root and Hoskins 1977). This zone is believed by the authors to be a fundamental fracture of the continental plate, end has been traced from the Blue Ridge, across the Appalachian Plateau a few miles south of the BVPS-2 site. The fault is believed to have been active prior to the opening of the Atlantic and was rejuvenated at that time. No seismic activity is now associated with the zone, and it probably has been inactive at least since the Jurassic. The existence of the zone in the Appalachian Plateau is somewhat conjectural, based only on anomalous aeromagnetic patterns (Popenoe et al 1964; Beck and Mattick 1964) and a proximity to one of Wagner and Lytle's (1976) zones of structural discontinuity.

Other investigators have recently proposed the existence of similar geologic features beneath the Appalachian Plateau. Wagner (1976) hypothesizes "growth faults" based on the confined subsurface distribution of certain rock units of Cambrian and Lower Ordovician age. The faults were to have been active during the Cambrian and Ordovician periods. Root (1978) proposes similar down-to-the-east basement faults recurrently active during the Paleozoic and Mesozoic eras. Parrish and Lavin (1982) propose that kimberlite intrusions of Mississippian to mid-Jurassic age were emplaced at the intersection of basement faults along the edge of the Rome Trough with crossstructural lineaments. These cross-structural lineaments were identified from gravity and magnetic anomaly terminations, pronounced magnetic gradients, changes in gravity patterns, structural discontinuities and satellite imagery. The basement faults are believed to be Triassic to Jurassic in age. Harris (1978) proposes border faults to the Rome Trough in Kentucky and West Virginia as being active during early and middle Paleozoic time.

There appears to be no correlation between the Transylvania faults, Wagner or Root's growth faults, or those faults and lineaments proposed by Parrish and Lavin. All of the features described above are believed to be at least Mesozoic in age, show no history of seismic activity, and pose no threat to the safety of the BVPS-2 site.

Other investigators have recently proposed the existence of similar geologic features beneath the Appalachian Plateau. Wagner (1976) hypothesizes "growth faults" based on the confined subsurface distribution of certain rock units of Cambrian and Lower Ordovician The faults were to have been active during the cambrian and age. Ordovician periods. Root (1978) proposes similar down-to-the-east basement faults recurrently active during the Paleozoic and Mesozoic. Parrish and Lavin (1982) proposal that kimberlite instrusions of Mississippian to mid-Jurassic age were emplaced at the intersection of basement faults along the edge of the Rome Trough with cross structural lineaments. These cross structural lineaments were identified from gravity and magnetic anomaly terminations, pronounced magnetic gradients, changes in gravity patterns, structural discontinuities, and satellite imagery. The basement faults are believed to be Triassic to Jurassic in age. Harris (1978) proposes border faults to the Rome Trough in Kentucky and West Virginia as being active during early and middle Paleozoic time. There appears to be no correlation between the Transylvania fault, Wagneror Root's growth faults or those faults and lineaments proposed by Parrish and Lavin. All of the features described above are believed to be at least Mesozoic in age, show no history of seismic activity, and pose no threat to the safety of the BVPS-2 site.

Ver Steeg (1944) describes several minor faults in eastern Ohio. Between Wilkesville and Clarion in Vinton County, there is a northsouth striking fault with a throw of 7 feet; in Harrison County, a north-northeasterly striking reverse fault has a throw of 1.5 feet. He also describes a vertical fault with a throw of 15 feet dipping to the west at 45 to 50 degrees. The age of these minor faults is not known, but they may be related to Alleghenian folding. They occur 140 and 42 miles southwest of the site, respectively.

Janssens et al (1976) report overthickened Salina units and folding of the younger sediments in Guernsey County, Ohio, approximately 60 miles from the site. They postulate thrusting above the E salt unit of about 1 mile toward the northwest, with parallel folding of the preceding units. They also postulate a major tear fault in eastcentral Washington County, Ohio, similar to others found in the Appalachian basin associated with movement along salt beds during the Allegheny orogeny, approximately 95 miles from the site. (Janssen 1977, personal communication)

2.5.1.1.3.2 Structures of the Central Stable Region

<u>Folds</u>

The Cincinnati arch is the dominant basement structure in the Central Stable region of Ohio. This feature marks the change in dip from the easterly component of the Appalachian basin to the westerly component of the Michigan and Illinois basins. It served to divide the basins into separate depositional environments as well as to separate the mobile Appalachian basin from the stable intra-cratonic interior.

The Cincinnati arch is the northern extension of the Nashville-Jessamine dome, located in Tennessee and Kentucky. At its northern end, the arch bifurcates into the northwesterly trending Kankakee arch, and the north-easterly trending Findlay arch. The earliest development of the arch took place near the end of the Early Ordovician, with evidence of erosion of the Lower Ordovician and Upper Cambrian in northern Ohio. A complex sequence of uplift, stability, and subsidence relative to the basins followed, continuing into post-Permian to pre-Late Triassic time when the final uplift occurred (Janssens 1967, 1973). The arches represent Precambrian basement structural highs which have remained inactive since approximately the beginning of the Mesozoic.

The arch system continues into Canada and extends to the Canadian Shield as the Algonquin axis. A cross structure, the Chatham sag, separates the Findlay arch from the Algonquin axis. The feature in Canada is believed to have a similar, related history. The arch system is believed to have served as the anvil during lateral compression of the Paleozoic sediments from the southeast and east during the Allegheny orogeny (Rodgers 1970).

Faults

Ver Steeg (1944) describes a minor fault in Delaware County, Ohio, within the Central Stable Region. It strikes N2OE and dips 58 degrees to the east with a displacement of 8.5 feet. Jacoby (1969) reports two minor faults in the International Salt Mine in Cleveland. Both are nearly vertical gravity faults striking N7OW, and dipping to the north. The throws on the faults are 4 to 4.5 feet and 47 feet. The origin of the faulting is unknown.

Owens (1967) reports a small displacement, north-south trending, gravity fault, downthrown to the east on the Precambrian surface in Clinton-Fayette-Pickaway Counties, Ohio. The fault is based on an east-west seismic cross section. He also states the feature may be the result of erosion on the Precambrian surface.

Two other faults in Ohio were recognized during site investigations for Perry Nuclear Power Plant - Unit 1 and Unit 2, Lake County, Ohio, approximately 90 miles northwest of the site. The Warners Creek and Hell Hollow faults were found to be relatively small features, probably younger than 35,000 years and related to slumping of rock masses along joint planes or thrust faults which resulted from loading effects and ice movements during the Pleistocene (Cleveland Electric Illuminating Company 1974). Evidence for deep-seated faulting in the area was not present. The largest fault known in Ohio is the Bowling Green fault, approximately 180 miles from the BVPS-2 site. It is a high angle reverse fault and trends north-northwest across the top of the Cincinnati arch through Hancock, Wood, and Lucas Counties into Michigan. It displaces the Trenton oil horizon some 200 feet. The fault is believed to become a monocline at both ends (Ver Steeg 1944; Janssens 1973). The bedrock surface across the fault was leveled by erosion during the Mesozoic and Cenozoic eras, and it appears that no movement has occurred on the fault since about the end of the Paleozoic (Woodward-Moorhouse and Associates, Inc. 1974). The majority of structural features in northwestern Ohio lie in a northwest-southeast direction paralleling the trends in the Michigan basin, to which they are probably related.

Several faults have been identified in Ontario in the vicinity of the Chatham sag. They include the Electric fault, the Dawn fault, the Clearville fault, the Kimball-Colinville fault, and the Willey fault (Brigham 1972). These faults are 140 miles from the site at their nearest known location and are believed to be Ordovician to Devonian in age, with no evidence of post-Devonian activity.

2.5.1.1.3.3 Structures of the Valley and Ridge Province

Sedimentary strata of the Valley and Ridge have been deformed into a close succession of anticlines and synclines, each several miles across. Most folds are asymmetrical and have been overturned or steepened on the northwest limb. Progressive deformation has resulted in southeast dipping thrust faults developing along the overturned limb in places. Faulting is common in the Valley and Ridge of Virginia and Tennessee, while folding dominates in Pennsylvania. The deformation has been related to thin-skinned tectonics which took place during the Allegheny orogeny (Gwinn 1964, 1970; Rodgers 1963, 1970).

Some of the larger structures closest to the site are the Wills Mountain anticline, Broad Top syncline, and the Nitanny anticline, 110 to 125 miles southeast of the site. All of these structures have been shown to be the result of folding and thrusting along relatively shallow, weak stratigraphic units during the Allegheny orogeny. The Birmingham and Sinking Valley thrusts are exposed 115 miles southeast of the site, and are associated with development of the Nitanny anticlinorium. A major fault, the Little North Mountain thrust, is present 150 miles southeast of the site where it coincides with the mid-province structural front, the zone of vertical and overturned beds (Rodgers 1970). It is just one of several thrusts in the Valley and Ridge of Virginia and Maryland, which developed approximately in the same manner and at the same time. 2.5.1.1.3.4 Structures of the Blue Ridge and Piedmont Provinces

A small part of the Blue Ridge and Piedmont provinces lies within 200 miles of the BVPS-2 site. The history and geologic structures of the two provinces are similar and are discussed together.

The Blue Ridge and Piedmont provinces are characteristically composed of metamorphosed Precambrian and Lower Cambrian eugeosynclinal sediments, which have been intensely folded and faulted. Involvement during two deformations distinguishes these provinces from the Valley and Ridge to the west. Major features of the northern end of the provinces, within 200 miles of the site, are the South Mountain-Blue Ridge anticlinorium and the Catoctin Border fault. The former is the major anticlinal feature of the Blue Ridge, and exposes metamorphosed Precambrian basement rocks in its core with volcanic rocks unconformably overlying them. Plate tectonic theory and interpretation of recent seismic profiling indicate that the Blue Ridge and Piedmont rocks are allochthonous, having been thrust a minimum of 35 miles over Valley and Ridge sedimentary rocks. Deformation and westward transport is believed to have started during the Ordovician Taconic orogeny and culminated in the Pennsylvania to Permian, Allegheny orogeny. Orogenic compressive stresses ceased with the initiation of continental rifting during the Triassic and Jurassic, creating the present Atlantic Ocean basin (Cook 1983; Cook et al 1979; Harris and Bayer 1979; Cook, Brown, and Oliver 1980.)

The Catoctin Border fault forms the western boundary of the Piedmont province in Pennsylvania, Maryland, and northern Virginia. It is a normal fault, downthrown to the east, and borders the Gettysburg-Culpepper basin of Triassic age.

2.5.1.1.3.5 Lineaments Within the Site Region

Several studies have included the identification and analysis of lineaments in the area surrounding the BVPS-2 site (Gwinn 1964; Wagner and Lytle 1976; Kowalik and Gold 1976; Briggs and Kohl 1976; Saunders and Hicks 1976; Bench et al 1977; Colton 1977). Some studies were based on satellite imagery, while others used subsurface geologic information. Several lineaments were identified in the immediate site vicinity, but none are believed to correspond to bedrock fault traces. Most lineaments can be shown to correspond to topographic features and segments of rivers and streams. Many lineaments which represent valley traces are parallel or normal to fold axes. Joint patterns in this area also tend to be parallel or normal to these axes (Briggs and Kohl 1976). Although the relationship between joints and straight stream valleys is still in question, the valley lineaments do suggest some relation to Allegheny orogeny folding. Most, if not all, lineaments identified in Pennsylvania can be shown to have originated before the Cenozoic. The very low level of seismic activity in the area precludes development of any lineaments as a result of recent fault movement.

2.5.1.1.4 Regional Geologic History

The geologic history of the region about the site can best be understood by tracing the history of development of the central Appalachian basin. The Appalachian basin is an elongated sedimentary basin which extends from the Canadian Shield in southern Quebec and Ontario southwestward to central Alabama. It includes here the area of typical Valley and Ridge structures, as well as the rocks of the Appalachian Plateau. Most of the site region (200-mile radius) falls within this area, except for a small section of Blue Ridge and Piedmont rocks to the southeast.

The general configuration and character of the Precambrian basement beneath the basin is fairly well known from oil and gas exploration activities and gravity and aeromagnetic data. Rocks of the Canadian Shield are believed to continue as a peneplain surface beneath the Paleozoic cover, and slope at a very low angle to the south and southeast (Beck and Mattick 1964; King 1977; Kulander and Dean 1978). The depth to the basement is generally equal to the thickness of the overlying sediments and ranges from 2,000 feet over the Cincinnati arch to about 35,000 feet in the geosynclinal portion of the Appalachian basin. The basement surface reflects a long period of erosion which began in Late Precambrian time, and continued into the Middle Cambrian. It resulted in a marked angular unconformity between the metamorphic and igneous complex of the basement, and the sedimentary pile within the basin. Subsidence of the basement in the late Middle Cambrian and Late Cambrian initiated deposition of marine sediments, consisting mostly of sandstone and sandy dolomite, the basal clastic sequence. The sequence is wedge-shaped with the greatest thickness being along the eastern edge of the basin, indicating transgression from the southeast. This sequence was followed by an episode of carbonate deposition during the Late Cambrian and Early Ordovician during a prolonged period of crustal quiescence. Evidence suggests that the carbonate deposits ended abruptly southeastward at a continental shelf bordering the eugeosyncline (Rodgers 1968). This sequence ended after deposition of the Trenton Limestone.

Gentle epeirogenic uplift during the early Middle Ordovician resulted in an erosional disconformity in the upper part of the Cambrian-Ordovician carbonate section in some parts of the basin. Clastic sediments of Late Ordovician to Middle Silurian age conformably overlie the Trenton, although in many areas this episode began in the Middle Ordovician (King 1959). They were derived from erosion of emergent land to the southeast and southwest (Eardley 1962; King 1959). The first part of the later clastic deposits forms the Middle and Upper Ordovician Normanskill and Martinsburg Formations of New York and Pennsylvania, considered to be flysch deposits (King 1977). Evidence exists for a major deformation in Late Ordovician time, mostly from the northeastern part of the basin. The Taconic orogeny left its imprint in central and eastern Pennsylvania as a distinct angular unconformity between highly deformed Upper Ordovician rocks, only slightly deformed Lower Silurian rocks. A thick wedge of elastic sediments, centered in east-central Pennsylvania, began to accumulate in Late Ordovician time, and continued through the Silurian and Devonian, with the greatest development during the Silurian and Devonian. The source of these sediments is believed to have been an upland to the east which was developing as a result of continental plate convergence (King 1977).

Early Silurian clastics were followed by a thick carbonate and shale sequence during the Middle Silurian. Overlying these are the evaporites of the Salina Group and the Bass Islands carbonates. The top of the Bass Islands is a widespread Early Devonian unconformity, while the top of the overlying Helderberg Formation is a widespread regional unconformity. This unconformity was overlain by the transgressive sequence of the Oriskany and Bois Blanc Formations and was followed by carbonate deposition (Onondaga). This was followed by a thin sequence of clastics with intermittent carbonates up to the end of the Devonian, known as the Catskill Delta. Deposition of the Pocono Sandstone and Mauch Chunk Shale followed conformably on top of the marine Catskill deposits in some areas, but unconformably in most of the basin during the Early Mississippian. They indicate thickening toward the eastern geosynclinal trough which coincided with the Valley and Ridge province.

Another gentle uplift occurred in Late Mississippian or Early Pemsylvanian time in the northwest part of the basin and resulted in an erosional disconformity between the Mississippian and Pennsylvanian (Colton 1970). The Pennsylvanian strata are distinctly clastic, and include the great coal-bearing formations of the Appalachian Plateau and Valley and Ridge provinces, deposited in a restricted basin. The Pottsville, Allegheny, Conemaugh, and Monongahela Formations in the site vicinity were deposited at this time.

Conformably overlying the Monongahela in an oval area of southwest Pennsylvania, eastern Ohio, and West Virginia, is the Dunkard Group of Early Permian age. It is composed of shale, sandstone, and a few thin coal beds.

The age of deformation of this area is not clearly defined, but, presumably, it was later than the youngest rocks present, and before deposition of Late Triassic sediments in the Newark-Gettysburg basin, 165 miles east of the site. The Allegheny orogeny completely changed the character of the Appalachian basin from a predominantly depositional environment into an emergent mountain range and plateau.

The basin can be separated into two structural provinces, the Appalachian foldbelt on the east, and the Appalachian Plateau on the west. The boundary between the two closely coincides with the Allegheny Front. The foldbelt was subjected to intense deformation during the Allegheny orogeny which resulted in folding and faulting and generation of the Valley and Ridge mountains in central Pennsylvania and Virginia. Intensity of deformation decreases rapidly west of the Allegheny Front.

The Appalachian Plateau was also subject to the lateral compressive forces of this orogenic episode, and shows mild deformation within the higher stratigraphic units. The presence of Salina salt beds underlying a large portion of the area apparently had a major effect on controlling the deformation in this area. The salt beds greatly reduced the resistance to the lateral compressive stresses, and facilitated thin-skinned tectonic movements over a large area (Gwinn 1964; Rodgers 1963). In some parts of the Plateau, it can be shown that other weak sedimentary units may have acted in a similar manner (Rodgers 1970). These deposits were the major influence for lateral, northwestern thrusting of the orogenically-disturbed sequences, folding, and faulting of strata above the weak zones, and plastic flow and decollement deformations within the zones. Most of the folds are asymmetrical and steepest on the northwest flank. Thrust faults are the dominant structure in the southern Appalachians, but die out in southern Pennsylvania in a belt of anticlines and synclines.

The tectonic forces which resulted in the Allegheny orogeny are believed by many to be a continuation of earlier Paleozoic continental collision. The forces have been extinct since Late Paleozoic-Early Mesozoic time when, it is believed, the continental plates were rifted apart and generated the present Atlantic Ocean basin. Since that time, the Appalachian basin area has been subjected to moderate epeirogenic movements, which have provided the relief necessary to produce the geomorphologic dissection present today.

Outside the Appalachian basin, but within the site region in eastern Pennsylvania, is an area of Mesozoic deformation and sedimentation. The Newark-Gettysburg-Culpepper basin is a series of long, narrow, fault-bound basins of Late Triassic and Early Jurassic deposits, which developed as a result of continental rifting. The deposits are chiefly clastic, and predominantly red in color, being fanglomerates, conglomerates, sandstones, arkoses, and mudstones. Basalt flows, and diabase dikes and sills are voluminous within the deposits. The total thickness of the deposits is about 20,000 feet, believed to be derived from granitic and gneissic terrain to the southeast (Eardley 1962). The faulting that marked the beginning of the basin deposition indicates the beginning of the Palisades orogeny. It started in late Triassic time, and probably ended before the Early Jurassic (Rodgers 1970).

The final phase in the geologic history of the site region was that of the Pleistocene glaciations, which covered the entire northern half of the region. The tectonic results of the glaciations were down-warping of the area overlain by and adjacent to ice, followed by rebound after removal of the ice load. The low-level seismic activity that still occurs in the northeastern United States and eastern Canada is traditionally attributed to this rebound. Periglacial events, associated with the development of outwash terraces along the Ohio River, are as yet incompletely understood.

2.5.1.2 Site Geology

2.5.1.2.1 Site Physiography

The site is located on the south bank of the Ohio River in the town of Shippingport, 0.5 mile southeast of the town of Midland, Pennsylvania, and adjacent to Beaver Valley Power Station - Unit 1 (BVPS-1). It is situated near the center of the Appalachian Plateau physiographic province as outlined by Fenneman (1938). As previously described in Section 2.5.1.1.1, this province is characterized as an extensively dissected peneplain underlain by nearly flat-lying, undeformed Paleozoic sediments. The dissected topography has been somewhat subdued beneath several glacial drifts beginning about 20 miles north of the site and extending northward.

The site is situated on the uppermost Pleistocene outwash terrace of the Ohio River, which has an average elevation of approximately 735 feet in the main plant area (Figure 2.5.4-1). A younger terrace exists between the upper terrace and the present flood plain of the Ohio at an elevation of approximately 688 feet. The mean pool elevation of the river at the site is approximately 664 feet 6 inches. The upper terrace rises gently southward for a distance of approximately 1,500 feet before ending abruptly against a series of steep-sided, flat-topped hills with a top elevation of approximately 1,200 feet. The Ohio River is between 1,000 feet to 1,400 feet wide near the site, including the present flood plain.

Figure 2.5.4-50 is a top of rock contour map. As indicated on Figure 2.5.1-2, flat-lying sedimentary rocks of the Allegheny Group of Pennsylvanian age immediately underlie the 100-foot thick terrace. The Allegheny Group consists of a sequence of interbedded shales, sandstones, coal seams, underclays, and a limestone bed. It is estimated to be 350 feet thick in the site area, and contains one minable coal bed, the Upper Freeport coal, which outcrops above plant grade at approximately el 900 feet. No coal seam has been mined in the plant area at elevations below that of the plant, and no seam is considered to have commercial potential beneath the plant (Patterson 1963). The site is drained by a small northwesterly flowing stream, Peggs Run, which enters the Ohio River near the east end of the site. The stream was diverted in conjunction with construction of BVPS-1, and is now culverted or lined for its entire run through the site.

In the immediate area, there are no surface features indicative of actual or potential landsliding, or surface or subsurface subsidence, due to mining or cavernous conditions.

2.5.1.2.2 Site Stratigraphy

The area within 5 miles of the site is underlain by predominantly flat-lying, or gently-dipping sedimentary rocks, varying in age from the Middle Cambrian to the Late Pennsylvanian. Formation descriptions are from Gray (et al 1960), Wagner (et al 1975), and Fettke (1950).

2.5.1.2.2.1 Precambrian

Crystalline Precambrian rock is believed to unconformably lie beneath the thick Paleozoic sequence. Little is known about the rocks which comprise the basement complex as none are exposed within the area, nor have they been encountered in drill holes. From other areas, they have been found to be composed of various metamorphosed sedimentary and igneous materials, which have been intruded by various other igneous bodies. Beck and Mattick (1964) indicate that the basement may be between 10,000 and 11,000 feet deep in the site area, based on an aeromagnetic survey. The eroded basement surface is believed to dip southeasterly, averaging 85 ft/mile.

2.5.1.2.2.2 Cambrian

Early and Middle Cambrian stratigraphy in the site area is incompletely known from a few deep wells located in adjacent states. Stratigraphic relations indicate a sea transgressing from the southeast during this time which deposited a thick clastic wedge on the Precambrian surface. The westward extent of this wedge is not fully known, and it is somewhat speculative whether Lower and early Middle Cambrian rocks are represented beneath the site. Deposition may have begun in late Middle Cambrian with the Pleasant Hill and Warrior Formations, or even the Potsdam sandstone, but is definitely known to have occurred by the Late Cambrian with the Gatesburg Formation. The Gatesburg is known from a deep well in Butler County, Pennsylvania, to be a fine-grained, crystalline, light brownish gray dolomite, sandy dolomite, or dolomitic sandstone in excess of 350 feet thick (Fettke 1950).

2.5.1.2.2.3 Ordovician

Carbonate sedimentation continued in the Early and Middle Ordovician with deposition of Beekmantown Group rocks, composed of fine- to medium-grained, crystalline, light gray dolomite. The thickness of the Beekmantown varies in western Pennsylvania from 0 to 200 feet. A major unconformity occurs at the base of the Middle Ordovician section and is nearly basinwide in extent with the magnitude increasing to the northwest (Colton 1970). Approximately 4,500 feet of Lower Ordovician strata, present in central Pennsylvania, are missing in western Pennsylvania. It is not known to what extent rocks beneath the site were affected.

A sequence of predominantly noncalcareous clastic sedimentation began in the Middle Ordovician and continued into the Early Silurian. The Utica Formation, Reedsville Shale, Oswego Sandstone, and Queenston Shale are believed to exist beneath the site. The Utica is a black shale 100 to 300 feet thick while the Reedsville is a gray shale between 700 and 800 feet thick in western Pennsylvania. The Oswego Sandstone is a very-fine-grained, gray sandstone 0 to 60 feet thick, and the Queenston is a red shale, in part silty and sandy, and may be between 850 and 1,200 feet thick.

2.5.1.2.2.4 Silurian

Clastic deposition continued through the Early and Middle Silurian with deposition of the Tuscarora and Rose Hill Formations, the Keefer Sandstone, the Rochester Shale and the McKenzie Formation. The Tuscarora is a white-to-gray, fine-grained sandstone, conglomeratic in part, and may be 500 to 700 feet thick. The Rose Hill Formation is a reddish purple-to-greenish gray, fossiliferous shale with some hematitic sandstone lenses, and may be up to 875 feet thick. The overlying Rochester Shale is dark gray and calcareous and may be from 0 to 60 feet thick. Moderate gas and minor oil production has been realized from the Tuscarora Formation in northwestern Pennsylvania, Ohio, and West Virginia. The McKenzie Formation is a greenish gray shale interbedded with gray, fossiliferous limestone, and may be up to 330 feet thick.

Upper Silurian rocks were also predominantly clastic comprising the Bloomsburg, Wills Creek, and Tonoloway Formations. The Bloomsburg is a red interbedded shale and siltstone with lenses of sandstone and limestone. The unit varies in thickness from a few hundred to over 1,500 feet thick. The Wills Creek Formation is typically a greenish gray, fissile shale with local limestone and sandstone lenses. It may be up to 475 feet thick. The Tonoloway Formation is a gray, laminated, argillaceous limestone up to 575 feet thick. Deposition of the Keyser Formation marked the completion of the shift to carbonate sedimentation. The Keyser is a dark gray, fossiliferous, crystalline to nodular limestone and may be up to 300 feet thick.

2.5.1.2.2.5 Devonian

Carbonate deposition continued into Early Devonian time beginning with the Coeymanns Limestone, New Scotland Formation, and the Mandata Shale. The Coeymanns Limestone is a dark gray, crystalline limestone which may be sandy and shaly in places with some chert nodules. The thickness varies from 0 to 75 feet. The New Scotland Formation is a dark gray, cherty, fossiliferous limestone with some sandstone lenses and may be between 25 and 80 feet thick. The Mandata is a dark gray, calcareous shale between 20 and 150 feet thick. Overlying these are the Shriver Chert and the Ridgeley Sandstone. The Shriver is a dark gray, cherty limestone with some interbeds of shale and sandstone and may be up to 165 feet thick. The Ridgeley Sandstone (Oriskany) is a white to brown, partly calcareous, fossiliferous sandstone from 0 to 110 feet thick. There has been some production of gas in Beaver County from the Ridgeley Sandstone.

Middle Devonian rocks are predominantly clastic, and include the Needmore Shale, Selinsgrove Limestone, Marcellus Formation, and Mahantango Formation. The Needmore Shale is a greenish blue, thin bedded shale, and the Selinsgrove is a blue to black, medium bedded limestone. The Marcellus Formation is a black, carbonaceous shale, while the Mahantango is a brown to olive shale with interbedded sandstones and may be highly fossiliferous. The four units comprise the Hamilton Group, which varies from 140 to 2,000 feet thick in western Pennsylvania.

Late Middle and Upper Devonian rocks are represented by the Tully Limestone, Harrell Shale, Brallier, Chemung, Canadaway, Conneaut, Cattaraugus, and Oswayo Formations. The Tully is an argillaceous limestone 0 to 150 feet thick. The Harrell is a dark gray to black shale, while the Brallier Formation consists of interbeds of gray shale, siltstone, and sandstone. The Chemung is an irregularlybedded gray siltstone, sandstone, and shale, displaying abundant primary sedimentary features. The Canadaway consists of alternating gray shales and brown sandstones, and the Conneaut consists of alternating gray, brown, greenish, and purplish shales and siltstones. Red, gray, and brown shales and sandstones make up the Cattaraugus Formation, while the uppermost Devonian Oswayo Formation consists of greenish gray to gray shales, siltstones, and sandstones. The thickness of the Upper Devonian section, from the Harrell Shale through the Oswayo Formation, is between 3,000 and 6,000 feet from northwest to southeast Pennsylvania.

2.5.1.2.2.6 Mississippian

Pocono Group rocks of Early Mississippian age are predominantly gray, massive, cross-bedded sandstones and conglomerates with minor amounts of shale. They are between 570 and 900 feet thick in western Pennsylvania, and are important oil and gas reservoirs. The overlying sandy Loyalhanna Limestone is between 0 and 80 feet thick. The Upper Mississippian Mauch Chunk Formation consists of red shales with brownto-greenish gray, flaggy sandstones and is 0 to 100 feet thick.

2.5.1.2.2.7 Pennsylvanian

An erosional unconformity separates the Upper Mississippian and Lower Pennsylvanian systems in western Pennsylvania. The Pottsville Group probably rests on the Mauch Chunk in the site area. The Pottsville is typically a light gray-to-white, coarse-grained sandstone and conglomerate with minor shale beds. It is between 120 and 230 feet thick in the site area and contains some minable coal beds. The Middle Pennsylvanian, Allegheny Group, overlies the Pottsville, and is the subsurface bedrock at the site. It is also the dominant rock unit exposed in the site area. The Allegheny consists of cyclic sequences of sandstone, shale, limestone, and coal.

Differentiating the several coal beds has been accomplished by utilizing the distinctive Vanport Limestone. The type locality is approximately 6 miles northeast of the site, in the town of Vanport.

This limestone has been described as having abundant fossils, being very brittle, and breaking into irregular fractures. It is gray to blue in color, and interbedded with calcareous shale. Its total maximum thickness is 19 feet (Woolsey 1905). Above the Vanport Limestone is an interval which includes several coal beds, beginning with the Lower, Middle, and Upper Kittanning. Their respective thicknesses have been reported as 0 to 36 inches, 14 to 24 inches, and generally less then 6 inches (Patterson 1963).

The Lower and Upper Freeport Coal occur above the Upper Kittanning Coal, and have been or are currently mined within the site area. The thickness of the lower Freeport coal seam is 14 to 48 inches, while the upper seam thickness averages 36 inches. They are separated by approximately 45 feet of sandstone and shale.

The Lower Freeport coal is presently being open cut, auger mined, at Kelly Mine No. 1 in a 41-inch bed along Wolf Run in Industry. The

Upper Freeport has been both strip mined and underground mined along Peggs Run in a 48-inch average thickness bed. This coal seam also serves as the boundary between the Allegheny and Conemaugh Groups.

The Upper Pennsylvanian Conemaugh Group outcrops in the site area and continues the cyclic sedimentation sequence begun in the Allegheny. The Conemaugh Group has been divided into two mappable formations in the site area, the Glenshaw and the Casselman Formations (Wagner et al 1975). The Glenshaw contains cyclic sequences of red shales, sandstones, thin coal beds, and several thin marine limestones. The Ames Limestone, which forms the boundary between the two formations, is the most distinctive marker horizon within the Conemaugh Group. The weathered surface, where exposed, is covered with numerous projections of crinoid stems. It is a very persistent bed with an average thickness of 3 feet (Woolsey 1905). Where exposed, it is light brownish gray, being dark bluish gray on a fresh surface.

Between the Ames Limestone and the Morgantown Sandstone (the probable upper limit of rock types found in the site area) are approximately 40 feet of variegated shale or shaly limestone, and a thin coal seam (Woolsey 1905). The Casselman also contains cyclic sequences of red sandstones and shales with thin limestones and coal beds. The Glenshaw is 300 to 350 feet thick in the site vicinity, while the Casselman varies between 200 and 400 feet thick. Rocks younger than the Conemaugh do not outcrop within the site area, but are found 10 miles south of the site. They belong to the Upper Pennsylvanian Monongahela Group and continue the cyclic sedimentation sequences.

2.5.1.2.2.8 Pleistocene

Pleistocene deposits in the site area exist as terraces above the larger streams, and consist of unconsolidated sand and gravel deposits with varying amounts of clay and silt. Thicknesses of greater than 150 feet are known. The terrace on which the plant is situated averages 100 feet in thickness. It resulted from the ancestral Ohio River depositing enormous volumes of glacial outwash which was being carried away from the ice margins during the Late Pleistocene. The terrace at the site has not been correlated with any one of the seven known ice advances into Pennsylvania, but is probably the product of several of them.

Recent alluvial materials exist in the site area as floodplain deposits, primarily adjacent to the present Ohio River, but also mantling the intermediate terrace. The intermediate terrace is the result of flood control projects which lowered the river level during the 1930s.

2.5.1.2.3 Site Structure

The site area geologic investigation consisted of field-checking the existing published geologic literature within approximately 7 miles of the site. The original work, upon which all later information is based, was performed by L. H. Woolsey between 1902 and 1905, and was

published as the Beaver Folio. Coal, oil, and gas explorations since 1905 have only slightly modified Woolsey's original interpretations.

The field verification took place over a 2-week period in the fall of 1978 and relied primarily on recent road cuts, coal mining activities, and cliff exposures. The results of our investigation are presented subsequently, and are in complete agreement with the efforts of Woolsey (1905) and of Wagner (et al 1975) which indicate that the bedrock in the area is relatively flat-lying and undeformed. No offsets of stratigraphic marker beds were detected, based on exposures several miles apart, and no bedrock faults were identified within 5 miles of the site.

2.5.1.2.3.1 Structure as Determined from Coal Mining

The Peggs Run Coal Company Mine No. 2 is located approximately 8,500 feet south of the BVPS-2 site. The mine was operated in the Upper Freeport Coal seam, with an average thickness of 48 inches and ranged in elevation between 939 and 906 feet. The dip of the seam has been calculated from mine elevations to be less than 1 degree (33 /4,000 feet) to the northeast in one area, to less than 0.25 degree (13 /3,000 feet) to the southwest in another area. A strip mine operated by Peggs Run Coal Company, and located 2,600 feet southeast of the site, removed Upper Freeport Coal at an average elevation of 920 feet.

A similar example of the local structural dip variation (deviation from the regional southwest dip) has been extracted from Woolsey (1905). He had reported a 30-foot decrease in elevation of the Upper Freeport between the two adjacent towns of Vanport (el 938 feet) and Beaver (el 908 feet) located at least 10 miles to the northeast of the Peggs Run mine referenced previously. Based on both the absolute value and range of elevations at the two locations, the regional dip is found to be imperceptible over a 10-mile distance.

The Lower Kittanning Coal seam was also observed to maintain a nearly imperceptible dip from one side of the Ohio River to the other at Midland.

2.5.1.2.3.2 Structure as Determined from Limestone Horizons

The Vanport Limestone was verified at Merrill along Fourmile Run at el 720 feet, as reported by Woolsey (1905).

In addition, the Ames Limestone was found both north of the site in Midland, and south of the site near Hookstown. The two exposures of the Ames in Midland were noted as loose, detached blocks, at approximately el 1,200 feet, as indicated by Woolsey's Areal Geology map (1905).

Along U.S. Route 30, 1.4 miles west of the State Route 151 junction, the Ames Limestone outcrops at approximately el 1,200 feet, once again consistent in elevation and location, as reported by Woolsey.

These two limestone marker beds nearly bound the exposure of the Allegheny and Conemaugh Groups in the site area.

2.5.1.2.3.3 Jointing and Bedding

Four joint sets were identified from outcrops within the 5-mile radius of the site; two sets strike roughly northeast and two strike northwest; all are near vertical. This is in fair agreement with the results of Bench (et al 1977) in this area, which indicated the joints are the result of tectonic stresses, and subsequent stress adjustments produced during the erosion and unloading that affected the Plateau. The two dominant joint systems, which strike N76W and N57W, are believed to be the bedrock expression of the principal stress trends, and possibly relate to a structural weakness which resulted from the Allegheny folding and thrusting.

2.5.1.2.3.4 Faulting

No bedrock faulting was identified within 5 miles of the BVPS-2 site. A fault was identified, however, about 0.25 mile outside the radius at the Stewart Hill road cut along U.S. Route 30 in West Virginia. The fault strikes N35E and dips 17 degrees northwest and appears to be a gravity fault. The amount of displacement is indeterminable, as an 18-inch marker coal seam (Elk Lick coal) does not outcrop west of the fault, but presumably occurs beneath the ground surface.

The time of last movement has been determined stratigraphically by an overlying, horizontal sandstone bed, which appears to be continuous across the fault plane at the top of the exposure. Slumping of sediments along the fault plane during deposition resulted in upturned beds being overlain by the horizontal sandstone, as seen at the west end of the exposure. A portion of the fault is shown on Figure 2.5.1-8.

A similar example of deformation during sedimentation is cited by Wagner (et al 1970) and occurs in a railroad cut in the Casselman Formation near McKeesport. Here, a faulted sandstone is overlain by a claystone which is not offset.

The shallow dip angle of the fault in West Virginia is indicative of near-surface failure, probably as a submarine slump of semiconsolidated material. The upper surface of the slumped block was then eroded, leveled, and subsequently overlain by the thick sand layer during the Pennsylvanian.

2.5.1.2.4 Site History

The geologic history of the site area is similar to the history of the Appalachian basin discussed in Section 2.5.1.1.4 and is briefly summarized here.

An extended period of erosion on the Precambrian basement complex, which began during the Late Precambrian, ended during the Middle or Late Cambrian when a sea, transgressing from the southeast, deposited

a basal clastic sequence in the site area. This was followed throughout most of the Paleozoic Era by an alternating sequence of carbonate and clastic sedimentation, punctuated by three orogenic events: 1) the Taconic during the Middle Ordovician, 2) the Acadian during the Middle Devonian, and 3) the Alleghenian during the Permian. The latter had the most significant effect in the site area by producing a series of very gentle folds within the Paleozoic strata, and ended the sedimentation cycle. The site lies on the west limb of a troughlike basin, known as the Pittsburgh-Huntington basin. Dips of strata into the basin are gentle, usually less than 3 degrees, and are nearly imperceptible in the site area. Obviously, diastrophic deformation has not played a major role in the history of the site area. Tectonic forces have been inactive in the site area probably since the end of the Allegheny orogeny, 250 million years ago. Periodic epeirogenic uplifts, isostatic adjustments, and erosion since the Paleozoic have produced the well-dissected plateau present in the site area today.

2.5.1.2.5 Plot Plan

Location plans of borings performed at the site are shown on Figures 2.5.4-10, 2.5.4-13 and 2.5.4-15. Plant structures are superimposed upon these figures.

2.5.1.2.6 Geologic Profiles of Plant Foundations

Geologic profiles are presented in Section 2.5.4.

2.5.1.2.7 Extent of Backfill and Excavation

Excavation end backfill at the site are discussed in Section 2.5.4.5.

2.5.1.2.8 Engineering Geology Evaluation

2.5.1.2.8.1 Dynamic Behavior During Prior Earthquakes

Site investigations show no features or conditions indicative of disturbance during prior earthquakes, such as flow structures, fissures, or slumps in the unconsolidated deposits.

2.5.1.2.8.2 Description and Evaluation of Deformational and Weathered Zones

No zones of severe weathering, structural deformation, or lithologic weakness were identified in the site area based on core borings, seismic velocity measurements, and site area geologic reconnaissance. Seismic refraction studies indicate that hard, intact rock with compressional wave velocities of 12,000 feet per second (fps) underlies the site. No low velocity or anomalous zones were indicated.

The underlying rock is slightly weathered for the first few feet, with weathering effects decreasing rapidly with depth. All

structures are founded on select granular fill or natural soil deposits, with bedrock lying at least 55 feet deep.

Pomeroy (1979) mapped recent and older landslides and identified areas most susceptible to sliding within Beaver County, Pennsylvania. It was noted that most landslides that had been observed occurred in colluvial soils and weathered rock derived from mudstone, claystone, shale, and siltstone. It was further noted that most recent landslides had been generated by construction activities. The site topography is shown on Figure 2.5.4-13. While not specifically identified as an area of potential instability, the steep slopes to the south of the BVPS plant site are colluvial in nature. They are, however, sufficiently removed from the main plant area to present no potential safety problem in the event of a landslide. The stability analysis of the colluvial slopes to the south of the emergency outfall structure (EOS), located at the far western end of the site, is described in a summary report submitted separately (SWEC 1983). It was concluded that although there may be a potential for movement of the upper portion of the colluvial slope above el 780 ft, any slope movements would not affect the EOS.

2.5.1.2.8.3 Unrelieved Residual Stresses in Rock

Unrelieved residual stresses in rock were considered to have no influence on the design and operation of the plant due to the thickness of founding overburden.

2.5.1.2.8.4 Evaluation and Description of Natural Soils

The characteristics of the in situ surficial materials are described in Section 2.5.4.2 and the Soil Densification Program Report (DLC 1976).

2.5.1.2.8.5 Description of Man's Activities at the Site

Some oil and natural gas have been recovered within 5 miles of the site, mostly from the Pocono Group of Mississippian age. No wells have been drilled within the site boundary, nor are any anticipated. Extraction of natural gas or oil is not likely to produce consolidation and subsequent surface subsidence of the well lithified rocks beneath the site. The rocks in the site area are Permian or older and have not been susceptible to consolidation upon withdrawal of fluids from them.

Coal has been recovered at several locations within 5 miles of the site by underground and surface mining methods, mostly from the Upper Freeport coal seam.

No coal mining has taken place beneath the site nor is any anticipated. The limited quantity, low quality, and depth below the surface of the underlying coal seams precludes development during the expected lifetime of the plant. Maps indicating the areas underlain by coal deposits, and oil and gas, are presented on Figures 2.5.1-9 and 2.5.1-10, respectively.

Withdrawal of groundwater in the site area is discussed in Section 2.4.13.

2.5.1.2.9 Site Ground-water Conditions

The ground-water conditions in the site area are discussed in detail in Sections 2.4.13 and 2.5.4.6.

2.5.1.3 References for Section 2.5.1

Beck, M.E. and Mattick, R.E. 1964. Interpretation of an Aeromagnetic Survey in Western Pennsylvania and Parts of Eastern Ohio, Northern West Virginia, and Western Maryland. Pennsylvania Topographical and Geological Survey, Information Circular 52.

Bench, B.M.; Diamond, W.P.; and McCullough, C.M. 1977. Methods of Determining the Orientations of Bedrock Fracture Systems in Southwestern Pennsylvania and Northern West Virigina. U.S. Bureau of Mines, Report of Investigation No. 8217.

Brent, R.A. and Delong, R.M. 1960. Coal Resources of Ohio. Ohio Geological Survey Bulletin 58.

Briggs, R.P. and Kohl, W.R. 1976. Map showing Major Fold Axes, Satellite Imagery Lineaments, Elongate Aeroradioactivity Anomalies, and Lines of Structural Discontinuity in Southwestern Pennsylvania and Vicinity. U.S. Geological Survey Miscellaneous Field Studies, Map MF-815.

Brigham, R.J. 1972. Structural Geology of Southwestern Ontario and Southern Michigan, Ontario Department of Mines and Northern Affairs, Petroleum Resources Section, Paper 71-2.

Cardwell, D.H.; Erwin, R.B.; and Woodward, H.P. 1968. Geologic map of West Virginia. West Virginia Geological and Economic Survey, Scale 1:250,000.

Cleveland Electric Illuminating Co. 1974. Perry Nuclear Power - Units 1 and 2, Preliminary Safety Analysis Report. Docket Nos. 50-440, 50-441.

Colton, G.W. 1970. The Appalachian Basin - Its Depositional Sequences and their Geological Relationships. In: Studies in Appalachian Geology - Central and Southern. Editors: Fisher, G.W.; Pettijohn, F.J.; Reed, J.C., Jr.; Weaver, K.N. John Wiley and Sons, New York, N.Y.

Colton, G.W. 1977. Gasfield in Devonian Black Shale and Landsat Lineaments in parts of Ohio, Kentucky, West Virginia, and Pennsylvania, U.S. Geological Survey, Open-File Report 77-864.

Cook, Frederick A.; Albaugh, Dennis S.; Brown, Larry D.; Kaufman, Sidney; Oliver, Jack E.; and Hatcher, Jr., Robert D. 1979. Thin-Skinned Tectonics in the Crystalline Southern Appalachians; COLORP Seismic-Reflection Profiling of the Blue Ridge and Piedmont, <u>Geology</u>, Vol. 7, pp 563-567.

Cook, Frederick A.; Brown, Larry D.; and Oliver, Jack E. 1980. The Southern Appalachians and the Growth of Continents, <u>Scientific</u> <u>American</u>, Vol. 243, pp 156-168.

Cook, Frederick A. 1983. Some Consequences of Palinspastic Reconstruction in the Southern Appalachians, <u>Geology</u>, Vol. II, pp 86-89.

Duquesne Light Company. Report on the Soil Densification Program-Beaver Valley Power Station - Unit 2. Prepared by Stone & Webster Engineering Corporation, Boston, Mass.

Eardley, A.J. 1962. Structural Geology of North America. Second Edition, Harper and Row, New York, N.Y.

Fenneman, N.M. 1938. Physiography of the Eastern United States. McGraw Hill Book Co., N.Y.

Fenneman, N.M. 1946. Physical Division of the United States: in cooperation with Physiographic Committee of the U.S. Geological Survey.

Fettke, C.R. 1950. Summarized Record of Deep Wells in Pennsylvania. Bulletin of Pennsylvania Topographical and Geological Survey No. M31. Harrisburg, Pa.

Gray, C.; Shepps, V.C.; Conlin, R.R.; Hoskins, D.M.; Shaffner, M.N.; Socolow, A.A.; McLauglin, D.B.; Geyer, A.R.; Cate, A.S.; Lytle, W.S.; Bergsten, J.M.; Arndt, H.H.; Kehn, T.M.; Van Olden, A.E. 1960. Geologic Map of Pennsylvania. Bulletin of the Pennsylvania Topographical and Geological Survey Scale 1:250,000 Harrisburg, Pa.

Gwinn, V.E. 1964. Thin-Skinned Tectonics in the Plateau and Northwestern Valley and Ridge Provinces of the Central Appalachians. Bulletin of the Geological Society of America, Vol. 55.

Gwinn, V.E. 1970. Kinematic Patterns and Estimates of Lateral Shortening, Valley and Ridge and Great Valley Provinces, Central Appalachians, South Central Pennsylvania. In: Studies in Appalachian Geology - Central and Southern. Editors: Fisher, G.W.; Pettijohn, F.J.; Reed, J.C., Jr.; and Weaver, K.N. John Wiley and Sons, New York, N.Y.

Harris, Leonard D. 1978. The Eastern Interior Avlacogen and its Relation to Devonian Shale-Gas Production, Second Eastern Gas Shales Symposium, Vol. 2. pp 55-70.

Harris, Leonard D., and Bayer, Kenneth C. 1979. Sequential Development of the Appalachian Orogen Above a Master Decollement - A Hypothesis, <u>Geology</u>, Vol. 7, pp 568-572.

2.5.1-27

Jacoby, C.H. 1969. Correlation, Faulting, and Metamorphism of the Michigan and Appalachian Salt Basin. Bulletin of the American Association of Petroleum Geologists, Vol. 53.

Janssens, A. 1967. Analysis of the Paleozoic Movements of the Cincinnati Arch. Unpublished PhD Thesis, Ohio State University, Columbus, Ohio.

Janssens, A. 1973. Stratigraphy of the Cambrian and Lower Ordovician Rocks in Ohio. Ohio Department of Natural Resources, Division of Geological Survey, Bulletin 64.

Janssens, A; Deyling, T.H.; and Ott, R.R. 1976. Relation between Salina (Upper Silurian) Oil Production and Shallow Structure in East-Central Ohio. Abstract, Bulletin of the American Association of Petroleum Geologists, Vol. 60.

Johnson, D. 1931. Stream Sculpture on the Atlantic Slope: A Study in the Evolution of Appalachian Rivers. Columbia University Press.

King, P.B. 1959. The Tectonics of North America - A Discussion to Accompany the Tectonic Map of North America. U.S. Geological Survey Professional Paper No. 628.

King, P.B. 1977. The Evolution of North America. Revised Edition, Princeton University Press, Princeton, N.J.

Kowalik, W.S. and Gold, D.P. 1976. The Use of Landsat-1 Imagery in Mapping Lineaments in Pennsylvania. In: Proceedings of First International Conference on the New Basement Tectonics. Editors: Bodgson, R.A.; Gay, S.P.; and Benjamin, J.Y. Utah Geological Association, Publication 5.

Kulander, B.R. and Dean, S.L. 1978. Gravity, Magnetics, and Structure: Allegheny Plateau/Western Valley and Ridge in West Virginia and Adjacent States, West Virginia Geological and Economic Survey, Report of Investigation No. RI-27.

Lamborn, R.E. 1951. Limestones of Eastern Ohio, Ohio Department of Natural Resources, Bulletin of the Division of Geological Survey, No. 49.

New York Department of Environmental Conservation 1977. New York State Oil and Gas Fields. Bureau of Mineral Resources.

Ohio Department of Natural Resources 1975. Oil and Gas Fields of Ohio. State of Ohio, Division of Geological Survey.

Ownes, G. L. 1967. The Precambrian Surface of Ohio. Ohio Department of Natural Resources, Division of Geological Survey, Report of Investigation No. 64. Parish, Jay B., and Lavin, Peter M. 1982. Tectonic Model for Kimberlite Emplacement in the Appalachian Plateau of Pennsylvania, <u>Geology</u>, Vol. 10, pp 344-347.

Patterson, E.D. 1963. Coal Resources of Beaver County, Pennsylvania. Bulletin of U. S. Geological Survey No. 1143-A.

Pennsylvania Department of Environmental Resources undated. Distribution of Pennsylvania Coals. Topographic and Geologic Survey, Map 11, Commonwealth of Pennsylvania.

Pennsylvania Department of Environmental Resources undated. Oil and Gas Fields Map of Pennsylvania. Topographic and Geologic Survey, Map 10, Commonwealth of Pennsylvania.

Pennsylvania Topographic and Geographic Survey 1975. Greater Pittsburgh Regional Geologic Map. Map #42, plate 1.

Pomeroy, F.W.; Mowak, T. A.; and Fakundiny, R. H. Clarendon-Linden Fault System of Western New York - A Vibroseis Seismic Study. In press.

Pomeroy, J.S. 1979. Map Showing Landslides and Areas Most Susceptible to Sliding in Beaver County, Pennsylvania. U.S. Geological Survey Map 1-1160.

Popenoe, P.; Petty, A.J.; and Tyson, N.S. 1964. Aeromagnetic Map of Western Pennsylvania and Parts of Eastern Ohio, Northwest Virginia, and Western Maryland. U.S. Geological Survey Map GP-445.

Rodgers, J. 1963. Mechanics of Appalachian Foreland Folding in Pennsylvania and West Virginia. Bulletin of American Association of Petroleum Geologists, Vol. 47.

Rodgers, J. 1968. The Eastern Edge of the North American Continent During the Cambrian and Early Ordovician. In: Studies of Appalachian Geology: Northern and Maritime. Editors: Zen, E-an; White, W.S.; Hadley, J.B; and Thompson, J.B., Jr. John Wiley and Sons, New York, N.Y.

Rodgers, J. 1970. The Tectonics of the Appalachians. Wiley Interscience, New York, N.Y.

Root, S.I. and Hoskins, D.M. 1977. Latitude 40°N Fault Zone, Pennsylvania: A New Interpretation. Geology, Vol. 5.

Root, S.I. 1978. Possible Recurrent Basement Faulting, Pennsylvania Part 1, Geologic Framework, Pennsylvania Geol. Surv. Open-File Rpt.

Saunders, D.F. and Hicks, D.E. 1976. Regional Geomorphic Lineaments on Satellite Imagery - Their Origin and Applications. Proceedings of Second International Conference on New Basement Tectonics, Newark, Del. Sbar, M.L. and Sykes, L.R. 1977. Seismicity and Lithospheric Stress in New York and Adjacent Areas. Journal Geophysical Research, Vol. 82.

Sheppa, V.C.; White, G.W.; Droste, J.B.; and Sitter, R.F. 1959. Glacial Geology of Northwestern Pennsylvania. Bulletin of Pennsylvania Topography and Geology Survey, No. G-32.

Stone & Webster Engineering Corp. (SWEC) 1983. Stability of Slopes at the Emergency Outfall Structure. Beaver Valley Power Station -Unit 2, Shippingport, Pennsylvania.

Stone & Webster Engineering Corp. 1978. Regional Geology of the Salina Basin, ONWI/SUB-E512-00600/1, Vol. 1.

United States Geological Survey 1974. Geological Map of the United States, 1:2,500,000

Van Tyne, A.M. 1976. Subsurface Investigation of the Clarendon-Linden Structure, Western New York. Empire State Geogram, Vol. 12, p. 13.

Ver Steeg, K. 1944. Some Structural Features of Ohio. Journal of Geology, Vol. 52.

Wagner, W.R.; Heyman, L; Gray, R. E.; Belz, D. J.; Lund, R; Cate, A.J.; and Edgerton, C.D. 1970. Geology of the Pittsburgh Area. Bulletin of the Pennsylvania Topography and Geology Survey, General Geology Report No. G-59.

Wagner, N.R.; Craft, J.L.; Heyman, L.; and Harper, J.A. 1975. Greater Pittsburgh Region Geologic Map and Cross-Sections. Bulletin of the Pennsylvania Topographical and Geological Survey, Map 42, Harrisburg, Pa.

Wagner, W.R. 1976. Growth Faults in Cambrian and Lower Ordovician Rocks of Western Pennsylvania, AAPG Bull, Vol. 60, No. 3, pp 414-427.

Wagner, N.R. and Lytle W.S. 1976. Greater Pittsburgh Region Revised Surface Structure and Its Relation to Oil and Gas Fields. Pennsylvania Topography and Geology Survey. Information Circular No. 80.

West Virginia Geological and Economic Survey undated. Generalized Geologic Map of the Coal Fields of West Virginia.

Woodward Moorhouse and Associates, Inc. 1974. Geology Seismology, Subsurface Conditions, and Geotechnical Design Criteria. Appendix 2C, Davis-Besse Nuclear Power Station - Units 2 and 3, PSAR, Docket Nos. 50-500 and 50-501.

Woolsey, L.H. 1905. Description of the Beaver Quadrangle, U.S. Geological Survey, Folio 134.



FIGURE 2.5.1-1 REGIONAL PHYSIOGRAPHIC PROVINCES BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





 \mathcal{X}





FIGURE 2.5.1-2 SITE AREA GEOLOGIC MAP BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT

CONTOUR INTERVAL 20 FEET







.

NOTE:

.

Modified from: Johnson D.,1931.

FIGURE 2.5.1-4 GENERALIZED GEOLOGIC CROSS-SECTION ACROSS PENNSYLVANIA AND EASTERN OHIO BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT




LEGEND:

• II - III

• IV-V

N N

VII OR GREATER AS NOTED

NO INTENSITY DATA

SHALLOW EARTHQUAKE

(1885) DATE

0	25	50	75	100				
SCALE-MILES								

NOTE:

1

INTENSITIES ARE MODIFIED MERCALLI (MM) SEE TABLE 2.5.2-1

FIGURE 2.5.1-5 EPICENTERS & TECTONIC PROVINCES WITHIN 200 MILES OF THE SITE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



TONOLOWAY FM BASS IS. WILLS CREEK FM. UPPER SALINA GP. BLOOMSBURG FM. WILLIAMSPORT SS SILURIAN LOCKPORT DOL. MCKENZIE FM CLINTON ROCHESTER SH ROCHESTER SH. GP.10 MIDDLE **KEEFER SS.** DAYTON LS. ROSE HILL FM ដ LOWER MEDINA TUSCARORA FM. QUEENSTON FM. JUNIATA FM. RICHMOND GP. MAYSVILLE GP. EDEN GP. UPPER OSWEGO BALD EAGLE FM SS. **REEDSVILLE SH.** REEDS-MARTINSBURG VILLE (MARTINS-UTICA SH. BURG SHALE SH. **COBURN-SALONA** TRENTON TRENTON ORANDA FMS. ORANDA FM ORDOVICIAN NEALMONT LS. NEALMONT CHAMBERS-BURG MERCERSBURG FM MIDDLE BENNER-SNYDER BLACK RIVER BLACK RIVER HATTER SHIPPENSBURG FM, WELLS CREEK LINCOLNSHIRE CHAZY NEW MARKETLS LOYSBURG FM. ST. PAUL GP. ROW PARK LS. PINESBURG STA. ₩0 BELLEFONTE FM. LAMBS CHAPEL BEEKMANT(GP. NITTANY FM. ROCKDALE RUN FM. LOWER LARKE FM. STONEHENGE LS. KNOX MINES FM. KERBEL GATESBURG FM. UPPER CONOCOCHEAGUE FM. CONASAUGA WARRIOR FM. ROME PLEASANT HILL FM. ELBROOK FM. MIDDLE MT. SIMON CAMBRIAN WAYNESBORO FM. TOMSTOWN DOL LOWER ANTIETAM FM HARPERS FM. WEVERTON-LOUDOUN FMS. 1? CATOCTIN FM. SWIFT RUN FM. PRECAMBRIAN MCRYSTALLINE MROCKS PRECAMBRIAN 117 ミハノー

FIGURE 2.5.1-6 REGIONAL STRATIGRAPHIC CORRELATION CHART BEAVER VALLEY POWER STATION-UNIT: FINAL SAFETY ANALYSIS REPORT

N



FIGURE 2.5.1-7 GEOLOGIC STRUCTURES AND TECTONIC PROVINCES WITHIN 200 MILES OF THE SITE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



EGEND
THRUST OR REVERSE FAULT
NORMAL FAULT
ANTICLINAL AXIS WITH PLUNGE
SYNCLINAL AXIS
LINE OF STRUCTURAL DISCONTINUITY







BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



FIGURE 2.5.1-10 OIL AND GAS FIELDS OF THE SITE REGION BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT







42°



2.5.2 Vibratory Ground Motion

The site region is characterized by a low level of earthquake activity. The Ohio River Valley in which the site is located has been well settled since the early 1800's. During the past 180 years, there have been few earthquakes within 200 miles of the site and only three within 50 miles. There are two centers of activity within 200 miles of the site - one near Attica, New York, and one near Anna, Ohio. Moderate size earthquakes have occurred in these centers of activity. The maximum earthquake potential at the site is an earthquake of Modified Mercalli (MM), Intensity VI, occurring near the site, corresponding to a peak horizontal ground acceleration of 0.07g. The safe shutdown earthquake (SSE) has been specified as 0.125g and the operating basis earthquake (OBE) has been specified as 0.06g.

2.5.2.1 Seismicity

Most of the information concerning earthquake activity in the eastern United States is based on historical reports, old diaries, and newspaper accounts. These earthquakes are classified on the basis of intensity corresponding to the Modified Mercalli (MM) scale. This scale, shown in Table 2.5.2-1, is based on observations of effects of earthquakes and damage to structures. The instrumental monitoring of earthquakes in the eastern United States began in the mid 1920's. Since that time, the number of seismograph stations has greatly increased. Historical reports of earthquakes and information obtained from instrumental coverage in recent years form the basis of the examination of the seismicity of the site region.

2.5.2.1.1 Totality and Reliability of Earthquake Catalog

Even though major historical catalogs carry entries dating back almost to the 1800's, the coverage of this long a period is not homogeneous. The completeness and reliability of the data are related to population distribution and to the seismograph network coverage. Therefore, the accuracy of epicentral coordinates and the assigned maximum intensities have to be evaluated carefully.

For the earlier historical events, epicenters near dense settlements are probably located incorrectly due to the absence of felt reports from the true epicentral area. The intensity of an earthquake at a given location depends not only on accurate and complete human observations but also on foundation conditions, structure, design, type, and quality of construction. Construction practices, particularly of chimneys in the earlier centuries, were certainly not those envisaged in the MM scale. Interpretation of historical damage reports, without consideration of construction practices, may result in overestimated intensities. Furthermore, the tendency of early settlers to build structures near rivers, where soil conditions often amplify the ground motion, resulted in a biased sampling of earthquake damage and overestimated intensities. Seismological information for the instrumental period (post-1900) must also be evaluated carefully. Seismic instrumentation began in the early 1900's in the United States and Canada and progressively improved the quality of earthquake data. Epicentral locations based on felt reports were complemented and somewhat controlled by instrumental data. From the 1900's until the 1960's, only a few seismographs operated in the eastern United States. Most of these stations were part of the regional network operated by the Jesuit Seismological Association. In the early decades, numerous factors were potential sources of errors such as the type of instrumental response, lack of accurate time control, awkward configuration, use of graphical methods and limited knowledge of crustal velocities. These produced large uncertainties in the epicentral coordinates which in many cases, amounted to tens of kilometers. Since the 1960's, increased interest in understanding local seismicity has resulted in the installation of dense seismographic networks.

2.5.2.1.2 Earthquake History

A chronological list of all earthquakes known to have occurred within 200 miles of the site is provided in Table 2.5.2-2. The basic information for this seismic data base was taken largely from the earthquake catalog developed by Barstow et al (1981) which lists 4289 events which have occurred in the eastern and central United States and portions of Canada between 1534 and April 1978. Several additional events not noted in Barstow et al (1981) were taken from the data bases prepared by Weston Geophysical (1972), Pomeroy and Faukundiny (1976), and Stover et al (1981).

To account for any additional events which may have occurred more recently then 1978, a search was made of the Regional Seismic Network Bulletins prepared by St. Louis University between January 1978 and March 1983 and the Northeastern United States Seismic Network Bulletins for January 1978 through September 1981 (Chiburis et al). Use was also made of a computerized earthquake catalog developed by the National Oceanic and Atmospheric Administration (NOAA) which lists worldwide events for the period between 1900 and the end of 1979 (SWEC 1980).

Figure 2.5.2-1 shows the location of the earthquakes listed in Table 2.5.2-2; Figures 2.5.2-2 and 2.5.1-5 show the location of the earthquakes in relation to geologic structures and tectonic boundaries, respectively. These figures show that BVPS-2, situated within the Appalachian Plateau Tectonic Province, is located in an almost aseismic area.

The cumulative historical seismicity data (Figure 2.5.2-1) reveal the presence of two areas of concentrated seismic activity. They are Attica, New York and Anna, Ohio. These will be addressed in this section in terms of their location, areal extent, and level of historical seismicity. The tectonic frame work of these sources as inferred from current research will be discussed in Section 2.5.2.3.

Activity in Attica, New York

The Attica area has been the site of a significant amount of historical seismic activity and includes the August 12, 1929 event discussed in Section 2.5.2.1.3. Recent low level seismic activity has been correlated with high-pressure

fluid injection operations in brine fields which are developed in the area (Fletcher and Sykes 1977).

Activity in Anna, Ohio Area

A localized concentration of seismic activity exists in Shelby County, Ohio near the town of Anna. Several moderately damaging earthquakes have occurred in the area. These include the earthquakes of June 18, 1875, Modified Mercalli Intensity (VII), September 19, 1884 (VI), September 30, 1930 (VII), March 2, 1937 (VII), and March 9, 1937 (VII-VIII) (Bradley and Bennett 1965). Understanding of the seismicity and tectonics of the Anna, Ohio region will be improved by the data being gathered by the microearthquake network recently installed in the area by the University of Michigan.

The cumulative historical seismicity data, carefully interpreted, can yield valuable information on the spatial and temporal distribution of larger and more significant earthquakes and the location of zones of concentrated activity. In the northeastern United States, several years of operation of the seismographic network have produced a complete record of accurately located events of magnitude 1.8 through 2.0 and larger in the region. Sbar and Sykes (1977) have noted that the spatial distribution of instrumental seismicity closely tracks the distribution of less accurately located historical events, thus reinforcing confidence that older events are fairly well located and that areas of seismic activity are stationary. Although the midwestern United States is not as densely monitored as the northeastern region, we can assume that a similar analogy exists. The site region can then be assumed to exhibit in reality very low levels of seismicity except in the Attica, New York and Anna, Ohio areas.

The low level of seismicity in the site region is also evident in several other comprehensive studies of earthquake hazard in the eastern United States. A portion of the seismic frequency map prepared by Hadley and Devine (1974) is shown in Figure 2.5.2-10. The contours were drawn to differentiate the areal distribution of earthquake epicenters for earthquakes having epicentral intensities \geq II (MM), on the basis of the total number of earthquakes per 10,000 km² during the time period between 1800 and 1972. Hadley and Devine (1974) point out that the contours are considerably generalized and are drawn only as a guide for estimating regional seismicity. Figure 2.5.2-10 shows that BVPS-2 is situated in an area that has experienced less than four earthquakes per 10,000 km² during the historical period studied.

A study similar to that of Hadley and Devine (1974) was conducted by Barstow et. al. (1981). They developed an earthquake catalogue for the eastern and central United States covering the period between 1800 and 1977. The beginning date, 1800, was chosen since a more uniform demographic coverage of the study area was achieved. An epicenter map was then computer-plotted which showed the location of all earthquakes with a Modified Mercalli Intensity \geq III or with a magnitude \geq 2.0. A uniform, rectilinear coordinate system with grid points 85 km apart was superimposed on the epicenter map and fixed at 96°W longitude and 39°N latitude. A computer program was used to count and plot the number of earthquake epicenters within a radius of 61 km (11,689 km²) from each grid point. A seismic frequency contour map was then hand drawn, a portion of which is shown on Figure 2.5.2-11. It shows that BVPS-2 is located in an area with an earthquake frequency less than four per 11,689 km² during the historic period used for the study.

Although the contours are drawn somewhat differently in Figures 2.5.2-10 and 2.5.2-11, they do illustrate the extremely low level of seismicity in the vicinity of the site and that BVPS-2 is located in one of the least seismic areas in the eastern United States.

2.5.2.1.3 Earthquakes Felt at the Site

In order to determine earthquake hazard to BVPS-2, it is necessary to examine how severely the site has been affected by large earthquakes in the past. This examination is based on available historical records. A discussion of these earthquakes follows:

<u>New Madrid, Missouri Earthquakes, 1811 and 1812</u>

The New Madrid, Missouri earthquakes of December 16, 1811, January 23, 1812, and February 7, 1812 (location - 36.6°N, 89.6°W - Intensity XI-XII), were felt over most of the eastern two-thirds of the United States, an affected area of at least 2,000,000 mi². Topographic changes including uplifts, landslides, and fissures took place over an area of 30,000 to 50,000 mi², principally along the Mississippi and The Beaver Valley Power Station (BVPS) site is located Ohio Rivers. 408 miles from the presently accepted Northern limit of the New Madrid fault zone at Vincennes, Indiana (USNRC 1982). The nearest report of significant damage from these earthquakes came from the Cincinnati, Ohio area about 330 miles from the epicenter and 250 miles from the site. In the Cincinnati area, the tops of chimneys were thrown down and some walls were cracked, indicating a probable Intensity VI (MM), perhaps low VII (MM), when considering the type and quality of construction and the foundation conditions. Fuller (1912) reports that "the earthquake was severe at Pittsburgh, being greater than any previously experienced. Many persons left their houses." Eppley (1965) reports that the earthquake was "strongly felt in Butler County, Pennsylvania." Butler, in the center of Butler County, is about 35 miles east-northeast of the site. Nuttli (1973) has reevaluated ground motion at various locations in the eastern United States and published an isoseismal map of this earthquake which is reproduced on Figure 2.5.2-3. Based on the available data and Nuttli's re-evaluation, the intensity at the site is estimated at low to middle V (MM).

Charleston, S.C. Earthquake August 31, 1886

This earthquake (location - 32.9°N, 80.0°W - Intensity IX-X) was felt over a 2,000,000-mi² area of the eastern United States. In the epicentral area, located a few miles north and west of Charleston, South Carolina, chimneys and fireplaces collapsed, railroad tracks were bent and laterally displaced, and fissures occurred in the ground with ejection of some water, sand, and mud. The area within 100 miles of the epicenter was strongly affected with damage to plaster and chimneys. C.E. Dutton (1889) conducted a thorough investigation of the effects of this earthquake in the epicentral area and throughout the eastern United States. Dutton prepared an isoseismal map which showed a Rossi-Forel Intensity of V (MM) in the vicinity of the site. Reports from Pittsburgh and other towns in the site area indicated a similar intensity except along and near the rivers where somewhat stronger effects were noted. In towns located along rivers, dishes were thrown from shelves and clocks were stopped, indicating an approximate intensity of low V (MM). Bollinger (1977) has reevaluated ground motion at various locations in the eastern United States and published an isoseismal map for this earthquake which is reproduced on Figure 2.5.2-4. The site, located adjacent to the Ohio River, may have experienced Intensity IV-V (MM).

St. Lawrence River Earthquake February 28, 1925

The epicenter was located in the St. Lawrence River Valley (47.6°N, 70.1°W - Intensity IX - Magnitude 7.0) northeast of Quebec City, a distance of 700 miles from the site. The earthquake was felt over an area of approximately 2,000,000 mi², extending south to Virginia and west to the Mississippi River. Important damage was confined to a narrow belt along the St. Lawrence River Valley. Isoseismals prepared by the Dominion Observatory and the United States Coast and Geodetic Survey (Figure 2.5.2-5) show that the estimated intensity at the site was II (MM).

Other earthquakes of Intensity IX and X (MM) have originated in the St. Lawrence River Valley near the epicenter of the February 28, 1925 earthquake. Nearly all of these earthquakes took place during colonial times when reporting of earthquake effects may be accurate in some cases and inaccurate and exaggerated in others. Based on attenuation data and the effects of the February 28, 1925 earthquake, it is estimated that some of these historical earthquakes may have had an intensity of III (MM) in the site area.

Attica, N.Y. Earthquake, August 12, 1929

This earthquake was centered near Attica, New York, (location - 42.9°N, 78.3°W - Intensity VIII - Magnitude 5.8) about 180 miles northeast of the site. It was originally assigned an Intensity VIII (MM) by Coffman and Von Hake (1973), but a re-evaluation by Fox and Spiker (1977) suggested that the epicentral intensity was about VII (MM). The earthquake was felt over a

100,000-mi² area of the northeastern United States and Ontario, Canada, extending from Cleveland, Ohio and Port Huron, Michigan on the west; to Montreal and the Connecticut River Valley on the east. The maximum intensity was confined to the eastern part of the city of Attica and the immediate area to the east, where many chimneys were thrown down and some buildings were structurally damaged. Intensity VI (MM) or greater was noted at Batavia, Dale, East Bethany, Johnsonburg, Warsaw, and Wyoming, New York. All of these localities are within 10 miles of the epicenter.

In the vicinity of the site, intensities ranged from IV (MM) at New Castle (25 miles north) and Butler (35 miles northeast) where windows rattled, to III (MM) at Pittsburgh (25 miles southeast) where the earthquake was only slightly felt. Similar intensities are estimated for the site (U.S. Geological Survey 1974). Figure 2.5.2-6 shows that the site is near the western boundary of the area affected by this earthquake.

Timiskaming, Quebec Earthquake, November 1, 1935

The epicenter was located approximately 425 miles north of the site, near Timiskaming Station, Quebec, (location - 46.8°N, 79.1°W -Intensity VII - Magnitude 6.25) where some damage was reported. The earthquake was felt over a 1,000,000-mi² area of the northeastern United States and eastern Canada. The earthquake was felt as far south as Virginia and Kentucky and as far west as Wisconsin. Damage in the epicentral region was relatively small when compared to the large area affected. Isoseismals prepared by the Dominion Observatory of Canada and the U.S. Coast and Geodetic Survey (1968) (Figure 2.5.2-7) show that the intensity in the vicinity of the site was III (MM).

Anna, Ohio Earthquake, March 8, 1937

This earthquake occurred in western Ohio in the vicinity of Anna (location - 40.6°N, 84.0°W - Intensity VII-VIII) where walls of brick buildings cracked, chimneys were thrown down, and furniture was upset. The earthquake was felt over a 150,000-mi² area including all of Ohio, most of Indiana and adjacent areas of Michigan, Kentucky, West Virginia, and southeastern Ontario, Canada. The site is located at the eastern limit of the perceptible area and may possibly have experienced an intensity of II (MM) (Westland and Heinrich 1940) (Figure 2.5.2-8).

Massena, N.Y. Earthquake, September 4, 1944

The epicenter was located in the vicinity of Massena, New York and Cornwall, Ontario (location - 44.95°N, 74.9°W - Intensity VIII -Magnitude 5.9) about 405 miles northeast of the site. Damage was estimated at two million dollars. The earthquake was felt over an estimated area of 175,000 mi². Isoseismals prepared by the Dominion Observatory of Canada (Figure 2.5.2-9) show that the area of damage (Intensity VI (MM) or greater) was elongated along the St. Lawrence River Valley. The isoseismals show that the intensity in the vicinity of the site was II (MM).

The review shows that the site has experienced vibratory ground motion primarily from large, distant earthquakes, most notably the 1811-1812 earthquakes near New Madrid, Missouri and the 1886 Charleston earthquake. The Attica, New York and Anna, Ohio earthquakes were barely perceptible at the site.

2.5.2.2 Geologic Structures and Tectonic Activity

Portions of four tectonic provinces are located within a 200-mile radius of the BVPS-2 site (Figure 2.5.1-7). The provinces have been defined on the basis of the following criteria:

- 1. Style and degree of deformation,
- 2. Age of the relationships with the basement rock, and
- 3. Age of the orogenic or tectonic activity found within the province.

The four provinces, from northwest to southeast are:

- 1. Central Stable Region,
- 2. Appalachian Plateau Province,
- 3. Valley and Ridge Province, and
- 4. Piedmont-Blue Ridge Province.

They are defined in accordance with 10 CFR 100, Appendix A, which defines a Tectonic Province as "A region of the North American continent characterized by a relative consistency of the geologic structural features contained therein." The conclusions are in general agreement with those of Rodgers (1970), Hadley and Devine (1974), and King (1969). The names for the tectonic provinces are taken from the physiographic provinces with which they generally correspond.

2.5.2.2.1 Appalachian Plateau Province

The BVPS-2 site is located on the Ohio River within the Appalachian Plateau Tectonic Province. Geologically, the province is a broad, gentle synclinal basin whose youngest rocks are the Dunkard Group of probable Early Permian age (Eardley 1962). The basin forms the western part of the former Appalachian geosyncline, with sediments thickening generally southeastward from the Cincinnati-Findlay Arch. Precambrian basement dips beneath the province in the same direction. Deformation in the province occurred primarily during the Permian Allegheny orogeny. The same type of structure exist in both the Appalachian Plateau and Valley and Ridge Provinces, the principal difference being a gradual decrease in intensity of deformation from east to west. "Thin-skinned" tectonics was the dominant mode of deformation in the Appalachian Plateau with movement occurring mainly along sole thrusts in Silurian salt beds and Cambro-Ordovician shales (Rodgers 1964, 1970; Gwinn 1964, 1970). Deep drilling has not as yet delineated the regional extent of thrusting in the Appalachian Plateau. Mild epeirogenic uplift has been the only tectonic event to affect the province since Late Paleozoic time.

Orogenic stresses were persistent and extensive during Permian time and affected all the rocks of the former Appalachian geosyncline. The Valley and Ridge region was the most intensely deformed, with effects diminishing north and west of the Allegheny Front. The Appalachian Plateau region shows only mild deformation and only within the higher stratigraphic units, generally above Silurian evaporites. These effects are recognized in east central Ohio as the Parkersburg-Lorain Syncline and the Cambridge Arch (Figure 2.5.1-7). The syncline can be traced from Parkersburg, West Virginia, north-northwest to Lorain County on Lake Erie. It is a structural trough parallel to the Cambridge Arch to the east and is nearly 5 miles in width with a structural relief of 300 feet (Lamborn 1951). The folds are known to affect the Devonian shale sequence above the Delaware limestone but they have not been investigated at depth (Janssens 1977). These folds are believed to have been formed because of their stratigraphic proximity directly above the low shear resistance zone of Salina salt (Janssens 1977). The thick salt beds are believed to have reduced the resistance to lateral compressive stresses during Permian time, facilitating "thin-skinned" movement of post-Salina rocks (decollement) over a very large area (Gwinn, 1964, 1970; Rodgers 1964, 1970).

Geiser and Engelder (1983) summarize the results of their work on evidence for Allegheny orogenic deformations in New York and eastern Pennsylvania. They believe that the "layer parallel shortening fabrics," which are identified in the rocks of the area "reflect the presence of deeper hidden or blind detachments." The subsurface thrust zones can then be mapped on the basis of the presence of the layer shortening fabrics. Maps showing the limit of the layer parallel shortening fabrics in central and western New York closely coincide with the limit of Silurian evaporite deposits in the Salina Basin.

Unfortunately, similar fabrics were not developed in the Carboniferous rocks on the surface in western Pennsylvania and eastern Ohio. Ver Steeg (1942), however, has identified a form of cleavage in the coal fields of eastern Ohio which controls mining techniques and direction. A set of perpendicular joints, or coal "cleats", is seen to be arranged in a broad arc from north-central Ohio to southern Ohio. The arc is convex to the west and corresponds to the curve of the Appalachian fold belt in eastern Pennsylvania. Ver Steeg (1942) believes that the joints were formed at the same time as the folds.

In order to include all of the rocks which might have been affected by the Allegheny orogenic events, the western boundary of the Appalachian Plateau province is drawn along the mapped limit of Silurian salt (Clifford 1973; SWEC 1978) except where other Allegheny evidence is known to exist further to the west, such as the Parkersburg - Lorain syncline. The present extent of the salt beneath Lake Erie is unknown but most authors indicate continuous salt across the Findlay Arch into the Michigan Basin. Since Allegheny deformation is not known to exist in the Michigan Basin, the boundary of the Appalachian Plateau Province is inferred where it is drawn beneath Lake Erie. South of Ohio, the limit of salt swings eastward, oulining the southeast edge of the Silurian Salina Basin. The Appalachian Plateau Province boundary, however swings westward to include unnamed, low amplitude anticlines and synclines east of Huntington, West Virginia, which parallel similar Allegheny deformational features further east (Rodgers 1970; Cardwell 1977).

The Appalachian Plateau Province is characterized in general by infrequent earthquakes of low intensity. Figures 2.5.2-10 and 2.5.2-11 show that a large part of the Province in the vicinity of the site has experienced less than four earthquakes per 10,000 to 11,680 km². Figure 2.5.2-1 shows that during the entire period of historical reports, only about 18 earthquakes, with epicentral intensities between II and VI (MM), have occurred in the Appalachian Plateau Province, indicating that it is in one of the least seismic areas in the eastern United States.

Barstow, et. al. (1981), using the results of the earthquake frequency study discussed in Section 2.5.2.1.2, developed a contour map of the cumulative strain released by earthquakes that occurred in the eastern United States between 1800 and 1977. A portion of the map is shown on Figure 2.5.2-12. Also shown is a table indicating the amount of strain released by a single event of a given intensity. By comparison with the contours of cumulative strain during the entire historical period, most of the Appalachian Plateau Province has experienced strain release corresponding to an earthquake of Intensity V (MM) or less. Only in the vicinity of Cleveland, Ohio; northwest of the site, is the cumulative strain equivalent to an earthquake of Intensity VII-VIII (MM).

The largest seismic events to occur in the Appalachian Plateau Province, not associated with tectonic structures (Figure 2.5.2-2), are two Intensity VI (MM) events. One is the April 9, 1900 shallow event near Cleveland, Ohio, and the other is the July 13, 1935 event in Blair County, Pennsylvania.

2.5.2.2.2 Central Stable Region

The Central Stable Region bounds the Appalachian Plateau on the north and west and begins about 85 miles from the site. The northern boundary of the Region is the Canadian Shield. Westward, it extends to the east flank of the Rocky Mountains and includes a wide variety of morphology and structure. The Coastal Plain overlaps the region to the south. The Central Stable Region is made up of a foundation of Precambrian crystalline rock with a veneer of sedimentary cover, which varies widely in thickness. It represents the craton or central stable area of the North American crustal plate. Deformation within the region has been restricted to the development of several broad basins, arches, and similar features, mostly during the Paleozoic. Several of the basins have in excess of 10,000 feet of strata in them, while some of the arches expose Precambrian crystallines. Movements since the Paleozoic have been mostly a series of epeirogenic uplifts and downwarps, followed by long episodes of erosion. The largest earthquakes to occur in this province are the March 9, 1937 event near Anna, Ohio, with Intensity VII-VIII (MM) (Coffman and Von Hake 1973), and the Intensity VIII (MM) event of August 12, 1929 near Attica, New York.

2.5.2.2.3 Valley and Ridge Provinces

The Northern and Southern Valley and Ridge Provinces contain the major portions of the sediments which were deposited in the Appalachian geosyncline, of which they comprise the southeastern part. They are bounded on the north and west by the Appalachian Plateau Province, on the south and east by the Piedmont-Blue Ridge Province, and on the northeast end by the New England Province. The Valley and Ridge Provinces are characterized by unmetamorphosed Paleozoic sediments that were tightly folded and thrust faulted during the Allegheny orogeny, approximately 250 million years ago. Intense pressure from the southeast folded the sediments into large synclines and anticlines, some overturned to the northwest. Thrust faults were commonly developed with the horizontal attitude of the sediments barely disturbed. The division of the Valley and Ridge Province into northern and southern sections is based on the difference in structural styles. The northern section is typified by folding, whereas the southern section is characterized by thrust faulting. The southern section has historically experienced a higher level of seismic activity south of an east-west line through central Virginia (Hadley and Devine 1974). This line is somewhat indistinct, but would fall outside the 200-mile site region.

The largest earthquake to occur in the northern province is the February 21, 1954 event near Wilkes Barre, Pennsylvania, approximately 235 miles from the site. This event was estimated to be Intensity VIII (MM) by Sbar and Sykes (1977), and Intensity VIII (MM) by Barstow et. al (1981) and Stover et. al (1981). Damage was restricted to a five-block area, suggesting that the earthquake resulted from the collapse of an abandoned mine and occurred at very shallow depths.

The Northern Valley and Ridge Province lies approximately 105 miles east of the site at its closest approach.

2.5.2.2.4 Piedmont-Blue Ridge Province

The Piedmont-Blue Ridge Province is characterized by metamorphosed Precambrian and early Paleozoic eugeosynclinal rocks which were deformed during the Taconic and Allegheny orogenies and may have been recrystallized during the Acadian orogeny. It includes the Blue Ridge Anticlinorium, a relatively narrow belt of folded and faulted Upper Precambrian crystalline schists and gneisses, which were thrust westward several kilometers over the rocks of the Valley and Ridge. Terrains of intrusive igneous rocks are notable in the Piedmont of Virginia and North Carolina. The eastern part of the province was also effected by the Allegheny orogeny. Long narrow graben structures filled with continental deposits of Late Triassic age are superimposed intermittently on the crystallines from Pennsylvania to South The effects that each orogeny had on the rocks in the Carolina. Piedmont are not yet fully understood, due to the lack of outcrop, lack of fossils, and the strong recrystallization. The

Piedmont-Blue Ridge Province is bounded on the northwest by the Northern and Southern Valley and Ridge Provinces. The southern and eastern boundary of the province is drawn at the present westward limit of Cretaceous Coastal Plain deposits. Piedmont geology certainly continues beneath the Coastal Plain for some distance, but the line where Coastal Plain becomes dominant is presently not well established. The northern boundary of the Piedmont with the New England Province is hidden beneath the Triassic Newark-Gettysburg Basin.

The largest earthquake to occur in this province is the January 1, 1913, event in Union County, South Carolina. Different sources have assigned Intensity VI-VII (MM) (Coffman and Von Hake 1973) or VII-VIII (NM) (Bollinger 1973, 1975). The most recent reference assigns it Intensity VII (MM) (Barstow et. al., 1981).

2.5.2.3 Correlation of Earthquake Activity With Geologic Structure Or Tectonic Provinces

The relationship between earthquake locations and geologic structures is important in assessing earthquake hazard to a particular site. Figure 2.5.2-2 reveals no direct spatial relationship between earthquake epicenters and known geologic or tectonic structures within 200 miles of the site, except in the area of the Clarendon-Linden fault zone in western New York. No evidence of tectonically induced faulting has been reported or inferred to have displaced Cenozoic age deposits in the Appalachian Plateau Province, and no rupture of the ground surface or man-made structures resulting from tectonic faulting has been recorded anywhere in the eastern United States (York and Oliver 1976).

2.5.2.3.1 Correlation with Geologic Structures

Clarendon-Linden Fault Zone

The Clarendon-Linden fault system has been traced from near Lake Ontario in the Central Stable Region to the northern part of Allegheny County in the Appalachian Plateau Province. A significant amount of seismic activity has taken place along the zone (Sbar and Sykes 1977; Pomeroy et al, In Press). Van Tyne (1975, 1976) reports that the Clarendon-Linden fault is not a single fault but a zone consisting of several parallel basement faults which become surface flexures. Most of the movement is believed to be confined to formations below the Silurian deposits. Movement is believed to have been initially downthrown to the east, reversing later to become now downthrown 100 feet on the west. Recent low-level seismic activity has been correlated with high-pressure fluid injection operations in brine fields which are developed in the area (Fletcher and Sykes 1977), and may be relieving stress along the fault system. These events are small but may be felt locally and number up to 80 per day. This activity and the occurrence of several small earthquakes near Attica indicate that the Clarendon-Linden structure is active and therefore is considered as a localized source. The Clarendon-Linden fault zone lies approximately 160 miles northeast of the BVPS-2 site at its closest approach.

Anna Fault System

2.5.2.3.2 Correlation with Tectonic Provinces

In accordance with 10 CFR 100, Appendix A, earthquakes in the site region which are not correlated to the Clarendon-Linden fault system are presumed to be associated with the tectonic provinces in which they occur. The seismicity of all tectonic provinces within the site region is discussed in Section 2.5.2.2.

2.5.2.4 Maximum Earthquake Potential

Maximum earthquake potential for the site is evaluated by utilizing maximum earthquakes associated with all nearby tectonic provinces and geologic structure. This analysis is made for two different sets of conditions. First, actual site intensities resulting from the larger historical earthquakes are determined. Second, the maximum potential site intensities resulting from hypothetical events are specified as arising from the largest known earthquakes in each adjoining tectonic province, postulated to occur at the point where the province or structure most closely approaches the site.

2.5.2.4.1 Maximum Historical Site Intensity

As discussed earlier, in the site region there are sources of earthquake activity at Attica, New York and Anna, Ohio. The largest earthquakes in each of these sources are discussed in Section 2.5.2. These earthquakes were barely perceptible at the site. Other than these two sources, Figure 2.5.2-2 shows that the largest earthquake in the site region took place on November 6, 1926 in southeastern Ohio, approximately 130 miles southwest of the site. The epicentral intensity of this earthquake was VI-VII (MM). Chimneys were toppled at Keno and Pomeroy in Meigs County, Ohio, and a stove was overturned in Pomeroy (Von Hake 1976). As indicated in Table 2.5.2-2, this event had a small felt area of only 300 square miles and was probably not felt at the site. Three earthquakes have been reported within 50 miles of the site (Figure 2.5.2-1). One was reported at Sharon, Pennsylvania, approximately 40 miles north of the site, on August 17, 1873. Limited details have resulted in an estimated Intensity III-IV (MM) for this event. Using the attenuation relationship of Gupta and Nuttli (1976) given in Section 2.5.2.4.2, this event would have attenuated to about an Intensity II at the site; it is not likely that it was felt. On September 26, 1885, an Intensity III (MM) earthquake occurred near Pittsburgh, about 30 miles southeast of the site. It was probably not felt. Another was the Intensity V (MM)

event of October 29, 1927, approximately 45 miles northwest of the site. This event would have attenuated to approximately Intensity III-IV at the site and may have been felt at the site.

This examination of ground motion effects of earthquakes within 200 miles of BVPS-2 indicates that the site has not experienced ground motion exceeding Intensity III-IV (MM). It is likely that in the last 180 years, only the October 29, 1927 earthquake may have been felt at the site. This is clearly a reflection of the low intensity of earthquakes which are reported to have occurred within 200 miles of BVPS-2.

The BVPS-2 site has, however, experienced more severe ground motion from larger, but more distant earthquakes. Examination of isoseismal maps of larger earthquakes felt in the eastern United States shows that the New Madrid events of 1811-1812 probably caused a maximum ground motion at the site corresponding to Intensity low to middle V (MM). Therefore, the maximum historical intensity at the site was probably Intensity low to middle V (MM).

2.5.2.4.2 Maximum Earthquake Potential From Tectonic Province Approach

The BVPS-2 site is located near the center of the Appalachian Plateau Tectonic Province. As discussed in Section 2.5.2.2.1, two earthquakes with epicentral Intensity VI (MM) have occurred in the province. No earthquake with epicentral intensity greater than VI (MM) has been reported to have occurred in the province. None of these earthquakes appear to be associated with a particular known geologic or tectonic structure, so that it is assumed that a similar random event of this intensity could occur anywhere within the province. Consequently, if such an earthquake occurred near the site, it would generate ground motion corresponding to Intensity VI (MM). The magnitude of an earthquake can be estimated on the basis of the Nuttli and Herrmann (1978) relationship:

 $I = 2m_b - 3.5$

where: I = epicentral intensity

 $m_b = body wave magnitude$

Using this relationship, an earthquake of Intensity VI would have a magnitude of 4.75.

The maximum earthquake potential at BVPS-2 from earthquakes in other tectonic provinces is computed by attenuating the largest known earthquake in each province from the point of nearest approach to the site in that province. If the size of the earthquake is specified in terms of epicentral intensity, Io (MM), then the expected intensity at the site can be computed by using the Gupta and Nuttli (1976) attenuation relationship.

Rev. 0

 $I(R) = I_0 + 3.7 - 0.0011R - 2.7 \log_{10}R$

where: I(R) = Intensity at site

 I_{\circ} = Epicentral intensity

R = Epicentral distance (kilometers), for $R \ge 20$ kilometers

If the size of the earthquake is specified in terms of bodywave magnitude, m_b , then ground motion at the site can be computed in terms of peak horizontal acceleration using the Nuttli and Herman (1981) relationship:

 $\log a = 0.57 + 0.50 m_b - 0.83 \log (R^2 + h^2)^{1/2} - 0.0016R$

where: a = peak horizontal acceleration, cm/sec²

 $m_b = body wave magnitude$

R = epicentral distance kilometers

h = focal depth, kilometers.

The largest event to occur in the Central Stable Region was the 1937 Anna, Ohio earthquake which had an epicentral intensity of VII-VIII (MM) and a magnitude, m_b , of 5.3 (Table 2.5.2-2). The minimum distance from the site to the boundary of the Central Stable Region is about 85 miles (136 km). A repeat of the 1937 Anna earthquake at this minimum distance from the site would result in an Intensity V-VI (MM) at the site if the Gupta and Nuttli (1976) attenuation relationship is used. Assuming zero focal depth, an earthquake with a magnitude of 5.3, at a distance of 85 miles, would cause a peak horizontal acceleration of 0.017g at the site according to the Nuttli and Herman (1981) relationship.

The largest earthquake to occur in the Northern Valley and Ridge Province was the February 21, 1954, Wilkes Barre, Pennsylvania event of Intensity VIII (MM) (Coffman and Von Hake 1973). The Northern Valley and Ridge Province lies approximately 105 miles east of the site at its closest approach. This earthquake was felt only within a five block area, suggesting that the earthquake resulted from the collapse of an abandoned mine. Earthquakes with epicentral Intensity VII (MM) occurring in the Northern Valley and Ridge Province at a distance of 105 miles (168 km) will cause ground motions at the site corresponding to Intensity IV-V (MM).

The Piedmont Province is at a minimum distance of 165 miles (264 km) east of the site. The largest earthquake in this province was the January 1, 1913 event with Intensity VII (MM) in Union County, South Carolina. Attenuation of this event to the site results in an Intensity IV (MM) at the site. If it were assumed that the 1929 Attica, New York event, which had an Intensity VII (MM), reoccurred on the Clarendon-Linden fault zone at its closest approach to the site, it would produce ground motions corresponding to Intensity IV (MM) at the site.

Hence, the maximum earthquake for the BVPS-2 site is equivalent to an Intensity VI event occurring in the Appalachian Plateau Province near the site.

2.5.2.5 Seismic Wave Transmission Characteristics of the Site

The amplification characteristics of the soil at the BVPS-2 site were originally discussed by Whitman (1968), and his results led to the development of the BVPS-1 response spectra and, as later modified, to the BVPS-2 response spectra. The BVPS-2 response spectra for the SSE is shown on Figure 3.7B-1.

2.5.2.6 Safe Shutdown Earthquake

The maximum earthquake expected at the site is described in Section 2.5.2.4 and results in ground motion corresponding to Intensity VI (MM). Trifunac and Brady (1975) developed the following correlation between intensity and acceleration:

 $\log a = 0.30I_{mm} + 0.014$

where:

a = Peak horizontal acceleration (cm/sec²)
I_{mm} = Modified Mercalli intensity

Using this correlation, an Intensity VI (MM) earthquake would produce a horizontal acceleration of 0.07g.

Murphy and O'Brien (1977), using a larger data base, determined the following relationship between acceleration and intensity:

 $\log a = 0.25I_{mm} + 0.25$

Using this relationship an Intensity VI (MM) produces an acceleration of 0.06g.

On the basis of these empirical relationships, the peak horizontal ground surface acceleration corresponding to a random Intensity VI (MM) earthquake occurring near the site would be 0.07 g. BVPS-2 has been designed for an SSE corresponding to a peak horizontal ground surface acceleration of 0.125 g, slightly greater than the midpoint acceleration between Intensity VI-VII (MM).

2.5.2.7 Operating Basis Earthquake

An OBE equivalent to one-half the SSE is used. The value of the OBE for BVPS-2 is 0.06.g.

2.5.2.8 References for Section 2.5.2

Algermissen, S.T. and Perkins, D.M., Thenhaus, D.C., Hansen, S.R., and Bender, B.L. 1982. Probabilistic Estimates of Maximum Acceleration and Velocity in Rock in the Contiguous U.S. USGS Open File Report 82-1033.

Barstow, N.L.; Brill, K.G.; Nuttli, O.W.; and Pomeroy, P.W. An Approach to Seismic Zonation for Siting Nuclear Electric Power Generating Facilities in the Eastern United States. Appendix B1. NUREG/CR-1577.

Bollinger, G. A. 1973. Seismicity of Southeastern United States. Bulletin of the Seismological Society of America, Vol. 63, No. 5, p. 1,785 - 1,808.

Bollinger, G.A. 1975. A Catalog of Southeastern United States Earthquakes 1754 through 1974. Virginia Polytechnic Institute and State University. Research Division Bulletin 101.

Bollinger, G. A. 1977. Reinterpretation of the Intensity Data for the 1886 Charleston, South Carolina Earthquake - In: Studies Related to the Charleston, South Carolina Earthquake of 1886 - A Preliminary Report. U.S. Geological Survey Professional Paper 1028, p. 17-32.

Bradley, E. A. and Bennett, T. J. 1965. Earthquake History of Ohio. Bulletin of the Seismological Society of America, Vol. 55, No. 4.

Cardwell, D.H. 1977. Oil and Gas Fields of West Virginia. West Virginia Geological and Economic Survey. Mineral Resources Series No. 7.

Chiburis, E.F.; Ahner, R.O.; and Graham, T. Seismicity of the Northeastern United States. Northeastern United States Seismic Network Bulletins. Nos. 10 through 24. Published by Weston Observatory of Boston College. January 1978 through September 1981.

Clifford, M.J. 1973. Silurian Rock Salt of Ohio. Ohio Geological Survey Report of Investigation No. 90.

Coffman, J. L. and Von Hake, C. A. 1973. Earthquake History of the United States, NOAA Publications 41-1. Revised Edition.

Dutton, Captain C.E. 1889. The Charleston Earthquake of August 31, 1886. Ninth Annual Report, 1887-1888. U.S. Geological Survey.

Eardley, A.J. 1962. Structural Geology of North America. Second Edition. Harper and Row, New York.

Eppley, R. A. 1965. Earthquake History of the United States, Part 1; Stronger Earthquakes of the United States (exclusive of California and Western Nevada). U.S. Coast and Geodetic Survey No. 41-1. Revised Edition through 1963, p. 120.

Fletcher, J. B. and Sykes, L. R. 1977. Earthquakes Related to Hydraulic Mining and Natural Seismic Activity in Western New York. Journal of Geophysical Research, Vol. 82.

Fox, F. L. and Spiker, C. T. 1977. Intensity Rating of the Attica (New York) Earthquake of August 12, 1929. A Proposed Earthquake Reclassification. Earthquake Notes, Vol. 48, Nos. 1-2, p. 37-46.

Fuller, M. L. 1912. The New Madrid Earthquake. Bulletin of U.S. Geological Survey No. 494, U.S. Government Printing Office, Washington, D.C.

Geiser, P. and Engelder, T, 1983. The Distribution of Layer Parallel Shortening Fabrics in the Appalachian Foreland of N.Y. and PA. Evidence for Two Non-Coaxial Phases of the Appalachian Orogeny. Geological Society of America, Memoir 158.

Gupta, I. N. and Nuttli, O. W. 1976. Spatial Attenuation of Intensities for Central United States Earthquakes. Bulletin of Seismological Society of America, Vol. 66, p. 743-751.

Gwinn, V.E. 1964. Thin-Skinned Tectonics in the Plateau and Northern Valley and Ridge Provinces of the Central Appalachians. Bulletin of the Geological Society of America. Vol. 55.

Hadley, J. B. and Devine, J. F. 1974. Seismotectonic Map of the Eastern United States. U.S. Geological Survey Miscellaneous Field Studies, Map MF-620.

Janssens, A. 1977. Oil and Gas in Ohio - Past, Present, and Future, In Janssens, A,, Editor Seminar on Industrial Self-help Programs for Natural Gas Supplies, November 26, 1976: Ohio Dept. Nat. Res., Geol. Surv. Spec. Pub, p. 3-25.

King, P. B. 1969. The Evolution of North America. Princeton University Press, Princeton, N.J.

Lamborn, R.E. 1951. Limestones of Eastern Ohio. Ohio Dept. of Natural Resources. Bulletin of the Division of Geological Survey. No. 49.

Lamont - Doherty Geological Observatory 1974, 1975. Regional Seismicity Bulletin of the Lamont - Doherty Network, Columbia University, Palisades, N.Y.

Murphy, J. R. and O'Brien, L. J. 1977. The Correlation of Peak Ground Acceleration Amplitude with Seismic Intensity, and Other Physical Parameters. Bulletin of the Seismological Society of America, Vol. 67, No. 3, p. 877-915. Nuttli, O. W. 1973. The Mississippi Valley Earthquakes of 1811 and 1812 - Intensities, Ground Motion, and Magnitudes. Bulletin of the Seismological Society of America, Vol. 63, p. 227-248.

Nuttli, O.W. Catalog of Central United States Earthquakes Since 1800 of $M_b \ge 3$. Contained in Barstow et al (1981) Appendix B2.

Nuttli, O. W. and Herrmann, R. B. 1978. State of the Art for Assessing Earthquake Hazards in the United States; Credible Earthquakes for the Central United States, Miscellaneous Paper S-73-1, Report 12, U.S. Army Corps of Engineers, Vicksburg, Mississippi.

Pomeroy, P. W.; and Fakundiny, R. H. 1976. Chronological List of Earthquakes in New York State and Adjacent States. New York State Museum and Science Service, Albany, N.Y.

Pomeroy, P. W.; Nowak, T. A.; and Fakundiny, R. H. Clarendon-Linden Fault System of Western New York - A Vibroseis Seismic Study. In Press.

Rodgers, J. 1970. The Tectonics of the Appalachians. Wiley Interscience, New York, N.Y.

St. Louis University. Regional Seismic Network Bulletins Nos. 19 through 35. January 1979 through March 1983.

Sbar, M. L. and Sykes, L. R. 1977. Seismicity and Lithospheric Stress in New York and Adjacent Areas. Journal of Geophysical Research, Vol. 82.

Smith, W. E. T. 1966. Earthquakes of Eastern Canada and Adjacent Areas, 1928-1959. Publications of the Dominion Observatory, Ottawa, Vol. 32, No. 3. Canada Department of Mines and Technical Surveys.

Stone & Webster Engineering Corp. (SWEC). 1978. Regional Geology of the Salina Basin. Prepared for Office of Nuclear Waste Isolation, ONWI/SUB-E512-00600/1, Vol. 1.

Stone & Webster Engineering Corporation (SWEC). 1980 Earthquake Data Retrieval, VOLO, GT038. (Based upon data tape prepared by National Oceanic & Atmospheric Administration).

Stover, C.W.; Reagor, B.G.; and Algermissen, S.T. 1981. Seismicity Map of the State of Pennsylvania. Miscellaneous Field Studies Map No. MF-1280. Department of the Interior, United States Geological Survey.

Trifunac, M. D. and Brady, A. G. 1975. On the Correlation of Seismic Intensity Scales with the Peaks of Recorded Strong Motion. Bulletin of Seismological Society of America, Vol. 65, No. 1, p. 139-162.

United States Coast and Geodetic Survey 1968, 1946-1974. United States Earthquakes 1928-1935, 1936-1940, 1941-1945. Annual Editions. U.S. Government Printing Office, Washington, D.C.

United States Geologic Survey 1974. Geological Map of the United States 1:2,500,000

U.S. Nuclear Regulatory Commission (USNRC) 1982. Safety Evaluation Report, Clinton Power Station - Unit 1, Illinois Power Company, Docket No. 50-461, July 31, 1982.

Van Tyne, A. M. 1975. Clarendon-Linden Structure, Western New York, N.Y. State Geological Survey Open File.

Van Tyne, A. M. 1976. Subsurface Investigation of the Clarendon-Linden Structure, Western New York. Empire State Geogram, Vol. 12, p. 13.

Ver Steeg, K. 1942. Jointing in the Coal Beds of Ohio. Economic Geology - V37.

Von Hake, C.A. 1976. Earthquake History of Ohio. Bulletin of Earthquake Information, Vol. 8, No. 1, p. 28-30.

Westland, A.J. and Heinrich, R.R. 1940. A Macroscopic Study of the Ohio Earthquake of March 1937. Bulletin of the Seismological Society of America, Vol. 30, P. 251-260.

Whitman, R.V. 1968. Effect of Local Soil Conditions Upon Seismic Threat to Beaver Valley Power Station. Appendix 2D, BVPS-1 Preliminary Safety Analysis Report.

York, J. E. and Oliver, J. E. 1976. Cretaceous and Cenozoic Faulting in Eastern North America. Bulletin of the Geological Society of America, Vol. 87, p. 1,105-1,114. Tables for Section 2.5.2

TABLE 2.5.2-1

MODIFIED MERCALLI INTENSITY SCALE OF 1931

- I Not felt except by a very few under especially favorable circumstances.
- II Felt only by a few persons at rest, especially on upper floors of buildings. Delicately suspended objects may swing.
- III Felt quite noticeably indoors, especially on upper floors of buildings, but many people do not recognize it as an earth-quake. Standing motor cars may rock slightly. Vibration like passing of truck. Duration estimated.
- IV During the day felt indoors by many, outdoors by few. At night some awakened. Dishes, windows, doors disturbed; walls make cracking sound. Sensation like heavy truck striking building Standing motor cars rocked noticeably.
- V Felt by nearly everyone, many awakened. Some dishes, windows, etc., broken; a few instances of cracked plaster; unstable objects overturned. Disturbances of trees, poles, and other tall objects sometimes noticed. Pendulum clocks may stop.
- VI Felt by all, many frightened and run outdoors. Some heavy furniture moved; a few instances of fallen plaster or damaged chimneys. Damage slight.
- VII Everyone runs outdoors. Damage negligible in buildings of good design and construction; slight to moderate in well built ordinary structures; considerable in poorly built or badly designed structures; some chimneys broken. Noticed by persons driving motor cars.
- VIII Damage slight in specially designed structures; considerable in ordinary substantial buildings with partial collapse; great in poorly built structures. Panel walls thrown out of frame structures. Fall of chimneys, factory stacks, columns, monuments, walls. Heavy furniture overturned. Sand and mud ejected in small amounts. Changes in well water. Persons driving motor cars disturbed.
 - IX Damage considerable in specially designed structures; well designed frame structures thrown out of plumb; great in substantial buildings, with partial collapse. Buildings shifted off foundations. Ground cracked conspicuously. Underground pipes broken.

- X Some well built wooden structures destroyed; most masonry and frame structures destroyed with foundations; ground badly cracked. Rails bent. Landslides considerable from river banks and steep slopes. Shifted sand and mud. Water splashed (slopped) over banks.
- XI Few, if any, (masonry) structures remain standing. Bridges destroyed. Broad fissures in ground. Underground pipelines completely out of service. Earth slumps and land slips in soft ground. Rails bent greatly.
- XII Damage total. Practically all works of construction are damaged greatly or destroyed. Waves seen on ground surface. Lines of sight and level are distorted. Objects are thrown upward into the air.

TABLE 2.5.2-2

EARTHQUAKE CATALOG BEAVER VALLEY POWER STATION - UNIT 2 200-MILE RADIUS

Year	Date Month	Day	Origin Time	Latitude (°N)	Longitude (°W)	Intensity (MM)	Depth (km)	Magnitude	Felt Area (x10 ³ mi ²)	Location
1776				39.9	82.0	VI	<u> </u>	<u> </u>	<u> </u>	
1796	12	26	1100	42.9	79.0	VI			75	
1823	05	30	1100	41.5	81.0	IV		3.8	1.0	
1824	07	15	1620	39.7	80.5	IV		0.0		
1836	07	08	1020	41.5	81.7	IV		3.8		
1840	09	10		43.2	79.9	V		0.0		Hamilton ON
1845	00	10		41 1	84.2	, II		3.0		
1846	10	18	2100	39.4	77.8			0.0		
1846	10	10	0200	30.3	77.9					
1850	10	01	1001	41 4	82.3	IV/		3.8		
1853	01	30	1001	38.0	78.5	10		5.0		
1953	03	13	1000	43.1	70.5	11 \/				St. Catharinas, ON
1055	05	02	0020	43.1	79.4	V			72.0	St. Cathannes, ON
1000	05	16	0920	30.3	79.0	V IV			72.0	
1850	01	10	0300	39.3	78.2			4.0		
1857	03	01		41.7	81.2	IV-V		4.0		
1857	12	10	2200	37.8	80.4					
1857	12	11	0300	37.8	80.5					
1858	01	15		43.1	79.1	Ш				Niagara Falls, ON
1858	04	16	1200	41.7	81.3	IV		3.8		
1867	01	13		41.5	81.7	III		3.4		
1869	04	09		42.7	80.8	111		3.4		

Year	Date <u>Month</u>	Day	Origin <u>Time</u>	Latitude <u>(°N)</u>	Longitude <u>(°W)</u>	Intensity <u>(MM)</u>	Depth <u>(km)</u>	<u>Magnitude</u>	Felt Area <u>(x10³ mi²)</u>	Location
1872	07	23		41.4	82.1	IV		3.8		
1873	04	30	PM	43.3	79.9	IV				Hamilton, ON
1873	08	17	1400	41.2	80.5					Sharon, PA
1873	07	06	1430	43.0	79.5	VI				Welland, ON
1875	06	18	1343	40.2	84.0	VII		5.3	38.6	
1876	02	27		42.4	83.2	Ш		3.0		
1876	06			40.4	84.2	V		4.2		
1877	08	17	1650	42.3	83.3	IV-V*		4.0 (3.2)	0.2	
1879	08	21	0800	43.2	79.2	V				
1881	07	31	0400	39.1	83.4					
1881	08	30	0500	39.2	83.7	III		3.4		
1882	02	09	2000	40.4	84.2	V		4.2	0.1	
1882	04	02		38.6	78.5	Ш				
1882	11	27	2330	43.0	79.2	IV				Welland, ON
1882	12	04	2330	43.0	79.2	Ш				Welland, ON
1883	04	01		43.3	79.9	III				Hamilton, ON
1883	05	23	0430	38.4	82.6	IV		3.8		
1884	09	19	2014	40.7	84.1	VI		4.7	123.6	
1884	12	23	2300	40.4	84.2	111		3.4		
1885	01	02	2116	39.2	77.5	V			3.5	
1885	01	18	1030	41.1	81.4	(IV)		(3.8)		
1885	01	18	1130	41.3	81.1	111		3.4		
1885	08	15	0505	41.3	81.1	Ш		3.2		

Year	Date <u>Month</u>	Day	Origin <u>Time</u>	Latitude <u>(°N)</u>	Longitude (°W)	Intensity <u>(MM)</u>	Depth <u>(km)</u>	<u>Magnitude</u>	Felt Area <u>(x10³ mi²)</u>	Location
1886	05	03	0300	39.5	82.1	III-IV		3.6	0.4	
1886	09	02		43.2	79.2	Ш				St. Catharines, ON
1889	09	00		40.4	84.2	III		3.4		
1885	09	26	2030	40.3	80.1					
1896	03	15	0700	40.3	84.2	IV		3.8		
1897	03	07		43.1	79.2	IV				Niagara Falls, ON
1898	10	24		41.5	81.7	III-IV		3.6		
1899	11	12	1400	39.3	83.0	IV		3.8		
1900	04	09	1400	41.4	81.8	VI*		4.7 (3.8)		
1901	05	17	0700	39.3	82.5	V		4.2	9.7	
1902	03	10	1030	39.6	77.1					
1902	03	11		39.6	77.1					
1902	06	14	0700	40.3	81.4	IV-V		4.0		
1903	01	01	1730	39.6	77.1	Ш				
1903	01	01	2045	39.6	77.1	I				
1906	04	20	1730	41.5	81.7	(III)		(3.4)		
1906	04	20	1830	41.5	81.7	IV		3.8		
1906	04	23	0712	40.7	83.6	V		4.2		
1906	06	27	1210	40.4	81.6	V*		4.2 (3.4)	0.4	
1906	06	27	2210	41.4	81.6	V				
1907	01	10	1000	41.2	77.1	IV				Williamsport, PA
1907	04	12		41.5	81.7	111		3.0		
1909	04	02	0725	39.4	78.0	VI			2.5	

Voar	Date	Dav	Origin			Intensity	Depth	Magnitude	Felt Area (x10 ³ mi ²)	Location
<u>1001</u>	<u>inontri</u>	Day	<u>mile</u>	<u>(N)</u>	<u>(vv)</u>		<u>(KIII)</u>	Magintade	<u>(xio iii)</u>	Location
1910	01	23	2120	39.6	77.0	Ш				
1910	02	08	0900	38.7	78.7	IV			1.1	
1910	02	25	PM	43.2	79.8	IV				Hamilton, ON
1912	03	27	1252	43.2	79.7	V				Hamilton, ON
1914				40.4	84.2	111		3.4		
1918	04	09	1808	38.5	79.0	П				
1918	04	16	1340	38.6	78.5	П				
1919	09	05	2146	38.8	78.2	VI				
1919	09	06	0246	38.8	78.2	VI				
1920	07	24		38.7	78.4	IV				
1921	09	27	0432	42.1	80.2	111				Erie, PA
1922	03	16	0930	43.0	82.5	111		3.4		
1923	12	31	2400	39.2	78.0	V				
1924	01	01		39.2	78.0	IV				
1924	01	01	0500	39.1	78.1	111				
1925	03	27	0406	39.5	83.9	V		4.2		
1925	10			40.4	84.2	111		3.4		
1926	10	28	0842	41.7	83.6	111		3.6		
1926	10	28	1100	41.7	83.6	IV		3.8		
1926	11	05	1553	39.1	82.1	VI-VII*		4.0 (3.4)	0.3	
1927	01	17	0530	40.7	82.5	IV		3.8		
1927	02	17	0600	40.7	82.5	П		3.0		
1927	06	10	0716	38.0	79.0	V		2.9		

	Date		Origin	Latitude	Longitude	Intensity	Depth		Felt Area	
Year	<u>Month</u>	Day	Time	<u>(°N)</u>	<u>(°W)</u>	<u>(MM)</u>	<u>(km)</u>	<u>Magnitude</u>	<u>(x10³ mi²)</u>	Location
1927	10	29		40.9	81.2	V		4.2		
1927	11	12		43.1	79.1	IV				Niagara Falls, ON
1927	11	13	0050	43.1	79.1	IV				Niagara Falls, ON
1928	09	09	2100	41.5	82.0	V		4.2	1.5	
1928	10	27		40.4	84.1	III		3.4	0.1	
1929	03	08	0906	40.6	84.2	V		4.2	5.0	
1929	08	12	1124	42.9	78.4	VIII		5.8		Attica, NY
1929	09	17	1900	41.5	81.5	III		3.0		
1929	12	02	2214	42.8	78.3	V				Attica, NY
1929	12	03	1250	42.8	78.3	V				
1930	01	17		42.8	78.3	III				
1930	02	16	1217	42.8	80.5	111				Simcoe, ON
1930	06	26	2145	40.5	84.0	IV		3.8		
1930	06	27	0723	40.5	84.0	IV		3.8		
1930	07	11	0015	40.6	83.2	IV		3.8		
1930	09	29	2115	40.4	84.2	111		3.4		
1930	09	29	2250	40.3	84.2	111				
1930	09	30	2040	40.3	84.3	VII*		5.3 (4.2)		
1930	10			40.4	84.2	III-IV		3.6		
1930	11	20		42.6	83.2	III		3.4		
1931	03	21	1548	40.4	84.2	III		3.4		
1931	04	01	0015	40.4	84.0	111		3.4		
1931	04	22		42.9	78.9	IV				Buffalo, NY
Year	Date <u>Month</u>	Day	Origin <u>Time</u>	Latitude <u>(°N)</u>	Longitude <u>(°W)</u>	Intensity <u>(MM)</u>	Depth <u>(km)</u>	<u>Magnitude</u>	Felt Area <u>(x10³ mi²)</u>	Location
------	----------------------	-----	-----------------------	-------------------------	--------------------------	--------------------------	----------------------	------------------	---	----------------
1931	06	10	0830	41.3	84.0	V		4.2	1.5	
1931	09	20	2305	40.4	84.2	VII		5.3	46.3	
1931	10			40.4	84.2	III				
1931	10	08	1430	40.4	84.2	III		3.4		
1932	01	22		41.1	81.5	V*		4.2 (3.6)		
1933	02	23	0320	40.3	84.2	IV		3.8	1.9	
1934	10	29	2007	42.0	80.2	V				Erie, PA
1934	11	05	2000	41.8	80.3	III				
1935	07	13		40.5	78.5	VI				Blair Co., PA
1935	11	01	0330	38.9	79.9	V				
1935	11	01	2030	39.9	79.9	V				
1936	01	31	0630	41.1	83.2	III				
1936	01	31	1930	41.2	83.2	IV		3.8		
1936	01	31	2000	41.2	83.2	П		3.0		
1936	08	26	0900	41.4	80.4	111				Greenville, PA
1937	03	02	1447	40.4	84.2	VII		5.3	108.1	
1937	03	03	0950	40.7	84.0	V		4.2	0.2	
1937	03	03	0955	40.7	84.0	111		3.4		
1937	03	09	0544	40.4	84.2	VII-VIII		5.3	193.1	Anna, OH
1937	04	23	1715	40.7	84.0	111		3.4	0.3	
1937	04	27	1700	40.7	84.0	111		3.4	0.3	
1937	05	02	1705	40.7	84.0	IV		3.8		
1938	03	13	1610	42.4	83.2	(IV)		(3.8)		

Year	Date Month	Dav	Origin Time	Latitude (°N)	Longitude	Intensity (MM)	Depth (km)	Magnitude	Felt Area (x10 ³ mi ²)	Location
1029	02	16	1000	42.4	<u></u>	<u>,</u>	<u>,</u>	2.0	<u></u>	
1938	03	10	1000	42.4	83.Z	IV		3.8		
1938	07	15	2245	40.7	78.4	V-VI				Blair Co., PA
1939	01	14	0810	43.3	79.9			3.3		Hamilton, ON
1939	02	24	0020	42.9	78.3	111				Attica, NY
1939	03	18		40.4	84.0	Ш		3.0		
1939	03	18	1403	40.4	84.0	III-IV		3.6	0.5	
1939	06	18	0320	40.3	84.0	IV		3.8	0.4	
1939	07	09	1250	40.3	84.0	II		3.0		
1939	11	26	0520	39.9	76.9					
1940	05	28	2006	40.3	76.9	II				Harrisburg, PA
1940	05	31	1700	41.1	81.5	Ш		3.0		
1940	06	16	0230	39.9	82.2	Ш				
1940	06	16	0430	40.9	82.3	IV		3.8		
1940	07	28	0930	40.9	82.3	III		3.4		
1940	08	15	1035	40.9	82.3	III		3.4		
1940	08	19	0330	39.9	82.2	II				
1940	08	20	0330	40.9	82.3	III		3.4		
1943	03	09	0325	42.2	80.9	V		4.7	84.9	
1944	02	26	2058	42.9	78.8	II				Buffalo, NY
1948	01	18		41.7	83.6	III		3.4		
1951	12	03	0200	41.6	81.4	IV		3.8	0.1	
1951	12	03	0702	41.6	81.4	(IV)		(3.2)	(0.1)	
1951	12	07		41.6	81.4	II		3.0		

Veen	Date		Origin	Latitude	Longitude	Intensity	Depth		Felt Area	Leasting
<u>Year</u>	Month	<u>Day</u>	lime	<u>(°N)</u>	<u>(°W)</u>	<u>(IMIM)</u>	<u>(ĸm)</u>	Magnitude	<u>(x10° mi⁻)</u>	Location
1951	12	21	2100	41.6	81.5	II				
1951	12	22	0400	41.6	81.4	II		3.0		
1952	06	20	0938	39.7	82.1	VI		4.7	5.0	
1953	05	07	2332	39.7	82.1	IV		3.8		
1953	06	12		41.7	83.6	IV		3.8		
1954	04	27	0214	43.1	79.2			4.1		Welland, ON
1955	05	26	1809	41.5	81.7	V (IV-V)*		3.8 (3.6)		
1955	06	29	0116	41.5	81.7	V (IV)*		3.8 (3.6)		
1955	08	16	0735	42.9	78.3	V				Attica, NY
1956	01	27	1203	40.4	84.2	V		4.2	1.9	
1957	06	29		42.9	81.3	IV		3.8		
1958	05	01	2247	41.5	81.7	IV-V		4.0		
1958	07	22	0146	43.0	79.5			4.3		Welland, ON
1958	08	04	2025	43.1	80.0	(IV)		3.9		Caledonia, ON
1958	08	04	2025	43.1	80.0	(IV)		3.9		Caledonia, ON
1958	08	22	1425	43.0	79.0			3.6		Niagara Peninsula, ON
1959	02	09	0200	43.0	81.0			2.4		London, ON
1959	02	09		43.0	81.0	(IV)		3.8		
1961	02	22	0645	41.2	83.3	V		4.2	5.0	
1961	02	22	0844	41.0	83.6	III				
1961	02	22	0945	41.2	83.3	V		(4.0)	(5.0)	
1962	03	27	0635	43.0	79.3	V		3.0		Niagara Falls, NY
1962	03	27	0737	42.9	79.0	V				

	Date		Origin	Latitude	Longitude	Intensity	Depth		Felt Area	
<u>Year</u>	<u>Month</u>	Day	<u>Time</u>	<u>(°N)</u>	<u>(°W)</u>	<u>(MM)</u>	<u>(km)</u>	<u>Magnitude</u>	<u>(x10³ mi²)</u>	Location
1962	09	07	1400	39.7	78.2	IV				
1963	02	27	0600	43.2	79.6			3.0		Grimsby, ON
1963	10	10	1500	39.8	78.2		15.0			
1964	02	13		40.4	78.2	VI		4.6		Blair Co., PA
1964	02	13	1946	40.4	78.2			5.2		Non-tectonic event **
1965	07	16	1100	42.9	78.2	IV		3.5		Attica, NY
1965			0157	42.9	78.2	IV				Attica, NY
1965	08	27		42.9	78.2	IV				Attica, NY
1965	10	08	0217	40.1	79.8			3.3		Southwestern PA
1966	01	01	1030	42.8	78.2					Attica, NY
1966	01	01	1129	42.8	78.3			3.0		Attica, NY
1966	01	01	1323	42.8	78.2	VI*		4.7(4.6)		Attica, NY
1966	09	28	2059	39.3	80.4	IV		(3.8)		
1967	04	08	0541	39.6	82.5	V		4.2	3.9	
1967	06	13	1908	42.9	78.2	VI*		3.9(4.4)		Attica, NY
1968	07	26		40.4	84.2	11-111		3.2		
1969	05	22	1500	39.7	78.2					
1969	08	13		42.9	78.2	IV		2.5		Attica, NY
1970	05	27	1800	39.7	78.2					
1970	08	11	0614	38.4	82.3	IV		3.8		
1970	12	13	0536	42.7	78.7			2.0		SW of Hamburg, NY
1971	02	18	1930	39.7	78.2					
1971	03	05	1719	40.7	78.0					Non-tectonic event**
1972	09	12	1715	39.7	79.9					

	Date		Origin	Latitude	Longitude	Intensity	Depth		Felt Area	
<u>Year</u>	<u>Month</u>	<u>Day</u>	<u>Time</u>	<u>(°N)</u>	<u>(°W)</u>	<u>(MM)</u>	<u>(km)</u>	<u>Magnitude</u>	<u>(x10³ mi²)</u>	Location
1973	02	09	0446	42.8	78.3			2.7		SE of Attica, NY
1974	03	23	0947	38.9	77.8			2.5		
1974	09	29	0226	41.2	83.4	II		3.0		
1974	10	10	2146	42.3	77.7			2.2		Hornell, NY
1974	10	20	1514	39.1	81.6	V		3.4		
1974	11	27	1028	43.3	79.1			3.3		
1975	02	03	1031	41.3	83.2	(IV)		(3.8)		
1975	02	16	2322	39.0	82.4	(IV)		(3.8) 3.3		
1975	06	30	2015	43.4	79.8	111		3.0		
1975	07	01	0010	42.8	78.6			2.4		Lancaster, NY
1975	08	30	0614	42.7	78.1			2.1		S of Warsaw, NY
1975	10	31	0026	42.8	78.2					Attica, NY
1975	11	20	1502	42.9	78.2				1.5	Attica, NY
1975	11	20	1504	42.9	78.2					Attica, NY
1975	11	29	1222	42.8	78.2					Attica, NY
1975	12	01	2341	42.8	78.2					Attica, NY
1976	01	01	2118	42.9	78.2					Attica, NY
1976	01	10	2114	42.8	78.2					Attica, NY
1976	01	14	2008	42.8	78.2					Attica, NY
1976	01	30	1859	39.7	78.2		15	2.8		
1976	02	02	2114	42.0	82.7	(111)		3.4		Leamington, ON
1978	04	26	1930	89.7	78.2		15	3.1		
1978	05	13	2156	42.8	78.3		16	2.8		W of Attica, NY
1978	05	13	2209	42.8	78.3		6	2.6		W of Attica, NY

	Date		Origin	Latitude	Longitude	Intensity	Depth		Felt Area	
Year	<u>Month</u>	Day	Time	<u>(°N)</u>	<u>(°W)</u>	<u>(MM)</u>	<u>(km)</u>	Magnitude	<u>(x10³ mi²)</u>	Location
1978	10	15	2237	43.2	80.5		OR	2.2		Drumbo, ON
1978	10	26	2154	42.7	77.8		6	2.6		Mount Morris, NY
1980	01	21	0616	43.3	79.8		5	2.5		E. Hamilton, ON
1980	08	20	0935	42.1	83.1	V	18R	3.3		Lake Erie, OH
1980	10	14	0059	43.1	80.6	Felt	18R	3.5		ON
1981	01	07	0503	43.2	80.4		6.7	2.8		Near Brantford, ON
1981	03	31	0541	42.9	78.3		4.9	1.4		Attica, NY
1981	03	31	2105	42.9	78.3		6.2	2.8		Attica, NY
1981	08	28	1051	43.2	80.6	111	1R	3.3		ON
1981	09	05	0547	42.7	81.4		9	1.9		15 km, S of DLA, ON
1981	09	05	0549	42.8	81.5		9R			15 km S of DLA, ON
1981	09	05	0549	42.8	81.4		9R	3.1		7 km S of DLA, ON
1981	09	11	1625	43.4	79.8		2.7	2.2		Burglington, ON

NOTES:

*Indicates shallow earthquake per Nuttli (1981).

Data in parentheses taken from Nuttli (1981).

**Stover, Reagor, and Algermission (1981).





LEGEND:

- п-ш
- I<u>V</u> <u>V</u>

VII OR GREATER AS NOTED

NO INTENSITY DATA

SHALLOW EARTHQUAKE

(1885) DATE

0	25	50	75	100
Ĺ		1		
	SC	ALE-MIL	ES	

NOTE:

INTENSITIES ARE MODIFIED MERCALLI (MM) SEE TABLE 2.5.2 -1

FIGURE 2.5.2-1 EPICENTERS AND FELT REPORTS WITHIN 200 MILES OF THE SITE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





WITHIN 200 MILES OF THE SITE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



NOTE: Nuttli 1973

> FIGURE 2.5.2-3 ISOSEISMAL MAP NEW MADRID EARTHQUAKE, 1811 BEAVER VALLEY POWER STATION-UNIT-2 FINAL SAFETY ANALYSIS REPORT











BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT









2.5.3 Surface Faulting

No surface faulting exists at or near the site. There are no known active or capable faults within 150 miles of the site. There has been no mining, hydrocarbon extraction, or other activity beneath the site which could cause ground rupture at the site.

2.5.3.1 Geologic Conditions of the Site

The geologic conditions in the region and at the site are described in Sections 2.5.1.1 and 2.5.1.2, respectively.

2.5.3.2 Evidence of Fault Offset

There is no evidence of fault offset at the ground surface within 5 miles of the site or in any subsurface boring taken at the site.

2.5.3.3 Earthquakes Associated with Capable Faults

There is no seismic or geologic evidence of capable faulting within 5 miles of the site.

2.5.3.4 Investigation of Capable Faults

There are no known capable faults within 5 miles of the site requiring investigation.

2.5.3.5 Correlation of Epicenters with Capable Faults

There is no seismic evidence to indicate capable faulting within 5 miles of the site to correlate with epicenters.

2.5.3.6 Description of Capable Faults

There are no known capable faults within 5 miles of the site.

2.5.3.7 Zone Requiring Detailed Faulting Investigation

No zone requiring a detailed faulting investigation has been identified within 5 miles of the site.

2.5.3.8 Results of Faulting Investigation

A fault zone requiring a detailed investigation has not been identified in the site area.

2.5.4 Stability of Subsurface Materials and Foundations

This section presents the results of investigations and studies conducted to evaluate the stability of subsurface soils at the site and the foundations which they support. Engineering properties of soils were determined based on detailed field and laboratory investigations described herein. The evaluation of subsurface conditions and soil properties and the results of stability analyses are presented under the following specific headings.

2.5.4.1 Geologic Features

The major structures of Beaver Valley Power Station (BVPS) are located on the highest of three terraces along the south side of the Ohio River. They are composed predominantly of alluvial deposits derived from the cyclic aggradation and degradation of local materials and glacial outwash by the ancestral Ohio River drainage system during the Pleistocene period. Figure 2.5.4-1 is a typical north-south cross section through the site showing the terraces.

The Upper Pleistocene terrace slopes gently toward the Ohio River from about el 760 feet to 735 feet. The soils of this terrace consist predominantly of interbedded sands, gravels, and silty sands and gravels.

A zone of loose granular material from approximately el 640 feet to 660 feet was discovered in the plant area during the excavation for the Beaver Valley Power Station - Unit 2 (BVPS-2) containment foundation. The loose zone was present under approximately the northern portion of the containment and extended east and west beneath most of the Category I structures. The loose zone was successfully densified using the pressure injected footing technique. The extent of the densified area is shown on Figure 2.5.4-15. It has been determined that the densified in situ soil will be stable under all anticipated loading and environmental conditions. The densification program and its evaluation are fully described in the Duquesne Light Company (DLC 1976) Report on Soil Densification Program.

The near surface soils of the intermediate terrace (original ground surface el 685 feet to 700 feet) and the present floodplain (original ground surface el 675 feet) consist of medium stiff to soft clays and silts. These recent river silts and clays extend to approximately el 655 feet where they are underlain by sands and gravels to bedrock. The intermediate terrace is overlain in part by fill placed during the construction of Shippingport Atomic Power Station (SAPS) and Beaver Valley Power Station - Unit 1 (BVPS-1).

The bedrock in the area of the site is Pennsylvanian in age and belongs to the Allegheny group which consists of interbedded sandstones, shales, coal seams, and occasional limestones. The rock underlying the plant site is a dark gray carbonaceous shale which dips gently southeastward about 15 to 20 ft/mi. The rock is slightly weathered for the first few feet with weathering effects decreasing rapidly with depth. A top of rock contour map is provided on Figure 2.5.4-50. The computer program SURFACE II (Sampson 1975) was used as an aid in developing the map. Input to SURFACE II was the top of rock elevations from the irregularly spaced exploratory borings. SURFACE II establishes a regularly spaced, orthogonal grid and interpolates top of rock elevations at these grid points utilizing a data fitting algorithm specified by the user. The computer develops the contour map from the interpolated grid elevations.

A check was made to verify that the computer-drawn map was an accurate representation of the top of rock elevations from the boring data. It was necessary to manually re-contour the computer-generated 800 to 875-feet contours because the computer-drawn contours did not adequately represent the top of rock elevations in an area of known rock surface geometry. Only the exploratory borings were used to evaluate the top of rock elevation; the verification borings performed for several densification programs in the main plant area provided redundant data and were not used.

As discussed in Section 2.5.1.2, there are no areas of actual or potential subsurface subsidence at the plant site.

The ground-water conditions at the site are discussed in detail in Sections 2.4.13 and 2.5.4.6.

Unrelieved residual stresses in rock were not considered to have an influence on the design and operation of the plant due to the thickness of the founding overburden.

2.5.4.2 Properties of Subsurface Materials

The BVPS is founded on the highest of three alluvial terraces. These terraces are described in more detail in Section 2.5.4.1.

Borings drilled within the main plant area indicate a subsurface profile consisting of about 115 feet of medium dense to dense granular soils underlain by shale bedrock with a top surface at about el 620 feet. A zone of loose granular material between about el 640 feet and 660 feet was discovered in the BVPS-2 plant area and was subsequently densified using the pressure injected footing technique (DLC 1976).

A lens of very stiff, silty clay was uncovered along the northern edge of the reactor containment excavation at about el 679 feet, the presence of which was not noted during the original subsurface investigation. The clay lens was removed from within the containment area and replaced with compacted structural fill. It extends eastward and is present beneath the northern portions of the safeguards area and the refueling water storage tank (RWST). Its areal extent is discussed further in Section 2.5.4.7.1.

Subsurface profiles are presented on Figures 2.5.4-2, 2.5.4-3, 2.5.4-4, 2.5.4-5, 2.5.4-6, 2.5.4-7, 2.5.4-8 and 2.5.4-9, Figures 2.5.4-51, 2.5.4-52, 2.5.4-53, 2.5.4-54 and 2.5.4-55, and Figure 2.5.4-60, the locations of which are shown on Figures 2.5.4-10 and 2.5.4-13. The profiles are based upon the data from the boring logs which are referenced in Table 2.5.4-1.

An extensive laboratory testing effort was undertaken to establish the engineering and index properties of the intermediate and lower terrace silts and clays. The results are presented and summarized in Appendix 2.5D. Included are grain size analyses performed on samples of the in situ sands and gravels and the results of in place density

tests conducted during the documentation of the soil conditions at the reactor containment founding elevation (Section 2.5.4.5).

Attempts to obtain undisturbed samples of the in situ sands and gravels were unsuccessful. Consequently, the engineering properties of the sands and gravels developed for design purposes were based upon accepted conservative, empirical correlations of engineering properties to subsurface conditions determined by geophysical surveys, test borings, and field testing.

A plot of relative density versus blow count for the in situ sands and gravels is shown on Figure 2.5.4-11 for borings outside of the area densified by the pressure-injected footing technique. (Borings within the densified area are discussed in DLC'S 1976 Report on Soil Densification Program.) The relative density of most of the samples fall within the range of 50-80 percent, classifying the in situ sands and gravels as medium dense to dense (Terzaghi and Peck 1967). Based on a correlation with relative density (U.S. Department of the Navy 1971), the angle of internal friction of the in situ sands and gravels may range between about 33 and 40 degrees; an angle of 30 degrees was conservatively chosen for design purposes.

The dry unit weight of the in situ sands and gravels was taken as 117 pcf, based on an average of in place density tests performed during excavation for plant structures.

The specific gravity was taken as 2.65 and was based upon laboratory determination (Appendix 2.5D).

The void ratio was computed to be 0.4 from the equation:

$$e = \frac{G_{\gamma_w} - \gamma_d}{\gamma_d} \tag{2.5.4-1}$$

where:

e = void ratio G = specific gravity γ_d = dry unit weight (pcf) γ_w = unit weight of water (62.4 pcf)

The saturated unit weight below the ground-water table was taken as 136 pcf from the equation:

$$\gamma_T = \frac{G + Se}{1+e} \bullet \gamma_w \tag{2.5.4-2}$$

where:

 γ_T = total unit weight S = degree of saturation Above the ground-water table, the total unit weight was taken as 125 pcf assuming an average water content of 7 percent.

The low strain shear moduli of the in situ sand and gravel, used in the estimation of building settlements, were determined using Equation 2.5.4-3.

$$G_{\max} = \frac{1,230 \ (2.97 \ -e)^2 \ (\overline{\sigma}_o) \ 0.5}{(1+e)}$$
(2.5.4-3)

where:

$$G_{max}$$
 = shear modulus (psi)
 σ_{o} = effective octahedral stress (psi)

Shear moduli determined from in situ seismic velocity measurements compared quite well with those computed using this relationship as shown on Figure 2.5.4-12. Section 2.5.4.10.2 provides an in-depth discussion of the determination of elastic properties of the in situ sands and gravels used in the estimation of building settlement.

The following tests, the results of which are presented in Appendix 2.5D, were performed on undisturbed block samples of the stiff silty clay lens which was encountered below the reactor containment excavation:

- 1. Atterberg limits and natural water contents,
- 2. Constant rate of strain (CRSC) and incrementally loaded (IC) consolidation tests,
- 3. Unconsolidated undrained (UU) triaxial compression tests, and
- 4. Consolidated undrained (CIU) triaxial compression tests.

Classification tests show that the silty clay has a liquid limit of 50, a plastic limit of 23, and a natural water content of 23 percent. The natural water content being equal to the plastic limit is an indication that the clay has been precompressed. The presence of small fissures with discoloration along their surfaces suggests that the precompression may have been due to dessication.

Consolidation tests show that the clay has been preloaded to a maximum past pressure ranging between 9.5 and 18 ksf. The estimated overburden pressure prior to the excavation for the containment foundation was approximately 7.5 ksf, indicating an overconsolidation ratio (OCR) of 1.3 to 2.4. The recompression ratio (RR) is approximately 0.02 and the compression ratio (CR) is approximately 0.12. The coefficient of consolidation (c_v) varies from approximately 5X10 to 1.8x10 cm/sec in the overconsolidated region and is approximately 2.5x10 cm/sec in the normally consolidated region. From the incremental consolidation test, the

average coefficient of secondary consolidation (c $_{\alpha}$) ranges between 5X10⁻⁴ and 2X10⁻³ in/in/log cycle of time.

The effective friction angle (ϕ) as determined from the *CIU* triaxial test was 25.7 degrees, assuming that the effective cohesion intercept was zero. The undrained shear strength measured in the UU tests is approximately 4.3 ksf.

In situ shear wave velocity measurements are discussed in Section 2.5.4.4. Field measurements of permeability are discussed in Section 2.5.4.6. Dynamic engineering properties of the soils underlying the site are discussed in Section 2.5.4.7.3.

2.5.4.3 Exploration

Site specific exploration activities at the BVPS site, for the purpose of evaluating subsurface conditions, consisted of drilling exploratory borings, installing piezometers, and performing geophysical surveys.

2.5.4.3.1 Exploratory Borings

A total of 298 exploratory borings were performed under the supervision of Stone & Webster Engineering Corporation (SWEC) at the BVPS site for the construction of SAPS, BVPS-1 and BVPS-2. The locations of the exploratory borings are shown on Figures 2.5.4-10 and 2.5.4-13. A list of borings, along with the dates that they were drilled and the locations of the boring logs for reference purposes, is provided in Table 2.5.4-1. Boring logs which have not been published in previous documents are included in Appendix 2.5B.

The primary functions of the exploratory borings were to establish the nature of the overburden soils and rock, to study the geology of the site area, and to obtain representative samples to develop engineering properties for design purposes.

A series of borings, PL-1 through PL-66, was performed by others in conjunction with the construction of a sludge pipeline system for the Bruce Mansfield Plant. Borings TH-1 through TH-13 were performed by others for the BVPS emergency response facility. The borings are shown on Figure 2.5.4-13 and are referenced in Table 2.5.4-1. Site subsurface profiles within the BVPS-2 area, based on data derived from the borings, are shown on Figures 2.5.4-2, 2.5.4-3, 2.5.4-4, 2.5.4-5, 2.5.4-6, 2.5.4-7, 2.5.4-8 and 2.5.4-9, Figures 2.5.4-51, 2.5.4-52, 2.5.4-53, 2.5.4-54 and 2.5.4-55; and on Figure 2.5.4-60.

Six piezometers were installed, at the locations shown on Figure 2.5.4-14, for the purpose of studying ground-water conditions at the site. Discussion of the data obtained is contained in Section 2.5.4.6.

2.5.4.3.2 Verification Borings

A total of 154 verification borings were performed to evaluate the results of the in situ densification program in the BVPS-2 area described in the Report on Soil Densification Program, (DLC 1976). A boring location plan is given on Figure 2.5.4-15, which also outlines the area densified. Each boring was evaluated on a sample-by-sample basis to verify that the desired degree of densification had been achieved. The logs of the verification borings can be found in Appendix I of the Report on Soil Densification Program, (DLC 1976).

Borings 501 through 562 were performed to verify the effectiveness of a vibroflotation densification program of the soil underlying the river water and service water system pipelines during the construction of BVPS-1. The limits of densification are shown on Figure 2.5.4-16, and Figure 2.5.4-54 presents the subsurface conditions beneath the 30inch service water system lines from the valve pit to the intake structure. The purpose of the program was to remove the potential for liquefaction of the underlying sands and gravels. This work is described in the Response to USNRC Questions 2.26 and 2.27 of the BVPS-2 PSAR, (DLC 1972e).

Borings 537T through 577T (Figure 2.5.4-13) were performed to verify the effectiveness of the Terra Probe densification program around the main intake structure. It was postulated that the nondensified soils, should they liquefy, could block the intake structure. The Terra Probe densification program was undertaken to prevent such an occurrence. This work is described in Section 2.5.4.12. The boring logs are included in Appendix 2.5B.

2.5.4.4 Geophysical Surveys

Geophysical surveys were conducted at the site by Weston Geophysical Engineers, Inc., to measure the in situ compression and shear wave velocities of the soil and rock. Appendix 2G of the BVPS-2 PSAR, (DLC 1972f) presents the results of measurements taken in 1968 in the area of the reactor containment for BVPS-1. Subsequent to the soil densification program at BVPS-2, additional crosshole seismic velocity measurements were made in the densified area. The results of this study are presented in the Report on Soil Densification Program, (DLC 1976). A summary of the information obtained from the geophysical surveys is given on Figure 2.5.4-17. A comparison of the measured in situ shear wave velocities on Figure 2.5.4-18 shows little difference before and after densification.

The field work involved with the measurement of the postdensification, in-situ, seismic velocities was performed between June 9 and June 22, 1977. From piezometer data presented in Appendix 2.5A, the ground-water levels within the terrace sands and gravels of the main plant area during this period averaged approximately el 665.7 feet. Consequently, the water level of el 652 presented by Weston Geophysical Engineers appears to be incorrect. The soil layer between approximately el 652 and el 667 with a compressive wave velocity of 3,000 ft per second was mostly below the ground-water table at the time of the seismic survey.

Since water is relatively incompressible, as compared to the soil skeleton, measurements of compressive (P) wave velocities below the ground-water table are more representative of the water than the soil. The compressive wave velocity of water is about 5,000 ft per second; therefore, the reported compressive wave velocity of 3,000 ft per second between el 652 and el 667 is anomolous.

This value of compressive wave velocity from the seismic survey was not used in analyzing the behavior of safety-related plant structures, systems, or components.

2.5.4.5 Excavations and Backfill

A comprehensive onsite quality control program was instituted at BVPS-2 to ensure compliance with excavation, material, and compaction requirements as specified by SWEC. This program was under the control of Duquesne Light Company Site Quality Control (DLC-SQC).

Field inspection and testing, as required, were performed by Dick Corporation Field Quality Control (DC-FQC), which reported directly to DLC-SQC.

2.5.4.5.1 Excavation

Permanent Seismic Category I excavations are not present at the site. All excavations are temporary and are related to the construction of the plant structures. They will be backfilled as required prior to plant operation. The areal extent of excavations in the plant area is shown on Figure 2.5.4-19. Profiles through the plant area are given on Figures 2.5.4-2, 2.5.4-3, 2.5.4-4, 2.5.4-5, 2.5.4-6, 2.5.4-7, 2.5.4-8 and 2.5.4-9.

To ensure the suitability of the excavated foundation levels for Category I structures, buried piping, and duct lines, the DC-FQC inspector verified the following (DLC 1979, SWEC 1978):

- 1. Excavated areas were within the limits shown on the drawings,
- All excavated or bedding areas or fill areas, as applicable, were graded to within 0.2 foot of the grades shown on the drawings,
- Tests were performed on granular excavated material and/or founding elevation, as applicable, as required in Table 2.5.4-2,
- 4. Any soft spots at the bottom of excavations were removed and backfilled at the direction of the Geotechnical Engineer, and
- 5. The founding elevations or prepared surfaces, as applicable, were approved by the Geotechnical Engineer.

The excavation for the reactor containment was made within a steel sheetpile cofferdam driven to el 671 feet. Upon completion of the excavation to el 679 feet, a foundation documentation program was conducted which consisted of the following:

- 1. Establishing a 25-foot square grid over the floor of the containment excavation,
- 2. Photographing the floor of the containment excavation,
- 3. Performing in-place density tests at each grid intersection, and
- 4. Obtaining a bag sample of the founding soil at each grid intersection for classification.

Grain size analyses performed on each bag sample and the results of the in-place density tests are presented in Appendix 2.5D.

The foundation documentation program revealed the presence of a lens of stiff silty clay along the northern perimeter of the containment excavation at el 679 feet; the presence of which was not encountered during the original subsurface investigation. The lens extends eastward beneath roughly the northern half of the safeguards area and the RWST. The areal extent of the clay lens is discussed further in Section 2.5.4.7.1. Soil profiles beneath the safeguards area and the RWST are shown on Figures 2.5.4-8 and 2.5.4-9. Laboratory tests performed on undisturbed block samples recovered from the containment excavation are presented in Appendix 2.5D and are summarized in Section 2.5.4.2.

The clay was removed from within the containment excavation and replaced with compacted structural fill; it was not removed from beneath the safeguards area and the RWST. Estimates of the settlements of the safeguards area and the RWST were found to be within tolerable limits (Figure 2.5.4-20) and the stiff clay was not considered to be a concern to the stability of the structures insofar as a bearing capacity failure was concerned due to the overlying thickness of compacted structural fill. The area beneath the northern portion of the safeguards area and the RWST was excavated to el 690 feet during the soil densification program (Section 2.5.4.12) and then backfilled with compacted structural fill.

Portions of the intermediate terrace and present floodplain (Figure 2.5.4-1) have been overlain by uncontrolled fill and nonstructural fill placed during the construction of SAPS and BVPS-1. The limit of the uncontrolled fill in the plant area was determined from a comparison of the original ground topography that existed prior to the construction of SAPS with the topography after the completion of BVPS-1. The excavation to el 690 feet north and east of the containment shown on Figure 2.5.4-19 was conducted to remove this material. The excavated area was then backfilled with compacted select granular fill.

Measures to control ground-water levels during excavation were not required. The ground-water level reflects the Ohio River water level which has a normal pool elevation of 665 feet. With the exception of a local area within the containment cofferdam, the bottom of all excavations were well above el 665 feet.

2.5.4.5.2 Backfill

<u>Specifications</u>

Structural or select granular fill for use beneath and adjacent to Category I structures consisted of well-graded sand and gravel, which conformed to the following grain size requirements:

<u>Sieve size</u>	Percent passing <u>by dry weight</u>
6 (inches)	100
No. 200	0-15 (nonplastic fines)

Figure 2.5.4-21 shows the upper and lower gradation limits Of 115 grain size analyses on material used for structural fill. Compaction tests performed on these samples according to ASTM 1557, Method D, indicated a mean maximum dry density of 136.9 pcf with a mean optimum water content of 7 percent.

The material was placed in loose lifts of 6 to 12 inches and compacted to a minimum of 95 percent of the maximum dry unit weight obtained from compaction tests performed in accordance with ASTM D1557, Method D, with a minimum required in-place density of 130 pcf.

Material testing requirements were as given in Table 2.5.4-2 (SWEC 1978).

Granular borrow material meeting the gradation requirements for structural/select granular fill was obtained from the suppliers listed in Table 2.5.4-6. Also shown are quantities of backfill provided by each supplier.

In situ soils removed from on-site excavations were not used as structural fill beneath or around Category I structures.

Soil Properties

The dry unit weight of compacted structural fill was taken as 130 pcf, corresponding to 95 percent of the mean maximum dry density from 115 moisture density tests.

The specific gravity was taken as 2.65.

The void ratio was computed to be 0.27.

The saturated unit weight below the ground-water table was taken as 144 pcf from the equation:

$$\gamma_T = \frac{G + Se}{1 + e} \bullet \gamma_W \tag{2.5.4-4}$$

where:

γ_T	= total unit weight (pcf)
G	= specific gravity
S	= degree of saturation, decimal (100%)
е	= void ratio
γ_w	= unit weight of water = 62.4 pcf

Above the ground-water table, the total unit weight was taken as 136 pcf assuming an average water content of 5 percent.

The angle of internal friction of compacted structural fill was conservatively assumed to be 36 degrees.

Low strain shear moduli were estimated using Equation 2.5.4-5 as follows (Hardin and Drenewich 1972):

$$G = \frac{1,230 \ (2.97 - e)^2 \ (\overline{\sigma}_o)^{0.5}}{1 + e}$$
(2.5.4-5)

where:

G = shear modulus (psi) σ_{\circ} = effective octahedral stress (psi) (2.5.4-5)

The vertical coefficient of subgrade reaction for buried pipe was computed according to the following equation (Vesie 1961, 1961a):

$$k_{\nu} = \frac{0.65^{12}}{D_{o}} \sqrt{\frac{E_{s} D_{o}^{4}}{E_{p} I_{p}}} \left(\frac{E_{s}}{1 - \nu^{2}}\right)$$

(2.5.4-5a)

where :

 $k_v = vertical coefficient of subgrade reaction (lb/in³)$ $D_o = outside diameter of pipe (in)$ $E_s = Young's modulus of soil (lb/in²)$ $E_p = Young's modulus of pipe (lb/in²)$ $I_p = moment of inertia of pipe section (in⁴)$ v = Poisson's ratio of soil

An average, low strain value of shear modulus, G, was estimated using equation 2.5.4-5 for two ranges of pipe embedment depth, H_e :

Using these values of shear modulus, Young's modulus, with a reduction to account for strain, was estimated as:

$$E_s = 2 \ (1+\nu) \ \frac{G}{3} \tag{2.5.4-5b}$$

Vertical coefficient of subgrade reaction is shown on Figure 2.5.4-62 as a function of depth of embedment and pipe diameter.

The horizontal coefficient of subgrade reaction for buried pipe was determined according to the empirical procedure described by Audibert and Nyman (1977). An analytical procedure was developed to determine the horizontal load-displacement (p-y) curve for any size pipe embedded at any given depth. Considering the horizontal coefficient of subgrade reaction as the amount of soil pressure reaction generated by a given amount of horizontal displacement (that is, as a secant to the p-y curve), the coefficient of horizontal subgrade reaction can be expressed by:

$$k_h = \frac{p}{y} = \frac{1}{A' + B'y}$$
(2.5.4-5c)

where:

$$\begin{array}{ll} k_{\rm h} & = \mbox{ horizontal coefficient of subgrade reaction (lb/in3)} \\ p & = \mbox{ pressure (lb/in2)} \\ y & = \mbox{ displacement (in)} \\ A' & = \frac{0.145 \ yu}{q_u} & (in^3 / lb) \\ B' & = \frac{0.855}{q_u} & (in^2 / lb) \\ y_{\rm u} & = \mbox{ ultimate displacement (in)} \\ q_{\rm u} & = \mbox{ ultimate soil resistance (lb/in2)} \end{array}$$

Considering the buried pipe as horizontal footing, the ultimate soil resistance, \mathbf{q}_{u} is computed as:

$$q_u = \gamma Z N_q$$

where:

The bearing capacity factor is given on Figure 2.5.4-63. The ultimate displacement, y_u , was evaluated from Figure 2.5.4-63. The iterative procedure used to calculate displacements assumes an initial value of displacement in order to compute an initial value of k_h . Then, using this initial value of k_h , an actual displacement is computed. This procedure continues until the iterative values converge at a final displacement.

2.5.4.6 Ground-water Conditions

Regional and local aquifer characteristics are described in detail in Section 2.4.13.

At the bottom of the excavation for the reactor containment foundation, four temporary observation wells were installed at the locations shown on Figure 2.5.4-22. These observation wells were abandoned when the reactor containment foundation mat was placed. Ground-water level readings as well as the Ohio River elevation were recorded daily from March 19, 1976 until May 21, 1976. The data are shown on Figures 2.5.4-23, 2.5.4-24, 2.5.4-25 and 2.5.4-26 along with | installation details of the observation wells. As can be seen from the data, there is essentially no hydrodynamic time lag between the elevation of the Ohio River and the ground-water level in the observation wells.

Falling head permeability tests were conducted in three of the wells in order to estimate the coefficient of permeability. The results were:

(2.5.4-5d)

Well <u>No.</u>	Coefficient of permeability <u>(x10⁻³ cm/see)</u>
1	1.3-3.9
3	0.9-1.7

In 1977, as part of the settlement monitoring program that is described in Section 2.5.4.13, six piezometers were installed at the locations shown on Figure 2.5.4-14. A typical piezometer detail is shown on Figure 2.5.4-27 and installation data are given in Appendix 2.5A. Tip elevations range between el 646 feet and 651 feet, and all of the piezometers are located within the in situ sand and gravel.

Piezometer data and Ohio River elevation data were recorded during construction on a weekly basis since mid-1977 and are included in Appendix 2.5A. With the exception of one period during February 1979, the ground-water levels recorded in the piezometers show very good correlation with the Ohio River elevations. During February 1979, the river rose to el 681 feet and the piezometer data indicate an apparent time lag. However, the piezometers were only read weekly during this period and in the interim between readings the water level in the piezometers may have risen higher thereby reducing the apparent elevation difference between the ground-water level and the Ohio River elevation indicated by the data.

For the purpose of design, the ground-water level in the plant site area can be expected to reflect the various stages of the Ohio River as discussed in Section 2.4 and repeated as follows:

<u>River stage</u>	Elevation <u>(feet)</u>
Normal water level	665
Ordinary high water	675
Twenty-five year flood	690
Standard project flood	705
Probable maximum flood	730

The design basis for substructure hydrostatic loading is discussed in Section 2.4.13.5.

As stated in Section 2.5.4.1, dewatering for the control of ground water in the plant area during excavation was not required.

The exterior surfaces of the reactor containment shell and foundation mat are protected from water seepage during flood stages caused by the standard project flood and the probable maximum flood by a continuous waterproof membrane. In the event that leakage should occur through the membrane, a supplementary water relief system is provided in the containment to prevent the buildup of water pressure under and behind the steel liner. This system is described in Section 3.8.1.1.1.

2.5.4.7 Response of Soil and Rock to Dynamic Loading

This section describes the dynamic engineering properties of the soils underlying the site and the method used to determine relative displacement between two structures during a seismic event.

Soil structure interaction analyses and the response of buried pipe to dynamic loading is discussed in Sections 3.7B.2 and 3.7B.3, respectively. A discussion of the liquefaction and dynamic settlement potential of the soils at the site is given in Section 2.5.4.8.

2.5.4.7.1 Subsurface Conditions

Subsurface exploration activities conducted at the site are described in Section 2.5.4.3. Soil profiles through the main plant area with the major structures superimposed are shown on Figures 2.5.4-2, 2.5.4-3, 2.5.4-4, 2.5.4-5, 2.5.4-6, 2.5.4-7, 2.5.4-8 and 2.5.4-9. The geology of the site is described in Sections 2.5.1.2 and 2.5.4.1. Soil profiles under Category I pipelines are shown on Figures 2.5.4-51, 2.5.4-52, 2.5.4-53, 2.5.4-54 and 2.5.4-55.

The major structures of BVPS-2 are located on the highest of three alluvial terraces along the south side of the Ohio River. These terraces are described in more detail in Section 2.5.4.1. Foundation information for Category I structures is given in Table 3.7B-2. Category I structures are founded on compacted select granular fill overlying dense in situ granular soil which extends to rock or directly on the in situ granular soil with one exception. The soil in the vicinity of the safeguards area was excavated to el 690 feet BVPS-2 UFSAR

as shown on Figure 2.5.4-8 to remove uncontrolled fill placed during the construction of SAPS and BVPS-1. Beneath the northern portions of the safeguards area and the RWST, underlying the select granular fill which was placed subsequent to the excavation, is a layer of stiff silty clay with a top surface at approximately el 688 feet. The layer is about 20 feet thick at the northern edge of the safeguards area and about 10 feet thick at the northern edge of the RWST. The layer thins to the south and is no longer present at about the east-west centerline of the safeguards area (Figures 2.5.4-8 and 2.5.4-9).

A zone of loose, in situ granular material located from approximately el 640 feet to 660 feet was discovered in the BVPS-2 plant area during the excavation for the reactor containment. This zone was successfully densified using the pressure-injected footing technique.

Site investigations did not detect evidence of features or conditions indicative of disturbance during prior earthquakes.

2.5.4.7.2 In Situ Seismic Velocity Measurements

The results of geophysical surveys conducted at the site are presented in Section 2.5.4.4.

2.5.4.7.3 Dynamic Soil Properties

Measurements were made of the in-situ shear wave velocity of the terrace sands and gravels by Weston Geophysical (DLC 1976). The results are tabulated on Figure 2.5.4-17 and plotted on Figure 2.5.4-18. The 1968 survey was performed in the vicinity of the BVPS-1 reactor containment building and the 1977 survey was performed after the soil densification program in the vicinity of the BVPS-2 fuel building. The terrace soil conditions at the two locations are very similar. The shear wave velocities in the densified zone are similar to those in the vicinity of the BVPS-1 reactor containment. This suggests that the soil densification program increased the density and modulus of the formerly loose sands and gravels such that they are similar to the density and modulus of the soils outside the densified area.

A finite element, soil-structure interaction analysis was performed to evaluate the effect of the densified zone on the response of the reactor containment structure. The densified zone is present beneath about one-half of the containment mat. A benchmark analysis was performed without the densified layer present and was compared with an analysis of an unsymmetric case which included the densified layer. In the latter analysis, conditions were exaggerated because the shear modulus value assigned to the densified zone was very much greater than that computed from the shear wave velocity measurements. The results indicated very little difference in the computed response of the containment structure for the two cases and it was therefore concluded that assuming properties within
the densified zone consistent with those outside the densified zone was justified.

As discussed in Section 2.5.4.2, and shown on Figure 2.5.4-12, the low strain shear moduli of the in-situ sands and gravels determined from the measured shear wave velocities outside the densified area compare quite well with those determined using the empirical relationship of Hardin and Drenevich (1972).

Shear strains generated by earthquake motions cause a reduction in the low strain value of shear modulus. Using test data presented by Seed (et al 1975), the low strain values of shear modulus were reduced for anticipated strain levels during the safe shutdown earthquake (SSE) as shown on Figure 2.5.4-12.

The low strain shear modulus of compacted structural fill was determined using the empirical relationship of Hardin and Drenevich (1972) as given in Section 2.5.4.5.

Cyclic triaxial tests to investigate the susceptibility of compacted structural fill to liquefaction were not performed. As stated in Section 2.5.4.5.2, structural fill was placed and compacted to 95 percent of the maximum dry density indicated by ASTM 1557 Method D, with a minimum required in-place dry density of 130 pcf, and this, coupled with the gradation of the material, was considered sufficient to preclude liquefaction. Liquefaction analyses are performed assuming, as a minimum, a groundwater level coincident with the 25year flood, which for BVPS is at el 690 feet. With the exception of an area beneath the northern portion of the reactor containment foundation, as shown on Figure 2.5.4-19, all structural fill within the main plant area was placed above el 690 feet.

Attempts to obtain undisturbed samples of the in situ sands and gravels suitable for dynamic triaxial testing were unsuccessful. The

resistance to liquefaction of the in situ sands and gravels at the site was investigated by two methods:

- 1. Based on dynamic triaxial tests on sands susceptible to liquefaction and
- 2. Based on the observed behavior of sand deposits in previous earthquakes (DLC 1976).

The results of dynamic triaxial tests upon Sacramento River sand, considered to be extremely susceptible to liquefaction, are presented on Figure 2.5.4-28, based on the response to USNRC Question 2.26 and 2.27 of the BVPS-2 FSAR. The figure shows the relationship between shearing stresses, expressed as a ratio of shear stress to effective stress, to the number of cycles necessary to cause initial liquefaction for this sand at several relative densities. It was used to evaluate the liquefaction potential of the soils within the main plant area as described in Section 2.6.5.2 of the BVPS-2 PSAR. This approach was conservative since the Sacramento River sand was considered especially susceptible to liquefaction in comparison to the sands and gravels at the site.

After the discovery of the loose zone in the main plant area and its subsequent densification, a liquefaction analysis was performed for soils within the densified zone (DLC 1976). The shear stress required to cause liquefaction of the in situ sands and gravels was evaluated using Figure 2.5.4-29. This figure presents a lower bound envelope for sites where liquefaction has occurred during earthquakes of Richter Magnitude 5.5 or less, correlated with corrected standard penetration resistance, N_1 , of the sand deposit. This figure was used to evaluate the resistance to liquefaction of the soils in the vicinity of the intake structure as well. Further discussion is presented in Section 2.5.4.8.

The maximum Rayleigh wave velocity used in the analysis of buried pipe was determined to be 3,000 ft/sec, using the procedure described below.

Ewing et.al. (1957) presented data, reproduced on Figure 2.5.4-31, which showed that Rayleigh wave velocity in a layered system was a complicated function of the depth of soil, the shear wave velocity of soil and rock, and the frequency wave length of the Rayleigh wave.

Using Figure 2.5.4-31, for $C_2/C_1=4.5$, the variation of Rayleigh wave velocity with frequency for the in situ soil conditions in the main plant area was determined and is shown on Figure 2.5.4-64. Rayleigh wave velocity is seen to vary widely depending on frequency. Because an earthquake is likely to produce Rayleigh waves of many frequencies, the selection of a control value of Rayleigh wave velocity was based upon a consideration of the predominant frequency likely to be produced by an earthquake occurring near the site. BVPS-2 UFSAR

The peaks of Fourier spectra for earthquake time histories represent frequencies at which large amounts of energy are released by the earthquake. Housner (1970) compared Fourier spectra with velocity response spectra and found that the peaks occurred at about the same frequencies. Accordingly, a predominant frequency of 2-3 Hz was determined from response spectra presented in SWEC (1984). These response spectra were computed for real earthquake time histories, with magnitudes corresponding to the BVPS-2 SSE, that were amplified through the BVPS-2 soil profile. From Figure 2.5.4-64, a frequency of 2-3 Hz corresponds to a Rayleigh wave velocity of about 3000 ft/sec.

2.5.4.7.4 Relative Displacements

Methods used to evaluate the relative displacement of the closely spaced main plant structures during a seismic event for input to pipe stress analyses are discussed in this section. One method considers the theoretical behavior of surface waves and the effect of site layering on the wave velocities. It uses a simple and conservative model which assumes that the wave motion is propagated horizontally without any change in wave form and with no effect of structure ridigity or other interference. The maximum displacement between two points is enveloped by the first term of a Taylor series expansion of the displacement. The second method, used in only two cases, uses an earthquake acceleration time history to determine a relative displacement time history from which the maximum value is determined. The mass and rigidity of the structures involved and of adjacent structures are not considered in either method. The motion of the ground surface, upon which is founded a massless structure, is estimated. The two methods estimate the magnitude of the relative displacement, but not the direction of movement of the individual structures involved. The earthquake time history method results in smaller estimates of relative displacement, but it represents a more realistic approach, since it is based upon actual recorded ground motions.

SIMPLE MODEL

It is assumed that relative displacement results from the horizontal propagation of seismic waves with little or no change in wave form. It is further assumed that the maximum particle motions produced by each wave occur simultaneously, and that the foundations behave as rigid bodies.

For soil sites such as BVPS, relative displacements are caused by Rayleigh waves and Love waves. The particle motion for the Rayleigh wave occurs in the vertical plane and is elliptical and retrograde with respect to the direction of propagation. By their nature, Rayleigh waves cause horizontal push-pull (R_x) and vertical (R_y) displacements. The particle motion of Love waves is transverse to the direction of propagation and as a result, they are the cause of translational (R_y) displacements.

For wave propagation parallel to the axis between structures under consideration, the horizontal push-pull and translational displacements between two points A, and B, as shown on Figure 2.5.4-30 are given as follows (Christian 1976):

$$R_x = V_m \frac{b}{C_r}$$
$$R_y = V_m \frac{b}{C_L}$$

where :

$R_{\rm X}$	= push-pull displacement along axis between
	two points
Vm	= peak velocity of design earthquake
b	= centroidal distance between structures
Cr	= Rayleigh wave velocity
Ry	= horizontal displacement perpendicular to the axis between two points
Ст	= Love wave velocity

The peak velocity of the design earthquake was determined from the empirical formulation that a V_m of 48 in/sec corresponds to a peak acceleration, a_m of lg; defined by Christian (1976) as:

 $V_m = 48a_m$

Generally, the vertical ground acceleration is taken as $2/3\ \text{of}\ a_h$ and therefore:

$$R_z = 2/3 R_x$$

where:

For oblique waves, the maximum relative displacement occurs when the direction of propagation is at an angle of 45 degrees to the line between two points. The relative displacement between points A and B (Figure 2.5.4-30) is determined by resolving the relative displacement of points A' and B' into components parallel and perpendicular to line AB. Both the Rayleigh wave and the Love wave contribute to the relative displacement between A and B.

For the Rayleigh wave:

$$R_{xr} = R_{yr} = \left(48a_m \frac{b}{C_r}\right) \sin^2 45^\circ$$

(2.5.4-6)

 R_z = vertical displacement

For the Love wave:

$$R_{xL} = R_{yL} = \left(48a_m \frac{b}{C_L}\right) \sin^2 45^\circ$$
(2.5.4-7)

Symbols are as defined previously.

The vertical displacement ${\tt R}_{\rm z}$ is a result of the Rayleigh wave only; therefore:

$$R_z = 2/3 \left(48a_m \frac{b}{C_r} \right) \sin^2 45^{\circ}$$
(2.5.4-8)

Soil sites underlain by stiffer materials such as denser soil or rock will have Rayleigh wave and Love wave velocities that are affected by the wave length, soil and rock characteristics, and depth of overburden to the stiffer layer. The relationship between the parameters already mentioned used to evaluate the Rayleigh wave velocity is given on Figure 2.5.4-31 (Ewing et al 1957). The wavelength was taken as twice the distance between points under consideration. The Love wave velocity was determined from the following expression (Bullen 1963):

$$G_{2}\left(1-\frac{C_{L}}{C_{2}^{2}}\right)^{1/2} - G_{1}\left(\frac{C_{L}^{2}}{C_{1}^{2}}-1\right) \tan\left[\frac{2\pi H}{\lambda}\left(\frac{C_{L}^{2}}{C_{1}^{2}}-1\right)^{1/2}\right] = 0$$
(2.5.4-9)

where:

G_2	=	shear modulus of lower denser layer (rock)
C ₂	=	shear wave velocity of lower denser layer
G_1	=	shear modulus of upper layer (in situ sand and gravel)
C ₁ H	=	shear wave velocity of upper layer depth of upper layer
λ	=	<pre>wavelength = 2b for parallel waves;</pre>
	=	2bsin 45° for oblique waves

There is a unique solution for the Rayleigh and Love wave velocities for each value of b.

The shear moduli, G_1 , and G_2 , of the in situ sand and underlying rock were computed from the measured in situ shear wave velocities (DLC 1976). Average values of 1,250 fps and 5,000 fps were chosen for the soil and rock, respectively.

The relative displacements for parallel waves and for oblique waves determined by this approach for the SSE are shown on Figure 2.5.4-30. For the operating basis earthquake (OBE), the values are one-half those shown for the SSE.

Earthquake Time History Method

The relative displacement of the centroids of two building foundations are computed using the computer program CORD2B, a shortened acronym for "Calculation of Relative Displacements between Two Buildings during an Earthquake" (SWEC 1982). The program does not consider the mass and rigidity of the structures; but rather, computes the displacements of points under massless, rigid buildings caused by the passing of an earthquake wave. It is assumed that each foundation is subjected to the same time history; i.e., that the wave shape is not changed while passing between the centroids. CORD2B filters the acceleration time history, averaging the effect of the accelerations to account for the time required for the earthquake to pass beneath an individual building. This averaging accounts for the fact that different parts of the foundation are subjected to different accelerations as the wave passes, and, in effect, the acceleration time history for each structure is integrated twice to determine a displacement time history. The two displacement time histories are then shifted by the time required for the wave to pass between the centroids, namely:

t = b/c

where:

```
t = time lag
```

- b = centroid to centroid distance
- c = Rayleigh wave velocity (Figure 2.5.4-31)

Subtracting the time shifted displacement time histories results in a relative displacement time history and identification of the peak relative displacement.

The push-pull and transverse relative displacement is computed as the mean peak horizontal displacement determined from the eighteen horizontal ground surface acceleration time histories listed in Table

BVPS-2 UFSAR

2.5.4-9. These time histories were recorded at soil sites for which soil properties and profiles are matched as closely as possible by BVPS-2 (SWEC 1985). They are scaled to the SSE ground surface acceleration of 0.125 g by CORD2B before computing displacements. The vertical relative displacement is assumed to be two-thirds of the horizontal.

The procedure is used to compute the relative motion between the auxiliary building and the main steam and cable vault and between the safeguards area and the main steam and cable vault. The results are summarized in Table 2.5.4-10.

2.5.4.8 Liquefaction Potential and Dynamic Settlement

2.5.4.8.1 Liquefaction Potential

Main Plant Area

A zone of loose, potentially liquefiable, granular soil was identified in the BVPS-2 main plant area during the excavation for the reactor containment foundation. The loose zone was successfully densified and a liquefaction analysis of soils within the densified zone indicated adequate factors of safety (DLC 1976). For further discussion of the densification program, refer to Section 2.5.4.12.

The extent of the loose zone was determined from the exploratory borings as a zone within a given boring containing a significant number of samples with N_1 values less than 10 determined using the data of Gibbs and Holtz (1957). N_1 is the measured standard penetration test resistance, N, corrected to an overburden pressure of 1 ton/ft 2 . This criterion was determined from an analysis of liquefaction potential. The analysis was performed with the groundwater table assumed at el 705 feet. A constant value of N_1 was assumed within the free field soil profile of the main plant area and at a number of depths. Δn allowable shear stress was determined from Figure 2.5.4-29 for this (Further discussion of Figure 2.5.4-29 is assumed value of N_1 . provided in the following section concerning the main intake structure.) At the same depths, the applied shear stress was determined from Figure 2.5.4-39 and the factor of safety was determined as the ratio of the allowable shear stress to the applied It was determined that an N_1 value of 5 would result in shear stress. a safety factor against liquefaction of 1.0. As an N_1 of 10 resulted in a factor of safety of about 2, it was conservative to define the loose zone in terms of N_1 values less than 10.

The applied shear stresses given on Figure 2.5.4-39 are based on BVPS-2 PSAR Figure 2.6-6, which shows free field shear stresses for the N69W component of the 1952 Taft accelerogram scaled to an estimate of the SSE bedrock acceleration, and then amplified through a BVPS-2 soil profile by a mass-spring-dashpot computer model (DLC, 1972i). The average stress for the 10 largest peaks has been increased slightly to account for differences between current

building loads and soil unit weights and those used in the original analysis described in DLC (1972i). The peak ground surface acceleration associated with PSAR Figure 2.6-6, and, therefore, Figure 2.5.4-39 is 0.098g. This is lower than the SSE ground surface acceleration of 0.125g, but the shear stresses given by Figure 2.5.4-39 are shown on Figure 2.5.4-39a to represent a very conservative estimate of shear stresses at the site during the SSE.

Figure 2.5.4-39a presents a comparison of the stresses shown on Figure 2.5.4-39 with those computed as an average of those from 28 rock outcrop time histories amplified through a BVPS-2 free field soil model using the computer program SHAKE (Schnabel, et al., 1972). The analysis technique is similar to the soil response analysis described by SWEC (1985), which used the same 28 earthquake records to estimate a BVPS-2 site dependent response spectrum.

The rock outcrop time history is input at the bedrock level of a BVPS-2 free field soil model and scaled to the SSE magnitude, using a procedure described by SWEC (1985). The peak acceleration contained in the record is not considered; magnitude is the design parameter. The scaled time history is then amplified through the soil model by SHAKE to compute peak shear stresses as a function of elevation. An average stress is taken as 65 percent of the peak value as recommended by Seed and Idriss (1971). An average of the responses of the 28 time histories is computed and is shown on Figure 2.5.4-39a. The groundwater table is taken at el 690 ft, the level of the 25 year flood. Shear stress from this analysis technique is seen to be very much lower than the original, but similar analysis based on only one earthquake record. Therefore, use of Figure 2.5.4-39 to evaluate shear stresses in the soil generated by the SSE is very conservative.

Using the criterion of a minimum N_1 of 10, the borings in the main plant area were examined and a thickness contour map of the zone requiring densification was prepared. This map is presented in Appendix Figure B-10 of the Report on the Soil Densification Program (DLC 1976). The extent of the area densified by the pressure-injected footing technique is shown on Figure 2.5.4-15. To the north, east, and west of the main plant area, the densified zone was extended beyond the foundation limits of the plant structures in order to provide continued support to the foundations in the event of the liquefaction of the adjacent, nondensified soil.

 N_1 values from borings under the main plant structures outside of the densified zone are presented on Figure 2.5.4-59. From a total of 541 samples, 15 show N values less than 10 and none are less than 5. Figure 2.5.4-11 shows relative density versus standard penetration test N values for the same boring as those on Figure 2.5.4-59. The mean relative density for the sand and gravel is 77.3 percent and the means relative density less one standard deviation is 62.9 percent.

Main Intake Structure

The main intake structure is located as shown on Figure 2.5.4-32 and is common to both BVPS-1 and BVPS-2. The structure is directly adjacent to the Ohio River with founding elevation varying between el 634 feet 6 inches and 640 feet 6 inches. A sheetpile cofferdam driven to rock was used to facilitate the construction of the 85-foot by 88foot structure. Extending along the river to the east and west of the intake structure are two rows of sheetpile walls that are tied together (Figure 2.5.4-32). The area directly north of the structure was dredged to el 645 feet, with an average side slope of approximately 3.5:1. A simplified north-south section through the intake structure is shown on Figure 2.5.4-66.

The onshore and offshore areas east and west of the intake structure, and the area to the south beneath the BVPS-1 river water lines and the BVPS-2 service water lines were densified. The limits of densification are shown on Figure 2.5.4-16. The purpose of densifying the sands around the intake structure was to prevent the soils from liquefying during the SSE and blocking the intake channel. A detailed discussion of the densification program is presented in Section 2.5.4.12. The areas north of and below the intake structure were not densified. No adverse effects to the structure, slopes, or wingwalls are anticipated. Stability of soils at the intake structure is considered in three parts: a. liquefaction analysis of soils around the intake structure, b. liquefaction analysis of soils below the intake structure and, c. stability of intake channel slopes.

a. Soils Around Intake Structure

Liquefaction analysis of the soils around the intake structure considers three separate areas: 1. the offshore densified area, 2. the onshore densified area, and 3. the intake channel. The factor of safety against liquefaction is taken as the ratio of the shear stress required to cause liquefaction of the soil and the shear stress induced by the design earthquake. The minimum factor of safety is 1.1. Ohio River elevations are assumed coincident with normal river conditions at el 665 ft, and with the 25-year flood at el 690 ft.

The shear stress required to cause liquefaction is evaluated using the relation developed by Seed et al (1975a) that is shown on Figure 2.5.4-29. This curve is described as a lower bound envelope for sites where liquefaction has occurred during earthquakes with Richter magnitudes between 5 and 6, correlated with the corrected penetration resistance, N_1 , of the sand deposit involved. N_1 is the measured standard penetration test (SPT) resistance, N, corrected to an overburden pressure of 1 ton/ft².

The values of N_1 are computed using the computer program RELDEN (SWEC 1979). The measured SPT resistance, N, is corrected to a maximum N_1 value of 42 by RELDEN, although the actual value could be higher. Since the resistance to liquefaction is related to N_1 , the computed factor of safety against liquefaction will be conservative for those samples with an N_1 value greater than 42.

The applied shear stress is calculated from the equation presented by Seed (1976):

$$\tau_{app} = 0.65 \ x \ \frac{\sigma_{vo}}{g} \ x \ a_{\max} \ x \ r_d \tag{2.5.4-10}$$

where :

- τ_{app} = applied shear stress
- σ_{vo} = total overburden pressure on sand layer under consideration
 - g = acceleration of gravity
- a_{max} = maximum acceleration at the ground surface, 0.125g for SSE
- rd = a stress reduction factor varying from a value of 1 at the ground surface to a value of 0.9 at depth of about 30 feet (Seed and Idriss 1971)

Although this equation was developed to predict applied shear stress for a horizontal ground surface, it is used for both horizontal and sloping ground conditions in this analysis. Since the intake channel side slopes are graded to a shallow slope of about 3.5:1, the use of this equation gives a reasonable approximation of the applied shear stresses.

Liquefaction analyses are performed on a sample-by-sample basis, using soil data obtained from the borings shown on Figures 2.5.4-13 and 2.5.4-32. Borings performed after the densification program was completed are used in the analysis of the onshore and offshore densified zones east and west of the intake structure. Borings were not drilled in the channel area immediately north of the structure and since this area was not densified, borings drilled offshore prior to densification are taken as representative of soil conditions in the intake channel.

In the analysis of the onshore areas, the entire soil profile is assumed to be saturated for cases with river level at or above el 665 feet. Therefore, changes in water level from normal to flood conditions do not change the results. Offshore, the total stress at a given sample is computed as the product of saturated weight of the soil and the depth below the river bottom to the soil sample. The total stress is computed neglecting the weight of the overlying water, since the water cannot transmit shear stress. As a result, the computed total stress and therefore the applied shear stress are unaffected by fluctuating river levels. Since the effective stress used to evaluate the allowable shear stress (Figure 2.5.4-29) is likewise not affected, the factor of safety against liquefaction offshore does not change with changes in the Ohio River elevation.

Safety factors for the onshore densified areas and the offshore densified areas are shown on Figures 2.5.4-33 and 2.5.4-35, respectively. The onshore densified areas south of the riverward sheetpile walls have satisfactory factors of safety against liquefaction with all alues at or above 1.6. The offshore densified

soils are not susceptible to liquefaction as shown by the preponderance of samples having factors of safety greater than 1.1. Two samples at a depth of less than 5 feet in two different borings have factors of safety less than 1.1, but this is neither significant nor unusual due to low confining stress at shallow depths.

Safety factors for the undensified intake channel area north of the intake structure are shown on Figure 2.5.4-34. Ten samples between el 645 ft and el 634 ft had factors of safety less than 1.1. Most of these samples occur within the top 5 to 10 feet of the soil profile. One sample at approximately el 623 ft was unsatisfactory. A similar analysis performed for samples above el 645 ft along the intake channel slopes outside the densified area shows that the upper 10 feet of soil is loose and may liquefy. Therefore, in the dynamic slope stability analysis of the intake channel described in Section c., the upper 10 feet along the slopes outside of the densified zone and below the dredge line in front of the intake structure were assumed to be liquefied at the end of the seismic event.

b. Analysis of Soils Beneath Intake Structure

Examination of the soil conditions below the intake structure indicates the possible presence of some low blow count granular soil with an average N_1 of about 7 blows/ft. The liquefaction potential of this granular soil and the underlying more dense soil are evaluated using a different approach than that used for the soils adjacent to the structure. The results indicate that the factor of safety against liquefaction within any low blow count material beneath the intake structure is between 1.2 and 1.6, which is acceptable.

A cross section through the intake structure showing soil conditions that existed prior to the Terra Probe densification program is given on Figure 2.5.4-67. It was developed using data from borings performed in 1954 and 1974 prior to the densification program.

Examination of the corrected blow counts indicates that low blow count materials could have existed below and on either side of the intake structure location between about el 635 ft and 640 ft prior to construction and the subsequent insitu densification by Terra Probe and vibroflotation adjacent to the structure. N₁ values between 4 and 10, with an average N₁ of about 8 are indicated on Figure 2.5.4-67. In contrast, most of the soil has N₁ values between 11 and 50, and averaging about 21. The majority of samples are described as gravelly sand, and the presence of gravel could cause N₁ values to be high. The Report on the Soil Densification Program, BVPS-2 (DLC 1976) indicates that a 13 percent reduction on average to N₁ is appropriate to account for gravel. Applying this correction reduces the average N₁ value for the low blow count material to 7, and to 18 for the deeper material. Using the Marcusson and Bieganouski (1976)

data, N_1 values of 7 and 18 correspond to relative densities of about 40 percent and 60 percent, respectively.

The low blow counts were eliminated on either side of the intake structure within the limits of the Terra Probe densification. Excavation for the intake structure mat was accomplished using a clamshell within the confines of a heavily braced steel sheetpile cofferdam. There were three vertical layers of cross bracing placed in two orthogonal directions. Horizontal spacing of the bracing was at about 17 feet on center. Considering the difficulty involved with using a clamshell in this confined space, it is unlikely that the bottom of the excavation would have conformed to the shape of the mat indicated on Figure 2.5.4-66. It is more likely that the bottom was overexcavated to about el 634.5 feet, corresponding to the bottom of the gravel layer at the south end of the mat, and then backfilled to the required elevation. This would have effectively eliminated the presence of the low blow count material below the structure.

A liquefaction analysis for materials beneath the intake structure is described below for the SSE occurring coincident with the 25-year flood at el 690 ft. Applied shear stresses determined by the computer program SHAKE (Schnabel et al 1972) are compared to shear stresses required to cause liquefaction determined using the two relationships shown on Figure 2.5.4-29a and from cyclic triaxial tests performed on reconstituted samples.

The procedure used to determine the applied shear stresses below the intake structure is shown schematically on Figure 2.5.4-68. It represents an extension of the method used by SWEC (1985) to compute site dependent ground surface response spectra.

A rock outcrop time history is input at the base of a BVPS-2 free field soil profile model. It is amplified through the profile by SHAKE to compute a ground surface time history which is scaled to the SSE magnitude. The scaling procedure, described by SWEC (1985), does not consider the acceleration level of the earthquake; magnitude is the design parameter. The scaled time history is then deconvoluted by SHAKE through the free field profile to compute a site and magnitude consistent bedrock motion, which is input below the intake structure to compute applied shear stresses.

The three rock outcrop records listed in Table 2.5.4-7 are used in this analysis. They were selected from the 28 rock outcrop records used by SWEC (1985). The basis for their selection is that applied shear stresses generated in the free field by these three records approximate the average of the shear stresses from all 28 records. Note, that for the purpose of earthquake record selection, the free field shear stresses are computed by scaling the rock outcrop time histories to the SSE magnitude prior to amplifying them through the free field profile. The free field soil model is shown on Figure 2.5.4-69. Shear wave velocities are those suggested by Whitman (1968), and are based upon insitu measurements. Soil unit weights are taken from Section 2.5.4.2. The soil model below the intake structure is shown on Figure 2.5.4-70. The structure is represented as a pseudo-soil with a unit weight and shear wave velocity compatible with characteristics of the structure. The shear wave velocity of the pseudo-soil layer is computed from the equation for the first harmonic natural period of the structure as:

$$V_s = \frac{4H_e}{T}$$

where :

 V_s = equivalent shear wave velocity

H_e = height or thickness of equivalent soil layer

T = natural period of structure

The equivalent height of the pseudo-soil is taken as the ground surface elevation around the structure minus the founding elevation. The liquefaction analysis is performed for the 25-year flood at el 690 ft, which is higher than the ground surface at el 675ft. Since SHAKE cannot handle a free water surface, the ground water table in the model is input at the ground surface. The equivalent unit weight of the pseudo-soil layer is selected so that the effective stress at the bottom of the layer is the same as the effective contact pressure of the structure for a water level at el 690 ft. Shear wave velocities of the soil layers below the structure were calculated from low strain shear moduli determined using equation 2.5.4-3. The void ratio of the low blow count layer estimated to be 0.68, based on data from the loose zone in the main plant area (DLC 1976). The void ratio of the denser layer is taken as 0.4 from Section 2.5.4.2.

SHAKE iterates to compute modulus and damping values compatible with strain levels induced by earthquake ground motions. Strain dependent variations of shear modulus and damping used in the analysis are based on data presented by Seed and Idriss (1970), and are shown on Figure 2.5.4-71.

Using the procedure shown on Figure 2.5.4-68, applied shear stresses are determined for each of the three selected rock outcrop motions and then averaged. Since SHAKE computes peak values of applied shear stress, an equivalent uniform shear stress is taken as 65 percent of the peak value (Seed and Idriss 1971). The results are shown in Table 2.5.4-8.

The resistance to liquefaction of the soils below the intake structure is evaluated using the two relationships shown on Figure 2.5.4-29A. One shows a correlation with N_1 , developed by Seed et al (1975a). It is described as a lower bound envelope for sites where liquefaction has occurred during earthquake having magnitudes between 5 and 6. The BVPS-2 SSE is approximately equivalent to a

BVPS-2 UFSAR

magnitude 5.0 earthquake, and, therefore, the cyclic stress ratio required to cause liquefaction shown by the Seed (1975a) curve is somewhat low for BVPS. Seed et al (1983), considering more recent earthquake data, discuss a procedure which accounts for the effect of varying magnitude on the cyclic stress ratio. Using this procedure, a curve for a magnitude 5 earthquake was determined and is shown on Figure 2.5.4-29A. It shows a higher cyclic stress ratio required to cause liquefaction for the same value of N_1 . The allowable shear stress at the center of the soil layers in the intake structure model, determined using the two relationships shown on Figure 2.5.4-29A.

The factor of safety against liquefaction, defined as the ratio of the allowable shear stress to the applied shear stress, is between 1.2 and 1.6 for the low blow count material, and between 3.3 and 4.6 for the higher blow count material. The minimum acceptable factor of safety is 1.1.

The resistance to liquefaction of the low blow count layer is also found to be satisfactory, based on the results of cyclic triaxial testing. The evaluation is described below.

During a study of the liquefaction potential of the soils at BVPS-1, and at the request of the U.S. NRC, a laboratory testing program was conducted to estimate the cyclic shear strength of the soil samples recovered from the freeze hole excavated in the low blow count zone beneath the BVPS-2 containment (SWEC 1977).

Cyclic triaxial testing was performed on eight reconstituted samples that were prepared from material finer than the No. 10 sieve and compacted to dry densities approximating the insitu dry density. The grain size curve of the minus No. 10 fraction is parallel to the grain size curve of the complete sample, indicating that the materials should have essentially the same mechanical properties when compacted to the same dry densities.

Samples were anistropically consolidated and tested isotropically at two confining pressures, one corresponding to the average depth of the loose zone and the other to the deepest level of the loose zone below BVPS-2. From the test results, a cyclic stress ratio was selected to define liquefaction as 2.5 percent double amplitude strain in 8 cycles as:

$$\frac{(\sigma_1 - \sigma_3) \text{ cy}}{2\overline{\sigma}_c} = 0.155$$

where:

 $(\sigma_1 - \sigma_3)$ cy = cyclic deviator stress

 σ_c = effective confining pressure

Seed (1979) contends that the cyclic simple shear test is more representative of actual field stress conditions than the cyclic triaxial test. The results of the two tests are related by a factor, CR, as shown below:

$$\left(\frac{\tau_h}{\overline{\sigma}_v}\right)$$
 simples hear = $CR \bullet \frac{(\sigma_1 - \sigma_3)_{cy}}{2\overline{\sigma}_c}$

where: τ_h = shear stress on horizontal plane

 σ_v = vertical effective stress

From the data presented by Seed (1979) and assuming a $K_{\rm o}$ of 0.5 for the low blow count soil, CR is 0.74.

Seed (1979) also recommends reducing laboratory simple shear test results an additional 10 percent to account for multidirectional shaking in the field. Therefore, an allowable field shear stress ratio, based on the results of the cyclic triaxial tests is:

$$\left(\frac{\tau_h}{\sigma_v}\right)_{field} = (0.74)(0.9)(0.155) = 0.104$$

Based on this relation, the factor of safety of any low blow count material below the intake structure is 1.4.

In summary, low blow count soils may have been present below the intake structure location, but were probably removed during excavation and foundation preparation. Should they still be present, liquefaction analyses based on blow count data and cyclic triaxial test data give satisfactory factors of safety for the soil below the intake structure.

c. Main Intake Structure: Sliding and Slope Stability

Figure 2.5.4-65 presents the loading diagram used to calculate the factor of safety against sliding of the main intake structure. The water level within the intake structure is the same as the river level. During plant operation, a maximum of one bay can be dewatered which would reduce the frictional resisting force along the base of the structure. During a seismic event, undrained shear behavior will govern sliding stability of the intake structure. Changes in vertical stresses at the soil structure interface will cause a

corresponding change in pore pressure. Therefore, the effective contact pressure will remain constant and equal to the effective building weight (total building weight minus static buoyant force). Consequently, only the horizontal component of inertial force is considered in the sliding stability analysis. Under the conservative conditions of the SSE plus standard project flood and one intake bay empty, the factor of safety against sliding is 1.3, which is satisfactory. The dynamic sliding stability analysis of the intake structure was conservatively performed without taking into account the passive resistance of the soil.

Two cross-sections of the intake channel slope at the locations shown on Figure 2.5.4-32 were analyzed for dynamic slope stability using the computer program LEASE II (SWEC 1980). One section was taken adjacent to the intake structure through the densified zone while the other section was taken approximately 100 feet from the intake structure beyond the densified zone.

The upper 10 feet of loose soil along the undensified slope and below the dredge line is susceptible to liquefaction. The pore pressure buildup in the loose zone during the seismic event is accounted for by reducing the friction angle from 25° for the drained case to 17° for the undrained case. This is conservative and assumes the pore pressure parameter equals 1, which is appropriate for loose soils (Lambe & Whitman 1969). A static, post-earthquake slope stability analysis was performed assuming that the liquefied soil would have completely liquefied at the end of a seismic event of short duration and therefore would have weight but not strength (where: $\phi = 0, c = 0$). The minimum acceptable factor of safety for the dynamic and postearthquake cases is 1.1. This is considered adequate since the liquefied soil will regain strength with time due to the dissipation of excess pore pressures generated by earthquake shaking.

The results of the dynamic slope stability analysis of the section through the densified portion of the intake channel slope presented on Figure 2.5.4-57 show satisfactory factors of safety of 1.4 for a failure circle passing behind the tied-back sheetpile wall under normal water conditions, and 1.1 for a shallow failure surface along the slope. The static, post-earthquake case with the water at the 25-year flood level and the upper 10 feet of the river bottom assumed to have liquefied ($\phi = 0$, c = 0) resulted in satisfactory factors of safety. Changing the water level from normal water at el 665 feet to the 25-year flood at el 690 feet has no significant effect on the results of the slope stability analysis since the slope is almost totally submerged under normal conditions.

Figure 2.5.4-37 presents the cross-section and soil properties assumed for the stability analysis of the intake channel slope in natural soil outside of the densified area. No borings were performed along the slope outside of the densified area. Therefore, the borings performed before densification, as shown on Figure 2.5.4-32, were used to develop the soil profile used in the analysis. The static analysis of the slope resulted in a satisfactory minimum factor of safety of 2.

As discussed previously, the upper 10 feet of the river bottom to the north of the intake structure between el 645 feet and el 635 feet, as well as the upper 10 feet along the undensified channel slopes, may The dynamic analysis, including earthquake forces, used a liquefy. reduced friction angle for the loose silty sand of 17° to account for pore pressure buildup prior to liquefaction. The results of the dynamic analysis indicated a family of failure surfaces extending below the loose silty sand layer with factors of safety less than the minimum acceptable value of 1.1. It was hypothesized that the 10-foot layer of loose silty sand along the surface of the intake channel would liquefy and flow downslope until it stabilized at about a 10:1 to 15:1 slope. The denser soils underlying the liquefied soils will remain stable as will the densified zone immediately adjacent to the intake structure, with perhaps some localized sloughing in areas directly adjacent to liquefied soils. These densified areas on either side of the intake structure will serve to prevent liquefied soil from moving directly towards the intake bays.

The intake structure draws water from 646 feet to el 659.5 feet; the pit floor at the pump intakes is at el 640 feet. The New Cumberland Lock and Dam maintains the Ohio River at a normal water elevation of 664.5 feet. A single failure of the dam during minimum river flow | conditions would result in an extreme low water level at the intake bays of el 648.6 feet. Even at this water level, there is a water depth of about 9 feet at the pump intakes. The minimum flow requirements for safe plant shutdown following the design basis accident were evaluated and found adequate for this extreme low water condition, (Sections 2.4.11 and 9.2). A proposed Technical | Specification discussed in Section 2.4.14 limits the operation of BVPS-2 to a minimum Ohio River level of el 654 feet.

Considering the geometry of the intake structure and adjacent densified areas, and the flow requirements which are adequate for safe shutdown even under extreme low water conditions, it is unlikely that the volume of soil on the intake channel slopes which may flow until stabilized at a very shallow angle would be sufficient to block the intake channel such that the safe shutdown of the plant would be jeopardized.

2.5.4.8.2 Dynamic Settlement

Ground vibrations during an earthquake tend to densify cohesionless soil and thereby cause settlement of structures founded upon them. The dynamic subsidence or settlement potential of the granular soil at the site was evaluated using concepts and data presented by Lee and Albaisa (1974). The dynamic settlement of saturated sand results from volumetric strain following the dissipation of excess pore pressures developed during cyclic loading. The magnitude of the volumetric strain is a function of several variables, including the peak pore pressure ratio developed during cyclic loading, the grain size distribution, and the relative density of the soil. The peak pore pressure ratio is the ratio of the peak excess pore pressure, Δ_u , to the effective confining pressure, $\overline{\sigma_c}$. From the results of cyclic triaxial tests, it has been found that $\Delta_u/\overline{\sigma_c}$ is primarily a function of the cycle ratio, N_c/N_L , which is the ratio of the number of significant applied cycles of loading, N_c , to the number of cycles required to cause liquefaction of the soil, N_L .

The steps used to compute dynamic settlement are outlined as follows:

- 1. The soil profile beneath the structures under consideration was divided into layers according to N_1 values, the standard penetration resistance corrected to an overburden pressure of 1 ton/ft².
- 2. An applied shear stress at the center of a given layer was determined from Figure 2.5.4-39 assuming a free field condition and the ratio of applied shear stress to total vertical stress was computed.
- 3. The applied shear stress at the center of the layer beneath the structure was computed as the applied shear stress ratio determined above for the free field case multiplied by the total vertical stress beneath the structure. Applied shear stresses below the intake structure were taken from Table 2.5.4-8.

- 4. The number of cycles to cause initial liquefaction, N , was estimated from Figure 2.5.4-38.
- 5. The volumetric strain for each layer was determined from Figure 2.5.4-40 which is based on test data reported by Lee and Albaisa (1974). A value of eight cycles was used for N for the site SSE as mentioned in Section 2.5.4.9 (DLC 1976).
- 6. Vertical strain was assumed equal to volumetric strain, implying that dynamic settlement is one-dimensional. The volumetric strain was multiplied by the layer thickness to determine the settlement of an individual layer, and the sum of these settlements for all layers beneath the structure was taken as the total settlement due to earthquake vibration.

This method was applied to soils above and below the ground-water table. It was found that the ground-water table had little effect on the calculated settlements and for the purpose of calculating dynamic settlement, it was taken at el 665 feet, or normal water level.

Considering the free field case, the average applied cyclic stress ratio is approximately 0.07. If an N_1 value within compacted structural fill beneath a given structure was as low as 10, N_1 determined from Figure 2.5.4-38 would be greater than 1,000, resulting in a volumetric strain determined from Figure 2.5.4-40 that is negligibly small. Consequently, the compacted structural fill was not considered as contributing to the dynamic settlement.

A summary of predicted dynamic settlements is given in Table 2.5.4-3.

2.5.4.9 Earthquake Design Basis

The seismicity of the Appalachian Plateau Province, of which the site is a part, is discussed in Section 2.5.2. The maximum earthquake expected at the site is an Intensity VI (MM), with a horizontal ground acceleration of 0.07g. The body-wave magnitude, m_b , of the SSE is 4.75 (SWEC 1985). The plant is designed for an SSE with a horizontal ground acceleration of 0.125g, which is slightly greater than the midpoint acceleration between Intensity VI-VII (MM). The horizontal acceleration for the OBE is 0.06g. Vertical accelerations are twothirds of the corresponding horizontal accelerations.

The BVPS-2 response spectra for the SSE are shown on Figure 3.7B-1. Analyses described by SWEC (1984) and SWEC (1985) clearly demonstrate that the BVPS-2 design response spectra are appropriate when compared to site dependent response spectra determined by current state-of-the-art methods.

To facilitate the analysis of liquefaction potential and dynamic settlement at the site, eight equivalent uniform stress cycles are used to represent the irregular acceleration-time history of the SSE.

Seed et al (1975) describe a statistical analysis of western United States earthquake time histories that is used to develop a relationship between earthquake magnitude and number of cycles of uniform motion. The BVPS-2 SSE is shown by SWEC (1985) to be equivalent to a western United States earthquake with a local (Richter) magnitude of 4.95. Seed et al (1975) shows that three to four cycles on average are representative of a magnitude 5 earthquake.

2.5.4.10 Static Stability

Foundation analyses related to the static stability of Category I structures included evaluation of bearing capacity, estimate of settlement, and the development of design lateral earth pressure parameters.

2.5.4.10.1 Bearing Capacity

All Category I structures are founded on mat foundations.

The design of mat foundations, particularly those on dense sands and gravels, is generally limited by a consideration of maximum tolerable settlements rather than by ultimate bearing capacity, since the factor of safety against a bearing capacity type failure is typically quite high. Estimated static settlements of plant structures are presented in Section 2.5.4.10.2. However, for completeness, the bearing capacity of the foundations of Category I structures and the factors of safety against a bearing capacity type failure have been computed for both static and dynamic loading conditions and are presented in Table 2.5.4-4.

The ultimate bearing capacity of the supporting soil is a function of the soil properties, the size and shape of the foundation, the depth of embedment and the depth to the ground-water table. The equation used for computing ultimate static bearing capacity is:

Square or rectangular footings:

$$q_{ult} = cN_c \left(1 + 0.3\frac{B}{L}\right) + \gamma DN_q + 0.4\gamma BN_{\gamma}$$

Circular footings: radius = R

$$q_{ult} = 1.3 \ cN_c + \gamma DN_q + 0.6 \ \gamma RN_{\gamma}$$
 (2.5.4-11)

where:

 $\begin{array}{rll} q_{ult} &= ultimate \mbox{ bearing capacity} \\ C &= \mbox{ cohesion} \\ D &= \mbox{ depth to base of mat foundation} \\ \gamma &= \mbox{ unit weight of soil} \\ B &= \mbox{ width of foundation} \\ L &= \mbox{ length of foundation} \\ N_c, N_q, N\gamma &= \mbox{ bearing capacity factors} \end{array}$

The following assumptions were made in computing the ultimate static bearing capacity:

- 1. Each structure was considered individually, ignoring increases in confinement due to adjacent structures.
- 2. Each structure was assumed to be founded on the in situ sand and gravel with the following properties:

friction angle	=	30°
cohesion	=	0
unit weight	=	125 pcf above ground-water table
	=	136 pcf below ground water table

3. The ground-water table was taken as that corresponding to probable maximum flood conditions at el 730 feet.

As discussed in Section 2.5.4.7, a portion of the safeguards area and the RWST is underlain by a layer of stiff silty clay with a top surface at approximately el 688 feet. Soil profiles depicting the conditions underlying these structures are shown on Figures 2.5.4-8 and 2.5.4-9. This stiff clay was not considered to be a concern to the stability of the structure insofar as a bearing capacity failure is concerned due to the thickness of the overlying compacted structural fill. The bearing capacities given in Table 2.5.4-4 for the safeguards area and the RWST were computed for their respective foundations on compacted fill with the preceding assumptions.

The ultimate static bearing capacity was also used as the ultimate dynamic bearing capacity when computing the factor of safety against a bearing capacity failure for dynamic loading conditions. The ultimate dynamic bearing capacity is conservatively represented by the computed ultimate static bearing capacity. Tests reported by Vesic et al (1965) for both dry and saturated dense sands, performed at various loading rates, showed a slight drop in bearing capacity with increased loading rate, followed by a steady slow increase. The observed minimum dynamic bearing capacities were about 30 percent lower than the static bearing capacities, which corresponds to a 2 degree decrease in the angle of internal friction. The in situ sands and gravels at the BVPS-2 site have an internal friction angle which ranges between 33 and 40 degrees (Section 2.5.4.2), while a 30 degree value was conservatively chosen for design purposes. Since a 2-degree reduction in the actual minimum internal friction angle of the in situ soils would result in a friction angle still higher than that used for design, the actual dynamic bearing capacity is higher than the computed static bearing capacity shown in Table 2.5.4-4. Therefore, the ultimate dynamic bearing capacity is conservatively represented by the computed ultimate static bearing capacity.

2.5.4.10.2 Settlement

This section describes the calculation procedure used to estimate the static settlement of selected points on plant structures. The same procedure is used to estimate a profile of settlement along buried, safety-related piping that extends from the structures out into the yard. The settlement profile is used to evaluate stresses imposed on the piping system using procedures described in Section 3.7B.3.12.3. This section also describes the calculation procedure used to estimate the differential settlements between the closely spaced main plant area structures that are used for pipe stress analysis. Dynamic settlements during a seismic event are discussed in Section 2.5.4.8.2.

A summary of the estimated total static settlements of the plant structures is provided on Figure 2.5.4-20. Observed settlements as of January 1, 1984 are shown on Figure 2.5.4-46.

Foundation soils in the main plant area consist of compacted select granular fill and medium dense to dense in situ granular soils. The northern portions of the safeguards area and RWST are underlain by a layer of stiff silty clay as discussed in Section 2.5.4.7. Site subsurface profiles within the plant area are shown on Figures 2.5.4-2, 2.5.4-3, 2.5.4-4, 2.5.4-5, 2.5.4-6, 2.5.4-7, 2.5.4-8 and 2.5.4-9.

The ground-water level was assumed to coincide with normal river level at el 665 feet.

Total static settlement of the plant structures founded on granular soils was assumed to consist of two components: an elastic component and a time-dependent component which was assumed to be equal in magnitude to the elastic component (Swiger 1974).

The elastic settlement of the structures in the main plant area was calculated using the computer program SETTLE II (Jubenville 1976). This program computes the elastically distributed stress with depth and computes the compression of each layer in the soil profile beneath a selected point on a given structure due to the load imposed on the soil by that structure along with any adjacent structures. The stresses induced by the loaded areas can be calculated using either Boussinesq or Westergaard solutions; the Boussinesq solution was used in this analysis. The foundation configurations, structural loads, and founding elevations of the plant structures are shown on Figure 2.5.4-41.

Certain assumptions accompany the use of SETTLE II in determining settlement. These are: 1) the load imposed by a structure was placed instantaneously, 2) the loads on all structures were placed simultaneously, and 3) settlements occurred simultaneously with load application.

In calculating settlement, the program sums the vertical strains between the founding elevation and the top of the rock according to Equation 2.5.4-12:

$$\rho = \int_{0}^{z} \varepsilon_{v} dz = \sum_{i=1}^{n} \frac{\Delta q_{i} \Delta z_{i}}{D_{i}}$$
(2.5.4-12)

where:

The constrained modulus was calculated according to the equation:

$$D_i = \frac{E_i (1-\mu)}{(1+\mu) (1-2\mu)}$$

(2.5.4 - 13)

where:

$$E_i$$
 = Young's modulus of layer $_i$
 μ = Poisson's ratio = 0.3

To account for the change in constrained modulus that occurs with changes in effective stress as construction continues and additional load is applied, an average value of constrained modulus was used to estimate the elastic settlement. Typically, an initial value of constrained modulus was computed based on the in situ stress conditions after excavation but before the structural loads were applied. SETTLE II was then used to determine the change in stresses at the center of each layer due to structural loads (including loads imposed by adjacent structures). Using these stress changes, values of the final constrained modulus were determined for each layer. Average values of the initial and final constrained moduli were then used in SETTLE II to calculate the settlement of the structures.

Young's modulus was determined by Equation 2.5.4-14:

$$E = 2G(1 + \mu)$$
 (2.5.4-14)

where:

E = Young's modulus G = Shear modulus μ = Poisson's ratio

Low strain shear moduli were estimated using the following Hardin and Black equation (Hardin and Drenevich 1972):

$$G = \frac{1,230 \ (2.97 - e)^2 \ (\overline{\sigma}_o)^{0.5}}{(1 + e)}$$
(2.5.4-15)

where:

Shear moduli determined from in situ seismic velocity measurements compared quite favorably with those computed using the Hardin and Black equation as shown on Figure 2.5.4-12. Standard penetration test N values in the densified zone showed a marked increase after densification as compared to before (DLC 1976). However, in situ seismic velocity measurements that were made after densification do not show the same marked increase (Figure 2.5.4-18), suggesting that the elastic properties of the densified zone are similar to those of the naturally dense in situ soil. Consequently, for the purpose of computing elastic properties for use in the analysis of settlement, no differentiation was made between soils within and outside the densified zone.

The value of low strain shear modulus was reduced by a factor of three to account for the reduction of shear modulus with strain (Swiger 1974).

The settlement of isolated structures outside of the BVPS-2 main plant area were calculated manually using published elastic solutions generally of the form (Poulos and Davis 1974):

$$\rho = \frac{IpB}{E}$$

(2.5.4 - 16)

where:

ρ	=	elastic settlement
I	=	influence factor which accounts for the
		shape of the loaded area and the position
		of the point for which settlement is calculated
р	=	foundation loading
B	=	characteristic dimension of structure
Е	=	Young's modulus

As with the analysis of settlement using the computer program SETTLE II, the value of moduli used was the average of the moduli determined for the initial and final stress conditions.

The settlement of the clay layer underlying the northern portion of the safeguards and the RWST was analyzed using one-dimensional consolidation theory. The estimated total settlement included both the clay layer consolidation and the elastic settlement of the in situ sand and compacted fill computed using SETTLE II. The properties of the stiff silty clay layer for use in the settlement analysis were developed from consolidation tests presented in Appendix 2.5D.

The active and passive earth pressure coefficients were computed for the case of a vertical wall, horizontal backfill, and no soil/wall friction according to the Rankine equations (Bowles 1977):

$$K = \tan^2 (45 - /2)$$

$$K = \tan^2 (45 + /2)$$

where:

К	=	coefficient of active earth pressure
K	=	coefficient of passive earth pressure
	=	effective friction angle of soil

The lateral earth pressure on a rigid wall, which experiences no appreciable deflections, is governed by the at-rest earth pressure coefficient, K_0 , computed as (Bowles 1977):

 $K = 1 - \sin \overline{\phi}$

(2.5.4 - 17)

For in situ sands and gravels with $\phi = 30^{\circ}$, K_{\circ} is 0.5. For compacted select granular fill with $\phi = 36^{\circ}$, the computed value of K_{\circ} is 0.41. From empirical correlations of strength characteristics such as that presented in U.S. Dept. of the Navy (1971), for a well-graded sand and gravel compacted to the density specified in Section 2.5.4.5.2, ϕ may be in excess of 40°, which corresponds to a K_{\circ} of 0.36 or less. A conservative K_{\circ} of 0.6 was generally used; however, lateral pressures on the walls of the lower pump cubicles on the east side of reactor containment were evaluated for a K_{\circ} of 0.36 ($\phi = 40^{\circ}$), and for the southwest wall of the control room extension and for the north and south walls of the adjoining electric cable tunnel, a K_{\circ} of 0.45 ($\phi = 33^{\circ}$) was used. These values are consistent with the estimate of ϕ for the compacted select granular fill.

The active, passive, and at-rest earth pressure coefficients for the in situ sands and the compacted select granular fill are given in Table 2.5.4-5. No safety factors have been applied to the coefficients presented.

Equations for determining the static and dynamic lateral earth and groundwater pressure distributions against unyielding walls are shown on Figure 2.5.4-42. Walls are designed for the combination of static and dynamic lateral earth and groundwater pressures and, in some cases, for compaction-induced lateral earth pressures.

Dynamic lateral earth pressures are those developed by Mononobe-Okabe and described by Seed and Whitman (1970). Hydrodynamic groundwater pressures are taken as 70 percent of the free water pressures determined by Westergaard (1933). Lateral loads on the reactor containment for seismic conditions are determined as described in Section 2.5.5.4 of the BVPS-2 PSAR (DLC 1972g). Compaction-induced lateral earth pressures are considered for the control room extension and the electrical cable tunnel. The earth pressure diagram shown on Figure 2.5.4-42 is computed using the procedure presented by Broms (1971) for the compaction equipment used at BVPS-2. The compaction-induced lateral earth pressure is added to the static lateral earth pressures computed for $K_o = 0.45$.

Design basis for structure hydrostatic loading is discussed in Section 2.4.13.5.

The differential settlements for safety-related piping that spans the shake spaces between adjacent main plant area structures are estimated by using the settlement data obtained from the settlement monitoring program described in Section 2.5.4.13. The observed settlement data is used to make a prediction of the total settlement of the two adjacent structures that are penetrated by the pipe. An average line is drawn through the log-time plots of observed settlement and extrapolated over an assumed 40-year plant life. The total settlement at the end of 40 years is reduced by the settlement that occurred prior to the date of the final weld connecting the pipe to the structures. Since settlement markers are typically not located at piping penetrations, it is necessary to interpolate between adjacent markers to estimate the total settlement at the penetration. The differential settlement of the pipe is the difference between the total settlements of the two adjacent structures at the piping penetration points subsequent to the final weld. An analysis is made of the stresses imposed on the piping system by this differential movement. For the purpose of pipe stress analysis, a minimum differential settlement of 0.5 inch has been used for initial analysis. If this assumption proves to be too conservative, the predicted differential settlement is used instead.

2.5.4.11 Design Criteria

State-of-the-art methods were used in the analysis of foundation stability of Category I structures. Methods used to evaluate bearing capacity, settlement, and lateral earth pressure are discussed in Section 2.5.4.10. The liquefaction potential and an estimate of the dynamic settlement of the granular soils at the site are discussed in Section 2.5.4.8. Soil properties used in the analyses are provided in Sections 2.5.4.2 and 2.5.4.5.

Minimum design factors of safety are as follows:

Bearing capacity	3.0 for all loading conditions.
Slope stability	<pre>1.5 for all permanent loading conditions; 1.1 for SSE loading conditions and for construction slopes.</pre>

Hydrostatic uplift 1.1 for maximum water levels.

Sliding

1.5 for all permanent loading conditions; 1.1 for SSE loading conditions.

A discussion of loads and load combinations used in the design of Category I structures is provided in Sections 3.8.1.3 and 3.8.4.3.

2.5.4.12 Techniques to Improve Subsurface Conditions

A zone of loose granular material from approximately el 640 to 660 feet was discovered in the BVPS-2 area during the excavation for the containment foundation. The extent of the loose zone was conservatively defined from exploratory borings as a zone containing a significant number of samples having N_1 values less than 10, as determined by the Gibbs and Holtz (1957) relationship. A discussion of the criteria used to establish the limits of the densified zone is provided in Section 2.5.4.8. Subsequent investigation revealed that the loose zone was present under roughly the northern half of the containment and extended east and west beneath most of the Category I structures. The loose zone was successfully densified using the pressure injected footing technique. The densification program and its evaluation are fully described in the Report on Soil Densification (DLC 1976). Plots of N_1 values obtained during Program the verification program are presented on Figures 3-29, 3-30, and 3-31 of These plots show that all samples of the loose sand and the report. gravel zone have been densified to obtain N_1 values greater than 10. Figure 3-30 shows five data points with N_1 values that are less than 10; however, these samples are not sand.

The removal of uncontrolled fill that was placed during the construction of SAPS and BVPS-1 is discussed in Section 2.5.4.5. The removal of a lens of stiff silty clay found during the containment excavation is also discussed in Section 2.5.4.5.

The approximate limits of densification of the lower terrace sands and gravels beneath the BVPS-1 circulating water lines and river water lines (WR) and the BVPS-2 service water lines (SWS) are shown on Figure 2.5.4-16. This densification program is described in responses to USAEC questions 2.26 and 2.27 in the BVPS-2 PSAR (DLC 1972e).

Initially, BVPS-1 had been designed with a once-through cooling system with an intake structure located near the present location of the BVPS-1 cooling tower. The Category I river water lines for BVPS-1 had been located directly adjacent to the 108-inch circulating water lines leading to this intake structure. Concern had been expressed that in the event of the liquefaction of soils along the circulating water lines leading to this intake structure. Concern had been expressed that in the event of the liquefaction of soils along the circulating water lines, erosion resulting from their possible rupture could disturb the adjacent river water lines and it was decided to densify the sands and gravels beneath the circulating water lines using vibroflotation to preclude this problem. After completion of the densification program, but before the installation of the circulating water lines, the decision was made to change from a once-through cooling system to closed-cycle cooling towers. Due to space limitations on the site, it was necessary that the intake structure be relocated to its present location as shown on Figure 2.5.4-16.

The soil conditions underlying the service water and river water lines from the point where they cross the circulating water lines to the present location of the intake structure are similar to the previous location and are typical of the low level terrace. The subsurface profile extending from the valve pit to the intake structure is provided on Figure 2.5.4-54. Although the results of a liquefaction analysis of the soils north of the circulating water lines to the intake indicated an adequate factor of safety against liquefaction (Appendix 2H, BVPS-1 PSAR), it was decided to densify the sands and gravels beneath the river water and service water system lines also. A typical section through the densified zone is shown on Figure 2.5.4-58.

The granular soil to the south of the densified zone on the intermediate terrace will not be subject to liquefaction (DLC 1976). If zones of granular material underlying the lower terrace outside of the densified zone were liquefied, flow slides are improbable since the rock surface does not slope toward the river appreciably but remains relatively flat at el 620 feet.

Although some surface subsidence of the soils outside of the densified zone could occur, major movements affecting the support of the service water and river water lines are not likely. The densified area would be constrained against movement towards the river by the densified area adjacent to the intake structure itself.

The limits of densification of the lower terrace sands and gravels beneath the Category II circulating water lines and the Category I service water lines to the intake structure are shown on Figure 2.5.4-As shown, the soil was densified to the top of rock using 16. vibroflotation under two of the circulating water lines from just west of the service water lines eastward to near the cooling tower. The soil underlying the service water lines to the intake structure was also densified. The subsurface profile extending from the valve pit to the intake structure is presented on Figure 2.5.4-54. The locations of verification borings 537 through 562 are shown on Figure 2.5.4-13. The results of the verification borings are presented on Figure 2.5.4-56. The minimum allowable relative density for this area was 75 percent. Only two of 178 sand and gravel samples show relative densities less than 75 percent; therefore, the program was successful. The mean relative density indicated by the verification boring data was 97.7 percent and the mean-less-one-standard-deviation relative density was 91.4 percent. The densification under the circulating water lines was done because the intake structure was originally planned for a different location,

near the present BVPS-1 cooling tower, and the service water lines were to run parallel to the circulating water lines. This work is described in the BVPS-2 PSAR response to USNRC Questions 2.26 and 2.27 addressed in Appendix 2A of the BVPS-2 FSAR.

There was a concern that the nondensified granular soils adjacent to the main intake structure, should they liquefy during an SSE, could block the intake channel and/or clog the pumps. To prevent this from occurring, two areas approximately 75 feet by 90 feet on the east and the west side of the main intake structure were densified in 1974 by the L. B. Foster Company of Union, New Jersey using the Terra Probe method. The approximate limits of the densification program are shown on Figure 2.5.4-16. The Terra Probe consists of a vibratory pile driving hammer to which a 30-inch diameter open-ended tubular probe is attached. The unit is suspended from a crane and vibrated into the soil. Densification occurs as the vibrating probe is withdrawn from the soil.

Forty-six verification borings were performed to evaluate the effectiveness of the densification program, the locations of which are shown on Figure 2.5.4-32. The median relative density at each boring location was required to be not less than 75 percent in the sands and gravels as determined using the Gibbs and Holtz relationship. In any one boring, not more than one sample point within the sands and gravels was allowed a relative density less than 70 percent and none were allowed to be less than 65 percent. If these criteria were not met, the area around the failing boring was redensified.

A test program was conducted to determine the optimum grid spacing for the Terra Probe. Three borings, TH-1 through TH-3, were performed before the test densification and three borings, TH-4 through TH-6, were performed afterwards. It was decided that a 5-foot grid spacing would be adequate to achieve the densification requirements.

Prior to beginning the production densification program, 12 borings were performed to allow a comparison of relative densities before and after densification. (Figure 2.5.4-32). A summary plot of relative densities before and after densification is given on Figure 2.5.4-43. Relative density plots of each individual verification boring are provided in Appendix 2.5C. Boring logs are provided in Appendix 2.5B.

Upon the completion of the initial series of borings, both of the areas were densified using the 5-foot grid spacing. Prior to densification, the material between the sheetpile walls was excavated to expose the tie rods at approximately el 663 feet to facilitate the insertion of the Terra Probe. After densification, backfill material was placed before performing the verification borings. Verification borings performed subsequent to the initial densification revealed that the desired densities were not being achieved in all cases. A second test panel was conducted in which a single Terra Probe was inserted and withdrawn from the soil. Three verification borings were performed, one in the center of the probe location and two at increasing distances from the probe (Borings 559T, 560T, 561T). It was found that in this particular area, densification was occurring within the probe itself and for a distance of about 8 inches outside the probe.

Selected areas offshore were redensified using a 5-foot grid spacing which overlapped the original densification pattern. The onshore areas were redensified using a 2.5-foot grid spacing. Figure 2.5.4-32 shows the approximate areas in which each of the densification patterns were performed.

BVPS-2 UFSAR

Eleven borings performed after the final densification program indicated that the densification requirements had been achieved. The boring locations are given on Figure 2.5.4-32 and summary plots of relative density before and after densification are given on Figure 2.5.4-43. The densification program required that the mean relative density at each boring location be not less than 75 percent for the sands and gravels as determined by the Gibbs and Holtz (1957) relationships. In any one boring, not more than one sample within the sands and gravels was allowed a relative density less than 70 percent and none were allowed to be less than 65 percent.

The results of the after-densification borings are summarized on Figure 2.5.4-43. Only three of 93 sand and gravel samples have relative densities less than 65 percent, and of these, two are very close to the soil surface. Thus, it is concluded that adequate densification of the sands and gravels was achieved with a mean relative density of 92.3 percent and a mean-less-one-standard-deviation relative density of 79.8 percent.

2.5.4.13 Surface and Subsurface Instrumentation

In 1977, a comprehensive settlement monitoring program was established for BVPS-2. The settlement of each BVPS-2 Category I structure was monitored during construction, and is monitored through the plant's life until the settlement of a particular structure has been determined to be stable as defined by the settlement monitoring program. For such structures, settlement monitoring is then discontinued.

Differential settlements along buried, safety-related piping that extends from the structures out into the yard and differential settlements of piping that spans the shake spaces between the closely spaced main plant area structures are not monitored as part of the settlement monitoring program. Section 2.5.4.10.2 describes the calculation procedures used to estimate the differential settlements that are used for pipe stress analysis.

During construction, settlement markers are monitored monthly, but, after construction when the structures are fully loaded and their settlement profiles begin to level out, the period between readings will be increased. Permanent bench marks are installed at various locations around the site to provide reliable survey reference points. Several piezometers monitor changes in ground-water elevation to evaluate possible correlations between settlement data and changes in ground-water elevation. In each structure several settlement markers are installed during construction, and are located so that they can be monitored during and after construction. The locations of the bench marks and piezometers are shown on Figure 2.5.4-14 and the locations of the settlement markers installed at present are shown on Figures 2.5.4-44 and 2.5.4-45.

The observed settlements to date (Figure 2.5.4-46) can be compared with the predicted total static settlements shown on Figure 2.5.4-20.

2.5.4.13.1 Bench Marks

Six permanent bench marks were installed at the locations shown on Figure 2.5.4-14. A typical bench mark installation detail is shown on Figure 2.5.4-47. It consists of a 2-inch diameter extra strong steel pipe anchored into bedrock inside of a 3 1/2-inch diameter casing extending to the top of rock. The bench marks are identified by a brass monument inscribed with the bench mark number, elevation, coordinates, and date of initial survey.

The elevations of the bench marks were checked at three-month intervals for the first year after installation and once per year thereafter. In addition, the elevations of bench marks in the

immediate vicinity of construction activities are monitored monthly and any bench mark that is disturbed or is suspected of being disturbed is resurveyed.

Bench marks are checked by running one or a series of leveling loops within the established bench marks. If, by comparison with the elevation measured during the original survey, it has been determined that a bench mark has been disturbed, a new brass monument is installed and the bench mark resurveyed.

All survey work performed in conjunction with checking and reestablishing bench marks is done using first order vertical control.

2.5.4.13.2 Piezometers

Six stand pipe piezometers were installed at the locations shown on Figure 2.5.4-14. Typical piezometer installation details are shown on Figure 2.5.4-27 and specific installation data are given in Appendix 2.5A. Tip elevations range between el 646 and el 651 feet and all of the piezometers are located within the in situ sand and gravel.

Piezometer data and Ohio River elevation data are recorded weekly and are included in Appendix 2.5A. With the exception of one period during February 1979, the ground-water levels recorded in the piezometers show very good correlation with the Ohio River elevations. During February 1979, the river rose to el 681 feet and the piezometer data indicate an apparent time lag. However, the piezometers were only read weekly during the period of high water and in the interim between readings the water level in the piezometers may have continued to rise, thereby reducing the apparent elevation difference between the ground-water levels and the Ohio River elevation.

2.5.4.13.3 Settlement Markers

The locations of the currently installed settlement markers are shown on Figures 2.5.4-44 and 2.5.4-45. Details of the several types of markers are shown on Figure 2.5.4-48. Construction activity in certain structures requires that settlement markers be relocated periodically in order to provide continuing access to the markers. In such structures, temporary markers have been installed instead of permanent markers. Temporary settlement markers have been installed on the reactor containment building, the safeguards area, the fuel and decontamination building, and the cooling tower. When construction activity diminishes to the point that markers are no longer subject to periodic relocation, the temporary settlement markers are replaced with permanent ones.
2.5.4.13.4 Data Processing

Data processing is accomplished using a SWEC computerized data storage system entitled Settlement Monitoring System (IS-233). The settlement marker elevations are input into the computer storage files and a computer printout providing the complete settlement record of each marker is produced. A specimen page of output is given on Figure 2.5.4-49.

For each settlement marker, settlement versus log-time plots have been prepared. These plots are not included herein but are provided in the report on Settlement Monitoring Program (DLC 1980). A summary of the observed settlements to date is shown on Figure 2.5.4-46.

The Ohio River elevation and piezometer data are included in Appendix 2.5A.

2.5.4.14 Construction Notes

The removal of uncontrolled fill placed during the construction of SAPS and BVPS-1 is discussed in Section 2.5.4.5. The removal of a lens of stiff silty clay found during the reactor containment excavation is also discussed in Section 2.5.4.5.

A zone of loose granular material was discovered in the BVPS-2 area during the excavation for the reactor containment excavation. It was densified using the pressure injected footing technique. The densification program and its evaluation are fully described in the Report on Soil Densification Program, (DLC 1976).

2.5.4.15 References for Section 2.5.4

Audibert, J.M.E. and Nyman, K.J. 1977. Soil Restraint Against Horizontal Motion of Pipes. Journal of the Geotechnical Engineering Division. ASCE October, 1977.

Bowles, J. E. 1977. Foundation Analysis and Design. McGraw-Hill Book Company, New York, N.Y.

Broms, B. 1971. Lateral Earth Pressures Due to Compaction of Cohesionless Soils. Proceedings, 4th Conference on Soil Mechanics, Budapest.

Bullen, K. E. 1963. An Introduction to the Theory of Seismology. Cambridge University Press, Cambridge, England.

Christian, J. T. 1976. Relative Motion Between Two Points During an Earthquake. Journal of the Geotechnical Engineering Division, Vol. 102, No. GT11. November, ASCE.

Dravo Corporation 1974. Subsurface Investigation Routing of Sludge Transportation Pipes Around Beaver Valley Power Station, Little Blue Run Development Area. Prepared by General Analytics Inc., Monroeville, Pa.

Duquesne Light Company (DLC) 1972a. Preliminary Safety Analysis Report - Beaver Valley Power Station - Unit 2, Appendix 2E. Prepared by Stone & Webster Engineering Corporation (SWEC), Boston, Mass.

Duquesne Light Company 1972b. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2, Appendix 2F. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972c. Final Safety Analysis Report - Beaver Valley Power Station - Unit 1, Appendix 2H. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972d. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2 response to USNRC question 2.24 (logs not published). Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972e. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2. Response to USNRC Questions 2.26 and 2.27. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972f. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2, Appendix 2G. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972g. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2, Section 2.6.5.4. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972h. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2. Volume 10 Appendix 2A, Response to Draft Questions Received January 21, 1976. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1972i. Preliminary Safety Analysis Report -Beaver Valley Power Station - Unit 2. Appendix 2D. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1976. Report on the Soil Densification Program - Beaver Valley Power Station - Unit 2. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1979, Inspection Manual - Earthwork IP-6.1. Beaver Valley Power Station - Unit 2.

Duquesne Light Company 1979. Soil Analysis of Turbine Building and Northern Yard Area. Prepared by SWEC, Boston, Mass.

Duquesne Light Company 1980. Report on Settlement Monitoring Program - Beaver Valley Power Station - Unit 2. Prepared by SWEC, Boston, Mass.

Ewing, W.; Jardetsky, W.; and Press, F. 1957. Elastic Waves in Layered Media. McGraw-Hill Book Company, New York, N.Y. Gibbs, H.

J. and Holtz, W. H. 1957. Research on Determining the Density of Sands by Spoon Penetration Testing. Fourth International Conference on Soil Mechanics and Foundation Engineering. Volume 1. Butterworth, London.

Hardin, B. O. and Drenevich, V. P. 1972. Shear Modulus and Damping in Soils, Design Equations. Journal of the Soil Mechanics and Foundation Division, Vol. 98, SM-7, ASCE.

Housner, G.W. 1970. Strong Ground Motion. Contained in Weigel, R.L. Earthquake Engineering. Prentice Hall, Inc. Englewood Cliffs, N.J.

Jubenville, D. M. 1976. Settle II. A Computer Program to Calculate Settlements. Geotechnical Engineering Software Activity, University of Colorado Computing Center, Boulder, Colo.

Lambe, T.W. and Whitman, R.V. 1969. Soil Mechanics. John Wiley and Sons, New York, N.Y.

Lee, K. L. and Albasia, A. 1974. Earthquake Induced Settlements in Saturated Sands. Journal of the Geotechnical Engineering Division, Vol. 100, GT4, ASCE.

Marcuson, W. F. and Bieganouski, W. A. 1977. SPT and Relative Density in Coarse Sands. Journal of the Geotechnical Engineering Division, Vol. 103, GT11, ASCE.

Poulos, H. G. and Davis, E. H. 1974. Elastic Solutions for Soil and Rock Mechanics. John Wiley and Sons, New York, N.Y.

Sampson, R. J. 1975. SURFACE II Graphics System. Kansas Geological Survey, Lawrence, Kansas.

Schnabel, P.B.; Lysmer, J.; and Seed, H.B. 1972. SHAKE: A Computer Program for Earthquake Response Analysis of Horizontally Layered Sites. Report EERC-72-12. University of California at Berkeley.

Seed, H. B. and Whitman, R. V. 1970. Design of Earth Retaining Structures for Dynamic Loading. Speciality Conference on Lateral Stresses and Design of Earth Retaining Structures, ASCE, New York, N.Y.

Seed, H.B. and Idriss, I.M. Soil Moduli and Damping Factors for Dynamic Response Analysis. Report EERC-70-10. College of Engineering, University of California at Berkeley, 1970.

Stone & Webster Engineering Corporation 1979. Relden, Relative Density from Standard Penetration Tests. GT-004. Version 04, Level 01. Seed, H. B. and Idriss, I. M. 1971. A Simplified Procedure for Evaluation of Soil Liquefaction Potential. Journal of Soil Mechanics and Foundations Division, ASCE, Vol. 97, No. SM9.

Seed, H. B.; Arango, I.; and Chan, C. K. 1975a. Evaluation of Soil Liquefaction Effects During Earthquakes. College of Engineering, University of California, Berkeley, Report No. EERC 7528.

Seed, H.B.; Idriss, I.M.; Makdisi, F.; and Banerjee, N. 1975b. Representation Of Irregular Stress Time Histories by Equivalent Uniform Stress Series in Liquefaction Analyses. Report No. EERC 75-29. Earthquake Engineering Research Center, University of California at Berkeley, October.

Seed, H. B. 1976. Evaluation of Soil Liquefaction Effects on Level Ground During Earthquakes, Liquefaction Problems in Geotechnical Engineering. ASCE, New York, N.Y.

Stone & Webster Engineering Corporation (SWEC) 1977. Soils Study -Category I Structures (Response to NRC Letter - November 17, 1976), Beaver Valley Power Station - Unit 1. Supplement No. 2, February 14.

Stone & Webster Engineering Corporation (SWEC) 1978. Excavation and placement of Fill Under Structures and Final Backfilling Around Structures. Specification No. 2BVS-928. Beaver Valley Power Station - Unit 2.

Stone & Webster Engineering Corporation (SWEC) 1979. Relden, Relative Density from Standard Penetration Tests. GT-004. Version 04, Level 01.

Stone & Webster Engineering Corporation (SWEC) 1980. Lease II Limiting Equilibrium Analysis in Soil Engineering. GT-018, Version 02, Level 00.

Stone & Webster Engineering Corporation (SWEC) 1982. Calculation of Relative Displacement between Two Buildings during an Earthquake. GT-037.

Stone & Webster Engineering Corporation (SWEC) June 1984. Seismic Design Response Spectra. Beaver Valley Power Station - Unit 2. Prepared for Duquesne Light Company, Pittsburgh, Pa.

Stone & Webster Engineering Corporation (SWEC) February 1985. Site Dependent Response Spectra, Beaver Valley Power Station - Unit 2. Prepared for Duquesne Light Company, Pittsburgh, Pa.

Swiger, W. F. 1974. Evaluation of Soil Moduli. Analysis and Design in Geotechnical Engineering, ASCE, New York, N.Y.

Terzaghi, K. and Peck, R. 1967. Soil Mechanics in Engineering Practice. John Wiley and Sons, New York, N.Y.

Trifunae, M.D. and Lee, V. 1973. Routine Computer Processing of Strong Motion Accelerograms, Volume II, Strong Motion Earthquake Accelerograms, Digitized and Plotted Data. Report EERL-73-03. California Institute of Technology, Earthquake Engineering Research Laboratory.

U. S. Department of the Navy 1971. Design Manual, Soil Mechanics, Foundations, and Earth Structures. NAUFAC DM-7.

Vesic, A.B. 1961a. Beams on Elastic Subgrade and Winkler's Hypothesis. Proc. 5th International Conference on Soil Mechanics and Foundation Engineering, Paris pp. 845-850.

Vesic, A.B. 1961b. Bending of Beams Resting on Isotropic Elastic Solid. Journal of the Engineering Mechanics Division. ASCE Vol. 87, No. EM2 April, pp 35-53.

Vesic, A.B.; Banks, D.C.; and Woodward, J.M. 1965. An Experimental Study of Dynamic Bearing Capacity of Footings on Sand, Proceedings, Sixth International Conference on Soil Mechanics and Foundation Engineering, Montreal, Canada, Vol. II, pp 209-213.

Westergaard, H.M. 1933. Water Pressures on Dams during Earthquakes. Transactions of ASCE. Vol. 98.

Whitman, R.V. 1968. Effect of Local Soil Conditions on Seismic Threat to Beaver Valley Power Station. Contained in Stone & Webster Engineering Corporation 1968. Preliminary Safety Analysis Report, Beaver Valley Power Station - Unit 1. Docket No. 50-334. BVPS-2 UFSAR

Tables for Section 2.5.4

BORING LOG INDEX

Project	Boring Number	Date Drilled	Purpose	Boring Log <u>Reference</u>
Shippingport Atomic Power Station	1 - 23 29 - 33 A - H J - N & P	July 1954 April 1955 April 1955 April 1955	General Site Investigation	1 1 1 1
Beaver Valley Power Station - Unit 1	101 - 117 301 - 310 401 - 404 501 - 518 519 - 536 537 - 562 537T - 577T TH1 - TH6 601, 602 608 - 613 650 - 652 701 - 718 H1 - H5 AB1 - AB12	March/April 1968 July 1969 Nov./Dec. 1969 June 1970 Nov./Dec. 1970 Jan./Feb. 1973 March/April 1974 March/April 1974 May 1971 May 1971 May 1971 August 1971 Feb./March 1974	General Site Investigation General Site Investigation General Site Investigation Vibroflotation Program Vibroflotation Verification Terra Probe Verification Terra Probe Verification Cooling Tower Location Study Cooling Tower Location Study Cooling Tower Location Study Cooling Tower Location Study Highway Embankment Stability Turbine Building Stability	2 3 4 2 2 5 5 4 2 4 4 4 6 7
Beaver Valley Power Station - Unit 2	801 - 843 854, 855 901 - 916 918 - 980 1000 - 1030 OF1 - OF6 RH1 Z1 PL1 - PL3 CT1 - CT3 SWS 1 - SWS 5 Figure 2.5.4-15 SEO-1 - SEO-5 EOS-1 - EOS-10 TH-1 - TH-13	Nov. 1971 - April 1972 July 1974 March 1974 July - Sept. 1976 May - June 1976 July 1976 August 1976 November 1976 Feb. 1977 July 1977 August 1977 May 1976 - July 1977 October 1981 May - June 1982 May - June 1982	General Site Investigation General Site Investigation Auxiliary Intake Structure Loose Zone - Plant Area Loose Zone - Containment Office Building General Site Investigation General Site Investigation Parking Lot Cooling Tower Service Water System Soil Verification Borings Office Building Emergency Outfall Structure Emergency Response Facility	2 5 8 8 8 8 8 8 8 8 8 8 8 8 5 5 5 5 6
Bruce Mansfield Power Plant <u>NOTES:</u> 1. Duquesne Light Company 2. Duquesne Light Company 3. Duquesne Light Company 4. Duquesne Light Company 5. Appendix 2.5B.	PL1 - PL66 9 1972a 9 1972b. 9 1972c. 9 1972d.	May - July 1974 6. Not published 7. Duquesne Light Company 1979. 8. Duquesne Light Company 1976. 9. Dravo Corporation 1974.	Slurry Pipelines	9

BVPS-2 UFSAR

TABLE 2.5.4-2

MATERIALS TESTING REQUIREMENTS AND FREQUENCY

Materials	Type Of Test	Test Designation	Minimum Frequency Of Testing	Remarks	
Excavated material	In-place density	Washington densometer ASTM D2167 and/or Nuclear densometer ASTM D2922	One per 5 ft of depth	Testing to be on exca- vated material which is to be stockpiled for later use.	
	Moisture density	ASTM D1557 Method D	Whenever visual inspection indicates a significant change in material gradation.		
	Sieve analysis	ASTM D422 and ASTM D1140	One per moisture density test.		
Founding elevation material	In-place density	Washington densometer ASTM D2167 and/or Nuclear densometer ASTM D2922	As directed by the Geotechnical Engineer.		
Select granular/ structural fill	In-place density	Washington densometer ASTM D2167 and/or Nuclear densometer ASTM D2922	 Areas greater than 1,000 ft², one per 1,000 yd³ placed or one per alternate lift, whichever results in a greater frequency. Areas less than 1,000 ft², one for every 2.5 ft of compacted fill. For each 200 linear ft of trench, one per alternate lift of compacted fill, or as directed by the Geotechnical Engineer. 		
	Moisture density	ASTM D1557 Method D	 One per new source. One per 5,000 yd³ placed Whenever visual inspection indicates a significant change in material gradation. 	Material for moisture density test shall be taken adjacent to an in-place density test.	
	Sieve analysis	ASTM D422 and ASTM D1140	One for each moisture density test	Material for sieve analysis shall be taken from material sampled for the moisture den- sity test	

MATERIALS TESTING REQUIREMENTS AND FREQUENCY

Materials

Type Of Test

Minimum Frequency Of Testing

Remarks

Specific gravity

ASTM C127

Test Designation

- One per new source.
 One per every 50 moisture density tests.

SUMMARY OF PREDICTED DYNAMIC SETTLEMENTS

<u>Structure</u>	Dynamic Settlement (in)
Auxiliary building	0.12
Control room extension	0.11
Demineralized water tank	0.16
Diesel generator building	0.13
Fuel building	0.14
Main steam and cable vault	0.14
Reactor containment	0.09
Refueling water tank	0.16
Safeguards area	0.14
Service building	0.16
Valve pit	0.14
Emergency outfall structure	0.03
Main intake structure	0.10

BEARING CAPACITY - CATEGORY I STRUCTURES

	Approximate Dimensions of Contact Area (ft)	Approximate Foundation Depth (ft)	Ultimate Bearing Capacity (ksf)	<u>Static</u> Approximate* Load (ksf)	Factor of Safety	<u>Dynamic</u> Approximate* Load (ksf)	Factor of Safety
	<u>, , ,</u>						
Auxiliary building	120 x 146	32	129	5.7	32	10.6	15
Control room extension	65 x 81	32	97	3.5	54	5.6	25
Decontamination building	33 x 33	5.5	33	6.3	5	11.5	3
Demineralized water tank	38 x 38	4.7	35	3.4	10	10.9	3
Diesel generator building	81 x 83	22	90	3.1	45	5.9	19
Emergency outfall structure	25 x 30	25	60	3.2	10	8.0	9
Fuel building	44 x 110	17.7	61	6.8	10	11.9	5
Main intake structure	84 x 89	39.5	115	8.9	19	6.7**	24**
Main steam and cable vault	90 x 135	22.5	74	3.7	28	7.1	12
Reactor containment	142 dia.	54	157	7.5	36	12.4	17
Refueling water storage tank	57 x 58	4.7	45	3.5	13	8.8	5
Safeguards area	60 x 98	20.5	76	3.2	35	4.7	21
Service building	55 x 186	9.5	54	4.0	15	4.6	13
Valve pit	25 x 37	18.8	50	1.6	31	3.8	17

NOTES:

- * Foundation load does not include buoyant effect of water. Bearing capacity calculated assuming ground-water level at el 730 feet corresponding to PMF conditions. Dynamic load evaluated for groundwater level at el 665 feet, corresponding to normal water conditions.
- **

SUMMARY OF LATERAL EARTH PRESSURE COEFFICIENTS

Coefficient of <u>Earth Pressure</u>	<u>In Situ Soil</u>	Compacted Select Granular Fill
Active, K _a	0.33	0.26
Passive, K _p	3.0	3.85
At rest, K _o	0.50	0.6
		0.45*
		0.36**

NOTES:

* For control room extension and electric cable tunnel only. ** For lower pump cubicles of the reactor containment only.

STRUCTURAL FILL SUPPLIER AND QUANTITIES PROVIDED

Supplier	Quanity <u>(cu yds)</u>
X & L Sand & Gravel Midland, Pennsyvania	287,695
X & L Sand & Gravel Negley, Pennsylvania	192,557
Mahoning Sand & Gravel	306
Georgetown Sand & Gravel Georgetown, Pennsylvania	175,745
Dravo Corporation Georgetown, Pennysylvania	24,855
Supplier X & L Sand & Gravel Midland, Pennsyvania X & L Sand & Gravel Negley, Pennsylvania Mahoning Sand & Gravel Georgetown Sand & Gravel Georgetown, Pennsylvania Dravo Corporation Georgetown, Pennysylvania Dravo/Kabuta Kabuta, Pennsylvania	<u>62,835</u>
Kabuta, Pennsylvania	743,993

LIQUEFACTION ANALYSIS AT INTAKE STRUCTURE EARTHQUAKE RECORDS

Year	<u>Mo.</u>	<u>Day</u>	Earthquake Name	Local <u>Magnitude</u>	Scaling Factor	Recording <u>Station</u>	<u>Component</u>	Record <u>No.</u>
1935	10	31	Helena, MT	6.0	0.271	Carroll College Helena, MT	EW	B-025
1975	09	27	Oroville, CA Aftershock	4.6	1.452	Oroville, CA CDMG No. 8	SOOE	8-234
1979	08	06	Coyote Lake, CA	5.9	0.307	Coyote Creek San Martin, CA	160°	SM-879

MAIN INTAKE STRUCTURE LIQUEFACTION ANALYSIS

Soil		_	Shear <u>Applied</u>	Stress <u>All</u>	(psf) owable**	<u>Safety</u>	Factor
Layer*	<u>N</u> 1	σ_v (psf)	A	<u> </u>	C	<u>B/A</u>	<u>C/A</u>
3	7	4,210	309	380	491	1.2	1.6
4	18	4,740	339	1,101	1,422	3.3	4.2
5	18	5,280	346	1,226	1,584	3.5	4.6

NOTES:

* Soil model shown on Figure 2.5.4-70

** Allowable shear stress on Figure 2.5.4-29

B τ allow = 0.0129N₁ σ_v (Seed et al 1975)

C τ allow = 0.01667N₁ σ_v (Seed et al 1983)

SITE MATCHED GROUND SURFACE EARTHQUAKE RECORDS*

Date		Epicentral	Recording	CIT Record		
Year	<u>Month</u>	<u>Day</u>	Location	<u>Station</u>	<u>Component</u>	<u>No. **</u>
1954	12	21	Eureka, CA	Federal Bldg. Eureka, CA	N79E S11E	A-008
1957	03	22	San Francisco	State Bldg.	N09E	A-016
			CA	San Francisco, CA	S81W	
				Alexander	N09W	A-104
				San Francisco, CA	N81E	
1957	03	22	San Francisco	Alexander	N09W	V-323
			CA	San Francisco, CA	N81E	
				City Hall, Oakland, CA	N26E S64E	A-017
1962	09	04	Northern CA	Federal Bldg. Eureka, CA	N79E S11E	V-330
1965	07	15	Southern CA	Old Ridge Rte. Castaic, CA	E S	V-331
1970	09	12	Lytle Creek, CA	6074 Park Dr. Wrightwood, CA	S65E S25W	W-334
1971	02	09	San	Old Ridge Rte.	N21E	D-056
			CA	Castaic, CA	N69W	

NOTES:

*Per SWEC (1985)

**California Institute of Technology reference number, Trifunae
and Lee (1973)

RELATIVE DISPLACEMENT OF SELECTED STRUCTURES USING THE EARTHQUAKE TIME-HISTORY METHOD

Structures		Centroid	Relative Displacement (in.)		
From	То	Distance (ft)	Horizontal	Vertical	
Main Steam & Cable Vault	Auxiliary Building	115	0.29	0.19	
Main Steam & Cable Vault	Safeguards Area	150	0.32	0.21	













-

.



-

~~~~

 $\sim$ 

AN AND

(803) SOUTH - 840 - 820 -780 CONDENSATE -760 POLISHING BUILDING WASTE HANDLING BUILDING 740 720 - 700 -680 -660 -640 



FIGURE 2.5.4-6 SUBSURFACE PROFILE E-E' BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





FIGURE 2.5.4-7 SUBSURFACE PROFILE F-F<sup>1</sup> BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

1



SOUTH





 $\widehat{\cdot}$ 









# Removed in Accordance with RIS 2015-17

FIGURE 2.5.4-14 BENCHMARK AND PIEZOMETER LOCATION PLAN BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





| -                                         | 1968 SURVE                          |                            |                                             |                                   |                    |
|-------------------------------------------|-------------------------------------|----------------------------|---------------------------------------------|-----------------------------------|--------------------|
| UNDIS                                     | TURBED IN SITU                      | SOIL                       | DENSIFIED IN SITU SOIL                      |                                   |                    |
| EL. 740'-                                 | "P" WAVE<br>VELOCITY (FPS)          | "S" WAVE<br>VELOCITY (FPS) | "P" WAVE<br>VELOCITY (FPS)                  | "S" WAVE<br>VELOCITY(FPS)         | - EL. 740'         |
| AP                                        | PROX. GROUND SU                     | RFACE                      |                                             |                                   |                    |
| EL. 720'—                                 | 1,500<br>(some 1,000)               | 900 -<br>(Some 600)        | APPROX. GROUND                              | SURFACE                           | - EL. 720'         |
|                                           | 2,000                               | 900-1,200-                 |                                             |                                   |                    |
| EL. 700'—                                 |                                     |                            | NOTE:<br>NO SEISMIC<br>ABOVE EL.            | MEASUREMENTS TA                   | - EL. 700'<br>AKEN |
| EL.680'—                                  | 2,000-                              | 1,050±                     | 2,000 - 2,500<br>??<br>2,400 - 2,500        | 700 – 800?<br>? ?<br>1,000        | — EL. 680'         |
|                                           | PROX. WATER TABL                    | E EL.665 FT.               |                                             |                                   |                    |
| EL. 660'—                                 | 6,000                               | 1,300                      | 3,000<br>Approx. water                      | 1,000 - 1,200<br>TABLE EL.652 FT. | - EL.660'          |
| EL. 640'—                                 | 6,000                               | I, 300 <sup>—</sup>        | 6,300-6,500                                 | 1,500 — 1,800                     | — EL. 640'         |
| EL. 620'—                                 |                                     |                            |                                             |                                   | — EL. 620'         |
| EL. 600'—                                 | 12,000                              | 6,000-                     | 12,000                                      | 4,400-5,800?                      | - EL. 600'         |
| NOTE:<br>1. DLC 197<br>2. REFER<br>SECTIO | 76<br>TO DISCUSSION IN<br>N 2.5.4-4 |                            | FIGURE 2.5.4-<br>GENERALIZED<br>WAVE VELOCI | 17<br>) "P" AND "S"<br>TY VALUES, |                    |

WAVE VELOCITY VALUES, 1968 AND 1977 SURVEYS BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



GROUND SURFACE AT EL. 735 FT FOR 1968 SURVEY AND AT EL. 715 FT FOR 1977 SURVEY.

DLC 1976

FIGURE 2.5.4-18 COMPARISON OF IN SITU SHEAR WAVE VELOCITIES BEFORE AND AFTER DENSIFICATION BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

• • •








STANDARD 2. MOISTURE DENSITY TESTS (ASTM 01557 METHOD D) PERFORMED ON DEVIATION MEAN SAMPLES REVEALED THE FOLLOWING: MAXIMUM DRY UNIT WEIGHT(PCF) 136.9 1.6 OPTIMUM WATER CONTENT 7. 0 1.4

BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





| ELEVATION<br>FR. FLEVATION                                         |
|--------------------------------------------------------------------|
|                                                                    |
| SEAL                                                               |
|                                                                    |
| EN                                                                 |
|                                                                    |
|                                                                    |
| 10 1 <b>5</b> 20<br>MAY 1976                                       |
| FIGURE 2.5.4-23                                                    |
| OBSERVATION WELL DATA                                              |
| BEAVER VALLEY POWER STATION-UNIT 2<br>FINAL SAFETY ANALYSIS REPORT |
|                                                                    |
|                                                                    |



OBSERVATION WELL DATA OW-2 BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

1 1 15 20 MAY 1976

NO INSTALLATION DATA AVAILABLE



15 20 FIGURE 2.5.4-25 OBSERVATION WELL DATA 0W-3 BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT











LEGEND:

- $\tau$  = SHEAR STRESS
- $\bar{\sigma}_v$  = VERTICAL EFFECTIVE STRESS
- M = MAGNITUDE

FIGURE 2.5.4-29A CORRELATION BETWEEN  $\tau/\overline{\sigma_v}$ CAUSING LIQUEFACTION AND NI BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





NOTE:

EWING ET AL, 1957.









FIGURE 2.5.4 -32 TERRA PROBE DENSIFICATION MAIN INTAKE STRUCTURE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





SOIL DATA FROM BORINGS PRIOR TO DENSIFICATION: 537T-542T

N, DETERMINED USING MARCUSSON & BIEGANOUSKI (1977) DATA

FIGURE 2.5.4-34 LIQUEFACTION ANALYSIS AT MAIN INTAKE STRUCTURE-INTAKE CHANNEL BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



.

\_\_\_\_\_





## NOTE

(1) FAILURE CIRCLES WITH RADII LESS THAN THOSE SHOWN HAVE DYNAMIC FACTORS OF SAFETY LESS THAN 1.1.

|              | SOIL PROPERTIES                |                    |                                    |                                 |                  |  |  |  |  |  |
|--------------|--------------------------------|--------------------|------------------------------------|---------------------------------|------------------|--|--|--|--|--|
| SOIL<br>UNIT | TOTAL UNIT<br>WEIGHT<br>γt,pcf | COHESION<br>C, psf | FRICTION<br>UNDRAINED<br>Ø DEGREES | N ANGLE<br>DRAINED<br>ゆ DEGREES | SOIL DESCRIPTION |  |  |  |  |  |
| 1            | 120                            | 0                  | 7                                  | 25                              | SILTY SAND       |  |  |  |  |  |
| 2            | 110                            | 350                | 0                                  | 0                               | CLAY             |  |  |  |  |  |
| 3            | 136                            | 0                  | 30                                 | 30                              | SAND AND GRAVEL  |  |  |  |  |  |
| 4            | 120                            | 0                  | 17                                 | 25                              | LOOSE SILTY SAND |  |  |  |  |  |
| L            | L                              |                    | 1                                  |                                 |                  |  |  |  |  |  |
|              |                                |                    |                                    |                                 |                  |  |  |  |  |  |



FIGURE 2.5.4 - 37 MAIN INTAKE CHANNEL SLOPE STABILITY SECTION I-I BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT











FIGURE 2.5.4-41 PLANT FOUNDATION ELEVATION AND LOAD DATA BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

120 40 80 SCALE-FEET

1. (719.0, 1.9)- FOUNDATION ELEVATION IN FEET, LOAD IN KSF

REFUELING WATER STORAGE TANK

PLANT X



FIGURE 2.5.4-42 LATERAL EARTH PRESSURES ON RIGID WALLS BEAVER VALLEY POWER STATION-UNIT2 FINAL SAFETY ANALYSIS REPORT

ah = HORIZONTAL SEISMIC COEFFICIENT = 0.125g (SSE)





FIGURE 2.5.4-44 SETTLEMENT MARKER LOCATION PLAN BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT





## NOTES

•

 $\frown$ 

 $\frown$ 

 $\frown$ 

- I. OBSERVED DATA SHOWN REPRESENTS THE SETTLEMENT (HEAVE) OF A GIVEN SETTLEMENT MARKER ESTIMATED BY AN AVERAGE LINE THROUGH THE SURVEY DATA AS OF JAN. 1, 1984. IF NO DATA IS GIVEN, INSUFFICIENT SURVEY DATA WAS AVAILABLE WITH WHICH TO ESTIMATE SETTLEMENT (HEAVE) OF THE MARKER.
- 2. LETTERED DESIGNATION OF SETTLEMENT MARKERS GIVEN IN FIGURE 2.5.4-44 AND 2.5.4-45.
- 4. APPROXIMATE PERCENTAGE OF TOTAL STRUCTURAL LOAD, INCLUDING MAJOR PIECES OF EQUIPMENT, AS OF JAN. 1, 1984. DOES NOT INCLUDE WEIGHT OF WATER FOR STRUCTURES 17, 18 AND 21.

3. 0.024 = SETTLEMENT, INCHES; -0.024 = HEAVE, INCHES.







| CONSTRUCTION (4) |    |                                      |
|------------------|----|--------------------------------------|
| PROGRESS (%)     |    |                                      |
| 95               | I  | VALVE PIT                            |
| 99               | 2  | ALTERNATE INTAKE STRUCTURE           |
| 94               | 3  | COOLING TOWER PUMPHOUSE              |
| 92               | 4  | CONTROL ROOM EXTENSION               |
| 93               | 5  | ELEC. CABLE TUNNEL                   |
| 95               | 6  | FUEL & DECON. BUILDING               |
| 95               | 7  | REACTOR CONTAINMENT                  |
| 98               | 8  | SAFEGUARDS AREA                      |
| 97               | 9  | AUXILIARY BUILDING                   |
| 9 <b>8</b>       | 10 | MAIN STEAM & CABLE VAULT             |
| 93               | 11 | SERVICE BUILDING                     |
| 93               | 12 | DIESEL GENERATOR                     |
| 95               | 13 | PIPE TUNNEL                          |
| 97               | 14 | CONDENSATE POLISHING BUILDING        |
| 95               | 15 | WASTE HANDLING BUILDING              |
| 91               | 16 | TURBINE BUILDING                     |
| 80               | 17 | REFUELING WATER TANK                 |
| 80               | 18 | DEMINERALIZED WATER TANK             |
| 100              | 19 | SANITARY TREATMENT BUILDING (BVPS-I) |
| 100              | 20 | ALTERNATE ACCESS FACILITY (BVPS-I)   |
| 99               | 21 | COOLING TOWER                        |
|                  | Þ  | VERTICAL MARKER                      |

- VERTICAL MARKER (TEMPORARY)
- HORIZONTAL MARKER
- O SLEEVE TYPE MARKER

FIGURE 2.5.4-46 SUMMARY OF OBSERVED SETTLEMENTS BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT





FIGURE 2.5.4-48 SETTLEMENT MARKER INSTALLATION DETAILS BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

| IAN 28, 1980<br>Detailed Settlement Report<br>Beaver Valley 2: 12241 |         |         |                        |                         |                             |                   |                                 |  |
|----------------------------------------------------------------------|---------|---------|------------------------|-------------------------|-----------------------------|-------------------|---------------------------------|--|
| GT-030-100                                                           |         |         |                        |                         |                             |                   |                                 |  |
| MARKER NO.                                                           | X-COORD | Y-COORD | REFERENCE<br>BENCHMARK | BASE<br>ELEV.<br>(FEET) | SURVEYED<br>ELEV.<br>(FEET) | DATE OF<br>SURVEY | TOTAL<br>SETTLEHENT<br>(INCHES) |  |
| T1                                                                   | .000    | . 000   | B2                     | 730.561                 | 730.561                     | 07/06/77          | .000                            |  |
|                                                                      |         |         |                        |                         | 730.561                     | 08/02/77          | .000                            |  |
|                                                                      |         |         |                        |                         | 730.559                     | 09/01/77          | .024                            |  |
|                                                                      |         |         |                        |                         | 730.565                     | 10/01/77          | 048                             |  |
|                                                                      |         |         |                        |                         | 730.565                     | 11/01/77          | 048                             |  |
|                                                                      |         |         |                        |                         | 730.567                     | 12/06/77          | 072                             |  |
|                                                                      |         |         |                        |                         | 730.569                     | 01/05/78          | 096                             |  |
|                                                                      |         |         |                        |                         | 730.572                     | 02/01/78          | 132                             |  |
|                                                                      |         |         |                        |                         | 730.570                     | 03/01/78          | 108                             |  |
|                                                                      |         |         |                        |                         | 730.556                     | 04/04/78          | .060                            |  |
|                                                                      |         |         |                        |                         | 730.551                     | 05/04/78          | .120                            |  |
|                                                                      |         |         |                        |                         | 730.543                     | 06/03/78          | .216                            |  |
|                                                                      |         |         | ,                      |                         | 730.540                     | 07/06/78          | .252                            |  |
|                                                                      |         |         |                        |                         | 730.532                     | 08/03/78          | . 348                           |  |
|                                                                      |         |         |                        |                         | 730.535                     | 09/05/78          | 312                             |  |
|                                                                      |         |         |                        |                         | 730.536                     | 10/02/78          | .300                            |  |
|                                                                      |         |         |                        |                         | 730.541                     | 11/07/78          | .240                            |  |
|                                                                      |         |         |                        |                         | 730.537                     | 12/08/78          | .288                            |  |
|                                                                      |         |         |                        |                         | 730.537                     | 01/08/79          | .288                            |  |
|                                                                      |         |         |                        |                         | 730.541                     | 02/01/79          | .240                            |  |
|                                                                      |         |         |                        |                         | 730.548                     | 03/09/79          | .156                            |  |
|                                                                      |         |         |                        |                         | 730.546                     | 04/03/79          | .180                            |  |
|                                                                      |         |         |                        |                         | 730.540                     | 05/03/79          | . 252                           |  |
|                                                                      |         |         |                        |                         | 730.538                     | 06/06/79          | .276                            |  |
|                                                                      |         |         |                        |                         | 730.516                     | 07/09/79          | .540                            |  |
|                                                                      |         |         |                        |                         | 730.531                     | 08/08/79          | .360                            |  |
|                                                                      |         |         |                        |                         | 730.528                     | 09/05/79          | .396                            |  |
|                                                                      |         |         |                        |                         | 730.532                     | 10/02/79          | - 346                           |  |
|                                                                      |         |         |                        |                         | 730.536                     | 11/07/79          | .300                            |  |
|                                                                      |         |         |                        |                         | 730.531                     | 12/16/79          | .360                            |  |
|                                                                      |         |         |                        |                         | 730.537                     | 01/02/80          | .288                            |  |
| T2                                                                   | .000    | .000    | 82                     | 725.728                 | 725.728                     | 08/02/77          | .000                            |  |
|                                                                      |         |         |                        |                         | 725.729                     | 09/01/77          | 012                             |  |
|                                                                      |         |         |                        |                         | 725,726                     | 10/01/77          | .024                            |  |
|                                                                      |         |         |                        |                         | 725.727                     | 11/01/77          | .012                            |  |
|                                                                      |         |         |                        |                         | 725.719                     | 12/06/77          | .108                            |  |
|                                                                      |         |         |                        |                         | 725.724                     | 01/05/78          | .048                            |  |
|                                                                      |         |         |                        |                         | /25./16                     | 04/04/78          | .144                            |  |
|                                                                      |         |         |                        |                         | 725.710                     | 05/04/78          | .216                            |  |
|                                                                      |         |         |                        |                         | 725.709                     | 00/05/78          | .228                            |  |
|                                                                      |         |         |                        |                         | 725.708                     | 0//06//8          | .240                            |  |
|                                                                      |         |         |                        |                         | /25.699                     | 08/03/78          | .348                            |  |

FIGURE 2.5.4-49 TYPICAL SETTLEMENT MONITORING DATA REPORT BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT





 $\widehat{}$ 



 $\widehat{}$ 

-

(0-87) (0-95) (0-88) (c-104) (P-64) ( 0-89 ) T-62) 740 🖵 C 8 SWS LINES 720 700 FEET 00000 v -00000000 020000 ELEVATION 680 660 640 620 L

,

1

C



4

720

Ш

EVATION-F

Ш

680



SUBSURFACE PROFILE K-K" BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT






 $\sim$ 

SELECT GRANULAR BACKFILL



BEDROCK

UNCONTROLLED FILL-SILTY CLAY-GRAVELLY CLAY, SANDY CLAY, CLAYEY SAND

NORMAL GROUNDWATER LEVEL

NOTES

1. LOCATION OF SECTION IS SHOWN ON FIGURE 2.5.4-16. 2. N - STANDARD PENETRATION TEST BLOW COUNT (BLOWS/FT.).

3. \* - INDICATES USE OF 300LB, HAMMER.

4 CIRCULATING WATER LINES IN CONCRETE ENCASEMENT AT THIS LOCATION ONLY.

20 SCALE-FEET

> FIGURE 2.5.4-54 SUBSURFACE PROFILE L-L" BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



-

 $\sim$ 





SLOPE STABILITY SECTION A-A RIVERWARD SLOPE ANALYSIS BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



 $\sim$ 

 $\overline{}$ 



**- 680** 



LEGEND

DENSIFICATION FOR SERVICE WATER SYSTEM PIPELINES BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT





| <b>60</b> % | GRAVELLY SAND-SANDY GRAVEL<br>Some Silty Sand-Sand                     | NOTE<br>LOCATION OF SECTION IS SHOWN |
|-------------|------------------------------------------------------------------------|--------------------------------------|
|             | SELECT GRANULAR BACKFILL                                               | ON FIGURE 2.5.4-16                   |
|             | SILTY CLAY-SOME SILTY SAND,<br>Sandy Clay                              |                                      |
| 你您          | UNCONTROLLED FILL-SILTY CLAY-GRAVELLY<br>CLAY, SANDY CLAY, CLAYEY SAND |                                      |
|             | BEDROCK                                                                |                                      |
| <u> </u>    | NORMAL GROUNDWATER LEVEL                                               |                                      |

# FIGURE 2.5.4-60 SUBSURFACE PROFILE N-N' BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



\_\_\_\_

|                                               | ٦ | 730          |  |
|-----------------------------------------------|---|--------------|--|
|                                               | 1 | 720          |  |
|                                               | 1 | 710          |  |
|                                               | 1 | 700          |  |
|                                               |   | 690 LL<br>1  |  |
|                                               | - | 680 <b>C</b> |  |
|                                               | - | 670 L        |  |
|                                               | _ | 660          |  |
| နိုင်ငံခွင့်<br>တို့လိုင်ငံ<br>တို့လိုလ်နှင့် | _ | 650          |  |
|                                               | - | 640          |  |
| 02 20 20 20 20<br>20 2 2 0 0 0 0 0 0 0 0 0    | _ | 630          |  |
|                                               |   | 620          |  |









ZONE SUBJECT TO LIQUEFACTION



|              | SOIL PROPERTIES               |                    |                                         |                               |                           |  |  |  |
|--------------|-------------------------------|--------------------|-----------------------------------------|-------------------------------|---------------------------|--|--|--|
| SOIL<br>UNIT | TOTAL UNIT<br>WEIGHT<br>γ,pcf | COHESION<br>C, psf | FRICTION<br>UNDRAINED<br>\$\Phi DEGREES | IANGLE<br>DRAINED<br>¢DEGREES | SOIL DESCRIPTION          |  |  |  |
| 1            | 120                           | 0                  | 17                                      | 25                            | SILTY SAND - SANDY SILT   |  |  |  |
| 2            | 140                           | 0                  | 36                                      | 36                            | COMPACTED FILL            |  |  |  |
| 3            | 136                           | 0                  | 30                                      | 30                            | SAND AND GRAVEL           |  |  |  |
| 4            | 140                           | 0                  | 36                                      | 36                            | DENSIFIED SAND AND GRAVEL |  |  |  |
| 5            | 120                           | 0                  | 7*                                      | 25                            | LOOSE SILTY SAND          |  |  |  |

 $\phi$  EQUALS O° FOR POST EARTHQUAKE CASE



FIGURE 2.5.4-61 MAIN INTAKE CHANNEL SLOPE STABILITY SECTION 2-2 BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT















| 736        | LAYER<br>No. | LAYER<br>THICKNESS<br>(FT) | UNIT*<br>WEIGHT<br>(PCF) | SHEAR WAVE<br>VELOCITY **<br>(FT./SEC) |                       |
|------------|--------------|----------------------------|--------------------------|----------------------------------------|-----------------------|
| 7 33       | 1            | 10                         | 125                      | 600                                    |                       |
| 715        | 2            | 10                         | 125                      | 800                                    |                       |
| 715        | 3            | 10                         | 125                      | 950                                    |                       |
| 705        | 4            | 10                         | 125                      | 950                                    |                       |
| 695        | 5            | 5                          | 125                      | 1100                                   |                       |
|            | 6            | 5                          | 136                      | 1100                                   |                       |
| ш 685<br>L | 7            | 10                         | 136                      | 1100                                   | SAND<br>C &<br>GRAVEL |
| VIION C    | 8            | 10                         | 136                      | 1100                                   |                       |
|            | 9            | 7.5                        | 136                      | 1200                                   |                       |
| ш 650 L    | 10           | 7.5                        | 136                      | 1200                                   |                       |
| 000        | 11           | 10                         | 136                      | 1200                                   |                       |
| 640        | 12           | 10                         | 136                      | 1200                                   |                       |
| 630 -      | 13           | 10                         | 136                      | 1200                                   |                       |
| 620 L      | 4            | HALF<br>Space              | 160                      | 5000                                   |                       |
|            |              |                            |                          |                                        | 4                     |

#### NOTES

\* UNIT WEIGHT FROM BVPS-2 FSAR SECTION 2.5.4

\*\* SHEAR WAVE VELOCITY FROM FIGURE 6-2 (SWEC 1984) IN SITU: NATURAL FREQUENCY = 2.3 Hz

> FIGURE 2.5.4-69 SOIL MODEL-FREE FIELD BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

|            |       | LAYER<br>No. | Nı | LAYER<br>THICKNESS<br>(FT) | UNIT<br>WEIGHT<br>(PCF) | SHEAR WAVE<br>VELOCITY<br>(FT/SEC) |      |                 |
|------------|-------|--------------|----|----------------------------|-------------------------|------------------------------------|------|-----------------|
| <b>L</b> - | 675   | ſ            | _  | 16.75                      | 181                     | 670                                |      |                 |
| N - FEE    | 58.25 | 2            | _  | 16.75                      | 181                     | 670                                |      | PSEUDO-<br>SOIL |
| TIC        | 641.5 | 3            | 7  | 6.5                        | 136                     | 675                                |      |                 |
| E VA       | 600   | 4            | 18 | 8                          | 136                     | 879                                |      |                 |
| Ш          | 021   | 5            | 18 | 7                          | 136                     | 915                                |      |                 |
|            | 620   | <br>6        | -  | HALF<br>SPACE              | 160                     | 5000                               | - 55 | ROCK            |

FIGURE 2.5.4-70 SOIL MODEL - INTAKE STRUCTURE BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT



BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT

#### 2.5.5 Slope Stability

Both static and dynamic stability analyses of the riverward slope involving the service water pipelines leading to the intake structure were performed and are described in DLC (1976). The factors of safety were found to be acceptable.

The dynamic stability analysis of this slope was supplemented using more conservative seismic coefficients as indicated below. Additional failure surfaces through the silty clay layer in the soil profile were also considered. The results are presented in Figure 2.5.4-57 and were acceptable. (Refer to Figure 2.5.4-16 for location of section.)

Two methods of analysis were employed: the Simplified Bishop method and the Morgenstern-Price method. The Simplied Bishop method assumes a circular arc failure surface while the Morgenstern-Price method allows for an arbitrary shaped failure mass, which, in this case, was assumed to be a sliding wedge with straight line failure surfaces. The stability analyses were performed using the computer program LEASE II (SWEC 1980). LEASE II uses a pseudo-static approach to dynamic stability analysis in which a constant force is applied to each slice and is computed as the weight of the slice multiplied by a seismic coefficient. The horizontal seismic coefficient was taken as 0.125, corresponding to the ground surface acceleration for the SSE; the vertical seismic coefficient was taken as 0.083. This analysis was considered conservative since the applied pseudostatic force was constant and no consideration was given to the variation of acceleration with time, direction, or with depth in the soil profile.

The analysis of the intake channel slopes is discussed in Section 2.5.4.8.

The analysis of the stability of the slopes in the vicinity of the emergency outfall structure is fully described in Appendix 2.5E.

2.5.5.1 Reference for Section 2.5.5

Duquesne Light Company 1976. Report on the Soil Densification Program. Beaver Valley Power Station - Unit 2. Prepared by SWEC, Boston, Mass.

Stone & Webster Engineering Corporation (SWEC) 1980. Slope Stability Analysis (LEASE II), GT-108.

### 2.5.6 Embankments and Dams

Seismic Category I embankments and dams are not utilized at Beaver Valley Power Station - Unit 2.

# APPENDIX 2.5A

## OHIO RIVER ELEVATIONS

# AND

# PIEZOMETER DATA

# BEAVER VALLEY POWER STATION

# APPENDIX 2.5A-1

### LIST OF FIGURES

Figure No. <u>Title</u>

| 2.5A-1  | OHIO RIVER | ELEVATIO | )N, 19  | 77      |       |     |
|---------|------------|----------|---------|---------|-------|-----|
| 2.5A-2  | OHIO RIVER | ELEVATIO | N, 19   | 78      |       |     |
| 2.5A-3  | OHIO RIVER | ELEVATIO | N, 19   | 79      |       |     |
| 2.5A-4  | OHIO RIVER | ELEVATIO | N, 19   | 80      |       |     |
| 2.5A-5  | OHIO RIVER | ELEVATIO | N, 19   | 81      |       |     |
| 2.5A-6  | PIEZOMETER | DATA, 19 | , AN    | D P- AN | ID P- |     |
| 2.5A-7  | PIEZOMETER | DATA, 19 | 978, A  | ND P-1  | AND   | P-2 |
| 2.5A-8  | PIEZOMETER | DATA, 19 | 979, A  | ND P-1  | AND   | P-2 |
| 2.5A-9  | PIEZOMETER | DATA, 19 | 980, A  | ND P-1  | AND   | P-2 |
| 2.5A-10 | PIEZOMETER | DATA, 19 | 981, A  | ND P-1  | AND   | P-2 |
| 2.5A-11 | PIEZOMETER | DATA, 19 | 977, Al | ND P-3  | AND   | P-4 |
| 2.5A-12 | PIEZOMETER | DATA, 19 | 978, A  | ND P-3  | AND   | P-4 |
| 2.5A-13 | PIEZOMETER | DATA, 19 | 979, Al | ND P-3  | AND   | P-4 |
| 2.5A-14 | PIEZOMETER | DATA, 19 | 980, A  | ND P-3  | AND   | P-4 |
| 2.5A-15 | PIEZOMETER | DATA, 19 | 981, A  | ND P-3  | AND   | P-4 |
| 2.5A-16 | PIEZOMETER | DATA, 19 | 977, Al | ND P-6  | AND   | P-7 |
| 2.5A-17 | PIEZOMETER | DATA, 19 | 978, A  | ND P-6  | AND   | P-7 |
| 2.5A-18 | PIEZOMETER | DATA, 19 | 979, Al | ND P-6  | AND   | P-7 |
| 2.5A-19 | PIEZOMETER | DATA, 19 | 980, A  | ND P-6  | AND   | P-7 |
| 2.5A-20 | PIEZOMETER | DATA, 19 | 980, A  | ND P-6  | AND   | P-7 |

Tables for Appendix 2.5A

### TABLE 2.5A-1

#### PIEZOMETER INSTALLATION DATA

| Piezometer<br>No. | Ground Surface<br><u>Elevation (ft)</u> | Tip<br><u>Elevation</u><br><u>(ft)</u> | a<br><u>(ft)</u> | b<br><u>(ft)</u> |
|-------------------|-----------------------------------------|----------------------------------------|------------------|------------------|
| P-1               | 730.9                                   | 646.4                                  | 85.5             | 68.0             |
| P-2               | 729.6                                   | 646.9                                  | 83.7             | 68.7             |
| P-3               | 728.2                                   | 645.2                                  | 84.0             | 70.0             |
| P-4               | 731.7                                   | 651.2                                  | 81.5             | 67.5             |
| P-6               | 705.8                                   | 647.1                                  | 59.7             | 44.2             |
| P-7               | 733.0                                   | 650.0                                  | 84.0             | 71.0             |











\_\_\_\_

•















PIEZOMETER DATA, 1978 BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT








\_\_\_\_









PIEZOMETER DATA, 1981 BEAVER VALLEY POWER STATION-UNIT 2 FINAL SAFETY ANALYSIS REPORT

APPENDIX 2.5B

## BORING LOGS

## BEAVER VALLEY POWER STATION

Tables for Appendix 2.5B

## TABLE 2.5B-1

## LIST OF BORING LOGS

| Boring<br>No. | Boring<br>No. | Boring<br>No. | Boring<br>No. | Boring<br>No. |
|---------------|---------------|---------------|---------------|---------------|
| 0.5.4         |               |               |               |               |
| 854           | TH - 1        | 5371          | SEO-1         | EOS-1         |
| 855           | TH-2          | 5381          | SEO-1A        | EOS-1A        |
| 901           | TH-3          | 5391          | SEO-2         | EOS-2         |
| 902           | TH-4          | 540T          | SEO-3         | EOS-3         |
| 903           | TH-5          | 541T          | SEO-4         | EOS-4         |
| 904           | TH-6          | 542T          | SEO-5         | EOS-4A        |
| 905           |               | 543T          |               | EOS-5         |
| 906           |               | 543A'I'       |               | EOS-6         |
| 907           |               | 544'1'        |               | EOS-7         |
| 907           |               | 5451          |               | EOS-7A        |
| 908           |               | 546'1'        |               | EOS-9         |
| 908           |               | 5471          |               | EOS-10        |
| 909           |               | 548'1'        |               |               |
| 910           |               | 5491          |               |               |
| 911           |               | 550T          |               |               |
| 912           |               | 551T          |               |               |
| 913           |               | 552T<br>552T  |               |               |
| 914           |               | 5531          |               |               |
| 915           |               | 5541          |               |               |
| 910           |               | 5551<br>EECT  |               |               |
|               |               | 5561          |               |               |
|               |               |               |               |               |
|               |               | 5501          |               |               |
|               |               | 560T          |               |               |
|               |               | 561T          |               |               |
|               |               | 562T          |               |               |
|               |               | 563T          |               |               |
|               |               | 564T          |               |               |
|               |               | 565T          |               |               |
|               |               | 566T          |               |               |
|               |               | 567T          |               |               |
|               |               | 568T          |               |               |
|               |               | 569T          |               |               |
|               |               | 570T          |               |               |
|               |               | 571T          |               |               |
|               |               | 572T          |               |               |
|               |               | 573T          |               |               |
|               |               | 574T          |               |               |
|               |               | 575T          |               |               |
|               |               | 576T          |               |               |
|               |               | 577T          |               |               |

| ITE           | В                           | EAVER VALLEY                 | POWER STATIC                            |                               | J.O. NOBORING NO854                                                                                                                     |
|---------------|-----------------------------|------------------------------|-----------------------------------------|-------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|
| YPE OF        | BORING                      | SPLIT SPOON<br>JULY 23, 19   | LOCATION                                | V DRIL                        | LED BY AMERICAN LOGGED BY D.F.P.                                                                                                        |
| UMMAR         | Y OF BO                     | DRING                        | -                                       |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
| ET .          | H                           | OVERALL<br>WEATHERING<br>AND | SAMPLE                                  | UH C                          | SOIL OR ROCK DESCRIPTION                                                                                                                |
| E             | DEP                         | RQD<br>0 25 50 75 100        | TYPE                                    | A Q                           | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOU OR JOINTING, BEDDING AND FAULTING AND TEXTURE                        |
|               |                             |                              |                                         | ပ                             | DESCRIPTIONS                                                                                                                            |
| 715.8         | r - 1                       |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              | 19 1                                    |                               | CLAYEY SILT, MODERATELY TO HIGHLY PLASTIC ,3-7% VERY FINE SAND, WI ROOTS, DARK BROWN.                                                   |
|               |                             |                              |                                         |                               |                                                                                                                                         |
| 710           | <b>^</b> -                  |                              | 32 2                                    |                               | SAND, UNIFORM, FINE/VERY FINE, 2-3% MEDIUM SAND, 4-8% SLICHTLY TO                                                                       |
|               | 10                          |                              |                                         |                               | OF CLAYEY SILT CONTAINING SOME ROCTS.                                                                                                   |
|               | 10 -                        |                              | 20 3                                    |                               | SAND, UNIFORM, FINE 5-8% MODERATELY PLASTIC FINES, LIGHT BROWN,                                                                         |
|               | T T                         |                              |                                         |                               | LESS THAN 1% GRAVEL TO 0.9 INCH MAXIMUM.<br>(SP)                                                                                        |
|               | /3<br>15                    |                              |                                         |                               |                                                                                                                                         |
| 700           |                             |                              | 23 4                                    |                               | (SP)                                                                                                                                    |
|               | 3                           |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              | 33 5                                    |                               | SAND, POCRLY GRADED, FINE TO COARSE, 5-10% MEDIUM AND COARSE SAND<br>4-6% GRAVEL TO 1.5 INCH MAXIMUM, 3-6% SLIGHTLY PLASTIC FINES, LIGH |
|               | 35 -                        |                              |                                         |                               | DECUN.<br>(SP)                                                                                                                          |
| 690           | <b>95</b>                   |                              | 52 6                                    |                               | SAND, POORLY GRADED, FINE TO COARSE, MOSTLY FINE AND MEDIUM, 5-10%                                                                      |
| ·             |                             |                              |                                         |                               | GRAVEL TO I.O INCH MAXIMUM, 4-8% SLIGHTLY TO MODERATELY PLASTIC<br>FINES, LICHT BROWN.<br>(SP)                                          |
|               |                             |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              | ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ |                               | END OF BORING AT 36.5                                                                                                                   |
|               | 35                          |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | =                           |                              |                                         |                               |                                                                                                                                         |
|               | _                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
|               | -                           |                              |                                         |                               |                                                                                                                                         |
|               |                             | 1                            |                                         |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
|               | -                           | <b>₽</b><br><b>−</b> ].      |                                         |                               |                                                                                                                                         |
|               |                             |                              |                                         |                               |                                                                                                                                         |
| 1. FIG<br>SOI | URES IN                     | BLOW OF RELE DENOTE TH       | COVERY CO                               | LUMN OF<br>OF BLOW            | POSITE<br>Is of A                                                                                                                       |
| A 2<br>FIG    | ים אח<br>" OD SA<br>URES SH | MPLE SPOON<br>IOWN OPPOSIT   | 30" REQU<br>12" OR TH<br>E ROCK CO      | IRED TO<br>E DISTA<br>RES DLA | DRIVE<br>NCE SHOWN.                                                                                                                     |
| THE<br>2. ₩2  | PERCEN<br>INDICAT           | T OF CORE A<br>ES LOCATION   | OF UNDIS                                | TURBED                        | SAMPLE. BORING LOG 854                                                                                                                  |
| ٥P            | INDICAT<br>WITH NO          | ES LOCATION<br>RECOVERY.     | OF SAMPL                                | ING ATT                       | EMPT BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                           |
| SUB<br>Num    | SCRIPT<br>BER.<br>INDICAT   | NEXT TO SYM                  | OR NATURA                               | ATES SA<br>Al grod            | MPLE SHIPPINGPORT, FRAMELIVANIA                                                                                                         |
| 3⊻_ `         |                             |                              | AT UNITORS                              |                               |                                                                                                                                         |

)

.

.

.

 $\sim$ 

|                           |                             |                                      |                                                                  |                                 | DIIQIII         | SH_1 OF 1                                                                                                                                         |
|---------------------------|-----------------------------|--------------------------------------|------------------------------------------------------------------|---------------------------------|-----------------|---------------------------------------------------------------------------------------------------------------------------------------------------|
| SITE<br>TYP<br>DAT<br>SUM | E<br>PE OF<br>TE DI<br>IMAR | BEAVE<br>BORING<br>RILLED<br>Y OF BI | R VALLEY POWER<br>S SPLIT SPOON<br>JULY 25.<br>ORING             | STATION<br>LOCATION             | DRIL            | J.O. NO. 12241 BORING NO. 855<br>GROUND ELEV. 694.4<br>LED BY AMERICAN LOGGE BY D.P.F.                                                            |
| ELEV.                     | FEET                        | DEPTH<br>FEET                        | OVERALL<br>WEATHERING<br>AND<br>RQD<br>0 25 50 75 100<br>1 1 1 1 | BLOWS<br>BLOWS<br>RECOV<br>TYPE | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHO<br>OH JOINTING BEDDING AND FAULTING<br>DESCRIPTIONS |
| 694.                      | 4.7                         |                                      |                                                                  | 1                               |                 |                                                                                                                                                   |
| 690                       |                             |                                      |                                                                  | 57 🔰                            |                 | SAND, UNIFORM, FINE 2-3% MEDIUM AND COARSE SAND, 4-6% SLIGHTLY<br>PLASTIC FINES, DARK BROWN WITH 3-5% GRAVEL TO 1.2 INCH MAXIMUM.<br>(SP)         |
|                           |                             | 10                                   |                                                                  | 6 2                             |                 | SILTY CLAY, MODERATELY PLASTIC, 2-3% VERY FINE SAND, MOTTLED,<br>LIGHT GRAY AND DARK BROWN, CONTAINING A TRACE OF MICA.<br>(CH)                   |
| 680                       | <u></u>                     | 15                                   |                                                                  | 8 3                             |                 | STIJLAR TO ABOVE.<br>(CH)                                                                                                                         |
|                           |                             | 20 -                                 | -                                                                | 5 4                             |                 | SIMILAR TO SS #2<br>(CH)                                                                                                                          |
| 670                       | ,                           | 25                                   |                                                                  | 3 5                             |                 | SPAILAR TO SS#2<br>(CH)                                                                                                                           |
|                           |                             | 30-                                  | 4                                                                |                                 | ,               |                                                                                                                                                   |

э.

1

and the second second

· · ·

)

|                 |                    |                                  | 2 🎽 6                  | 6                   | CH)           | ) SS#2  |                                       |                     |               |           |       |
|-----------------|--------------------|----------------------------------|------------------------|---------------------|---------------|---------|---------------------------------------|---------------------|---------------|-----------|-------|
|                 | _                  |                                  |                        | Ĩ                   |               |         |                                       |                     |               |           |       |
| 660             | _                  |                                  |                        |                     |               |         |                                       |                     |               |           | _     |
|                 | 35                 |                                  | 5 7                    | 6                   | ILTY CLAT     | , MODER | ATELY PLASTIC,                        | 3-5% VERY F         | INE SAND,     | DARK BRO  | . KI  |
|                 |                    |                                  |                        |                     | Сн)           |         | , , , , , , , , , , , , , , , , , , , |                     | ******        |           | -     |
|                 |                    |                                  | ľ                      |                     |               |         |                                       |                     |               |           | 1     |
|                 |                    |                                  |                        |                     | END OF BO     | RING AT | 36.5'                                 |                     |               |           | _     |
|                 |                    |                                  | 1                      |                     |               |         |                                       |                     |               |           |       |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           | -     |
|                 |                    |                                  |                        |                     |               |         |                                       |                     |               |           | -     |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           | 4     |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           | -4    |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           | 1     |
|                 |                    |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  | 1                                |                        |                     |               |         |                                       |                     |               |           | 1     |
|                 |                    |                                  |                        |                     |               |         |                                       |                     |               |           |       |
|                 | -                  | 4                                |                        |                     |               |         |                                       |                     |               |           | -     |
|                 |                    |                                  |                        |                     |               |         |                                       |                     |               |           | コ     |
|                 | -                  | 4                                |                        |                     |               |         |                                       |                     |               |           |       |
| 1               | -                  | 4                                |                        |                     |               |         |                                       |                     |               |           | _     |
| Ì               | _                  | ]                                |                        |                     |               |         |                                       |                     |               |           | _     |
|                 | -                  | 1                                |                        |                     |               |         |                                       |                     |               |           | -     |
|                 |                    | 4                                |                        |                     |               |         |                                       |                     |               |           | 7     |
|                 | -                  | -                                |                        |                     |               |         |                                       |                     |               |           | -     |
|                 | -                  |                                  |                        |                     |               |         |                                       |                     |               |           | -1    |
| L               | <u> </u>           | L                                |                        |                     | <u>L</u>      | -       |                                       |                     |               |           |       |
| 1. FIG          | DRES II            | BLOW OR REC                      | OVERY COL              | UMN OPP             | OSITE         |         |                                       |                     |               |           |       |
| S011<br>140     | L SAMPI<br>LB HAI  | LE DENOTE THE                    | NUMBER OF<br>30" REOUT | F BLOWS<br>RED TO   | OF A<br>DRIVE |         |                                       |                     |               |           | Į     |
| A 2             | " OD S             | MPLE SPOON 1                     | 2" OR THE              | DISTAN              | CE SHOW       | 1.      |                                       |                     |               |           |       |
| THE             | PERCEI             | VT OF CORE RE                    | COVERED.               | es deno.            | 1 M           |         |                                       |                     |               |           |       |
| 2. 12           | INDICA'<br>INDICA' | TES LOCATION                     | OF UNDIST<br>OF SPLIT- | JRBED S.<br>Sdoon S | AMPLE.        |         |                                       | BORING LO           | <u>G 855</u>  |           | -     |
|                 | INDICA             | ES LOCATION                      | OF SAMPLI              | NG '″ L             | 4PT           | ,┝──┤   | BEAVER V                              | ALLEY POWER         | STATION -     | UNIT NO.1 | 1     |
| SUB             | SCRIPT             | NEXT TO SYMB                     | OL INDICA              | TES SAM             | PLE           |         | SH                                    | IPPINGPORT,         | PENNSYLVA     | NIA       | ļ     |
|                 | BER.<br>[NDICA]    | TES LOCATION                     | OF NATURA              | L GROUN             | D WATER       | 2       |                                       | DUQUESNE LIG        | HT COMPAN     | Y         |       |
| 4. ROD          | CABLE.             | OUALITY DES                      | IGNATION.              |                     |               | WIII    | STONE & WE                            | BSTER ENG           | NEERING       | CORPOR    | ATION |
| 5. 11<br>6. DAT | UNDICAT            | TES DEPTH & LI<br>MEAN SEA LEVEL | ENGTH OF I             | NX CORII            | NG RUN        | 111     | À                                     | 122 <u>4</u> 1 - GS | <b>K -</b> 12 |           |       |

| 676       Better Build Dody Database       Dody Build Dody Database         677       Better Build Dody Database       Dody Build Dody Build Dody Database         678       Better Build Dody                                                                                                                                                                                                                                                | DUQUESNE LIGHT COMPANY SH1 OF 1 |                                                                                    |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------|------------------------------------------------------------------------------------|--------------------------------|---------------|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
| DATE Ded(12) (2003) 11/2/2         OP(12) (2003) 11/2/2         OP(12) (2003) 11/2/2         OP(12) (2003) 11/2/2           2         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1 </td <td>SITE<br/>TYPE OF</td> <td colspan="10">TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 677.21</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | SITE<br>TYPE OF                 | TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 677.21 |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| 2       3       3       3       3       3       3       3       3       3       3       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1 <th1< th=""> <th1< th=""> <th1< th=""></th1<></th1<></th1<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | SUMMARY OF BORING               |                                                                                    |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| Solution restances of the second seco                                                                                                               |                                 |                                                                                    |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| GE         GE <thge< th="">         GE         GE         GE<!--</td--><td>EV.<br/>ET</td><td>РТН<br/>ЕТ</td><td>OVERALL<br/>WEATHERING</td><td>8AM</td><td></td><td>SOIL OR ROCK DESCRIPTION</td></thge<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | EV.<br>ET                       | РТН<br>ЕТ                                                                          | OVERALL<br>WEATHERING          | 8AM           |                         | SOIL OR ROCK DESCRIPTION                                                                                                               |  |  |  |  |  |
| 677.2       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1 <td>EL<br/>FE</td> <td>DE</td> <td>RQU<br/>0 25 50 75 100</td> <td></td> <td>GRA<br/>GRA</td> <td>FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br/>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br/>DESCRIPTIONS</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | EL<br>FE                        | DE                                                                                 | RQU<br>0 25 50 75 100          |               | GRA<br>GRA              | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |  |  |
| Image: Second                                                                                                                                       | 677.2                           |                                                                                    |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| 2       4       2         600       3       3         10       3       3         10       3       3         11       3       3         12       3       3         13       3       3         14       3       3         15       2       1         15       2       1         15       2       1         15       2       1         15       2       1         15       2       1         15       2       1         15       2       1         160       2       1         160       2       1         160       2       1         17       2       1         18       2       1         19       2       1         10       1       1         11       1       1         12       7       2       1         14       12       12       12         15       3       2       10       2         160       12 </td <td>*</td> <td></td> <td></td> <td>1</td> <td>1</td> <td>ORGANIC SILT, MODERATELY PLASTIC, 5-10% FINE SAND, VERY SOFT,<br/>DARK EROWN, CONTAINS ROOT OR TWIG FRAGMENTS, DAMP.<br/>(OL)</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | *                               |                                                                                    |                                | 1             | 1                       | ORGANIC SILT, MODERATELY PLASTIC, 5-10% FINE SAND, VERY SOFT,<br>DARK EROWN, CONTAINS ROOT OR TWIG FRAGMENTS, DAMP.<br>(OL)            |  |  |  |  |  |
| 670       10       5       2         10       10       5       2         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       3         10       5       5       5         10       5       5       5         10       5       5       5         11       5       5       5         11       5       5       5         12       7       5       5         13       12       7       5       5         14       5       5       5       5         14       5       5       5       5         14       5       5       5       5       5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                 | -<br>5 -                                                                           |                                |               |                         | -                                                                                                                                      |  |  |  |  |  |
| 10       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         15       2       3         160       30       30         12       7       30         12       7       30         12       7       30       30         12       7       30       30         12       7       30       30       30         12       7       30       30       30       30         12       7       30       30       30       30         14       7       30       30       30       30       30         12       7       30       30       30       30       30                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <b>67</b> 0                     |                                                                                    |                                | 5             | 2                       | SILTY SAND, UNIFORM, FINE TO VERY FINE, 10-15% SLIGHTLY PLASTIC<br>ORGANIC FINES, DAMP, MEDIUM TO DARK EROWN.<br>(SM)                  |  |  |  |  |  |
| 13     2     2       14     2     2       15     2     2       160     2     2       15     2     2       160     2     2       160     2     2       17     2     2       18     2     2       19     2     2       10     2     2       11     16     2       11     16     2       11     16     2       12     11     16       11     16     2       12     11     16       14     17     2       15     2     2       16     11     16       16     2     11       16     2     11       17     2     10       18     2     10       19     2     10       10     11     16       11     16     2       12     17     2       14     17     2       15     2     10       16     2     10       16     2     10       16     2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                 | 10 —                                                                               |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| 15       2       2       2         660       20       10       20       10         660       20       11       66       11       66         20       11       66       11       66       11       66         20       11       66       11       66       11       66         20       11       66       11       66       11       66         30       12       11       66       11       66       11       66         30       12       11       66       11       66       11       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10       10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                 |                                                                                    |                                | <b>`</b> '    | - 3                     | PLASTIC ORGANIC FINES, MOIST, MEDIUM TO DARK BROWN, OCCASIONAL<br>PEBELES TO 0.75 INCHES.                                              |  |  |  |  |  |
| 20       20       20         20       20       21.27.9.00, PUPCHA, FIRE TO VERT FIRE, 10-155 KUTLASTIC FIRES, 107, 107, 107, 107, 108, 100, 100, 107, 107, 108, 100, 100, 100, 100, 100, 100, 100                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 660                             | 15 —<br>—<br>—                                                                     |                                | 2             |                         | SILTY SAND, UNIFORM, FINE TO VERY FINE, 15-20% SLIGHTLY PLASTIC<br>FINES, MOIST TO WET, MEDIUM BROWN.<br>(SM)                          |  |  |  |  |  |
| 40     5       60     11       60     11       60     11       60     11       60     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       7     12       14     12       14     13       14     14       14     14       15     14       160     14       161     14       162     14       163     14       164     14       165     14       165     14       166     14       167     14       168     14       169     14       160     14       161     14       162     14       163     14       164     14       165                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 600                             | 20 1                                                                               |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| 25       11       6         500       30       12       7         30       12       7       NO RECOVERT CLAY, SLICETLY PLACTIC, 20-304 VENT FIRE SAMD, FIRM, NEULON         640       12       7       NO RECOVERT         55       32       7       NO RECOVERT         640       14       7       NO RECOVERT       COLOREST TO FIRE, 1-36 ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 1-37 ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, NOIST, MEDINE TO RECOVERT COLOREST TO FIRE, 3-5% ENDELASTIC FIRES, 3-5% ENDELASTIC FIRES, 3-5% ENDELASTIC FIRES, 3-5% ENDELASTIC FIREST COLOREST TO FIREST COLOREST TO FIREST COLOREST FIRESTOR FIREST COLOREST TO FIREST CLARAST TO RECOVER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                 | -                                                                                  |                                | WOH           | 5                       | (SILTY SAND, UNIFORM, FINE TO VERY FINE, 10-15% NONPLASTIC FINES, WET, MEDIUM GRAY.<br>(SM)                                            |  |  |  |  |  |
| 650       30       12       7       NO REDOVERT         30       12       7       NO REDOVERT         31       33       78       SILTY SAME, NOSTLY UNIFORM, FINE, 5-10% NONFLASTIC FINES, NOIST, NEDIMEDIM GROW, ONE 1 INCH PERCH.         40       14       79       SILTY SAME, NOSTLY UNIFORM, ONE 1 INCH PERCH.         40       14       79       SILTY SAME, NOSTLY UNIFORM, ONE 1 INCH PERCH.         40       14       79       SILTY SAME, NOSTLY UNIFORM, ONE 1 INCH PERCH.         45       32       10       SILTY SAME, NOSTLY UNIFORM, UNIT COMPLET TO FINES, NOIST, NEDIMERTARY, NET COMPLET TO FINES, NOIST, NEDIMERTARY, NET COMPLET TO FINE, 3-5% KONFLASTIC FINES, NOIST, NEDIMERTARY, NET COMPLET TO FINE, 3-5% KONFLASTIC FINES, SAME AND TO MADIM GAY.         50       52       52       53         50       55       52       52         50       55       52       52         50       52       53       54         50       55       52       52         50       55       52       52         50       52       53       54         55       52       52       55         55       52       52       53         56       52       54       55 </td <td></td> <td>25 -</td> <td></td> <td></td> <td></td> <td>SANDY CLAY SLIGHTLY PLASTIC, 20-30% VERY FINE SAND, FIRM, MEDIUM</td>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                 | 25 -                                                                               |                                |               |                         | SANDY CLAY SLIGHTLY PLASTIC, 20-30% VERY FINE SAND, FIRM, MEDIUM                                                                       |  |  |  |  |  |
| 30       12       73         35       33       78         36       33       78         37       78       31         38       33       78         39       78       31         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         40       14       79         41       79       100         50       22       11         50       22       11         50       22       11         50       22       11         50       22       12         50       22       12         50       22       12                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 650                             |                                                                                    |                                | **            |                         | GRAY, SOME SMALL PEBBLES; SWET.<br>(CL)                                                                                                |  |  |  |  |  |
| 35       33       78         360       33       78         40       14       79         40       14       79         40       14       79         40       14       79         41       79       32         42       14       79         45       32       10         45       32       10         45       32       10         50       25       11         50       25       11         50       25       11         51       52       52         52       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         56       52       52         55       52       52         56       52       52         56       52       52         57       52       52         58       54       54                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                 | 30 -                                                                               |                                | 12            | $\overline{7}$          | NO RECOVERY                                                                                                                            |  |  |  |  |  |
| 35       33       8       SILTY SAND, WOTLY DEPORT PIRE 5-10% NORLASTIC FIRES, WET, MEDIA         40       14       9       SALD, WEITORN, ONE 1 HED PERLE.       5-10% NORLASTIC FIRES, NET, MEDIA         630       14       9       SALD, WEITORN, ONE 1 HED PERLE.       100 FIRES, NET, MEDIA         630       14       9       SALD, WEITORN, ORE 1 HED PERLE.       100 FIRES, NET, MEDIA         630       14       9       SALD, WEITORN, ORE 1 HED PERLE.       100 FIRES, NET, MEDIA         630       50       25       11       SALD, MOSTLUTUTION, VERT COARSE TO FIRE, 3-5% NORFLASTIC FIRES, MEDIA         630       50       25       11       SALD, MOSTLUTUTION, VERT COARSE TO FIRE, 3-5% NORFLASTIC FIRES, MEDIA         640       50       25       11       SALD, MOSTLUTUTION, VERT COARSE TO FIRE, 3-5% NORFLASTIC FIRES, MEDIA         650       50       25       11       SALD, MOSTLUTUTION OF NO KADIN GAT.       100 FIRE, 3-5% NORFLASTIC FIRES, MEDIA         660       25       11       SALE ALLY COARSE TO FIRE, 3-5% NORFLASTIC FIRES, MEDIA       100 FIRES, 100 FIRES                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                 |                                                                                    |                                |               |                         | -                                                                                                                                      |  |  |  |  |  |
| 640       40       14       9       MED THM SEMAN, ONE I INCH PERSAC.         640       14       9       14       9         630       14       9       14       9         630       14       9       14       9         630       14       9       14       9         630       14       9       14       9         630       14       9       14       9         630       14       9       14       14         630       14       9       14       14         630       14       14       14       14         630       14       14       14       14         14       14       14       14       14         15       14       14       14       14         14       15       14       14       14         15       14       14       14       14         15       14       14       14       14         15       14       14       14       14         15       14       14       14       14         15       14                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                 | 35 <del>-</del>                                                                    |                                | 33            | 8                       | SILTY SAND, MOSTLY UNIFORM, FINE, 5-10% NONPLASTIC FINES, WET,                                                                         |  |  |  |  |  |
| 40       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       14       15       15       15       15       16       16       16       16       16       16       16       16       16       16       16       16       16       16       16       16       16 <td< td=""><td>640</td><td>_</td><td></td><td></td><td></td><td>(SM)</td></td<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 640                             | _                                                                                  |                                |               |                         | (SM)                                                                                                                                   |  |  |  |  |  |
| 45 - 32 10<br>50 - 25 11<br>50 - 25 11<br>51 - 52 - 52 10<br>52 - 52 - 52 10<br>55 - 52 - 52 10<br>50 - 52 - 52 - 52 10<br>50 - 52 - 52 - 52 - 52 - 52 - 52 - 52 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                 | 40 -                                                                               |                                | 14            | 9                       | SAND, UNIFORM, COARSE TO FINE, 1-3% NONPLASTIC FINES, MOIST, MEDIUM                                                                    |  |  |  |  |  |
| 42       32       10         630       32       10         50       50       25         50       25       11         50       25       11         51       52       52         52       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         55       52       52         56       52       52         55       52       52         56       52       52         56       52       52         56       52       53         56       52       54         56       55       55         57       56       57         58       58       58         59       50       57         50       58       58         50       58       59         50       58       59                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                 |                                                                                    |                                |               |                         | (SP)                                                                                                                                   |  |  |  |  |  |
| <ul> <li>(SP)</li> <li< td=""><td>630</td><td>47 -</td><td></td><td>32</td><td>10</td><td></td></li<></ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 630                             | 47 -                                                                               |                                | 32            | 10                      |                                                                                                                                        |  |  |  |  |  |
| 25       11       3A.D., UNIFORM, VENT COARSE TO FINE, 3-5% NONPLASTIC FINES, SATURATED, NEDIUM TO DARK GRAT.         55       62       2       SATURATED, NEDIUM TO DARK GRAT.         56       62       2       SATURATED, NEDIUM TO DARK GRAT.         56       62       2       SAND, SANE AS ABOVE.         56       62       72       SAND, SANE AS ABOVE.         560       62       72       SANE AS ABOVE.         501L SANFLE BOOK 12       70       60       70         501L SANFLE BOOK 12       70       70       70       70         501L SANFLE BOOK 12       70       70       70       70                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 0,0                             |                                                                                    |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| 55       62       72         560       62       72         560       62       70P OF ROCK AT 58.0'         600       GRAT SHALR         700       GRAT SHALR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                 | 111                                                                                |                                | 25            | iı                      | SAND, UNIFORM, VERY COARSE TO FINE, 3-5% NONPLASTIC FINES,<br>SATURATED, MEDIUM TO DARK GRAY.                                          |  |  |  |  |  |
| 620       TOP OF NOCK AT 58.0'         60       GRAY SHALR         8MD OF BORLING AT 60.0'         9       GRAY SHALR         9       GRAY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                 | -<br>55 -                                                                          |                                | 62            |                         | SAND. SAME AS ABOVE.                                                                                                                   |  |  |  |  |  |
| 60       ORAT SHALE         Image: Construct of the state of the stat                                                                                                                                                                                 | 620                             |                                                                                    |                                |               |                         | TOP OF ROCK AT 58.01                                                                                                                   |  |  |  |  |  |
| <ul> <li>I. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br/>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br/>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br/>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br/>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br/>THE PERCENT OF CORE RECOVERED.</li> <li>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br/>[]//INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br/>[]//INDICATES LOCATION OF SAMPLING ATTEMPT<br/>WITH NO RECOVERY.</li> <li>3. 4 INDICATES LOCATION OF NATURAL GROUND WATER<br/>TABLE.</li> <li>4. ROD - ROCK QUALITY DESIGNATION.</li> <li>5. LI INDICATES DEPTH &amp; LEMOTH OF NX CORING RUN</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                 |                                                                                    |                                | <b> </b>      |                         | GRAY SHALE -                                                                                                                           |  |  |  |  |  |
| <ul> <li>1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br/>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br/>140 LB HAMMER FALLING 30° REQUIRED TO DRIVE<br/>A 2° OD SAMPLE SPOON 12° OR THE DISTANCE SHOWN.<br/>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br/>THE PERCENT OF CORE RECOVERED.</li> <li>2. WIDICATES LOCATION OF UNDISTURBED SAMPLE.<br/>VIDICATES LOCATION OF SAMPLING ATTEMPT<br/>WITH NO RECOVERY.<br/>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br/>NUMBER.</li> <li>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATES<br/>TABLE.<br/>A ROD - ROCK QUALITY DESIGNATION.<br/>5. UL INDICATES DEPTH &amp; LENGTH OF NX CORING RUN<br/>5. DATUM IS MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                 |                                                                                    |                                |               |                         | END OF DURING AT OU.U'                                                                                                                 |  |  |  |  |  |
| <ul> <li>1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br/>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br/>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br/>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br/>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br/>THE PERCENT OF CORE RECOVERED.</li> <li>2. #2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br/>[][7 INDICATES LOCATION OF SAMPLING ATTEMPT<br/>WITH NO RECOVERY.<br/>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br/>NUMBER.</li> <li>3. # INDICATES LOCATION OF NATURAL GROUND WATER<br/>TABLE.</li> <li>4. RGD - ROCK QUALITY DESIGNATION.</li> <li>5. [] INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>6. DATUM IS MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                 | -                                                                                  |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| <ul> <li>1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br/>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br/>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br/>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br/>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br/>THE PERCENT OF CORE RECOVERED.</li> <li>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br/>[]/INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br/>[]/INDICATES LOCATION OF SAMPLING ATTEMPT<br/>WITH NO RECOVERY.<br/>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br/>NUMBER.</li> <li>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATEF<br/>TABLE.</li> <li>4. RQD - ROCK QUALITY DESIGNATION.</li> <li>5. []. INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>6. DATUM IS MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                 |                                                                                    |                                |               | ~                       |                                                                                                                                        |  |  |  |  |  |
| <ul> <li>1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br/>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br/>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br/>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br/>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br/>THE PERCENT OF CORE RECOVERED.</li> <li>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br/>[]/INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br/>[]/INDICATES LOCATION OF SAMPLING ATTEMPT<br/>WITH NO RECOVERY.<br/>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br/>NUMBER.</li> <li>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATER TABLE.</li> <li>4. RQD - ROCK QUALITY DESIGNATION.</li> <li>5. []. INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>6. DATUM IS MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                 | -                                                                                  |                                |               |                         |                                                                                                                                        |  |  |  |  |  |
| A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br>THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>F 6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>NUMBER.<br>3. 4 INDICATES LOCATION OF NATURAL GROUND WATER<br>TABLE.<br>4. RQD - ROCK QUALITY DESIGNATION.<br>5. 1. INDICATES DEPTH & LENGTH OF NX CORING RUN<br>6. DATUM IS MEAN SEA LEVEL<br>CONTACT OF CORE OF CORES OF CORPORATION<br>CONTACT OF CORE OF CORES OF CORPORATION<br>2. 2 INDICATES DEPTH & LENGTH OF NX CORING RUN<br>CONTACT OF CORE OF CORPORATION<br>2. 2 INDICATES DEPTH & LENGTH OF NX CORING RUN<br>CONTACT OF CORE OF CORPORATION<br>3. 4. RQD - ROCK QUALITY DESIGNATION.<br>5. 1. INDICATES DEPTH & LENGTH OF NX CORING RUN<br>CONTACT OF CORPORATION<br>CONTACT OF CORPORATION<br>CONTACT OF CORE OF CORPORATION<br>CONTACT OF CONTACT OF CORPORATION<br>CONTACT OF CONTACT OF CONT | 1. FIGU<br>SOIL                 | RES IN<br>SAMPL                                                                    | BLOW OR REC<br>E DENOTE THE    | OVERY         | COLUMN OF<br>ER OF BLOW | PPOSITE<br>VS OF A                                                                                                                     |  |  |  |  |  |
| <ul> <li>2. ■ 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.</li> <li>✓ 6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.</li> <li>□ / INDICATES LOCATION OF SAMPLING ATTEMPT</li> <li>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE</li> <li>3. ▼ INDICATES LOCATION OF NATURAL GROUND WATER</li> <li>3. ▼ INDICATES LOCATION OF NATURAL GROUND WATER</li> <li>4. RQD - ROCK QUALITY DESIGNATION.</li> <li>5. □. INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>6. DATUM IS MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 2"<br>Figu<br>Thp               | OD SA                                                                              | MPLE SPOON 1<br>OWN OPPOSITE   | 2" OR<br>ROCK | THE DISTA               |                                                                                                                                        |  |  |  |  |  |
| <ul> <li>WITH NO RECOVERY.</li> <li>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE</li> <li>WITH NO RECOVERY.</li> <li>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE</li> <li>INDICATES LOCATION OF NATURAL GROUND WATER</li> <li>INDICATES LOCATION OF NATURAL GROUND WATER</li> <li>RQD - ROCK QUALITY DESIGNATION.</li> <li>INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>INDICATES DEPTH &amp; LENGTH OF NX CORING RUN</li> <li>INDICATES MEAN SEA LEVEL</li> </ul>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2. <b>■2</b> I<br><b>▼6</b> I   | NDICATI<br>NDICATI                                                                 | ES LOCATION<br>ES LOCATION     | OF UNI        | DISTURBED               | SAMPLE. BORING LOG 901                                                                                                                 |  |  |  |  |  |
| 3. ¥ INDICATES LOCATION OF NATURAL GROUND WATER?<br>TABLE.<br>4. ROD - ROCK QUALITY DESIGNATION.<br>5. □ INDICATES DEPTH & LENGTH OF NX CORING RUN 10/0/12<br>6. DATUM IS MEAN SEA LEVEL<br>11700 - GSK -152                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                 | ITH NO<br>CRIPT N<br>ER.                                                           | RECOVERY.<br>NEXT TO SYMB      | OF SAN        | TPLING ATT              | EMPT     BEAVER VALLEY POWER STATION - UNIT NO. 1       MPLE     SHIPPINGPORT, PENNSYLVANIA       DUQUESNE LIGHT COMPANY               |  |  |  |  |  |
| 5. IL INDICATES DEPTH & LENGTH OF NX CORING RUN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 3• ¥ I<br>T.<br>4. ROD          | NDICATH<br>ABLE.<br>- ROCK                                                         | OUALITY DES                    | OF NAT        | URAL GROU               | ND WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                                     |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 5. 11 II<br>6. DATU             | NDICATE<br>M'IS MI                                                                 | ES DEPTH & LI<br>EAN SEA LEVEL | ENGTH         | OF NX COR               | ING RUN I 200 - GSK -152                                                                                                               |  |  |  |  |  |

)

 $\bigcirc$ 

 $\left( \begin{array}{c} \\ \end{array} \right)$ 

.

|                                                                                                                                                     | DEGITESHE LIGHT COMPANY SH_1 OF 1                                                                                                                                                              |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
|-----------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------|-----------------------------------------|--------------------------------------|-------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE BRAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 902<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND FLEV 678.3' |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
| DATE D                                                                                                                                              | DATE DRILLED MARCH 15, 1972 DRILLED BY AMERICAN DRILLING LOGGED BY F.P.V.                                                                                                                      |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
| × + <                                                                                                                                               | Ξ⊢                                                                                                                                                                                             | OVERALL<br>WEATHERING                                                      | SAM                                     | PLE                                  | HIC                                             | SOIL OR ROCK DESCRIPTION                                                                                                               |  |  |  |  |
| ELE<br>FEE                                                                                                                                          | DEP'                                                                                                                                                                                           | RQD<br>• 25 50 75 100                                                      | BLOWS<br>RECOV                          | TYPE                                 | G R API                                         | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, SEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |  |
| 678.3                                                                                                                                               |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
|                                                                                                                                                     | +                                                                                                                                                                                              |                                                                            |                                         |                                      |                                                 | APPROXIMATELY 3.5' OF FILL PLACED TOLLEVEL DRILL.<br>(NO SAMPLE)                                                                       |  |  |  |  |
| <b>67</b> 0 —                                                                                                                                       | 5 -                                                                                                                                                                                            |                                                                            | 6                                       | 1                                    |                                                 | ORGANIC SILT, SLIGHTLY PLASTIC, 10-15% FINE SAND, SOFT, DAMP,<br>DARK BROWN, SOME ROOT FRAGMENTS.<br>(OL)                              |  |  |  |  |
|                                                                                                                                                     | 10                                                                                                                                                                                             |                                                                            | 7                                       | 2                                    |                                                 | SANDY SILT, MODERATELY PLASTIC, 15-20% FINE SAND, SOFT, WET,<br>DARK EROWN, SOME FINES MAY BE ORGANIC.<br>(ML)                         |  |  |  |  |
| 660                                                                                                                                                 |                                                                                                                                                                                                |                                                                            | 4                                       | 3                                    |                                                 | SILTY SAND, UNIFORM, FINE TO VERY FINE, 20-25% SLIGHTLY PLASTIC<br>FINES, WET, MEDIUM BROWN.<br>(SM)                                   |  |  |  |  |
|                                                                                                                                                     | 20                                                                                                                                                                                             |                                                                            | 3                                       | 4                                    |                                                 | SILTY SAND, UNIFORM, FINE TO VERY FINE, 20-25% SLIGHTLY PLASTIC<br>FINES, WET, MEDIUM BROWN WITH TRACES OF MEDIUM GRAY.<br>(SM)        |  |  |  |  |
| 650 —                                                                                                                                               | ×2                                                                                                                                                                                             |                                                                            | 2                                       | 5                                    |                                                 | CLAYEY SAND, UNIFORM, FINE TO VERY FINE, 20-25% SLIGHTLY PLASTIC                                                                       |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            | 12                                      | 6                                    |                                                 | SAND, UNIFORM, COMPASE TO FINE, 1-3% NONPLASTIC FINES, WET, MEDIUM                                                                     |  |  |  |  |
| 640                                                                                                                                                 |                                                                                                                                                                                                |                                                                            | 31                                      | 7                                    |                                                 | SAND, UNIFORM, FINE, 1-3% NONPLASTIC FINES, HOIST, MEDIUM BROWN,<br>ONE 0.75 INCH PEBHLE.<br>(SP)                                      |  |  |  |  |
|                                                                                                                                                     | 40                                                                                                                                                                                             |                                                                            | 35                                      | 8                                    |                                                 | SAND, UNIFORM, FINE, 1-3% NONPLASTIC FINES, DAMP, MEDIUM BROWN.                                                                        |  |  |  |  |
| 630                                                                                                                                                 |                                                                                                                                                                                                |                                                                            | . 15                                    | 9                                    |                                                 | SAND, POORLY GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC FINES,<br>DAMP TO MOIST, MEDIUM BROWN.<br>(SP)                              |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            | 20                                      | 10                                   |                                                 | SAND, UNIFORM, COARSE TO FINE, 1-3% NONPLASTIC FINES, SATURATED,<br>MEDIUM GRAY.<br>(SP)                                               |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            | 21                                      | 11                                   |                                                 | SANDY GRAVEL, GAP GRADED, VERY COARSE TO FINE, 1% NONPLASTIC FINES,<br>SATURATED, MEDIUM GRAY, SANDSTONE FRAGMENTS TO 1 INCH.          |  |  |  |  |
| 620                                                                                                                                                 | 60 -                                                                                                                                                                                           |                                                                            |                                         |                                      |                                                 | GRAY SHALE                                                                                                                             |  |  |  |  |
|                                                                                                                                                     | -                                                                                                                                                                                              |                                                                            |                                         |                                      |                                                 | END OF BORING AT 60.01                                                                                                                 |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
|                                                                                                                                                     |                                                                                                                                                                                                |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |
| 1. FIG<br>SOI<br>140<br>A 2<br>FIG                                                                                                                  | JRES IN<br>L SAMPL<br>LB HAM<br>" OD SA<br>JRES SH                                                                                                                                             | BLOW OR REC<br>E DENOTE THE<br>MER FALLING<br>MPLE SPOON 1<br>OWN OPPOSITE | OVERY<br>NUMB<br>30" R<br>2" OR<br>ROCK | COL<br>ER OF<br>EQUIN<br>THE<br>CORF | JMN OPP<br>BLOWS<br>RED TO<br>DISTAN<br>ES DENO | OSITE<br>OF A<br>DRIVE<br>CE SHOWN.<br>TE                                                                                              |  |  |  |  |
| 2. <b>2</b><br><b>76</b><br><b>76</b>                                                                                                               | INDICAT<br>INDICAT<br>INDICAT<br>INDICAT                                                                                                                                                       | I OF CORE RE<br>ES LOCATION<br>ES LOCATION<br>ES LOCATION<br>RECOVERY      | OF UN<br>OF SP<br>OF SA                 | ed.<br>DISTU<br>LIT-S<br>MPLIN       | JRBED S<br>Spoon S<br>Ig Atte                   | AMPLE.<br>AMPLE.<br>MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                     |  |  |  |  |
| SUBS<br>NUME<br>3. ¥ I                                                                                                                              | SCRIPT<br>SER.<br>[NDICAT]                                                                                                                                                                     | NEXT TO SYMB                                                               | OL IN<br>OF NA                          | DICAT<br>TURAL                       | 'ES SAM<br>, groun                              | PLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                               |  |  |  |  |
| 4. ROD<br>5. []. I<br>6. DAT                                                                                                                        | TABLE.<br>4. RQD - ROCK QUALITY DESIGNATION.<br>5. I. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL<br>STONE & WEBSTER ENGINEERING CORPORATION<br>11700 - GSK - 153 |                                                                            |                                         |                                      |                                                 |                                                                                                                                        |  |  |  |  |

)

\*

| DUQUESNE LIGHT COMPANY SH 1 OF 1                               |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|----------------------------------------------------------------|--------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------|--------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 903 |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
| TYPE OF BORING SPLIT SPOON LOCATION                            |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
| SUMMAR                                                         | Y OF B                                                 | ORING                                                                                                                                                                                                                               |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        | OVERALL                                                                                                                                                                                                                             | SAMPLE                               | U                  | SOU OF POCK DESCRIPTION                                                                                                                         |  |  |  |
| LEV.<br>EET                                                    | EET                                                    | WEATHERING                                                                                                                                                                                                                          | PE Nov                               | Hay                | SUL OR ROCK DESCRIPTION                                                                                                                         |  |  |  |
| ᄡᄄ                                                             | 8-                                                     | 0 25 50 75 100                                                                                                                                                                                                                      |                                      | G R<br>G R         | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOG<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS            |  |  |  |
| 572.8                                                          |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
| 670                                                            |                                                        |                                                                                                                                                                                                                                     | 3 1                                  |                    | SILTY SAND, UNIFORM, FINE TO VERY FINE, 15-20% SLIGHTLY PLASTIC<br>ORGANIC FINES, MEDIUM TO DARK BROWN, SOME ROOT FRAGMENTS.                    |  |  |  |
| 010                                                            |                                                        |                                                                                                                                                                                                                                     |                                      | -                  | (SM)                                                                                                                                            |  |  |  |
|                                                                | -                                                      |                                                                                                                                                                                                                                     | 2 2                                  |                    | SILTY SAND, POORLY GRADED, COARSE TO FINE, 15-20% SLIGHTLY PLASTIC<br>FINES SOME FINES OFGANIC MEDIUM TO DARK BROWN WITH TRACE OF GRAY.         |  |  |  |
|                                                                | -                                                      |                                                                                                                                                                                                                                     |                                      |                    | (SM)                                                                                                                                            |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     | 7 3                                  |                    | SILTY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 10-15%                                                                                  |  |  |  |
| 660                                                            |                                                        |                                                                                                                                                                                                                                     |                                      |                    | NONPLASTIC FINES, REDDISH BROWN TO MEDIUM BROWN, FEW PEBBLES TO<br>1.5 INCHES.                                                                  |  |  |  |
|                                                                | 15 —                                                   |                                                                                                                                                                                                                                     |                                      |                    | (SM)                                                                                                                                            |  |  |  |
|                                                                |                                                        | a de la companya de l<br>La companya de la comp | 2 4                                  |                    | CLAYEY SAND, UNIFORM, FINE TO VERY FINE, 25-30% MODERATELY PLASTIC<br>FINES, MEDIUM GRAY WITH SOME REDDISH BROWN.                               |  |  |  |
|                                                                | 20 —                                                   |                                                                                                                                                                                                                                     | _                                    |                    | (SC)                                                                                                                                            |  |  |  |
| :                                                              |                                                        |                                                                                                                                                                                                                                     | 9 5                                  |                    | SILTY SAND, WELL GRADED, COARSE TO FINE, 10-15% SLIGHTLY PLASTIC<br>FINES, MEDIUM GRAY CHANGING TO ORANGE BOTTOM ONE THIRD OF RUN.              |  |  |  |
| 690                                                            |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                | 25 —                                                   |                                                                                                                                                                                                                                     | 25 6                                 |                    | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, 5-10% NONPLASTIC                                                                                  |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    | FINES, MEDIUM BROWN, PEBBLES TO 1.5 INCHES.<br>(SP)                                                                                             |  |  |  |
|                                                                | 30 —                                                   |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
| 40                                                             |                                                        |                                                                                                                                                                                                                                     | 17 🚩 7                               |                    | SAND, UNIFORM, FINE, 1-3% NONPLASTIC FINES, DAMP, MEDIUM BROWN.                                                                                 |  |  |  |
| ••                                                             | 35 —                                                   |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
| •                                                              |                                                        |                                                                                                                                                                                                                                     | 11 8                                 |                    | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, LESS THAN 1% NONPLAS                                                                              |  |  |  |
| •                                                              |                                                        |                                                                                                                                                                                                                                     |                                      |                    | (SP)                                                                                                                                            |  |  |  |
|                                                                | <b>2</b> 0                                             |                                                                                                                                                                                                                                     | 19 9                                 |                    | SANDY GRAVEL. POORLY GRADED. COARSE TO FENE. 1% NONPLASTIC FINES.                                                                               |  |  |  |
| 30                                                             | -                                                      |                                                                                                                                                                                                                                     |                                      |                    | SATURATED, MEDIUM EROWN.<br>(GP)                                                                                                                |  |  |  |
|                                                                | 45 —                                                   |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     | 31 10                                |                    | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC<br>FINES, WET, MEDIUM GRAY, SOME PEBBLES TO 0.5 INCHES.                       |  |  |  |
|                                                                | <br>50                                                 |                                                                                                                                                                                                                                     |                                      |                    | (SW)<br>GRAVELLY SAND. POORLY GRADED. COARSE TO FINE. LESS THAN 1% FINES.                                                                       |  |  |  |
|                                                                | 51.5                                                   |                                                                                                                                                                                                                                     | r00/0                                |                    | WET, MEDIUM GRAY-BROWN, OCCASIONAL PEBBLES TO 1 INCH, CHANGES TO:<br>GRAY SHALE - BOTTOM 4 INCHES OF RUN.                                       |  |  |  |
| 20                                                             |                                                        |                                                                                                                                                                                                                                     |                                      |                    | (SP)<br>END OF BORING AT 51.5'                                                                                                                  |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                |                                                        |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                | -                                                      |                                                                                                                                                                                                                                     |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                | -                                                      | 1                                                                                                                                                                                                                                   |                                      |                    |                                                                                                                                                 |  |  |  |
|                                                                | ••••••••                                               | 1                                                                                                                                                                                                                                   |                                      |                    |                                                                                                                                                 |  |  |  |
| L. FIGU                                                        | JRES IN                                                | BLOW OR RE                                                                                                                                                                                                                          | COVERY COL                           | LUMN OP            | POSITE                                                                                                                                          |  |  |  |
| 140<br>A 2'                                                    | LB HAM                                                 | MER FALLING                                                                                                                                                                                                                         | 30" REQUI                            | IRED TO<br>CONSTAN | DRIVE<br>NCE SHOWN.                                                                                                                             |  |  |  |
| FIGU<br>THE                                                    | JRES SH<br>PERCEN                                      | OWN OPPOSIT                                                                                                                                                                                                                         | E ROCK COP<br>ECOVERED.              | RES DEN            | DTE BORING LOG 903                                                                                                                              |  |  |  |
| ć. <b>–</b> – –                                                | NDICAT                                                 | ES LOCATION                                                                                                                                                                                                                         | OF SPLIT-                            | SPOON S            | SAMPLE.                                                                                                                                         |  |  |  |
|                                                                | NDICAT                                                 | ES LOCATION                                                                                                                                                                                                                         | OF SAMPLE                            |                    |                                                                                                                                                 |  |  |  |
|                                                                | INDICAT<br>VITH NO<br>SCRIPT<br>SER.                   | RECOVERY.<br>NEXT TO SYM                                                                                                                                                                                                            | BOL INDICA                           | TES SAN            | APLE A LZEZZ                                                                                                                                    |  |  |  |
|                                                                | NDICAT<br>VITH NO<br>SCRIPT<br>BER.<br>NDICAT<br>ABLE. | RECOVERY.<br>NEXT TO SYMI                                                                                                                                                                                                           | OF SAMPLI<br>BOL INDICA<br>OF NATURA | TES SAN            | APLE MEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY<br>STONE & WERSTER ENGINEERING CORRECTION |  |  |  |

 $\cdots)$ 

•



|                  |                                              |                                     | NONFLASTIC FINES, SATURATED, MEDIUM EROWN, FEW PEBBLES TO 0.5 INCH.<br>(SP)                                                                                            |
|------------------|----------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                  |                                              |                                     |                                                                                                                                                                        |
| 630              | 40                                           | 20 9                                | <u>SILTY SAND</u> , WELL GRADED, COARSE TO VERY FINE, 5-10% SLIGHTLY PLASTIC<br>FINES, WET, MEDIUM BROWN CHANGING TO MEDIUM GRAY BOTTOM TWO THIRDS.<br>OF RUN.<br>(SW) |
|                  | 45                                           | 45 10<br>100/0                      | SAND, UNIFORM, FINE, 3-5% SLIGHTLY PLASTIC FINES, DAMP, MEDIUM<br>GRAY, SOME GRAY SANDSTONE FRAGMENTS AT BOTTOM OF RUN.                                                |
| 620              |                                              |                                     | END OF BORING AT 47.5 <sup>†</sup>                                                                                                                                     |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     | -                                                                                                                                                                      |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     |                                                                                                                                                                        |
| ×                |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     |                                                                                                                                                                        |
|                  |                                              |                                     | -                                                                                                                                                                      |
|                  |                                              |                                     |                                                                                                                                                                        |
| 1. F)<br>S(      | GURES IN BLOW OR REDIL SAMPLE DENOTE TH      | COVERY COLUMN OP                    | POSITE * DENOTES USE OF 300 LB. HAMMER                                                                                                                                 |
| 11<br>A          | +O LB HAMMER FALLING<br>2" OD SAMPLE SPOON   | 30" REQUIRED TO<br>12" OR THE DISTA | DRIVE<br>NCE SHOWN.                                                                                                                                                    |
| F]<br>TH<br>2. ■ | UURES SHOWN OPPOSIT                          | E ROCK CORES DEN<br>ECOVERED.       |                                                                                                                                                                        |
|                  | 6 INDICATES LOCATION<br>7 INDICATES LOCATION | OF ONDISTURBED<br>OF SPLIT-SPOON    | SAMPLE.                                                                                                                                                                |
| SU               | WITH NO RECOVERY.<br>JESCRIPT NEXT TO SYM    | BOL INDICATES SA                    | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA                                                                                                 |
| 3• <del>≩</del>  | INDICATES LOCATION<br>TABLE.                 | OF NATURAL GROUP                    | ND WATEF 2 DUQUESNE LIGHT COMPANY                                                                                                                                      |
| 4. RC<br>5. ∐    | D - ROCK QUALITY DE<br>INDICATES DEPTH &     | SIGNATION.<br>LENGTH OF NX COR      | ING RUN I 401 STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                  |
| 6. DA            | TUM IS MEAN SEA LEVE                         | L                                   |                                                                                                                                                                        |

SH 1 OF 1 DUQUESNE LIGHT COMPANY J.O. NO. \_\_\_\_\_\_\_ BORING NO. \_\_\_\_ 905 BEAVER VALLEY POWER STATION SITE \_\_\_\_ 670.0 TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA \_ GROUND ELEV. AMERICAN LOGGED BY \_\_\_\_\_F.P.V. DATE DRILLED \_\_\_\_\_ARCH 20, 1974 \_\_\_\_\_ DRILLED BY \_\_ SUMMARY OF BORING \_\_\_\_ OVERALL SAMPLE SOIL OR ROCK DESCRIPTION O DEPTH FEET ELEV. FEET WEATHERING RAPHI LOG AND BLOWS L.J ТҮР RQD SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS 0 25 50 75 100 G 670.0 ORGANIC SILT, MODERATELY PLASTIC, 3-5% FINE SAND, VERY SOFT, DAMP, CONTAINS ROOT FRAMMENTS, DARK GRAY BROWN. 1/12" (OL)5 1/18" ORGANIC SILT, MODERATELY PLASTIC, 20-25% FINE SAND, VERY SOFT, WET, SOME WOOD FRAGMENTS, DARK GRAY BROWN TO BLACK. (OL)10 660 SILTY SAND, GAP GRADED, VERY COARSE TO VERY FINE, MOSTLY FINE, 31 10-15% NON PLASTIC FINES, WET, MEDIUM BROWN, FEW PEBBLES TO 1". (SM) 15 GRAVELLY SAND, POORLY GRADED, VERY COARSE TO FINE, 5-10% SLIGHTLY 31 PLASTIC FINES, WET, MEDIUM GRAY, PEBBLES TO 1 1/2". (SW) 20 650 19 GRAVELLY SAND, WELL GRADED, VERY COARSE TO FINE, LESS THAN 1% NON 5 PLASTIC FINES, MOIST, MEDIUM ORANGE BROWN, PEBBLES TO 1/2". (SW) 25 18 SAND, UNIFORM, FINE, 1-3% NON PLASTIC FINES, DAMP, MEDIUM GRAYISH 6 BROWN. (SP)30 640 26 SAND, MOSTLY UNIFORM, FINE, 1-3% NON PLASTIC FINES, DAMP, MEDIUM 7 GRAYISH BROWN, FEW PEBBLES TO 1/2". (SP) 35 8 11 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, MOSTLY COARSE, SATURATED,

|                              | -                                                                                                         |                                                                                      | (SW) - (SW)                                                                                                                      |
|------------------------------|-----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|
| 630                          | 40 <b></b><br>-<br>-<br>-                                                                                 | 16 9                                                                                 | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC -<br>FINES, SATURATED, MEDIUM GRAY, PEBBLES TO 1".<br>(SW)  |
|                              | 45 —<br>-<br>-                                                                                            | 22 10<br>100/0 <sup>m</sup>                                                          | SAND, MOSTLY UNIFORM, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC FINES,<br>MOIST, MEDIUM GRAY, CONTAINS 1/4" GRAY CLAY LAYER.<br>(SP) |
| 620                          | 50                                                                                                        |                                                                                      | IND RECOVERY (REFILSAL)                                                                                                          |
|                              |                                                                                                           |                                                                                      |                                                                                                                                  |
|                              |                                                                                                           |                                                                                      |                                                                                                                                  |
|                              |                                                                                                           |                                                                                      |                                                                                                                                  |
|                              |                                                                                                           |                                                                                      |                                                                                                                                  |
|                              |                                                                                                           |                                                                                      | -                                                                                                                                |
|                              |                                                                                                           |                                                                                      |                                                                                                                                  |
| 1. FIG<br>SOI<br>140<br>A 2' | JRES IN BLOW OR REG<br>L SAMPLE DENOTE THE<br>LB HAMMER FALLING<br>' OD SAMPLE SPOON 1                    | COVERY COLUMN OF<br>S NUMBER OF BLOW<br>30" REQUIRED TO<br>L2" OR THE DISTA          | PPOSITE<br>WS OF A<br>D DRIVE<br>ANCE SHOWN.                                                                                     |
| F160<br>THE<br>2. ■21<br>761 | PERCENT OF CORE RE<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION<br>WITH NO RECOVERY. | S ROCK CORES DEN<br>ECOVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON<br>OF SAMPLING ATT | SAMPLE.<br>SAMPLE.<br>TEMPT BEAVER VALLEY POWER STATION - UNIT NO. 1                                                             |
|                              | SCRIPT NEXT TO SYME<br>BER.                                                                               | BOL INDICATES SA                                                                     | MPLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                        |
| ן ¥ָ•נ<br>[                  | ABLE.                                                                                                     | OF NATURAL GROU                                                                      | IND WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                              |
| 5. ∐ 1<br>6. DAT             | - ROCK QUALITY DES<br>NDICATES DEPTH & L<br>IM IS MEAN SEA LEVEL.                                         | ENGTH OF NX COR                                                                      | LING RUN                                                                                                                         |

|                           | DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                                                                                                                    |                       |                                                       |                        |                 |                                                                                                                                                                          |  |  |  |  |  |
|---------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------|-------------------------------------------------------|------------------------|-----------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
| SITI<br>TYF<br>DAT<br>SUN | SITE <u>BEAVER VALLEY POWER STATION</u> J.O. NO. <u>11700</u> BORING NO. <u>906</u><br>TYPE OF BORING <u>SPLIT SPOON</u> LOCATION <u>SHIPPINGPORT, PENNSYLVANIA</u> GROUND ELEV. <u>49.4</u><br>DATE DRILLED <u>MARCH 22, 1974</u> DRILLED BY <u>AMERICAN</u> LOGGED BY F.P.V.<br>SUMMARY OF BORING |                       |                                                       |                        |                 |                                                                                                                                                                          |  |  |  |  |  |
| ELEV.                     | FEET                                                                                                                                                                                                                                                                                                | DEPTH<br>FEET         | OVERALL<br>WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | BLOWS<br>BECOV<br>TYPE | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>ON JOINTING BEDDING AND FAULTING AND TEXTURE                        |  |  |  |  |  |
| 68                        | 689.4                                                                                                                                                                                                                                                                                               |                       |                                                       |                        |                 |                                                                                                                                                                          |  |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                     |                       |                                                       | 35 1                   |                 | SAND, POORLY GRADED, MEDIUM COARSE TO FINE, MOSTLY FINE, 5-10%<br>SLIGHTLY PLASTIC FINES, DAMP, MEDIUM BROWN, ONE 1 INCH PEBELE.<br>(ROAD FILL) (SP)                     |  |  |  |  |  |
|                           | 1.4 - 51.6 1                                                                                                                                                                                                                                                                                        |                       |                                                       |                        |                 | SILTY SAND, UNIFORM, FINE TO VERY WINE, 20-25% MODERATELY PLASTIC<br>FINES, DAMP, MEDIUM BROWN.<br>(SM)                                                                  |  |  |  |  |  |
| 680                       |                                                                                                                                                                                                                                                                                                     | 10 <u>-</u><br>-<br>- |                                                       |                        |                 | <u>SILTY SAND</u> , SAME AS ABOVE.<br>(SM)                                                                                                                               |  |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                     | 15 —<br>—<br>—        |                                                       |                        | *               | NO RECOVERY<br>SILTY SAND, UNIFORM, FINE TO VERY FINE, 20-25% MODERATELY PLASTIC                                                                                         |  |  |  |  |  |
| 670                       |                                                                                                                                                                                                                                                                                                     | 20                    |                                                       | 9 2<br>13 3            | 2               | FINES, WET, MEDIUM BROWN.<br>(SM)<br><u>SILTY SAND</u> , POORLY GRADED, COARSE TO VERY FINE, MOSTLY FINE,<br>20-25% MODERATELY PLASTIC FINES, WET, MEDIUM BROWN.<br>(SM) |  |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                     | 25 —<br><br>          |                                                       | 38                     | •               | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, 5-10% SLIGHTLY<br>PLASTIC FINES, MOIST, MEDIUM BROWN, PEBBLES TO 1 1/4 INCH.<br>(SP)                                       |  |  |  |  |  |
| 660                       | ······································                                                                                                                                                                                                                                                              |                       |                                                       | 23                     |                 | SAND, WELL GRADED, COARSE TO FINE, 3-5% NONPLASTIC FINES, WET,<br>MEDIUM BROWN, FEW PEBBLES TO 3/4 INCH.                                                                 |  |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                     |                       |                                                       |                        |                 |                                                                                                                                                                          |  |  |  |  |  |



|                           |                                                                                                                                                                                                                                                                                                          |                |                                                       |                                  | ~               |                                                                                                                                                                                            |  |  |  |  |
|---------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------|-------------------------------------------------------|----------------------------------|-----------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                           | DUQUESNE LIGHT COMPANY SH 1 OF 2                                                                                                                                                                                                                                                                         |                |                                                       |                                  |                 |                                                                                                                                                                                            |  |  |  |  |
| SITI<br>TYP<br>DAT<br>SUM | SITE <u>BEAVER VALLEY POWER STATION</u> J.O. NO. <u>11700</u> BORING NO. <u>907</u><br>TYPE OF BORING <u>SPLIT SPOON</u> LOCATION <u>SHIPPINGPORT, PENNSYLVANIA</u> GROUND ELEV. <u>715.01</u><br>DATE DRILLED <u>MARCH 26-27, 1974</u> ORILLED BY <u>AMERICAN</u> LOGGED BY J.E.P.<br>SUMMARY OF BORING |                |                                                       |                                  |                 |                                                                                                                                                                                            |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          |                |                                                       |                                  |                 |                                                                                                                                                                                            |  |  |  |  |
| ELEV.                     | FEET                                                                                                                                                                                                                                                                                                     | DEPTH<br>FEET  | OVERALL<br>WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | BAMPLE<br>Some<br>BECOK<br>BECOK | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>OR JOINTING BEDDING AND FAULTING<br>DESCRIPTIONS                                                                         |  |  |  |  |
| 715                       | 715                                                                                                                                                                                                                                                                                                      |                |                                                       |                                  |                 |                                                                                                                                                                                            |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          |                |                                                       |                                  |                 | SILTY SAND, UNIFORM, FINE, 20-25% NONPLASTIC FINES, DRY, COMPACT,<br>MEDIUM BROWN.<br>(SM)                                                                                                 |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          | 5              |                                                       | 2                                |                 | <u>CLAYEY SAND</u> , SIMILAR TO SH #1, EXCEPT FINES ARE SLIGHTLY PLASTIC,<br>DAMP.<br>(SC)                                                                                                 |  |  |  |  |
| 705                       |                                                                                                                                                                                                                                                                                                          | 10<br>         |                                                       | 3                                |                 | <u>CLAYET SAND</u> , SAME AS SH #2.                                                                                                                                                        |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          | 15 —<br>—<br>— |                                                       | 4                                |                 | CLAYET SAND, WIDELY GRADED, 10-20% ROUNDED GRAVEL TO 1.0 INCH<br>MAXIMUM, COARSE TO FINE, MOSTLY FINE, 20-25% SLIGHTLY PLASTIC FINES,<br>COMPACT, DAMP, MEDIUM BROWN, LARGE PIECE OF WOOD. |  |  |  |  |
| 695                       |                                                                                                                                                                                                                                                                                                          | 20 —<br>-      | -                                                     | 5                                |                 | NO RECOVERY                                                                                                                                                                                |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          |                |                                                       | 23                               |                 | SILTY SAND, WIDELY GRADED, MEDIUM TO FINE, MOSTLY FINE, 10-15%                                                                                                                             |  |  |  |  |
| 685                       |                                                                                                                                                                                                                                                                                                          |                |                                                       | 31 2                             |                 | SILTY SAND, WIDELY GRADED, 8-12% ROUNDED GRAVEL TO 1.0 INCH MAXIMUM<br>COARSE TO FINE, MOSTLY FINE, 10-20% NONPLASTIC FINES, STIFF, MOIST, -<br>MEDIUM BROWN, TRACE COAL.<br>(SM)          |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          | -              | -                                                     | 16 3                             |                 | SAND, UNIFORM, FINE, 3-8% NONPLASTIC FINES, COMPACT, MOIST, MEDIUM<br>EROWN.<br>(SP)                                                                                                       |  |  |  |  |
|                           |                                                                                                                                                                                                                                                                                                          | 35             | ]                                                     | 18                               |                 | GRAVELLY SAND, POORLY GRADED, 5-10% ROUNDED GRAVEL TO 1.0 INCH<br>MAXIMUM, COARSE TO FINE SAND, MOSTLY FINE, 3-8% NONPLASTIC FINES,                                                        |  |  |  |  |



SH 2 OF 2 DUQUESNE LIGHT COMPANY SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 \_\_\_\_ BORING NO. 907 GROUND ELEV. 715.01 SHIPPINGPORT, PENNSYLVANIA TYPE OF BORING SPLIT SPOON LOCATION DRILLED BY \_\_AMERICAN DATE DRILLED MARCH 26-27, 1974 LOGGED BY \_\_\_\_\_\_ SUMMARY OF BORING . RAPHIC LOG OVERALL SAMPLE OR ROCK DESCRIPTION DEPTH FEET SOIL ELEV. WEATHERING FEET BLOWS RECOV. AND ΤΥΡΕ RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 TS 100 6 645-70 SAND, POORLY GRADED, 3-8% ANDULAR GRAVEL TO 0.6 INCH MAXIMUM, COARSE TO FINE, MOSTLY FINE, 1-5% NONPLASTIC FINES, VERY DENSE, 54 11 MEDIUM BROWN. (SP) 75 SILTY SAND, WIDELY GRADED, COARSE TO FINE, MOSTLY FINE, 10-15% NON-58 12 PLASTIC FINES, VERY DENSE, MEDIUM BROWN. (SM) 635 ~ 80 SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-5% NONPLASTIC FINES, VERY DENSE, MEDIUM BROWN. 50 13 (SP) 85 SAME AS S #13. 56 <u>SAND</u>, 17 (SP) 625 90 GRAVELLY SAND, WIDELY GRADED, 15-25% SUBROUNDED GRAVEL TO 1.1 INCH-MAXIMUM, COARSE TO FINE, MOSTLY FINE, 10-15% NONPLASTIC FINES, 147 15 VERY DNESE, GREENISH BROWN. (SP) 95 -GRAVELLY SAND, POORLY GRADED, 15-25% ROUNDED GRAVEL TO 1.0 INCH 16 MAXIMUM, COARSE TO FINE, MOSTLY FINE, 1-5% NONPLASTIC FINES, VERY 186 DENSE, GREENISH BROWN. (SP) TOP DE ROCK AT 96.01 100 0# NO RECOVERY 17 615 100 ENDOOE BORING AT 100.01

| 1                             |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|-------------------------------|-----------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------|
| 1                             |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               | -                                                                                                                           |                                                                              | i                                   | `                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
| · · · · · ·                   |                                                                                                                             |                                                                              |                                     | _                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               | -                                                                                                                           |                                                                              |                                     |                                                                                                  |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               | 4                                                                                                                           |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | _                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               |                                                                                                                             |                                                                              | 2                                   |                                                                                                  |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     |                                                                                                  |
|                               | -                                                                                                                           |                                                                              |                                     | ·                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | -                                                                                                |
|                               |                                                                                                                             |                                                                              |                                     | <u> </u>                                                                                         |
|                               |                                                                                                                             |                                                                              | -                                   |                                                                                                  |
| 1. FIG<br>SOII<br>140<br>A 2' | URES IN BLOW OR RECO<br>L SAMPLE DENOTE THE<br>LB HAMMER FALLING 3<br>" OD SAMPLE SPOON 12                                  | OVERY COLUMN OPPO<br>NUMBER OF BLOWS<br>O" REQUIRED TO 1<br>OR THE DISTANC   | OSITE<br>OF A<br>DRIVE<br>CE SHOWN  |                                                                                                  |
| FIGU<br>THE<br>2. <b>2</b> 1  | JRES SHOWN OPPOSITE<br>PERCENT OF CORE REC<br>INDICATES LOCATION O                                                          | ROCK CORES DENOT                                                             | MPLE.                               | BORING LOG 907                                                                                   |
| UPI<br>SUBS<br>NUME<br>3• ₽ I | INDICATES LOCATION OF<br>INDICATES LOCATION OF<br>ITH NO RECOVERY.<br>SCRIPT NEXT TO SYMBO<br>BER.<br>INDICATES LOCATION OF | F SPLIT-SPOON SA<br>F SAMPLING ATTEM<br>L INDICATES SAMP<br>F NATURAL GROUND | MPLE.<br>MPT<br>S<br>PLE<br>MATER 2 | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY |
| 4. <u>RQ</u> D                | ABLE.<br>- ROCK QUALITY DEST                                                                                                | GNATION.                                                                     |                                     | STONE & WEBSTER ENGINEERING CORPORATION                                                          |
| 5. 11 I<br>6. DATO            | NDICATES DEPTH & LEI<br>IM IS MEAN SEA LEVEL                                                                                | NGTH OF NX CORIN                                                             | G RUN                               | 11700 - GSK - 158A                                                                               |

|                                   |                                                                                                                                                                                                                                                    |                    |       |                |      | DUQ            | UESNE LIGHT COMPANY SH_1 OF 2                                                                                                                                                           |  |  |  |
|-----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|-------|----------------|------|----------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE 1<br>TYPE C<br>DATE<br>SUMMA | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 908<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 718.5 7/8.5<br>DATE DRILLED MARCH 28-29, 1974 DRILLED BY AMERICAN LOGGED BY JEP<br>SUMMARY OF BORING |                    |       |                |      |                |                                                                                                                                                                                         |  |  |  |
| <u>ب &lt;</u>                     | TH<br>T                                                                                                                                                                                                                                            | OVERAL<br>WEATHERI |       | SAM            | IPLE | ЧС<br>ИС       | SOIL OR ROCK DESCRIPTION                                                                                                                                                                |  |  |  |
| FEE                               | FEE                                                                                                                                                                                                                                                | RQD<br>• 25 50 T   | 5 100 | BLOWS<br>RECOV | TYPE | G R A P<br>LO( | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                  |  |  |  |
| 010                               |                                                                                                                                                                                                                                                    |                    |       |                |      |                |                                                                                                                                                                                         |  |  |  |
| 118-                              |                                                                                                                                                                                                                                                    |                    | F     | PUSH           |      |                | CUTTINGS - <u>SILTY SAND</u> , WIDELY GRADED, COARSE TO FINE, MOSTLY FINE,<br>20-30% NONPLASTIC TO SLIGHTLY PLASTIC FINES, COMPACT, DAMP, MEDIUM -<br>BROWN, SOME WOOD AND ROOTS, (SM). |  |  |  |
| <b>P11</b> A                      | 5                                                                                                                                                                                                                                                  |                    |       | 41-38'         | 1    |                | GRAVELLY SAND, POORLY GRADED, 25-30% ANGULAR GRAVEL TO 1.0 INCH<br>MAXIMUM COARSE TO FINE SAND, 5-10% NONPLASTIC FINES, VERY DENSE,<br>DAMP, LIGHT BROWN, (SP).                         |  |  |  |
| Υ <b>Τ</b> Ο                      | 10                                                                                                                                                                                                                                                 |                    |       | 101            | 2    |                | <u>GRAVELLY SAND</u> , SIMILAR TO S#1 EXCEPT 15-20% SUBANGULAR GRAVEL TO 1.2 INCH MAXIMUM, (SP).                                                                                        |  |  |  |
|                                   | 15                                                                                                                                                                                                                                                 |                    |       | 79             | 3    |                | GRAVELLY SAND, WIDELY GRADED, 10-15% SUBANGULAR GRAVEL TO 0.8<br>INCH MAXIMUM, COARSE TO FINE SAND, MOSTLY FINE, 5-10% MONPLASTIC<br>FINES, VERY DENSE, MOIST, MEDIUM BROWN, (SP).      |  |  |  |
| 700-                              | 20                                                                                                                                                                                                                                                 |                    |       | 85             | 4    |                | <u>GRAVELLY SAND</u> , SIMILAR TO S#3 EXCEPT DRY, (SP).                                                                                                                                 |  |  |  |
|                                   | 25 <u>-</u><br>-                                                                                                                                                                                                                                   |                    |       | 51             | 5    |                | GRAVELLY SAND, SIMILAR TO S#3 EXCEPT 15-25% ANGULAR GRAVEL TO 1.1<br>INCH MAXIMUM, (SP).                                                                                                |  |  |  |
| 690                               | 30 -                                                                                                                                                                                                                                               |                    |       | 59             | 6    |                | GRAVELLY SAND, WELL GRADED, 10-20% ANGULAR GRAVEL TO 0.9 INCH<br>MAXIMUM, COARSE TO FINE SAND, 3-8% NONPLASTIC FINES, MOIST, VERY<br>DENSE, MEDIUM BROWN, (SW).                         |  |  |  |
|                                   | 35 -                                                                                                                                                                                                                                               |                    |       | 59             | 7    |                | GRAVELLY SAND, POORLY GRADED, 10-15% SUBROUNDED GRAVEL TO 0.8 INCH<br>MAXIMUM, COARSE TO FINE SAND, MOSTLY FINE, 3-8% NONPLASTIC FINES,                                                 |  |  |  |



| DUQUESNE LIGHT COMPANY SH 2 OF 2             |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|----------------------------------------------|----------------------------------------------------------------|-------------------------------------|--------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|
| SITE BEAU                                    | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 908 |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| TYPE OF BO<br>DATE DRIL                      | RING <u>SPLIT SPOON</u><br>ED <u>MARCH 28-29</u>               | _ LOCATION                          | SHIPP:<br>DRIL     | LED BY AMERICAN LOGGED BY JP                                                                                                           |  |  |  |  |  |  |
| SUMMARY O                                    | F BORING                                                       | <u> </u>                            |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | OVERALL                                                        | SAMPI F                             | ()                 |                                                                                                                                        |  |  |  |  |  |  |
| EC.<br>PTH                                   | WEATHERING                                                     | SX NICL                             | PHI(<br>0G         | SOIL OR ROCK DESCRIPTION                                                                                                               |  |  |  |  |  |  |
|                                              |                                                                | BLO<br>REC                          | GRA<br>U           | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| 718.5                                        | _                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| 650                                          |                                                                |                                     |                    | <u></u>                                                                                                                                |  |  |  |  |  |  |
| 70                                           |                                                                | 36                                  |                    | SILTY SAND, SAME AS S#13, (SM).                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| . 75                                         |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | 4                                                              | 42 15                               |                    | SAND, SKIP GRADED, COARSE AND FINE, MOSTLY FINE, 1-5% NONPLASTIC<br>FINES, 1-5% NONPLASTIC FINES, DENSE, MEDIUM BROWN, (SP).           |  |  |  |  |  |  |
| 640                                          |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                | 40 16                               |                    | SAND, SAME AS S#15, (SP).                                                                                                              |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| 85                                           |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| 630                                          |                                                                | 113 * 17                            |                    | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-5% NONPLASTIC<br>FINES, VERY DENSE, DARK BROWN, POCKETS OF SILTY SAND, (SP).       |  |  |  |  |  |  |
| 90                                           |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | -                                                              | 167 18                              |                    | SILTY SAND, UNIFORM, FINE, 10-20% NONPLASTIC FINES, VERY DENSE,<br>DARK BROWN, FEW SEVERELY WEATHERED GREEN SANDSTONE FRAGMENTS, (SM). |  |  |  |  |  |  |
| 95                                           |                                                                |                                     |                    | TOP OF WEATHERED ROCK AT 95.0'                                                                                                         |  |  |  |  |  |  |
|                                              |                                                                | <u>100</u> 19<br>3"                 |                    | SANDY CHAY MODERATELY PLASTIC, 10-20% FINE SAND, VERY DENSE, -                                                                         |  |  |  |  |  |  |
| 620 - 100                                    |                                                                | 100 20                              |                    | 95-100' - GRAY SHALE IN WATER RETURN                                                                                                   |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    | NO RECOVERY. TOP OF ROCK AT 100.0'                                                                                                     |  |  |  |  |  |  |
|                                              | -                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | -                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | -                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | <b>-</b>                                                       |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    | · -                                                                                                                                    |  |  |  |  |  |  |
|                                              | ]                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | -                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | -                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | 1                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | ]                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | 1                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              |                                                                |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
|                                              | 1                                                              |                                     |                    |                                                                                                                                        |  |  |  |  |  |  |
| 1. FIGURES                                   | IN BLOW OR RE                                                  | COVERY COI                          | JUMIN OPP          | OSITE * DENOTE'S USE OF 300 LB HAMMER.                                                                                                 |  |  |  |  |  |  |
| SOIL SA<br>140 LB<br>A 2" OT                 | MPLE DENOTE THE<br>HAMMER FALLING                              | E NUMBER (<br>30" REQUI             | DF BLOWS           | OF A<br>DRIVE<br>CE SHOWN                                                                                                              |  |  |  |  |  |  |
| FIGURES<br>THE PER                           | SHOWN OPPOSIT                                                  | E ROCK COF                          | ES DENO            |                                                                                                                                        |  |  |  |  |  |  |
| <. ■2 INDI <b>F6</b> INDI <b>F6</b> INDI     | CATES LOCATION<br>CATES LOCATION<br>CATES LOCATION             | OF UNDIST<br>OF SPLIT-<br>OF SAMPLE | URBED S<br>SPOON S | AMPLE.<br>AMPLE.<br>BEAVER VALLEY FOWER STATION - UNIT NO. 1                                                                           |  |  |  |  |  |  |
| WITH<br>SUBSCRI                              | NO RECOVERY.<br>PT NEXT TO SYM                                 | BOL INDICA                          | TES SAM            | BHIPPINGPORT, PENNSYLVANIA                                                                                                             |  |  |  |  |  |  |
| 3. ¥ INDI<br>TABL                            | CATES LOCATION<br>E.                                           | OF NATURA                           | L GROUN            | D WATER 2 DUQUESNE LIGHT COMPANY                                                                                                       |  |  |  |  |  |  |
| 4. <u>RQD</u> - R<br>5. ∐ INDI<br>6. DATUM I | OCK QUALITY DES<br>CATES DEPTH & 1<br>S MEAN CEA TENNE         | SIGNATION.<br>LENGTH OF             | NX CORII           | NG RUN                                                                                                                                 |  |  |  |  |  |  |

)

-

)

SH 1 OF 1 DUQUESNE LIGHT COMPANY \_ BORING NO. \_\_\_\_909 11700 SITE \_\_\_\_\_ BEAVER VALLEY POWER STATION J.O. NO. \_\_ SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 670.71 TYPE OF BORING SPLIT SPOON LOCATION \_ DATE DRILLED APRIL 17, 1974 LOGGED BY \_\_\_\_\_F.P.V. DRILLED BY \_\_\_\_AMERICAN SUMMARY OF BORING . OVERALL SAMPLE SOIL OR ROCK DESCRIPTION Q I DEPTH WEATHERING RAPHI FEET Ž G BLOWS RECOV AND ΤΥΡΕ Ŝ Ш RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY 0 25 50 75 100 6 670.7 670 SILTY SAND, UNIFORM, VERY FINE, 35-40% SLIGHTLY PLASTIC FINES, 5 1 DAMP, MEDIUM DARK BROWN, SOME FINES ORGANIC, SOME ROOT FRAGMENTS. (SM) 5 SANDY SILT, SLIGHTLY PLASTIC, 20-30% VERY FINE SAND, VERY SOFT, 2 2 MEDIUM BROWN WITH SOME REDDISH BROWN AND GRAY. (ML) 10 660 SILTY SAND, MOSTLY UNIFORM, FINE TO VERY FINE, 25-30% SLIGHTLY PLASTIC FINES, WET, MEDIUM BROWN CHANGING TO MEDIUM DARK GRAY, 10 3 BOTTOM 1/3 OF RUN. (SM) 15 SILTY SAND, MOSTLY UNIFORM, FINE, 10-15% SLIGHTLY PLASTIC FINES, WER, MEDIUM GRAY, FEW PEBBLES TO 1/2 INCH. 19 (SM) 20 650 SAND, UNIFORM, FINE, 1-3% NONPLASTIC FINES, DAMP, MEDIUM BROWN. 11 5 SAND, SAME AS SS #5. (SP) CHANGING AT APPROXIMATELY 26.0' TO: GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC 25 12 6 FINES, WET, MEDIUM GRAY-BROWN, PEBBLES TO 3/4 INCH. (SW) 30 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES, 14 7 640 WET, MEDIUM GRAY-BROWN, PEBBLES TO 3/4 INCH. (SW) 35 ·

|                                                                                                  | 16 🖡 8                                      | SAND, MOSTLI UNIFORM, FINE, 1-3% NONPLASTIC FINES, SATURATED,<br>MEDIUM GRAY-EROWN                                                              |
|--------------------------------------------------------------------------------------------------|---------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------|
| 630 40                                                                                           | 28 9                                        | <u>GRAY SANDSTONE</u> , VERY WEATHERED, 20-25% MEDIUM GRAY SAND.                                                                                |
| 45 -                                                                                             | 55 10                                       | <u>GRAVELLY SAND</u> , WELL GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES,<br>MOIST, MEDIUM GRAY, PEBBLES TO 1/2 INCH, SOME SANDSTONE FRAGMENTS |
|                                                                                                  |                                             |                                                                                                                                                 |
| 620                                                                                              |                                             | END OF BORING AT 49.5'                                                                                                                          |
|                                                                                                  |                                             |                                                                                                                                                 |
| 55                                                                                               |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
|                                                                                                  |                                             |                                                                                                                                                 |
| 1. FIGURES IN BLOW OR RECOV                                                                      | ERY COLUMN OPPO                             | )SITE                                                                                                                                           |
| SOIL SAMPLE DENOTE THE N<br>140 LB HAMMER FALLING 30                                             | UMBER OF BLOWS<br>" REQUIRED TO D           | OF A<br>DRIVE                                                                                                                                   |
| FIGURES SHOWN OPPOSITE R<br>THE PERCENT OF CORE RECO                                             | OR THE DISTANC<br>OCK CORES DENOT<br>VERED. | B C C C C C C C C C C C C C C C C C C C                                                                                                         |
| 2. 2 INDICATES LOCATION OF<br><b>76</b> INDICATES LOCATION OF<br><b>17</b> INDICATES LOCATION OF | UNDISTURBED SA<br>SPLIT-SPOON SA            | MPLE. BORING LOG 909                                                                                                                            |
| WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL<br>NUMBER.                                         | SAMPLING ATTEM                              | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                |
| 3. ¥ INDICATES LOCATION OF<br>TABLE.<br>4. RQD - ROCK QUALITY DEST                               | NATURAL GROUND                              | WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                                                 |
| 5. IL INDICATES DEPTH & LENG<br>6. DATUM IS MEAN SEA LEVEL                                       | GTH OF NX CORING                            | G RUN. 101 11700 - GSK - 160                                                                                                                    |

|                        |                            | OVERALL                                    | SAMPLE                        |                      |                                                                                                                                                             |
|------------------------|----------------------------|--------------------------------------------|-------------------------------|----------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEV.<br>FEET          | DEPTH<br>FEET              | WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | BLOWS<br>OR<br>RECOV.<br>TYPE | G R APHI<br>LOG      | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LI<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                              |
| <u>669_0</u>           |                            |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            | 2 1                           |                      | SILTI SAND, UNIFORM, FINE TO VERI FINE, 35-406 SLIGHTLI FIXSI<br>FINES, SOME ORGANIC, MOIST, MEDIUM DARK BROWN, SOME ROOT FRAGE<br>(SM)                     |
|                        | 5 _                        |                                            |                               |                      | SILTY SAND, UNIFORM, FINE TO VERY FINE, 30-35% SLIGHTLY PLAST                                                                                               |
|                        |                            |                                            | 9 📂 2                         |                      | FINES, SOME ORGANIC, MOIST, MEDIUM DARK BROWN, SOME ROOT FRAGE<br>PIECES OF ROTTED WOOD AT BOTTOM OF RUN.<br>(SM)                                           |
| 660                    | - 10                       |                                            |                               | ,                    |                                                                                                                                                             |
|                        |                            |                                            | 9 3                           |                      | SILTY SAND, UNIFORM, FINE TO VERY FINE, 15-20% SLIGHTLY PLAST<br>FINES, WET, DARK GRAY TO BLACK, SOME COAL.<br>(SM)                                         |
|                        | - 15 -                     |                                            |                               | ,                    |                                                                                                                                                             |
|                        |                            |                                            | 22 🔰 4                        |                      | SILTI SAND, MOSTLY UNIFORM, FINE TO VERY FINE, 10-15% NONPLAS<br>FINES, WET, MEDIUM BROWN WITH TRACE OF ORANGE BROWN, FEW PEBBI<br>1 INCH AT BOTTOM OF RUN. |
| 650                    | 20 _                       |                                            | -                             |                      | (SM)<br>SAND, UNIFORM, FINE, LESS THAN 1% NONPLASTIC FINES, MOIST, MI                                                                                       |
|                        |                            |                                            | 10 5                          |                      | BROWN, FEW PEBBLES TO 1/2 INCH AT BOTTOM OF RUN. (SP)                                                                                                       |
|                        | <br>25                     |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            | 12 📂 6                        |                      | GRAVELLY SAND, WIDELY GRADED, VERY COARSE TO FINE, 10-15% NON<br>FINES, WET, MEDIUM BROWN, PEBBLES TO 1 1/4 INCH, SOME SANDSTON<br>FRAGMENTS.               |
| 640                    | 30 <u>-</u>                |                                            |                               |                      | (SW)                                                                                                                                                        |
|                        |                            |                                            | 22 7                          |                      | <u>GRAVELLY SAND</u> , SAME AS SS #6.<br>(SW)                                                                                                               |
|                        | <br>35 —                   |                                            |                               | -                    |                                                                                                                                                             |
|                        |                            |                                            | 23 8                          |                      | SAND, MOSTLY UNIFORM, FINE, LESS THAN 1% NONPLASTIC FINES, DA<br>MEDIUM GRAY, FEW PEBBLES TO 1/2 INCH.<br>(SP)                                              |
| 630 —                  | -<br>40                    |                                            |                               | •                    |                                                                                                                                                             |
|                        | -                          |                                            | 26 9                          |                      | SAND, SAME AS ABOVE.<br>(SP)                                                                                                                                |
|                        | 45 -                       |                                            |                               |                      |                                                                                                                                                             |
|                        | -                          |                                            | 51 10                         |                      | SAND, SAME AS ABOVE, LAYER OF SILTY SAND AT BOTTOM OF RUN WITS SOME GRAY CLAY.                                                                              |
|                        | 50                         |                                            | 100 11                        | -                    |                                                                                                                                                             |
|                        |                            |                                            |                               |                      | END OF BORING AT 49.5'                                                                                                                                      |
|                        | -                          |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
|                        | -                          |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
|                        |                            |                                            |                               |                      |                                                                                                                                                             |
| 1. FIGU<br>SOII<br>140 | RES IN<br>SAMPLI<br>LB HAM | BLOW OR REC<br>DENOTE THE<br>MER FALLING   | OVERY CO<br>NUMBER            | LUMN OPP<br>OF BLOWS | OSITE<br>S OF A<br>DBI <b>VE</b>                                                                                                                            |
| A 2"<br>FIGU<br>THE    | OD SAN                     | APLE SPOON 1<br>DWN OPPOSITE               | 2" OR TH<br>ROCK CO           | E DISTAN<br>RES DENO | TE                                                                                                                                                          |
| 2. <b>2</b> 1          | NDICATE                    | S LOCATION                                 | OF UNDIS                      | TURBED S             | BORING LOG 910                                                                                                                                              |

 $\cdot$ 

)

 $\mathbf{r}$ 

SH 1 OF 1 DUQUESNE LIGHT COMPANY 911 SITE BEAVER VALLEY POWER STATION J.O. NO. \_\_\_\_\_\_11700 BORING NO. . \_ GROUND ELEV. \_\_\_\_683 683 SHIPPINGPORT. PENNSYLVANIE TYPE OF BORING SPLIT SPOON LOCATION LOGGED BY F.P.V. DATE DRILLED \_\_\_\_\_ APRIL 19, 1974 ORILLED BY AMERICAN SUMMARY OF BORING \_ OVERALL RAPHIC LOG SAMPLE OR ROCK DESCRIPTION SOIL DEPTH FEET WEATHERING ELEV. FEET BLOWS RECOV. AND TYPE RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 Ø 683 GRAVELLY SAND, FILL, TOP 1/3 OF RUN. CHANGING TO: SILTY SAND, UNIFORM, FINE TO VERY FINE, 20-25% MODERATELY PLASTIC 9 FINES, SOME ORGANIC, DAMP, MEDIUM BROWN. 680 (SM) 5 SANDY SILT, MODERATELY PLASTIC, 15-20% FINE SAND, VERY SOFT, DAMP, MEDIUM BROWN WITH SOME ORANGE BROWN, SOME ORGANIC. 3 (ML)10 SILTY SAND, UNIFORM, FINE TO VERY FINE, 15-20% SLIGHTLY PLASTIC FINES, MOIST, MEDIUM BROWN. 7 (SM) 670 15 SILTY SAND, SAME AS ABOVE. 8 (SM)20 2 SILTY SAND, SAME AS ABOVE. 5 (SM) 660 25 1 1 1 SILTY SAND, POORLY GRADED, COARSE TO VERY FINE, 15-20% SLIGHTLY PLASTIC FINES, MET, MEDIUM BROWN, FEW PEBBLES - ONE 1 1/4 INCH. 14 6 (SM)30 -GRAVELLY SAND, WELL GRADED, VERY COARSE TO VERY FINE, 10-15% NONPLASTIC FINES, SATURATED, MEDIUM BROWN. 22 7 (SW) 650 35 -GRAVELLY SAND, GAP GRADED, VERY COARSE TO FINE, 3-5% SLIGHTLY PLASTIC FINES, WET, MEDIUM BROWN, PEBBLES TO 3/4 INCH. 23

()

|                                    |                                                                                                                                                                                                 |                                                     |                                     | (SW)                                                                                                                                                                   |  |  |  |  |  |  |
|------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|
|                                    | 40 -                                                                                                                                                                                            |                                                     |                                     |                                                                                                                                                                        |  |  |  |  |  |  |
| 640                                |                                                                                                                                                                                                 |                                                     | 34 9                                | SAND, UNIFORM, FINE, LESS THAN 1% NONPLASTIC FINES, MOIST, MEDIUM<br>GRAYISH BROWN.<br>(SP)                                                                            |  |  |  |  |  |  |
|                                    |                                                                                                                                                                                                 |                                                     | 22 10                               | GRAVELLY SAND, POORLY GRADED, VERY COARSE TO FINE, MOSTLY FINE,<br>5-10% NONPLASTIC FINES, MOIST, MEDIUM GRAVISH BROWN, PEBBLES TO<br>1 1/4 INCH.<br>(SW)              |  |  |  |  |  |  |
| 630                                | 50                                                                                                                                                                                              |                                                     | 29                                  | SAND, WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC FINES,<br>WET, MEDIUM GRAY BROWN, FEW PEBELES TO 3/8 INCH.<br>(SP)                                           |  |  |  |  |  |  |
|                                    | -<br>55 -<br>-<br>-<br>-                                                                                                                                                                        |                                                     | 28 12                               | SAND, WELL GRADED, MEDIUM TO FINE, 3-5% SLIGHTLY PLASTIC FINES,<br>WET, MEDIUM GRAY<br>(SP)                                                                            |  |  |  |  |  |  |
| 620                                |                                                                                                                                                                                                 |                                                     | 49 13<br><u>100</u>                 | GRAVELLY SAND, POORLY GRABED, VERY COARSE TO VERY FINE, 5-10%<br>SLIGHTLY PLASTIC FINES, WET, MEDIUM GRAY, PEBELES TO 1/2 INCH,<br>MOSTLY SANDSTONE FRAGMENTS.<br>(SW) |  |  |  |  |  |  |
|                                    | 65                                                                                                                                                                                              |                                                     | 0" 14                               | END OF BORING AT 65.0'                                                                                                                                                 |  |  |  |  |  |  |
|                                    |                                                                                                                                                                                                 |                                                     |                                     |                                                                                                                                                                        |  |  |  |  |  |  |
|                                    |                                                                                                                                                                                                 |                                                     |                                     |                                                                                                                                                                        |  |  |  |  |  |  |
|                                    |                                                                                                                                                                                                 |                                                     |                                     |                                                                                                                                                                        |  |  |  |  |  |  |
| 1. FIG<br>SOI<br>140<br>A 2<br>FIG | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                                                     |                                     |                                                                                                                                                                        |  |  |  |  |  |  |
| THE 2. 2                           | PERCENT O<br>INDICATES                                                                                                                                                                          | OF CORE REC<br>LOCATION OF                          | OVERED.<br>F UNDIST                 | URBED SAMPLE.                                                                                                                                                          |  |  |  |  |  |  |
|                                    | INDICATES<br>INDICATES<br>WITH NO RE<br>SCRIPT NEX<br>BER.                                                                                                                                      | LOCATION OF<br>LOCATION OF<br>COVERY.<br>T TO SYMBO | F SPLIT-S<br>F SAMPLII<br>L INDICAT | SPOON SAMPLE.<br>NG ATTEMPT<br>TES SAMPLE<br>TES SAMPLE<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY            |  |  |  |  |  |  |
| 3.¥ 1                              | INDICATES                                                                                                                                                                                       | LOCATION OF                                         | F NATURAI                           | L GROUND WATEF 2                                                                                                                                                       |  |  |  |  |  |  |
| 4. ROD<br>5. ∐. J<br>6. DAT        | - ROCK QU<br>INDICATES<br>JM IS MEAL                                                                                                                                                            | ALITY DESIC<br>DEPTH & LEN<br>N SEA LEVEL           | GNATION.<br>NGTH OF N               | NX CORING RUN                                                                                                                                                          |  |  |  |  |  |  |

| DIQUESNE LIGHT COMPANY SH 1 OF 1 |                                                                                                |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|----------------------------------|------------------------------------------------------------------------------------------------|----------------------------------------------|---------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
| SITE BEA                         | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 912                                 |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
| TYPE OF                          | BORING                                                                                         | 5 <u>SPLIT SPOON</u><br><u>MAY 1, 1974</u>   |                                 | ON <u>_Shtpp</u><br>Dril            | LED BYAMERICAN LOGGED BYIDG                                                                                                                |  |  |  |  |  |
| SUMMAR                           | Y OF B                                                                                         |                                              |                                 | ··· <b>·</b>                        |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                | OVERALL                                      | SAMPI                           |                                     |                                                                                                                                            |  |  |  |  |  |
| ET                               | PTH<br>EET                                                                                     | WEATHERING<br>AND                            |                                 | DHI<br>DHI                          | SOIL OR ROCK DESCRIPTION                                                                                                                   |  |  |  |  |  |
| 33                               | DE<br>FE                                                                                       | KQU<br>a 25 50 75 160                        | BLOV<br>REC                     | 6 RA                                | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS      |  |  |  |  |  |
|                                  | <u> </u>                                                                                       |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
| 710.9<br>710 —                   |                                                                                                |                                              | 15                              | 1                                   | SILTY SAND, TRACE OF GRAVEL TO 1.25 INCH MAXIMUM, UNIFORM, FINE,                                                                           |  |  |  |  |  |
|                                  |                                                                                                |                                              |                                 |                                     | SAND, 20-30% NONPLASTIC FINES, COMPACT, DAMP, BROWN MOTTLED WITH<br>YELLOW BROWN, (SM).                                                    |  |  |  |  |  |
|                                  | 5                                                                                              | Į                                            | ~                               | ĸ                                   |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                |                                              |                                 | ~                                   | 10-15% NONPLASTIC FINES, LOOSE, DAMP, BROWN, (SM).<br>BOTTOM 7" <u>CLAYEY SAND</u> , 25-35% POORLY GRADED, COARSE TO VERY FINE,            |  |  |  |  |  |
|                                  | -<br>10                                                                                        |                                              |                                 |                                     | MOSTLY VERY FINE SAND, SLIGHTLY TO MODERATELY PLASTIC GRAY TO PINK -<br>CLAY, LOOSE, DAMP, GRAY BROWN, (SC) TRACE OF GRAVEL TO 0.50 INCH - |  |  |  |  |  |
| <b>700</b>                       | -                                                                                              |                                              | 6                               | 3                                   | CLAYEY SILT, SLIGHTLY PLASTIC, TRACE OF MEDIUM TO FINE SAND, FIRM,<br>BROWN TO GRAY BROWN, (MC).                                           |  |  |  |  |  |
|                                  | -                                                                                              |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  | 15                                                                                             |                                              | 13                              | 4                                   | SANDY SILT, SLIGHTLY PLASTIC, 10-15% COARSE TO FINE SAND, FIRM,                                                                            |  |  |  |  |  |
|                                  | -                                                                                              | 4                                            |                                 |                                     | DARK GRAY.                                                                                                                                 |  |  |  |  |  |
| (00                              | 20 _                                                                                           |                                              | 15                              |                                     | STIT NONDIASTIC 2 54 COARGE TO FINE SAND FIDM DOUDI (MT)                                                                                   |  |  |  |  |  |
| 690                              |                                                                                                |                                              | 15                              | 2                                   | SILI, NONPLASIIC, 5-5% COARSE TO FINE SAND, FIRM, BROWN, (ML).                                                                             |  |  |  |  |  |
|                                  | 25                                                                                             |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  | -                                                                                              |                                              | 11                              | 6                                   | GRAVELLY SAND, 15-20% SUBROUNDED GRAVEL TO 0.75 INCH MAXIMUM, POORLY _ GRADED, COARSE TO FINE, MOSTLY FINE, SAND, 8-12% NONPLASTIC FINES,  |  |  |  |  |  |
|                                  | -                                                                                              |                                              |                                 |                                     | COMPACT, SATURATED, LIGHT BROWN, (SP-SM).                                                                                                  |  |  |  |  |  |
| 680 —                            | <sup>30</sup> -                                                                                | 4                                            | 21                              | 7                                   | SANDY GRAVEL, SUBANGULAR GRAVEL TO 1.5 INCH MAXIMUM, 30-40% POORLY                                                                         |  |  |  |  |  |
|                                  | -                                                                                              | 1                                            |                                 |                                     | COMPACT, SATURATED, LIGHT BROWN, (GP).                                                                                                     |  |  |  |  |  |
|                                  | 35 _                                                                                           |                                              | 17                              |                                     | SANDY STUT NONDLASTIC 10-154 VERY FINE SAND FIRM LIGHT BROWN                                                                               |  |  |  |  |  |
|                                  | -                                                                                              |                                              |                                 | Ŭ                                   | (MS-ML).                                                                                                                                   |  |  |  |  |  |
|                                  | 40                                                                                             |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
| 670 -                            | -                                                                                              |                                              | 9                               | 9                                   | GRAVELLY SAND, 10-15% SUBROUNDED GRAVEL TO 0.50 INCH MAXIMUM, POORLY<br>GRADED, COARSE TO FINE, MOSTLY FINE SAND, 8-12% NONPLASTIC FINES,  |  |  |  |  |  |
|                                  | -<br>-                                                                                         | 4                                            |                                 |                                     | LOOSE, SAIURAIED, LIGHI BROWN, (SP-SM).                                                                                                    |  |  |  |  |  |
|                                  | 47 _                                                                                           |                                              | 7                               | 10                                  | GRAVELLY SAND, SIMILAR TO ABOVE.                                                                                                           |  |  |  |  |  |
|                                  |                                                                                                | -                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
| 660 —                            | 50                                                                                             |                                              | 28                              | 11                                  | GRAVELLY SAND, SIMILAR TO S#9 EXCEPT 15-20% GRAVEL TO 1.0 INCH                                                                             |  |  |  |  |  |
|                                  |                                                                                                |                                              | 1                               |                                     | END OF BORING AT 51.51                                                                                                                     |  |  |  |  |  |
|                                  | -                                                                                              | 4                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                | -                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                | 4                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  | -                                                                                              | 1                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                | 1                                            |                                 |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  | -                                                                                              | -                                            | Ì                               |                                     |                                                                                                                                            |  |  |  |  |  |
|                                  |                                                                                                |                                              | 1                               |                                     |                                                                                                                                            |  |  |  |  |  |
| 1. FIG<br>SOI<br>140             | URES I<br>L SAMP<br>LB HA                                                                      | N BLOW OR RE<br>LE DENOTE TH<br>MMER FALLING | COVERY (<br>E NUMBE)<br>30" RF4 | COLUMN OP<br>R OF BLOW<br>DUIRED TO | POSITE<br>S OF A<br>DRIVE                                                                                                                  |  |  |  |  |  |
| A 2<br>FIG                       | " OD S.<br>URES S                                                                              | AMPLE SPOON<br>HOWN OPPOSIT                  | 12" OR<br>E ROCK                | THE DISTA<br>CORES DEN              | NCE SHOWN.                                                                                                                                 |  |  |  |  |  |
| 2. <b>■2</b><br><b>▼6</b>        | FERCE<br>INDICA<br>INDICA                                                                      | NI OF CORE R<br>TES LOCATION<br>TES LOCATION | COVERE<br>OF UND<br>OF SPL      | U.<br>ISTURBED<br>IT-SPOON          | SAMPLE.                                                                                                                                    |  |  |  |  |  |
|                                  | INDICA<br>WITH N                                                                               | TES LOCATION<br>0 RECOVERY.                  | OF SAM                          | PLING ATT                           | BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                   |  |  |  |  |  |
| 308<br>NUM<br>3• ¥               | BER.<br>INDICA                                                                                 | NEAL TO SYM                                  | OF NAT                          | JRAL GROU                           | ND WATER 2 DUQUESNE LIGHT COMPANY                                                                                                          |  |  |  |  |  |
| 4. ROD                           | TABLE.<br>- ROCI                                                                               | K QUALITY DES                                | SIGNATI(                        |                                     | STONE & WEBSTER ENGINEERING CORPORATION                                                                                                    |  |  |  |  |  |
| 6. DAT                           | 2. LI INDICATES DEPTH & LENGTH OF NX CORING RUN<br>6. DATUM IS MEAN SEA LEVEL 11700 - GSK - 63 |                                              |                                 |                                     |                                                                                                                                            |  |  |  |  |  |

|                          |                                                                                                                                                                                                                                                                                                                                        |               |                                                       |                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | DUQ             | UESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                                                                                                                                                                            |  |  |  |  |
|--------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------|-------------------------------------------------------|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SIT<br>TYI<br>DA1<br>SUN | SITE       BEAVER VALLEY POWER STATION       J.O. NO. 11700       BORING NO. 913         TYPE OF BORING       SPLIT SPOON       LOCATION       SHIPPINGPORT, PENNSYLVANIA       GROUND ELEV. 725_6         DATE       DRILLED       APRIL 30, 1974       DRILLED BY       AMERICAN       LOGGED BY       JDG         SUMMARY OF BORING |               |                                                       |                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                 |                                                                                                                                                                                                                                                                                                                                                          |  |  |  |  |
| ELEV.                    | FEET                                                                                                                                                                                                                                                                                                                                   | DEPTH<br>FEET | OVERALL<br>WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | BLOWS<br>BLOWS<br>RECOV | IPLE<br>I J d X L                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>OR JOINTING, BEDDING AND FAULTING<br>DESCRIPTIONS                                                                                                                                                                                                                                      |  |  |  |  |
|                          |                                                                                                                                                                                                                                                                                                                                        | -             |                                                       |                         | _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 |                                                                                                                                                                                                                                                                                                                                                          |  |  |  |  |
| 725.                     | δ                                                                                                                                                                                                                                                                                                                                      | -             |                                                       | 10                      | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | SILTY SAND, UNIFORM, VERY FINE, SAND, 30-40% NONPLASTIC FINES,<br>LOOSE, DAMP, BROWN, (SM).                                                                                                                                                                                                                                                              |  |  |  |  |
| 720                      |                                                                                                                                                                                                                                                                                                                                        | 5 _           |                                                       | 19                      | 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | CLAYEY SAND, POORLY GRADED, COARSE TO VERY FINE, MOSTLY VERY FINE,<br>SAND, 10-15% NONPLASTIC FINES, 15-25% MODERATELY PLASTIC FINES,<br>DAMP, COMPACT, BROWN, (0.1" LAYERS OF YELLOW AND GREEN CLAY,<br>ALTERNATING WITH 0.1'-0.3' SILTY SAND, (SM-SC).                                                                                                 |  |  |  |  |
|                          |                                                                                                                                                                                                                                                                                                                                        |               |                                                       | 12                      | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | TOP 4" <u>SILTY SAND</u> UNIFORM, VERY FINE SAND, 25-30% SLIGHTLY PLASTIC<br>FINES, COMPACT, MOIST, BROWN, (SM).<br>BOTTOM 12"- <u>SAND</u> , GAP GRADED MEDIUM AND VERY FINE, MOSTLY VERY FINE, -<br>3-5% NONPLASTIC FINES, COMPACT, DAMP, LIGHT BROWN, (SP) TRACE OF<br>MEDIUM SAND.                                                                   |  |  |  |  |
| 710                      |                                                                                                                                                                                                                                                                                                                                        | 15            | 1                                                     | 9                       | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | TOP 7" <u>SILTY SAND</u> , UNIFORM, VERY FINE SAND, 15-20% NONPLASTIC FINES,<br>8-12% MODERATELY PLASTIC FINES, LOOSE, DAMP BROWN WITH GRAY<br>CLAY. (SM).                                                                                                                                                                                               |  |  |  |  |
|                          |                                                                                                                                                                                                                                                                                                                                        | 20 _          |                                                       | 68                      | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | BOTTOM 11" - SAND, UNIFORM, VERY FINE SAND, 3-8% NONPLASTIC FINES,<br>LOOSE, DAMP, BROWN. TRACE OF BROKEN GRAVEL FRAGMENTS TO 0.75 INCH<br>MAXIMUM IN SHOE, (SP).<br><u>SANDY GRAVEL</u> , ANGULAR GRAVEL TO 1.5 INCH MAXIMUM, POORLY GRADED,<br>COARSE TO FINE, MOSTLY FINE SAND, 3-8% SLIGHTLY PLASTIC FINES,<br>VERY DENSE, MOIST, LIGHT BROWN, (GP). |  |  |  |  |
| 700                      | )                                                                                                                                                                                                                                                                                                                                      | 25 —<br>-     |                                                       | 50                      | 6                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | SANDY GRAVEL, SAME AS ABOVE, (GP).                                                                                                                                                                                                                                                                                                                       |  |  |  |  |
|                          |                                                                                                                                                                                                                                                                                                                                        | 30 -          |                                                       | 20                      | 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | SANDY GRAVEL, ANGULAR GRAVEL TO 1.25 INCH MAXIMUM, 25-35% POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, 3-8% NONPLASTIC FINES, COMPACT, DAMP, LIGHT YELLOW BROWN, (GP).                                                                                                                                                                              |  |  |  |  |
| 1                        | 1                                                                                                                                                                                                                                                                                                                                      | 35 -          | 1                                                     | 1                       | a series a la series de la seri | (               | -                                                                                                                                                                                                                                                                                                                                                        |  |  |  |  |

| 090                                          |                                                                                                                                  | 25                                                                                            | GRAVELLY SAND, 10-15% SUBROUNDED GRAVEL TO 1.0 INCH MAXIMUM,<br>POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, 5-8% NONPLASTIC<br>FINES, COMPACT, SATURATED, LIGHT BROWN, (SP).                                                 |
|----------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                              | 40 <b>-</b><br>-<br>-<br>-                                                                                                       | 22 <b>7</b> 9<br>44 <b>1</b> 0                                                                | 1ST ATTEMPT - NO RECOVERY - 1.4' PIECE OF GRAVEL LODGED IN SHOE.                                                                                                                                                                   |
| 680                                          | 45 <b></b><br>-<br>-                                                                                                             | 25 /11<br>38 /12                                                                              | NONPLASTIC FINES, DENSE, DAMP, LIGHT BROWN, (SP).<br>NO RECOVERY.<br><u>GRAVELLY SAND</u> - SAME AS SAMPLE #10 EXCEPT 30-40% GRAVEL TO 1.0<br>INCH MAXIMUM, (SP).                                                                  |
|                                              | 50 —<br>                                                                                                                         | 98                                                                                            | SANDI GRAVEL, ANGULAR TO SUBBOUNDED GRAVEL TO 0.80 INCH MAXIMUM<br>30-40% POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, 3-8%<br>NONPLASTIC FINES, VERY DENSE, MOIST, GRAY BROWN, MOTTLED WITH BROWN<br>AND ORANGE BROWN, (GP). |
|                                              | 55                                                                                                                               |                                                                                               |                                                                                                                                                                                                                                    |
|                                              |                                                                                                                                  |                                                                                               |                                                                                                                                                                                                                                    |
|                                              |                                                                                                                                  |                                                                                               |                                                                                                                                                                                                                                    |
|                                              |                                                                                                                                  |                                                                                               |                                                                                                                                                                                                                                    |
| 1. FIG<br>SOI<br>140<br>A 2<br>FIG           | JRES IN BLOW OR REC<br>L SAMPLE DENOTE THE<br>LB HAMMER FALLING<br>' OD SAMPLE SPOON 1<br>JRES SHOWN OPPOSITE                    | OVERY COLUMN OPP<br>NUMBER OF BLOWS<br>30" REQUIRED TO<br>2" OR THE DISTAN<br>ROCK CORES DENO | POSITE<br>OF A<br>DRIVE<br>ICE SHOWN.                                                                                                                                                                                              |
| 2. ■2<br><b>76</b><br>□7<br>SUBS             | PERCENT OF CORE RE<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION<br>WITH NO RECOVERY.<br>SCRIPT NEXT TO SYMB | COVERED.<br>OF UNDISTURBED S<br>OF SPLIT-SPOON S<br>OF SAMPLING ATTE<br>OL INDICATES SAM      | BORING LOG 913<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA                                                                                               |
| NUMI<br>3. ¥ 1<br>4. ROD<br>5. ∏ 1<br>6. DAT | BER.<br>INDICATES LOCATION<br>ABLE.<br>- ROCK QUALITY DES<br>INDICATES DEPTH & L<br>JM IS MEAN SEA LEVEN                         | OF NATURAL GROUN<br>IGNATION.<br>ENGTH OF NX CORI                                             | DUQUESNE, LIGHT COMPANY<br>STONE & WEBSTER ENGINEERING CORPORATION<br>NG RUN I III 11700 - GSK - 64                                                                                                                                |

|                                     |                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | DUQU            | ESNE LIGHT COMPANY SH_1 OF_1                                                                                                                                                                                 |  |  |  |  |  |  |  |
|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|--|
| SITE<br>TYPE OF<br>DATE D<br>SUMMAR | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 914<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV<br>DATE DRILLED APRIL 30, 1974 DRILLED BY AMERICAN LOGGED BY JDG<br>SUMMARY OF BORING |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                 |                                                                                                                                                                                                              |  |  |  |  |  |  |  |
| ELEV.<br>FEET                       |                                                                                                                                                                                                                                    | ALL<br>RING<br>D<br>75 100<br>AL<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>C<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMP | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>OR JOINTING BEDDING AND FAULTING<br>DESCRIPTIONS                                                                                           |  |  |  |  |  |  |  |
|                                     | 5                                                                                                                                                                                                                                  | 14 1<br>22 22                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                 | SILTY SAND, MOSTLY UNIFORM, VERY FINE, 20-25% SLIGHTLY PLASTIC<br>FINES, SOME ORGANIC, DAMP, MEDIUM BROWN WITH SOME GRAY, 1 1/4"<br>SLAG PEBBLE, (SM).                                                       |  |  |  |  |  |  |  |
|                                     |                                                                                                                                                                                                                                    | 14 73                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                 | SILTY SAND, WELL GRADED, COARSE TO FINE, 10-15% NONPLASTIC FINES,                                                                                                                                            |  |  |  |  |  |  |  |
|                                     |                                                                                                                                                                                                                                    | 55 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                 | GRAVELLY SAND, PCORLY GRADED, VERY COARSE TO VERY FINE, 15-20%<br>NONPLASTIC FINES, DAMP, MEDIUM BROWN, LAYERS OF VARIED SILT AT<br>TOP OF RUN, PEBBLES TO 1", (SP-SM).                                      |  |  |  |  |  |  |  |
|                                     |                                                                                                                                                                                                                                    | 50 <b>5</b><br>31 <b>6</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                 | SAND, WELL GRADED, COARSE TO FINE, 3-5% NONPLASTIC FINES, SAND<br>LIGHT TO MEDIUM BROWN, FEW PEBBLES TO 3/4", (SP).                                                                                          |  |  |  |  |  |  |  |
|                                     | 25 -<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-                                                                                                                                     | 58 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                 | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES,<br>MOIST, MEDIUM TO DARK BROWN, PEBBLES TO 1/2", MOSTLY SANDSTONE<br>FRAGMENTS, PIECE OF WEATHERED GRAY SANDSTONE AT BOTTOM OF RUN, (SP). |  |  |  |  |  |  |  |
|                                     | 35 -                                                                                                                                                                                                                               | 53 🔽 8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                 | <u>GRAVELLY SAND</u> , SAME AS ABOVE, PEBBLES TO 1 1/4", (SP).                                                                                                                                               |  |  |  |  |  |  |  |

--·

|   |                               | -<br>-<br>40                  |                                                | 63                         | 9                                |                               | <u>GRAVEL</u>                      | LY SAND:                           | SAME AS ABOVE, (SP)                                                                                                                                                 |
|---|-------------------------------|-------------------------------|------------------------------------------------|----------------------------|----------------------------------|-------------------------------|------------------------------------|------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|   |                               | -<br>-<br>45 -                |                                                | 41                         | 10                               |                               | SAND,<br>TOP 1/<br>FINE,<br>TO 1/2 | UNIFORM,<br>3 OF RUN,<br>LESS THAN | FINE, 5-10% NONPLASTIC FINES, MOIST, MEDIUM BROWN<br>(SP-SM), CHANGING TO <u>SAND</u> , WELL GRADED, COARSE TO<br>1% NONPLASTIC FINES, MOIST, MEDIUM BROWN, PEBBLES |
|   |                               | -<br>-<br>50                  |                                                | 32                         | Í                                |                               | SAND,                              | UNIFORM,                           | FINE, 5-10% NONPLASTIC FINES, WET, MEDIUM BROWN, (SP).                                                                                                              |
|   |                               | -                             | 4                                              |                            |                                  |                               | END                                | OF BORING                          | AT 50.0'                                                                                                                                                            |
|   | -                             |                               |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   |                               |                               |                                                |                            |                                  |                               |                                    |                                    | -                                                                                                                                                                   |
|   |                               | -                             |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   |                               | -                             | 4                                              |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   |                               | -                             |                                                |                            |                                  | -                             |                                    |                                    |                                                                                                                                                                     |
|   |                               | -                             |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   |                               | -                             |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   |                               | - 1                           |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
| Ļ |                               |                               |                                                |                            |                                  |                               |                                    |                                    |                                                                                                                                                                     |
|   | 1. FIGU<br>SOII<br>140<br>A 2 | JRES II<br>L SAMPI<br>LB HAN  | I BLOW OR REC<br>LE DENOTE THE<br>IMER FALLING | 0VER<br>NUM<br>30"<br>2" 0 | Y COL<br>BER O<br>REQUI<br>8 THE | UMN OPP<br>F BLOWS<br>RED TO  | OSITE<br>OF A<br>DRIVE             | 1.JNI                              |                                                                                                                                                                     |
|   | FIGU                          | JRES SH<br>PERCEN             | IOWN OPPOSITE                                  | ROC                        | K COR<br>RED.                    | ES DENO                       | TE SHU                             |                                    |                                                                                                                                                                     |
|   | 2, ∎2]<br>16]<br>[][7]        | INDICAT<br>INDICAT<br>INDICAT | ES LOCATION<br>ES LOCATION<br>ES LOCATION      | OF U<br>OF S<br>OF S       | NDIST<br>PLIT-<br>Ampi.t         | URBED S<br>SPOON S<br>NG ATTE | AMPLE.<br>AMPLE.<br>MPT            |                                    | BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                                            |
|   |                               | VITH NO                       | NEXT TO SYMB                                   | OL I                       | NDICA                            | TES SAM                       | PLE                                | 3                                  | SHIPPINGPORT, PENNSYLVANIA                                                                                                                                          |
|   | NUME<br>איש איש<br>זי         | NDICAT                        | ES LOCATION                                    | OF N.                      | ATURA                            | L GROUN                       | D WATE                             | 2                                  | DUQUESNE LIGHT COMPANY                                                                                                                                              |
|   | +. RQD<br>5. ∐ I<br>5. DATU   | - ROCK<br>NDICAT              | QUALITY DES<br>ES DEPTH & LI<br>MEAN SEA LEV   | I GNA'<br>ENGTI<br>EL      | TION.<br>H OF                    | NX CORI                       | NG RUN                             | Melet 20                           | STONE & WEBSTER ENGINEERING CORPORATION<br>11700 - GSK - 65                                                                                                         |

|             | DEGRESSE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
|-------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------|---------------------------------|--------------------|------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE        | BEAVER V.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ALLEY POWER ST        | TION                            |                    | J.O. NO BORING NO                                                                                                                  |  |  |  |  |
| TYPE OF     | BORING                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | SPLIT SPOON           |                                 | •                  | GROUND ELEV. 686.81                                                                                                                |  |  |  |  |
| DATE D      | RILLED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | JUNE 7, 1974          | <u> </u>                        | ORIL               | LED BY LOGGED BY D.F.P.                                                                                                            |  |  |  |  |
| SUMMAN      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 | <u> </u>           |                                                                                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| ゾー          | Т                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | OVERALL<br>WEATHERING | SAMPLE                          |                    | SOIL OR ROCK DESCRIPTION                                                                                                           |  |  |  |  |
| ש ש<br>ש וב |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | RQD                   | × ₽ C • S                       |                    | FIELD AND LABORATORY TERT REQUETS. SOLL REPATA DESCRIPTION, LITHOLOGY                                                              |  |  |  |  |
| w u         | ۵ ۳                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 0 25 50 75 100        | μ<br>β<br>β<br>β<br>β<br>β<br>β | 8<br>8             | OR JOINTING BEDDING AND FAULTING AND TEXTURE DESCRIPTIONS                                                                          |  |  |  |  |
|             | •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| 686.8       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     | 1                               |                    |                                                                                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             | 5 _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                       | -                               |                    | -                                                                                                                                  |  |  |  |  |
| (m —        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       | 5 1                             |                    | SILT, MODERATELY PLASTIC, 6-10% VERY FINE SAND, DARK BROWN,                                                                        |  |  |  |  |
| 080         | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | (SM)                                                                                                                               |  |  |  |  |
| Į           | 10 _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       | 9 2                             |                    | SILTY SAND, UNIFORM, VERY FINE, 15-20% MODERATELY PLASTIC FINES, DAR                                                               |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | BROWN.<br>(SM)                                                                                                                     |  |  |  |  |
|             | 15 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| 670 —       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     | 11 3                            |                    | SAND, UNIFORM, FINE, LESS THAN 3% MEDIUM AND COARSE SAND, 5-7% FINES                                                               |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | (SP) -                                                                                                                             |  |  |  |  |
|             | 20                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                       |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       | 4 4                             |                    | CLAYEY SAND, UNIFORM, VERY FINE, 8-10% SLIGHTLY TO MODERATELY PLASTIC                                                              |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | (SC) -                                                                                                                             |  |  |  |  |
|             | 25 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| 660         | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       | 6 5                             |                    | CLAYEY SAND, SIMILAR TO SS #4.                                                                                                     |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             | 30 _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1                     |                                 |                    |                                                                                                                                    |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | -                     | 37 6                            |                    | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, MOSTLY MEDIUM AND                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | -                     |                                 |                    | COARSE, 10-15% GRAVEL TO 1.9 INCH MAXIMUM, 4-6% FINES, LIGHT BROWN.<br>(SP)                                                        |  |  |  |  |
|             | 35 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| 650         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | -                     | 27 7                            |                    | SAND, UNIFORM, FINE, LESS THAN 4% MEDIUM SAND, 3-4% FINES, LIGHT                                                                   |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       |                                 |                    | BHOWN.<br>(SP)                                                                                                                     |  |  |  |  |
|             | 40 _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                       |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 4                     | 22 8                            |                    | SAND, SIMILAR TO ABOVE EXCEPT SAMPLE CONTAINS 8-10% GRAVEL TO 2.0                                                                  |  |  |  |  |
|             | .                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | -                     |                                 |                    | INCH MAXIMUN.                                                                                                                      |  |  |  |  |
|             | 45 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1                     |                                 |                    | -                                                                                                                                  |  |  |  |  |
| 640         | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                       | 21 9                            |                    | SANDY GRAVEL, FOORLY GRADED TO 1.9 INCH MAXIMUM, 10-15% FINE TO                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 4                     |                                 |                    | (GP)                                                                                                                               |  |  |  |  |
|             | 50                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4                     |                                 |                    |                                                                                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     | 24                              |                    | DANDI GRAVEL, FOURLY GRADED TO 1.75 INCH MAXIMUM, 12-18% FINE TO .<br>COARSE SAND, MOSTLY FINE, 3-7% FINES, LUGHT YELLOWISH BROWN. |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | -                     | 1                               |                    |                                                                                                                                    |  |  |  |  |
|             | 55 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | -                     |                                 |                    | SIED THITPODE STUD I BOD BOAN IS LODATING AND ALLES A LOT                                                                          |  |  |  |  |
| 630 —       | -  :                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 4                     | 58 11                           |                    | GRAVEL TO 2.0 INCH MAXIMUM, LESS THAN 5% FINES, BLUEISH GRAY.                                                                      |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
|             | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4                     |                                 |                    | SAND, UNIFORM, FINE, 4-6% MEDIUM SAND, 6-10% GRAVEL TO 1.75 INCH -                                                                 |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     | 48 12<br>100/2.5"               | •                  | BLUE SHALE FRAGMENTS AT BOTTOM OF SHOE                                                                                             |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 |                    | END OF BORING AT 62.4                                                                                                              |  |  |  |  |
|             | - 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 4                     |                                 |                    |                                                                                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     |                                 |                    | -                                                                                                                                  |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                     |                                 |                    |                                                                                                                                    |  |  |  |  |
|             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 4                     |                                 |                    |                                                                                                                                    |  |  |  |  |
|             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                       |                                 |                    |                                                                                                                                    |  |  |  |  |
| 1. FIG      | URES I                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | N BLOW OR RE          | COVERY CO                       | LUMN OP            | POSITE                                                                                                                             |  |  |  |  |
| 140         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | MMER FALLING          | 30" REQU                        | IRED TO            | DRIVE<br>NCE SHOLM                                                                                                                 |  |  |  |  |
| FIC         | JURES S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | HOWN OPPOSIT          | E ROCK CO.                      | E DISTA<br>RES DEN | OTE                                                                                                                                |  |  |  |  |
| 2. <b>2</b> | INDICA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | TES LOCATION          | OF UNDIS                        | TURBED             | SAMPLE. BORING LOG 915                                                                                                             |  |  |  |  |
|             | INDICA<br>UNDICA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | TES LOCATION          | OF SAMPL                        | -SPOON<br>ING ATT  | EMPT BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                      |  |  |  |  |
| SUE         | SCRIPT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | NEXT TO SYM           | BOL INDIC.                      | ATES SA            | MPLE SHIPPINGFORT, PENNSYLVANIA                                                                                                    |  |  |  |  |
| 3. ¥        | INDICA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | TES LOCATION          | OF NATUR.                       | AL GROU            | ND WATER 2 DUQUESNE LIGHT COMPANY                                                                                                  |  |  |  |  |
| 4. ROD      | $\frac{1}{1} = \frac{1}{1} $ | K QUALITY DE          | SIGNATION                       | *<br>117 000       | THE RUN A STONE & WEBSTER ENGINEERING CORPORATION                                                                                  |  |  |  |  |
| 6. DAT      | 5. II INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL 11700 - CSK -66                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                       |                                 |                    |                                                                                                                                    |  |  |  |  |

))

 $( \widehat{ } )$ 



)

| 650                                                   | -                                                                                                                                                                                                                                                                                                                                       | 40 <b>F</b> (                 | 3-6% FINES, DARK HROWN.                                                                                                |  |  |  |  |  |  |  |  |  |
|-------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------|------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|--|--|--|
|                                                       | 40                                                                                                                                                                                                                                                                                                                                      | 18 8                          | NO RECOVERY                                                                                                            |  |  |  |  |  |  |  |  |  |
| 640 <sup>.</sup>                                      | 45                                                                                                                                                                                                                                                                                                                                      | 42 9                          | SAND, UNIFORM, FINE, 4-7% MEDIUM SAND, 5-7% SLIGHTLY PLASTIC FINES,                                                    |  |  |  |  |  |  |  |  |  |
|                                                       | 50 -                                                                                                                                                                                                                                                                                                                                    | 52 10                         | SAND, UNIFORM, FINE, 4-6% MEDIUM SAND, LESS THAN 1% GRAVEL TO 2.0 -<br>INCH MAXIMUM, LESS THAN 5% FINES, ELUEISH GRAY. |  |  |  |  |  |  |  |  |  |
| 630                                                   | 55                                                                                                                                                                                                                                                                                                                                      | 52 11                         | SAND, UNIFORM, FINE, 3-5% MEDIUM AND COARSE SAND, 2-4% FINES,<br>HLUEISH GRAY.<br>(SP)                                 |  |  |  |  |  |  |  |  |  |
|                                                       | 60 <b>-</b><br>-<br>-<br>-                                                                                                                                                                                                                                                                                                              | <b>85</b> 12<br><u>100</u> 13 | SAND, SIMILAR TO SS #11.                                                                                               |  |  |  |  |  |  |  |  |  |
| 620                                                   | -<br>65<br>-<br>-<br>-<br>-                                                                                                                                                                                                                                                                                                             | 2.5*                          | INE OF BORING AT 64.8'                                                                                                 |  |  |  |  |  |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU                | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.                                                                                                                                         |                               |                                                                                                                        |  |  |  |  |  |  |  |  |  |
| THE<br>2. 21<br>761<br>071<br>SUBS                    | FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br>THE PERCENT OF CORE RECOVERED.<br>2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>F6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>[][] INDICATES LOCATION OF SAMPLING ATTEMPT<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE |                               |                                                                                                                        |  |  |  |  |  |  |  |  |  |
| NUME<br>3. ¥ I<br>4. <u>R</u> QD<br>5. ∏ I<br>6. DATU | NUMBER.<br>V INDICATES LOCATION OF NATURAL GROUND WATER 2<br>TABLE.<br>RQD - ROCK QUALITY DESIGNATION.<br>I INDICATES DEPTH & LENGTH OF NX CORING RUN<br>DATUM IS MEAN SEA LEVEL<br>DUQUESHE LIGHT COMPANY<br>STONE & WEBSTER ENGINEERING CORPORATION<br>11700 - GSK - 67                                                               |                               |                                                                                                                        |  |  |  |  |  |  |  |  |  |

SH\_1 OF\_1 DUQUESNE LIGHT COMPANY \_ J.O. NO. \_\_\_\_\_\_\_ BORING NO. \_\_\_\_\_\_ BEAVER VALLEY POWER STATION SITE \_\_\_\_\_ \_\_\_\_\_ GROUND ELEV. \_\_\_\_\_675.9 TYPE OF BORING SPLIT SPOON LOCATION \_\_\_\_\_ SHIPPINGPORT, PENNSYLVANIA \_\_\_\_\_ LOGGED BY \_\_\_\_\_ J.E.P. DATE DRILLED MARCH 29-30, 1974 DRILLED BY AMERICAN SUMMARY OF BORING \_\_\_\_ OVERALL SAMPLE Q SOIL OR ROCK DESCRIPTION DEPTH FEET WEATHERING RAPHI LOG ž E AND BLOWS OR RECOV TYPE <u>ה</u> RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING, BEDDING AND FAULTING DESCRIPTIONS 501L STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 G 675.9 5 **67**0 10 NO RECOVERY 5 SILTY SAND, WIDELY GRADED, MEDIUM TO FINE, MOSTLY FINE, 10-20% NONPLASTIC FINES, LOOSE, MOIST, DARK BROWN. 7 (SM)15 SILTY SAND, UNIFORM, FINE, 15-20% NONPLASTIC FINES, LOOSE, DAMP, 660 DARK BROWN. 3 (MR) 20 SILTY SAND, WIDELY GRADED, 10-15% ANGULAR GRAVEL TO 0.8 INCH MAX-IMUM, COARSE TO FINE SAND, MOSTLY FINE, 15-20% NONPLASTIC FINES, 135 VERY DENSE, DAMP, GREENISH BROWN. (M2)25 GRAVELLY SAND, WIDELY GRADED, 15-25% ANGULAR TO ROUNDED GRAVEL TO -650 1.0 INCH MAXIMUM, COARSE TO FINE SAND, MOSTLY FINE, 5-10% NONPLASTIC-12 FINES, COMPACT, SATURATED, DARK BROWN. (SP) 30 -SAND, POORLY GRADED, 3-8% SUBROUNDED GRAVEL TO 0.7 INCH MAXIMUM, COARSE TO FINE SAND, MOSTLY MEDIUM, 1-5% NONPLASTIC FINES, COMPACT, 14 DARK BROWN. (SP) TOP 14 INCHES: <u>SAND</u>, UNIFORM, MEDIUM, COMPACT, BROWN. (WASH?) 35 (SP)

)

|                                |                                                                                                                                                                                                |                                                 | 21               | 7                     | н<br>С<br>म<br>С | OTTOM<br>0.8 INC<br>(INES,<br>(SM) | 4 INCHI<br>CH MAXIN<br>COMPACI | es:<br>Mum,<br>I, M | SILTY SAND, WIDELY GRADED, 3-8% ROUNDED GRAVEL TO                      |  |
|--------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|------------------|-----------------------|------------------|------------------------------------|--------------------------------|---------------------|------------------------------------------------------------------------|--|
|                                | 40 <b></b><br><br>                                                                                                                                                                             | Ý                                               | 28               | 8                     | 5<br>F           | AND,<br>INES,<br>SP)               | POORLY<br>COMPACI              | GRAI<br>F, M        | DED, MEDIUM TO FINE, MOSTLY FINE, 1-5% NONPLASTIC                      |  |
| 630                            | -<br>45 <del>-</del><br>-                                                                                                                                                                      |                                                 | 27               | 9                     | s<br>(           | AND,<br>SP)                        | SAME AS                        | S <b>&amp;S</b>     | #8                                                                     |  |
|                                |                                                                                                                                                                                                |                                                 | 23               | 10                    | c.               | SAND,                              | SAME AS                        | s ss                | #8.                                                                    |  |
| <b>6</b> 20                    | -<br>-<br>55 -                                                                                                                                                                                 |                                                 | <u>164</u><br>7" | 4                     | 1<br>0<br>E      | TOP 5 I<br>DENSE,<br>BOTTOM        | INCHES:<br>MEDIUM<br>2 INCHI   | SIL<br>BRO<br>ES:   | TY SAND, UNIFORM, FINE, 10-15% NONPLASTIC FINES,                       |  |
|                                | -                                                                                                                                                                                              |                                                 |                  |                       | E                | end of                             | BORING                         | AT                  | 56.6'                                                                  |  |
|                                | 60 -                                                                                                                                                                                           |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
|                                |                                                                                                                                                                                                |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
| 1. FIGU<br>SOII<br>140<br>A 2' | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OF THE DISTANCE SHOLD |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
| FIGU<br>THE                    | JRES SHO                                                                                                                                                                                       | DWN OPPOSITE                                    | ROCK             | CORI                  | es denot         | E                                  |                                | -                   | BORING LOG TH-1                                                        |  |
| 2. 21<br>761                   | NDICATE<br>NDICATE                                                                                                                                                                             | ES LOCATION C<br>ES LOCATION O<br>ES LOCATION O | )F UND<br>)F SPL | DIST<br>DIST<br>DIT-S | JRBED SA         | MPLE.<br>MPLE.                     | <b> </b>                       |                     |                                                                        |  |
|                                | VITH NO<br>SCRIPT N                                                                                                                                                                            | RECOVERY.<br>IEXT TO SYMBO                      | L INE            | DICAT                 | TES SAMP         | FT<br>LE                           | 3                              |                     | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA |  |
| NUME<br>3• <del>♀</del> I      | NUMBER.<br>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATER 2                                                                                                                                   |                                                 |                  |                       |                  |                                    |                                |                     |                                                                        |  |
| 4. ROD<br>5. ∏. I              | ABLE.<br>- ROCK<br>NDICATE                                                                                                                                                                     | QUALITY DESI<br>CS DEPTH & LE                   | GNATI<br>NGTH    | ON.<br>OF N           | IX CORIN         | G RUN                              | M 4/7/                         | ZZ                  | STONE & WEBSTER ENGINEERING CORPORATION                                |  |
| U. DAIG                        |                                                                                                                                                                                                | PARTIC PRINT                                    |                  |                       |                  |                                    | T bess                         |                     |                                                                        |  |

| DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|--------------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. TH-2<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 676.5                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
| DATE DRILLED MARCH 30-APRIL 2,1974 DRILLED BY AMERICAN LOGGED BY J.E.P.                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | I.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | OVERALL                                 | SAMP                     | EU                  | SOIL OR ROCK DESCRIPTION                                                                                                                                              |  |  |  |  |
| ELEY<br>FEET                                                                                                                                                                                    | DEPT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | AND<br>RQD<br>0 25 50 75 100            | BLOWS<br>BLOWS<br>RECOV. | G RAPH<br>LOG       | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                 |  |  |  |  |
| 676.5                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     | •                                                                                                                                                                     |  |  |  |  |
|                                                                                                                                                                                                 | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
| <b>67</b> 0                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         | 7                        |                     | NO RECOVERY                                                                                                                                                           |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         | 4                        | 2                   | NO RECOVERY                                                                                                                                                           |  |  |  |  |
|                                                                                                                                                                                                 | 10<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                         | 2                        | 3                   | SILTY SAND, WIDELY GRADED, MEDIUM TO FINE, MOSTLY FINE, 15-20%<br>NONPLASTIC FINES, VERY LOOSE, SATURATED, DARK BROWN, MANY WOOD<br>PIECES, FEW CLAY POCKETS.<br>(SM) |  |  |  |  |
|                                                                                                                                                                                                 | 15 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
| 660                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         | 2                        | 4                   | SMELL, ROOTS AND FIBERS.<br>(OL)                                                                                                                                      |  |  |  |  |
| х.                                                                                                                                                                                              | 20                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                         |                          |                     | SANDY GRAVEL, POORLY GRADED, ANGULAR TO ROUNDED TO 1.1 INCH MAXIMUM                                                                                                   |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         | 15                       | 2                   | AND BROWN, OILY SMELL.<br>(GP)                                                                                                                                        |  |  |  |  |
| 650                                                                                                                                                                                             | 25 <u>-</u><br>-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                         | 19                       | 6                   | SANDY GRAVEL, SIMILAR TO SS #5, EXCEPT NO BLACK OR OILY SMELL.                                                                                                        |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | 30 —                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                         | 12                       | 7                   | SAND, POORLY GRADED, 5-10% ROUNDED GRAVEL TO 0.8 INCH MAXIMUM,<br>COARSE TO FINE SAND, MOSTLY FINE, 1-5% NONPLASTIC FINES, COMPACT,<br>DARK BROWN.                    |  |  |  |  |
|                                                                                                                                                                                                 | 35 —                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                         |                          | Z                   |                                                                                                                                                                       |  |  |  |  |
| 640                                                                                                                                                                                             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                         | 9                        | 9                   | GRAVELLY SAND, SIMILAR TO SS #7, EXCEPT 10-20% ROUNDED GRAVEL<br>TO 0.9 INCH MAXIMUM<br>(SP)                                                                          |  |  |  |  |
|                                                                                                                                                                                                 | 40 —<br>-<br>-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                         | 17                       | 10                  | SILTY SAND, UNIFORM, FINE, 15-20% NONPLASTIC FINES, COMPACT,<br>LIGHT BROWN.<br>(SM)                                                                                  |  |  |  |  |
| 630                                                                                                                                                                                             | 45 <u>-</u><br>-                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                         | 25                       | 11                  |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1                                       |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | - ייי                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ]                                       | 30                       | 12                  | $\frac{\text{SAND}}{(\text{SW})},  \text{SAME AS SS #11.}$                                                                                                            |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                                       | -                        |                     | TOP & INCHES, SAND SAME AS SS #11                                                                                                                                     |  |  |  |  |
| 620                                                                                                                                                                                             | 55 <del>-</del>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         | 129                      | 13                  | (SW)<br>BOTTOM 2 INCHES: LICHT GRAY SHALE, HIGHLY WEATHERED.                                                                                                          |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     | END OF BORING AT 56.7'                                                                                                                                                |  |  |  |  |
|                                                                                                                                                                                                 | 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
| 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
| THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |
|                                                                                                                                                                                                 | NDICAT<br>NDICAT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ES LOCATION<br>ES LOCATION<br>RECOVERY. | OF SPLI<br>OF SAMP       | T-SPOON SPLING ATTH | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA                                                                                                |  |  |  |  |
| NUME<br>3• ¥ I                                                                                                                                                                                  | NDICAT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ES LOCATION                             | OF NATU                  | CATES SAN           | DUQUESNE LIGHT COMPANY                                                                                                                                                |  |  |  |  |
| 4. <u>RQ</u> D<br>5. ∐. I<br>6. DATO                                                                                                                                                            | 4. RQD - ROCK QUALITY DESIGNATION.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. []. []. []. []. []. []. []. []. []. [] |                                         |                          |                     |                                                                                                                                                                       |  |  |  |  |

.

>

| DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                                                                                                               |                |                                                                |               |                 |                                                                                                                                                                                                                                                                                   |  |  |  |  |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------|----------------------------------------------------------------|---------------|-----------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE <u>BEAVER VALLEY POWER STATION</u> J.O. NO. <u>11700</u> BORING NO. <u>TH-3</u><br>TYPE OF BORING <u>SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA</u> GROUND ELEV. <u>676.7</u><br>DATE DRILLED <u>MARCH 30, 1974</u> DRILLED BY <u>AMERICAN</u> LOGGED BY F.P.V.<br>SUMMARY OF BORING |                |                                                                |               |                 |                                                                                                                                                                                                                                                                                   |  |  |  |  |
| ELEV.<br>FEET                                                                                                                                                                                                                                                                                  | DEPTH<br>FEET  | OVERALL S<br>WEATHERING<br>AND S<br>RQD S<br>0 25 50 75 100 10 | RECOK<br>TYPE | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>ON JOINTING BEDDING AND FAULTING<br>DESCRIPTIONS                                                                                                                                                                |  |  |  |  |
| .676.7                                                                                                                                                                                                                                                                                         |                |                                                                |               |                 |                                                                                                                                                                                                                                                                                   |  |  |  |  |
| 670                                                                                                                                                                                                                                                                                            | -<br>-<br>5 -  | 1                                                              |               |                 | <u>GRAVELLY SAND</u> , POORLY GRADED, COARSE TO VERY FINE, DAMP, MEDIUM<br>EROWN, PEBBLES TO 1 1/2 INCH (FILL), 3-5% MENPLASTIC FINES.<br>(SP)<br><u>GRAVELLY SAND</u> , POORLY GRADED, COARSE TO VERY FINE, 5-10% NONPLASTIC<br>FINES, WET, MEDIUM BROWN, FEW PEBBLES TO 1 INCH. |  |  |  |  |
|                                                                                                                                                                                                                                                                                                |                |                                                                | 2             |                 | SILTY SAND, UNIFORM, FINE TO VERY FINE, 10-15% NONPLASTIC FINES,                                                                                                                                                                                                                  |  |  |  |  |
| 660                                                                                                                                                                                                                                                                                            | 15 —<br><br>   |                                                                | 1 74          |                 | SANDY SILT, MODERATELY PLASTIC, 15-20% VERY FINE SAND, VERY SOFT,                                                                                                                                                                                                                 |  |  |  |  |
|                                                                                                                                                                                                                                                                                                | 20             |                                                                | 7 5           |                 | NO RECOVERY.                                                                                                                                                                                                                                                                      |  |  |  |  |
| 650                                                                                                                                                                                                                                                                                            | 25 —<br>-<br>- |                                                                | 32 7          |                 | MODERATELY PLASTIC FINES, WET, DARK GRAY CHANGING TO MEDIUM GRAY                                                                                                                                                                                                                  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                | 30 -<br>       |                                                                | 10            |                 | SANDY GRAVEL, GAP GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES,<br>WET, MEDIUM BROWN, PEBBLES TO 1 INCH.<br>(GP)                                                                                                                                                                 |  |  |  |  |
| 640                                                                                                                                                                                                                                                                                            | 35 -<br>-<br>- |                                                                |               |                 | NO RECOVERY.                                                                                                                                                                                                                                                                      |  |  |  |  |

• )

Ì

| 630                                                            | -<br>40<br>-<br>-<br>45<br>45<br>-                                                                                                                                        | 33 10<br>15 11<br>31 12                                                                                                                     | SAND, UNIFORM, MEDIUM TO FINE, LESS THAN 1% NONPLASTIC FINES,<br>MOIST, MEDIUM BROWN.<br>(SP)<br>GRAVELLY SAND, UNIFORM, COARSE TO MEDIUM, LESS THAN 1% NONPLASTIC<br>FINES, WET, MEDIUM GRAY-EROWN.<br>(SP)<br>SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NONPLASTIC<br>FINES, SATURATED, MEDIUM BROWN, FEW PEBHLES TO 1/2 INCH. |
|----------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                                                | <br>50<br><br>                                                                                                                                                            | 23                                                                                                                                          | GRAVELLY SAND, WELL GRADEN, COARSE TO FINE, 1-3% NONPLASTIC FINES,<br>WET, MEDIUM BROWN, PEBBLES TO 1/2 INCH.                                                                                                                                                                                                                                 |
| 620                                                            |                                                                                                                                                                           | 44 14                                                                                                                                       | SAND, UNIFORM, MEDIUM TO FINE, LESS THAN 1% NONPLASTIC FINES,<br>WET, MEDIUM BROWN, 1/2 INCH GRAY CLAY SEAM NEAR BOTTOM OF RUN.                                                                                                                                                                                                               |
|                                                                |                                                                                                                                                                           | 100/0"                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                               |
|                                                                |                                                                                                                                                                           |                                                                                                                                             | END OF BORING AT 58.0'                                                                                                                                                                                                                                                                                                                        |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU<br>THE<br>2. ■2I<br>¥6I | IRES IN BLOW OR REC<br>SAMPLE DENOTE THE<br>LB HAMMER FALLING<br>OD SAMPLE SPOON I<br>IRES SHOWN OPPOSITE<br>PERCENT OF CORE RE<br>NDICATES LOCATION<br>NDICATES LOCATION | COVERY COLUMN OP<br>NUMBER OF BLOW<br>30" REQUIRED TO<br>2" OR THE DISTA<br>CROCK CORES DEN<br>COVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON | POSITE<br>US OF A<br>DRIVE<br>NCE SHOWN.<br>OTE<br>SAMPLE.<br>SAMPLE.                                                                                                                                                                                                                                                                         |
| UPI<br>SUBS<br>NUME<br>3. ₹ I<br>4. RQD<br>5. ∐. I<br>6. DATU  | NDICATES LOCATION<br>ITH NO RECOVERY.<br>CRIPT NEXT TO SYME<br>DER.<br>NDICATES LOCATION<br>ABLE.<br>- ROCK QUALITY DES<br>NDICATES DEPTH & L<br>M IS MEAN SEA LEVEL      | OF SAMPLING ATT<br>OL INDICATES SA<br>OF NATURAL GROU<br>IGNATION.<br>ENGTH OF NX COR                                                       | EMPT  MPLE  MOVER VALLEY POWER STATION - UNIT NO. 1  SHIPPINGPORT, PENNSYLVANIA  DUQUESNE LIGHT COMPANY  STONE & WEBSTER ENGINEERING CORPORATION  ING RUN.  11700 - GSK - 6                                                                                                                                                                   |



|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |        |          | (SW)                                                                                                                                                               |  |  |  |  |  |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------|-----------------------------------------|--------|----------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 40 -<br>-<br>-<br>-                                                                                                                                                                                                                                                                                      | _             | 47                                      | 10     |          | <u>GRAVELLY SAND</u> , WELL GRADED, VERY COARSE TO FINE, 3-5% NONPLASTIC<br>FINES, WET, MEDIUM BROWN, PEBBLES TO 1 INCH, SOME SANDSTONE FRAG-<br>MENTS.<br>(SW)    |  |  |  |  |  |
| 630                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 45 —<br>-<br>-                                                                                                                                                                                                                                                                                           |               | 31                                      |        |          | GRAVELLY SAND, POORLY GRADED, VERY COARSE TO FINE, MOSTLY FINE,<br>LESS THAN 1% NONPLASTIC FINES, MOIST, LIGHT TO MEDIUM GRAY-BROWN,<br>PEBBLES TO 1 INCH.<br>(SP) |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 50 —<br><br>                                                                                                                                                                                                                                                                                             |               | 57                                      | 12     |          | SAND, WELL GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC FINES,<br>WET, MEDIUM GRAY-BROWN.<br>(SW)                                                                 |  |  |  |  |  |
| 620                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                          |               | <u>200</u> ::                           |        |          | SAND, MOSTLY UNIFORM, MEDIUM TO FINE, LESS THAN 1% NONPLASTIC<br>FINES, WET, MEDIUM BROWN.                                                                         |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               | 6"                                      | 13     |          | CRAY SHALE : BOTTOM 1 INCH OF SAMPLE.                                                                                                                              |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          | LAD OF BORLING KI 97.0                                                                                                                                             |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 60                                                                                                                                                                                                                                                                                                       |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          | -                                                                                                                                                                  |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               | -                                       |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | -                                                                                                                                                                                                                                                                                                        |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | _                                                                                                                                                                                                                                                                                                        |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          | -                                                                                                                                                                  |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
|                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                          |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FIGURES SHOWN OPPOSITE BOCK CORES DENOTE                                                              |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
| 2. <b>■</b> 21<br><b>7</b> 6 I<br><b>1</b> 71<br><b>1</b> | THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>F 6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>WITH NO RECOVERY.<br>BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                       |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
| SUBS<br>NUME                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | SCRIPT ]<br>SER.                                                                                                                                                                                                                                                                                         | NEXT TO SYMBO | DL IN                                   | IDICAT | TES SAMP | CLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                                                           |  |  |  |  |  |
| <sup>3</sup> • ¥ I<br>T                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 3. ¥ INDICATES LOCATION OF NATURAL GROUND WATER 2                                                                                                                                                                                                                                                        |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |
| 4. <u>RO</u> D<br>5. ∏. I<br>6. DATU                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 4. RQD - ROCK QUALITY DESIGNATION.<br>5. [] INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL<br>5. DATUM IS MEAN SEA LEVEL |               |                                         |        |          |                                                                                                                                                                    |  |  |  |  |  |

|                                                                                                           | DUQUESNE LIGHT COMPANY SH 1 OF 1                                                               |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|-----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|-----------------|----------------------------------------------------------------------------------------------------------------|------------|-----------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. TH-5                                           |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| DATE DRILLED APRIL 15, 1974 DRILLED BY AMERICAN LOGGED BY J.P.D.                                          |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| SUMMARY OF BORING                                                                                         |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                |                 | OVERALL                                                                                                        | SAMPLE     |                       | SOUL OF POCK DESCRIPTION                                                                                                                                                                                        |  |  |  |
| LEV.                                                                                                      | EET                                                                                            | EET             | WEATHERING<br>AND<br>ROD                                                                                       | P E K      | PHI<br>DG             | JUL ON NOON DESCRIPTION                                                                                                                                                                                         |  |  |  |
| Ш                                                                                                         |                                                                                                | 8 L             | 0 25 50 75 100                                                                                                 | BLO<br>REC | G R/                  | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                          |  |  |  |
| 6776                                                                                                      | 0                                                                                              |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           | ••                                                                                             | _               | · · · · · · · · · · · · · · · · · · ·                                                                          |            |                       | _                                                                                                                                                                                                               |  |  |  |
|                                                                                                           |                                                                                                | -               |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| 670                                                                                                       |                                                                                                | 5               |                                                                                                                | 14         | 71                    | PUSHED COBELE (FROM 5'-10')                                                                                                                                                                                     |  |  |  |
| 0/0                                                                                                       |                                                                                                | -               |                                                                                                                | 25         | 72                    | -                                                                                                                                                                                                               |  |  |  |
|                                                                                                           |                                                                                                | - 10 -          |                                                                                                                | 4          | <sup>7</sup> 3        | SANDY SILT, SLIGHTLY PLASTIC, 8-12% GRAVEL TO 1.75 INCH MAXIMUM.                                                                                                                                                |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                | 2          | 4                     | LESS THAN 5% REDDISH CLAYEY MATERIAL, TRACE OF ORGANIC MATTER -<br>THROUGHOUT, SMALL ROOTS. (W.O.H KEPT SINKING)                                                                                                |  |  |  |
|                                                                                                           |                                                                                                | 15 -            |                                                                                                                |            |                       | (MI)<br>TOH 12 INCHES: <u>SANDY SILT</u> , SAME AS ABOVE, NO GRAVEL.                                                                                                                                            |  |  |  |
| 660                                                                                                       |                                                                                                |                 |                                                                                                                |            | 1                     | BOTTOM 6 INCHES: HLACK ORGANIC SANDY SILT, SLIGHTLY PLASTIC, 20-30%<br>FINE SAND, VERY SOFT, HLACK, SMALL ROOTS, ORGANIC GILY SMELL.                                                                            |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                | 28         | 6                     | (OH)<br>ORGANIC SANDY WIDELY GRADED SAND. COARSE TO FINE. 8-125 GRAVEL TO                                                                                                                                       |  |  |  |
|                                                                                                           |                                                                                                | 20              |                                                                                                                |            | Ĭ                     | 1.6 INCH MAXIMUM DIAMETER, SUB-ROUNDED, 20-30% SLICHTLY PLASTIC<br>FINES, MOIST, BROWN BLACK, ORGANIC MATERIAL THROUGHOUT.                                                                                      |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                |            | _                     | (SP-OL) -                                                                                                                                                                                                       |  |  |  |
| ( "                                                                                                       |                                                                                                | 25 -            |                                                                                                                | 25         | 7                     | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 10-15% GRAVEL TO 1.4-<br>INCH MAXIMUM DIAMETER, SUB-ANGULAR, 20-25% COARSE SAND, SUB-ANGULAR,<br>NO. 15% NONTRACTICE REPORTS COMPACT MOIST TO SATURATED SEPARATE  |  |  |  |
| 050                                                                                                       |                                                                                                |                 |                                                                                                                |            |                       | DARK YELLOW AND DARK GREEN COLORS, OILY SMELL.                                                                                                                                                                  |  |  |  |
|                                                                                                           | :                                                                                              | 30 —            |                                                                                                                |            | -                     | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 15-20% GRAVEL TO -                                                                                                                                                |  |  |  |
|                                                                                                           | 1                                                                                              | -               |                                                                                                                | 33         | 0                     | UNIFORM SAND, 8-12% NONPLASTIC FINES, MOIST, DENSE, MEDIUM BROWN,<br>OILY SMELL.                                                                                                                                |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                | 10 17      | 7                     | (SP-SW)                                                                                                                                                                                                         |  |  |  |
| <b>64</b> 0                                                                                               |                                                                                                | 35 —            |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                | -               |                                                                                                                |            | ,                     |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           | ۵<br>,                                                                                         | 40              |                                                                                                                | 40 1       | .0                    | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 15-20% GRAVEL TO -<br>7/8 INCH MAXIMUM DIAMETER, SUB-ROUNDED TO ANGULAR, 20% COARSE SAND,<br>ANGULAR ANGULAR FINE TO MEDIUM SAND LESS THAN 5% NONPLASTIC          |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                |            |                       | FINES, DAMP, DENSE, LIGHT BROWN, OILY SMELL.                                                                                                                                                                    |  |  |  |
|                                                                                                           |                                                                                                | -<br>45         |                                                                                                                | 101        | 11                    | SAND, POORLY GRADED, COARSE TO FINE, 1 PIECE OF GRAVEL, 1.4 INCH<br>DIAMETER, SUE-ROUNDED, 15-25% COARSE TO MEDIUM SAND, FINE UNIFORM, -<br>SUB-ROUNDED SAND, LESS THAN 5% NONPLASTIC FINES, DAMP, VERY DESSE;- |  |  |  |
| 630                                                                                                       |                                                                                                | -               |                                                                                                                |            |                       | LIGHT BROWN, OILY SMELL.                                                                                                                                                                                        |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                | 70         | 2                     | SAND, WIDELY GRADED, COARSE TO FINE, 1 PIECE OF GRAVEL, 1 7/8<br>INCH DIAMETER, ANGULAR, 10% EOARSE SAND, DAMP, VERY DENSE, LESS                                                                                |  |  |  |
|                                                                                                           | 1                                                                                              | 1               |                                                                                                                |            |                       | THAN 5% NONPLASTIC FINES, ROUNDED AND SUB-ANGULAR MEDIUM SAND,<br>MEDIUM BROWN, OILY SMELL.                                                                                                                     |  |  |  |
|                                                                                                           |                                                                                                | -               |                                                                                                                |            | •                     | (SP-SW)<br>SAND, UNIFORMLY GRADED, FINE SAND, LESS THAN 5% MEDIUM SAND,                                                                                                                                         |  |  |  |
| <b>6</b> 20                                                                                               |                                                                                                | 55 —            |                                                                                                                | 78 🚩 :     | 13                    | DAMP, VERY DENSE, LESS THAN 3% NONPLASTIC FINES, LIGHT BROWN,<br>TRACE OF COAL, OILY SMELL.                                                                                                                     |  |  |  |
| <b> </b>                                                                                                  |                                                                                                |                 | and free the second |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                | 60 -            |                                                                                                                |            |                       | END OF BURLING AT 57.57                                                                                                                                                                                         |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                | -               |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
|                                                                                                           |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| ļ                                                                                                         |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| 1.                                                                                                        | FIGU<br>SOIL                                                                                   | RES IN<br>SAMPL | BLOW OR REC<br>E DENOTE THE                                                                                    | OVERY CO   | )LUMN OPP<br>OF BLOWS | OSITE<br>Of A                                                                                                                                                                                                   |  |  |  |
|                                                                                                           | 140<br>A 2"<br>FTC"                                                                            | LB HAM          | MER FALLING<br>MPLE SPOON 1<br>OWN OPPOSITE                                                                    | 30" REQU   | JIRED TO<br>HE DISTAN | DRIVE<br>CE SHOWN.                                                                                                                                                                                              |  |  |  |
| THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.                          |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| ·                                                                                                         | VOINDICATES ECCATION OF SPLIT-SPOON SAMPLE.                                                    |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE SHIPPINGPORT, PENNSYLVANIA<br>NUMBER.<br>JUQUESNE LIGHT COMPANY |                                                                                                |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| 3• ·<br>4.                                                                                                | TABLE.<br>4. ROD - ROCK QUALITY DESIGNATION.                                                   |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |
| 5.<br>6.                                                                                                  | 5. LI INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL 11700 - GSK - 8 |                 |                                                                                                                |            |                       |                                                                                                                                                                                                                 |  |  |  |

.

• ~

.

-


|                                                                                                                                                                          | -<br>40<br>-                                                                                                                                                                                                                                                                      |                                                | 19                               |                      | <u>GRAVELLY SAND</u> , WIDELY GRADED, COARSE TO FINE, 20-30% GRAVEL TO 1.0-<br>INCH MAXIMUM DIAMETER, SUB-ROUNDED, 10-15% COARSE SAND, SUB-ROUNDED,<br>8-12% NONPLASTIC FINES, COMPACT, MOIST, LIGHT BROWN.                   |  |  |  |  |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------|----------------------------------|----------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| 630                                                                                                                                                                      | 45                                                                                                                                                                                                                                                                                |                                                | 23 10                            |                      | <u>GRAVELLY SILTY SAND</u> , WIDELY GRADED, COARSE TO FINE, 10-15% GRAVEL -<br>TO 1 3/8 INCH MAXIMUM DIAMETER, SUB-ANGULAR, MOSTLY FINE TO MEDIUM-<br>SAND, 12-15% NONPLASTIC FINES, MEDIST, COMPACT, LIGHT BROWN.<br>(SM-SP) |  |  |  |  |
|                                                                                                                                                                          | 50 -<br>                                                                                                                                                                                                                                                                          |                                                | 59                               |                      | SAND, WELL GRADED, COARSE TO FINE, 8-125 GRAVEL TO 1.25 INCH<br>MAXIMUM DIAMETER, SUB-ROUNDED TO ANGULAR, MOIST, VERY DENSE, 8-125<br>NONPLASTIC FINES, MEDIUM BROWN.<br>(SW)                                                 |  |  |  |  |
| 620                                                                                                                                                                      |                                                                                                                                                                                                                                                                                   |                                                | 44 12                            |                      | SAND, FINE UNIFORM SAND, 8-125 GRAVEL TO 1 3/8 INCH MAXIMUM<br>DIAMETER, SUB-ANGULAR, LESS THAN 5% NONPLASTIC FINES, DAMP, LIGHT<br>BROWN.<br>(SP)                                                                            |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      | END OF BORING AT 56.5'                                                                                                                                                                                                        |  |  |  |  |
|                                                                                                                                                                          | 60 -                                                                                                                                                                                                                                                                              |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
| 1. FIG<br>SOI                                                                                                                                                            | URES IN<br>L SAMPL                                                                                                                                                                                                                                                                | BLOW OR RECO                                   | VERY COL                         | UMN OPPO             | OSITE<br>OF A                                                                                                                                                                                                                 |  |  |  |  |
| 140<br>A 2                                                                                                                                                               | LB HAM<br>" OD SA                                                                                                                                                                                                                                                                 | MER FALLING<br>MPLE SPOON 12                   | SO" REQUI                        | RED TO I<br>DISTAN   | DRIVE<br>CE SHOWN.                                                                                                                                                                                                            |  |  |  |  |
| F1G<br>THE<br>2. <b>2</b>                                                                                                                                                | URES SH<br>PERCEN<br>INDICAT                                                                                                                                                                                                                                                      | OWN OPPOSITE<br>T OF CORE REC<br>ES LOCATION C | ROCK COR<br>COVERED.<br>F UNDIST | es denot<br>Urbed sa | BORING LOG TH-6                                                                                                                                                                                                               |  |  |  |  |
|                                                                                                                                                                          | FORTHOLOGIES LOCATION OF UNDISTURBED SAMPLE.<br>FORTHOLOGIES LOCATION OF SPLIT-SPOON SAMPLE.<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DVOUDEND 1 TO SYMBOL INDICATES SAMPLE |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |
| 3• ₹                                                                                                                                                                     | INDICAT                                                                                                                                                                                                                                                                           | ES LOCATION O                                  | F NATURA                         | L GROUND             | WATEF 2                                                                                                                                                                                                                       |  |  |  |  |
| 4. RQD - ROCK QUALITY DESIGNATION.<br>5. IL INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL<br>STONE & WEBSTER ENGINEERING C<br>11700 - GSK - 9 |                                                                                                                                                                                                                                                                                   |                                                |                                  |                      |                                                                                                                                                                                                                               |  |  |  |  |

SH\_1 OF\_1 DUQUESNE LICHT COMPANY BORING NO. 537 C 11700 BEAVER VALLEY POWER STATION J.O. NO. \_\_ SITE \_\_\_\_ 657.8 SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. TYPE OF BORING SPLIT SPOON LOCATION \_\_\_ AMERICAN DATE DRILLED MARCH 18, 1974 LOGGED BY \_\_\_\_. DRILLED BY\_ SUMMARY OF BORING \_\_ OVERALL SAMPLE OR ROCK DESCRIPTION  $\mathbf{Q}$ SOIL DEPTH FEET WEATHERING RAPHI N. E G AND BLOWS RECOV. LUI ğ TΥΡ Ц Ш RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING, BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 ଓ 657.8 SILTY SAND, UNIFORMLY GRADED, FINE, SUB-ROUNDED PARTICLES, 30% 2 1 SLIGHTLY PLASTIC FINES, VERY LOOSE, SATURATED, GRAY BROWN. (SM)5 GRAVELLY SAND, GAP-GRADED, FINE AND COARSE SAND, COARSE ANGULAR AND SUB-ROUNDED FINE PARTICLES, 30% GRAVEL TO 1 1/4" DIAMETER, SUB-ANGU-**65**0 53 LAR, MOIST, 8% SLIGHTLY PLASTIC FINES, COMPACT, MEDIUM BROWN. (SP) 10 NO RECOVERY 15 NO RECOVERY GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE PARTICLES, SUB-ROUNDED; 640 5% NONPLASTIC FINES, SATURATED, VERY LOOSE, BROWN, TWO PIECES OF ANGULAR GRAVEL IN SHOE, 1 1/4" DIAMETER. 20 (SW) GRAVELLY SAND, POORLY GRADED, FINE TO MEDIUM, SAND, ANGULAR AND SUB-ROUNDED, 3 PIECES OF GRAVEL TO 1" DIAMETER, ANGULAR, LESS THAN 3% FINES, MOIST, COMPACT, GRAY BROWN. (SP) 15 6" - SAND, POORLY GRADED, FINE SAND PARTICLES, 15% MEDIUM SAND SIZES, 25 SUB-ROUNDED, 10% NONPLASTIC FINES, MOIST, COMPACT, LIGHT BROWN. . (SM) 15 12" - SAND, WELL GRADED FROM FINE TO MEDIUM SAND PARTICLES, SUB-ROUNDED AND ANGULAR PARTICLES, 50% MEDIUM SAND, 40% FINE SAND, 10% 630 NONPLASTIC FINES, MOIST, COMPACT, BROWN. (SW) SAND, WELL GRADED FROM FINE TO MEDIUM SIZE PARTICLES, 50% MEDIUM 30 SAND, SUB-ROUNDED AND ANGULAR, 405 FINE SAND, SUB-ROUNDED, 10% 18 8 NONPLASTIC FINES, MOIST, LIGHT BROWN, LOOSE. (SW) SAND, SAME AS ABOVE, EXCEPT LESS THAN 10% MEDIUM SAND, MORE THAN 35 22 10% FINE SAND. (SP)

•

| 620            |                     |                                          | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE SAND, 10% NOMPLASTIC<br>FINES, PARTICLES ROUNDED AND ANGULAR, MOIST, COMPACT, MEDIUM BROWN,<br>15% SMALL GRAVEL. (SP) |
|----------------|---------------------|------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                | 40                  |                                          | END OF BORING AT 39.4                                                                                                                                              |
|                | -                   |                                          |                                                                                                                                                                    |
|                |                     |                                          | -                                                                                                                                                                  |
|                | -                   |                                          |                                                                                                                                                                    |
|                | 45 —                |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                | _                   |                                          |                                                                                                                                                                    |
|                | 50                  |                                          |                                                                                                                                                                    |
|                | _                   |                                          | -                                                                                                                                                                  |
|                |                     |                                          |                                                                                                                                                                    |
|                | -                   |                                          |                                                                                                                                                                    |
|                | 55 —                |                                          |                                                                                                                                                                    |
|                |                     |                                          | -                                                                                                                                                                  |
|                |                     |                                          |                                                                                                                                                                    |
|                | -                   |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                | -                   |                                          |                                                                                                                                                                    |
|                | -                   |                                          | -                                                                                                                                                                  |
|                |                     |                                          |                                                                                                                                                                    |
|                | _                   |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
|                |                     |                                          |                                                                                                                                                                    |
| 1. FIGU        | RES IN BLOW OR F    | RECOVERY COLUMN OF                       | POSITE                                                                                                                                                             |
| 140            | LB HAMMER FALLIN    | THE NUMBER OF BLOY<br>IG 30" REQUIRED TO | DRIVE                                                                                                                                                              |
| A 2"           | OD SAMPLE SPOON     | 12" OR THE DIST                          | NCE SHOWN.                                                                                                                                                         |
| THE            | PERCENT OF CORE     | TE ROCK CORES DEN<br>RECOVERED.          |                                                                                                                                                                    |
| 2. <b>2</b> 4  | NDICATES LOCATIO    | N OF UNDISTURBED                         | SAMPLE. BORING LOG 537 Z                                                                                                                                           |
|                | NDICATES LOCATIO    | N OF SPLIT-SPOON<br>N OF SAMPLING ATT    |                                                                                                                                                                    |
|                | ITH NO RECOVERY.    |                                          | BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                                           |
| NUMB           | CRIFT NEXT TO SY    | MBOL INDICATES SA                        | MPLE SHIPPINGPORT, PENNSYLVANIA                                                                                                                                    |
| 3•葉 ፲          | NDICATES LOCATIO    | N OF NATURAL GROU                        | ND WATER 2                                                                                                                                                         |
| 4. <u>RQ</u> D | - ROCK QUALITY D    | ESIGNATION.                              | Auch TA STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                    |
| 5. []. I       | NDICATES DEPTH &    | LENGTH OF NX COR                         | ING RUN 11700 - GSK - 10                                                                                                                                           |
| O. DATO        | M LO FILAN OLA LEVI |                                          |                                                                                                                                                                    |

|                   | DUQUESNE LIGHT COMPANY SH 1 OF 1                                 |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|-------------------|------------------------------------------------------------------|--------------------------------------------|-----------------------|------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE              | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 538 - |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
| DATE D            | RILLED                                                           | MARCH 20.                                  | 1974                  | DRIL       | LED BY AMERICAN LOGGED BY J.P.D.                                                                                                                                                                               |  |  |  |  |
| SUMMAR            | Y OF B                                                           | ORING                                      |                       | <u> </u>   |                                                                                                                                                                                                                |  |  |  |  |
|                   |                                                                  | OVERALL                                    | SAMPLE                | ι <u>υ</u> |                                                                                                                                                                                                                |  |  |  |  |
| LEV.<br>EET       | EET                                                              | WEATHERING<br>AND<br>ROD                   | PE V                  | H B B H    | SOIL OR ROCK DESCRIPTION                                                                                                                                                                                       |  |  |  |  |
| шш                | Ğ.                                                               | 0 25 50 75 160                             |                       | G R/       | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                          |  |  |  |  |
| 655.3             |                                                                  |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   |                                                                  |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   |                                                                  |                                            | 1                     |            | GRAVELLY SAND, GAP GRADED, COARSE TO FINE SAND, ROUNDED IN ANGULAR,<br>40% ANGULAR, GRAVEL TO 1 3/4 INCH DIAMETER, 20% COARSE SAND, ANGULAR<br>30% FINE SAND SUB-ROUNDED 10% NONPLASTIC FINES. SANDRATED, VERY |  |  |  |  |
| <b>65</b> 0 —     | 5 -                                                              |                                            |                       |            | LOOSE, DARK GREEN.<br>(SP)                                                                                                                                                                                     |  |  |  |  |
|                   |                                                                  |                                            | 60 2                  |            | GRAVELLY SAND, GAP GRABED, COARSE TO FINE SAND, 1 LARGE PIECE OF<br>GRAVEL, ANGULAR, 2 INCH DIAMETER, 15% SMALLER GRAVEL, 50% COARSE                                                                           |  |  |  |  |
|                   | 10                                                               |                                            |                       |            | SUB-ROUNDED SAND, 30% FINE SAND, 5% NONPLASTIC FINES, MOIST, COMPACT<br>GRAY GREEN.<br>(SP)                                                                                                                    |  |  |  |  |
|                   |                                                                  |                                            | 5 7                   | ,          |                                                                                                                                                                                                                |  |  |  |  |
| 640 —             | 15 _                                                             |                                            |                       |            | SAND, WIDELY GRADED, COARSE TO FINE, SUB-ROUNDED, 30% COARSE, 30%<br>MEDIUM, 30% FINE SAND, 5% NONPLASTIC FINES, DAMP, VERY LOOSE, LIGHT                                                                       |  |  |  |  |
|                   |                                                                  |                                            |                       |            | (SP-SW)<br>GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE SAND, GRAVEL TO 1 INCH                                                                                                                                 |  |  |  |  |
|                   | 20 -                                                             | 1                                          | 7 75                  |            | DIAMETER, 15% GRAVEL, 20% COARSE SAND, 20% MEDIUM SAND, 30% FINE SAND<br>5% NONPLASTIC FINES, MOIST, VERY LOOSE, BROWN.<br>(SP-SW)                                                                             |  |  |  |  |
|                   | -                                                                |                                            |                       |            | GRAVELLY SAND SAME AS A BOVE EXCEPT 1 OF NONDLASTIC ETNES AND                                                                                                                                                  |  |  |  |  |
| 630               | 25 -                                                             |                                            | 19 6                  |            | GRAVEL ONLY TO 3/4 INCH DIAMETER.                                                                                                                                                                              |  |  |  |  |
|                   | -                                                                | 1                                          |                       | 1          |                                                                                                                                                                                                                |  |  |  |  |
|                   | 20 -                                                             |                                            | 22                    |            | GRAVELLY SAND, SAME AS SAMPLE #6.                                                                                                                                                                              |  |  |  |  |
|                   | - 1                                                              | 4                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            | 11 8                  |            | (SP-SW)<br>LAST 4 INCHES: <u>GRAVELLY SAND</u> , WIDELY GRADED, FROM FINE TO COARSE -                                                                                                                          |  |  |  |  |
| 620               | 35 -                                                             | 1                                          |                       | 2          | SAND SIZES, GRAVEL TO 1 1/2 INCHES, 25% GRAVEL, ANGULAR, 30% MEDIUM_<br>AND COARSE SAND, 10% NONBLASTIC FINES, 35% FINE SAND, DAMP, COMPACT,<br>LIGHT BROWN. (SP)                                              |  |  |  |  |
|                   |                                                                  |                                            | 100/3"                |            | TOP 2.5 INCHES: <u>GRAVELLY SAND</u> , SAME AS ABOVE. (SP)<br>BOTTOM 2.5 INCHES: DECOMPOSED <u>CLAYEY SHALE</u> , HIGHLY PLASTIC, DAMP,                                                                        |  |  |  |  |
|                   | 40 -                                                             |                                            |                       |            | END OF BORING AT 39.3'                                                                                                                                                                                         |  |  |  |  |
|                   | -                                                                | ]                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                | 4                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                | 1                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   |                                                                  | ]                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                | 1                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   | -                                                                |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |
|                   |                                                                  | 1                                          |                       |            |                                                                                                                                                                                                                |  |  |  |  |
| 1. FIG            | JRES IN                                                          | BLOW OR REC                                | COVERY COL            | LUMN OPF   | POSITE                                                                                                                                                                                                         |  |  |  |  |
| 140<br>A 2'       | LB HAM<br>OD SA                                                  | MER FALLING<br>MPLE SPOON 1                | 30" REQUI             | RED TO     | DRIVE<br>ICE SHOWN.                                                                                                                                                                                            |  |  |  |  |
| THE<br>2. 21      | PERCEN<br>NDICAT                                                 | T OF CORE RE<br>ES LOCATION<br>ES LOCATION | OF UNDIST             | URBED S    | AMPLE. BORING LOG 538 Z                                                                                                                                                                                        |  |  |  |  |
|                   | NDICAT                                                           | ES LOCATION<br>RECOVERY.                   | OF SAMPLI             | ING ATTE   | MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                                                                                 |  |  |  |  |
| NUME<br>3• ¥ I    | BER.                                                             | ES LOCATION                                | OF NATURA             | LES SAM    | D WATEF 2 SHIPPINGPORT, PENNSTLVANIA<br>DUQUESNE LEGHT COMPANY                                                                                                                                                 |  |  |  |  |
| 4. ROD<br>5. ∐. I | - ROCK<br>NDICAT                                                 | QUALITY DES<br>ES DEPTH & L                | IGNATION.<br>ENGTH OF | NX CORI    | NG RUN                                                                                                                                                                                                         |  |  |  |  |
| 6. DATU           | 5. IL INDICATES DEPTH & LENGTH OF NX CORING RUN                  |                                            |                       |            |                                                                                                                                                                                                                |  |  |  |  |

}

| DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                               |                                                                                                                                              |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------|--------------------------------------------------|--------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO BORING NO 539 C<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 640.75 |                                                                                                                                              |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
| DATE D                                                                                                                                         | RILLED                                                                                                                                       | MARCH 22,                                                  | 1974                                             | DRIL               | LED BY LOGGED BYJ.P.D.                                                                                                                                                                                                                                                                                                   |  |  |  |
|                                                                                                                                                |                                                                                                                                              | ·                                                          |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
| Ч.                                                                                                                                             | H                                                                                                                                            | OVERALL<br>WEATHERING                                      | SAMPLE                                           | HIC<br>6           | SOIL OR ROCK DESCRIPTION                                                                                                                                                                                                                                                                                                 |  |  |  |
| ELE<br>FEE                                                                                                                                     | DEP'<br>FEE                                                                                                                                  | RQD<br>0 25 50 75 100                                      | BLOWS                                            | G R A P<br>LO(     | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                                                                                                                                   |  |  |  |
| 640.75                                                                                                                                         |                                                                                                                                              | -                                                          |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            | PUSH                                             | ~                  | ÷                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                | - 5                                                                                                                                          |                                                            | 5 2<br>9 3                                       |                    | GRAVELLY SAND, UNIFORMLY GRADED, FINE SAND, SUBROUNDED; 25% GRAVEL,<br>SUBROUNDED, TO 1 <sup>1</sup> / <sub>2</sub> " DIA., 15% NONPLASTIC FINES, SATURATED, LOOSE TO<br>VERY LOOSE, GRAY BROWN.<br>(SP)                                                                                                                 |  |  |  |
| 630 —                                                                                                                                          | 10 —<br>—<br>—<br>15 —                                                                                                                       | •                                                          | 21 <b>4</b><br>18 <b>5</b>                       |                    | TOP 4"- <u>SAND</u> , WIDELY GRADED, FINE TO MEDIUM SAND, SUBROUNDED ANGULAR-<br>PARTICLES, 5% FINES, NONPLASTIC, DAMP, LOOSE BROWN (SP); BOTTOM 8"- <u>SAND</u> , WELL GRADED, EVEN DISTRIBUTION OF PARTICLE SIZE AND SHAPE, 10%<br>GRAVEL TO 3/4" DIA., ANGULAR; 10% NONPLASTIC FINES, SATURATED, LOOSE<br>BROWN. (SW) |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            | 19 🚩 6                                           |                    | SAME AS ABOVE EXCEPT ALL DAMP                                                                                                                                                                                                                                                                                            |  |  |  |
| 620                                                                                                                                            | 20 -                                                                                                                                         |                                                            | 21 7                                             |                    | SAME AS ABOVE, WIDELY GRADED. (SP)<br>BOTTOM 3"- <u>SAND</u> , WIDELY GRADED, MEDIUM TO COARSE SAND, SUBROUNDED,<br>10% SLIGHTLY PLASTIC FINES, LIGHT BROWN.<br>(SM)                                                                                                                                                     |  |  |  |
|                                                                                                                                                | 25                                                                                                                                           |                                                            | 33 8                                             |                    | TOP 3"-SAND, WIDELY GRADED, SUBROUNDED AND ANGULAR, MOIST, BROWN<br>MIDDLE 12" - <u>SAND</u> , UNIFORMLY GRADED, COARSE SAND, 5% NONPLASTIC FINE<br>ANGULAR, MOIST, BROWN. (SP)                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            | <u>100</u> 9<br>2"                               |                    | 15% SLIGHTLY PLASTIC FINES, BROWN, MOIST. (SP)                                                                                                                                                                                                                                                                           |  |  |  |
|                                                                                                                                                | 30 -                                                                                                                                         |                                                            |                                                  |                    | END OF HEHLING AT 28.3'                                                                                                                                                                                                                                                                                                  |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    | -                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    | -<br>-<br>-                                                                                                                                                                                                                                                                                                              |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    | -                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    | -                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    | -<br>                                                                                                                                                                                                                                                                                                                    |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    | -                                                                                                                                                                                                                                                                                                                        |  |  |  |
|                                                                                                                                                | _                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                | -                                                                                                                                            |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
|                                                                                                                                                |                                                                                                                                              |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2'                                                                                                                 | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |
| FIGU<br>THE<br>2. <b>2</b> 1<br><b>76</b> 1                                                                                                    | JRES SH<br>PERCEN<br>INDICAT<br>INDICAT                                                                                                      | OWN OPPOSITE<br>T OF CORE RE<br>ES LOCATION<br>ES LOCATION | E ROCK COR<br>COVERED.<br>OF UNDIST<br>OF SPLIT- | URBED S            | AMPLE. BORING LOG 539 Z                                                                                                                                                                                                                                                                                                  |  |  |  |
|                                                                                                                                                | NDICAT<br>NITH NO<br>SCRIPT<br>BER.                                                                                                          | ES LOCATION<br>RECOVERY.<br>NEXT TO SYME                   | OF SAMPLI                                        | NG ATTE<br>TES SAM | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                                                                                                                                                                         |  |  |  |
| 3. ¥ I<br>T<br>4. ROD<br>5. ∐ I                                                                                                                | NDICAT<br>ABLE.<br>- ROCK<br>NDICAT                                                                                                          | ES LOCATION<br>QUALITY DES<br>ES DEPTH & L                 | OF NATURA<br>IGNATION.<br>ENGTH OF               | L GROUN            | NG RUN I                                                                                                                                                                                                                                                                                                                 |  |  |  |
| J. DAIO                                                                                                                                        | 6. DATUM IS MEAN SEA LEVEL.                                                                                                                  |                                                            |                                                  |                    |                                                                                                                                                                                                                                                                                                                          |  |  |  |

Contraction of the local distribution of the

and the second second second

And the second second

.

SH1\_OF\_1 DUQUESNE LIGHT COMPANY 11700 540 🗲 BEAVER VALLEY POWER STATION BORING NO. SITE \_ \_ J.O. NO. \_ TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 646.1 DRILLED BY AMERICAN MARCH 25, 1974 LOGGED BY \_\_\_\_\_J.E.P. DATE DRILLED . SUMMARY OF BORING ..... OVERALL SAMPLE  $\overline{\mathbf{O}}$ SOIL OR ROCK DESCRIPTION DEPTH FEET Ц С WEATHERING RAPHI LOG Ц AND BLOWS RECOV. TYPE Е E RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 G 646.1 NO RECOVERY. WOH SILTY SAND, WIDELY GRADED, 8-12% SUBROUNDED GRAVEL TO 1.0 IN. MAX., 10 2 COARSE TO FINE, MOSTLY COARSE AND FINE, 15-20% NONPLASTIC FINES, 5 LOOSE, SATURATED, DARK BROWN. (SM)640 SAND, POORLY GRADED, MEDIUM AND FINE, MOSTLY MEDIUM, 1-5% NONPLASTIC 14 FINES, COMPACT, DARK BROWN. (SP) 10 GRAVELLY SAND, POORLY GRADED, 5-10% ROUNDED GRAVEL TO 1.2 IN. MAX., 16 COARSE AND MEDIUM SAND, COMPACT, DARK BROWN. (SP) 15 630 GRAVELLY SAND, SIMILAR TO S #4, EXCEPT POCKET OF LIGHT BROWN SILTY 26 5 SAND IN MIDDLE SAMPLE. 20 (SP) SAND, POORLY GRADED, COARSE TO FINE, MOSTLY MEDIUM, 1-5% NONPLASTIC-FINES, COMPACT, DARK BROWN, POCKET OF LIGHT BROWN SILTY SAND AT 37 6 BOTTOM OF SAMPLE. 25 (SP) 620 <u>100</u> 2" NO RECOVERY. 7 END OF BORING AT 27.7' 30

| 1. FIGURES IN BLOW OR REC<br>SOIL SAMPLE DENOTE THE<br>140 LB HAMMER FALLING                        | COVERY COLUMN OPPOSITE<br>E NUMBER OF BLOWS OF A<br>30" BEQUIRED TO DRIVE                                                                                                           | E<br>A                                                                |  |  |  |  |  |  |  |  |  |
|-----------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------|--|--|--|--|--|--|--|--|--|
| A 2" OD SAMPLE SPOON<br>FIGURES SHOWN OPPOSITE<br>THE PERCENT OF CORE RE<br>2. 2 INDICATES LOCATION | L2" OR THE DISTANCE SH<br>E ROCK CORES DENOTE<br>ECOVERED.                                                                                                                          | BORING LOG 540 C                                                      |  |  |  |  |  |  |  |  |  |
| V6 INDICATES LOCATION                                                                               | OF SPLIT-SPOON SAMPLE<br>OF SAMPLING ATTEMPT                                                                                                                                        | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT DENNSYLVANIA |  |  |  |  |  |  |  |  |  |
| NUMBER.<br>3. X INDICATES LOCATION<br>TABLE.                                                        | SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>NUMBER.<br>. VINDICATES LOCATION OF NATURAL GROUND WATEF 2<br>TABLE.<br>POD DOCK CHALLEY DECLONATION                                   |                                                                       |  |  |  |  |  |  |  |  |  |
| 4. HQD - ROCK QUALITY DES<br>5. 1. INDICATES DEPTH & I<br>6. DATUM IS MEAN SEA LEVEL.               | TABLE.<br>ROD - ROCK QUALITY DESIGNATION.<br>I INDICATES DEPTH & LENGTH OF NX CORING RUN<br>DATUM IS MEAN SEA LEVEL.<br>STONE & WEBSTER ENGINEERING CORPORATION<br>11700 - GSK - 13 |                                                                       |  |  |  |  |  |  |  |  |  |

| DU(                                                                                                                         | QUESNE LIGHT COMPANY SH 1 OF 1                                                                                                             |
|-----------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|
| SITE BEAVER VALLEY POWER STATION                                                                                            | J.O. NO. 11700 BORING NO. 5412                                                                                                             |
| DATE DRILLED MARCH 26, 1974 DRIL                                                                                            | LED BYAMERICAN LOGGED BY                                                                                                                   |
| SUMMARY OF BORING                                                                                                           |                                                                                                                                            |
| TOVERALL SAMPLE U                                                                                                           | SOIL OR ROCK DESCRIPTION                                                                                                                   |
|                                                                                                                             | FIELD AND LABORATORY TEST RESULTS. SOLL STRATA DESCRIPTION & LITHOLOGY                                                                     |
|                                                                                                                             | OR JOINTING BEDDING AND FAULTING AND TEXTURE DESCRIPTIONS                                                                                  |
| 650.9                                                                                                                       |                                                                                                                                            |
| 650 WOH                                                                                                                     | NO RECOVERY                                                                                                                                |
|                                                                                                                             | NO RECOVERY<br>GRAVELLY SAND, POORLY GRADED, 15-20% SUBANGULAR GRAVEL TO 1.2 INCH                                                          |
| 19 4                                                                                                                        | SATURATED, DARK BROWN. (SP)<br>GRAVELLY SAND, SIMILAR TO SS #3, EXCEPT COMPACT, 2 INCH POCKET                                              |
|                                                                                                                             | (SP)                                                                                                                                       |
|                                                                                                                             | SILTY SAND, WIDELY GRADED, 5-10% ANGULAR GRAVEL TO 0.8 INCH<br>MAXIMUM. COARSE TO FINE SAND. MOSTLY FINE, 15-25% NONPLASTIC FINES,         |
|                                                                                                                             | COMPACT, DARK AND LIGHT BROWN.<br>(SM)                                                                                                     |
|                                                                                                                             | NO RECOVERS                                                                                                                                |
|                                                                                                                             | SILTY SAND, WIDELY GRADED, 5-10% ANGULAR GRAVEL TO 1.2 INCH MAXIMUM<br>COARSE TO FINE SAND, MOSTLY FINE, PO-15% NONPLASTIC FINES, COMPACT, |
| 20 - <u>100</u> 8<br>2"                                                                                                     | DARK BROWN. (SM)<br>NO RECOVERY                                                                                                            |
|                                                                                                                             | END OF BORING AT 21.0'                                                                                                                     |
| 25                                                                                                                          |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             | -                                                                                                                                          |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
|                                                                                                                             |                                                                                                                                            |
| 1. FIGURES IN BLOW OR RECOVERY COLUMN OPP                                                                                   | OSITE                                                                                                                                      |
| SOIL SAMPLE DENOTE THE NUMBER OF BLOWS<br>140 LB HAMMER FALLING 30" REQUIRED TO<br>A 2" OD SAMPLE SPOON 12" OF THE STORAGE  | OF A<br>DRIVE<br>CE SHOLW                                                                                                                  |
| FIGURES SHOWN OPPOSITE ROCK CORES DENO<br>THE PERCENT OF CORE RECOVERED.                                                    | TE BORING LOG KULT                                                                                                                         |
| 6 INDICATES LOCATION OF UNDISTURBED S<br>[76 INDICATES LOCATION OF SPLIT-SPOON S<br>[71 INDICATES LOCATION OF SAMPLING ATTE | AMPLE.<br>AMPLE.<br>MPT BRAVER VALLEY POWER STATION _ INTY NO 1                                                                            |
| WITH NU RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAM<br>NUMBER.                                                      | PLE DUQUESNE LIGHT COMPANY                                                                                                                 |
| 3. ¥ INDICATES LOCATION OF NATURAL GROUN<br>TABLE.<br>4. ROD - ROCK QUALITY DESIGNATION                                     | D WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                                          |
| 5. 1. INDICATES DEPTH & LENGTH OF NX CORI<br>6. DATUM IS MEAN SEA LEVEL                                                     | NG RUN 1 2 2 2 11700 - GSK - 14                                                                                                            |

•• )

| <u></u>                                                          | DUQUESNE LIGHT COMPANY SH 1 OF 1    |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|------------------------------------------------------------------|-------------------------------------|----------------------------------------------|------------------------------------|----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 542 C |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
| DATE D                                                           | BORING                              | MARCH 27, 1                                  | 974                                | DRIL                             | LED BY AMERICAN LOGGED BY J.E.P.                                                                                                       |  |  |  |
| SUMMAR                                                           | Y OF B                              | ORING                                        |                                    |                                  | ٢                                                                                                                                      |  |  |  |
|                                                                  |                                     | OVERALL                                      | SAMPLE                             | <u>د</u>                         | SOUL OR ROCK DESCRIPTION                                                                                                               |  |  |  |
| ELEV.<br>FEET                                                    | DEPTH                               | WEATHERING<br>AND<br>RQD<br>0 28 50 75 100   | BLOWS                              | 6 RAPHI<br>LOG                   | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |
| 652 2                                                            |                                     |                                              | <b></b>                            |                                  |                                                                                                                                        |  |  |  |
| 053.3                                                            | -                                   |                                              | WOH 1                              |                                  | NO RECOVERY                                                                                                                            |  |  |  |
| 650 <del>- "</del>                                               |                                     |                                              | $^{2}$                             | 2                                | NO RECOVERY                                                                                                                            |  |  |  |
|                                                                  | 5                                   |                                              | 3                                  | 3                                | GRAVELLY SAND, POORLY GRADED, 10-20% ANGULAR GRAVEL TO 1.0 INCH                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  | MAXIMUM, CORRECTO FINE SAND, MOSTLY COARSE, 1-3% NONPLASTIC FINES,<br>SATURATED, LOOSE, DARK BROWN.                                    |  |  |  |
|                                                                  | 10 —                                |                                              |                                    | 5                                | NO RECOVERY                                                                                                                            |  |  |  |
| 640                                                              | -                                   |                                              | 10                                 | 7                                | SAND, WELL GRADED, COARSE TO FINE, 1-5% NONPLASTIC FINES, COMPACT,<br>DARK BROWN. (SW)                                                 |  |  |  |
| 040                                                              | <br>15                              |                                              | 11                                 | 8                                | SILTY SAND, WEDELY GRADED, COARSE TO FINE, MOSTLY FINE, 5-10%                                                                          |  |  |  |
|                                                                  | _                                   |                                              |                                    |                                  | (SM_SP) · _                                                                                                                            |  |  |  |
|                                                                  |                                     |                                              | 17                                 | 9                                | SILTY SAND, SAME AS SS #8.                                                                                                             |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
| 630 —                                                            | -                                   |                                              | 19                                 | •                                | SILTY SAND, POORLY GRADED, MEDIUM AND FINE, MOSTLY FINE, 10-15%                                                                        |  |  |  |
|                                                                  | 25                                  |                                              |                                    |                                  | NONPLASTIC FINES, COMPACT, DARK BROWN.                                                                                                 |  |  |  |
|                                                                  |                                     |                                              |                                    | •                                | STUTY SAND SIMILAR WYCEPT 15-20% NONPLASTIC FINES.                                                                                     |  |  |  |
|                                                                  | 30 <b></b><br>-                     |                                              | 19 🕶 1                             |                                  | (SM)                                                                                                                                   |  |  |  |
|                                                                  | _                                   |                                              | $\frac{100}{67}$ 1                 | 2                                | DARK GREEN SANDSTONE.                                                                                                                  |  |  |  |
|                                                                  | 35 —                                |                                              |                                    |                                  | END OF BORING AT 33.5'                                                                                                                 |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  | -                                                                                                                                      |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  | -                                   |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  | -                                   |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  | -                                                                                                                                      |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    | <i>6</i> .                       |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
|                                                                  |                                     |                                              |                                    |                                  |                                                                                                                                        |  |  |  |
| 1. FIG<br>SOII<br>140                                            | JRES IN<br>SAMPL<br>LB HAM          | BLOW OR REC<br>E DENOTE THE<br>MER FALLING   | OVERY CO<br>NUMBER<br>30" REQU     | LUMIN OPP<br>OF BLOWS<br>IRED TO | OSITE<br>OF A<br>DRIVE                                                                                                                 |  |  |  |
| A 2"<br>FIGU<br>THE                                              | ' UD SA<br>JRES SH<br>PERCEN        | MPLE SPOON 1<br>OWN OPPOSITE<br>T OF CORE RE | 2" OR TH<br>ROCK CO                | E DISTAN<br>RES DENO             | CE SHOWN.                                                                                                                              |  |  |  |
| 2. <b>■2</b> I<br><b>76</b> I                                    | NDICAT                              | ES LOCATION<br>ES LOCATION                   | OF UNDIS                           | TURBED S.<br>-SPOON S.           | AMPLE. BORING LOG 542 T                                                                                                                |  |  |  |
|                                                                  | ITH NO<br>CRIPT                     | RECOVERY.<br>NEXT TO SYMB                    | OF SAMPLI                          | ING ATTE:<br>Ates sam            | MPT<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE.LIGHT COMPANY                                |  |  |  |
| 3. ¥ I<br>T<br>4. <u>RO</u> D<br>5. ∐ I                          | NDICAT<br>ABLE.<br>- ROCK<br>NDICAT | ES LOCATION<br>QUALITY DES<br>ES DEPTH & L   | OF NATURA<br>IGNATION.<br>ENGTH OF | AL GROUN<br>NX CORTI             | NG RUN I OTH                                                                                                                           |  |  |  |
| 6. DATU                                                          | MISM                                | EAN SEA LEVEL                                |                                    |                                  | 11700 - GSK - 15                                                                                                                       |  |  |  |

· · ·)

|                                                                  | DUQUESNE LIGHT COMPANY SH 1 OF 1     |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|------------------------------------------------------------------|--------------------------------------|------------------------------------------------|---------------------------------------|---------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 543 2 |                                      |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
| DATE D                                                           | RILLED                               | MARCH 26-27,                                   | 1974                                  | DRIL                            | LED BY LOGGED BY J.E.P.                                                                                                                                                                    |  |  |  |
|                                                                  |                                      |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
| <u>у</u> н                                                       | ΞĻ                                   | OVERALL<br>WEATHERING                          | SAMPLE                                | С<br>Ч                          | SOIL OR ROCK DESCRIPTION                                                                                                                                                                   |  |  |  |
|                                                                  | DEPT<br>FEE                          | AND<br>RQD<br>0 25 50 75 100                   | BLOWS<br>RECOV.<br>TYPE               | G R API                         | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                      |  |  |  |
| 672.44                                                           | •                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
| 670                                                              | -                                    |                                                | 31 1                                  |                                 | SILTY SAND, WIDELY GRADED, 8-12% SUBROUNDED GRAVEL TO 1.0 IN. MAX.,<br>COARSE TO FINE SAND, MOSTLY FINE, 10-15% NONPLASTIC FINES, DRY,<br>DENSE, MEDIUM BROWN AND BLACK, MUCH COAL<br>(SM) |  |  |  |
|                                                                  | 5<br><br>                            |                                                | 12 🗾 2                                |                                 | SANDY GRAVEL, POORLY GRADED, ANGULAR TO 1.0 IN. MAX., COARSE TO FINE<br>SAND, MOSTLY COARSE, 1-5% NONPLASTIC FINES, SATURATED, COMPACT, DARK<br>BROWN.                                     |  |  |  |
| 660                                                              | 10 —<br>-                            |                                                | 7 3                                   |                                 | (GP)<br><u>SANDY SILT</u> , NONPLASTIC, 20-30% COARSE TO FINE SAND, MOSTLY FINE,<br>FIRM, DARK BROWN, 1.2 IN. GRAVEL AT TOP.<br>(ML)                                                       |  |  |  |
| 000                                                              |                                      |                                                |                                       |                                 | END OF BORING AT 12.0'                                                                                                                                                                     |  |  |  |
|                                                                  | -                                    |                                                | 3                                     |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
| -                                                                | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 4                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      | 4                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      | •                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      | ·                                              | r<br>T                                |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | ]                                              |                                       |                                 | -                                                                                                                                                                                          |  |  |  |
|                                                                  | -                                    |                                                |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      | 4                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 4                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 1                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 1                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | ]                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  |                                      | 1                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 4                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
|                                                                  | -                                    | 1                                              |                                       |                                 |                                                                                                                                                                                            |  |  |  |
| 1. FIG<br>SOI<br>140                                             | URES IN<br>L SAMPI<br>LB HAN         | N BLOW OR REALE DENOTE THIS                    | COVERY COI<br>E NUMBER (<br>30" REQUI | LUMN OP<br>DF BLOWS<br>IRED TO  | POSITE<br>S OF A<br>DRIVE                                                                                                                                                                  |  |  |  |
| FIG<br>THE                                                       | URES SH<br>PERCEN                    | IOWN OPPOSITI                                  | E ROCK COF                            | RES DENC                        | BORTNE LOG 5/3 7                                                                                                                                                                           |  |  |  |
|                                                                  | INDICAT<br>INDICAT<br>INDICAT        | TES LOCATION<br>TES LOCATION<br>TES LOCATION   | OF UNDIST<br>OF SPLIT-<br>OF SAMPLI   | URBED S<br>-SPOON S<br>ING ATTE | SAMPLE.                                                                                                                                                                                    |  |  |  |
| SUB<br>NUM<br>3• <del>Ş</del>                                    | WITH NO<br>SCRIPT<br>BER.<br>INDICAT | ) RECOVERY.<br>NEXT TO SYMP<br>TES LOCATION    | BOL INDICA<br>OF NATURA               | TES SAN                         | MPLE BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSELVANIA<br>DUQUESNE LIGHT COMPANY                                                                                      |  |  |  |
| 4. <u>ROD</u><br>5. ∐.<br>6. DAT                                 | - ROCK<br>INDICAT                    | QUALITY DES<br>CES DEPTH & I<br>MEAN SEA LEVEL | SIGNATION.<br>LENGTH OF               | NX CORI                         | ING RUN I STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                          |  |  |  |

• )

SH\_\_\_OF\_\_\_ DUQUESNE LIGHT COMPANY J.O. NO. \_\_\_\_\_\_\_\_\_ BORING NO. 543 At BEAVER VALLEY POWER STATION SITE \_\_ TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. \_\_\_\_\_ 672.8 LOGGED BY \_\_\_\_\_\_ J.E.P. AMERICAN DATE DRILLED MARCH 27-29, 1974 DRILLED BY SUMMARY OF BORING \_ OVERALL SAMPLE ပ OR ROCK DESCRIPTION DEPTH FEET SOIL Ň WEATHERING RAPHI LOG ET AND BLOWS RECOV. ΤΥΡΕ Ц Ē RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY 0 25 50 75 100 G 672.8 670 5 27 SILTY SAND, WIDELY GRADED, 10-15% ANGULAR GRAVEL TO 1.0 IN. MAX., COARSE TO FINE SAND, 15-20% NONPLASTIC FINES, SATURATED, COMPACT, DARK BROWN. (SM)10 SANDY SILT, NONPLASTIC TO SLIGHTLY PLASTIC, 25-35% MEDIUM TO FINE 6 2 SAND, MOSTLY FINE, FIRM, DARK BROWN, TRACE COAL. 660 (ML) 15 SANDY SILT, NONPLASTIC, 20-30% FINE SAND, SOFT, BLACK AND BROWN, OILY SMELL. 3 3 (ML) 20 GRAVELLY SAND, WIDELY GRADED, 10-20% ANGULAR GRAVEL TO 1.0 IN. MAX., COARSE TO FINE, MOSTLY FINE, 5-10% NONPLASTIC FINES, DENSE BLACK 33 4 650 AND GRAY. (SP) 25 SILTY SAND, WIDELY GRADED, COARSE TO FINE SAND, MOSTLY FINE, 10-20%-16 NONPLASTIC FINES, COMPACT, MEDIUM BROWN. (SM)30 6 GRAVELLY SAND, WELL GRADED, 5-10% ROUNDED GRAVEL TO 1.0 IN. MAX., 11 COARSE TO FINE SAND, 1-5% NONPLASTIC FINES, COMPACT, DARK BROWN. (SW) 640 35

|                                      | -                                                              |                                                 | 10 🚩 7                             |                                | SILTY SAND, UNIFORM, FINE, 10-15% NONPLASTIC FINES, LOOSE, LIGHT<br>BROWN.<br>(SM)                                                                     |  |  |  |
|--------------------------------------|----------------------------------------------------------------|-------------------------------------------------|------------------------------------|--------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| 630                                  | 40 -                                                           |                                                 | 18 8                               |                                | <u>GRAVELLY SAND</u> , WELL GRADED, 10-15% ROUNDED GRAVEL TO 1.0 IN. MAX.,<br>COARSE TO FINE, 3-8% NONPLASTIC FINES, VERY DENSE, MEDIUM BROWN.<br>(SW) |  |  |  |
|                                      | 45 —<br>—<br>—                                                 |                                                 | 29                                 |                                | <u>GRAVELLY SAND</u> , SAME AS S #8.<br>(SW)                                                                                                           |  |  |  |
| 620                                  | 50 —<br>                                                       |                                                 | 30 10                              | 2                              | SAND, WELL GRADED, 3-8% SUBANGULAR GRAVEL TO 0.7 IN. MAX., COARSE<br>TO FINE SAND, 3-8% NONPLASTIC FINES, COMPACT, DARK BROWN.<br>(SW)                 |  |  |  |
| 020                                  | 55                                                             |                                                 | <u>100</u><br>3"                   |                                | NO RECOVERY.                                                                                                                                           |  |  |  |
|                                      |                                                                |                                                 |                                    |                                | TOP OF ROCK AT 55.0'<br>END OF BORING AT 55.3'                                                                                                         |  |  |  |
|                                      | 1                                                              |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
|                                      | -                                                              |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
|                                      |                                                                |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
|                                      | -                                                              |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
|                                      |                                                                |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
| 1. FIGU<br>SOII                      | JRES IN<br>SAMPL                                               | BLOW OR RECO<br>E DENOTE THE                    | VERY COL<br>NUMBER O               | UMN OPP                        | OSITE<br>OF A                                                                                                                                          |  |  |  |
| A 2"<br>FIGU                         | LB HAM<br>OD SA<br>JRES SH                                     | MER FALLING 3<br>MPLE SPOON 12<br>OWN OPPOSITE  | 30" REQUI<br>2" OR THE<br>ROCK COR | RED TO (<br>DISTAN<br>ES DENO: | DRIVE<br>CE SHOWN.<br>CE                                                                                                                               |  |  |  |
| 2. ₩21<br>761                        | PERCEN<br>NDICAT<br>NDICAT                                     | T OF CORE REC<br>ES LOCATION C<br>ES LOCATION C | OVERED.<br>DF UNDIST<br>DF SPLIT-  | URBED SA                       | AMPLE. BORING LOG 543 A T                                                                                                                              |  |  |  |
|                                      | WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE |                                                 |                                    |                                |                                                                                                                                                        |  |  |  |
| 3• <del>¥</del> I<br>T               | NDICATI                                                        | ES LOCATION O                                   | F NATURAI                          | L GROUNT                       | WATER 2                                                                                                                                                |  |  |  |
| 4. <u>RO</u> D<br>5. ∐. I<br>6. DATU | - ROCK<br>NDICATH<br>M IS MH                                   | QUALITY DESI<br>ES DEPTH & LE<br>SAN SEA LEVEL. | GNATION.<br>NGTH OF 1              | NX CORIN                       | IG RUN.                                                                                                                                                |  |  |  |

| DUQUESNE LIGHT COMPANY SH 1 OF 1 |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|----------------------------------|------------------------------------------------------------------|---------------------------|-------------------------------------------------|-----------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE                             | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 544 2 |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
| DATE D                           | RILLE                                                            | D MARCH 30                | LOCATIO                                         |                             | LED BY AMERICAN LOGGED BY JEE.P.                                                                                                                                              |  |  |  |  |
|                                  | IY OF                                                            | BORING                    |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  | OVERAL                    | L SAMPLE                                        | U                           | COUL OD DOCK DESCONDION                                                                                                                                                       |  |  |  |  |
| LEV.                             | PTH<br>FFT                                                       |                           |                                                 | Hd<br>90                    | SUL OR RUCK DESCRIPTION                                                                                                                                                       |  |  |  |  |
|                                  |                                                                  | 0 25 50 75                |                                                 | C R4                        | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                         |  |  |  |  |
| (                                |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
| 0/4.1/                           |                                                                  | -                         |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
| (70                              |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
| 070                              | 5.                                                               |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  | 4                         |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  | 10                                                               |                           |                                                 |                             | -                                                                                                                                                                             |  |  |  |  |
|                                  |                                                                  | -                         | 5 1                                             |                             | SILTY SAND, WIDELY GRADED, 8-12% ANGULAR GRAVEL TO 0.7 IN. MAX.,<br>COARSE TO FINE SAND, MOSTLY FINE, 20-30% NONPLASTIC FINES, LOOSE,<br>MOIST DARK BROWN AND BLACK, ORGANIC. |  |  |  |  |
| 660 —                            | 15                                                               | ]                         |                                                 |                             | (SM)                                                                                                                                                                          |  |  |  |  |
|                                  |                                                                  | 4                         | 1 2                                             |                             | ORGANIC SILT, NONPLASTIC, 25-35% FINE SAND, VERY LOOSE, SATURATED,                                                                                                            |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             | (OL)                                                                                                                                                                          |  |  |  |  |
|                                  | 20                                                               |                           | 70 3                                            |                             | SILTY SAND, WIDELY GRADED, 5-10% ANGULAR GRAVEL TO 1.0 IN. MAX., -                                                                                                            |  |  |  |  |
|                                  |                                                                  | ]                         |                                                 |                             | COARSE TO FINE SAND, MOSTLY FINE, 15-20% NONPLASTIC FINES, VERY<br>DENSE, LIGHT GRAY AND DARK BROWN<br>(SM)                                                                   |  |  |  |  |
| 650 -                            | 25                                                               |                           |                                                 |                             | (341)                                                                                                                                                                         |  |  |  |  |
|                                  |                                                                  |                           | 10 🕨 4                                          |                             | GRAVELLY SAND, POORLY GRADED, 8-12% SUBANGULAR TO 1.1 IN. MAX.,<br>COARSE TO FINE SAND, MOSTLY FINE, 3-8% NONPLASTIC FINES, LOOSE,<br>DARK BROWN                              |  |  |  |  |
|                                  | 30                                                               |                           | _                                               |                             | (SP)                                                                                                                                                                          |  |  |  |  |
|                                  |                                                                  |                           | 5 5                                             |                             | SILTY SAND, WIDELY GRADED, MEDIUM TO FINE, MOSTLY FINE, 8-12% NON-                                                                                                            |  |  |  |  |
| 640 -                            | 25                                                               | 4                         |                                                 |                             | (SM-SP) –                                                                                                                                                                     |  |  |  |  |
|                                  | 22                                                               |                           | 11 6                                            |                             | GRAVELLY SAND, FOORLY GRADED, 10-20% ROUNDED GRAVEL TO 1.1 IN. MAX.,                                                                                                          |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             | COARSE TO FINE SAND, MOSTLY FINE, 3-8% NONPLASTIC FINES, COMPACT, -<br>MEDIUM BROWN<br>(SP)                                                                                   |  |  |  |  |
|                                  | 40                                                               |                           |                                                 | ,                           |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  | 4                         | 18                                              |                             | IN. MAX.                                                                                                                                                                      |  |  |  |  |
| 630                              | 45                                                               | -                         |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  |                           | 19 6                                            |                             | GRAVELLY SAND, SAME AS S #7.<br>(SP)                                                                                                                                          |  |  |  |  |
|                                  | 50                                                               |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  |                           | 34 9                                            |                             | SILTY SAND, WIDELY GRADED, 3-8% ROUNDED GRAVEL TO 0.7 IN. MAX.,                                                                                                               |  |  |  |  |
| 620 —                            | K E                                                              |                           | <u>100</u>                                      |                             | DARK BROWN.<br>(SM)<br>SAND. POORLY GRADED. COARSE TO FINE. MOSTLY FINE 1-5% NONPLASTIC                                                                                       |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             | FINES, VERY DENSE, DARK BROWN.                                                                                                                                                |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             | TOP OF ROCK AT 55.4'                                                                                                                                                          |  |  |  |  |
|                                  |                                                                  | -                         |                                                 |                             | END OF BORING AT 55.4                                                                                                                                                         |  |  |  |  |
|                                  |                                                                  | ]                         |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  | -                         |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
|                                  |                                                                  |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |
| 1. FIG                           | URES (<br>L SAM                                                  | IN BLOW OR<br>PLE DENOTE  | RECOVERY COL<br>THE NUMBER C                    | UMN OPP<br>DF BLOWS         | OSITE<br>OF A                                                                                                                                                                 |  |  |  |  |
| A 21<br>FIG                      | URES :                                                           | SAMPLE SPO<br>SHOWN OPPO: | ING JO" REQUI<br>ON 12" OR THE<br>SITE ROCK COP | LED TO<br>DISTAN<br>ES DENO | DRIVE<br>CE SHOWN.<br>TE                                                                                                                                                      |  |  |  |  |
| THE<br>2. 21                     | PERCI<br>INDICA                                                  | ENT OF COR<br>TES LOCAT   | E RECOVERED.<br>ION OF UNDIST                   | URBED S.                    | AMPLE. BORING LOG 544 Z                                                                                                                                                       |  |  |  |  |
|                                  | INDIC/                                                           | TES LOCAT                 | ION OF SAMPLI                                   | ING ATTE                    | MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                                                |  |  |  |  |
| SUB3<br>NUMI<br>3. 포 1           | SCRIPJ<br>BER.<br>INDIC/                                         | NEXT TO S                 | SYMBOL INDICA                                   | TES SAM                     | PLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                                                                      |  |  |  |  |
| 4. <u>RO</u> D                   | ABLE.                                                            | K QUALITY                 | DESIGNATION.                                    |                             | STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                       |  |  |  |  |
| 6. DAT                           | 5. IL INDICATES DEPTH & LENGTH OF NX CORING RUN.                 |                           |                                                 |                             |                                                                                                                                                                               |  |  |  |  |

 $\cdot \cdot \cdot )$ 

| <u>, , , , , , , , , , , , , , , , , , , </u> | DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|-----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------|---------------------------------------------------------------|----------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE                                          |                                                                                                                                                                                                 | ALLEY POWER ST                           |                                                               | SHIT     | J.O. NO. 11700 BORING NO. 545 C                                                                                                                                                                                                  |  |  |  |  |
| DATE DE                                       | RILLED                                                                                                                                                                                          | APRIL 1, 19                              | 74                                                            | DRIL     | LED BY LOGGED BY                                                                                                                                                                                                                 |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          | · · · · · · · · · · · · · · · · · · ·                         | ,        |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               | ਸ਼ੑੑੑ                                                                                                                                                                                           | OVERALL<br>WEATHERING                    | SAMPLE                                                        | HC I     | SOIL OR ROCK DESCRIPTION                                                                                                                                                                                                         |  |  |  |  |
| ELE<br>FLE                                    | DEP1<br>FEE                                                                                                                                                                                     | AND<br>RQD<br>0 25 50 75 100             | BLOWS<br>BECOV.<br>TYPE                                       | G RAPI   | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                                           |  |  |  |  |
| 671.5                                         |                                                                                                                                                                                                 | ·                                        |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          |                                                               |          | NO SAMPLES FIRST 10'                                                                                                                                                                                                             |  |  |  |  |
|                                               | -                                                                                                                                                                                               |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
| 66 <u>0</u>                                   | 10                                                                                                                                                                                              |                                          | $\begin{array}{c c} & & & \\ & & & \\ 2 & & & \\ \end{array}$ |          | NO RECOVERY<br>TOP 8 INCHES: <u>CLAYEY ORGANIC SILT</u> , SLIGHTLY TO MODERATELY PLASTIC, -<br>8-126 FINE SAND, VERY SOFT, MOIST, BROWN.<br>LAST 10 INCHES: BAME AS ABOVE, EXCEPT BLACK.                                         |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          | 2 3                                                           |          | CLAYEY ORGANIC SILT, SIMILAR TO ABOVE, EXCEPT SATURATED.                                                                                                                                                                         |  |  |  |  |
| 6 <u>50</u>                                   | 20                                                                                                                                                                                              |                                          | 39 4                                                          |          | SAND, POORLY GRADED, FINE SUB-BOUNDED SAND, LESS THAN 5% COARSE<br>SAND, LESS THAN 5% SLIGHTLY PLASTIC FINES, DAMP, COMPACT, DRY<br>POCKET OF BLUE-GREEN FINE SAND, HLUE BROWN.<br>(SP)                                          |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          | 20 5                                                          |          | SANDY GRAVEL, POORLY GRADED, 60% GRAVEL TO 1.75 INCH DIAMETER,<br>15% COARSE SAND, 18-22% FINE SAND, 5% NONPLASTIC FINES, ELUE-BROWN,<br>DAMP, 1/2 INCH ON BOTTOM SATURATED GRAVELLY SAND, 40% GRAVEL, 60%<br>FINE SAND.<br>(GP) |  |  |  |  |
| 64 <u>0</u>                                   |                                                                                                                                                                                                 |                                          | 6 6                                                           | L        | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 40% GRAVEL TO 1.75<br>INCH DIAMETER, 30% COARSE SAND, 25% MEDIUM TO FINE SAND, LESS THAN<br>5% NONPLASTIC FINES, VERY LOOSE, DAMP, MEDIUM BROWN.<br>(SP)                           |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          | 15 7                                                          |          | SAND, WIDELY GRADED, COARSE TO FINE, EVENLY DISTRIBUTED, 10% NONPLAS-<br>TIC FINES, DAMP, MEDIUM BROWN.<br>(SP-SW)                                                                                                               |  |  |  |  |
| 63 <b>0</b>                                   | 40 —<br>—<br>—                                                                                                                                                                                  |                                          | 12 8                                                          | T        | <u>GRAVELLY SAND</u> , UNIFORM, FINE, SAND, 20% GRAVEL TO 1.5 INCH DIAMETER,<br>10% NONPLASTIC FINES, DAMP, LOOSE, BROWN.<br>(SP)                                                                                                |  |  |  |  |
|                                               | 45 —<br><br>                                                                                                                                                                                    |                                          | 24                                                            |          | GRAVELLY SAND, WIDELY GRADED, 15-25% GRAVEL TO 3/4 INCH DIAMETER,<br>SAND EVENLY DISTRIBUTED, 5-8% NONPLASTIC FINES, DAMP, COMPACT,<br>ELUE-BROWN.<br>(SP)                                                                       |  |  |  |  |
| 620                                           | 50 —<br><br>                                                                                                                                                                                    |                                          | 31 10                                                         |          | <u>GRAVELLY SAND</u> , SIMILAR TO ABOVE.                                                                                                                                                                                         |  |  |  |  |
|                                               | <u> </u>                                                                                                                                                                                        |                                          | 1 17                                                          |          | END OF BORING AT 55.0'                                                                                                                                                                                                           |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               | -                                                                                                                                                                                               |                                          |                                                               |          | · ·                                                                                                                                                                                                                              |  |  |  |  |
|                                               |                                                                                                                                                                                                 |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               | -                                                                                                                                                                                               |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
|                                               | _                                                                                                                                                                                               | ł                                        |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
| 1. FIG<br>SOII<br>140<br>A 2'<br>FIG          | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                                          |                                                               |          |                                                                                                                                                                                                                                  |  |  |  |  |
| THE<br>2. 21                                  | PERCEN<br>INDICAT                                                                                                                                                                               | T OF CORE RE<br>ES LOCATION              | OF UNDIST                                                     | URBED S  | BORING LOG 545 Z                                                                                                                                                                                                                 |  |  |  |  |
|                                               | INDICAT<br>VITH NO<br>SCRIPT<br>SER.                                                                                                                                                            | ES LOCATION<br>RECOVERY.<br>NEXT TO SYME | OF SAMPLI                                                     | ING ATTE | CMPT<br>CPLE<br>ID WATER OF THE DUQUESNE LIGHT COMPANY                                                                                                                                                                           |  |  |  |  |
| 4. ROD<br>5. []. I                            | ABLE.<br>- ROCK                                                                                                                                                                                 | QUALITY DES<br>ES DEPTH & L              | IGNATION.<br>ENGTH OF                                         | NX CORI  | ING RUN I III STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                                                            |  |  |  |  |
| LO. DATE                                      |                                                                                                                                                                                                 | MCAN SEA LEVEL                           |                                                               |          | Dra Ma                                                                                                                                                                                                                           |  |  |  |  |

 $\cdot \cdot )$ 



|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      | DEMP, LOOSE, BLUE-BROWN.<br>(SP) -                                                                                                                                                          |  |  |  |  |
|------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------|-----------------------------------------------------------------------------|----------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| 635                                      | 40                                                                                                                                                                                              |                                                              | 10 7                                                                        |                      | <u>GRAVELLY SAND</u> , WIDELY GRADED, COARSE TO FINE SAND, 25-35% GRAVEL<br>TO 1.25 INCH DIAMETER, ANGULAR, MOSTLY FINE SAND, 5% NONPLASTIC<br>FINES, SATURATED, LOOSE, BLUE-BROWN.<br>(SP) |  |  |  |  |
|                                          | 45                                                                                                                                                                                              |                                                              | 30 8                                                                        |                      | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, 10-20% COARSE TO<br>MEDIUM SAND, 10-20% ANUGLAR GRAVEL TO 1 INCH DIAMERER, 5-10%<br>NONPLASTIC FINES, DAMP, COMPACT, MEDIUM BROWN.<br>(SP)    |  |  |  |  |
| 625                                      | 50 —<br>                                                                                                                                                                                        |                                                              | 27                                                                          |                      | SAND, POORLY GRADED, MEDIUM TO FINE SAND, 5-10% ANGULAR GRAVEL<br>5% OF COARSE, ANGULAR SAND, DAMP, COMPACT, 5% NONPLASTIC FINES,<br>LICHT BROWN.<br>(SP)                                   |  |  |  |  |
| an a | 55 -                                                                                                                                                                                            | a tha an                 | ىرىنى مەركىيى بىرىكىيى بىرىكىيى بىرىكىيى بىرىكىيى بىرىكىيى بىرىكىيى بىرىكىي |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          | -                                                                                                                                                                                               |                                                              |                                                                             |                      | END OF BORING AT 55.0'                                                                                                                                                                      |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          | -                                                                                                                                                                                               |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
|                                          |                                                                                                                                                                                                 |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2'<br>FIGU   | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                                                              |                                                                             |                      |                                                                                                                                                                                             |  |  |  |  |
| THE<br>2. <b>2</b> 1                     | PERCEN<br>INDICAT                                                                                                                                                                               | T OF CORE REG                                                | OVERED.                                                                     | URBED S              | BORING LOG 546 C                                                                                                                                                                            |  |  |  |  |
| V61<br>DV1<br>SUBS                       | INDICAT<br>INDICAT<br>WITH NO<br>SCRIPT                                                                                                                                                         | ES LOCATION (<br>ES LOCATION (<br>RECOVERY.<br>NEXT TO SYMBO | OF SPLIT-                                                                   | SPOON SA<br>NG ATTER | MPLE.<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA                                                                                                             |  |  |  |  |
| NUME<br>3• ¥ I                           | BER.<br>Indicat                                                                                                                                                                                 | ES LOCATION C                                                | F NATURA                                                                    | L GROUNI             | DUQUESNE LIGHT COMPANY                                                                                                                                                                      |  |  |  |  |
| 4. ROD                                   | ABLE.                                                                                                                                                                                           | QUALITY DESI                                                 | GNATION.                                                                    |                      | MUNICA STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                              |  |  |  |  |
| <b>6.</b> DAT                            | JM IS                                                                                                                                                                                           | ES DEPTH & LE<br>MEAN SEA LEVEL                              | NGTH OF 1                                                                   | NX CORIN             | 11700 - GSK - 20                                                                                                                                                                            |  |  |  |  |

|                                     | <u> </u>                                                                                                                                                                                                                                                                                             |                                                 | <u>.</u>                        | DU              | QUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |  |
|-------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|---------------------------------|-----------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE<br>TYPE OF<br>DATE E<br>SUMMAR | SITE <u>BEAVER VALLEY POWER STATION</u> J.O. NO. <u>11700</u> BORING NO. <u>547</u><br>TYPE OF BORING <u>SPLIT SPOON</u> LOCATION <u>SHIPPINGPORT, PENNSYLVANIA</u> GROUND ELEV. <u>676'</u><br>DATE DRILLED <u>APRIL 3-4, 1974</u> DRILLED BY <u>AMERICAN</u> LOGGED BY J.P.D.<br>SUMMARY OF BORING |                                                 |                                 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| ELEV.<br>FEET                       | DEPTH<br>FEET<br>C<br>MO                                                                                                                                                                                                                                                                             | WERALL<br>ATHERING<br>AND<br>RQD<br>S SO TS 100 | BLOWS<br>BLOWS<br>RECOV<br>TYPE | G RAPHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>OR JOINTING BEDDING AND FAULTING<br>DESCRIPTIONS                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
| 676                                 |                                                                                                                                                                                                                                                                                                      |                                                 |                                 |                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| 665                                 |                                                                                                                                                                                                                                                                                                      |                                                 | 21 1                            |                 | GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 35-40% GRAVEL TO<br>1.5 INCH DAIMETER, SUB-ROUNDED AND ANGULAR, 15-20% COARSE SAND,<br>5-10% SLIGHTLY PLASTIC FINES, MOIST, COMPACT, LIGHT BROWN.<br>(SP)<br>SAND, POORLY GRADED, COARSE TO FINE, 15-20% COARSE SAND, ANGULAR;<br>VERY ANGULAR FINE SAND, 5-10% NONPLASTIC FINES, MOIST TO ALMOST<br>SATURATED, VERY LOOSE, BLUE-BROWN.<br>(SP)<br>GRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 25-35% GRAVEL TO<br>CRAVELLY SAND, WIDELY GRADED, COARSE TO FINE, 25-35% GRAVEL TO |  |  |  |
| 655                                 |                                                                                                                                                                                                                                                                                                      |                                                 | 34 3<br>35 4                    |                 | 1.6 INCH DIAMETER, SUB-ROUNDED TO ANGULAR, 15-20% COARSE SAND,<br>8-12% SLIGHTLY TO MODERATELY PLASTIC FINES, DAMP, DENSE, GREEN<br>EROWN, SAMPLE HAD 1 INCH LAYER OF DENSE, FINE, LIGHT BLUE SAND IN<br>MIDDLE OF SPOON.<br>(GP)<br><u>SANDY GRAVEL</u> , POORLY GRADED, WIDELY GRADED SAND, GRAVEL TO 1.5<br>INCH DIAMETER, ANGULAR AND SUB-ROUNDED, 35-40% COARSE TO FINE SAND<br>8-12% SLIGHTLY PLASTIC FINES, 15% COARSE SAND, MOIST TO SATURATED,<br>DENSE, HLUE-GREEN.<br>(GP)                                             |  |  |  |
| 645                                 | 30                                                                                                                                                                                                                                                                                                   |                                                 | 19 5                            |                 | GRAVELLY SAND, POORLY GRADED, MEDIUM TO FINE SAND, 30-40% GRAVEL<br>TO 1.4 INCH DIAMETER, ANGULAR AND SUB-ANGULAR, 15-20% MEDIUM SAND,<br>SUB-ROUNDED, 6-9% NONPLASTIC FINES, DAMP, COMPACT, MEDIUM BROWN.<br>(COBBLE IN PATH OF SPOON, SMALL RECOMERY).<br>(SP)                                                                                                                                                                                                                                                                  |  |  |  |



|                               |                              |                                             |                          | DUQI                 | UESNE LIGHT COMPANY SH_1 of 1                                                                                                    |
|-------------------------------|------------------------------|---------------------------------------------|--------------------------|----------------------|----------------------------------------------------------------------------------------------------------------------------------|
| SITE B                        | CAVER VAL                    | LLEY POWER STAT                             | TION                     | GUTDDT               | J.O. NO. 11700 BORING NO. 548 4                                                                                                  |
| TYPE OF<br>DATE D             | BORING                       | APRIL 4, 1974                               |                          | DRILI                | LED BY AMERICAN LOGGED BY JPD                                                                                                    |
| SUMMAR                        | Y OF B                       | ORING                                       |                          |                      |                                                                                                                                  |
|                               |                              | OVERALL                                     | SAMPLE                   | <u>ບ</u> [           | SOIL OR ROCK DESCRIPTION                                                                                                         |
| ELEV<br>FEET                  | EPT!                         | RQD                                         | OWS<br>BR<br>COV.<br>YPE | H APH<br>LOG         | FIELD AND LABORATORY TEST REBULTS; SOIL STRATA DESCRIPTION; LITHOLOGY                                                            |
|                               | 0 *                          | 0 25 50 75 100                              | BL<br>BL                 | 9<br>В               | OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                    |
| 675,3                         |                              |                                             |                          |                      |                                                                                                                                  |
|                               |                              |                                             | 3 1                      |                      | OF BLACK FINE MATERIAL, PROBABLY ORGANIC.                                                                                        |
| 670                           | 5                            |                                             |                          |                      | OPCANTO GANDY CTUD MONDE CONTO DO 250 ETNE IDITEODM COMP. HEDV                                                                   |
| 070                           |                              |                                             | 2 2                      |                      | SOFT, MOIST, OILY ORGANIC SMELL; DARK BLUE GRAY; STREAKS OF BLACK<br>ORGANIC MATERIAL THROUGHOUT.<br>(OL)                        |
|                               | 10                           |                                             | 13 🗾 3                   |                      | SANDY GRAVEL; POORLY GRADED, GRAVEL TO 1.5 DIAMETER, ANGULAR AND SUBROUNDED, 6-8% FINE SAND, 3% NONPLASTIC FINES, COMPACT, DAMP; |
|                               |                              |                                             |                          |                      | BLUE BROWN.<br>(GP)                                                                                                              |
| 660 -                         | 15 _                         |                                             | 66 <b>V</b> I            |                      | SANDY GRAVEL, POORLY GRADED; GRAVEL TO 1.75" DIAMETER, ANGULAR                                                                   |
| · · ·                         | -                            |                                             |                          |                      | 30-35% WIDELY GRADED SAND, COARSE TO FINE; 6-8% NONPLASTIC FINES,<br>VERY DENSE; MOIST, BLUE GRAY.<br>(GP)                       |
|                               | 20 _                         |                                             | 14 75                    |                      | SAND, MOSTLY UNIFORM, FINE, LESS THAN 1% NONPLASTIC FINES, DAMP,                                                                 |
|                               |                              |                                             |                          |                      | MEDIUM GRAYISH BROWN, FEW PEBBLES TO 1/2", (SP)                                                                                  |
| 650                           | 25 _                         |                                             |                          |                      | GRAVELLY SAND WELL GRADED COLOSE TO FINE 1-3% SLICHTEN PLASTIC                                                                   |
|                               | -                            | •                                           | 17 0                     |                      | FINES, MOIST, MEDIUM ORANGE BROWN, PEBBLES TO 1", (SN).                                                                          |
|                               | 30 _                         |                                             |                          |                      |                                                                                                                                  |
|                               | -                            |                                             | 13 7                     |                      | FINES, WET, MEDIUM GRAYISH BROWN, PEBLES TO 1 1/4", (SW).                                                                        |
| 610                           | 35 -                         | 1                                           |                          |                      |                                                                                                                                  |
| 040                           | -                            | 4                                           | 24 8                     |                      | MEDIUM GRAYISH BROWN, FEW PEBBLES TO 1/2", (SM).                                                                                 |
|                               | 40 -                         |                                             |                          |                      |                                                                                                                                  |
|                               |                              | -                                           | 49 9                     |                      | SAND, UNIFORM, MEDIUM TO FINE, LESS THAN 1% NONPLETIC FINES, WET,<br>MEDIUM BROWN, (SF).                                         |
| 630                           | 45 -                         |                                             | 100                      |                      | SAND, POORLY GRADED, COLRSE TO FINE, MOSTLY FINE, 3-5% JLIGHTLY                                                                  |
|                               | -                            |                                             | <u> </u>                 |                      | PEBBLES TO 1/2", (SP).<br>END OF BORING AT 56.0'                                                                                 |
|                               | -                            |                                             |                          |                      | -                                                                                                                                |
|                               |                              | -                                           |                          |                      | -<br>-                                                                                                                           |
|                               | -                            |                                             |                          |                      | -                                                                                                                                |
|                               | -                            |                                             |                          |                      |                                                                                                                                  |
|                               | -                            | 4                                           |                          |                      |                                                                                                                                  |
|                               |                              |                                             |                          |                      |                                                                                                                                  |
|                               |                              | 4                                           |                          |                      |                                                                                                                                  |
|                               |                              |                                             |                          |                      |                                                                                                                                  |
|                               |                              | -4                                          |                          |                      |                                                                                                                                  |
|                               | -                            |                                             |                          |                      |                                                                                                                                  |
| 1. FIC<br>SOT                 | URES I                       | N BLOW OR RE<br>LE DENOTE TH                | COVERY CO                | LUMN OPI             | POSITE<br>S OF A                                                                                                                 |
| 140<br>A 2<br>FT0             | D LB HA                      | MMER FALLING<br>AMPLE SPOON<br>HOWN OPPOSIT | 30" REQU<br>12" OR TH    | IRED TO<br>E DISTAN  | DRIVE<br>NCE SHOWN.                                                                                                              |
| THE<br>2. ■2                  | PERCE                        | NT OF CORE R<br>TES LOCATION                | ECOVERED.<br>OF UNDIS    | IURBED S             | BORING LOG 548 C                                                                                                                 |
|                               | VITH N                       | TES LOCATION<br>D RECOVERY.                 | OF SAMPL                 | ING ATTE             | BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT. PENNSYLVANTA                                                           |
| 508<br>NUM<br>3• <del>≩</del> | BER.<br>INDICA               | TES LOCATION                                | OF NATUR                 | ATES SAN<br>AL GROUN | DUQUESNE LIGHT COMPANY                                                                                                           |
| 4. ROD<br>5. []               | TABLE.<br>  - ROC]<br>INDICA | K QUALITY DE<br>IES DEPTH &                 | SIGNATION<br>LENGTH OF   | NX CORI              | ING RUN I ME I 11700 - GSK - 22                                                                                                  |
| 6. DAT                        | UM IS                        | MEAN SEA LEVEI                              | -<br>J                   |                      |                                                                                                                                  |

Ĵ

......

tion of the second s

|                              |                 |                                        |                                   | DUQUE                  | ESNE LIGHT COMPANY SH 1 OF 1                                                                                                           |
|------------------------------|-----------------|----------------------------------------|-----------------------------------|------------------------|----------------------------------------------------------------------------------------------------------------------------------------|
| SITE                         |                 | BEAVER VALL                            | ey power s                        | TATION                 | J.O. NO. 11700 BORING NO. 549 2                                                                                                        |
| TYPE OF<br>DATE DF           | BORING          | <u>MAY 2, 197</u>                      |                                   | IN DRILI               | LED BY AMERICAN LOGGED BY F.P.V.                                                                                                       |
| SUMMAR                       | Y OF BO         | ORING                                  |                                   |                        |                                                                                                                                        |
|                              |                 | OVERALL                                | SAMPLE                            | <u>ເບ</u>              | SOUL OF POCK DESCRIPTION                                                                                                               |
| LEV.<br>LET                  | EET             | WEATHERING                             | E K S                             | H B H                  | SOIL OR ROCK DESCRIPTION                                                                                                               |
|                              | ä u             | 0 25 50 75 100                         | BLO<br>REC                        | G R/                   | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |
| 646.9                        |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        | 14 1                              |                        | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC                                                                      |
|                              | -               |                                        |                                   |                        | (SW)                                                                                                                                   |
| (10                          | 5 -             |                                        | 11 2                              |                        | SAND, WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC FINES, -                                                                   |
| 640                          |                 |                                        |                                   |                        | MEDIUM GRAY, DEW PEBBLES TO 3/4".<br>(SW)                                                                                              |
|                              | 10 <b></b>      |                                        | 17                                | -                      | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% NON PLASTIC FINES.                                                                    |
|                              | -               |                                        |                                   |                        | MEDIUM GRAY BROWN, PEBBLES TO 1 1/4".<br>(SW)                                                                                          |
|                              | 15 _            |                                        |                                   | -                      |                                                                                                                                        |
| 630                          | -               |                                        | 25 🔰 4                            |                        | GRAVELLY SAND, AS ABOVE, MEDIUM BROWN.<br>(SW)                                                                                         |
|                              | 20 _            |                                        |                                   | -                      |                                                                                                                                        |
|                              | -               |                                        | 13 5                              |                        | SAND, WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC FINES,                                                                     |
|                              | 25              |                                        |                                   |                        | (SW)                                                                                                                                   |
| 620                          |                 |                                        | 14 6                              |                        | SAND, AS ABOVE, PEBBLES TO 3/4".                                                                                                       |
|                              | -               |                                        | <u>100</u>                        | -                      | SAND, WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC FINES,                                                                        |
|                              |                 | · · · · · · · · · · · · · · · · · · ·  | 5                                 |                        | END OF BORING @ 29.9' -                                                                                                                |
|                              | -               |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              | -               |                                        |                                   |                        |                                                                                                                                        |
|                              | _               |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              | -               |                                        |                                   |                        | -                                                                                                                                      |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              |                 | Í                                      |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
|                              | -               |                                        |                                   |                        |                                                                                                                                        |
|                              | -               |                                        |                                   |                        |                                                                                                                                        |
|                              | -               |                                        |                                   |                        |                                                                                                                                        |
|                              | -               | 1                                      |                                   |                        |                                                                                                                                        |
|                              |                 |                                        |                                   |                        |                                                                                                                                        |
| 1. FIGU                      | JRES IN         | BLOW OR REAL                           | COVERY C                          | OLUMN OPP              | POSITE<br>S OF A                                                                                                                       |
| 140<br>A 2"                  | LB HAM          | MER FALLING                            | 30" REQI                          | JIRED TO<br>HE DISTAN  | DRIVE<br>NCE SHOWN.                                                                                                                    |
| THE<br>2. 21                 | PERCEN          | T OF CORE RIVES LOCATION               | E RUCK CO<br>ECOVERED<br>OF UNDIS | JRES DENO<br>Sturbed s | SAMPLE.                                                                                                                                |
|                              | INDICAT         | ES LOCATION<br>ES LOCATION<br>RECOVERY | OF SPLIT                          | C-SPOON S<br>LING ATTE | EMPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                        |
|                              | SCRIPT<br>SER.  | NEXT TO SYM                            | BOL INDIC                         | CATES SAM              | MPLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                              |
| → ¥ I<br>T<br>4. <u>RQ</u> D | ABLE.<br>- ROCK | QUALITY DES                            | OF NATUR                          | KAL GROUN              | STONE & WEBSTER ENGINEERING CORPORATION                                                                                                |
| 5. ∏ I<br>6. DATU            | NDICAT<br>JM IS | ES DEPTH & I<br>MEAN SEA LEVE          | LENGTH OF                         | NX CORI                | ING RUN                                                                                                                                |

 $(\cdot)$ 

|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            |                                  | DUQUI                      | ESNE LIGHT COMPANY SH OF                                                                                       |  |  |
|------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------|----------------------------------|----------------------------|----------------------------------------------------------------------------------------------------------------|--|--|
| SITE                               |                                                                                                                                                                                                                                                                                                                                              | BEAVER VAL                                 | LEY POWE                         | R STATION                  | J.O. NO. 11700 BORING NO. 550 C                                                                                |  |  |
| TYPE OF                            | BORING                                                                                                                                                                                                                                                                                                                                       | S SPLIT SPOON<br>MAY 7, 19                 | LOCAT                            | ION <u>Shippi</u><br>Drili | INGPORT. PENNSYLVANTA GROUND ELEV. 650.6                                                                       |  |  |
| SUMMAR                             | Y OF B                                                                                                                                                                                                                                                                                                                                       |                                            |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            |                                  |                            |                                                                                                                |  |  |
|                                    | <b>-</b> '                                                                                                                                                                                                                                                                                                                                   | OVERALL                                    | SAMPL                            | ευ                         | SOU OR ROCK DESCRIPTION                                                                                        |  |  |
| ELEV.<br>FEET                      | DEPTH<br>FEET                                                                                                                                                                                                                                                                                                                                | WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | ALOWS<br>RECOV.<br>TYPF          | RAPHI                      | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY                                          |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            |                                  | 9                          | DESCRIPTIONS                                                                                                   |  |  |
| 650.6                              |                                                                                                                                                                                                                                                                                                                                              |                                            |                                  | ·                          |                                                                                                                |  |  |
| <b>65</b> 0 -                      |                                                                                                                                                                                                                                                                                                                                              |                                            | 11                               | 1                          | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 5-10%                                               |  |  |
|                                    | 5 -                                                                                                                                                                                                                                                                                                                                          |                                            |                                  |                            | (SP) —                                                                                                         |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            | 12                               | 2                          | <u>GRAVELLY SAND</u> , WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC<br>FINES. MEDIUM GRAY. PEBBLES TO 1" |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            |                                  |                            | (SW)                                                                                                           |  |  |
| 640 -                              | 10 —                                                                                                                                                                                                                                                                                                                                         |                                            |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            | 22                               |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            | 43 1                             | 2                          | (SW)                                                                                                           |  |  |
|                                    | 15 -                                                                                                                                                                                                                                                                                                                                         |                                            |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              |                                            | 24                               | 4                          | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, LESS THAN 1% NON                                             |  |  |
|                                    | 20 -                                                                                                                                                                                                                                                                                                                                         |                                            |                                  |                            | FLASTIC FINES, MEDIUM BROWN.<br>(SP)                                                                           |  |  |
| 630                                | - 20                                                                                                                                                                                                                                                                                                                                         |                                            |                                  |                            | _                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | ]                                          | 37                               | 5                          | SAND, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM                                              |  |  |
|                                    | 25 -                                                                                                                                                                                                                                                                                                                                         | 1                                          |                                  |                            | (SE)                                                                                                           |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          |                                  |                            | · –                                                                                                            |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          | 51                               | 6                          | SAND, AS ABOVE.                                                                                                |  |  |
| 620                                | 30 _                                                                                                                                                                                                                                                                                                                                         |                                            | <u>100</u><br><u>1"-</u>         | 7                          | NO RECOVERY (REFUSAL)                                                                                          |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | ]                                          |                                  | <b>'</b>                   | END OF BORING @ 30.31                                                                                          |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | ]                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | -                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | ł                                          |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 4                                          |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 4                                          |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            |                                            |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            |                                            |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            |                                            |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    |                                                                                                                                                                                                                                                                                                                                              | 1                                          |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | -                                          |                                  |                            | -                                                                                                              |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            |                                            |                                  |                            |                                                                                                                |  |  |
|                                    | -                                                                                                                                                                                                                                                                                                                                            | 1                                          |                                  |                            |                                                                                                                |  |  |
|                                    | [                                                                                                                                                                                                                                                                                                                                            |                                            |                                  |                            |                                                                                                                |  |  |
| 1. FIG                             | URES IN                                                                                                                                                                                                                                                                                                                                      | BLOW OR RE                                 | COVERY                           | COLUMN OPP                 | OSITE                                                                                                          |  |  |
|                                    | L SAMPI<br>LB HAM                                                                                                                                                                                                                                                                                                                            | LE DENOTE THE<br>(MER FALLING              | 5 NUMBEI<br>30" RE(              | H OF BLOWS<br>QUIRED TO    | SOF A<br>DRIVE                                                                                                 |  |  |
| A 2<br>FIG                         | URES SH                                                                                                                                                                                                                                                                                                                                      | IOWN OPPOSITI                              | LZ" OR 7<br>E ROCK (<br>ECOMPORT | THE DISTAN<br>CORES DENO   | TE                                                                                                             |  |  |
| 2. <b>2</b>                        | INDICAT                                                                                                                                                                                                                                                                                                                                      | ES LOCATION                                | OF UND                           | ISTURBED S                 | AMPLE. BORING LOG 558 C                                                                                        |  |  |
|                                    | INDICAT                                                                                                                                                                                                                                                                                                                                      | ES LOCATION                                | OF SAM                           | PLING ATTE                 | MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                 |  |  |
| SUB<br>NUM                         | SCRIPT<br>BER.                                                                                                                                                                                                                                                                                                                               | NEXT TO SYM                                | BOL IND                          | CATES SAM                  | PLE DUQUESNE LIGHT COMPANY                                                                                     |  |  |
| 3• ₽ 1                             | INDICAT<br>TABLE.                                                                                                                                                                                                                                                                                                                            | ES LOCATION                                | OF NATU                          | JRAL GROUN                 | D WATER 2                                                                                                      |  |  |
| 4. <u>RO</u> D<br>5. ∐ 1<br>6. DAT | 4. ROD - ROCK QUALITY DESIGNATION.<br>5. I INDICATES DEPTH & LENGTH OF NX CORING RUN<br>6. DATUM IS MEAN SEA LEVEL.<br>5. DATUM IS MEAN SEA LEVEL. |                                            |                                  |                            |                                                                                                                |  |  |

SH1\_ OF1 DUQUESNE LIGHT COMPANY 11700 BORING NO. 551 to SITE BEAVER VALLEY POWER STATION \_ J.O. NO. .... GROUND ELEV. 661.0 TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA AMERICAN DATE DRILLED \_\_\_\_\_ MAY 7, 1974 \_\_\_\_ ORILLED BY . LOGGED BY \_\_\_\_\_F.P.V. SUMMARY OF BORING \_ OVERALL SAMPLE  $\overline{\mathbf{O}}$ SOIL OR ROCK DESCRIPTION DEPTH FEET WEATHERING N U EI AND BLOWS RECOV. لعا Ц TYPI L L RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY 0 25 50 75 100 G 661.0 WOR 660 SILTY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 10-15% 1 MODERATELY PLASTIC FINES, SOME ORGANIC, DARK GRAY. (SM) 5 2 NO RECOVERY. 1 SILTY SAND, MOSTLY UNIFORM, FINE, 20-25% SLIGHTLY PLASTIC FINES, SOME ORGANIC, BLACK, OIL SMELL, 3 PEBBLES TO 1".(SM). SAND, MOSTLY UNIFORM, FINE, 3-5% SLIGHTLY PLASTIC FINES, SOME ORGANIC DARK GRAY TO ORANGE BROWN, FEW PEBBLES TO 1" (SP). 18 10 650 17 NO RECOVERY. 7 SAND, WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC FINES, MEDI-5 UM GRAY, FEW PEBBLES TO 3/8". (SW) 15 SANDY GRAVEL, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM BROWN, PEBBLES TO 1 1/4", 15-20% FINE SAND. 11 <u>6</u> (GW) 20 640 7 15 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM BROWN, PEBBLES TO 1 1/8". (SW) 25 20 8 GRAVELLY SAND, AS ABOVE. (SW) 30 12 ۳9 SAND, WELL GRADED, COARSE TO FINE, 3-5% NON PLASTIC FINES, MEDIUM 630 BROWN, FEW PEBBLES TO 1/2". (SW) 35 19 10 SAND, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM

i,

|                                        |                                                                                                                                                                                                 |                           | BROWN, FEW PEBBLES TO 3/4".                     |  |  |  |  |
|----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------|-------------------------------------------------|--|--|--|--|
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | 40 <b></b>                                                                                                                                                                                      | 100                       |                                                 |  |  |  |  |
| 620 —                                  | -                                                                                                                                                                                               | 3.5"                      | GRAY SHALE, WEATHERED                           |  |  |  |  |
|                                        |                                                                                                                                                                                                 | 11                        | END OF BORING @ 41 81                           |  |  |  |  |
|                                        | _                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        | 45 —                                                                                                                                                                                            |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           | -                                               |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           | -                                               |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           | -                                               |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           | _                                               |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           | -                                               |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | _                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        | ]                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                           |                                                 |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                           |                                                 |  |  |  |  |
| THE                                    | PERCENT OF COR                                                                                                                                                                                  | E RECOVERED.              |                                                 |  |  |  |  |
| 2. ∎2I<br>  /6I                        | NDICATES LOCAT                                                                                                                                                                                  | ION OF UNDISTURBE         | D SAMPLE. BURING LOG 551 T                      |  |  |  |  |
|                                        | NDICATES LOCAT                                                                                                                                                                                  | ION OF SAMPLING A         | BEAVER VALLEY POWER STATION - UNIT NO. 1        |  |  |  |  |
| SUBS                                   | CRIPT NEXT TO S                                                                                                                                                                                 | SYMBOL INDICAMPO          | SAMPLE                                          |  |  |  |  |
| NUMB                                   | ER.                                                                                                                                                                                             | JINDUL INDICATES          | DUQUESNE LIGHT COMPANY                          |  |  |  |  |
| ןֿ ¥ָ ינ                               | NDICATES LOCATI                                                                                                                                                                                 | ION OF NATURAL GR         | DUND WATEF 2                                    |  |  |  |  |
| <u>4. ROD</u>                          | - ROCK QUALITY                                                                                                                                                                                  | DESIGNATION.              | A WITTH STONE & WEBSTER ENGINEERING CORPORATION |  |  |  |  |
| 5.    I<br>6. DATU                     | NDICATES DEPTH<br>M IS MEAN SEA L                                                                                                                                                               | & LENGTH OF NX C<br>EVEL. | DRING RUN 11700 - GSK - 25                      |  |  |  |  |
|                                        |                                                                                                                                                                                                 | <i>€</i>                  |                                                 |  |  |  |  |

|                            | DUQUESNE LIGHT COMPANY SH_1 OF 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|----------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------|--------------------------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE                       | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 552                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                |                                      |                     |                                                                                                                                        |  |  |  |
| DATE D                     | RILLED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | MAY 8, 1                                       | 1974                                 |                     | LED BY LOGGED BY F.P.V.                                                                                                                |  |  |  |
| SUMMAR                     | TY OF B                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ORING                                          |                                      |                     | ······································                                                                                                 |  |  |  |
|                            | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | OVERALL                                        |                                      |                     |                                                                                                                                        |  |  |  |
| LEV.                       | PTH<br>EET                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | WEATHERING                                     | PE N                                 | DG<br>DG            | SOIL OR ROCK DESCRIPTION                                                                                                               |  |  |  |
|                            | DE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | G 25 50 75 100                                 | BLO<br>REC                           | GRA                 | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      | - <u></u>           |                                                                                                                                        |  |  |  |
| <u>651.3</u><br>650        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                | 4 1                                  |                     | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC -                                                                    |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     | FINES, SAME ORGANIC, MEDIUM TO DARK GRAY, SLIGHT OIL SMELL, PEBBLES -<br>TO 3/4".<br>(SW)                                              |  |  |  |
|                            | 5 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                |                                      |                     | SAND, UNIFORM, FINE, 1-3% SLIGHTLY PLASTIC FINES, MEDIUM GRAY.                                                                         |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                | 9 🖌 2                                |                     | (SP) CHANGING TO:<br>GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC<br>FINES, MEDIUM BROWN, PEBBLES TO 1 1/2".      |  |  |  |
|                            | 10 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                | <br>                                 |                     | (SW)                                                                                                                                   |  |  |  |
| 640 -                      | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                | 6 3                                  |                     | NO RECOVERY.                                                                                                                           |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                | 61/4                                 |                     | NO RECOVERY.                                                                                                                           |  |  |  |
|                            | - 17                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1                                              | 12 5                                 |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     | (SP)                                                                                                                                   |  |  |  |
| 630                        | 20 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                | 16 6                                 |                     | GRAVELLY SAND. AS ABOVE, LESS THAN 1% NON PLASTIC FINES.                                                                               |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     | (SP)                                                                                                                                   |  |  |  |
|                            | 25 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                |                                      |                     | -                                                                                                                                      |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                | 33 7                                 |                     | SAND, UNIFORM, FINE, LESS THAN 1% NON PLASTIC FINES, MEDIUM BROWN.                                                                     |  |  |  |
|                            | 30                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                | 100/0"                               |                     | NO RECOVERY (REFUSAL).                                                                                                                 |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 4                                              |                                      |                     | END OF BORING @ 29.5                                                                                                                   |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ]                                              |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     | -                                                                                                                                      |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     | ·<br>-                                                                                                                                 |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     | · · ·                                                                                                                                  |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | -                                              |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1                                              |                                      |                     | -                                                                                                                                      |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | مد 1                                           |                                      |                     |                                                                                                                                        |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     | -                                                                                                                                      |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1                                              |                                      |                     |                                                                                                                                        |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     |                                                                                                                                        |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1                                              |                                      |                     | -                                                                                                                                      |  |  |  |
|                            | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | ]                                              |                                      |                     |                                                                                                                                        |  |  |  |
|                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                |                                      |                     |                                                                                                                                        |  |  |  |
| 1. FIG                     | URES IN<br>L SAMPT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | BLOW OR REG                                    | COVERY COL                           | LUMN OPP            | OSITE<br>OF A                                                                                                                          |  |  |  |
| 140<br>A 2                 | LB HAN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | AMER FALLING                                   | 30" REQUI                            | RED TO<br>DISTAN    | DRIVE<br>ICE SHOWN.                                                                                                                    |  |  |  |
| F1G<br>THE<br>2. <b>■2</b> | PERCEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | IOWN OPPOSITI<br>IT OF CORE RI<br>SES LOCATION | S RUCK COP<br>SCOVERED.<br>OF UNDIST | es deno<br>Turbed s | AMPLE, BORING LOG 552 T                                                                                                                |  |  |  |
|                            | INDICAT<br>INDICAT<br>WITH NO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ES LOCATION<br>ES LOCATION<br>RECOVERY         | OF SPLIT-<br>OF SAMPLI               | SPOON S             | AMPLE. BEAVER VALLEY PORER STATION - UNIT NO. 1                                                                                        |  |  |  |
|                            | SCRIPT<br>BER.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | NEXT TO SYME                                   | BOL INDICA                           | TES SAM             | PLE DUQUESNE LIGHT COMPANY                                                                                                             |  |  |  |
| 3• ¥<br>4. <u>R</u> QD     | TABLE.<br>- ROCK                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | QUALITY DES                                    | OF NATURA                            | L GROUN             | D WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                                      |  |  |  |
| 5.∏<br>6. DAT              | 4. RQD - ROCK QUALITY DESIGNATION.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>6. DATUM IS MEAN SEA LEVEL.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN.<br>5. ☐ INDICATES DEPTH & LENGTH OF NX CORING RUN. |                                                |                                      |                     |                                                                                                                                        |  |  |  |

|                                     | DURIVESNE LIGHT COMPANY SH 1 OF 1                                |                                               |                                    |                              |                                                                                                                                       |  |  |
|-------------------------------------|------------------------------------------------------------------|-----------------------------------------------|------------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------|--|--|
| SITE B                              | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 553 C |                                               |                                    |                              |                                                                                                                                       |  |  |
| DATE D<br>SUMMAR                    | RILLED                                                           | MAY 8, 1                                      | 1974                               | DRIL                         | LED BYAMERICAN LOGGED BYF.P.V.                                                                                                        |  |  |
|                                     |                                                                  | · · · · · · · · · · · · · · · · · · ·         |                                    |                              |                                                                                                                                       |  |  |
| EV.<br>ET                           | РТН<br>ЕТ                                                        | OVERALL<br>WEATHERING<br>AND                  | SAMPLE<br>S Z W                    | PHIC<br>0G                   | SOIL OR ROCK DESCRIPTION                                                                                                              |  |  |
| EL<br>FE                            | DEI<br>FE                                                        | RQD<br>0 25 50 75 100                         | BLOW<br>RECC                       | G R A                        | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |
| 653.0                               |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     | 1                                                                | · · · · · · · · · · · · · · · · · · ·         | 8 1                                |                              | GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3%                                                                       |  |  |
| 650                                 | -<br>5_                                                          |                                               | 17 2                               |                              | (SP)<br>GRAVELLY SAND, AS ABOVE.                                                                                                      |  |  |
|                                     |                                                                  |                                               |                                    |                              | (SP)                                                                                                                                  |  |  |
|                                     |                                                                  |                                               | 19 3                               |                              | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC                                                                    |  |  |
| 610                                 | -                                                                |                                               |                                    |                              | (SW)                                                                                                                                  |  |  |
| 040                                 | -<br>-<br>15 -                                                   |                                               | 38 4                               |                              | SAND, WELL GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC FINES,                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              | (SW)                                                                                                                                  |  |  |
|                                     | 20 -                                                             |                                               | 123 5                              |                              | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% NON PLASTIC FINES,                                                                   |  |  |
| 630 -                               | -                                                                |                                               |                                    |                              | (SW)<br>NOTE: SPOON BENT WHILE DRIVING. MARKS ON SIDE INDICATE STRIKING<br>METALLIC OBJECT.                                           |  |  |
| 0,0                                 | 25                                                               |                                               | 37 6                               |                              | GRAVELLY SAND, AS ABOVE, PEBBLES TO 1".                                                                                               |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     | 30 _                                                             |                                               | 27 7                               |                              | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NON PLASTIC                                                                    |  |  |
| 620 -                               |                                                                  |                                               |                                    |                              | (SP)                                                                                                                                  |  |  |
|                                     | 35 -                                                             |                                               | 24 8                               |                              | SAND, UNIFORM, FINE, LESS THAN 1% NON PLASTIC FINES, MEDIUM BROWN.                                                                    |  |  |
| <u></u>                             |                                                                  |                                               | 100/19                             |                              |                                                                                                                                       |  |  |
|                                     | 40 _                                                             |                                               |                                    |                              | END OF BORING @ 37.1'                                                                                                                 |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     | -                                                                |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     | -                                                                |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
|                                     |                                                                  |                                               |                                    |                              |                                                                                                                                       |  |  |
| 1. FIGU<br>SOII                     | JRES IN<br>SAMPL                                                 | BLOW OR REC<br>E DENOTE THE                   | OVERY COL                          | JUMN OPP<br>)F BLOWS         | OSITE<br>OF A                                                                                                                         |  |  |
| 140<br>A 2'<br>Figu                 | LB HAM<br>' OD SA<br>JRES SH                                     | MER FALLING<br>MPLE SPOON 1<br>OWN OPPOSITE   | 30" REQUI                          | RED TO<br>DISTAN<br>RES DENO | DRIVE<br>CE SHOWN.<br>TE                                                                                                              |  |  |
| THE<br>2. ■21<br>▼61                | PERCEN<br>NDICAT<br>NDICAT                                       | T OF CORE RE<br>ES LOCATION<br>ES LOCATION    | COVERED.<br>OF UNDIST<br>OF SPLIT- | URBED S                      | AMPLE. BORING LOG 553 Z                                                                                                               |  |  |
|                                     | NDICAT                                                           | ES LOCATION<br>RECOVERY.<br>NEXT TO SYMB      | OF SAMPLI<br>OL INDICA             | NG ATTE<br>Tes sam           | MPT BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                  |  |  |
| NUME<br>3• ¥ I<br>T                 | SER.<br>NDICATI<br>ABLE.                                         | ES LOCATION                                   | OF NATURA                          | L GROUN                      | D WATER 2                                                                                                                             |  |  |
| 4. <u>ROD</u><br>5. ∐. I<br>6. DATU | - ROCK<br>NDICATI<br>M IS MI                                     | QUALITY DES<br>ES DEPTH & L<br>EAN SEA LEVEL. | IGNATION.<br>ENCTH OF              | NX CORII                     | NG RUN A 11700 - GSK - 27                                                                                                             |  |  |

SH\_1 OF\_1 DUQUESNE LIGHT COMPANY 5<u>54-C</u> BORING NO. \_ BEAVER VALLEY POWER STATION 11700 J.O. NO. ..... SITE \_\_ GROUND ELEV. 661.32 TYPE OF BORING SPLIT SPOON LOCATION \_\_\_\_\_SHTPPTNCPORT, PENNSYLVANIA F.P.V. \_ ORILLED BY \_\_\_\_AMERICAN DATE DRILLED \_\_\_\_\_MAY 10, 1974 LOGGED BY\_ SUMMARY OF BORING \_ OVERALL RAPHIC SAMPLE OR ROCK DESCRIPTION SOIL DEPTH FEET WEATHERING ELEV. E AND BLOWS RECOV TYPE Ē RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 G 661.32 <u>GRAVELLY SAND</u>, WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC FINES, MEDIUM GRAYISH BROWN, PEBBLES TO 3/4". 4 (SW) 5 17 🕨 2 GRAVELLY SAND, &S ABOVE, MEDIUM BROWN. 10 (SW) 650 45 🕨 3 SAND, WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC FINES, 15 MEDIUM GRAYISH BROWN, FEW PEBBLES TO 1/2". (SW) GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM BROWN, PEBBLES TO 1 1/4". 19 4 20 (SW)640 19 5 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC-25 FINES, MEDIUM GRAYISH BROWN, PEBBLES TO 3/4". (SW)39 6 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 1-3% SLIGHTLY PLASTIC 30 FINES, MEDIUM BROWN, PEBBLES TO 1/2". (SW) 630 15 7 SAND, UNIFORM, FINE, LESS THAN 1% NON PLASTIC FINES, MEDIUM BROWN. 35 (SP)

|                    |                                                                                                                                                                                          |                                                 | 100/1"                |          |                      |  |  |
|--------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|-----------------------|----------|----------------------|--|--|
|                    | 40 -                                                                                                                                                                                     |                                                 | 8                     |          | END OF BORING @ 37.5 |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    | 1                                                                                                                                                                                        |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
|                    |                                                                                                                                                                                          |                                                 |                       |          |                      |  |  |
| 1.                 | FIGURES IN                                                                                                                                                                               | BLOW OR RECO                                    | VERY COL              | UMN OPPO | OSITE                |  |  |
|                    | SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FLOURES SHOWN OPPOSITE POCK CORRESPONDE |                                                 |                       |          |                      |  |  |
| 2.                 | THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>F6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.                                                         |                                                 |                       |          |                      |  |  |
| ŝ                  | UPINDICATES LOCATION OF SAMPLING ATTEMPT<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE                                                                               |                                                 |                       |          |                      |  |  |
| 3                  | INDICATI                                                                                                                                                                                 | ES LOCATION C                                   | F NATURAL             | GROUNI   | WATER 2              |  |  |
| 4.<br>5. 1<br>6. 1 | ROD - ROCK<br>L INDICATE<br>DATUM IS ME                                                                                                                                                  | QUALITY DESI<br>ES DEPTH & LE<br>EAN SEA LEVEL. | GNATION.<br>NGTH OF N | IX CORIN | IG RUN               |  |  |

SH\_1 OF\_1 DUQUESNE LIGHT COMPANY BORING NO. 555 C BEAVER VALLEY POWER STATION 11700 J.O. NO. . SITE . TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA 675.3 GROUND ELEV. LOGGED BY \_\_\_\_\_F.P.V. MAY 8, 1974 AMERICAN DATE DRILLED \_\_ DRILLED BY SUMMARY OF BORING \_ RAPHIC LOG OVERALL SAMPLE OR ROCK DESCRIPTION SOIL DEPTH FEET WEATHERING FEET С С BLOWS RECOV TYPE Ш RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 G 675.3 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% NON PLASTIC FINES, 28 5 670 MEDIUM BROWN, DAMP, PEBBLES TO 1/2" (FILL). (SW) 2 GRAVELLY SAND, AS ABOVE, PEBBLES TO 1". (FILL). 12 10 (SW) ß 8 NO RECOVERY. 15 660 -GRAVELLY SAND, SAME AS SAMPLE #2, LAYER OF DARK GRAY SILT AT BOTTOM 6 OF RUN. (SW) 5 GRAVELLY SAND, POORLY GRADED, COARSE TO MEDIUM, TRACE OF FINES, GRAY, 52 20 PEBBLES TO 3/4". (SP) 37 6 GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC 25 FINES, MEDIUM GRAY, PEBBLES TO 1". 650 (SW) 28 7 GRAVELLY SAND, AS ABOVE, WITH SOME BLACK FINES. 30 (SW) 28 8 GRAVELLY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 10-13% 35

| 640                                    |                                                                                                                                                                                                 |                               | SLIGHTLY PLASTIC FINES, MEDIUM BROWN, PEBBLES TO 1".<br>(SP)                                                                |  |  |  |  |
|----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------|-----------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                                        | -<br>-<br>40<br>-                                                                                                                                                                               | 14 79                         | <u>GRAVELLY SAND</u> , WELL GRADED, COARSE TO FINE, LESS THAN 1% NON PLASTIC<br>FINES, MEDIUM BROWN, PEBBLES TO 1".<br>(SW) |  |  |  |  |
| 630                                    | 45                                                                                                                                                                                              | 31 10                         | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, LESS THAN 1% NON<br>PLASTIC FINES, MEDIUM BROWN.<br>(SP)                  |  |  |  |  |
|                                        |                                                                                                                                                                                                 | 42 11                         | SAND, MOSTLY UNIFORM, FINE, LESS THAN 1% NON PLASTIC FINES, MEDIUM<br>BROWN, FEW PEBBLES TO 1/2".<br>(SP)                   |  |  |  |  |
| 620                                    | 55                                                                                                                                                                                              | 38 12                         | SAND, AS ABOVE, TRACE OF BLACK FINES.<br>(SP)                                                                               |  |  |  |  |
|                                        | -                                                                                                                                                                                               | 100/0"                        | END OF BORING @ 57.5'                                                                                                       |  |  |  |  |
|                                        | ₩ <b>-</b>                                                                                                                                                                                      |                               |                                                                                                                             |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                               |                                                                                                                             |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                               |                                                                                                                             |  |  |  |  |
|                                        |                                                                                                                                                                                                 |                               |                                                                                                                             |  |  |  |  |
|                                        | -                                                                                                                                                                                               |                               | -                                                                                                                           |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN. |                               |                                                                                                                             |  |  |  |  |
|                                        | PERCENT OF CORE RI<br>NDICATES LOCATION                                                                                                                                                         | COVERED.<br>OF UNDISTURBED S  | BORING LOG 555 Z                                                                                                            |  |  |  |  |
|                                        | FOR INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>IF INDICATES LOCATION OF SAMPLING ATTEMPT<br>WITH NO RECOVERY.<br>BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA       |                               |                                                                                                                             |  |  |  |  |
| NUME<br>3• ¥ I                         | NUMBER.<br>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATER 2                                                                                                                                    |                               |                                                                                                                             |  |  |  |  |
| 4. <u>RO</u> D<br>5. ∐. I              | - ROCK QUALITY DES<br>NDICATES DEPTH & L                                                                                                                                                        | IGNATION.<br>ENGTH OF NX CORI | NG RUN I /// STONE & WEBSTER ENGINEERING CORPORATION                                                                        |  |  |  |  |
| 6. DATU                                | DATUM IS MEAN SEA LEVEL.                                                                                                                                                                        |                               |                                                                                                                             |  |  |  |  |



| 040                   | -                                                                                                                                            |                                        | FINES, MEDIUM BROWN, PEBBLES TO 3/4".<br>(SW)                                                                         |  |  |  |  |  |
|-----------------------|----------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------|-----------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
|                       |                                                                                                                                              | 35 8                                   | -<br>                                                                                                                 |  |  |  |  |  |
|                       |                                                                                                                                              |                                        | MEDIÚM BROWN.                                                                                                         |  |  |  |  |  |
| 630                   | -<br>45<br>-                                                                                                                                 | 74 9                                   | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NON PLASTIC<br>FINES, MEDIUM BROWN, YEW PEBBLES TO 1".<br>(SP) |  |  |  |  |  |
|                       | <br><br>50<br>                                                                                                                               | 93 10                                  | SAND AS ABOVE.<br>(SP)                                                                                                |  |  |  |  |  |
| 620                   | 55                                                                                                                                           | 82= 11                                 | SAND, AS ABOVE, FEW PEBBLES TO 1/2".                                                                                  |  |  |  |  |  |
|                       | _                                                                                                                                            | 100/0"                                 | END OF BORING @ 56.01                                                                                                 |  |  |  |  |  |
|                       | -                                                                                                                                            |                                        |                                                                                                                       |  |  |  |  |  |
|                       | <u> </u>                                                                                                                                     |                                        |                                                                                                                       |  |  |  |  |  |
|                       | -                                                                                                                                            |                                        |                                                                                                                       |  |  |  |  |  |
|                       |                                                                                                                                              |                                        |                                                                                                                       |  |  |  |  |  |
|                       |                                                                                                                                              |                                        |                                                                                                                       |  |  |  |  |  |
|                       | -                                                                                                                                            |                                        |                                                                                                                       |  |  |  |  |  |
|                       | _                                                                                                                                            |                                        |                                                                                                                       |  |  |  |  |  |
| 1. FIG<br>SOII<br>140 | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE |                                        |                                                                                                                       |  |  |  |  |  |
| FIGU<br>THR           | JRES SHOWN OPPOSITE<br>PERCENT OF CORF PE                                                                                                    | COVERED                                |                                                                                                                       |  |  |  |  |  |
| 2. <b>2</b> 1<br>761  | INDICATES LOCATION                                                                                                                           | OF UNDISTURBED S.<br>OF SPLIT-SPOON S. | AMPLE. BORING LOG 556 C                                                                                               |  |  |  |  |  |
|                       | INDICATES LOCATION                                                                                                                           | OF SAMPLING ATTE                       | MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                        |  |  |  |  |  |
| NUME<br>3. 목 I        | BER.<br>NDICATES LOCATION                                                                                                                    | OF NATURAL GROUND                      | DUQUESNE LIGHT COMPANY                                                                                                |  |  |  |  |  |
| <u>4. <u>R</u>QD_</u> | ABLE.<br>- ROCK QUALITY DES                                                                                                                  | IGNATION.                              | STONE & WEBSTER ENGINEERING CORPORATION                                                                               |  |  |  |  |  |
| 5. 11. 1<br>6. DATU   | DATUM IS MEAN SEA LEVEL.                                                                                                                     |                                        |                                                                                                                       |  |  |  |  |  |

|                           | DUQUESNE LIGHT COMPANY SH1_OF_1_ |                |                                                       |                                            |                  |                                                                                                                                                           |  |  |
|---------------------------|----------------------------------|----------------|-------------------------------------------------------|--------------------------------------------|------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| SITE<br>TYP<br>DAT<br>SUM | SITE                             |                |                                                       |                                            |                  |                                                                                                                                                           |  |  |
| ELEV.                     | FEET                             | DEPTH<br>FEET  | OVERALL<br>WEATHERING<br>AND<br>RQD<br>0 25 50 75 100 | SAMPLE<br>BLOWS<br>BLOWS<br>BLOWS<br>BLOWS | G R APHIC<br>LOG | SOIL OR ROCK DESCRIPTION<br>FIELD AND LABORATORY TEST RESULTS;<br>OR JOINTING, BEDDING AND FAULTING<br>DESCRIPTIONS                                       |  |  |
| 676                       | 676.1                            |                |                                                       |                                            |                  |                                                                                                                                                           |  |  |
| 670                       |                                  | 5 _            |                                                       | 4 1                                        |                  | <u>GRAVELLY SAND</u> , WELL GRADED, COARSE TO FINE, 3-5% NON PLASTIC FINES,<br>DAMP, MEDIUM TO DARK BROWN, PEBBLES TO 1 1/8". (FILL)<br>(SW)              |  |  |
|                           |                                  | 10 -           |                                                       | 25 2                                       |                  | <u>GRAVELLY SAND</u> , AS ABOVE, PEBBLES TO 1/2".<br>(SW)                                                                                                 |  |  |
| 660                       |                                  | -<br><br>15    |                                                       | 11 3                                       |                  | GRAVELLY SAND, SAME AS ABOVE, PEBBLES TO 1", WET.                                                                                                         |  |  |
|                           |                                  |                |                                                       | 22 🔽 4                                     |                  | <u>GRAVELLY SAND</u> , WELL GRADED, COARSE TO FINE, 5-10% MODERATELY PLASTIC<br>FINES, MEDIUM GRAYISH BROWN WITH SOME BLACK FINES, PEBBLES TO 1".<br>(SW) |  |  |
| 650                       | <b>)</b>                         | 25             |                                                       | 30 5                                       |                  | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC<br>FINES, MEDIUM BROWN, CONTAINS BROKEN GRAY SANDSTONE FRAGMENTS TO<br>1 1/4".<br>(SW) |  |  |
|                           |                                  | 30 -           |                                                       | 15 6                                       |                  | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 5-10% MODERATELY PLASTIC<br>FINES, MEDIUM BROWN WITH SOME GRAY, PEBBLES TO 1".<br>(SW)                        |  |  |
| 640                       | )                                | -<br>35 —<br>- |                                                       | 50 7                                       |                  | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NON PLASTIC<br>FINES, MEDIUM BROWN, FEW PEBBLES TO 1".<br>(SP)                                     |  |  |

|                                                    | _                                                                                                                                                                                                                                           |       |                                                                                                                                              |  |  |  |  |
|----------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------|----------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                                                    | 40                                                                                                                                                                                                                                          | 34 8  | SAND, WELL GRADED, COARSE TO FINE, 1-3% NON PLASTIC FINES, MEDIUM<br>GRAYISH BROWN, FEW PEBBLES TO 1/2".<br>(SP)                             |  |  |  |  |
| 630                                                | -<br>-<br>45<br>-                                                                                                                                                                                                                           | 28    | SAND, SAME AS SAMPLE 7.                                                                                                                      |  |  |  |  |
|                                                    | -<br>50<br>-                                                                                                                                                                                                                                | 26 10 | SAND, POORLY GRADED, OOARSE TO FINE, MOSTLY FINE, 3-5% SLIGHTLY<br>PLASTIC FINES, MEDIUM BROWN WITH SOME BLACK, FEW PEBBLES TO 1/2".<br>(SP) |  |  |  |  |
|                                                    | -<br>-<br>55 -                                                                                                                                                                                                                              | 29    | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NON PLASTIC                                                                           |  |  |  |  |
| 620                                                | _                                                                                                                                                                                                                                           | BIT   | TOP OF ROCK @ 56.0.<br>GRAY SHALE CUTTINGS.                                                                                                  |  |  |  |  |
|                                                    | <b>6</b><br><b>6</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b><br><b>1</b>                                                                                                |       | END OF BORING @ 58.0                                                                                                                         |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU             | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FIGURES SHOWN OPPOSITE BOCK CORES DENOTE |       |                                                                                                                                              |  |  |  |  |
|                                                    | THE PERCENT OF CORE RECOVERED.<br>2. 2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>76 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>NUMBER<br>NUMBER                      |       |                                                                                                                                              |  |  |  |  |
| 3. ¥ I<br><sup>4</sup> . RQD<br>5. ∐. I<br>6. DATU | NUMBER.<br>3. ¥ INDICATES LOCATION OF NATURAL GROUND WATEF 2<br>TABLE.<br>4. ROD - ROCK QUALITY DESIGNATION.<br>5. □. INDICATES DEPTH & LENGTH OF NX CORING RUN.                                                                            |       |                                                                                                                                              |  |  |  |  |

|                               |                   |                                           |                          | DUG                  | QUESNE LIGHT COMPANY SH_1 OF_1                                                                                                        |
|-------------------------------|-------------------|-------------------------------------------|--------------------------|----------------------|---------------------------------------------------------------------------------------------------------------------------------------|
| SITE                          | BE                | AVER VALLEY PO                            | WER STATION              | <u> </u>             | J.O. NO. 11700 BORING NO. 5587                                                                                                        |
| TYPE OF<br>DATE D             | BORING            | S SPLIT SPOON<br>MAY 9, 1                 | LOCATION<br>.974         | N SHII               | PPINGPORT, PENNSYLVANIA GROUND ELEV. 662.1   LED BY AMERICAN   LOGGED BY F.P.V.                                                       |
| SUMMAR                        | Y OF B            | ORING                                     |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
| ET .                          | ETH               | OVERALL<br>WEATHERING<br>AND              | SAMPLE<br>So > W         | рніс                 | SOIL OR ROCK DESCRIPTION                                                                                                              |
| EL<br>FE                      | DEF               | RQD<br>0 25 50 75 100                     | BLOW<br>0.8<br>AECO      | G R AI<br>O          | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |
| 662.1                         |                   |                                           |                          |                      |                                                                                                                                       |
| 660                           |                   |                                           | PUSHEI<br>BY             |                      |                                                                                                                                       |
| ••••                          |                   |                                           | HAND YI                  |                      | ERATELY PLASTIC FINES, MEDIUM TO DARK BROWN, FEW PEBBLES TO 3/4".<br>(SM)                                                             |
|                               | 5 -               |                                           |                          |                      | -                                                                                                                                     |
| :                             | -                 |                                           |                          |                      |                                                                                                                                       |
|                               | 10                |                                           | 2 🗾 2                    |                      | SILTY SAND, AS ABOVE, FEW PEBBLES TO 1 1/8".                                                                                          |
| 650                           |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           | 11 3                     |                      | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 10-15% MODERATELY PLASTIC<br>FINES, MEDIUM GRAY WITH TRACE OF BLACK, PEBBLES TO 1 1/4".   |
| -<br>-<br>-                   |                   |                                           |                          |                      | (SW)                                                                                                                                  |
|                               |                   |                                           | 13                       |                      | SAND, UNIFORM, FINE, 1-3% SLIGHTLY PLASTIC FINES, MEDIUM BROWN.                                                                       |
|                               | 20 _              |                                           |                          |                      | (SP)                                                                                                                                  |
| 640                           |                   |                                           |                          |                      | -                                                                                                                                     |
|                               | 25 —              |                                           | 14 75                    |                      | GRAVELLY SAND, WELL GRADED, COARSE TO FINE, 3-5% SLIGHTLY PLASTIC<br>FINES, MEDIUM BROWN, PEBBLES TO 1".                              |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               | 30 -              |                                           | 18 6                     |                      | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 1-3% NON PLASTIC                                                                    |
| 630                           |                   |                                           |                          |                      | (SP)                                                                                                                                  |
|                               |                   |                                           | 16 7                     |                      | SAND. WELL GRADED. COARSE TO FINE. 1-3% NON PLASTIC FINES. MEDIUM                                                                     |
|                               | 35                |                                           |                          |                      | BROWN.<br>(SW)                                                                                                                        |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               | 40                |                                           | 11 8                     |                      | SAND, AS ABOVE.                                                                                                                       |
| 620                           |                   |                                           | T00\0.                   |                      | END OF BORING @ 41.7'                                                                                                                 |
|                               | 45 —              |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      | -                                                                                                                                     |
|                               | -                 |                                           |                          |                      | -                                                                                                                                     |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      | -                                                                                                                                     |
|                               |                   |                                           |                          |                      | -                                                                                                                                     |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               |                   |                                           |                          |                      |                                                                                                                                       |
|                               | -                 |                                           |                          |                      |                                                                                                                                       |
| 1. FIGU                       | JRES IN           | BLOW OR REG                               | COVERY COI               | JUMN OPP             | POSITE                                                                                                                                |
| SOII<br>140<br>4 25           | LB HAM            | E DENOTE THI<br>MER FALLING<br>MPLE SPOON | E NUMBER O<br>30" REQUI  | F BLOWS<br>RED TO    | OF A<br>DRIVE                                                                                                                         |
| FIGU                          | TRES SH<br>PERCEN | OWN OPPOSITI                              | E ROCK COR<br>E COVERED. | E DISTAN<br>RES DENO | TE TE                                                                                                                                 |
| 2. <b>■2</b> 1<br><b>▼6</b> 1 | NDICAT<br>NDICAT  | ES LOCATION<br>ES LOCATION                | OF UNDIST<br>OF SPLIT-   | URBED S<br>SPOON S   | AMPLE. BORING LOG 558 C                                                                                                               |
|                               | ITH NO            | RECOVERY.<br>NEXT TO SYME                 | or Sampli<br>BOL INDICA  | NG ATTE              | PLE BEAVER VALLEY POWER STATION - UNIT NO. 1<br>SHIPPINGPORT, PENNSYLVANIA                                                            |
| NUME<br>3. ¥ I<br>T           | NDICATI<br>ABLE.  | ES LOCATION                               | OF NATURA                | L GROUN              | DUQUESNE LIGHT COMPANY                                                                                                                |
| 4. <u>R</u> 0D<br>5. ∐ I      | - ROCK<br>NDICATI | QUALITY DES<br>ES DEPTH & L               | IGNATION.<br>ENGTH OF    | NX CORI              | NG RUN                                                                                                                                |
| U. DATU                       | n 15 P            | MAR ODA LEVEL                             | •                        |                      | 11700 - UDA - 32                                                                                                                      |



|                                               | 40 <b>–</b>                                                                                                                                                                                                                                                                                                                                                                                                       | 41                                                                                             | GRAVELLY SAND; AS ABOVE.<br>(SW)                                                                 |                                        |  |  |  |  |
|-----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|----------------------------------------|--|--|--|--|
| 630                                           | 45 -                                                                                                                                                                                                                                                                                                                                                                                                              | 29                                                                                             | SAND: POORLY GRADED, COAN<br>FINES, MEDIUM GRAYISH BRO<br>(SP)                                   |                                        |  |  |  |  |
| 620                                           | -<br>50 <b>-</b><br>-                                                                                                                                                                                                                                                                                                                                                                                             | <b>39</b><br>100/0" 10                                                                         | <u>GRAVELLY SAND</u> : POOPLY GRA<br>NONPLASTIC FINES, MEDIUM<br>(SP)<br>NO RECOVERY. (REFUSAL). | DED, COARSE TO FINE, MOSTLY FINE, 1-3% |  |  |  |  |
|                                               | 55 -                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                | END OF BORING AT 53.5                                                                            |                                        |  |  |  |  |
|                                               | 60<br>60                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                |                                                                                                  |                                        |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU        | JRES IN BLO<br>L SAMPLE DE<br>LB HAMMER<br>OD SAMPLE<br>IBES SHOWN                                                                                                                                                                                                                                                                                                                                                | W OR RECOVERY CO<br>NOTE THE NUMBER<br>FALLING 30" REQU<br>SPOON 12" OR TH<br>OPPOSITE BOCK CO | N OPPOSITE<br>BLOWS OF A<br>D TO DRIVE<br>ISTANCE SHOWN.                                         | -                                      |  |  |  |  |
|                                               | FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br>THE PERCENT OF CORE RECOVERED.<br>2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>6 INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>17 INDICATES LOCATION OF SAMPLING ATTEMPT<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>NUMBER.<br>17 INDICATES LOCATION OF SAMPLING ATTEMPT<br>WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE |                                                                                                |                                                                                                  |                                        |  |  |  |  |
| 3. ¥ I<br>4. <u>R</u> QD<br>5. ∐ I<br>6. DATU | NDICATES L<br>ABLE.<br>- ROCK QUA<br>NDICATES D<br>IM IS MEAN S                                                                                                                                                                                                                                                                                                                                                   | OCATION OF NATUR<br>LITY DESIGNATION<br>EPTH & LENGTH OF<br>EA LEVEL                           | CORING RUN                                                                                       | & WEBSTER ENGINEERING CORPORATION      |  |  |  |  |

|                      | DUQUESNE LIGHT COMPANY SH 1 OF 1                                                                                                                      |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|----------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------|------------------------------------------------|------------------|--------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| SITE                 | SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 560 ±<br>TYPE OF BORING SPLIT SPOON LOCATION SHIPPINGPORT, PENNSYLVANIA GROUND ELEV. 673.9 |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
| DATE                 | DRIL                                                                                                                                                  | LED                  | MAY 10, 19                                     | 974              |                          |                              | LED BY LOGGED BY                                                                                                                       |  |  |  |  |
| SUMMA                |                                                                                                                                                       | OF B                 | ORING                                          | <u> </u>         |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       |                      | OVERAL                                         | <b>PA</b> 84     |                          | ()                           |                                                                                                                                        |  |  |  |  |
| E K                  | D T H                                                                                                                                                 | ΕT                   | WEATHERING                                     | 5 X              | ш<br>Ш                   | РНК                          | SOIL OR ROCK DESCRIPTION                                                                                                               |  |  |  |  |
| 3                    | l u                                                                                                                                                   | ц<br>Ц               | RQD<br>0 25 50 75 100                          | BLOW<br>RECC     | ТҮР                      | 3 R AI<br>LC                 | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          | <u> </u>                     |                                                                                                                                        |  |  |  |  |
| <u>673.9</u>         |                                                                                                                                                       |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       | _                    |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
| 670                  |                                                                                                                                                       |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                | 6                |                          |                              | -<br>                                                                                                                                  |  |  |  |  |
|                      | 10                                                                                                                                                    | )                    |                                                |                  |                          |                              | WET, MEDIUM BROWN, PEBBLES TO 1/2".<br>(SW)                                                                                            |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
| 660                  | -  1                                                                                                                                                  | , _                  | -<br>-                                         | W.O.H.           | 2                        |                              | SILTY SAND, : UNIFORM, FINE TO VERY FINE, 20-25% MODERATELY PLASTIC -<br>FINES, WET, DARK GRAY TO BLACK.                               |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              | <u>(</u> SM)                                                                                                                           |  |  |  |  |
|                      |                                                                                                                                                       | _                    |                                                | 25               |                          |                              | GRAVELLY SAND: WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC -                                                                   |  |  |  |  |
|                      |                                                                                                                                                       | )                    |                                                |                  |                          |                              | FINES, WET, MEDIUM BROWN WITH SOME GRAY AND BLACK, PEBBLES TO 1" (SW)                                                                  |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
| 650 -                | 2;                                                                                                                                                    |                      |                                                | 15               | 4                        |                              | GRAVELLY SAND: WELL GRADED, COARSE TO FINE, 3-5% NONPLASTIC FINES, -<br>MEDIUM BROWN, PEBBLES TO 3/4".                                 |  |  |  |  |
|                      |                                                                                                                                                       | -                    |                                                |                  |                          |                              | - (SW)<br>-                                                                                                                            |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                | 5.               | 5                        |                              | SAND: MOSTLY UNIFORM, FINE, 1-3% NONPLASTIC FINES, MEDIUM GRAYISH<br>BROWN WITH TRACE OF BLACK, FEW PEBBLES TO 1".                     |  |  |  |  |
|                      | 3                                                                                                                                                     | -                    |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       | -                    |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
| 640 -                | 35                                                                                                                                                    |                      |                                                | 7                | 6                        |                              | GRAVELLY SAND WELL GRADED, COARSE TO FINE, LESS THAN 1% NONPLASTIC<br>FINES, MEDIUM GRAY, FEW PEBBLES TO 1".                           |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                | 14               | 7                        |                              | SAND: UNIFORM. MEDIUM TO FINE, LESS THAN 1% NONPLASTIC FINES,                                                                          |  |  |  |  |
|                      |                                                                                                                                                       | -                    |                                                |                  |                          |                              | (SP)                                                                                                                                   |  |  |  |  |
| 630 -                |                                                                                                                                                       | -                    |                                                | 12               |                          |                              | -                                                                                                                                      |  |  |  |  |
| 0,0                  | 45                                                                                                                                                    |                      |                                                | 1.5              | 8                        |                              | SAND: AS ABOVE.                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       | -                    |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      | 50                                                                                                                                                    | _                    |                                                | 13               | $\overline{\mathcal{A}}$ |                              | NO RECOVERY.<br>SAND: WELL GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES, MEDIUM                                                       |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              | BROWN, FEW PEBBLES TO 1/2".<br>(SW)                                                                                                    |  |  |  |  |
| 620 —                | -                                                                                                                                                     |                      |                                                | 100/2            | -191                     |                              | TRACE OF <u>GRAY SHALE</u> AT BUTTOM OF RUN                                                                                            |  |  |  |  |
|                      | 55                                                                                                                                                    |                      | 4                                              |                  |                          |                              | END OF BORING AT 54.2                                                                                                                  |  |  |  |  |
|                      |                                                                                                                                                       | -                    | 1                                              |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      | I                                                                                                                                                     | _                    |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      |                                                                                                                                                       | -                    |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      |                                                                                                                                                       | -                    | ]                                              |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      |                                                                                                                                                       | _                    |                                                |                  |                          |                              | -                                                                                                                                      |  |  |  |  |
|                      |                                                                                                                                                       |                      |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
|                      |                                                                                                                                                       | _                    |                                                |                  |                          |                              |                                                                                                                                        |  |  |  |  |
| <b>, .</b>           |                                                                                                                                                       |                      |                                                |                  |                          | L                            |                                                                                                                                        |  |  |  |  |
| 1. FIC<br>SOJ<br>114 | GURE<br>IL S<br>) LP                                                                                                                                  | S IN<br>Ampi<br>Ham  | I BLOW OR REC<br>JE DENOTE THE<br>IMER FALLING | OVERY            | COL<br>ER O<br>EOUT      | UMN OPP<br>F BLOWS<br>BFD TO | OSITE<br>OF A<br>DRIVE                                                                                                                 |  |  |  |  |
| A 2<br>FIC           | 2" O<br>GURE                                                                                                                                          | D SA<br>S SH         | MPLE SPOON 1<br>IOWN OPPOSITE                  | 2" OR<br>ROCK    | THE<br>COR               | DISTAN<br>ES DENO            | CE SHOWN.<br>TE T                                                                                                                      |  |  |  |  |
| THE<br>2. <b>■2</b>  | E PE<br>IND                                                                                                                                           | RCEN<br>ICAT         | T OF CORE RE<br>ES LOCATION                    | OF UN            | ED.<br>DIST              | URBED S                      | AMPLE. BORING LOG 560 t                                                                                                                |  |  |  |  |
|                      | VIND<br>WIT                                                                                                                                           | ICAT<br>ICAT<br>H NO | ES LOCATION<br>RECOVERY.                       | OF SPI<br>OF SAI | MPLI                     | SPOON S<br>NG ATTE           | MPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                         |  |  |  |  |
|                      | BSCR                                                                                                                                                  | IPT                  | NEXT TO SYMB                                   | OL IN            | DICA                     | TES SAM                      | PLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                                                               |  |  |  |  |
| 3• ¥<br>  4. Ron     | IND<br>TAB                                                                                                                                            | LCAT<br>LE.<br>Rock  | ES LUCATION<br>QUALITY DES                     | OF NAT           | l'URA:                   | L GROUN                      | D WATER 2 LOB STONE & WEBSTER ENGINEERING CORPORATION                                                                                  |  |  |  |  |
| 5. ∏<br>6. DAT       | IND<br>MUT                                                                                                                                            | ICAT                 | ES DEPTH & L<br>(EAN SEA LEVEL                 | ENGTH            | OF                       | NX CORII                     | NG RUN 644 11700 - GSK - 134                                                                                                           |  |  |  |  |

•

|                       |                          |                                 |                                                                                                  | DU               | QUESNE LIGHT COMPANY SH 1 OF 1                                                                                       |
|-----------------------|--------------------------|---------------------------------|--------------------------------------------------------------------------------------------------|------------------|----------------------------------------------------------------------------------------------------------------------|
| SITE                  | BEAVER V                 | VALLEY POWER ST                 | FATION                                                                                           |                  | 10 NO 11700 BORING NO 561 亡                                                                                          |
| TYPE OF               | BORING                   | G SPLIT SPOON                   | LOCATIC                                                                                          | N SHIP           | PINGPORT, PENNSYLVANIA GROUND ELEV. 673.6                                                                            |
| DATE D                | RILLED                   | MAY 10, 1974                    | 4                                                                                                | DRIL             | LED BY LOGGED BY FPV                                                                                                 |
| SUMMAR                | YOFU                     | ORING                           |                                                                                                  |                  |                                                                                                                      |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |
| <u> </u>              | IL                       | OVERALL<br>WEATHERING           | SAMPLE                                                                                           | <u> </u>         | SOIL OR ROCK DESCRIPTION                                                                                             |
|                       | EEI                      | ROD                             | ₩<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S | H G              |                                                                                                                      |
| ليا ليـ               | ۳<br>۳                   | 0 25 50 75 100                  |                                                                                                  | 0<br>0<br>0<br>0 | OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                         |
|                       |                          |                                 | <b>b</b>                                                                                         | <b>P</b>         |                                                                                                                      |
| 673.6                 | r                        | r                               |                                                                                                  |                  |                                                                                                                      |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |
| 670                   | -                        |                                 |                                                                                                  |                  |                                                                                                                      |
|                       | 5                        |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |
|                       | -                        |                                 |                                                                                                  |                  | STUTY SAND. POORLY GRADED, COARSE TO FINE, MOSTLY FINE, 15-20%                                                       |
|                       | 10 —                     |                                 |                                                                                                  |                  | MODERATELY PLASTIC FINES, MOIST, MEDIUM DARK GRAYISH BROWN, FEW                                                      |
|                       |                          |                                 |                                                                                                  |                  | (SM)                                                                                                                 |
| 660                   | -                        | 1                               | W.O.H                                                                                            |                  | SANDY SILT: HIGHLY PLASTIC, 15-20% FINE SAND, VERY SOFT, MOIST,                                                      |
|                       | 15 -                     | ł                               | 2                                                                                                |                  | (ML)                                                                                                                 |
|                       | -                        | 1                               |                                                                                                  |                  | -                                                                                                                    |
|                       |                          | 4                               | 20                                                                                               | -                | GRAVELLY SAND: WELL GRADED, COARSE TO FINE, 5-10% SLIGHTLY PLASTIC                                                   |
|                       | 20 —                     |                                 | 3                                                                                                |                  | FINES, WET, MEDIUM BROWN, PEBBLES TO 1".                                                                             |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |
| 650                   |                          |                                 | 15                                                                                               | -                | NO RECOVERY.                                                                                                         |
|                       | 25 —                     | 1                               |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       |                          |                                 | 13                                                                                               | -                | GRAVELLY SAND: WELL GRADED, COARSE TO FINE, 1-3% NONPLASTIC FINES,                                                   |
|                       | 30 -                     |                                 | ~ 5                                                                                              |                  | MEDIUM BROWN, PEBBLES TO 1".<br>(SW)                                                                                 |
|                       | -                        |                                 |                                                                                                  |                  | -                                                                                                                    |
| 640                   | _                        |                                 | 17                                                                                               | -                | GRAVELLY SAND: AS ABOVE.                                                                                             |
|                       | 35 -                     |                                 | <b>•</b>                                                                                         |                  | (SW)                                                                                                                 |
|                       | -                        |                                 |                                                                                                  |                  | SAND; POORLY GRADED, COARSE TO FINE, MOSTLY FINE, LESS THAN 1%<br>NONPLASTIC FINES, MEDIUM BROWN, FEW PEBBLES TO 1". |
|                       | -                        | -                               | 19                                                                                               |                  | (SP) -                                                                                                               |
|                       | - 40                     |                                 | <u>× ′</u>                                                                                       |                  | END OF BORING AT 40.0                                                                                                |
|                       | -                        |                                 |                                                                                                  |                  | NOTE: HOLE TREMINGTED AT IO OF THE TO RESERVE HATTER AND THE OWNER DETERMINED                                        |
|                       | i                        | 4                               |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        | 1                               |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        |                                 |                                                                                                  |                  |                                                                                                                      |
|                       |                          |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        | 4                               |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        | 1                               |                                                                                                  |                  |                                                                                                                      |
|                       | -                        |                                 |                                                                                                  |                  |                                                                                                                      |
|                       | -                        |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        | ]                               |                                                                                                  |                  | -                                                                                                                    |
|                       | _                        |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        | 1                               |                                                                                                  |                  |                                                                                                                      |
|                       | -                        | ]                               |                                                                                                  |                  |                                                                                                                      |
|                       | _                        | 1                               |                                                                                                  |                  |                                                                                                                      |
|                       | -                        | }                               |                                                                                                  |                  | -                                                                                                                    |
|                       | -                        |                                 |                                                                                                  |                  | -                                                                                                                    |
|                       |                          | 4                               |                                                                                                  |                  | -                                                                                                                    |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |
| 1. FIG                | JRES IN                  | BLOW OR REG                     | COVERY C                                                                                         | DLUMN OPI        | POSITE                                                                                                               |
| SOI1<br>140           | L SAMPI<br>LB HAN        | LE DENOTE THI<br>IMER FALLING   | 30" REQI                                                                                         | OF BLOWS         | S OF A<br>DRI <b>VE</b>                                                                                              |
| A 2<br>FIG            | " OD SA<br>JRES SH       | AMPLE SPOON I<br>IOWN OPPOSITI  | L2" OR TH<br>E ROCK CO                                                                           | HE DISTAN        | ICE SHOWN.                                                                                                           |
| THE<br>2. <b>■2</b> 1 | PERCEN<br>INDICAT        | T OF CORE RECENTION             | SCOVERED.<br>OF UNDIS                                                                            | TURBED S         | BORING LOG 561t                                                                                                      |
| <b>76</b> 1<br>1171   | INDICAT<br>INDICAT       | ES LOCATION                     | OF SPLIT                                                                                         | SPOON S          | MPLE.                                                                                                                |
| SUBS                  | WITH <b>NG</b><br>SCRIPT | NEXT TO SYME                    | BOL INDIC                                                                                        | CATES SAM        | PLE DUOUESNE LICHT COMPANY                                                                                           |
| NUME<br>3• <u>목</u> 1 | BER.<br>INDICAT          | ES LOCATION                     | OF NATUR                                                                                         | AL GROUN         | DUGUESHE LIGHT CONFANT                                                                                               |
| 4. <u>R</u> QD        | ABLE.<br>- ROCK          | QUALITY DES                     | IGNATION                                                                                         | 34               | STONE & WEBSTER ENGINEERING CORPORATION                                                                              |
| 5• ∐ 1<br>6. DAT(     | NDICAT                   | 'ES DEPTH & L<br>Mean sea level | ENGTH OF                                                                                         | 'NX CORI         | NG RUN 11700 - GSK - 135                                                                                             |
|                       |                          |                                 |                                                                                                  |                  |                                                                                                                      |

|                               |                                                                         | DUQUESN                  | E LIGHT COMPANY                                                                                                                          |
|-------------------------------|-------------------------------------------------------------------------|--------------------------|------------------------------------------------------------------------------------------------------------------------------------------|
| CITE BRAVER VA                |                                                                         |                          | 10 No 31700 BORING No 562 C                                                                                                              |
| TYPE OF BORIN                 | IG SPLIT SPOON LOCATIO                                                  | N SHIPPING               | PORT. PENNSYLVANIA GROUND ELEV. 674.1                                                                                                    |
| DATE DRILLE                   | D 16, 1974<br>BORING                                                    | DRILLI                   | ED BYAMERICANLOGGED BYJDG                                                                                                                |
|                               |                                                                         |                          |                                                                                                                                          |
|                               | OVERALL SAMPLE                                                          |                          | SOUL OF POCK DESCRIPTION                                                                                                                 |
|                               |                                                                         |                          | JOIL ON ROCK DESCRIPTION                                                                                                                 |
|                               |                                                                         | G R A                    | FIELD AND LABORATORY TEST REBULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS   |
|                               |                                                                         | <u> </u>                 |                                                                                                                                          |
| 674.1                         | T T                                                                     |                          |                                                                                                                                          |
|                               |                                                                         |                          |                                                                                                                                          |
| 670 -                         | 21                                                                      |                          | SANDY GRAVEL, GRAVEL TO 1.25 INCH MAXIMUM, POORLY GRADED, COARSE                                                                         |
|                               |                                                                         |                          | TO FINE, MOSTLY FINE, SAND, 3-5% NONPLASTIC FINES, COMPACT, SATURATED, LIGHT BROWN, (GP).                                                |
|                               | -                                                                       | 7                        |                                                                                                                                          |
| 10 -                          |                                                                         |                          | NO RECOVERY.                                                                                                                             |
|                               |                                                                         |                          |                                                                                                                                          |
| 660 -                         |                                                                         |                          | SANDY SILT, SLIGHTLY PLASTIC. 3-5% UNIFORM FINE. SAND. SOFT. BLACK                                                                       |
| 15 -                          |                                                                         |                          | FEW PIECES OF COARSE SAND, (ML).                                                                                                         |
|                               | ן<br>איז 1/1 איז <b>בי</b>                                              |                          | TOP 12" - ORGANTO MATERIAL, HIGHLY PLASTIC SOFT BLACK (CH)                                                                               |
| 20 -                          | 26                                                                      |                          | BOTTOM 6" - SILT, MODERATELY PLASTIC, FIRM, GRAY BLACK.                                                                                  |
|                               | -                                                                       |                          | GRAVELLY SAND, 15-25% SUBROUNDED GRAVEL TO 0.75 INCH MAXIMUM,                                                                            |
| 650                           | 21                                                                      |                          | GRAVELLY SAND, SIMILAR TO ABOVE EXCEPT GRAVEL TO 1.5 INCH MAXIMUM, -                                                                     |
| 25 -                          |                                                                         |                          | (SF)                                                                                                                                     |
|                               |                                                                         | •                        |                                                                                                                                          |
| 30 •                          |                                                                         | 7                        | SAND, TRACE OF GRAVEL TO 0.25 INCH MAXIMUM, UNIFORM, FINE, LESS THAN-<br>3% NONPLASTIC FINES, COMPACT, SATURATED, GRAY GREEN, (SP)       |
|                               |                                                                         |                          | Z DISTINCT BLACK STRATA APPROXIMATELI 0.2" THICK UBVIOSSLI                                                                               |
| 640                           | 37                                                                      | 8                        | SANDY GRAVEL, WASHED OUT GRAVEL TO 1.0 INCH MAXIMUM, POORLY GRADED,                                                                      |
| 35 -                          |                                                                         |                          | (GP) (PIECES OF GRAVEL LODGED IN SHOE)                                                                                                   |
|                               |                                                                         |                          |                                                                                                                                          |
| 40                            |                                                                         | 9                        | GRAVELLY SAND, 15-20% SUBROUNDED TO ANGULAR GRAVEL, POURLY GRADED,<br>COARSE TO FINE, MOSTLY FINE, SAND, 3-5% NONPLASTIC FINES, COMPACT, |
|                               |                                                                         |                          |                                                                                                                                          |
| 630                           | - 25                                                                    |                          | SAND. TRACE OF GRAVEL TO 0.75 INCH MAXIMUM, POORLY GRADED, COARSE                                                                        |
| 45                            |                                                                         | 0                        | TO VERY FINE, MOSTLY UNIFORM, FINE, SAND, LESS THAN 2% NONPLASTIC                                                                        |
|                               |                                                                         |                          |                                                                                                                                          |
| 50                            |                                                                         | 1                        | GRAVELLY SAND, 10-15% SUBROUNDED GRAVEL TO 0.30 INCH MAAIMON,<br>POORLY GRADED, COARSE TO FINE, MOSTLY FINE SAND, 3-5% NONPLASTIC        |
|                               | -                                                                       |                          | SILTY SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, -                                                                          |
| .620 55                       | 60/61                                                                   |                          | 10-15% NONPLASTIC FINES, DENSE, SATURATED, LIGHT BROWN, (SM) - (SHALE CHIPS IN SHOE)                                                     |
|                               | ] ]                                                                     |                          | END OF BORING AT 54.5                                                                                                                    |
|                               | -                                                                       |                          | -                                                                                                                                        |
|                               | 7                                                                       |                          |                                                                                                                                          |
|                               | -                                                                       |                          |                                                                                                                                          |
|                               | ]                                                                       |                          |                                                                                                                                          |
|                               | -                                                                       |                          |                                                                                                                                          |
|                               | -                                                                       |                          |                                                                                                                                          |
|                               | -                                                                       |                          |                                                                                                                                          |
| l                             |                                                                         |                          |                                                                                                                                          |
| 1. FIGURES I<br>SOIL SAME     | IN BLOW OR RECOVERY C<br>PLE DENOTE THE NUMBER                          | OLUMN OPPO<br>OF BLOWS   | SITE<br>OF A                                                                                                                             |
| A 2" OD S                     | AMMER FALLING 30" REQ<br>SAMPLE SPOON 12" OR T<br>SHOWN OPPOSITE DOOR T | JIRED TO D<br>HE DISTANC | RIVE<br>E SHOWN.                                                                                                                         |
| THE PERCE                     | ENT OF CORE RECOVERED<br>ATES LOCATION OF UNDI                          | STURBED SAL              | MPLE. 4 BORING LOG 562 t                                                                                                                 |
|                               | TES LOCATION OF SPLI                                                    | S-SPOON SAL              | MPLE. BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                           |
| WITH N<br>SUBSCRIPT<br>NUMBER | NEXT TO SYMBOL INDI                                                     | CATES SAMP               | LE SHIPPINGPORT, PERNSYLVANIA                                                                                                            |
| 3. ¥ INDICA<br>TABLE.         | TES LOCATION OF NATU                                                    | RAL GROUND               | WATER 2 DUQUESHE LIGHT SOMPANY                                                                                                           |
| 4. RQD - ROC<br>5. ∐ INDICA   | K QUALITY DESIGNATION<br>TES DEPTH & LENGTH OF                          | I.<br>'NX CORING         | G RUN I /// 11700 - GST - 136                                                                                                            |
| O. DATOM 15                   | MEAN SEA LEVIEL                                                         |                          |                                                                                                                                          |

|                                         |                                      |                                                                   |                               | DIIQII          | ESNE LIGHT COMPANY SH SH OF                                                                                                                                                                                                                                                                                                                                                         |
|-----------------------------------------|--------------------------------------|-------------------------------------------------------------------|-------------------------------|-----------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SITE BE<br>TYPE OF<br>DATE DI<br>SUMMAR | BORING<br>BORING<br>RILLED<br>Y OF B | ALLEY POWER STAT<br>G <u>SPLIT SPOON</u><br>MAY 16, 1974<br>ORING | TION<br>LOCATION              | ISHTPP<br>DRILI | J.O. NO. 11700 BORING NO. 563 <del>C</del><br>INGPORT, PENNSYLWANIA GROUND ELEV. 674.4<br>LED BY AMERICAN LOGGED BY JDG                                                                                                                                                                                                                                                             |
| <u>у</u> н                              | Ξ_                                   | OVERALL<br>WEATHERING                                             | SAMPLE                        | ¥               | SOIL OR ROCK DESCRIPTION                                                                                                                                                                                                                                                                                                                                                            |
| FEE'                                    | DEPT                                 | AND<br>RQD<br>0 25 50 75 100                                      | BLOWS<br>OR<br>RECOV.<br>TYPE | G RAPH<br>LOG   | FIELD AND LABORATORY TEST REGULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                                                                                                                                                                                              |
|                                         |                                      |                                                                   |                               |                 |                                                                                                                                                                                                                                                                                                                                                                                     |
| 674.4                                   |                                      | T                                                                 |                               |                 |                                                                                                                                                                                                                                                                                                                                                                                     |
| 670 —                                   | 5                                    |                                                                   |                               |                 |                                                                                                                                                                                                                                                                                                                                                                                     |
|                                         | 10                                   |                                                                   | -                             |                 | (WASHED TO 13.5' - CASING SAME 1.5' WHILE CLEANING OUT.                                                                                                                                                                                                                                                                                                                             |
| 660                                     | 15<br>                               |                                                                   | 1/12"                         |                 | ORGANIC SILT, MODERATELY TO HIGHLY PLASTIC, 2-5% UNIFORM, VERY FINE _ SAND, BLACK (MH-OH)? (COAL DUST?)                                                                                                                                                                                                                                                                             |
|                                         | 20<br>-<br>-                         | 4                                                                 | PUSH 2                        |                 | ORGANIC SILT, SAME AS ABOVE (MH-OH?)                                                                                                                                                                                                                                                                                                                                                |
| 650                                     | 25                                   |                                                                   | 27                            |                 | GRAVELLY SAND, 35-45% SUBANGULAR GRAVEL TO 1.5 INCH MAXIMUM,<br>POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, 3-8% NONPLASTIC<br>FINES, COMPACT, SATURATED, YELLOW BROWN, (SP).                                                                                                                                                                                                 |
|                                         | -<br>30 -<br>-                       |                                                                   | 19 4                          |                 | TOP 10" - <u>SAND</u> , UNIFORM, FINE, LESS THAN 2% HOMPLASTIC FINES,<br>COMPACT, SATURATED, LAYERS OF LIGHT BROWN, ORANGE, BROWN AND<br>BLACK, (LAYERS UNDEFORMED), (SP). BOTTOM 2" - <u>GRAVELLY SAND</u> ,<br>SUBROUNDED GRAVEL TO 0.30 INCH MAXIMUM, POORLY GRADED, COARSE TO<br>FINE, MOSTLY FINE, SAND, 8-12% MOMPLASTIC FINES, COMPACT, SATURATED,-<br>LIGHT BROWN, (SP-SN). |
| 640                                     | 35 -                                 |                                                                   | 34                            |                 | GRAVELLY SAND, ONE PIECE OF GRAVEL TO 0.60 INCH MAXIMUM, AND A FEN<br>GRAINS OF COARSE SAND, (SP).                                                                                                                                                                                                                                                                                  |

•

· )

|                                        | 40 -                                                                                      |                                                                                                                 | 39 76                                                            |                                                   | SAND, TRACE OF SUBROUNDED GRAVEL TO 0.50 INCH MAXIMUM, POORLY<br>GRADED, COARSE TO FINE, MOSTLY FINE SAND, LESS THAN 3% NONPLASTIC<br>FINES, DENSE, SATURATED, LIGHT BROWN, (SP). |
|----------------------------------------|-------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|---------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 630                                    | 45 -                                                                                      |                                                                                                                 | 39                                                               |                                                   | SAND, SAME AS ABOVE EXCEPT TRACE OF GRAVEL TO 0.60 INCH MAXIMUM,                                                                                                                  |
|                                        | 50 -                                                                                      |                                                                                                                 | 38                                                               |                                                   | TOP 5" - SAND - UNIFORM, FINE, LESS THAN 3% NONPLASTIC FINES, DENSE                                                                                                               |
| 620                                    | 55 -                                                                                      |                                                                                                                 | 2" 9                                                             |                                                   | COARSE TO FINE MOSTLY FINE, SAND, LESS THAN 3% HOMPLASTIC, VERY                                                                                                                   |
|                                        |                                                                                           |                                                                                                                 |                                                                  |                                                   | END OF BORING AT 54.6                                                                                                                                                             |
| 1. FIGU<br>SOII<br>140<br>A 2"<br>FIGU | JRES IN<br>SAMPL<br>LB HAM<br>OD SAU<br>JRES SHO                                          | BLOW OR REC<br>E DENOTE THE<br>MER FALLING<br>MPLE SPOON 12<br>DWN OPPOSITE                                     | OVERY COL<br>NUMBER O<br>30" REQUI<br>2" OR THE<br>ROCK COR      | UMN OPP<br>F BLOWS<br>RED TO<br>DISTAN<br>ES DENO | POSITE<br>S OF A<br>DRIVE<br>NCE SHOWN.<br>OTE                                                                                                                                    |
|                                        | PERCEN<br>INDICAT<br>INDICAT<br>INDICAT<br>INDICAT<br>ITH NO<br>CRIPT I<br>BER.<br>NDICAT | T OF CORE RE(<br>ES LOCATION (<br>ES LOCATION (<br>ES LOCATION (<br>RECOVERY.<br>NEXT TO SYMB(<br>ES LOCATION ( | DVERED.<br>DF UNDIST<br>DF SPLIT-<br>DF SAMPLINDICA<br>DL INDICA | URBED S.<br>SPOON S.<br>NG ATTEL<br>TES SAMI      | BORING LOG 563 t<br>BORING LOG 563 t<br>BRAVER TALLEY, POWER STATICE - THET 19. 1<br>SHIPPINGPORT, PERNSYLVANIA<br>DUQUESNE LIGHT COMPANY                                         |
| 4. <u>ROD</u><br>5. ∏ I<br>6. DATU     | ABLE.<br>- ROCK<br>NDICATH<br>M IS ME                                                     | QUALITY DESI<br>CS DEPTH & LH<br>AN SEA LEVEL                                                                   | GNATION.<br>ENGTH OF 1                                           | NX CORI                                           | ING RUN                                                                                                                                                                           |



|                                           | 40                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                   | 61 🗾                                 |                              | GRAVELLY SAND, 5-10% SUBROUNDED GRAVEL TO 0.60 INCH MAXIMUM, POORLY<br>GRADED, COARSE TO FINE, MOSTLY FINE, SAND, LESS THAN 3% MOMPLASTIC<br>FINES, VERY DENSE, SATURATED, LIGHT BROWN, (SP).                                                                                                                             |  |  |  |  |
|-------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|--------------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| 630 -                                     | 45 ·                                                                                                                                                                                                                                                                                                                                                                                                | -                                                                 | 52                                   |                              | TOP 7" - <u>GRAVELLY SAND</u> , 5-10% SUBROUNDED GRAVEL TO 0.80 INCH<br>MAXIMUM, POORLY GRADED, COARSE TO FINE, MOSTLY FINE, SAND, LESS<br>THAN 5% MOMPLASTIC FINES, VERI DENSE, SATURATED, LIGHT BROWN, (SP)<br>BOTTOM 8" - <u>SAND</u> , UNIFORM, FINE, LESS THAN 3% MOMPLASTIC FINES,<br>DENSE, SATURATED, BROWN, (SP) |  |  |  |  |
|                                           | <b>5</b> 0 ·                                                                                                                                                                                                                                                                                                                                                                                        |                                                                   | 51 🛒                                 |                              | SAND, UNIFORM, FINE, ERSS THAN 3% NONPLASTIC FINES, VERY DENSE,                                                                                                                                                                                                                                                           |  |  |  |  |
| <b>6</b> 20                               | _                                                                                                                                                                                                                                                                                                                                                                                                   | -                                                                 |                                      |                              | TOP OF BOCK AT 53.51                                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                                           |                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                   |                                      |                              | END OF BORING AT 54.5'                                                                                                                                                                                                                                                                                                    |  |  |  |  |
| 1. FI<br>SC<br>14                         | GURES I<br>DIL SAMP<br>O LB HA                                                                                                                                                                                                                                                                                                                                                                      | N BLOW OR REC<br>LE DENOTE THE<br>MMER FALLING                    | OVERY COL<br>NUMBER O<br>30" REQUI   | UMN OPP<br>F BLOWS<br>RED TO | OSITE<br>OF A<br>DRIVE                                                                                                                                                                                                                                                                                                    |  |  |  |  |
| A<br>FI<br>2.<br>SU<br>NU                 | SOLL SAFFLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FIGURES SHOWN OPPOSITE ROCK CORES DENOTE<br>THE PERCENT OF CORE RECOVERED.<br>2 INDICATES LOCATION OF UNDISTURBED SAMPLE.<br>INDICATES LOCATION OF SPLIT-SPOON SAMPLE.<br>INTH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMBOL INDICATES SAMPLE<br>NUMBER |                                                                   |                                      |                              |                                                                                                                                                                                                                                                                                                                           |  |  |  |  |
| 3. <del>¥</del><br>4. RQ<br>5. ∐<br>6. DA | INDICA<br>TABLE.<br>D - ROC<br>INDICA<br>TUM IS                                                                                                                                                                                                                                                                                                                                                     | TES LOCATION<br>K QUALITY DES<br>TES DEPTH & LI<br>MEAN SEA LEVEL | OF NATURA<br>IGNATION.<br>ENGTH OF 1 | L GROUNI<br>NX CORIN         | WATER 2 DE DUQUESHE LIGHT COMPANY<br>STONE & WEBSTER ENGINEERING CORPORATION<br>I III 111 III III IIII IIII IIII IIIII IIIIII                                                                                                                                                                                             |  |  |  |  |

|              |                                  |                                                   |                                                                             |                                             |                                         |                                                      | DITQUESNE LIGHT COMPANY SH1 OF 1                                                                                                                                                   |
|--------------|----------------------------------|---------------------------------------------------|-----------------------------------------------------------------------------|---------------------------------------------|-----------------------------------------|------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SITE         |                                  | BEAVER                                            | VALLEY POWER                                                                | STATIO                                      | 1                                       |                                                      | J.O. NO. 11700 BORING NO. 565t                                                                                                                                                     |
| DATE         | OF<br>DI                         | BORING                                            | <u></u>                                                                     |                                             |                                         | DRIL                                                 | LED BY AMERICAN LOGGED BY JDG                                                                                                                                                      |
| SUMN         |                                  | Y OF B                                            | ORING                                                                       | ·····                                       |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  |                                                   | OVERALL                                                                     | SAM                                         | PLE                                     | <u>с</u>                                             | SOUL OR ROCK DESCRIPTION                                                                                                                                                           |
| ELEV.        | FEET                             | DEPTH                                             | WEATHERING<br>AND<br>RQD<br>0 25 50 75 100                                  | BLOWS<br>RECOV.                             | TYPE                                    | G R APHI<br>LOG                                      | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                              |
| 647.0        | 0                                |                                                   |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  |                                                   |                                                                             | 11                                          | 1                                       |                                                      | <u>GRAVELLY SAND</u> , 15-20% SUBROUNDED GRAVEL TO 0.75 IN. MAX., UNIFORM, -<br>FINE SAND, 10-15% SLIGHTLY PLASTIC FINES, COMPACT, SATURATED, BLACK<br>(SP-SM) (SOME COARSE SAND.) |
| 640          |                                  | <b>5</b> —<br>—<br>—                              |                                                                             | 58                                          | 2                                       |                                                      | GRAVELLY SAND, 15-20% FLAT TO SUBROUNDED GRAVEL TO 0.50 IN. MAX.,                                                                                                                  |
|              |                                  | - 10<br>                                          |                                                                             |                                             |                                         |                                                      | FINES, VERY DENSE, SATURATED, GRAY BROWN WITH GREEN STAINS. (SP)                                                                                                                   |
| 620          | ×                                | 15 -                                              |                                                                             | 51                                          | 73                                      |                                                      | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY FINE SAND, 5-10% NON-<br>PLASTIC, FINES, VERY DENSE, MOIST, BROWN. (SP)                                                                |
| 0,0          |                                  | -<br>20                                           |                                                                             | 74                                          | 4                                       |                                                      | GRAVELLY SAND, 5-10% SUBANGULAR GRAVEL TO 1.0 IN. MAX., POORLY<br>GRADED, COARSE TO FINE, MOSTLY FINE SAND, 3-8% NONPLASTIC FINES,<br>VERY DENSE, SATURATED, LIGHT BROWN. (SP)     |
|              |                                  | -<br>-<br>25 -                                    |                                                                             | 37                                          | 5                                       |                                                      | SAND, TRACE OF GRAVEL TO 0.25 IN. MAX., POORLY GRADED, COARSE TO<br>FINE, MOSTLY FINE SAND, 1-3% NONPLASTIC FINES, DENSE, SATURATED,<br>BROWN. (SP)                                |
| 620          |                                  |                                                   |                                                                             | <u>100</u><br>4"                            | 6                                       |                                                      | SHALE.                                                                                                                                                                             |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      | END OF BORING @ 27.8'.                                                                                                                                                             |
|              |                                  | -                                                 | •                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 | 4                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | _                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  |                                                   | 4                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 | 4                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | - I                                               | -                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 |                                                                             |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 | 1                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
| Ì            |                                  | -                                                 | 1                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | •                                                 | 1                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | ·                                                 | 1                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  | -                                                 | 1                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
|              |                                  |                                                   | 1                                                                           |                                             |                                         |                                                      | -                                                                                                                                                                                  |
|              |                                  |                                                   | ]                                                                           |                                             |                                         |                                                      |                                                                                                                                                                                    |
| 1. 1         | FIG<br>SOI<br>140<br>A 2'<br>FIG | URES II<br>L SAMP<br>LB HAI<br>" OD S.<br>URES SI | N BLOW OR RE<br>LE DENOTE TH<br>MMER FALLING<br>AMPLE SPOON<br>HOWN OPPOSIT | COVER<br>E NUM<br>30" 1<br>12" 01<br>E BOCI | Y CO<br>BER (<br>REQUI<br>R THI<br>C CO | LUMN OP<br>DF BLOW<br>IRED TO<br>E DISTA<br>RES DENI | POSITE<br>S OF A<br>DRIVE<br>NCE SHOWN.                                                                                                                                            |
| 2.           | THE<br>2                         | PERCE                                             | NT OF CORE R<br>TES LOCATION                                                | ECOVEI<br>OF UI                             | RED.                                    | URBED                                                | BORING LOG 565t                                                                                                                                                                    |
|              | <b>7</b> 61<br>071               | INDICA'<br>INDICA'<br>NITH NO                     | TES LOCATION<br>TES LOCATION<br>RECOVERY                                    | of Si<br>Of Si                              | PLIT.<br>AMPLI                          | -SPOON S<br>ING ATTI                                 | BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                                                           |
|              | SUBS<br>NUMI                     | SCRIPT<br>SER.                                    | NEXT TO SYM                                                                 | BOL IN                                      | NDICA                                   | ATES SAI                                             | MPLE SHIPPINGPORT, PENNSYLVANIA                                                                                                                                                    |
| 3. 4<br>4. 1 | ¥– ]<br>]<br>ROD                 | ABLE.                                             | CONALTY DE                                                                  | OF NA                                       | ATURA                                   | L GROUN                                              | ND WATER 2 STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                 |
| 5.  <br>6.   |                                  | INDICAT                                           | TES DEPTH & T<br>MEAN SEA LEVEL                                             | LENGTH                                      | I OF                                    | NX CORI                                              | ING RUN 11700 - GSK - 139                                                                                                                                                          |

`)

| ·          |                                                   |                       |                              |          |                                                                                                                                        |
|------------|---------------------------------------------------|-----------------------|------------------------------|----------|----------------------------------------------------------------------------------------------------------------------------------------|
|            |                                                   |                       |                              | DUQUESNE | LIGHT COMPANY SH 1 OF 1                                                                                                                |
| SITE       | BEAVER                                            | VALLEY POWER ST       | r <u>át</u> ion              |          | J.O. NO. 11700 BORING NO. 566 Z                                                                                                        |
| TYPE OF    | HIPPINGPORT, PENNSYLVANIA GROUND ELEV. 650.8650.8 |                       |                              |          |                                                                                                                                        |
| DATE DI    | RILLED                                            | JUNE 4, 1974          | +                            | DRIL     | LED BYAMERICANLOGGED BY                                                                                                                |
|            |                                                   |                       |                              |          |                                                                                                                                        |
|            |                                                   |                       |                              |          |                                                                                                                                        |
| ×Ε         | H                                                 | OVERALL<br>WEATHERING | SAMPL                        | 튁읖。      | SOIL OR ROCK DESCRIPTION                                                                                                               |
| ELE<br>FEE | FEE<br>DEP                                        | RQD<br>a zs so ts 100 | BLOWS<br>OR<br>RECOV<br>TYPE | G R API  | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS |
|            | . <u></u>                                         | ·····                 |                              |          |                                                                                                                                        |
| 650,8      |                                                   |                       |                              | <b></b>  |                                                                                                                                        |
| 650 —      | -                                                 | r.                    | 42 🔽                         | i        | GRAVELLY SAND, 10-15% GRAVEL TO 0.8 IN. MAX., UNIFORM, FINE SAND, -<br>8-10% SLIGHTLY PLASTIC FINES, BROWN. (SP)                       |
|            | -<br>5 _                                          |                       | 51                           | 2        | SANDY GRAVEL, POORLY GRADED TO 2.5 IN. MAX., 10-15% FINE TO MEDIUM SAND, MOSTLY FINE, 8-12% FINES, BROWN. (GP)                         |
|            |                                                   |                       |                              |          | -                                                                                                                                      |
|            | -                                                 | {                     |                              |          |                                                                                                                                        |
|            | 10 —                                              |                       | 65 💌                         | 3        | SAND, POORLY GRADED, FINE TO COARSE, 5-8% MEDIUM AND COARSE SAND,<br>8-10% GRAVEL TO 2.0 IN. MAX., 5-8% SLIGHTLY PLASTIC FINES, LIGHT  |
| 640        | -                                                 |                       |                              |          | BROWN. (SP)                                                                                                                            |
|            | -                                                 |                       | 108                          | <b>7</b> | -                                                                                                                                      |
|            | 15 —                                              |                       | 108                          | 4        | GRAVEL TO 1.0 IN. MAX., 4-7% FINES, LIGHT BROWN. (SP)                                                                                  |
|            | _                                                 |                       |                              |          | NOTE: DRILLER BELIEVED TO BE PUSHING COBBLE.                                                                                           |
|            |                                                   | 4                     | 67                           | 5        | SAND POORLY CRADED FINE TO COARSE MOSTLY FINE AND MEDIUM 2-54                                                                          |
| 630        | 20 —                                              | 4                     |                              |          | GRAVEL TO 0.9 IN. MAX., 3-5% FINES, LIGHT BROWN. (SP)                                                                                  |
|            | -                                                 |                       |                              |          |                                                                                                                                        |
|            |                                                   | 1                     | 57                           |          | SAND. UNIFORM. FINE. 3-5% FINES. LIGHT BROWN WITH A 2" LAVER OF                                                                        |
|            | 25 -                                              | 1                     |                              | -        | MEDIUM AND COARSE SAND.                                                                                                                |
|            | -                                                 | ł                     |                              |          |                                                                                                                                        |
|            | -                                                 | 1                     | 55                           | 7        | SAND, POORLY GRADED FINE TO COARSE, MOSTLY FINE AND MEDIUM. 3-5%                                                                       |
| 620        | 30 -                                              |                       |                              |          | GRAVEL TO 1.0 IN. MAX., LESS THAN 5% FINES, LIGHT BROWN. (SP) -                                                                        |
|            |                                                   |                       |                              |          | END OF BORING @ 31:25:                                                                                                                 |
|            |                                                   |                       |                              |          |                                                                                                                                        |
|            | 35 -                                              |                       |                              |          |                                                                                                                                        |

|                |                                             |                                                |                    | -                                        |
|----------------|---------------------------------------------|------------------------------------------------|--------------------|------------------------------------------|
|                | -                                           |                                                |                    | -                                        |
|                |                                             |                                                |                    | •                                        |
|                | -                                           |                                                |                    | •                                        |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    | •                                        |
|                |                                             |                                                |                    | -                                        |
|                | -                                           |                                                |                    | -                                        |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    |                                          |
| ,              |                                             |                                                | 1                  |                                          |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    | -                                        |
|                | -                                           |                                                |                    | -                                        |
|                |                                             |                                                |                    |                                          |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    | · · · · ·                                |
|                |                                             |                                                |                    | _                                        |
|                |                                             |                                                |                    | · · · · · · · · · · · · · · · · · · ·    |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    | -                                        |
|                |                                             |                                                |                    |                                          |
| L. FIG         | URES IN BLOW OR REC                         | OVERY COLIMN OPP                               | ACT TO D           |                                          |
| SOI            | L SAMPLE DENOTE THE                         | NUMBER OF BLOWS                                | OF A               |                                          |
| A 2            | " OD SAMPLE SPOON 1                         | 2" OR THE DISTAN                               | DRIVE<br>CE SHOWN. |                                          |
| FIG<br>THE     | URES SHOWN OPPOSITE<br>PERCENT OF CORE RE   | COVERED.                                       |                    |                                          |
| 2. <b>∎</b> 2  | INDICATES LOCATION<br>INDICATES LOCATION    | OF UNDISTURBED S.                              | AMPLE.             | BORING LOG 566t                          |
| DЙ.            | INDICATES LOCATION                          | OF SAMPLING ATTE                               | MPT                | BEAVER VALLEY POWER STATION - UNIT NO. 1 |
| SUB            | SCRIPT NEXT TO SYME                         | OL INDICATES SAM                               |                    | SHIPPING PORT, PENNSYLVANIA              |
| 3• ¥           | DER.<br>INDICATES LOCATION                  | OF NATURAL GROUN                               | D WATER            | DUQUESNE LIGHT COMPANY                   |
| 4. <u>R</u> OD | TABLE.<br>- ROCK QUALITY DES                | IGNATION.                                      | JAND 1             | STONE & WEBSTER ENGINEERING CORPORATION  |
| 5. ∐<br>6. DAT | INDICATES DEPTH & L<br>UM IS mean gea truet | ENGTH OF NX CORIN                              | NG RUN             | 11700 - GSK - 140                        |
| ·······        |                                             | والمراجع والمراجع والبرابية والمتكفية والمراجع |                    |                                          |

|                                                                                          |                                                                                                                          |                                                                                                          |                                                                                                      |                                                                                             |                                                                                                |                                                                                                            | SH 1 OF 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
|------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|                                                                                          | CITE BRAVER VALLEY POWER STATION 567 -                                                                                   |                                                                                                          |                                                                                                      |                                                                                             |                                                                                                |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| TYPE OF                                                                                  | BORING                                                                                                                   | SPLIT                                                                                                    | SPOOL                                                                                                | LOC                                                                                         | ATION                                                                                          | SHIPPI                                                                                                     | GPORT, PERSYLVANIA GROUND ELEV. 651.7 651.7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |  |  |
| DATE DI<br>SUMMAR                                                                        | RILLED                                                                                                                   | ORING _                                                                                                  | 5, 1974                                                                                              | ·                                                                                           | · · · ·                                                                                        | DRIL                                                                                                       | LED BY LOGGED BY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|                                                                                          |                                                                                                                          |                                                                                                          |                                                                                                      | ·····                                                                                       | - <u> </u>                                                                                     |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
|                                                                                          | x_                                                                                                                       | OVER                                                                                                     |                                                                                                      | SAM                                                                                         | PLE                                                                                            |                                                                                                            | SOIL OR ROCK DESCRIPTION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| ELE/<br>FEE1                                                                             | DEPT<br>FEE1                                                                                                             | AN<br>RQ<br>0 25 50                                                                                      | D<br>75 100                                                                                          | BLOWS                                                                                       | TYPE                                                                                           | G R APH<br>LOG                                                                                             | FIELD AND LABORATORY TEST RESULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING BEDDING AND FAULTING AND TEXTURE<br>DESCRIPTIONS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| 651 7                                                                                    |                                                                                                                          |                                                                                                          |                                                                                                      |                                                                                             |                                                                                                |                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |  |  |
| 650                                                                                      |                                                                                                                          |                                                                                                          |                                                                                                      | 22                                                                                          | 1                                                                                              |                                                                                                            | SAND, POCHLY GRADED, FINE TO COARSE MOSTLY FINE, LESS THAN 25<br>GRAVEL TO 0.8 INCH MAXIMUM, 6-125 SLIGHTLY PLASTIC FINES, DARK -<br>MONN, (SP-SM).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                                                                                          | 5 -                                                                                                                      |                                                                                                          |                                                                                                      | 35                                                                                          |                                                                                                |                                                                                                            | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, MOSTLY FINE AND -<br>MEDIUM, 10-16% GRAVEL TO 1.75 INCH MAXIMUM, 8-10% SLIGHTLY PLASTIC -<br>FINES, DARK BROWN, CONTAINING A 1/8" THICK LAYER OF LIGHT YELLOW -<br>CEMENTED SAND, (SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| <b>64</b> 0                                                                              | 15                                                                                                                       |                                                                                                          |                                                                                                      | 8                                                                                           | 3                                                                                              |                                                                                                            | SILTY SAND, UNIFORM, FINE, VERY FINE, 12-15% MODERATELY PLASTIC<br>FINES, DARK GRAY WITH ONE PIECE OF GRAVEL 1.5 INCH IN SIZE AND<br>LESS THAN 25 MEDIUM AND COARSE SAND. (SM) (SM)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|                                                                                          | <b>89</b><br>                                                                                                            |                                                                                                          |                                                                                                      | 112                                                                                         | <b>F</b> .                                                                                     |                                                                                                            | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, NOSTLY MEDIUM AND -<br>COARSE, 12-15% GRAVEL TO 1.5 INCH MAXIMUM, LESS THAN 5% FINES,<br>LIGHT BROWN, (SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |
| 630                                                                                      |                                                                                                                          |                                                                                                          |                                                                                                      | 102                                                                                         | <b>7</b> 5                                                                                     |                                                                                                            | SAND, POORLY GRADED, FINE TO COARSE, MOSTLY MEDIUM AND COARSE,<br>8-10% GRAVEL TO 0.9 INCH MAXIMUM, LESS THAN 5% FINES, YELLOWISH GRAY,<br>(SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |  |  |
|                                                                                          |                                                                                                                          |                                                                                                          |                                                                                                      | 80                                                                                          | 6                                                                                              |                                                                                                            | SAND, UNIFORM, FINE, 2-4,5 FINES, YELLOWISH BROWN, (SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
|                                                                                          | 30                                                                                                                       |                                                                                                          |                                                                                                      | <b>1.10</b><br>2 1/5                                                                        | 7                                                                                              |                                                                                                            | SAND, POORLY GRADED FINE TO COARSE, NOSTLY MEDIUM 6-125 GRAVEL TO<br>0.7 INCH MAXIMUM, 3-65 FINES YELLOWISH BROWN, (SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| 620                                                                                      |                                                                                                                          |                                                                                                          |                                                                                                      |                                                                                             |                                                                                                |                                                                                                            | HED OF BOREHS AT 31,31                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |
| 1. FIG<br>SOI<br>140<br>A 2'<br>FIG<br>THE<br>2. ■2<br>V6<br>UV<br>SUBS<br>NUM<br>3. ¥ 1 | URES IN<br>L SAMPI<br>LB HAM<br>OD SA<br>URES SH<br>PERCEN<br>INDICAT<br>INDICAT<br>WITH NO<br>SCRIPT<br>BER.<br>INDICAT | H BLOW<br>LE DENO<br>IMER FA<br>AMPLE S<br>IOWN OP<br>IT OF C<br>TES LOC<br>TES LOC<br>NES LOC<br>NEXT T | OR REG<br>TE THI<br>LLING<br>POON :<br>POSITI<br>ORE RI<br>ATION<br>ATION<br>ERY.<br>O SYMM<br>ATION | COVER<br>E NUM<br>30"<br>12" O<br>E ROC:<br>E ROC:<br>OF U<br>OF S<br>OF S<br>BOL I<br>OF N | Y COI<br>BER C<br>REQUI<br>R THE<br>K COF<br>RED.<br>NDIST<br>PLIT-<br>AMPLI<br>NDICA<br>ATURA | UMN OPI<br>DF BLOW<br>IRED TO<br>DISTAN<br>RES DENC<br>URBED S<br>SPOON S<br>NG ATTE<br>TES SAN<br>L GROUN | POSITE<br>S OF A<br>DRIVE<br>NCE SHOWN.<br>OTE<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE.<br>SAMPLE. |  |  |
| 4. <u>RQ</u> D<br>5. ∐ 1<br>6. DAT                                                       | - ROCK<br>INDICAT                                                                                                        | QUALI<br>ES DEP<br>MEAN SE                                                                               | TY DES<br>TH & I<br>A LEVEI                                                                          | BIGNAT<br>LENGTI                                                                            | TION.<br>H OF                                                                                  | NX CORI                                                                                                    | ENG RUN I AND THE A WEBSTER ENGINEERING CORPORATION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |

Ì



|                                                                                           | 40                                                                                               |                                | 37 8                                                                                                                   |                     | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, MOSTLY FINE, 8-12%<br>GRAVEL TO 1.75 INCH MAXIMUM, 3-7% FINES, LIGHT BROWN<br>(SP) |  |  |  |  |  |
|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------|------------------------------------------------------------------------------------------------------------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|
| 630                                                                                       | 45 —<br>-<br>-                                                                                   |                                | SAND, UNIFORM, FINE, 5-8% MEDIUM AND COARSE SAND, 2-4% FINES,<br>LIGHT BROWN, 1-2% GRAVEL TO 1.1 INCH MAXIMUM.<br>(SP) |                     |                                                                                                                                  |  |  |  |  |  |
|                                                                                           |                                                                                                  |                                | 32 10                                                                                                                  |                     | SAND, POORLY GRADED, FINE TO COARSE, MOSTLY MEDIUM, 2-4% GRAVEL<br>TO 0.9 INCH MAXIMUM, 2-3% FINES, BROWN.<br>(SP)               |  |  |  |  |  |
|                                                                                           | _55                                                                                              |                                | 100/1"                                                                                                                 |                     | NO RECOVERY REFUSAL AT 55.1'                                                                                                     |  |  |  |  |  |
| 620                                                                                       |                                                                                                  |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
|                                                                                           |                                                                                                  |                                |                                                                                                                        |                     | END OF BORING AT 56.5:                                                                                                           |  |  |  |  |  |
|                                                                                           |                                                                                                  |                                |                                                                                                                        |                     | -                                                                                                                                |  |  |  |  |  |
|                                                                                           | -                                                                                                |                                |                                                                                                                        |                     | _                                                                                                                                |  |  |  |  |  |
|                                                                                           | _                                                                                                |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
|                                                                                           | -                                                                                                |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
|                                                                                           | -                                                                                                |                                |                                                                                                                        |                     | -                                                                                                                                |  |  |  |  |  |
|                                                                                           | _                                                                                                |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
|                                                                                           |                                                                                                  |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
| l. FIGU                                                                                   | JRES IN                                                                                          | BLOW OR REC                    | OVERY COL                                                                                                              | UMN OPPO            | POSITE                                                                                                                           |  |  |  |  |  |
| S011<br>140                                                                               | . SAMPI<br>LB HAM                                                                                | E DENOTE THE<br>MER FALLING    | NUMBER O<br>30" REOUT                                                                                                  | F BLOWS<br>RED TO I | S OF A DRIVE                                                                                                                     |  |  |  |  |  |
| A 2"<br>FIGU                                                                              | OD SA                                                                                            | MPLE SPOON 1<br>OWN OPPOSITE   | 2" OR THE<br>ROCK COR                                                                                                  | DISTAN              | NCE SHOWN.                                                                                                                       |  |  |  |  |  |
| THE<br>2. <b>■2</b> I                                                                     | PERCEN                                                                                           | T OF CORE REC                  | COVERED.                                                                                                               | URBED SI            | BORING LOG 568 t                                                                                                                 |  |  |  |  |  |
|                                                                                           | NDICAT<br>NDICAT                                                                                 | ES LOCATION (<br>ES LOCATION ( | OF SPLIT-                                                                                                              | SPOON SI            | SAMPLE. BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                 |  |  |  |  |  |
| WITH NO RECOVERY.<br>SUBSCRIPT NEXT TO SYMPOL INDICATES SAMPLE SHIPPINGPORT, PENNSYLVANIA |                                                                                                  |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
| NUMB<br>3. ⊈ I                                                                            | NUMBER.<br>3. V INDICATES LOCATION OF NATURAL CROWND WARDED 1/1/1 DOQUENSE LIGHT COMPANY         |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
| 4. ROD                                                                                    | TABLE.<br>4. ROD - ROCK QUALITY DESIGNATION                                                      |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |
| 5. ∐ I<br>6. DATU                                                                         | 5. I INDICATES DEPTH & LENGTH OF NX CORING RUN. 1<br>6. DATUM IS MEAN SEA LEVEL 11700 - GSK -142 |                                |                                                                                                                        |                     |                                                                                                                                  |  |  |  |  |  |

SH\_1 OF1 DUQUESNE LIGHT COMPANY BORING NO. \_560 t J.O. NO. \_\_\_\_\_\_\_ SITE \_\_ BEAVER VALLEY POWER STATION GROUND ELEV. 671.01 TYPE OF BORING SPLIT SPOON LOCATION 6-14 DRILLED BY \_\_\_\_AMERICAN LOGGED BY D.F.P. DATE DRILLED \_\_\_\_\_\_ JUNE 12, 1974 SUMMARY OF BORING \_\_ RAPHIC OVERALL **SAMPLE** OR ROCK DESCRIPTION DEPTH FEET SOIL WEATHERING ELEV FEET ğ BLOWS RECOV. TYPE RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING BEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 O 671.0 670 5 GRAVELLY SAND, POCRLY GRADED, FINE TO COARSE, 10-15% GRAVEL TO 2.5 INCHES MAXIMUM, 5-7% FINES, LIGHT BROWN. 8 (SP) 10 SANDY SILT, MODERATELY PLASTIC, 15-18% VERY FINE SAND, BLACK. 3 2 660  $(\mathbf{M})$ 15 2 3 SIMILAR TO SS#2. 20 SAND, UNIFORM, FINE, 4-8% SLIGHTLY TO MODERATELY PLASTIC FINES, LIGHT BLUEISH GRAY, CONTAINING 5-8% GRAVEL TO 1.7 INCH MAXIMUM. 29 650 4 (SP) 25 SAND, UNIFORM, FINE CLEAN 1-2% FINES, 2-4% MEDIUM SAND, PALE BROWN. (SP) 14 5 30 16 6 SIMILAR TO SS #5 640 35

|                                        |                                                                                                                                                                                                                                             | 20 🕨 7                                                           | SAND, UNIFORM, FINE, CLEAN, 1-2% FINES, LIGHT BROWN.                                                              |  |  |  |  |  |  |  |
|----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|--|
| 630 —                                  | 40 -                                                                                                                                                                                                                                        | 18 8                                                             | SAND, UNIFORM, FINE, 3-5% MEDIUM AND COARSE, CLEAN, LESS THAN 2%<br>FINES, LIGHT BROWN.<br>(SP)                   |  |  |  |  |  |  |  |
|                                        | 45 <del>-</del><br>-<br>-<br>-                                                                                                                                                                                                              | 20 9                                                             | SAND, UNIFORM FINE, 3-5% MEDIUM SAND, 1-3% FINES, BROWN, WITH LESS<br>THAN 1% GRAVEL TO 0.6 INCH MAXIMUM.<br>(SP) |  |  |  |  |  |  |  |
| 620                                    | -<br>50 -<br>-                                                                                                                                                                                                                              | 29 10                                                            | SAND, UNIFORM, FINE, 4-6% MEDIUM AND COARSE SAND, LESS THAN 3%<br>GRAVEL TO 1.1 INCH MAXIMUM, 3-5% FINES BROWN.   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  | END OF BORING AT 52.5'                                                                                            |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        | -                                                                                                                                                                                                                                           |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        | -                                                                                                                                                                                                                                           |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
|                                        |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
| 1. FIGU<br>SOII<br>140<br>A 2'<br>FIGU | 1. FIGURES IN BLOW OR RECOVERY COLUMN OPPOSITE<br>SOIL SAMPLE DENOTE THE NUMBER OF BLOWS OF A<br>140 LB HAMMER FALLING 30" REQUIRED TO DRIVE<br>A 2" OD SAMPLE SPOON 12" OR THE DISTANCE SHOWN.<br>FIGURES SHOWN OPPOSITE BOCK COPES DENOTE |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |
| THE<br>2. ■21<br>761<br>[7]            | PERCENT OF CORE F<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION                                                                                                                                                         | ECOVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON<br>OF SAMPLING ATT | SAMPLE.<br>SAMPLE.<br>EMPT BEAVER VALLEY POWER STATION - UNIT NO. 1                                               |  |  |  |  |  |  |  |
| SUBS<br>NUME                           | SCRIPT NEXT TO SYM<br>BER.                                                                                                                                                                                                                  | BOL INDICATES SA                                                 | MPLE SHIPPINGPORT, PENNSYLVANIA                                                                                   |  |  |  |  |  |  |  |
| 3• ¥ I<br>T                            | NDICATES LOCATION                                                                                                                                                                                                                           | OF NATURAL GROU                                                  | ND WATER 2 20 STONE & WERSTER ENGINE ERING CORRORATION                                                            |  |  |  |  |  |  |  |
| 4. <u>ROD</u><br>5. ∐. I<br>6. DAT(    | – ROCK QUALITY DE<br>INDICATES DEPTH &<br>JM IS MEAN SEA LEVET                                                                                                                                                                              | SIGNATION.<br>LENGTH OF NX COR                                   | ING RUN I ALL 11700 - GSK - 143                                                                                   |  |  |  |  |  |  |  |
| Procession and the second second       |                                                                                                                                                                                                                                             |                                                                  |                                                                                                                   |  |  |  |  |  |  |  |

| DUDUESNE LICHT COMPANY SH 1 OF 1                                 |                              |                              |            |          |                    |                              |                                |                        |                                    |                             |
|------------------------------------------------------------------|------------------------------|------------------------------|------------|----------|--------------------|------------------------------|--------------------------------|------------------------|------------------------------------|-----------------------------|
| SITE BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 570 C |                              |                              |            |          |                    |                              |                                |                        |                                    |                             |
| DATE D                                                           | RILLED                       | JULI 13, 1974                |            | DRIL     | LED BY             | AMERI                        | CAN                            | LOGG                   | ED BY <u> </u>                     | <u> </u>                    |
| SUMMAR                                                           | Y OF B                       | ORING                        |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  |                              |                              |            |          |                    |                              |                                |                        |                                    |                             |
| ゾー                                                               | Ξ⊢                           | OVERALL<br>WEATHERING        | SAMPLE     | ¥<br>₽   |                    | S                            | OIL OR I                       | ROCK                   | ESCRIPTIO                          | <u>N</u>                    |
|                                                                  | E E E                        | RQD                          | YPE        | a g      | FIELD              | AND LABOR                    | ATORY TEST RE                  | EGULTS;                | SOIL STRATA DESC                   | RIPTION; LITHOLOGY          |
|                                                                  | Δ-                           | 0 25 50 75 100               |            | 9        | OR JOI<br>Descr    | NTING BEC                    | IDING AND FAUL                 | TING                   | AND TEXTURE                        | · •                         |
| 654 6                                                            |                              |                              |            |          |                    |                              |                                |                        |                                    |                             |
| 0, 10                                                            |                              |                              | T          |          |                    |                              |                                | ••••                   | <u> </u>                           |                             |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | 5-                           |                              | 30 1       |          | GRAVEL 1           | POORLY GE                    | ADED, FINE TO<br>CH MAXIMUM, ( | 0 MEDIUM,<br>37≸ SLIG  | MOSTLY FINE, LE<br>TLY PLASTIC FIN | SS THAN 35<br>ES, DARK GRAY |
|                                                                  |                              |                              |            |          | (SP)               |                              |                                |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                | ·                      |                                    | -                           |
|                                                                  | 10                           |                              | 44 2       |          | GRAVELLY<br>GRAVEL | <u>( SAND</u> ,<br>10 2.0 IN | POORLY GRADE                   | d, Fine W<br>35% Fines | ) MEDIUM, MOSTLY<br>5, DARK GRAY.  | FINE, 12-15%                |
| 640                                                              | -                            |                              |            |          | (SP)               |                              |                                |                        |                                    |                             |
|                                                                  | -                            |                              | 61 3       |          | SANDY GI           | RAVIE. F                     | OORLY GRADED                   | TA 10 TI               | <b>PU MAYTMIN 10.</b>              | -<br>                       |
|                                                                  |                              |                              |            |          | COARSE S<br>(GP)   | SAND, MOS                    | TLY MEDIUM A                   | ND COARSE,             | 4-7% FINES, DA                     | RK BROWN.                   |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | 20                           |                              | 67 4       |          | SANDY GI           | RAVEL, P                     | OORLY GRADED                   | TO 2.0 II              | CH MAXIMUM, 8-1                    | 25 FINE TO COARSE           |
| <sup>630</sup>                                                   | -                            |                              |            |          | (GP)               | AJIDI MOD                    | TOM AND GOAN                   | 55, 7 <b>-</b> 77 1    | IRES, MARA DRAW                    | A•                          |
|                                                                  |                              |                              |            |          |                    |                              |                                |                        |                                    | -                           |
| 1                                                                | 25                           |                              | 60 5       |          | SAND, N<br>(SW)    | IELL GRAD                    | ED, FINE TO                    | MEDIUM, CI             | EAN 1-2% FINES,                    | LIGHT BROWN.                |
|                                                                  |                              |                              | 100/1"_6   |          | NO RECOV           | ERY                          |                                |                        |                                    | -                           |
|                                                                  | 30 -                         |                              | ,          |          | END OF B           | ORING AT                     | 28.11                          |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | ۰<br>•                      |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | _                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | _                            | 4                            |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            |                              |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            | ]                            |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            | -                            |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            | ]                            |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            | 1                            |            |          |                    | <b>、</b>                     |                                |                        |                                    |                             |
|                                                                  | -                            | ]                            |            |          |                    |                              |                                |                        |                                    |                             |
|                                                                  | -                            | 1                            | •          |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  |                              | 1                            |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            | 4                            |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | -                            | 4                            |            |          |                    |                              |                                |                        |                                    | -                           |
|                                                                  | <u> </u>                     |                              |            |          |                    |                              |                                |                        |                                    |                             |
| 1. FIG                                                           | URES I                       | N BLOW OR RE                 | COVERY CO  | LUMN OP  | POSITE             |                              |                                |                        |                                    |                             |
| A 2" OD SAMPLE SPOON 12" OF THE DISTANCE SHOWN                   |                              |                              |            |          |                    |                              |                                |                        |                                    |                             |
| FIG                                                              | URES SH<br>PERCEN            | HOWN OPPOSIT                 | E ROCK CON | RES DEN  | DTE                |                              |                                | <u> </u>               |                                    |                             |
| 2. <b>■</b> 2<br><b>▼</b> 6                                      | INDICAT                      | TES LOCATION                 | OF UNDIS   | TURBED   | SAMPLE.            |                              | REAV                           | BOI                    | PONTER STATION                     |                             |
|                                                                  | INDICAT<br>WITH NO<br>SCRIPT | NEXT TO GVM                  | BOL INDIA  | ING ATT  | SMPT<br>NDT P      | 3                            |                                | SHIPPING               | PORT, PENNSYLVA                    | UNII NU. ]                  |
| NUM<br>3• <del>∡</del>                                           | BER.<br>Indicat              | TES LOCATION                 | OF NATURA  | AL GROUN | H DE<br>ND WATER   | A 12/2                       |                                | DUQUES                 | NE LIGHT COMPAN                    | ſ                           |
| 4. <u>R</u> OD                                                   | TABLE.<br>- ROCK             | QUALITY DE                   | SIGNATION  | •        |                    | MUDIZZ                       | STONE &                        | WEBSTER                | ENGINEERING                        | CORPORATION                 |
| 2• ∐<br>6. DAT                                                   | INDICAT                      | ES DEPTH &<br>Mean sea levei | LENGTH OF  | NX CORI  | ING RUN.           | 14                           |                                | 117                    | 00 - GSK -144                      |                             |
|                        |                               |                                              |                                  |                                  | SH 1 OF 1                                                                                                                                         |
|------------------------|-------------------------------|----------------------------------------------|----------------------------------|----------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------|
| SITE BE                | AVER VAL                      | LET POWER STAT                               |                                  |                                  | J.O. NO. 11700 BORING NO. 5717                                                                                                                    |
| TYPE OF                | BORING                        | SPLIT SPOON<br>JUNE 13, 197                  | LOCATIO                          | N <u>Shippi</u><br>Dril          | LED BY ANTRICAN LOGGED BY DEP                                                                                                                     |
| SUMMAR                 | Y OF B                        | ORING                                        |                                  |                                  |                                                                                                                                                   |
|                        |                               | OVERALL                                      | SAMPLE                           | ျ<br>ျ                           | SOUL OF POCK DESCRIPTION                                                                                                                          |
| LEV.<br>EET            | EET                           | WEATHERING                                   | TPE WS                           | HAPH BO                          | SULL UN RUCK DESCRIPTION LITHOLOGY                                                                                                                |
|                        | Ğ "                           | 0 25 50 75 100                               |                                  | 8<br>9                           | OR JOINTING BEDDING AND FAULTING AND TEXTURE                                                                                                      |
| 657.1                  |                               |                                              |                                  |                                  | WETGET OF BODS SARK SPOOR                                                                                                                         |
|                        |                               |                                              | PUSH                             |                                  | TOP 11" SANDY SILT, MODERATELY PLASTIC, 10-15% VERY FINE SAND,<br>BLACK, (SN).<br>BOTTOM 7" SAND, UNIFORM, FINE, 5-10% MEDIUM SAND, 6-8% SLIGHTLY |
|                        | 5                             |                                              | 2 🏹                              | 2                                | PLASTIC FINES, DARE GRAY, (SP).                                                                                                                   |
| 650                    | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              | 37                               | 3                                | SAND, UNIFORM, FINE, 3-45 MEDIUM SAND, 5-8% SLIGHTLY PLASTIC FINES, -                                                                             |
|                        |                               |                                              |                                  |                                  | LIGHT GRAI WITH LESS THAN 5% GRAVEL TO 1.0 INCH MAXIMON, (SP).                                                                                    |
|                        |                               |                                              | 37                               |                                  | SANDI GRAVEL, POORLY GRADED TO 1.8 INCH MAXIMON, 8-10% FINE TO -                                                                                  |
|                        | 15                            |                                              |                                  |                                  | COARSE SAND, MOSTLY FINE, 4-8% FINES, DARK GRAY, (GP)                                                                                             |
| 640 —                  |                               |                                              | 34                               | -                                | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, MOSTLY MEDIUM AND                                                                                   |
|                        | 20                            | 4                                            |                                  |                                  | COARSE, 10-12% GRAVEL TO 0.9 INCH MAXIMUM, 6-8% FINES, LIGHT BROWN,                                                                               |
|                        | -                             |                                              |                                  |                                  | SAND, POORLY GRADED FINE TO COARSE MOSTLY MEDTING & LOS OPAVET                                                                                    |
|                        | 25 _                          | 4                                            |                                  | '                                | TO 0.8 INCH MAXIMUN, CLEAN, 1-25 FINES, LIGHT BROWN, (SP).                                                                                        |
| 630                    | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        | 30 -                          |                                              | 49                               | 7                                | SAND, FOOBLI GRADED, FINE TO COARSE, MOSTLY MEDIUM AND COARSE, 8-125                                                                              |
|                        |                               | -                                            |                                  |                                  | <br>                                                                                                                                              |
|                        | -                             | 4                                            |                                  |                                  | REFUSAL AT 32-6*                                                                                                                                  |
|                        | -                             | 4                                            |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        |                               |                                              |                                  |                                  | -                                                                                                                                                 |
|                        | -                             | 4                                            |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        |                               |                                              |                                  |                                  |                                                                                                                                                   |
|                        | -                             |                                              |                                  |                                  |                                                                                                                                                   |
|                        |                               |                                              |                                  |                                  |                                                                                                                                                   |
|                        | -                             | -                                            |                                  |                                  |                                                                                                                                                   |
|                        |                               |                                              |                                  |                                  | -                                                                                                                                                 |
|                        | -                             | 4                                            |                                  |                                  |                                                                                                                                                   |
|                        |                               | -                                            |                                  |                                  |                                                                                                                                                   |
|                        |                               |                                              |                                  |                                  |                                                                                                                                                   |
| 1. FIG                 | URES I                        | N BLOW OR RE<br>LE DENOTE TH                 | COVERY C                         | OLUMN OP<br>OF BLOW              | PPOSITE<br>VS OF A                                                                                                                                |
| 140<br>A 2<br>FIC      | ) LB HA<br>?" OD S<br>JURES S | MMER FALLING<br>AMPLE SPOON<br>HOWN OPPOSIT  | 30" REQ<br>12" or T<br>'E Rock o | UIRED TO<br>HE DISTA<br>ORES DEN | DRIVE<br>NCE SHOWN.                                                                                                                               |
| THE<br>2. 22           | PERCENTINDICA                 | NT OF CORE R<br>TES LOCATION<br>TES LOCATION | ECOVERED                         | STURBED                          | SAMPLE.                                                                                                                                           |
|                        | INDICA<br>WITH N              | TES LOCATION<br>O RECOVERY.                  | OF SAMP                          | LING ATT                         | EMPT 3 BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                                   |
| NUM<br>3• <del>≩</del> | IBER.                         | TES LOCATION                                 | OF NATU                          | CATES SA<br>RAL GROU             | ND WATER 2 DOQUESHE LIGHT COMPANY                                                                                                                 |
| 4. ROD<br>5. ∏         | INDICA                        | K QUALITY DE<br>TES DEPTH &                  | SIGNATIO<br>LENGTH O             | N.<br>F NX COR                   | ING RUN I                                                                                                                                         |
| 0. DA1                 | UM 13 1                       | MEAN SEA LEVEL                               |                                  |                                  | 11700 - USK - 145                                                                                                                                 |

•

|                            |                    |                              |                                        |                    | ČH 1 AF 1                                                                                                                              |
|----------------------------|--------------------|------------------------------|----------------------------------------|--------------------|----------------------------------------------------------------------------------------------------------------------------------------|
|                            |                    |                              | <u></u>                                | DIIQI              | HESNELIGHT COMPANY                                                                                                                     |
| SITE                       | BEAVER I           | ALLEY POWER ST               | MOITA                                  |                    | J.O. NO. 11700 BORING NO. 572 t                                                                                                        |
| TYPE OF                    | BORING             | S SPLIT SPOON                | LOCATION                               |                    | GROUND ELEV. 668.51                                                                                                                    |
| SUMMAR                     | YOFB               | ORING                        | •••••••••••••••••••••••••••••••••••••• |                    |                                                                                                                                        |
| . <del></del>              |                    |                              |                                        |                    |                                                                                                                                        |
|                            |                    | OVERALL                      | SAMPLE                                 | U I                |                                                                                                                                        |
| E Y.                       | РТН<br>ЕТ          | WEATHERING                   | S NIN CL                               | HI 9               | SOIL OR ROCK DESCRIPTION                                                                                                               |
| EL<br>FE                   | DEF<br>FE          | RQD<br>0 25 50 75 100        |                                        | RA                 | FIELD AND LABORATORY TEST REBULTS; SOIL STRATA DESCRIPTION; LITHOLOGY<br>OR JOINTING, BEDDING AND FAULTING AND TEXTURE                 |
|                            |                    |                              |                                        | ບ                  |                                                                                                                                        |
| 668.5                      |                    |                              |                                        |                    |                                                                                                                                        |
|                            |                    |                              |                                        |                    |                                                                                                                                        |
|                            |                    |                              |                                        |                    | -                                                                                                                                      |
|                            | - 5                |                              |                                        |                    | WEIGHT OF HAMMER AND RODS ADVANCED SPOON 15!                                                                                           |
|                            | -                  |                              |                                        |                    |                                                                                                                                        |
| 660                        |                    |                              |                                        |                    |                                                                                                                                        |
|                            | - 10 -             |                              |                                        |                    | -                                                                                                                                      |
|                            |                    |                              |                                        |                    |                                                                                                                                        |
|                            | _                  | 4                            |                                        |                    |                                                                                                                                        |
|                            | 15 -               | }                            |                                        |                    |                                                                                                                                        |
|                            |                    | 1                            | 22 <b>1</b>                            |                    | SANDI GRAVEL, POURLI GRADED TO 2.0 INCH MAXIMUM, 5-7% FINE AND<br>MEDIUM SAND, 8-10% FINES, DARK GRAY.                                 |
| <b>65</b> 0                | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | 20                 | 1                            | 36 🗾                                   |                    | SAND, UNIFORM, FINE, LAS GRAVEL TO 1 25 THEIR MATTHEM 4-1 OF                                                                           |
|                            |                    | 1                            |                                        |                    | SLIGHTLY PLASTIC FINES, DARK GRAY.                                                                                                     |
|                            |                    | 4                            |                                        |                    |                                                                                                                                        |
|                            | 25 —               |                              | 91                                     |                    | GRAVELLY SAND. UNIFORM. FINE. 4-6% MEDIUM AND COARSE SAND. 8-10%                                                                       |
|                            | - 1                |                              |                                        |                    | GRAVEL TO 1.5 INCH MAXIMUM, 4-6% FINES, DARK BROWN.                                                                                    |
| 640                        | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | 30<br>-            | 4                            | 58 🗾 4                                 |                    | SAND, UNIFORM, FINE, 3-7% MEDIUM AND COARSE SAND, 5-8% GRAVEL TO                                                                       |
|                            | -                  |                              |                                        |                    | 0.7 INCH MAXIMUM, 2-45 FINES, DARK BROWN.                                                                                              |
|                            | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | - «                | ]                            | 48 5                                   |                    | SAND, POORLY GRADED, FINE TO MEDIUM, CLEAN, LESS THAN 2% FINES,                                                                        |
| 4.0.5                      | -                  |                              |                                        |                    | LIUTI DRUWA,                                                                                                                           |
| 630                        | -<br>              |                              |                                        |                    |                                                                                                                                        |
|                            | -                  | }                            | 49 6                                   |                    | SAND, POORLY GRADED, VERY FINE TO COARSE, MOSTLY MEDIUM AND FINE,<br>3-5% GRAVEL TO 0.6 INCH MAXIMUM, LESS THAN 2% FINES, LIGHT BROWN. |
|                            | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | 45 -               | 1                            |                                        |                    | _                                                                                                                                      |
|                            |                    |                              | 30 7                                   |                    | SAND, UNIFORM, FINE, CLEAN, 1-2% FINES, LIGHT BROWN.                                                                                   |
| 620 —                      | -                  | •                            |                                        |                    | BOTTON OF BORING AT 46.5                                                                                                               |
|                            | 50 -               |                              |                                        |                    | REFUSAL AT 49.0'                                                                                                                       |
|                            |                    | ]                            |                                        |                    |                                                                                                                                        |
|                            | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | -                  | 1                            |                                        |                    | -                                                                                                                                      |
|                            | -                  | <u> </u>                     |                                        |                    |                                                                                                                                        |
|                            | -                  |                              |                                        |                    |                                                                                                                                        |
|                            | -                  | 1                            | 1                                      |                    |                                                                                                                                        |
|                            |                    |                              |                                        |                    |                                                                                                                                        |
|                            | -                  | 1                            |                                        |                    |                                                                                                                                        |
|                            | -                  | ]                            |                                        |                    |                                                                                                                                        |
|                            | -                  | 1                            |                                        |                    |                                                                                                                                        |
|                            | _                  | 1                            |                                        |                    |                                                                                                                                        |
|                            |                    |                              |                                        |                    |                                                                                                                                        |
| 1. FIG                     | JRES 11            | N BLOW OR RE                 | COVERY COL                             | UMN OPF            | POSITE                                                                                                                                 |
| SOI:<br>140                | L SAMPI<br>LB HAN  | LE DENOTE TH<br>IMER FALLING | E NUMBER O<br>30" REQUI                | F BLOWS<br>RED TO  | GOF A<br>DRI <b>VE</b>                                                                                                                 |
| A 2<br>Fig                 | " OD SA<br>JRES SH | AMPLE SPOON<br>HOWN OPPOSIT  | 12" OR THE<br>E ROCK COR               | DISTAN<br>ES DENO  | ICE SHOWN.                                                                                                                             |
| THE<br>2. <b>■2</b>        | PERCEN<br>INDICAT  | NT OF CORE R<br>TES LOCATION | ECOVERED.<br>OF UNDIST                 | URBED S            | BORING LOG 572t                                                                                                                        |
| <b>₹6</b><br>[  <b> </b> 7 | NDICAT             | TES LOCATION                 | OF SPLIT-<br>OF SAMPLI                 | SPOON S<br>NG ATTE | BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                                               |
| SUB                        | NITH NC<br>SCRIPT  | NEXT TO SYM                  | BOL INDICA                             | TES SAM            | IPLE SHIPPINGPORT, PENNSYLVANIA                                                                                                        |
| N∪N<br>3•葉                 | NDICAT             | ES LOCATION                  | OF NATURA                              | L GROUN            | DUQUENSE LIGHT COMPANY                                                                                                                 |
| 4. ROD                     | - ROCK             | QUALITY DE                   | SIGNATION.                             | NY 2007            | STONE & WEBSTER ENGINEERING CORPORATION                                                                                                |
| 6. DAT                     | JM IS N            | EAN SEA LEVEL                | DEMOTR OF ]                            | MA CORI            | NU RUN - USE - 140                                                                                                                     |

•••



| 630 ——                               |                                                                                                               | 48 7                                                                                             | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, MOSTLY MEDIUM AND<br>COARSE, 8-12% GRAVEL TO 0.7 INCH MAXIMUM, 3-5% FINES, DARK BROWN. |
|--------------------------------------|---------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------|
| (                                    | 45                                                                                                            | 61 8<br>100/5"                                                                                   | SAND, UNIFORM, FINE, CLEAN, LESS THAN 2% FINES, LIGHT BROWN.                                                                         |
| 629                                  | 50                                                                                                            |                                                                                                  | END OF BORING AT 48.51                                                                                                               |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
|                                      |                                                                                                               |                                                                                                  |                                                                                                                                      |
| 1. FIG<br>SOII<br>140<br>A 2'<br>FIG | URES IN BLOW OR REG<br>L SAMPLE DENOTE THI<br>LB HAMMER FALLING<br>" OD SAMPLE SPOON 1<br>JRES SHOWN OPPOSITE | COVERY COLUMN OP<br>S NUMBER OF BLOW<br>30" REQUIRED TO<br>2" OR THE DISTAN<br>B BOCK CORES DENN | POSITE<br>S OF A<br>DRIVE<br>NCE SHOWN.                                                                                              |
| THE<br>2. ■21<br>▼61                 | PERCENT OF CORE RE<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION                          | COVERED.<br>OF UNDISTURBED (<br>OF SPLIT-SPOON (<br>OF SAMPLING ATT)                             | BORING LOG 573t<br>BEAVER VALLEY POWER STATION - UNIT NO. 1                                                                          |
|                                      | SCRIPT NEXT TO SYME<br>BER.                                                                                   | OL INDICATES SAN                                                                                 | MPLE SHIPPINGPORT, PENNSYLVANIA                                                                                                      |
| ·· ÷ 1<br>4. <u>R</u> QD             | ABLE.<br>- ROCK QUALITY DES                                                                                   | OF NATURAL GROUD                                                                                 | MATER Z STONE & WEBSTER ENGINEERING CORPORATION                                                                                      |
| 5• ∐ 1<br>6. DAT                     | INDICATES DEPTH & L<br>JM IS MEAN SEA LEVEL                                                                   | ENGTH OF NX COR                                                                                  | ING RUN                                                                                                                              |

and a straight



| 630                            |                                                               |                                                                                               |                                                                 |                                                  | 0.5 INCH MAXIMUM, 3              | -5% FINES, LIGHT BROWN.                                      |
|--------------------------------|---------------------------------------------------------------|-----------------------------------------------------------------------------------------------|-----------------------------------------------------------------|--------------------------------------------------|----------------------------------|--------------------------------------------------------------|
|                                | 40<br><br>                                                    |                                                                                               | 41 6                                                            |                                                  | SAND, UNIFORM, FINE              | , 3-4% FINES, LIGHT BROWN.                                   |
|                                | 45                                                            |                                                                                               | 30 <b>7</b><br>100/1*                                           |                                                  | SAND, UNIFORM, FINE              | , CLEAN, LESS THAN 2% FINES, LIGHT BROWN.                    |
| 620                            |                                                               |                                                                                               |                                                                 |                                                  | ERD OF BORING AT                 | 47.5 <sup>1</sup>                                            |
|                                | -<br>-<br>-<br>-                                              |                                                                                               |                                                                 |                                                  |                                  |                                                              |
|                                | -                                                             |                                                                                               |                                                                 |                                                  |                                  | -<br>-<br>                                                   |
|                                | -<br>-<br>-                                                   |                                                                                               |                                                                 |                                                  |                                  |                                                              |
|                                |                                                               |                                                                                               |                                                                 |                                                  |                                  | <br><br>                                                     |
| 1. FIGU<br>SOII<br>140<br>A 2' | IRES IN<br>L SAMPI<br>LB HAM<br>' OD SA                       | BLOW OR REC<br>E DENOTE THE<br>MER FALLING<br>MPLE SPOON 1                                    | VERY COLU<br>NUMBER OF<br>30" REQUIE<br>2" OR THE               | JMN OPPO<br>BLOWS<br>RED TO D<br>DISTANC         | SITE<br>DF A<br>RIVE<br>E SHOWN. |                                                              |
|                                | JRES SH<br>PERCEN<br>INDICAT<br>INDICAT<br>INDICAT<br>VITH NO | OWN OPPOSITE<br>T OF CORE RE(<br>ES LOCATION (<br>ES LOCATION (<br>ES LOCATION (<br>RECOVERY. | ROCK CORE<br>COVERED.<br>OF UNDISTU<br>OF SPLIT-S<br>OF SAMPLIN | IS DENOT:<br>JRBED SAI<br>Spoon Sai<br>Ig Attemi | \$<br>4PLE.<br>4<br>PLE.<br>3    | BORTING LOG 574t<br>BEAVER VALLEY POWER STATION - UNIT NO. 1 |
| SUBS<br>NUME<br>3•葉Ⅱ           | SCRIPT<br>SER.<br>INDICAT                                     | NEXT TO SYMBO<br>ES LOCATION (                                                                | DL INDICAT<br>DF NATURAL                                        | ES SAMPI                                         | E M 46/5Z<br>WATEF 2             | SHIPPINGPORT, PENNSYLVANIA<br>DUQUENSE LIGHT COMPANY         |
| 4. ROD<br>5. ∏ I<br>6. DATU    | - ROCK<br>NDICAT<br>JM IS 1                                   | QUALITY DESI<br>ES DEPTH & LH<br>ÆAN SEA LEVEL                                                | GNATION.<br>ENGTH OF N                                          | X CORING                                         | RUN . 1                          | TONE & WEBSTER ENGINEERING CORPORATION                       |

4

And the second second

and the second

the second of the second



|                             |                                                                                                                                |                                                                                      | BROWN, WITH LESS THAN 5% GRAVEL TO 0.5 INCH MAXIMUM                                                 |
|-----------------------------|--------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|
| 630                         | <br>40<br><br>                                                                                                                 | 68 5                                                                                 | BAND, WELL GRADED, FINE TO MEDIUM, 1-3% FINES, LIGHT BROWN, WITH                                    |
|                             | 45                                                                                                                             | 80 6                                                                                 | BAND, UNIFORM, FINE, 3-5% MEDIUM AND COARSE, 4-7% GRAVEL TO 0.5 INCH<br>MAXIMUM, 2-3% FINES. BROWN. |
|                             |                                                                                                                                |                                                                                      | END OF BORING 47.5'                                                                                 |
|                             |                                                                                                                                |                                                                                      |                                                                                                     |
|                             |                                                                                                                                |                                                                                      |                                                                                                     |
|                             |                                                                                                                                |                                                                                      |                                                                                                     |
|                             |                                                                                                                                |                                                                                      |                                                                                                     |
|                             |                                                                                                                                |                                                                                      |                                                                                                     |
| 1. FIG<br>SOI<br>140<br>A 2 | URES IN BLOW OR RE<br>L SAMPLE DENOTE TH<br>LB HAMMER FALLING<br>" OD SAMPLE SPOON                                             | COVERY COLUMN OP<br>E NUMBER OF BLOW<br>30" REQUIRED TO<br>12" OR THE DISTA          | POSITE<br>S OF A<br>DRIVE<br>NCE SHOWN.                                                             |
| FIG<br>THE<br>2. ■2<br>76   | DRES SHOWN OPPOSIT<br>PERCENT OF CORE R<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION<br>WITH NO RECOVERY. | E ROCK CORES DEN<br>ECOVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON<br>OF SAMPLING ATT | OTE<br>SAMPLE.<br>SAMPLE.<br>EMPT BEAVER VALLEY POWER STATION - UNIT NO. 1                          |
| SUB<br>NUM<br>3• ¥          | SCRIPT NEXT TO SYM<br>BER.<br>INDICATES LOCATION<br>FABLE.                                                                     | BOL INDICATES SAN<br>OF NATURAL GROUN                                                | MPLE SHIPPINGPORT, PENNSYLVANIA<br>ND WATEF 2 DUQUESNE LIGHT COMPANY<br>TONIC CHERCICE ENGINE       |
| 4. ROD<br>5. ∐.<br>6. DAT   | - ROCK QUALITY DES<br>INDICATES DEPTH & I<br>UM IS MEAN SEA LEVEN                                                              | SIGNATION.<br>JENGTH OF NX COR                                                       | ING RUN I III IIII IIIII IIIIIIIIIIIIIIIIII                                                         |



)

ć.`}

| 630 —                                  | <br>-<br>40                                                                                                                                                    | 42 5                                                                                      | GRAVELLY SAND, POORLY GRADED, FINE TO COARSE, 10-12% GRAVEL TO 0.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
|----------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                        |                                                                                                                                                                |                                                                                           | INCH MAXIMUM, 3-5% SLIGHTLY PLASTIC FINES, LIGHT BROWN.<br>(SP)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
|                                        | 45 <u>-</u><br>_<br>_                                                                                                                                          | 64                                                                                        | SAND, UNIFORM FINE, 4-8% GRAVEL TO 0.75 INCH MAXIMUM, CLEAN, LESS<br>THAN 2% FINES, LIGHT BROWN.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
|                                        |                                                                                                                                                                |                                                                                           | END OF BORING AT 48.0'                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|                                        |                                                                                                                                                                |                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |
| 1. FIGU<br>SOII<br>140<br>A 2'<br>FIGU | JRES IN BLOW OR REC<br>L SAMPLE DENOTE THE<br>LB HAMMER FALLING<br>' OD SAMPLE SPOON 1<br>JRES SHOWN OPPOSITE                                                  | OVERY COLUMN OF<br>NUMBER OF BLOW<br>30" REQUIRED TO<br>2" OR THE DISTA<br>ROCK CORES DEN | PPOSITE<br>WS OF A<br>D DRIVE<br>ANCE SHOWN.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|                                        | PERCENT OF CORE RE<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION<br>WITH NO RECOVERY.<br>SCRIPT NEXT TO SYMB<br>BER.<br>INDICATES LOCATION | COVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON<br>OF SAMPLING ATT<br>OL INDICATES SA        | SAMPLE.<br>SAMPLE.<br>MEMPT<br>MPLE<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER<br>MATER |
| 4. ROD<br>5. 11. 1<br>6. DAT(          | ABLE.<br>- ROCK QUALITY DES<br>INDICATES DEPTH & L<br>JM IS MEAN SEA LEVEL                                                                                     | IGNATION.<br>ENGTH OF NX COR                                                              | TING RUN AND THE STONE & WEBSTER ENGINEERING CORPORATION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |

SH1\_OF1\_ DUQUESNE LIGHT COMPANY BEAVER VALLEY POWER STATION J.O. NO. 11700 BORING NO. 577 t. SITE \_ \_ GROUND ELEV. \_669.01 TYPE OF BORINGSPLIT SPOON LOCATION DATE DRILLED \_\_\_\_\_\_\_ 19, 1974\_\_\_ \_ DRILLED BY \_AMERICAN \_\_\_\_ LOGGED BY D.F.P. SUMMARY OF BORING \_ G R APHIC LOG OVERALL SAMPLE SOIL OR ROCK DESCRIPTION DEPTH FEET WEATHERING ELEV. FEET BLOWS RECOV. TYPE RQD FIELD AND LABORATORY TEST RESULTS; OR JOINTING SEDDING AND FAULTING DESCRIPTIONS SOIL STRATA DESCRIPTION; LITHOLOGY AND TEXTURE 0 25 50 75 100 669.0 5 660 10-15 650 20 23 1 SANDY GRAVEL, POORLY GRADED TO 1.7 INCH MAXIMUM, 6-10% FINE SAND, 5-8% SLIGHTLY PLASTIC FINES, LIGHT GRAY. (GP) 25\* 53 2 SAND, UNIFORM, FINE, CLEAN, 1-2% FINES LIGHT BROWN. (SP) 640 30 SANDY GRAVEL, POORLY GRADED TO 1.5 INCH MAXIMUM, 10-15% FINE TO COARSE SAND, 3-5% SLIGHTLY PLASTIC FINES, LIGHT BROWN. 53 3 (GP) 35' TINT PODY FT NE. CLEAN.

)

| 630                                 |                                                                                                             | 091 4                                                                                | 1-2% FINES, LIGHT BROWN.                                                                                                     |
|-------------------------------------|-------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|
|                                     | 40<br>-<br>-<br>-                                                                                           | 65 5                                                                                 | SAND, UNIFORM, FINE 3-5% MEDIUM AND COARSE SAND, 2-4% GRAVEL TO<br>0.6 INCH MAXIMUM, CLEAN, 1-2% FINES, LIGHT BROWN.<br>(SP) |
| 620                                 | 45                                                                                                          | 91 6                                                                                 | SAND, UNIFORM, FINE; 2-4% MEDIUM SAND, CLEAN 1-2% FINES, LIGHT<br>BROWN.<br>(SP)                                             |
|                                     | >0                                                                                                          |                                                                                      | END OF BORING AT 49.5'                                                                                                       |
|                                     |                                                                                                             |                                                                                      |                                                                                                                              |
|                                     |                                                                                                             |                                                                                      |                                                                                                                              |
|                                     |                                                                                                             |                                                                                      |                                                                                                                              |
|                                     |                                                                                                             |                                                                                      |                                                                                                                              |
| 1. FIG<br>SOII<br>140<br>A 2'       | JRES IN BLOW OR REG<br>SAMPLE DENOTE THI<br>LB HAMMER FALLING<br>OD SAMPLE SPOON                            | COVERY COLUMN OF<br>E NUMBER OF BLOW<br>30" REQUIRED TO<br>12" OR THE DISTA          | POSITE<br>/S OF A<br>DRIVE<br>NCE SHOWN.                                                                                     |
| FIGU<br>THE<br>2. 21<br>761<br>[]7] | JRES SHOWN OPPOSITE<br>PERCENT OF CORE RI<br>INDICATES LOCATION<br>INDICATES LOCATION<br>INDICATES LOCATION | E ROCK CORES DEN<br>SCOVERED.<br>OF UNDISTURBED<br>OF SPLIT-SPOON<br>OF SAMPLING ATT | OTE<br>SAMPLE.<br>SAMPLE.<br>EMPT<br>BEAVER VALLEY POWER STATION - UNIT NO. 1                                                |
| SUBS<br>NUME<br>3. ¥ I<br>1         | CRIPT NO RECOVERY.<br>CRIPT NEXT TO SYME<br>DER.<br>NDICATES LOCATION<br>ABLE.<br>- ROCK QUALTY DEC         | BOL INDICATES SA<br>OF NATURAL GROU                                                  | MPLE SHIPPINGPORT, PENNSYLVANIA<br>DUQUESNE LIGHT COMPANY<br>STONE & WEBSTER ENGINEERING CORPORATION                         |
| 5. 11. 1<br>6. DATU                 | NDICATES DEPTH & I<br>IM IS MEAN SEA LEVEL                                                                  | ENGTH OF NX COR                                                                      | ING RUN                                                                                                                      |

|              |                 | 7.5437           | 20 174             | 1172 500        | ዋር ሰጥ አጥ          | 101               | BORING NO. SEO-1                                                                                                                                      |
|--------------|-----------------|------------------|--------------------|-----------------|-------------------|-------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|
|              |                 |                  | <u>ск ул</u><br>Ге | N3724-5         | <u></u>           | <u>iva -</u><br>P | 9230.0 GROUND FLEV (1) 727.3 SHEETOF                                                                                                                  |
|              |                 |                  | 50 .<br>1          |                 |                   | <br>Al            |                                                                                                                                                       |
|              |                 | : STAI           | RT/1               | FINISH          | 10/6/             | <u>81</u>         | / 10/8/81 CONTRACTOR / ORLLER EGER DRILLING/JARVIS                                                                                                    |
|              |                 | C GRO            | DUND               | WATER           | DEPT              | н70               | ATE 52.2 (FT) / 10/9/81 DRILL RIG TYPE                                                                                                                |
|              | EPT             | н то             | BED                | ROCK _          | 104.3             |                   | (FT) TOTAL DEPTH DRILLED 104.5 (FT)                                                                                                                   |
|              | ETH             | 00S:             |                    |                 |                   |                   |                                                                                                                                                       |
|              |                 | DRILLI           | NG                 | SOIL            | AH ROD            | 5, 3              | IN RULLER BIT, DRILLING MUD AND CASING                                                                                                                |
| 1            |                 | SAMPL            | . ING              | SOIL.           | 2.0 IN            | 0.0               | SPLIT BARREL                                                                                                                                          |
|              |                 | DRILL            | ING                | ROCK            | <u>N/A</u>        |                   |                                                                                                                                                       |
|              | PECI            | AL TE            | ESTI               | NG OR           | INSTR             | UME               | ITATION                                                                                                                                               |
|              |                 |                  |                    |                 |                   |                   |                                                                                                                                                       |
|              | :OMM            | IENTS            | GROUT              | NDWATER         | AT 51.3           | TT C              | N 10/12/81                                                                                                                                            |
|              |                 |                  | FILL               | TO APPR         | JAIMATE           | LI 43             |                                                                                                                                                       |
|              |                 |                  |                    |                 |                   |                   |                                                                                                                                                       |
| I I          | - :             | 3                |                    | 10              | i a               | 9                 |                                                                                                                                                       |
| ATI0<br>ET)( | PTH             |                  |                    | S a             |                   | 12 g              |                                                                                                                                                       |
| LEV<br>F     |                 | E IN F           | N S                | BLO             |                   | 18 X              |                                                                                                                                                       |
| . ш <b>і</b> |                 |                  |                    | <u> </u>        |                   |                   |                                                                                                                                                       |
|              |                 |                  |                    |                 |                   |                   |                                                                                                                                                       |
| 727.3        |                 |                  | 1,                 | 37-31-3         | 1 52              | Gu                | CANNY CRAUPT FOR POINTED TO ANOTH AR CANDETONE AND CTUTEOUT TRANSFORME                                                                                |
|              |                 | Ē                | 1                  | (10")           | 1 52              |                   | TO 1 IN, 40% COARSE TO FINE SAND, LESS THAN 5% NONPLASTIC FINES, SROWN.                                                                               |
|              | 5               | - S              | ] 2                | (3")            | 16                | SP                | GRAVELLY SAND, 302 ROUNDED TO SUBANGULAR CRAVEL TO 3/4 IN, LARGE 14 IN<br>GRAVEL AT BOTTOM, 60-702 FINE TO MEDIUM SAND, MOSTLY FINE, 52               |
|              | -               | -                | ]                  |                 |                   |                   | NONPLASTIC FINES, BROWN.                                                                                                                              |
| 720.0        |                 |                  | 3                  | 7-6-5<br>(4½")  | 11                | GP                | SANDY GRAVEL, MOSTLY MEDIUM TO FINE GRAVEL, 30-407 COARSE TO FINE SAND,<br>MOSTLY MEDIUM TO FINE, FEW FRAGMENTS TO 1 IN, 10% NOMPLASTIC FINES, BROWN. |
|              | 10              | -1-3-            | 4                  | 3-3-5           | 8                 | GP                | TOP 3 IN - SAME AS ABOVE.                                                                                                                             |
|              | 10              | 1                | 1                  |                 |                   | <sup>3</sup>      | 2 IN - FINE SAND, 5-107 NONPLASTIC FINES, RUSTY BROWN.                                                                                                |
|              |                 |                  | 5                  | 8-4-5           | 9                 | ମ୍ୟ               | SILTY SANDY GRAVEL, WELL-GRADED COARSE TO FINE GRAVEL, ROUNDED TO                                                                                     |
|              |                 |                  | 6                  | 3-1-1           | 2                 |                   | SUBANGULAR, FEW LARGE GRAVEL SIZES, 30% COARSE TO FINE SAND, MOSTLY<br>MEDIUM TO FINE, 15% NONPLASTIC FINES, BROWN.                                   |
|              | 15              |                  |                    | (0")            |                   | 5P                | BOTTOM 2 IN - FINE 5AND, LESS THAN 52 NONPLASTIC FINES, TRACE OF FINE - GRAVEL, WOOD FRAGMENTS NEAR BOTTON, RUSTY BROWN.                              |
| 710.0        |                 |                  | ,                  | WOH             |                   | SP                | GRAVELLY SAND 307 COARSE TO FINE GRAVEL TO 3/4 TH COARSE TO FINE SAND                                                                                 |
| 1010         |                 |                  | <u> </u>           | ()")            |                   | <b>_</b>          | MOSTLY MEDIUM TO FINE, 10X NONPLASTIC FINES, BROWN. (WASH?).                                                                                          |
|              | 20              | s                | 8                  | 3-0-1<br>(0")   | 1                 |                   |                                                                                                                                                       |
|              |                 | 3                | 9                  | 5-4-5<br>(14")  | 9                 | SP                | FINE SAND, 5+102 NONPLASTIC FINES, TRACE OF FINE GRAVEL, BROWN.                                                                                       |
|              |                 | - <u>s</u>       | 10                 | 5-3-4<br>(4")   | . 7               | SP                | SAME AS ABOVE.                                                                                                                                        |
|              | 25              | <b>-</b>   5     | 11                 | 6-6-5           | 11                | SM                | SILIT SAND, 10-15% FINE TO COARSE GRAVEL, MOSTLY FINE, MEDIUM TO FINE SAND -<br>10-15% NONPLASTIC FINES, TRACE OF BLACK ORGANICS AT BOTTOM            |
|              |                 |                  | 12                 | (9")<br>}2-10-7 | 17                | SP<br>GM          | TOP 6 IN, <u>FINE SAND</u> , LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BOTTOM: GRAVELLY SILT, 20-30% FINE TO MEDIUM GRAVEL SIZED SANDSTONE             |
| 700.0        |                 |                  | 1                  | (12")           |                   |                   | FRAGMENTS, 10% PLASTIC FINES, BROWN.                                                                                                                  |
|              | 30              | - S              | 13                 | 6-7-7<br>(10")  | 14                | SM                | SILTY SAND, 10% COARSE TO FINE WEATHERED ROCK FRAGMENTS AND GRAVEL, 10-15% -<br>PLASTIC FINES, COARSE TO FINE SAND.                                   |
| 1.           | DATU            | M IS M           |                    | SEA LEV         | EL                | 7.                | S-SPLIT BARREL SAMPLE                                                                                                                                 |
| 2.<br>3.     | ¥ un<br>BLOW    | S REQI           | WATE               | TO DAL          | VE.               |                   | BORING LOG                                                                                                                                            |
| ES           | 2"0.D.<br>DISTA | SAMPL            | .E SP<br>Howm      | OON 6"          | OR                |                   | BEAVER VALLEY POWER STATION                                                                                                                           |
| 10           | 140%            | HANNE            | RFA                | LLING 3         | 0 <sup>11</sup> . |                   | UNIT 2                                                                                                                                                |
| <u> </u>     | HANN            | IER. (           | ) IN(              | HES OF          | / ·O.             |                   | DUQUESNE LIGHT COMPANY                                                                                                                                |
| a 4.         | SAMP<br>% R     | LE RE(<br>DCK CO | COVE!              | RY.<br>IECOVERY | 1                 |                   | SHIPPINGPORT, PENNSYLVANIA                                                                                                                            |
| NU 5.        | ROCK<br>STD.    | QUALI            | TY C<br>Rati       | ESIGNAT         | ION.              |                   | STONE & WEBSTER ENG. CORP.                                                                                                                            |
| Ш.           | RESIS           |                  |                    | NS/FT.          | TICH              |                   | APPROVED DATE ROUMA NO SHEFT                                                                                                                          |
| · ·          | SYST            | EN.              |                    | HUGH' IV        |                   |                   | Dide / Jake M SED-1 1 OF 3                                                                                                                            |

|                          |             |                  |                       |                       |                  |                    |                     |                                                                                                                                                         |                                   |                                             | BORING N                                     | 0. 980-1              |
|--------------------------|-------------|------------------|-----------------------|-----------------------|------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------|---------------------------------------------|----------------------------------------------|-----------------------|
|                          |             |                  |                       |                       |                  |                    |                     |                                                                                                                                                         |                                   |                                             | SHEET 2                                      | OF                    |
| 5                        | BITE        | BEAVE            | R VALI                | LEY POW               | ER S'            | TATIO              | <u>v - v</u>        | NIT 2                                                                                                                                                   | J.Q.                              | NO. 12241                                   |                                              |                       |
| ELEVATION<br>(FEET)()62) | DEPTH       | (FEET)<br>SAMPLE | SAMPLE                | NUMBER<br>BLOWS (3)   | REC/RQD (4)      | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPL                                                                                                                                                   | LE DESCI                          | RIPTION                                     |                                              |                       |
|                          |             |                  |                       |                       | -                |                    |                     |                                                                                                                                                         |                                   |                                             |                                              |                       |
|                          |             |                  | ; I/                  | 7-9-<br>(10"          | 6<br>)           | 17                 | SM                  | <u>SILTY SAND, 10-157 MEDIUM TO</u><br>Shale Fragments, coarse to Fi                                                                                    | FINE GRAVEL<br>NE SAND, 10        | . SIZED WEAT                                | HERED SANDST(<br>Ly plastic f)               | INE AND               |
|                          |             | _                | - 1 I                 | s 11-1<br>(10"        | 1 <b>-1</b> 0    | 21                 | SM                  | SAME AS ABOVE, FRAGMENTS, BRO                                                                                                                           | WN, RUST, G                       | RAY.                                        |                                              | -                     |
| 690.0                    |             |                  |                       | 5 <b>15-1</b><br>(14" | ,<br>5-16<br>)   | 31                 | SM                  | SIMILAR TO ABOVE, SILTY SAND,<br>ROCK FRACMENTS, COARSE TO FIN                                                                                          | 10-15% COA                        | RSE TO FINE                                 | GRAVEL SIZEN<br>IO FINE, 15-2                | ) TO 1 IN             |
|                          | 60          |                  | ; 17                  | 1 17-1                | 8-14             | 32                 | SM                  | SAME AS ABOVE.                                                                                                                                          | t fines, um                       | ALL AND BROWN                               | n,                                           | _                     |
|                          |             |                  | -<br>-<br>-           | 3 8-8-<br>(14"        | )<br>7 .<br>)    | 15                 | SM                  | TOP 6 IN, SAME AS ABOVE.<br>BOTTOM 8 IN, SILTY SAND, COARL                                                                                              | SE TO FINE                        | SAND, MOSTL'                                | COARSE TO B                                  | ædium, =              |
|                          | 45          |                  | -<br>-<br>-<br>-<br>- | ) 5-10<br>(18"        | <b>-12</b><br>}  | 22                 | CL                  | TOP 5 IN, SAME AS ABOVE,<br>BOTTOM 13 IN, <u>STIFF SILTY CLA</u><br>FRAGMENTS. TRACE OF ROOTS. GR                                                       | <u>NY</u> , MODERATE              | LY PLASTIC,<br>(1.75 taf                    | TRACE OF ROO                                 |                       |
| 680.0                    |             |                  | ;2(                   | ) 15-2<br>(14"        | 2-19<br>)<br>) 2 | 41                 | CL<br>SM            | TOP 6 IN, SAME AS ABOVE.<br>MIDDLE 2 IN, LARGE SANDSTONE :<br>BOTTOM 6 IN, <u>SILTY SAND</u> , WITH                                                     | FRAGMENTS W                       | TTH COARSE SANDSTONE A                      | ,<br>To Medium Sai<br>ND Shale, Co/          | TD, BROWN             |
|                          | 50          |                  | <u>'</u>              | (8")                  | 13               | 20                 | CL                  | FINE GRAVEL SIZED, 102 SLICH<br>VERY STIFF <u>SILTY CLAY</u> , MODERA                                                                                   | TLY PLASTIC                       | C, 20-30Z S                                 | Y, OIL SMELL<br>ILT, MOTTLED                 | BROWN AND             |
|                          |             |                  |                       | ! 18-1<br>(9")        | 1-13             | 24                 | CL<br>SM            | GRAY, TRACE OF SHALE FRAGMENT<br>TOP 4 <sup>1</sup> / <sub>2</sub> in, similar to above, i<br>middle 1 <sup>1</sup> / <sub>4</sub> in, silty sand, with | NORE WEATHE<br>HORE WEATHE        | VEL SIZED, "<br>RED SHALE :<br>FRAGMENTS TO | TRACE OF OIL<br>Fragments.<br>D 3/4 in, gr/  | SMELL.                |
|                          | 55          |                  |                       | (5")                  | 1~11<br>7=16     | 22                 | SM                  | BOTTON 3 IN, SANDSTONE FRACME<br>SANDY SILT, 10-202 MEDIUM TO<br>SANDSTONE AND SHALE FRACMENTS                                                          | FINE SAND,1                       | FRACHENTS TO<br>5-20% SLIGHT                | D 14IN, GRAY.<br>Ly plastic f<br>large sands | INES , 10-15%         |
| 670.0                    |             |                  | 22                    | (14"<br>5 10-9        | -11              | 20                 | SM                  | FRAGMENT IN BOTTOM.                                                                                                                                     | D SHALE PRA                       | CHENTS TO 1                                 | IN, 10-152 S                                 | LIGHTLY               |
|                          | 60          | +-               | 1.                    | (12")                 |                  |                    |                     | MOTTLED BROWN AND GRAY.                                                                                                                                 | BO DOP OT                         | E OF HILR,                                  | TRACE OF BLAC                                | K URGANIC <u>S.</u>   |
|                          |             |                  |                       | (9")                  | ,<br>            | 15                 | SM                  | SILTY FINE SAND, 20-30% NONPL.                                                                                                                          | ASTIC FINES                       | , ZONES OF                                  | VEATHERED SAI                                | idstone -             |
|                          | 65          | ╉                |                       | (12")                 | >                | 20                 | SM                  | AND SHALE FRAGENTS, TRACE OF<br>SIMILAR TO ABOVE, 10-157 NONP<br>FINE SAND. ZONES OF WEATHERED                                                          | LASTIC TO S                       | N.<br>LIGHTLY PLA:<br>AND SHALE FI          | STIC FINES, (<br>RACMENTS.BROW               | COARSE TO             |
| 660.0                    |             |                  | 28                    | ) 15-1<br>(12"        | 5-10<br>)        | 26                 | SM                  | TOP 7 IN, SIMILAR TO ABOVE, 1<br>MIDOLE 3 IN, SILTY SAND, COARS                                                                                         | 0-201 NONPL<br>SE TO FINE         | ASTIC SLIGH<br>SAND, MOSTLY                 | TLY PLASTIC I<br>COARSE TO P                 | INES, BROWN-          |
|                          | 70          | 4                | 29                    | ) 10-1<br>(13"        | 4-14<br>)        | 28                 |                     | 10-15% NONPLASTIC FINES, TRAC<br>ORANGE,<br>BOTTOM 2 IN. SAME AS TOP.                                                                                   | E OF FINE G                       | RAVEL SIZE S                                | SANDSTONE FR                                 | AGMENTS, _            |
|                          |             |                  |                       |                       |                  |                    | un.                 | GRAVELLY SAND, 20-30% COARSE SHALE FRAGMENTS TO 1 IN, COAL<br>BROWN.                                                                                    | TO FINE GRA<br>RSE TO FINE        | VEL, ROUNDE<br>SAND, 5-10                   | D TO SUBANGUI<br>I NONPLASTIC                | AR, FEW FINES,        |
|                          | 75          |                  | 30                    | ) 9-8-<br>(14"        | 17               | 25                 | SM                  | SILTY SAND, WELL GRADED, COARS:<br>TRACE OF FINE GRAVEL, BROWN.                                                                                         | E TO FINE S                       | AND, 5-15%                                  | NONFLASTIC F                                 | INES,                 |
| 650.0                    |             | 4                |                       |                       |                  |                    |                     |                                                                                                                                                         |                                   |                                             |                                              |                       |
|                          | 80          |                  | 31                    | 13-1<br>(14"          | 5-14             | 29                 | SM                  | TOP 4 IN, SAME AS ABOVE.<br>MIDDLE 1 IN, <u>SILTY FINE SAND</u> ,<br>DARK BROWN.                                                                        | 10-15Z NON                        | PLASTIC FIN                                 | ES, TRACE OF                                 | <br>ЖІСА, –           |
|                          | 85          | <u> </u>         |                       |                       |                  |                    |                     | COARSE TO FINE SAND, MOSTLY MU<br>PLASTIC FINES.                                                                                                        | EDIUM TO FI                       | NE, 15-20X I                                | NONPLASTIC TO                                | SLICHTLY              |
| 640.0                    |             |                  | 32                    | : 19-2(<br>(12'')     | )-21             |                    | SM                  | SILTY SAND, SIMILAR TO ABOVE,<br>MENTS, ALL COLORS, 15-202 NONI<br>FINE SAND, MOSTLY MEDIUM TO F                                                        | 20-307 WEA<br>Plastic to<br>'Ine. | THERED SAND:<br>SLIGHTLY PL                 | STONE AND SHA                                | LE FRAG-<br>COARSE TO |
| NOTE: /                  | 90<br>For I |                  | SUMO                  | ARY A                 | ND               |                    | STON                | E & WEBSTER ENG. CORP.                                                                                                                                  | PPROVED                           | DATE                                        | BORING NO.                                   | SHEET                 |
| L                        | DGEN        | d NFQ            | SEE                   | SHEET                 | . I              |                    | SKET                | CH No. 12241-GSK-234B                                                                                                                                   | ROX/                              | 1/12/82                                     | SEO-1                                        | 2 OF 3                |

|                        |                  |          |                  |                              |                    |                    | BORING NOSZO                                                                                                                                                                                                                                          | 1 |
|------------------------|------------------|----------|------------------|------------------------------|--------------------|--------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|
| _                      |                  | 96.      | \VFP '           | ALLEY DOL                    | ER CT              | ATTOM              | SHEET _ 3 OF _ 3                                                                                                                                                                                                                                      | _ |
| <del>য</del><br>ম      | ие <u>—</u><br>Т |          |                  |                              | <u> </u>           |                    |                                                                                                                                                                                                                                                       |   |
| ELEVATION<br>(FEET)(IG | DEPTH<br>(FEET)  | SAMPLE   | SAMPLE<br>NUMBER | BLOWS (3<br>OR<br>REC/ROD (4 | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6 | SAMPLE DESCRIPTION                                                                                                                                                                                                                                    |   |
|                        |                  |          |                  |                              | <b></b> ,          | ·1                 |                                                                                                                                                                                                                                                       |   |
|                        |                  |          | 33               | A-15-14<br>(10")             | 29                 |                    | GRAVELLY SAND, 10-20% COARSE TO FINE GRAVEL, ROUNDED TO ANGULAR, COARSE<br>TO FINE SAND, 5-10% NONPLASTIC FINES, FEW FRAGMENTS WEATHERED SANDSTONE<br>AND SHALE FRAGMENTS.<br>AT 4 IN, SEAH OF DARK BROWN SILTY FINE SAND, TRACE OF MICA (1 IN THICK) |   |
| 630.N                  | 95 -             |          | 34               | 11-15-14<br>(10")            | 29                 |                    | TOP 5 IN, SAME AS ABOVE (NO SILTY SAND SEAM)<br>BOTTOM 5 IN, <u>SILTY FINE SAND</u> , 10-15% NONPLASTIC FINES, LIGHT BROWN.                                                                                                                           |   |
|                        | 100 -            | -<br>- s | 35               | 11-12-15                     | 27                 |                    | SAND, LESS THAN 5% FINE GRAVEL, COARSE TO FINE SAND, MOSTLY COARSE TO                                                                                                                                                                                 |   |
| 622.8                  |                  |          | 36               | (10")<br>105/.21<br>(0")     |                    |                    | MEDIUM, 5-7% NONPLASTIC FINES, BROWN.                                                                                                                                                                                                                 |   |
|                        | 105 •            |          |                  |                              |                    |                    | END OF BORING AT 104.5 FT.                                                                                                                                                                                                                            |   |
|                        |                  |          |                  | 1                            |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        | -                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        | .                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       | - |
|                        |                  |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       | - |
|                        | -                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        | -                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        | .                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       | - |
|                        |                  |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        |                  |          | -                |                              |                    |                    |                                                                                                                                                                                                                                                       |   |
|                        | -                |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       | - |
|                        |                  |          |                  |                              |                    |                    |                                                                                                                                                                                                                                                       | - |
| NOTE :                 | FOR B            | ORING    | SUMM<br>SEE      | ARY AND SHEET 1.             |                    | STO                | NE & WEBSTER ENG. CORP. APPROVED DATE M BORING NO. SHEET<br>TCH No. 12241-GSK-234C                                                                                                                                                                    | 3 |

,

| C<br>                     | OORDIA                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | s _                                                                                           |                                                                                            |                                                                             |                                           | •                                      | APING SROUND ELEV (1) 727.3                                                                                                                                                                     |
|---------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------|-------------------------------------------|----------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| II<br>S<br>D<br>M<br>S    | NCLINA<br>NATE :<br>TATIC<br>EPTH<br>IETHOD<br>DR<br>SA<br>DR<br>PECIAL                                                                                            | TION<br>STAR<br>GROU<br>TO U<br>S :<br>NLLIN<br>NLLIN<br>TE:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | T/F<br>UND<br>BED<br>BED<br>NG S<br>NG F<br>STIN                                              | FINISH<br>WATE<br>ROCK<br>SOL<br>SOL<br>ROCK                                               | 1<br>R C<br>                                                                | 10-8<br>DEPT<br>AW B<br>2.0<br>N/A<br>STR | . 8E<br>/81<br>H / D.<br>ODS.<br>IN 0. | ARING INSPECTOR Hector<br>/ 10/8/81 CONTRACTOR / DRILLEREGER DRILLING/JARVIS<br>ATE (FT) / DRILL RIG TYPE<br>(FT) TOTAL DEPTH DRILLED 17.5 (FT)<br>3 IN ROLLER BIT<br>D. SPLIT BARREL<br>TATION |
| c                         | OMMEN                                                                                                                                                              | ITS _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | <u>4</u> I                                                                                    | FT EAS                                                                                     | T OF                                                                        | SEO-                                      | -1                                     |                                                                                                                                                                                                 |
| ELEVATION<br>(FEET)(16.2) | DEPTH<br>(FEET)                                                                                                                                                    | SAMPLE<br>TYPE (7)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | SAMPLE<br>Number                                                                              | BLOWS (3)                                                                                  | REC/ROD (4)                                                                 | SPT N<br>VALUE (5)                        | GROUP<br>SYMBOL (6)                    | SAMPLE DESCRIPTION                                                                                                                                                                              |
| 727.3                     |                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                               |                                                                                            |                                                                             | <u> </u>                                  |                                        |                                                                                                                                                                                                 |
| 720.0                     | 5 1 4                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                               | 3                                                                                          |                                                                             |                                           |                                        | AUGERED TO 14.5 FT - NO SAMPLES                                                                                                                                                                 |
| 710.0                     | 15   1<br>20   1                                                                                                                                                   | 5<br>_ <u>S</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1<br>2                                                                                        | 4-4-<br>(18"<br>4-3-<br>(0")                                                               | 3                                                                           | 7                                         |                                        | SANDY SILT, 10-152 FINE SAND, 52 MEDIUM TO FINE GRAVEL, FEW SANDSTONE<br>AND SHALE FRAGMENTS, BROWN.<br>SANDSTONE FRAGMENT IN BOTTOM OF SAMPLER.<br>END OF BORING AT 17.5 FT.                   |
|                           | DATUM                                                                                                                                                              | 15 MI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | EAN                                                                                           | SEA L                                                                                      | EVE                                                                         | L                                         | 7.                                     | S-SPLIT BARREL SAMPLE                                                                                                                                                                           |
| GEND / NOTES              | CROCK Q<br>BLOWS<br>2"0.D. S<br>DISTAN(<br>14016. Ho<br>HAMMET<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE | UND V<br>REQU<br>AMPLI<br>CE SH<br>AMMEI<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>CATES<br>C | VATE<br>IIRED<br>E SP<br>IOWN<br>R FAI<br>USE<br>INC<br>OVEF<br>RE R<br>CY D<br>RATIO<br>BLOV | K LEN<br>TO (<br>OON (<br>USIN<br>LLING<br>CHES (<br>RY.<br>ECOVE<br>ESIGN<br>DN<br>VS/FT. | FEL<br>04174<br>5" OF<br>5<br>500<br>500<br>500<br>500<br>500<br>500<br>500 | E<br>R<br>1.                              |                                        | BORING LOG<br>BEAVER VALLEY POWER STATI<br>UNIT 2<br>DUQUESNE LIGHT COMPANY<br>SHIPPINGPORT, PENNSYLVANI<br>STONE & WEBSTER ENG, CORP.<br>SKETCH No. 12241-05K-235                              |

| -<br>-      |                                         | e N3743                                 | .24                         |                                               | E9373.28 GROWN FIEW (1) 728.7 SHEETOF                                                                                        |
|-------------|-----------------------------------------|-----------------------------------------|-----------------------------|-----------------------------------------------|------------------------------------------------------------------------------------------------------------------------------|
| С<br>14     | UURDINATE:                              | s <u></u>                               |                             | <br>                                          |                                                                                                                              |
| -<br>-<br>- | ATE · CTAD                              | T / EINIGH                              | 10/8/                       | . 90<br>81                                    | / 10/9/81 CONTRACTOR / DBN + ED EGER DRILLING/JARVIS                                                                         |
| с.          | ALE . JIAR                              |                                         |                             | <u>н</u> / п                                  | ATE 42 (FT) / 10/10/81 DBH + BIG TYDE                                                                                        |
| 5           | EDTH TO                                 |                                         | 105.0                       |                                               | (FT) TOTAL DEPTH DONLER 105.8 (FT)                                                                                           |
|             | EFTHODS '                               |                                         |                             |                                               |                                                                                                                              |
|             | DRILLI                                  | NG SOLL                                 | AW RO                       | DS, 3                                         | IN ROLLER BIT, DRILLING MUD AND CASING                                                                                       |
|             | SAMPL                                   | ING SOL                                 | 2.0 I                       | NCH (                                         | D.D. SPLIT BARREL                                                                                                            |
|             | DRILLI                                  | NG ROCK                                 | N/A                         |                                               |                                                                                                                              |
| S           | PECIAL TE                               | STING OR                                | INSTR                       | UMEN                                          |                                                                                                                              |
|             |                                         | 0.0000000000000000000000000000000000000 |                             | 1 1700                                        | ov 10/11/201                                                                                                                 |
| C           | OMMENTS _                               | GROUNDWAT                               | <u>ER AT 39</u><br>PPROXIMA | <u>.3 FT</u><br>TELV                          | <u>0N 10/12/81</u>                                                                                                           |
|             |                                         |                                         | III KOALIA                  |                                               | 42.11                                                                                                                        |
|             |                                         |                                         |                             |                                               | -                                                                                                                            |
| N (EE2      | н Г ш <sup>(</sup>                      | w a 🗊                                   | 3 3                         | 9                                             |                                                                                                                              |
| VATI<br>EET | EPT<br>SEPT<br>SEE                      | MPL<br>OWS                              | PT /                        | NOU NO                                        | SAMPLE DESCRIPTION                                                                                                           |
| ELE<br>F    | 3 5 5                                   | BLAK                                    | AEC S                       | ο Y G                                         |                                                                                                                              |
| 1           | · · ·                                   |                                         | L                           | L                                             | L                                                                                                                            |
| 728.7       |                                         |                                         |                             | <u>,                                     </u> |                                                                                                                              |
|             | S                                       | 1 29-24<br>(9")                         | /5" 24/5"                   | 1                                             | FILL, LARGE SANDSTONE FRAGMENTS, SILTY SAND, CONCRETE.<br>(HARD AUGERING TO 3 FT).                                           |
|             | s                                       | 2 18-9-                                 | 7 16                        | SM                                            | SILTY SAND, 10-20% COARSE TO FINE GRAVEL, ROUNDED TO ANGULAR, COARSE TO                                                      |
|             | 5                                       | (11")                                   |                             |                                               | FINE SAND, MOSTLY MEDIUM TO FINE, DARK BROWN.                                                                                |
|             | - S<br>                                 | 3 1-1-3<br>(11")                        | 4                           | SM                                            | SANDY SILT, 5-10% COARSE TO FINE GRAVEL, 10-20% COARSE TO FINE SAND,<br>MOSTLY FINE, PAPER, GRAY.                            |
| 720.0       | s                                       | 4 3-4-2                                 | 6                           |                                               | (HIT REBAR)<br>SAME AS ABOVE.                                                                                                |
|             | 10 -                                    | (18")                                   |                             |                                               |                                                                                                                              |
|             |                                         | 5 (18")                                 | 4                           | SM                                            | SIMILAR TO ABOVE, SANDY SILT, 4-6% COARSE TO FINE GRAVEL, TRACE OF ROOTS, DARK GRAY.                                         |
|             | S                                       | 6 3-1-3                                 | 4                           | รห                                            | SILTY SAND, LESS THAN 5% COARSE TO FINE GRAVEL. FEW FRAGMENTS TO 1 IN.                                                       |
|             | 15                                      | (16")                                   |                             |                                               | COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 10-15% NONPLASTIC TO SLIGHTLY                                                    |
|             |                                         | 7 2-3-4                                 | 7                           | SM                                            | SANDY SILT, 10-15% COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, NONPLASTIC                                                    |
| 710.0       | s                                       | 8 3-2-3                                 | 5                           | SM                                            | IO SLIGHTLY PLASTIC, TRACE OF GRAVEL, DARK GRAY.<br>SIMILAR TO ABOVE, FEW BROWN SANDSTONE FRACMENTS TO 1 IN, TRACE OF ROOTS. |
| 1010        | 20 -                                    | (18")                                   |                             |                                               |                                                                                                                              |
|             | <b>-</b> s                              | 9 3-3-4<br>(18")                        | 7                           | SM                                            | SAME AS ABOVE.                                                                                                               |
|             | s                                       | 10 3-6-8                                | 14                          | SM                                            | TOP 11 IN, SAME AS ABOVE.                                                                                                    |
|             | 25                                      | (18'')                                  |                             | SP                                            | BOTTOM 7 IN, <u>SAND</u> , COARSE TO FINE, MOSTLY MEDIUM TO FINE, 5-10% NONPLASTIC FINES, BROWN.                             |
|             | <u></u> s                               | 11 3-3-4                                | 7                           | SM                                            | SILTY SAND, 5% FINE GRAVEL, COARSE TO FINE SAND, 10-15% NONPLASTIC FINES,                                                    |
| 00.0        |                                         | 12 3-4-5                                |                             | 24                                            |                                                                                                                              |
| 00.0        | 30                                      | (18")                                   | 9                           | 3.4                                           | SARE AS ABOVE, I LARGE <u>SANDSIONE</u> FRAMENI.                                                                             |
| 1.          | DATUM IS MI                             | EAN SEA LI<br>Vater Lev                 | EVEL<br>El                  | 7.                                            | S-SPLIT BARREL SAMPLE                                                                                                        |
| 3.          | BLOWS REQU                              | HRED TO D                               | RIVE                        |                                               | BEAVER VALLEY POWER STATI                                                                                                    |
| S I I       | DISTANCE SH                             | IOWN USING                              | - 0R<br>                    |                                               |                                                                                                                              |
| 2           | HOID. HANNEL                            | N FALLING                               | 30 .<br>00 lb.              |                                               | DUQUESNE LIGHT COMPANY                                                                                                       |
| ~           | HAMMER. (<br>SAMPLE REC                 | OVERY.                                  | ) <b>r</b>                  |                                               | SHIPPINGPORT PENNSYI VANI                                                                                                    |
| 0 4.        | % ROCK CO                               | RE RECOVE                               | RY/                         |                                               | SHIFFINGFORT, FENINSTEVANIA                                                                                                  |
| Z           | ROCK QUALI                              | TY DESIGN                               | ATION.                      |                                               | A STONE & WEBSTER ENG CORP                                                                                                   |
| N 39 5.     | ROCK QUALI<br>STD. PENETI<br>RESISTANCE | TY DESIGN/<br>RATION<br>BLOWS/FT.       | ATION.                      |                                               | STONE & WEBSTER ENG. CORP.<br>SKETCH NO. 12241-GSK-236A                                                                      |

 $\overline{}$ 

|                          |                 |                    |                  |                                   |                    |                     | SHEET 2_ OF _3                                                                                                                                                                                                |   |
|--------------------------|-----------------|--------------------|------------------|-----------------------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|
| SI                       | TE <u>Be</u> /  | VER V              | ALLEY            | POWER ST.                         | ATION              | - UN                | IT 2 J.O. NO. 12241                                                                                                                                                                                           |   |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(Feet) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4)    | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                            |   |
|                          | -               | S                  | 13               | 4-4-5<br>(18")                    | 9                  | SM                  | SILTY SAND, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 10-20% NONPLAST<br>FINES, TRACE OF BLACK ORGANICS, ORANGE BROWN.                                                                                      | C |
|                          | 35 -            | S                  | 14               | 8-6-7<br>(14")                    | 13                 | SM                  | TOP 7 IN, <u>SANDY SILT</u> , 10% COARSE TO FINE GRAVEL, 10-20% COARSE TO FINE<br>SAND, NONPLASTIC, GRAY.<br>BOTTOM 7 IN, <u>SILTY SAND</u> , COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE,                     |   |
|                          | -               | S                  | 15               | 4-3-5<br>(18")                    | 8                  | SM                  | 10-20% NONPLASTIC FINES, TRACE OF GRAVEL, GRAY.<br>SAME AS ABOVE, STRONG OIL SMELL.                                                                                                                           |   |
| 690.0                    | 40 —            | s                  | 16               | 5-5-3<br>(18")                    | 8                  | SM                  | SILTY SAND, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 15-20%<br>NONPLASTIC TO SLIGHTLY PLASTIC FINES, TRACE OF FINE GRAVEL, OIL SMELL,<br>GRAY.                                                             |   |
|                          | -               | s                  | 17               | 3-4-12<br>(14")                   | 16                 | SM                  | TOP 8 IN, SAME AS ABOVE.<br>BOTTOM 6 IN, LARGE SANDSTONE FRAGMENTS TO 14 IN WITH GRAY SILTY CLAY.                                                                                                             |   |
|                          | -               | 5                  | 18               | 3-5-8<br>(18")                    | 13                 | ML                  | CLAYEY SILT, 5% FINE SAND, 7-12% SLIGHTLY PLASTIC TO NONPLASTIC FINES,<br>TRACE OF FINE GRAVEL, TRACE OF ROOTS, OIL SMELL, BROWNISH GRAY. (1 tef)                                                             |   |
|                          | 45 -            | s                  | 19               | 12-10-9<br>(18")                  | 19                 | SM                  | TOP 5 IN, SANDY SILT, 10% FINE SAND, GRAY.<br>BOTTOM 13 IN, WEATHERED BROKEN SANDSTONE FRAGMENTS TO 14 IN, WITH SILTY<br>COARSE TO FINE SAND, GRAY, BROWN, ORANGE                                             |   |
| 680.0                    | 50 -            | s                  | 20               | 11-10-8<br>(10")                  | 18                 | SM                  | SANDY SILT, WITH WEATHERED SANDSTONE AND SHALE FRAGMENTS, 10-15Z COARSE<br>TO FINE SAND, NONPLASTIC TO SLIGHTLY PLASTIC, GRAY AND BROWN.                                                                      |   |
| 570.0                    | 55              | S                  | 21<br>22         | 9-6-18<br>(18")<br>8-9-8<br>(12") | 24                 | SM<br>SM            | TOP 2 IN, SAME AS ABOVE.<br>MIDDLE 2 IN, SILT, 52 FINE SAND, NONPLASTIC, BROWN.<br>BOTTON 14 IN, SAME AS TOP.<br>SAME AS ABOVE, <u>SANDY SILT</u> AND WEATHERED SANDSTONE AND SHALE FRACMENTS.                |   |
|                          | 65 <b>-</b>     | <u>s</u>           | 23               | 10-9-6<br>(8")                    | 15                 | SM                  | TOP 6 IN, SAME AS ABOVE.<br>BOTTOM 2 IN, SILTY FINE SAND, 10-15% NONPLASTIC FINES, BROWN.                                                                                                                     |   |
| 660.0                    | 70              | S                  | 24               | 11-11-7<br>(3")                   | 18                 | sw                  | SAND, WELL-GRADED COARSE TO FINE, LESS THAN 5% COARSE TO FINE GRAVEL,<br>FEW LARGE PIECES, 5% NONPLASTIC FINES, BROWN.                                                                                        |   |
|                          | 75 -            | <u> </u>           | 25               | 14-11-13<br>(L2")                 | 24                 | ŚW                  | SIMILAR TO ABOVE<br>AT 75.3 FT SAND, OILY(?), NO SMELL,<br>2 IN THICK, BLACK STAINS                                                                                                                           |   |
| 650.0                    | 80 -            | S                  | 26               | 14-20-12<br>(16")                 | 32                 | GM                  | <u>SILTY GRAVEL</u> , 52 COARSE TO FINE SAND, MOSTLY FINE, 10-202 NONPLASTIC TO<br>SLIGHTLY PLASTIC FINES, ANGULAR TO SUBROUNDED WEATHERED SANDSTONE AND<br>SHALE FRAGMENTS, FEW SANDSTONE FRAGMENTS TO 1 IN. | 1 |
|                          | 85              | S                  | 27               | 2-10-21<br>(18")                  | 31                 | SM                  | TOP 12 IN, <u>SILTY FINE SAND</u> , 10-207 NONPLASTIC FINES, BROWN.<br>BOTTOM 6 IN, SAME AS S-26.                                                                                                             |   |
| \$40.0                   | 90              |                    |                  |                                   |                    |                     |                                                                                                                                                                                                               |   |

•

| SI          | TF B            | EAVER              | VALL             | EY POWER                       | STATI              | on -                | UNIT 2                                                                                                                | SHEET 3_0                        | F_3    |
|-------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------|----------------------------------|--------|
| (FEET)(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                    |                                  |        |
|             |                 | S                  | 28               | 20-20-24<br>(11")              | 44                 | GM                  | SILTY SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                 | ·····                            |        |
|             | 95 <b>   </b>   | S                  | 29               | 14-15-11<br>(6눅")              | 26                 | SW                  | GRAVELLY SAND, 10-20% COARSE TO FINE ROUNDED TO SUB.<br>COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, 5% NO<br>BROWN. | ANGULAR GRAVEI<br>NPLASTIC FINES | ,<br>, |
| 0.0         | 100             | S                  | 30               | 15-14-15<br>(7")               | 29                 | SW                  | SAME AS ABOVE, ONE SANDSTONE FRAGMENT TO 14 IN.                                                                       |                                  |        |
| 2.9         |                 | S                  | 31               | 26- <u>100</u><br>.3'<br>(2'') | <u>100</u><br>.3'  |                     | GRAY, SOFT CLAYEY SHALE.<br>END OF BORING AT 105.8 FT.                                                                |                                  |        |
|             |                 |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             |                 |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             |                 |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             |                 |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             | 1.1.1.1         |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             | 1111            |                    |                  |                                |                    |                     |                                                                                                                       | . •                              |        |
|             |                 |                    |                  |                                |                    |                     |                                                                                                                       |                                  |        |
|             | 4111            | •.                 |                  |                                |                    |                     |                                                                                                                       |                                  |        |

| S<br>C<br>IN<br>D<br>S<br>D<br>M<br>S<br>C                           | ITE BEAVER VALI<br>COORDINATES _<br>NCLINATION<br>ATE : START / F<br>TATIC GROUND<br>EPTH TO BEDI<br>DETHODS :<br>DRILLING S<br>SAMPLING<br>DRILLING F<br>PECIAL TESTIN                                                                            | EY POWER ST/<br>N3649,42<br>FINISH 10/11<br>WATER DE<br>ROCK<br>SOL<br>SOL<br>ROCK<br>HG OR INST                                                              | ATION                            | INIT 2<br>E9393.59 GROUND ELE<br>ARING IN:<br>/ _10/13/81 CONTRACTOR<br>ATE (FT) / D<br>(FT) TOTAL DEPTI<br>IN ROLLER BIT, DRILLING MUD AN<br>. SPLIT BARREL<br>ITATION<br>40 FT | JO. NO. <u>12241</u><br>EV. (I) <u>727.2</u><br>SPECTOR J.W. McCOY<br>R / DRILLER EGER DRI<br>RILL RIG TYPE<br>M DRILLED10<br>ND CASING | BORING NO. <u>SEC-3</u><br>SHEET_LOF 3<br>LLING/JARVIS                                                                                    |
|----------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162)                                             | DEPTH<br>(FEET)<br>Sample<br>TYPE (7)<br>Sample<br>Number                                                                                                                                                                                          | BLOWS (3)<br>OR<br>REC/ROD (4)                                                                                                                                | VALUE (5)<br>GROUP<br>SYMBOL (6) | SAMPI                                                                                                                                                                            | LE DESCRIPTION                                                                                                                          |                                                                                                                                           |
| 27.2                                                                 |                                                                                                                                                                                                                                                    |                                                                                                                                                               |                                  |                                                                                                                                                                                  |                                                                                                                                         |                                                                                                                                           |
|                                                                      |                                                                                                                                                                                                                                                    | 37-34-35<br>(16")<br>7-5-3                                                                                                                                    | 69<br>8 SW                       | ROAD FILL, SANDY COARSE TO FI                                                                                                                                                    | NE SLAG, GRAVEL TO I I                                                                                                                  | TEL AND SLAG, COARSE TO                                                                                                                   |
|                                                                      | 5                                                                                                                                                                                                                                                  | (12")                                                                                                                                                         |                                  | FINE SAND, GRAY.<br>BOTTOM 7 IN, <u>GRAVELLY SAND</u> , 1<br>SAND, BROWN.                                                                                                        | 0-15% COARSE TO FINE O                                                                                                                  | GRAVEL, COARSE TO FINE                                                                                                                    |
| 20.0                                                                 |                                                                                                                                                                                                                                                    | (15")                                                                                                                                                         | SM                               | SANDY SILT, 15-20% FINE SAND<br>GRAVEL, TRACE OF ROOTS, GRAY.                                                                                                                    | , NONPLASTIC TO SLIGHT                                                                                                                  | TLY PLASTIC, TRACE OF                                                                                                                     |
|                                                                      |                                                                                                                                                                                                                                                    | 1-1-2<br>(15")                                                                                                                                                | 3 SM                             | SAME AS ABOVE                                                                                                                                                                    |                                                                                                                                         |                                                                                                                                           |
|                                                                      | - S 5                                                                                                                                                                                                                                              | 1-1-2<br>(17")                                                                                                                                                | 3 SM                             | SIMILAR TO ABOVE, GRASS, ROOT                                                                                                                                                    | S, FEW SANDSTONE FRAGE                                                                                                                  | ænts. –                                                                                                                                   |
|                                                                      |                                                                                                                                                                                                                                                    | 1-2-1<br>(13")                                                                                                                                                | 3 SM                             | SAME AS ABOVE.                                                                                                                                                                   |                                                                                                                                         | -                                                                                                                                         |
| 10.0                                                                 | <b>. . . . .</b>                                                                                                                                                                                                                                   | 1-1-2<br>(13")                                                                                                                                                | 3 SM                             | SILTY SAND, 3-5% COARSE TO FI<br>COARSE TO FINE SAND, MOSTLY F                                                                                                                   | NE GRAVEL, 20-25% SLIG<br>TINE, TRACE OF ROOTS AN                                                                                       | GHTLY PLASTIC FINES,<br>ND WOOD, GRAY.                                                                                                    |
|                                                                      | 20 5 8                                                                                                                                                                                                                                             | 3-3-3<br>(9")                                                                                                                                                 | 6 SM                             | SAME AS ABOVE, SANDSTONE FRAG                                                                                                                                                    | MENTS AT BOTTOM.                                                                                                                        | -                                                                                                                                         |
|                                                                      | <b>S</b> 9                                                                                                                                                                                                                                         | 1-2-3<br>(18")                                                                                                                                                | 5 SM                             | SILTY FINE SAND, 5-7% COARSE<br>10-15% NONPLASTIC TO SLIGHTLY                                                                                                                    | TO FINE GRAVEL, FEW SA<br>PLASTIC FINES, TRACE                                                                                          | ANDSTONE FRAGMENTS,                                                                                                                       |
|                                                                      | - S 10                                                                                                                                                                                                                                             | 3-2-3<br>(16")                                                                                                                                                | 5 SM                             | SIMILAR TO ABOVE, 20-25% NONP<br>Smell, Gray.                                                                                                                                    | LASTIC TO SLIGHTLY PL                                                                                                                   | ASTIC FINES, ORGANIC                                                                                                                      |
| .0.0                                                                 | <b>S</b> 11                                                                                                                                                                                                                                        | 3-5-7                                                                                                                                                         | 12 SM                            | SIMILAR TO ABOVE, TRACE OF RO                                                                                                                                                    | OTS AND WOOD, BROWN                                                                                                                     | =                                                                                                                                         |
|                                                                      | 30 5 12                                                                                                                                                                                                                                            | 4-4-6<br>(18")                                                                                                                                                | 10 SM                            | SIMILAR TO ABOVE, 15-20% NONP<br>OF BLACK CINDERS, GRAY. (1 t                                                                                                                    | PLASTIC TO SLIGHTLY PLA<br>sf).                                                                                                         | ASTIC FINES, ROOTS, TRACE                                                                                                                 |
| 1.<br>2.<br>3.<br>94<br>94<br>94<br>94<br>95<br>94<br>95<br>94<br>96 | DATUM IS MEAN<br>GROUND WATE<br>BLOWS REQUIRED<br>2"Q.D. SAMPLE SP<br>DISTANCE SHOWN<br>HOID. HAMMER FA<br>HINDICATES USE<br>KAMPLE RECOVER<br>% ROCK CORE R<br>ROCK QUALITY D<br>STD. PENETRATIC<br>RESISTANCE BLOW<br>UNIFIED SOIL CL<br>SYSTEM. | SEA LEVEL<br>R LEVEL<br>TO DRIVE<br>OON 6" OR<br>USING<br>LLING 30".<br>OF 3001b.<br>HES OF<br>IV.<br>ECOVERY/<br>ESIGNATION.<br>DN<br>IS/FT.<br>ASSIFICATION | 7.                               | S-SPLIT BARREL SAMPLE                                                                                                                                                            | BORING<br>BEAVER VALLE<br>UN<br>DUQUESNE L<br>SHIPPINGPORT,<br>STONE & WEBS<br>SKETCH No. 12<br>APPROVED DATE M                         | B LOG<br>Y POWER STATION<br>IIT 2<br>IGHT COMPANY<br>PENNSYLVANIA<br>ITER ENG. CORP.<br>241-GSK- 237A<br>BORNIG NO. SHEET<br>SEO-3 L OF 3 |

| <b>.</b>                 |                 | A 1/E D            | VALT   | V DOLIDE CO                    | 4 *** ~ ~          | 1 •                 | BORING NO. SEO-3<br>SHEET 2 OF 3                                                                                                                                                                 |
|--------------------------|-----------------|--------------------|--------|--------------------------------|--------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(I62) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                               |
|                          | Γ               |                    |        |                                |                    |                     |                                                                                                                                                                                                  |
| :                        | -               | s                  | 13     | 3-3-4<br>(18")                 | 7                  | SM                  | SANDY SILT, 5-77 COARSE TO FINE GRAVEL, 20-30% FINE SAND, NONPLASTIC TO SLIGHTLY PLASTIC, TRACE OF ROOTS, BROWN.                                                                                 |
|                          | 35 —            | S                  | 14     | 3-4-4<br>(18")                 | 8                  | SM                  | SIMILAR TO ABOVE, 10-15% FINE SAND.                                                                                                                                                              |
| 590.0                    |                 | S                  | 15     | 3-4-5<br>(16")                 | 9                  | SM                  | SILTY FINE SAND, 5-7% COARSE TO FINE GRAVEL, LARGE SLAG PIECE NEAR<br>BOTTOM, SULFUR SMELL, 20-30% SLIGHTLY PLASTIC FINES, TRACE OF RED<br>CLAY SEAM. BROWN.                                     |
|                          | 40 —            | s                  | 16     | 5-5-8<br>(18'')                | 13                 | SM                  | SILTY FINE SAND, ORGANIC, 5-7% COARSE TO FINE GRAVEL, 15-20% NONPLASTIC<br>FINES, WOOD AT BOTTOM AND SANDSTONE FRAGMENTS.                                                                        |
|                          | -               | s<br>s             | 17     | <u>100</u><br>5"               | <u>100</u><br>5''  |                     | WOOD                                                                                                                                                                                             |
|                          | 45 -            | s                  | 19     | 4-3-3<br>(9")                  | 6                  | он                  | WOOD ON TOP<br>ORGANIC CLAYEY SILT, MODERATELY PLASTIC TO VERY PLASTIC, 5-107 FINE SAND<br>TRACE OF COARSE TO FINE GRAVEL, GRAY.                                                                 |
| 580.0                    |                 | <br>S              | 20     | 5-6-8<br>(11")<br>9-7-9        | 14                 | см                  | SILTY SANDY GRAVEL, COARSE TO FINE GRAVEL AND WEATHERED SANDSTONE AND SHAL<br>FRAGMENTS, FEW FRAGMENTS TO 14 IN, 6-12% COARSE TO FINE SAND, 10-15%                                               |
|                          | 50 -            | S                  | 21     | (11")<br>6-5-5<br>(18")        | 10                 |                     | NONPLASTIC TO SLIGHTLY PLASTIC FINES, TRACE OF COAL FRAGMENTS, BROWN.<br>(NO RECOVERY FIRST ATTEMPT).<br>SAND COARSE TO FINE 5-72 COARSE TO FINE GRAVEL, 5-73 NON-                               |
|                          |                 |                    |        | (10)                           |                    | SW<br>GM            | PLASTIC FINES, FEW SANDSTONE FRAGMENTS TO 14 IN, BROWN.<br>SILTY GRAVEL, SOME SAND, COARSE TO FINE GRAVEL AND WEATHERED SANDSTONE                                                                |
|                          | 55 -            | -                  |        |                                |                    | <b>C</b> Y          | AND SHALE, ANGULAR, 5-;0% COARSE TO FINE SAND, 15-20% NONPLASTIC TO SLIGHTLY PLASTIC FINES, GRAY.                                                                                                |
| 70.0                     |                 | <u>s</u>           | 22     | 7-13-14<br>(12")               | 27                 | GM                  | SIMILAR TO ABOVE, SLIGHT OIL SMELL, RED AND BROWN SHALE FRAGMENTS.                                                                                                                               |
|                          | 60 —            |                    |        | 6-7-9                          | 15                 | GM                  | SAME AS ABOVE (NO OTI SMETT)                                                                                                                                                                     |
|                          |                 |                    | 23     | (11")                          | 13                 | SW                  | BOTTOM 1 IN, <u>SAND</u> , COARSE TO FINE, 5-7% NONPLASTIC FINES, TRACE OF<br>GRAVEL, GRAY.                                                                                                      |
|                          | 65 —            |                    |        |                                | .,                 | ഷ                   | TOD 3 THE CANER COALER TO THE CANE BOUND                                                                                                                                                         |
| 660.0                    | -               |                    | 24     | (11")                          | 14                 | SP                  | BOTTOM 9 IN, FINE SAND, 5-7% COARSE TO FINE GRAVEL, 5-7% NONPLASTIC FINES<br>FEW SANDSTONE FRACMENTS, BROWN.                                                                                     |
|                          | 70 -            |                    |        |                                |                    | <b>611</b>          |                                                                                                                                                                                                  |
|                          | -               | S<br>              | 25     | 11-13-13<br>((11")             | 26                 | 5₩                  | TOP 5 IN, <u>GRAVELLY SAND</u> , 10-15% COARSE TO FINE GRAVEL, 5-7% NONPLASTIC<br>TO SLIGHTLY PLASTIC FINES, BROWN.<br>MIDDLE 5 IN, BROKEN SOFT SANDSTONE FRAGMENTS WITH COARSE SAND, GRAY.      |
|                          | 75 -            |                    |        |                                |                    | 24                  | BOTTOM I IN, SAME AS TOP.                                                                                                                                                                        |
| 650.0                    | -               | S                  | 26     | 10-12-13<br>(13'')             | 25                 | SW<br>GW            | TOP 5 IN, <u>GRAVELLY SAND</u> , 10-15% COARSE TO FINE GRAVEL, 5-7% NONPLASTIC<br>TO SLIGHTLY PLASTIC FINES, BROWN.<br>BOTTOM & IN, SANDY GRAVEL, COARSE TO FINE GRAVEL-SIZED WEATHERED SANDSTON |
| ,                        | 80 -            |                    |        |                                |                    |                     | AND SHALE, ANGULAR TO SUBROUNDED, 20-25% COARSE TO FINE SAND, 10-15% NONPLASTIC TO SLIGHTLY PLASTIC FINES, BROWN.                                                                                |
|                          |                 | . S                | 27     | 16-18-21<br>(15")              | 39                 | GW                  | SAME AS ABOVE, SANDY GRAVEL.                                                                                                                                                                     |
|                          |                 |                    |        |                                |                    |                     |                                                                                                                                                                                                  |
| 640 0                    | -               | s_                 | 28     | 21-15-21<br>(12")              | 37                 | SW                  | TOP 5 IN, BROKEN <u>SANDSTONE</u> FRAGMENTS AND <u>SAND</u> , GRAY.<br>BOTTOM, SAME AS S-27.                                                                                                     |
| 040.0                    | 90              |                    |        |                                |                    |                     |                                                                                                                                                                                                  |
| OTE : I                  | FOR BO          | RING               | SUMM   | ARY AND                        |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE M BORING NO. SHEET                                                                                                                                         |

|                           |                 |                    |                  |                      | -                    |                                  | BORING NO. <u>SEO-3</u><br>SHEET <u>1</u> OF <u>3</u>                                                                                                                                                                                        |
|---------------------------|-----------------|--------------------|------------------|----------------------|----------------------|----------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| S                         | TE              | EAVER              | VALL             | EY POWE              | R STAT               | ION - 1                          | J.O. NO. <u>12241</u>                                                                                                                                                                                                                        |
| ELEVATION<br>(FEET)(16.2) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)            | REC/NOD (4)<br>SPT N | VALUE (5)<br>GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                           |
|                           |                 | s                  | 29               | 12-10-               | 12 22                | GW                               | SAME AS ABOVE, <u>SANDY GRAVEL</u> .                                                                                                                                                                                                         |
|                           |                 | 1                  | ] .              |                      | ·                    |                                  |                                                                                                                                                                                                                                              |
| 30.0                      | 95 -            | <b>S</b>           | 30               | 18-15-<br>(11")      | 18 33                | SP                               | GRAVELLY SAND, 20-30% COARSE TO FINE ROUNDED TO ANGULAR GRAVEL AND<br>SOFT COAL FRAGMENTS, FEW SANDSTONE FRAGMENTS TO 14 IN, COARSE TO FINE<br>SAND, MOSTLY COARSE TO MEDIUM, 5-7% NONPLASTIC TO SLIGHTLY PLASTIC FINES,<br>BROWN AND BLACK. |
|                           | 100 —           | s                  | 31               | 16-14-<br>(10")      | 19 33                | SP                               | SAND, 5-102 COARSE TO FINE GRAVEL, SANDSTONE AND SHALE, ROUNDED TO<br>ANGULAR, TRACE OF COAL, COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM,<br>BROWN TO ORANGE BROWN.                                                                        |
|                           | 105 —           |                    | 32               | $\frac{100}{.2^{1}}$ | <u>10</u><br>.2      | <u></u>                          | SOFT WEATHERED THINLY BEDDED GRAY SILTSTONE                                                                                                                                                                                                  |
|                           |                 |                    |                  | <b>-</b>             |                      | T                                | END OF BORING AT 105.2 FT                                                                                                                                                                                                                    |
|                           |                 |                    |                  | ł                    |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 |                    |                  |                      |                      |                                  | -                                                                                                                                                                                                                                            |
|                           |                 |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | ]                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           | -               |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | 1                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | ]                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           | -               |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | ]                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           | -               | 1                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | 1                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 | ]                  |                  |                      | 1                    |                                  |                                                                                                                                                                                                                                              |
|                           | -               | 1                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           | -               |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           |                 |                    |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
|                           | -               | 4                  |                  |                      |                      |                                  |                                                                                                                                                                                                                                              |
| TE:                       | FOR BC          | RNG                | 5UM64            | ARY AN               | • 1                  | STO                              | NE & WEBSTER ENG. CORP. APPROVED DATE M BORNS NO. SHEET                                                                                                                                                                                      |

|                      | OORDI                                                                       | NATE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | <u>s</u> <u>n</u>                                                          | 3629.89                                                                            |                | E                | 9275.04 GROUND ELEV. (I) SHEETOF                                                                                                           |
|----------------------|-----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------|------------------------------------------------------------------------------------|----------------|------------------|--------------------------------------------------------------------------------------------------------------------------------------------|
| IN                   |                                                                             | TION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                            |                                                                                    |                | BE               | ARING INSPECTOR JACCOY                                                                                                                     |
| D                    | ATE :                                                                       | STAR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | T/F                                                                        |                                                                                    | 10/11/         | /81              | / 10/11/81 CONTRACTOR / DRILLEREGER_DRILLING/JARVIS                                                                                        |
| \$1                  | TATIC                                                                       | GRO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | UNDY                                                                       | NATER C                                                                            | EPT            | H / D/           | ATE 36 (FT) / 10/12/81 DRILL RIG TYPE                                                                                                      |
| DI                   | EPTH                                                                        | то                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | BEDR                                                                       | юск                                                                                | 103            |                  | (PT) TOTAL DEPTH DRILLED 103.1 (PT)                                                                                                        |
| M                    | ETHO                                                                        | os:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                            |                                                                                    |                |                  |                                                                                                                                            |
|                      | DI                                                                          | RILLI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | NG S                                                                       | SOIL                                                                               | AW ROL         | os <u>3</u>      | IN ROLLER BIT, DRILLING MUD AND CASING                                                                                                     |
|                      | S                                                                           | AMPL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ING                                                                        | SOIL                                                                               | 2.0 IX         | 10.D             | SPLIT BARREL                                                                                                                               |
| -                    | DI                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | NG R                                                                       | юск                                                                                | N/A            |                  |                                                                                                                                            |
| 51                   | PECIA                                                                       | LITE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | STIN                                                                       | G OR IN                                                                            | STRU           | JMEN             | TATION                                                                                                                                     |
|                      |                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            | TT TO APP                                                                          |                | -<br>TET V       | LE TA                                                                                                                                      |
|                      |                                                                             | N15 .                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | <u> </u>                                                                   | LL IV AFC                                                                          | KUALITU        | <u>AICLI</u>     | 45 FI                                                                                                                                      |
|                      |                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            |                                                                                    |                |                  |                                                                                                                                            |
| ন                    |                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            |                                                                                    | T              |                  |                                                                                                                                            |
| 10N                  | ΞĒ                                                                          | ت<br>س                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | <u>ш</u> е                                                                 | e S                                                                                | <u>,</u> 0     | 9                |                                                                                                                                            |
| FEET                 | DEPI                                                                        | AMPL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | AMPL                                                                       | OWS<br>OR                                                                          | PT I           | NBO              | SAMPLE DESCRIPTION                                                                                                                         |
| ELE                  |                                                                             | S.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | δŻ                                                                         | BL<br>REC                                                                          | o, 2           | З <sup>у</sup> с |                                                                                                                                            |
| ł                    |                                                                             | L                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                            | L                                                                                  | L              |                  | · · · · · · · · · · · · · · · · · · ·                                                                                                      |
| 726.4                |                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            |                                                                                    |                |                  |                                                                                                                                            |
|                      | -                                                                           | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 1                                                                          | 6-14-16<br>(12")                                                                   | 30             | SM               | TOP 3 IN SLAG FILL.<br>BOTTOM 9 IN, SILTY FINE SAND,10-15% NONPLASTIC FINES, TRACE OF GRAVEL,BRC                                           |
|                      | -                                                                           | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 2                                                                          | 6-4-4                                                                              | 8              | SM               | SILTY FINE AND, 5-10% NONPLASTIC FINES, FEW ROCK FRAGMENTS, TRACE OF                                                                       |
|                      | 5 -                                                                         | ╉─┤                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                            | (18")                                                                              |                |                  | ROCK FRAGMENTS, TRACE OF GRAVEL, TRACE OF BLACK CINDERS, BROWN.                                                                            |
| 720.0                | -                                                                           | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 3                                                                          | 5-3-3                                                                              | 6              | SM               | SILTY SAND, 10-12% COARSE TO FINE GRAVEL, FEW ROCK FRAGMENTS, COAL AND                                                                     |
|                      | -                                                                           | <b>_</b>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                            | (13.)                                                                              |                |                  | SLAG, COARSE TO FINE SAND, MOSILI MEDIUM TO FINE, 10-134 NUNFLASITO FINES<br>BROWN.                                                        |
|                      | 10 🛶                                                                        | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 4                                                                          | 1-1-2<br>(12")                                                                     | 3              | SM               | SAME AS ABOVE, FEW LARGE SANDSTONE FRAGMENTS TO 1 IN.                                                                                      |
|                      | •                                                                           | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 5                                                                          | 1-2-1                                                                              | 3              | SM               | SAME AS ABOVE.                                                                                                                             |
|                      | -                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            | (8")                                                                               |                | ~                |                                                                                                                                            |
|                      | 15 -                                                                        | -S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 6                                                                          | 2-1-3<br>(10")                                                                     | 4              | SM               | SIMILAR TO ABOVE, <u>SILTY FINE SAND</u> , 10-12% COARSE TO FINE GRAVEL, FEW SANDSTONE AND SLAG FRAGMENTS, 20-25% NONPLASTIC FINES, BROWN. |
| 710.0                | -                                                                           | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 7                                                                          | 8-9-8                                                                              | 17             | GW               | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SLAG, FEW FRAGMEN                                                                  |
|                      | -                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            | (18")                                                                              |                |                  | TO 14 IN, COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, 5-7% NONPLASTIC<br>FINES, TRACE OF COAL CINDERS, BROWN.                            |
|                      | 20 -                                                                        | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 8                                                                          | 4-4-2                                                                              | 6              | SM               | SILTY FINE SAND, 5-7% COARSE TO FINE GRAVEL AND SLAG, 15-20% NONPLASTIC                                                                    |
|                      | -                                                                           | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 9                                                                          | 3-4-7                                                                              | 11             | SM               | SAND, 5-7% COARSE TO FINE GRAVEL, SANDSTONE FRAGMENT AT BOTTOM, COARSE TO                                                                  |
|                      | -                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                            | (9")                                                                               |                |                  | FINE SAND, MOSTLY COARSE TO MEDIUM, 8-10% NONPLASTIC FINES, TRACE OF SLA<br>BROWN.                                                         |
|                      | - 25                                                                        | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 10                                                                         | 3-4-3                                                                              | 7              | SP               | SAME AS ABOVE.                                                                                                                             |
| 700 0                |                                                                             | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 11                                                                         | 4-3-4                                                                              | 7              | SP               | GRAVELLY SAND, 15-20% COARSE TO FINE ROUNDED GRAVEL, FEW FRAGMENTS TO 1                                                                    |
| /00.0                |                                                                             | 1—                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                            | (16")                                                                              |                |                  | COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, 5% NONPLASTIC FINES, BROWN                                                                   |
| i                    | 30                                                                          | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 12                                                                         | 4-3-3<br>(10")                                                                     | 6              | SP               | SAME AS ABOVE.                                                                                                                             |
|                      |                                                                             | IS M                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | EAN                                                                        | SEA LEVE                                                                           | L              | 7.               | S-SPLIT BARREL SAMPLE                                                                                                                      |
|                      | -                                                                           | UND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | WATE                                                                       | R LEVEL                                                                            | c .            |                  | BORING LOG                                                                                                                                 |
| 1.                   |                                                                             | REW                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | DIVED                                                                      |                                                                                    | R              |                  | BEAVER VALLEY POWER STAT                                                                                                                   |
| 1.<br>2.<br>3.       | BLOWS<br>2"O.D.                                                             | SAMPL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | E SP                                                                       | 00N 6 0                                                                            |                |                  |                                                                                                                                            |
| 1.<br>2.<br>3.       | BLOWS<br>2"O.D. :<br>DISTAN<br>14016. H                                     | SAMPLICE SI<br>IAMME                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | e sp<br>hown<br>r fai                                                      | USING<br>LLING 30                                                                  |                |                  | UNIT 2                                                                                                                                     |
| <br>2.<br>3.         | BLOWS<br>2"Q.D. :<br>DISTAN<br>1401b. H<br>HAMME                            | SAMPLICE SI<br>IAMME<br>ICATES<br>IR. (                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | E SP<br>HOWN<br>R FAI<br>B USE<br>) INC                                    | OON 6 0<br>USING<br>LLING 30<br>OF 3001<br>CHES OF                                 | ,<br>D.        |                  | UNIT 2<br>DUQUESNE LIGHT COMPANY                                                                                                           |
| 1.<br>2.<br>3.       | BLOWS<br>2"Q.D. :<br>DISTAN<br>HOID. H<br>HAMME<br>SAMPL<br>% RO(           | SAMPLICE SI<br>IAMME<br>ICATES<br>IR. (<br>E. RE(<br>CK - CO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | E SP<br>HOWN<br>R FAI<br>B USE<br>) INC<br>COVEF<br>IRE R                  | USING<br>USING<br>LLING 30 <sup>°</sup><br>COF 3001<br>CHES OF<br>RY.<br>HECOVERY/ | b.             |                  | UNIT 2<br>DUQUESNE LIGHT COMPANY<br>SHIPPINGPORT, PENNSYLVANI                                                                              |
| 1. 12. 3.<br>3. 4. 5 | BLOWS<br>2"O.D. :<br>DISTAN<br>HOID. H<br>HAMME<br>SAMPL<br>% ROCK<br>STD 6 | SAMPLICE SI<br>IAMME<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATES<br>ICATE | E SP<br>HOWN<br>R FAI<br>3 USE<br>) INC<br>COVEF<br>IRE R<br>TY D<br>BATIO | USING<br>USING<br>LLING 30<br>E OF 3001<br>CHES OF<br>RY.<br>DESIGNATION           | n<br>D.<br>XN. |                  | UNIT 2<br>DUQUESNE LIGHT COMPANY<br>SHIPPINGPORT, PENNSYLVAN                                                                               |

|                          |                 |                       |                  |                                |                    |                     | BORING NO<br>SHEETOF                                                                                                                                                                                                                                                                          |
|--------------------------|-----------------|-----------------------|------------------|--------------------------------|--------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SI                       |                 | EAVER                 | VALLE            | TY POWER S                     | TATIO              | <u>n - 1</u>        | NIT 2                                                                                                                                                                                                                                                                                         |
| ELEVATION<br>(FEET)(162) | OEPTH<br>(FEET) | SAMPLE<br>TYPE (7)    | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                            |
|                          | <u> </u>        |                       |                  |                                | <b></b>            |                     |                                                                                                                                                                                                                                                                                               |
| -                        |                 | S                     | 13               | 4-3-3<br>(11")                 | 6                  | SP                  | SIMILAR TO ABOVE, 20-30Z COARSE TO FINE ROUNDED GRAVEL. FEW FRAGMENTS TO<br>14 IN, 5-10Z NONPLASTIC FINES.                                                                                                                                                                                    |
|                          |                 | s                     | 14               | 6-7-6<br>(10")                 | 13                 | GM                  | GRAVELLY SILTY FINE SAND, 10-152 COARSE TO FINE ROUNDED TO SUBANGULAR<br>GRAVEL, 15-202 NONPLASTIC FINES, BROWN.                                                                                                                                                                              |
| 90.0                     |                 | s                     | 15               | 2-3-2<br>(9")                  | 5                  | SM                  | SILTY FINE SAND AND LARGE SANDSTONE FRAGMENTS TO 14 IN, 20-252 NONPLASTIC<br>TO SLIGHTLY PLASTIC FINES, 5-72 COARSE TO FINE GRAVEL, BROWN.                                                                                                                                                    |
|                          |                 | - s                   | 16               | 4-4-6                          | 10                 | SM                  | SAME AS ABOVE.                                                                                                                                                                                                                                                                                |
|                          | 40-             | • 5                   | 17               | (5")<br>2-5-8                  | 13                 | GP                  | LARGE GRAVEL, SOME SAND AND SILT, WET. (POSSIBLE WASH).                                                                                                                                                                                                                                       |
| 80 <b>.0</b>             | 45 •            | -<br>-<br>-<br>-<br>- | 18               | 15-11-13<br>(10")              | 23                 | ମ୍ୟ                 | TOP 4 IN, <u>GRAVELLY SANDY SILT</u> , 10-152 COARSE TO FINE GRAVEL, MOSTLY<br>MEDIUM TO FINE, 10-152 COARSE TO FINE SAND, SLIGHTLY PLASTIC TO MODERATELY<br>PLASTIC, ORGANICS, ROOTS, GRAY AND BROWN.<br>BOTTOM 6 IN, BROKEN SANDSTONE FRAGMENTS AND COARSE TO FINE SAND, GRAY AND<br>BROWN. |
|                          | 50              |                       | 19               | 10-10-23<br>(10")              | 33                 | GW                  | SANDY GRAVEL, COARSE TO FINE ANGULAR GRAVEL AND SANDSTONE AND SHALE<br>FRACMENTS TO 14 IN, 20-252 COARSE TO FINE SAND, 5-102 NONPLASTIC TO<br>SLIGHTLY PLASTIC FINES, BROWN AND GRAY,                                                                                                         |
| 70.0                     | 55 •            |                       | . 20             | 7-7-9<br>(14")                 | 16                 | GM                  | SILTY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE, FEW LARGE<br>FRAGMENTS, 7-10% COARSE TO FINE SAND, 15-20% SLIGHTLY PLASTIC FINES,<br>GRAY (FEW RED AND BROWN SHALE).                                                                                                           |
|                          | 60 -            |                       | 21               | 9-12-9<br>(13")                | 21                 | GH                  | GRAVELLY SILTY SAND, 20-25% COARSE TO FINE GRAVEL SIZED SANDSTONE AND<br>SHALE, FEW FRAGMENTS TO 14 IN, COARSE TO FINE SAND, 10-15% NONPLASTIC<br>TO SLIGHTLY PLASTIC FINES, GRAY (BROWN SANDSTONE).                                                                                          |
| 60.0                     | 65 -            |                       | 22               | 9-9-9<br>(14")                 | 18                 | GM                  | SAME AS ABOVE.                                                                                                                                                                                                                                                                                |
|                          | 70 <b>-</b>     | -<br>- <u>s</u><br>-  | 23               | <b>10-9-15</b><br>(13")        | 24                 | GW                  | SANDY GRAVEL, COARSE TO FINE ANGULAR SANDSTONE AND SHALE FRAGMENTS, FEW<br>TO 1½ IN, 15-202 COARSE TO FINE SAND, 5-72 NONPLASTIC TO SLIGHTLY<br>PLASTIC FINES, TRACE OF CARBON, SHALE, BROWN.                                                                                                 |
| 50.0                     | 75 -            |                       | 24               | 11-10 <del>-</del> 9<br>(12")  | 19                 | ĠŴ                  | SAME AS ABOVE.                                                                                                                                                                                                                                                                                |
|                          | 80-             |                       | 25               | 10-9-11<br>(14")               | 20                 | SM<br>GW<br>SP      | TOP 4 IN, <u>SILTY FINE SAND</u> , 20-30% NONPLASTIC TO SLIGHTLY PLASTIC FINES,<br>BROWN.<br>MIDDLE 5 IN, SAME AS S24.<br>BOTTOM 5 IN, <u>SAND</u> , LARGE SANDSTONE FRAGMENT AT BOTTOM, 5-7% NONPLASTIC<br>FINES, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, TRACE OF GRAVEL, BROWN.        |
| 40.0                     | 85-             | S                     | 26               | 27-23-22<br>(14")              | 55                 |                     | WEATHERED SANDSTONE FRAGMENTS TO 14 IN, WITH COARSE TO FINE SAND, BROWN<br>AND LIGHT GRAY, (TRACE OF NONPLASTIC FINES).                                                                                                                                                                       |
|                          | 90              | 1                     |                  |                                |                    |                     |                                                                                                                                                                                                                                                                                               |

|                          |                 |                    |                  |                                |                    |                     | BORING NO. SEC-4<br>SHEET OF                                                                                                          |
|--------------------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------|
| SI                       | TE <u>BE</u>    | AVER               | VALLI            | ey power s                     | TATIC              | N - (               | J.O. NO. 12241                                                                                                                        |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RGD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                    |
|                          |                 | s                  | 27               | 18-13-12                       | 25                 | eu                  | TOP 3 IN, GRAVELLY SAND, 15-202 COARSE TO FINE GRAVEL, FEW SANDSTONE                                                                  |
|                          |                 |                    |                  | (11")                          |                    |                     | FRAGMENTS TO 14 IN, COARSE TO FINE SAND, LESS THAN 52 NONPLASTIC FINES, BROWN.                                                        |
|                          |                 |                    |                  |                                |                    | SP                  | BOTTOM 8 IN, <u>SAND</u> , COARSE TO FINE, MOSTLY COARSE TO MEDIUM, LESS THAN 52<br>FINE GRAVEL, LESS THAN 52 NONPLASTIC FINES, GRAY. |
|                          | 95-             | s                  | 28               | 12-12-11                       | 23                 | SP                  | GRAVELLY SAND, 30-40% COARSE TO FINE ROUNDED TO ANGULAR GRAVEL, FEW                                                                   |
| 0.0                      | _               |                    | 1                | (9")                           |                    |                     | SANDSTONE FRAGMENTS TO 14 IN, COARSE TO FINE SAND MOSTLY COARSE TO<br>MEDIUM. GRAY.                                                   |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 100             | s                  | 29               | 15-18-32                       | 50                 | SP                  | SAND, 5-7% COARSE TO FINE GRAVEL, FEW SANDSTONE AND COAL FRAGMENTS,                                                                   |
| 3.3                      |                 | s                  | 30               | (12")<br>50/.1'                | 50/.1              | Į                   | SOFT THINLY BEDDED GRAY SILTSTONE.                                                                                                    |
|                          | 105             |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     | END OF BORING AT 103.1 FT.                                                                                                            |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 1               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | I               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | _               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 1               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 1 1             |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 1               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    | l                | [                              |                    |                     |                                                                                                                                       |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          | 1 1             |                    |                  |                                | ŀ                  |                     |                                                                                                                                       |
|                          | 1 1             |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                | .<br>              |                     |                                                                                                                                       |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    | l                |                                |                    |                     |                                                                                                                                       |
|                          |                 |                    |                  |                                |                    | 1                   |                                                                                                                                       |
|                          |                 | ч <u>.</u>         |                  |                                |                    |                     |                                                                                                                                       |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                       |
| TE: F                    | FOR BO          | RING               | SUMM             |                                |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE N BORNS NO. SHEET                                                                               |

| s            | SITE            | BEAVE                                         | R VAL        | LEY PO             | ER STA      | TION        | - UNIT 1                           |                             | J.O. NO1                            | 2241                       | BORING NO.                        | SE0-5_           |
|--------------|-----------------|-----------------------------------------------|--------------|--------------------|-------------|-------------|------------------------------------|-----------------------------|-------------------------------------|----------------------------|-----------------------------------|------------------|
| . c          | OORDI           | NATE                                          | 5 _ <u>`</u> | 13682 .79          | I           | <u> </u>    | 9320.89                            | GROUND                      | ELEV. (I)                           | .2                         |                                   | ' <u></u>        |
| H            | NCLINA          | TION                                          |              |                    |             | . 88        |                                    |                             | INSPECTOR                           | JWMcCare                   |                                   |                  |
| D            | ATE :           | STAR                                          | T/F          | INISH              | 10/10       | /81         | / _10/10/81                        | CONTRAC                     | TOR / DRILLER                       | EGER DR                    | ILLING/JARVIS                     | <b></b>          |
| S            | TATIC           | GRO                                           | UNDV         | VATER              | DEPT        | H/D         | ATE                                | /                           | DRILL RIG T                         | YPE                        |                                   |                  |
| D            | EPTH            | то                                            | BEDR         | OCK _              | 104.0       | <del></del> | <u>(FT)</u>                        | TOTAL DE                    | PTH DRILLED                         | 104.5                      |                                   | <u>(FT)</u>      |
| N            | ETHO            | )S :                                          |              |                    |             |             |                                    |                             |                                     |                            |                                   |                  |
|              | DF              |                                               | ig s         | ioll.              | AW ROD      | <u>s, 3</u> | CRITT BARDEN                       | DRILLING MUL                | AND CASING                          |                            |                                   |                  |
|              | DF              | AMPCI<br>All Lik                              | NG R         | OCK 2              | N/A         | <u> </u>    | <u></u>                            |                             |                                     |                            |                                   |                  |
| S            | PECIA           | TE:                                           | STIN         | GORI               | NSTR        | UMEN        |                                    |                             |                                     |                            |                                   |                  |
|              |                 |                                               |              |                    |             |             |                                    |                             |                                     |                            |                                   |                  |
| C            | OMME            | NTS _                                         | HOL          | E_CAVED,           | UNABL       | <u>E_TO</u> | OBTAIN GROUNDWAT                   | TER LEVEL                   |                                     |                            |                                   |                  |
|              | _               |                                               | FIL          | L TO APP           | ROXIMA      | TELY        | 40 FT                              |                             |                                     |                            |                                   |                  |
|              |                 |                                               |              |                    |             |             |                                    |                             |                                     |                            |                                   |                  |
| 162)<br>162) |                 | Ē                                             |              | (î)                |             | 9           |                                    |                             |                                     |                            |                                   |                  |
| ATIO         | EE T            |                                               | APLE<br>ABER | ) SM<br>(          |             | 2<br>2<br>2 |                                    | SAI                         |                                     | PTION                      |                                   |                  |
| 1<br>1<br>1  | <u> </u>        | I SAI                                         | SAN          |                    | g A         | SYNK GR     |                                    |                             |                                     |                            |                                   |                  |
| 4            | l               |                                               |              |                    |             |             | . <u>.</u>                         | ,<br>                       |                                     |                            |                                   |                  |
|              |                 | · · · · · ·                                   |              |                    |             |             |                                    |                             |                                     |                            |                                   | ,                |
| 7.2          |                 | S                                             | 1            | 26-22-2            | 0 42        |             | ROAD FILL, SILT                    | TY SAND, 10                 | GRAVEL, SLAG AN                     | D SANDSTONE                | FRAGMENTS, BRO                    | WN AND           |
|              | -               |                                               |              | (11")              |             |             | GRAY.                              |                             |                                     |                            |                                   | -                |
|              | 5 -             | s                                             | 2            | 11-7-5<br>(13")    | 12          | GM          | GRAY.                              | ELLY SILT,                  | U-30% COARSE TO                     | FINE GRAVE                 | L, 5% FINE SAN                    | D, LIGHT -       |
| -            | -               | S                                             | 3            | 2-1-2              | 3           | SP          | MENTS, COARSE 1                    | RAVELLY SAN<br>TO FINE SAN  | 2, 10-20% COARS<br>), MOSTLY MEDIUM | E TO FINE G<br>TO FINE, 5  | RAVEL, FEW LARG<br>X SILT, BROWN. | GE FRAG-         |
| 20.0         | -               |                                               |              | (18")              |             | GM          | GRAVELLY SILT,                     | 20-25% COAL                 | SE TO FINE GRAV                     | EL, ROUNDEL                | TO SUBANGULAR                     | , FEW            |
|              | 10 _            | s                                             | 4            | 3-4-4              | . 8         | GW          | SANDY GRAVEL,                      | COARSE TO F                 | NE GRAVEL, ROUN                     | DED TO SUBA                | NGULAR, FEW FR                    | AGMENTS          |
|              | -               | s                                             | 5            | 1-2-3              | 5           |             | TO 15 IN, 20-30<br>PLASTIC TO SLIC | DX COARSE TO<br>GHTLY PLAST | ) FINE SAND, MOS<br>IC FINES, BROWN | TLY MEDIUM                 | TO FINE, 10-152                   | K NON-           |
|              | -               |                                               | Ĭ            | (11")              |             | SP          | GRAVELLY SAND,                     | 20-307 COA                  | SE TO FINE GRAV                     | EL, ROUNDEL<br>7 NONDIASTI | TO SUBANGULAR                     | , COARSE         |
|              | 15 _            | s                                             | 6            | 5-4-2              | 6           | C.D.        | FINES, BROWN.                      |                             |                                     |                            |                                   |                  |
|              | -               | s                                             | ,            | 3-4-2              | 6           | Sr<br>ev    | SAME AS ABOVE.                     | -307 COAPSE                 | TO FINE SAND 5                      | T COARSE TO                | FINE CRAVET                       | -                |
| 10.0         | -               | ١ <u>ــــــــــــــــــــــــــــــــــــ</u> |              | (8")               | ľ           | Srt .       | SANDSTONE FRAG                     | MENTS TO 14                 | IN, NONPLASTIC                      | TO SLIGHTLY                | PLASTIC, BROW                     | N                |
|              | 20              | s                                             | 8            | 2-4-6              | 10          | SM          | SILTY FINE SAN                     | D, 5-7% COA                 | SE TO FINE GRAV                     | EL, FEW FRA                | GMENTS TO 1-11                    | , 15-20 <b>%</b> |
|              |                 | 5                                             | ,            | 2-2-2              | 4           | SM          | SAME AS ABOUT                      | GRAY                        |                                     | NATION DROW                | *** •                             |                  |
|              | -               | Ē                                             |              | (18")              |             |             |                                    | -1417 1                     |                                     |                            |                                   | -                |
|              | 25 _            | s                                             | 10           | 2-2-2<br>(5")      | 4           | SM          | SAME AS ABOVE.                     |                             |                                     |                            |                                   | -                |
|              | -               | <br>S                                         | <u>11</u>    | 2-3-3              | 6           | SM          | SAME AS ABOVE                      |                             |                                     |                            |                                   | -                |
| 0.00         | -               |                                               |              | (10")              |             | SM          | STMTI AP TO APO                    | עד כווייע די                | NE SAND HOOD                        |                            | TTC TDACE OF 4                    | -                |
|              | 30              |                                               | 12           | 3-5-7<br>(6")      |             |             | GRAY.                              | , <u>silli f</u>            |                                     | NOUIS, PLAS                | TIC, INALE OF (                   | JANVEL, 🕳        |
| 1.           |                 | IS ME                                         | EAN S        | BEA LEV            | EL          | 7.          | S-SPLIT BARRI                      | EL SAMPLE                   |                                     | BOBING                     | 106                               |                  |
| 3.           | BLOWS           | REQU                                          | IRED         | TO DRI             | VE          |             |                                    |                             |                                     |                            | DOWER C                           |                  |
|              | 2'O.D. S        | ANPLE<br>CE SH                                | SPC<br>OWN   | USING              | OR          |             |                                    |                             | DEAVER                              |                            | TOWERS                            | ATION            |
|              | 14015. H        | AMMER<br>CATES                                | FAL          | LING 30            | 0".<br>)16. |             |                                    |                             |                                     |                            |                                   |                  |
|              | HAMME           | R. (<br>E <b>rec</b> i                        | ) INCI       | HES OF             |             |             |                                    |                             |                                     | SNE LI                     | GHT COMP                          |                  |
| 4.           | % ROC<br>ROCK C | K COP                                         | RE RI        | ECOVERY<br>ESIGNAT | /<br>KONL   |             |                                    |                             | SHIPPIN                             | GPORT,                     | PENNSYL                           | VANIA            |
| 5.           | STD. P          |                                               | ATIO         | N<br>S/FT          |             |             |                                    |                             | SKET                                | CH NO. 12                  | 241-GSK-239A                      | ·                |
| <b>1</b> 6.  | UNIFIED         | SOIL                                          | CL/          | SSIFICA            | TION        |             |                                    |                             | APPROVED                            | DATEM                      | BORING NO. SI                     | EET              |

|                        |                 |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 | BORING NO.                                      | SEO-5                           |
|------------------------|-----------------|-------------------|------------------|-------------------------|--------------------|--------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|---------------------------------|
|                        |                 |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 | SHEET 2 C                                       | )F                              |
| SI                     | TE              | BEAV              | ER_VA            | LLEY POWER              | STAT               | TION -             | - UNIT 1 J.O. NO. 12241                                                                                                                                                                                         |                                                 |                                 |
| - 2                    |                 | 3                 |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 |                                 |
| ELEVATION<br>(FEET)(IE | DEPTH<br>(FEET) | SAMPLE (          | SAMPLE<br>NUMBER | BLOWS (3)<br>REC/RGD (4 | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6 | SAMPLE DESCRIPTION                                                                                                                                                                                              |                                                 |                                 |
|                        |                 |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 |                                 |
|                        |                 |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 | l                               |
|                        |                 | s<br>s            | 13<br>14         | 5-5-4<br>(10")<br>7-7-5 | 9<br>12            | SM<br>GW           | TOP 4 IN, SAME AS ABOVE.<br>BOTTOM 6 IN, <u>SANDY CRAVEL</u> , COARSE TO FINE GRAVEL, R<br>FEW FRAGMENTS TO 1 IN, 20-30% COARSE TO FINE SAND,<br>MEDIUM, 5% NONPLASTIC FINES, BROWN.                            | OUNDED TO SUB<br>MOSTLY COARSE                  | ANGULAR,<br>TO                  |
|                        | 35              |                   |                  | (9")                    |                    | SM                 | TOP 4 IN, SILTY FINE SAND, 15-20% NONPLASTIC TO S                                                                                                                                                               | LIGHTLY PLAST                                   | IC FINES,                       |
| 690.0                  |                 | S                 | 15               | 5-7-11<br>(11")         | 18                 | sw                 | BOTTOM 5 IN, <u>GRAVELLY SAND</u> , 20-30% COARSE TO FINE G<br>FRAGMENTS TO 1 IN, COARSE TO FINE SAND, 5% NONPLAST                                                                                              | RAVEL, FEW SA<br>IC FINES, BRO                  | NDSTONE                         |
|                        | 40 —            | S                 | 16               | 5-6-11<br>(15")         | 17                 | SM                 | SILTY FINE SAND, 15-20% NONPLASTIC TO SLIGHTLY PLAS<br>STONE FRAGMENTS TO 3/4 IN, TRACE OF GRAVEL, TRACE C                                                                                                      | TIC FINES, FE<br>F ROOTS, BROW                  | W SAND-                         |
|                        |                 | S                 | 17               | 14-25-20<br>(16")       | 45                 | SM                 | SILTY FINE SAND, 5-7% COARSE TO FINE GRAVEL AND COA<br>STONE FRAGMENTS TO 14 IN, 10-15% NONPLASTIC TO SLIC<br>GRAY AND BROWN.                                                                                   | L FRAGMENTS,<br>HTLY PLASTIC                    | FEW SAND-<br>FINES,             |
|                        | 45              | s                 | 18               | 6-8-9<br>(9")           | 17                 | SM                 | SILTY FINE SAND, 5% COARSE TO FINE GRAVEL, MOSTLY N<br>SANDSTONE FRACMENTS, 15-20% NONPLASTIC FINES, TRACE<br>FRACMENTS, BROWN AND GRAY.                                                                        | EDIUM TO FINE<br>OF COAL AND                    | WOOD                            |
| 680.0                  |                 | S                 | 19               | 6-10-15<br>(15")        | 25                 | SM                 | SIMILAR TO ABOVE, ROOTS AND WOOD, DARK GRAY.                                                                                                                                                                    |                                                 | -                               |
|                        |                 | s                 | 20               | 8-7-8                   | 15                 | SM<br>CM           | SIMILAR TO ABOVE, 5-10% COARSE TO FINE GRAVEL, BROW                                                                                                                                                             | N.                                              |                                 |
|                        | 50 -            |                   |                  | (13")                   |                    | 50                 | SLIGHTLY PLASTIC, TRACE OF COAL FRAGMENTS (CINDERS)                                                                                                                                                             | ) OIL SME                                       | L, GRAY.                        |
|                        |                 | s                 | 21               | 11-10-8<br>(5")         | 18                 | <b>ମ୍ଲ</b>         | GRAVELLY SILT, 20-25% COARSE TO FINE GRAVEL, LARGE<br>BOTTOM, 5-7% COARSE TO FINE SAND, SLIGHTLY PLASTIC                                                                                                        | SANDSTONE FRA                                   | AGMENT AT -                     |
|                        | 55 -            | \$                | 22               | 7-11-10<br>(14")        | 21                 | GM                 | GRAVELLY SILTY SAND, 10-15% COARSE TO FINE GRAVEL,<br>SHALE FRAGMENTS, 20-20% SLIGHTLY PLASTIC FINES, SL                                                                                                        | GHT OIL SMEL                                    | , GRAY.                         |
| 670.0                  |                 | S                 | 23               | 9-6-9<br>(9")           | 15                 | େମ୍ପ               | SILTY GRAVEL, COARSE TO FINE GRAVEL, FEW LARGE FRA<br>FINE SAND, 15-20% SLIGHTLY PLASTIC FINES, BROWN.                                                                                                          | ments, 10-13                                    | COARSE TO                       |
|                        | 60              | S                 | 24               | 7 <b>-6-8</b><br>(13")  | 14                 | <b>ଜ୍</b> ୟ        | SILTY GRAVEL, COARSE TO FINE GRAVEL-SIZED SANDSTON<br>ALL COLORS, RED, BROWN, GRAY, 5-10% COARSE TO FINE<br>SLIGHTLY PLASTIC TO MODERATELY PLASTIC FINES, GRA                                                   | E AND SHALE F<br>SAND, 15-20%<br>(.             | RACHENTS, —<br>—                |
|                        |                 | <u>.</u> <u>S</u> | 25               | 12-11-8<br>(12")        | 19                 | ଫ                  | SANDY GRAVEL, WEATHERED SANDSTONE AND SHALE, 8-102<br>PLASTIC FINES,15-202 COARSE TO FINE SAND, BROWN.                                                                                                          | NONPLASTIC T                                    | O SLIGHTLY-                     |
|                        | 65 -            | S                 | 26               | 9-7-9<br>(9")           | 16                 | CM                 | SAND, 5-7% COARSE TO FINE GRAVEL, ROUNDED TO ANGUL<br>SAND, 5-7% NONPLASTIC FINES, BROWN.                                                                                                                       | AR, COARSE TO                                   | FINE                            |
| 660.0                  |                 |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 |                                 |
|                        | ]               |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 | -                               |
|                        | 70-             | S                 | 27               | 9-10-10<br>(12")        | 20                 | G₩                 | SANDY GRAVEL, COARSE TO FINE GRAVEL-SIZED SANDSTON<br>SOME TO 14 IN, ANGULAR TO SUBANGULAR, 20-25% COARS<br>NONPLASTIC TO SLIGHTLY PLASTIC FINES, BROWN. (FEW<br>AND ORANGE).                                   | E AND SHALE F<br>E TO FINE SAN<br>SHALE FRAGMEN | RAGMENTS,<br>D, 5-8%<br>TS, RED |
| 650.0                  | 75              | S                 | 28               | 14-14-10<br>(15")       | 24                 | SP                 | GRAVELLY SAND, 20-25% COARSE TO FINE GRAVEL-SIZED<br>FRACMENTS, ALL COLORS, COARSE TO FINE SAND, MOSTLY<br>5-7% NONPLASTIC FINES, FEW SANDSTONE FRACMENTS TO                                                    | SANDSTONE AND<br>MEDIUM TO FI<br>1 IN, BROWN.   | SHALE<br>NE,                    |
|                        | 80-             |                   |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 | COADER TO                       |
|                        |                 | S                 | 29               | 8-10-15<br>(14")        | 25                 | SP<br>GW           | TOP 5 IN, <u>SAND</u> , 5% COARSE TO FINE GRAVEL, 5-7% NON<br>FINE SAND, MOSTLY FINE, BROWN.<br>BOTTOM 9 IN, <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL-S<br>SHALE FRAGMENTS, 15-20% COARSE TO FINE SAND, 5-7% | IZED SANDSTON                                   | E AND                           |
|                        | 85              |                   |                  |                         |                    | ~.                 | SLIGHTLY PLASTIC FINES, FEW SANDSTONE FRACMENTS TO                                                                                                                                                              | 1 IN, BROWN.                                    | -                               |
|                        |                 | s<br>             | 30               | 19-15-18<br>(9")        | 33                 | GW                 | SIMILAR TO ABOVE, <u>SANDY GRAVEL</u> .                                                                                                                                                                         |                                                 | -                               |
| 640.0                  | -               | •.                |                  |                         |                    |                    |                                                                                                                                                                                                                 |                                                 | -                               |
|                        | 90              |                   |                  |                         |                    | STO                | NE & WERSTER ENG CORP LADOROUTO DATE                                                                                                                                                                            |                                                 | SHEET                           |
| NOTE:                  | LEGEND          | NFQ.              | SEE              | SHEET L                 |                    | SKE                | TCH No. 12241-GSK-239B DD+ /12/8=                                                                                                                                                                               | SEO-5                                           | 3 07 3                          |

|                          |       |            |             |              |             |                    |                     | BORING NO. SEC<br>SHEET_3_OF                                                                                                                                          | -5  |
|--------------------------|-------|------------|-------------|--------------|-------------|--------------------|---------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|
| SI                       | TE    | BE/        | VER V       | ALLEY        | POWE        | ER STA             | TION                | - UNIT 1                                                                                                                                                              |     |
| ELEVATION<br>(FEET)(162) | DEPTH | SAMPLE TY  | SAMPLE      | BLOWS (3)    | REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                    |     |
|                          |       |            | <del></del> | r            |             |                    |                     |                                                                                                                                                                       |     |
|                          |       |            | 31          | 13-1<br>(12" | 0-12<br>'}  | 22                 | SW                  | GRAVELLY SAND, 15-252 COARSE TO FINE GRAVEL, ROUBLED TO SUBARGULAR<br>FEW PIECES TO 3/4 IN, COARSE TO FINE SAND, 5-72 NONPLASTIC TO<br>SLIGHTLY PLASTIC FINES, BROWN. | ]   |
|                          | 95 -  | 4          |             |              |             |                    |                     |                                                                                                                                                                       | 1   |
|                          |       |            | 32          | 8-9-<br>0-15 | -14<br>T    | 23                 | SW                  | NO RECOVERY FIRST ATTEMPT.                                                                                                                                            | -   |
| 630.0                    | Į     | 4          |             | 2"-          | 2ND         |                    |                     | SAME AS ABOVE                                                                                                                                                         | 4   |
|                          | 100 • |            | - 1         | 12-1         | 1-1         | 22                 | รษ                  | SAME AS ABOVE.                                                                                                                                                        | Ⅎ   |
|                          |       | ]          | 1″          | (10          | ')          |                    |                     |                                                                                                                                                                       | _   |
| 622.7                    |       | - <u>s</u> | =_ 34       | <u>-1'</u>   |             | <u>.1'</u>         | <br>                | SOFT THINLY BEDDED GRAY SILTSTONE.                                                                                                                                    |     |
|                          |       | Ξ          |             |              |             |                    |                     | FWD OF ROPING 104 5 FT                                                                                                                                                | -   |
|                          |       | -          |             |              |             |                    |                     |                                                                                                                                                                       | _   |
|                          |       | -          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 7          |             |              |             |                    |                     |                                                                                                                                                                       | ]   |
|                          |       | 7          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 7          |             |              |             |                    |                     |                                                                                                                                                                       | -   |
|                          |       | 7          |             |              |             |                    |                     |                                                                                                                                                                       | 7   |
|                          |       | 1          |             |              |             | ļ                  |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       | 1   |
|                          |       | 1          |             |              |             |                    | ļ .                 |                                                                                                                                                                       | 11  |
|                          |       |            |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             | ļ                  |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       | -   |
|                          | ŀ     |            |             |              |             | l                  | <b>.</b>            |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          | .     | 4          |             |              |             | 1                  |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 1          |             |              |             | l                  |                     |                                                                                                                                                                       |     |
|                          | .     | -          |             |              |             |                    |                     |                                                                                                                                                                       | ļ   |
|                          |       | -          |             |              |             |                    |                     |                                                                                                                                                                       |     |
|                          |       | 7          |             |              |             |                    |                     |                                                                                                                                                                       |     |
| OTE:                     | FOR E | BORING     | SUMM        |              | AND         |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE M BORNS NO. SHEL                                                                                                                | IT. |

· · · ·

|             | OORDI                                       | NATE                                 | s _           | N3843              |              |          | E6223 GROUND ELEV. (I) _741.0 PT SHEETOF                                                                                                       | <u>_</u> |
|-------------|---------------------------------------------|--------------------------------------|---------------|--------------------|--------------|----------|------------------------------------------------------------------------------------------------------------------------------------------------|----------|
| IN          |                                             | TION                                 | VER           | TICAL              |              | 8E       |                                                                                                                                                | _        |
| D           | ATE :                                       | STAR                                 | T/F           | INISH _            | 6-4-8        | 2        | / 6-7-82 CONTRACTOR / DRILLEREGER/JARVIS                                                                                                       |          |
| S           | TATIC                                       | GRO                                  | UNDV          |                    | DEPTI        | H / D.   | AT ERECORDED(FT) / DRILL RIG TYPE CHE 45                                                                                                       |          |
| D           | EPTH                                        | то                                   | BEDR          | юск                | 52           | .0       | (FT) TOTAL DEPTH DRILLED 52.0 (FT)                                                                                                             | n        |
| M           | ETHOD                                       | <b>S</b> :                           |               |                    |              |          |                                                                                                                                                |          |
|             | DF                                          |                                      | NG S          | <b>IOL</b> _       | 3-1/8        | IN RC    | DLLER BIT, 3-1/4 IN I.D. CASING, WATER                                                                                                         | -        |
|             | S                                           | MPL                                  | ING :         | sol _2             | 2 IN O       | .D. S    | SPLIT SPOON                                                                                                                                    | -        |
|             | Df                                          | RILLI                                | NG R          | оск _              | IONE         |          |                                                                                                                                                | -        |
| S           | PECIA                                       | LTE                                  | STIN          | g or in            | ISTRU        | JMEN     | TATION 2 FT POROUS STONE PIEZOMETER INSTALLED WITH TIP AT EL 718                                                                               | -        |
| _           | _                                           |                                      |               |                    |              | ***      |                                                                                                                                                |          |
| C           | OMME                                        | NTS 1                                | NO            | NE                 |              | <u> </u> |                                                                                                                                                |          |
|             | -                                           |                                      |               |                    |              |          |                                                                                                                                                |          |
|             |                                             |                                      |               |                    |              |          |                                                                                                                                                |          |
| N<br>(162)  | - 2                                         | Ē                                    |               | (F) (F)            | 6            | (9)      |                                                                                                                                                |          |
| ET)(        | :PTH                                        | L H                                  | IPL 6         | NS (<br>Vor        | E N<br>L     | ₽<br>2   |                                                                                                                                                |          |
| N B         | DE<br>FI                                    | SAL                                  | SAN           | ANG                | AL N         | Ϋ́Υ.     |                                                                                                                                                |          |
| ū           |                                             |                                      |               | RE L               |              |          |                                                                                                                                                |          |
|             |                                             |                                      |               |                    |              |          |                                                                                                                                                |          |
| 741.0       | 0 .                                         | s                                    | 1             | 1-3-5              | 8            | ML       | TOP 6 IN: SANDY SILT, DENSE, 10% FINE GRAVEL TO 3/8 IN, ANGULAR, 15-2                                                                          | 07       |
| 1           | -                                           |                                      |               | (12")              |              |          | COARSE TO FINE SAND, CONTAINS ROOTS AND ORGANIC MATTER, VERY SLIGHTL<br>MOIST, DARK BROWN AND BLACK.                                           | Y        |
|             |                                             |                                      |               |                    |              | CL       | BOTTOM 6 IN: SANDY CLAY, SLIGHTLY PLASTIC, STIFF, OCCASIONAL FINE GRA<br>12-15Z COARSE TO FINE SAND, ANGULAR, VERY SLIGHTLY MOIST, LIGHT BROWN | VEL      |
|             | -                                           | s                                    | 2             | 4-13-9             | 22           | CL       | SIMILAR TO S-1, BOTTOM 6 IN.                                                                                                                   |          |
|             |                                             | ┨──                                  |               | (12")              |              |          |                                                                                                                                                |          |
|             | -                                           | s                                    | 3             | 1-5-6              | 111          | CL       | SIMILAR TO S-1, BOTTOM 6 IN, GRAY BROWN.                                                                                                       |          |
|             | 5 -                                         | ł                                    |               | (12")              |              |          |                                                                                                                                                |          |
|             |                                             | s                                    | 4             | 5-6-8              | 14           | CL       | SILTY CLAY, MODERATELY PLASTIC, STIFF, 2% FINE SAND, SLIGHTLY MOIST, 1                                                                         | BRO      |
|             | -                                           | -                                    |               | (18")              |              |          | POCKETS OF SANDY CLAY WITH SOME COARSE AND MEDIUM SAND, TRACE SUBANGUL                                                                         | AR       |
|             |                                             |                                      |               |                    |              | -<br>-   | GRAVEL TO 0.5 IN MAXIMUM.                                                                                                                      |          |
|             | -                                           |                                      | )             | (16")              | 1            | ML       | SILI CLAI-CLAIEI SILI, SLIGHILI PLASTIC, MEDIUM STIFF, MUIST, BROWN.                                                                           |          |
|             |                                             | s                                    | 6             | 6-6-5              | 11           | ML       | SILT, NONPLASTIC TO SLIGHTLY PLASTIC, 5% VERY FINE SAND, MOIST, BROWN                                                                          | ۱.       |
|             | . •                                         | -                                    |               | (18")              |              |          |                                                                                                                                                |          |
| /31.0       | 10 -                                        | s                                    | 7             | 2-3-6              | 9            | ML       | SIMILAR TO S-6.                                                                                                                                |          |
|             | -                                           | 1                                    |               | (1 <b>3</b> ")     |              |          |                                                                                                                                                |          |
|             |                                             | s                                    | 8             | 4-5-5              | 10           | ML       | SIMILAR TO S-6, CONTAINS OCCASIONAL 5mm FINE SAND LENS.                                                                                        |          |
| ÷ .         | -                                           | 1                                    |               | (1/")              |              |          |                                                                                                                                                |          |
|             |                                             | s                                    | 9             | 6-7-9              | 16           | мг       | TOP 7 IN: <u>SANDY SILT</u> , NONPLASTIC TO SLIGHTLY PLASTIC, 30-40% FINE SAN                                                                  | nD,      |
|             |                                             | 1                                    |               |                    |              | SM       | BOTTOM 8 IN: SILTY SAND, UNIFORM, FINE, 10-15% NONPLASTIC FINES, BROW                                                                          | n.       |
| ╺╌┰──┦      | 15                                          | S                                    | 10            | <u> </u>           | 1            | L        | L                                                                                                                                              |          |
| 1.          |                                             | IS N                                 | EAN           | SEA LEV<br>R LEVFI | EL           |          | UNDISTURBED SAMPLES<br>US-SHELBY TUBE RORING LOG                                                                                               |          |
| 3.          | BLOWS                                       | REQ                                  | UIRED         | TO DRIV            | /E           |          | UO-OSTERBERG                                                                                                                                   |          |
| a i         | 2"0.D. S                                    | SAMPLICE S                           | .E SP<br>Hown | OON 6" 0<br>USING  | R            |          | BEAVER VALLEY POWER STATION U                                                                                                                  | NI.      |
| ŭ į         | 14016. H                                    |                                      | R FA          | LLING 30           | / <b>!</b> . |          | DUQUESNE LIGHT COMPANY                                                                                                                         | ,        |
|             |                                             |                                      |               | personal das las   |              |          | SHIPPINGPORT PENNSYLVANI                                                                                                                       | A        |
| NOTE:       | RECOV                                       | ERY.                                 |               |                    | -            | •        |                                                                                                                                                |          |
| 10 / NOTE:  | RECOV<br>STD. F<br>BLOWS                    | ERY.<br>ENET<br>/FT.                 | RATI          | ON RESIS           | TANCE        |          |                                                                                                                                                |          |
| END / NOTE: | RECOV<br>STD. F<br>BLOWS<br>UNIFIE<br>SYSTE | ERY.<br>PENET<br>/FT.<br>D SO!<br>M. | RATIO         | ON RESIS           | TANCE        | :        | STONE & WEBSTER ENG. CORP.                                                                                                                     |          |

|                          |                                                                              |                       |        |                             |                    |                     | BORING NO. <u>B05-1</u>                                                                                                                                                                      | -               |  |  |  |  |  |  |
|--------------------------|------------------------------------------------------------------------------|-----------------------|--------|-----------------------------|--------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------|--|--|--|--|--|--|
| S                        | SITE BEAVER VALLEY POWER STATION-UNIT 2, SHIPPINGPORT, PA. J.O. NO. 12241.00 |                       |        |                             |                    |                     |                                                                                                                                                                                              |                 |  |  |  |  |  |  |
| ELEVATION<br>(FEET)(162) | DEPTH                                                                        | (FEET)<br>SAMPLE      | SAMPLE | REC/ROD (4)                 | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                           |                 |  |  |  |  |  |  |
|                          | 15                                                                           |                       |        |                             | <u> </u>           | <b>—</b>            |                                                                                                                                                                                              | ╘               |  |  |  |  |  |  |
|                          |                                                                              |                       | - 10   | 6-5-7<br>(18")              | 12                 | ML-<br>Sh           | LAYERED SILT AND SILTY FINE SAND, SLIGHTLY PLASTIC FINES, CLAYEY SILT<br>CONTAINING COARSE TO FINE GRAVEL SIZED ROCK FRAGMENTS AT BOTTOM.                                                    | -               |  |  |  |  |  |  |
|                          |                                                                              | <b>]</b> <sup>s</sup> | 11     | (14")                       | 12                 | SM<br>ML            | TOP 2 IN: <u>Silty sand</u> , fine, few fine gravel.<br>Bottom 12 IN: <u>Silt</u> , nonplastic to very slightly plastic, moist, brown.                                                       | -               |  |  |  |  |  |  |
|                          |                                                                              | <b>J</b> s            | 12     | 3-5-5<br>(14")              | 10                 | ML.                 | SIMILAR TO S-11, BOTTOM 12 IN, CONTAINS FINE SAND LENSES ABOUT 2 = THICK.                                                                                                                    | 1               |  |  |  |  |  |  |
| 721.0                    | 20                                                                           | s                     | 13     | 4-2-3<br>(12")              | 5                  | SM<br>ML            | TOP 10 IN: <u>Silty Sand</u> , Fine, 10–15% Nonplastic Fines, Brown.<br>Bottom 2 In: <u>Silt</u> , Slightly Plastic, Brown.                                                                  |                 |  |  |  |  |  |  |
|                          |                                                                              | - s                   | 14     | 3-1-6<br>(17")              | 7                  | SM                  | TOP 8 IN AND BOTTOM 1 IN: <u>SILTY SAND</u> , FINE, 10-15% NONPLASTIC FINES, WET<br>ORANGE-BROWN.                                                                                            |                 |  |  |  |  |  |  |
|                          |                                                                              |                       | -      | 3-4-4                       | 8                  | ML<br>ML-           | MIDDLE 8 IN: <u>Silt</u> , Slightly Plastic, Gray-Brown.<br>Top 8 IN: <u>Layered Sandy Silt and Silty Fine Sand</u> , Nonplastic Fines, Brown.                                               | <b>ب</b>        |  |  |  |  |  |  |
|                          |                                                                              |                       | -      | (17")                       | 5                  | SM<br>SP            | BOTTOM 9 IN: <u>SILTY SAND</u> , FINE, 10-15% NONPLASTIC FINES, BROWN.                                                                                                                       | -               |  |  |  |  |  |  |
|                          |                                                                              | 7                     |        | (17")                       |                    |                     |                                                                                                                                                                                              | Ξ               |  |  |  |  |  |  |
|                          | 25                                                                           | - s                   | 17     | 2-2-2<br>(15")              | 4                  | SP                  | SIMILAR TO S-16.                                                                                                                                                                             |                 |  |  |  |  |  |  |
|                          |                                                                              | -                     | 18     | 1-5-6<br>(18")              | 11                 | SM<br>GP            | OF 12 IN: <u>SIMILAR TO S-16</u> .<br>IIDDLE 2 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED WEATHERED SHALE<br>RAGMENTS, ANGULAR.                                                   |                 |  |  |  |  |  |  |
|                          |                                                                              |                       | 19     | 3-4-3                       | 7                  | SP<br>SP            | BOTTOM 4 IN: <u>Sand</u> , Uniform, Fine, Moist, Brown.<br>Top 6 IN: <u>Sand</u> , Fine, Trace Silt, Brown.                                                                                  | -               |  |  |  |  |  |  |
|                          |                                                                              | ┨_                    |        | (13-)                       | Ļ                  | GP-<br>GW           | BOTTOM 7 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, 1 IN MAXIMUM, ANGULAR TO ROUNDI<br>20-30% COARSE TO FINE SAND, BROWN.                                                                     | <sup>ED</sup> , |  |  |  |  |  |  |
| /11.0                    | 30                                                                           | ]                     |        | (9")                        | Ľ                  | GP                  | NONPLASTIC FINES.<br>BOTTOM 4 IN: <u>GRAVEL</u> , COARSE TO FINE, 1 IN MAXIMUM, ANGULAR TO ROUNDED,<br>TRACE SAND, <u>WET.</u> GRAV AND BROWN, OFCANIC OILY SWELL AND FREE                   | -               |  |  |  |  |  |  |
|                          |                                                                              |                       | 21     | 5-3-3<br>(5")               | 6                  | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE, 1.5 IN MAXIMUM, ANGULAR TO ROUNDED, 15-202<br>COARSE TO FINE SAND, 5-82 NONPLASTIC FINES, TRACE IRON STAINING, BROWN,<br>CRAY OR MARKE                         | 11              |  |  |  |  |  |  |
|                          |                                                                              | - s                   | 22     | 4-3-5<br>(6")               | 8                  | GP<br>GW            | SIMILAR TO S-21.                                                                                                                                                                             |                 |  |  |  |  |  |  |
|                          | 35                                                                           | - s                   | 2:     | 5-5-5<br>(4")               | 10                 | GP-<br>GW           | SIMILAR TO S-21.                                                                                                                                                                             | -               |  |  |  |  |  |  |
| 1                        |                                                                              |                       | 24     | 7-4-5<br>(13")              | 9                  | GP                  | TOP 5 IN: <u>Sandy Gravel</u> , coarse to fine gravel sized sandstone fragments to<br>1.5 IN Maximum, 10–15% coarse to fine sand, less than 5% nonplastic fines<br>gray.                     | 。<br>-<br>-     |  |  |  |  |  |  |
|                          |                                                                              | 1                     |        |                             |                    | GP<br>GW            | BOTTOM 8 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ROUNDED, 20-30% COARSE TO FINE<br>SAND, LESS THAN 5% NONPLASTIC FINES, TRACE IRON STAINS, BROWN.                                          | -               |  |  |  |  |  |  |
|                          |                                                                              |                       | 2      | 9-8-13<br>(0 <sup>1</sup> ) | 21                 |                     | NO RECOVERY.                                                                                                                                                                                 |                 |  |  |  |  |  |  |
| 701.0                    | 40                                                                           |                       | 26     | 8-9-8                       | 17                 | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN,<br>ANGULAR, SOME ROUNDED GRAVEL, 15-20% COARSE TO FINE SAND, TRACE NONPLASTIC<br>FINES, IRON STAINS AND COAL, GRAY. |                 |  |  |  |  |  |  |
|                          |                                                                              | -s                    | 27     | 13-19-22<br>(14")           | 41                 | GP                  | <u>SIMILAR TO S-26</u> .<br>BLOWS/INCH: 2-2-3-2-2-2/3-3-3-3-4-3/4-3-4-4-3-4                                                                                                                  |                 |  |  |  |  |  |  |
|                          |                                                                              |                       | 28     | 9-11-20<br>(13")            | 31                 | SP                  | SAND, POORLY GRADED, MEDIUM TO FINE, 5-10% COARSE TO FINE GRAVEL,<br>SUBANGULAR TO ROUNDED, 1.5 IN SANDSTONE FRAGMENT AT TOP, TRACE NONPLASTIC<br>FINES, BROWN.                              | 1111            |  |  |  |  |  |  |
| NOTE :                   | 45<br>FOR                                                                    | BORIN                 | G SUN  | MARY AND                    |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET                                                                                                                                       |                 |  |  |  |  |  |  |

,

|                       |                  |                |                  |                             |                       |                   | BORING NO. BOS-1                                                                                                                                                                                                                                                        | ·   |  |  |  |  |  |  |
|-----------------------|------------------|----------------|------------------|-----------------------------|-----------------------|-------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----|--|--|--|--|--|--|
|                       | SHEET 3_OF3      |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
| SI                    | TE               | AVER           | VALLE            | EY POWER S                  | TATIO                 | N-UNI             | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                |     |  |  |  |  |  |  |
| 3                     |                  | E              |                  | a 3                         |                       | 6                 |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
| ELEVATION<br>(FEET)() | DEPTH<br>(feet)  | SAMPLE<br>TYPE | SAMPLE<br>NUMBER | BLOWS (1<br>OR<br>REC/ROD ( | SPT N<br>VALUE (S     | GROUP<br>SYMBOL ( | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                      |     |  |  |  |  |  |  |
|                       |                  |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | :   |  |  |  |  |  |  |
|                       | 45 -             | S              | 29               | 12-25-31<br>(16")           | 56                    | G₽                | SANDY GRAVEL, BROKEN COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE<br>FRAGMENTS TO 1.5 IN MAXIMUM, ANGULAR, FEW ROUNDED, 10-15% COARSE TO FINE<br>SAND, LESS THAN 5% NONPLASTIC FINES, TRACE COAL, BROWN AND GRAY.<br>BLOWS/INCH: 2-I-2-2-2-3/4-5-3-5-4-4/5-5-7-5-3-6 |     |  |  |  |  |  |  |
|                       |                  | S              | 30               | 23-34-111<br>(12")          | 145                   | GP                | SANDY GRAVEL, BROKEN COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO<br>1.5 IN MAXIMUM, ANGULAR, 30-40% COARSE TO FINE SAND, 10-13% NONPLASTIC<br>FINES, TRACE COAL AND IRON STAINING, BROWN.<br>BLOWS/INCH: 2-2-3-4-5-7/5-5-5-8-6-5/13-20-18-17-25                  |     |  |  |  |  |  |  |
| 691.0                 | -<br>50          | s              | 31               | 47-50-113<br>(18")          | 163                   | SP<br>GP          | TOP 2 IN: <u>Sand</u> , Fine, trace fine gravel, 5-102 nonplastic fines, orange-<br>brown.<br>Bottom 16 IN: <u>Sandy gravel</u> , coarse to fine gravel sized sandstone and                                                                                             | 1 1 |  |  |  |  |  |  |
|                       |                  | s              | 32               | 37- <u>105</u><br>3"        | <u>105</u><br>3"      |                   | SHALE FRACMENTS TO 1.5 IN, 20-30% COARSE TO FINE SAND, 10-15% SLIGHTLY<br>PLASTIC FINES, TRACE COAL, BROWN, GRAY, ORANGE-BROWN.<br>BLOWS/INCH: 3-4-6-4-5-25/18-14-5-5-3-5/20-27-15-17-13-21                                                                             |     |  |  |  |  |  |  |
|                       |                  | S              | 33               | 50<br>5"                    | 50<br><sup>1</sup> 2" | P.                | SILTY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACMENTS<br>TO 1.0 IN MAXIMUM, ANGULAR, 5-102 FINE SAND, 5-202 SLIGHTLY PLASTIC<br>FINES, TRACE COAL, IRON STAINS, ORANGE, BROWN, GRAY.                                                                  |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   | REFUSAL                                                                                                                                                                                                                                                                 |     |  |  |  |  |  |  |
|                       | _                |                |                  |                             |                       |                   | BOTTOM OF BORING AT 52 FT 1/2 IN                                                                                                                                                                                                                                        | -   |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   | ELEVATION GOD. 90 FI                                                                                                                                                                                                                                                    |     |  |  |  |  |  |  |
|                       |                  |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | ~                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | , l |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | l   |  |  |  |  |  |  |
|                       |                  |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       |                  | А. С.          |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       |                  |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | _                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       |                  |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
| 1                     | -                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       | _                |                |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       |                  | ľ              |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         | -   |  |  |  |  |  |  |
|                       |                  | i              |                  |                             |                       |                   |                                                                                                                                                                                                                                                                         |     |  |  |  |  |  |  |
| NOTE: I               | for bo<br>Legend | ring<br>NFO.   | SUMM<br>SEE      | ARY AND SHEET I.            |                       | SKE               | TCH NO. 12241-GSK-241C DAY 9/1/62 EOS-1 3 OF                                                                                                                                                                                                                            | 3   |  |  |  |  |  |  |

| _ C      | OOP                           | INATE          | S             | 6.5 FT    | SOUTH C | F EOS-1 | GROLIND FI                               | FLEW (1) 741,0 FT SHEET |          |               |             |  |  |
|----------|-------------------------------|----------------|---------------|-----------|---------|---------|------------------------------------------|-------------------------|----------|---------------|-------------|--|--|
| IN IN    | ICLIN                         | ATION          |               | ERTICAL   |         | BEAR    | G <u>NA</u> IN                           | SPECTOR J.W. P          | 10007    |               |             |  |  |
| D        | ATE                           | STAR           | T/F           | INISH     | 6/7/82  | /_      | / 6/7/82 CONTRACTOR / DRILLEREGER/JARVIS |                         |          |               |             |  |  |
| S        | TATIC                         | ; GRO          | UND           | NATER     | DEPTH   | /DATE   | (FT) / [                                 | DRILL RIG TYPE          |          | E 45          |             |  |  |
| D        | EPTH                          | то             | BEDF          | юск _     | N       | A       | (FT) TOTAL DEPT                          | TH DRILLED              | 2        | 2.0 FT        | <u>(FT)</u> |  |  |
| N        | IETH                          | DDS:           |               |           |         |         | . v                                      |                         |          |               |             |  |  |
|          | . 1                           | RILLI          | NG S          | BOIL 3    | -1/8 IN | O.D. AU | ER TO ADVANCE HOLE, 3 IN                 | O.D. SPLIT SPOON        | USED T   | O CLEAN OUT H | IOLE.       |  |  |
|          | •                             | SAMPL          | ING<br>NG 7   | SOL 3     | IONE    | 086     |                                          |                         | ·        |               |             |  |  |
| e        | PECI                          | AL TF          | STIN          | G OR I    | NSTRU   | ENTAT   | N                                        |                         |          |               |             |  |  |
| J        |                               |                |               |           |         |         |                                          |                         |          |               |             |  |  |
| c        | OMM                           | ENTS .         |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               |                |               | ······    |         |         |                                          |                         |          |               |             |  |  |
|          |                               |                |               |           |         |         |                                          |                         |          |               |             |  |  |
| (16.2)   | <b>T</b>                      | :£             |               | (3)       | 6       | (6)     |                                          |                         |          |               |             |  |  |
| EET)     | EPTI                          |                | MPL           | D/OR      | N U     |         | SAMP                                     | LE DESCRIPT             | ION      |               |             |  |  |
| E        | פֿיַ                          | -  S           | N S           | BLG<br>AN | R S     | B N N   |                                          |                         |          |               |             |  |  |
|          | L                             |                | L             | Č         |         |         | · · ·                                    |                         |          |               | · · · · ·   |  |  |
|          | 0                             | <u> </u>       |               |           |         |         |                                          |                         |          |               |             |  |  |
|          | Ŭ                             | 1              |               |           |         |         | NO SAMDI DE TO                           | 10 FT                   |          |               |             |  |  |
|          |                               | 1              |               |           |         |         | nu annilla IU                            | 10 21                   |          |               |             |  |  |
|          | ĺ                             | -              |               |           |         |         |                                          |                         |          |               |             |  |  |
| •        |                               | 7              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          | 5                             | -              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | -              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 7              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
| 0        | 10                            | <b></b>        |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               |                | 1             | (28")     |         | SII     | Y CLAY - CLAYEY SILT. ST                 | IGHTLY PLASTIC 4        | TTNF     | SAND, LICHT   | RAN         |  |  |
|          |                               | 1              |               | . =, /    |         | (SC     | EWHAT DILATIVE ON HANDLI                 | ING).                   |          |               |             |  |  |
|          |                               | -              | 2             | (0")      |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          |                               | 4              |               |           |         |         |                                          |                         |          |               |             |  |  |
|          | 15                            | 1              |               |           |         |         |                                          |                         |          |               |             |  |  |
| 1.       |                               | IS M           |               | SEA LE    | /EL     | UN      | ISTURBED SAMPLES                         |                         | <b>.</b> |               |             |  |  |
| 2.<br>3. |                               | S REQ          | WATE<br>JIRED | TO DRIN   | Æ       |         | S-SHELBY TUBE<br>O-OSTERBERG             | <u>BO</u>               | RING     | LOG           | •           |  |  |
|          | 2"0.D.                        |                | E SP          | OON 6" (  | R       |         |                                          |                         |          |               |             |  |  |
|          | HOID.                         | HAMME          | R FA          | LLING 30  | )".     |         |                                          | DEAVER VALLE            | T POW    | ER STATIO     | N UNIT-     |  |  |
|          |                               | VCHES          | OF S          | AMPLE     |         |         |                                          | DUQUES                  | NE LK    | STIL COMP/    | ANT .       |  |  |
| 4.       | () I<br>RECO                  | VERY.          |               |           |         |         |                                          | CUIDDINO                |          | DENNOVIN      |             |  |  |
| 4.<br>5. | ()  :<br>RECO<br>STD.<br>BLOW | VERY.<br>PENET | RATIO         | ON RESIS  | TANCE   |         |                                          | SHIPPING                | PORT,    | PENNSYLV      | ANIA        |  |  |

|                          |                 |                    |                  |                                |                    |                     |                                                   |         | BORING NO  | E05-1A |
|--------------------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|---------------------------------------------------|---------|------------|--------|
| SI                       | TE              | AVER               | VALLE            | Y POWER S                      | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO                    | 12241.0 | x          |        |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPT                                   | ION     |            |        |
| r                        | 15              |                    |                  |                                |                    |                     | · · · · · · · · · · · · · · · · · · ·             |         | <u></u>    |        |
|                          | -               | US                 | 3                |                                |                    |                     |                                                   |         |            | •      |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | 4               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  | 5<br>2<br>2                    |                    |                     |                                                   |         |            |        |
| 21.0                     | 20              | US                 | 4                | (23)                           |                    |                     |                                                   |         |            |        |
|                          | -               |                    |                  | 2                              |                    |                     |                                                   |         |            |        |
|                          | _               |                    |                  |                                |                    |                     |                                                   |         |            | -      |
|                          | -               |                    |                  |                                |                    |                     | BOTTOM OF BORING AT 22.0 PT<br>Elevation 719.0 PT |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            | -      |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | 4               |                    |                  |                                |                    |                     |                                                   |         |            | -      |
|                          | 4               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | -               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            | -      |
|                          | -               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | il              |                    |                  |                                |                    |                     |                                                   |         |            | •      |
|                          | -               |                    |                  | -<br>-                         |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | ]               |                    |                  |                                |                    |                     | κ.                                                |         |            |        |
|                          | -               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  | -<br>-<br>-                    |                    |                     |                                                   |         |            |        |
|                          | -               |                    |                  | -<br>                          |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          |                 |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | -               |                    |                  |                                |                    |                     |                                                   |         |            |        |
|                          | -               | 200                |                  |                                |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DA               | ATE     | BORING NO. | SHEET  |
| יז ביוי<br>נ             | EGEND           | NFQ                | SEE              | SHEET I.                       |                    | SKE                 | TCH No. 12241-GSK-2428 DDH 4/1                    | 182-    | EOS-1A     | 2 OF 2 |

|              | 9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ITE  | B           | EAVE        | R VAL          | LEY POWER | STAT         | ion -  | UNIT 2                                                      | J.O. NO. 12                                                 | 241                        | BORING N                              | ). <u>E05-2</u> |  |  |
|--------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------|-------------|-------------|----------------|-----------|--------------|--------|-------------------------------------------------------------|-------------------------------------------------------------|----------------------------|---------------------------------------|-----------------|--|--|
|              | C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | OORI | NIC         | ATE         | S              | N4000     |              |        | E6165 GROUND EL                                             | EV (I) 723.9                                                | 9                          |                                       | OF              |  |  |
|              | - 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |      | AT          | ION         |                | TERTICAL  |              | BE     | ARING II                                                    |                                                             | . W. MCCOY                 |                                       |                 |  |  |
|              | D                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ATE  | : s         | TAF         | RT / F         | INISH _   | 5/21/        | 82     | / CONTRACTO                                                 | R / DRILLER                                                 | EGER/JA                    | RVIS                                  | <u> </u>        |  |  |
|              | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | TATI | 5           | GRO         | UND            | WATER D   | )EPT         | H / D  | ATE 40'10" (FT) / 5/27/82                                   | DRILL RIG TY                                                | PE                         | 45                                    |                 |  |  |
|              | D                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | EPTI | • •         | то          | BEDR           | юск       | 60.0         | )      | (FT) TOTAL DEPT                                             | TH DRILLED                                                  | 60.                        | 3                                     | (PT)            |  |  |
|              | M                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ETH  | 003         | <b>s</b> :  |                |           |              |        |                                                             |                                                             |                            |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | DRI         |             | NG S           | SOL       | <u>3-1/8</u> | IN RO  | LLER BIT, 3-1/4 IN I.D. CASH                                | NG DRILLING MUD                                             | <b>,</b>                   |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | SÅ          | MPL         | .ING           | SOL       | 2 IN C       | ).p. 9 | PLIT SPOON                                                  |                                                             |                            |                                       |                 |  |  |
| l            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | DRI         | ILLI        | ING R          | юск 🔔     | IONE         |        |                                                             |                                                             |                            | <u> </u>                              |                 |  |  |
| I            | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | PEC  | AL          | TE          | STIN           | G OR IN   | STRU         | JMEN   | TATION                                                      |                                                             |                            | · · · · · · · · · · · · · · · · · · · |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             |             |                |           |              |        |                                                             |                                                             |                            |                                       |                 |  |  |
| İ            | C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | OMN  | EN          | TS .        |                | ·         |              |        | · · · · · · · · · · · · · · · · · · ·                       | <u></u>                                                     |                            |                                       | <del></del>     |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             |             |                |           |              | 8      |                                                             |                                                             |                            |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             |             |                |           |              |        |                                                             |                                                             |                            |                                       |                 |  |  |
|              | [2]                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |      |             | 3           |                |           |              | 6      |                                                             |                                                             |                            |                                       |                 |  |  |
| 9            | T)()E                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | Ē    | 2           | س<br>ہے تے  | <u>ן</u> ש מין | 500       | z 6          | ۳<br>۳ |                                                             |                                                             |                            |                                       |                 |  |  |
| I ₹          | Image: State of the state |      |             |             |                |           |              |        |                                                             |                                                             |                            |                                       |                 |  |  |
|              | )                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |      |             | φ1          | 07 2           |           | <b>``</b> >  | 5      |                                                             |                                                             |                            |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             |             |                | <b></b>   |              | •      |                                                             |                                                             |                            |                                       |                 |  |  |
| 7.7          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | -           |             | Τ.             |           |              |        | ALLO CONDER TO RIVE OBJURI                                  | CT220 10 200                                                |                            |                                       |                 |  |  |
| / <u>*</u> . |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             | 5           | 1              | 4-3-3     | ľ            | GP     | (FILL).                                                     | SIZED, 10-202                                               | LUARSE TO                  | FIRE SARD, I                          | skuwa, –        |  |  |
| ł            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | +           |             | 1              |           |              |        |                                                             |                                                             |                            |                                       | -               |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             |                |           |              |        |                                                             | AND WELL CRADED COARSE TO FINE 10-157 COARSE TO FINE GRAVE! |                            |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | -           | 5           | 2              | 2-3-2     | 5            | SW     | SAND, WELL GRADED, COARSE TO<br>ROUNDED, LESS THAN 5% NONPL | O FINE, 10-15%<br>ASTIC FINES, BR                           | COARSE TO                  | FINE GRAVEL                           | -               |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             | 4              |           |              |        |                                                             |                                                             |                            |                                       | -               |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 5    | -           |             |                |           |              |        |                                                             |                                                             |                            |                                       | -               |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | -    | 7           | S           | 3              | 2-2-3     | 5            | SP     | TOP 4 IN: SAND, UNIFORM, FI                                 | NE, 2-5% NONPLA                                             | STIC FINE                  | S, TRACE GRAM                         | EL, BROWN.      |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             |                | (14)      |              |        | VERY MOIST, LIGHT BROWN.                                    | VEL STOCK PRACE                                             | , LEJJ 1114<br>ØNTC        |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | _           |             |                |           |              | Gr     | BUILDED IN: COAL, FIRE GRA                                  | VEL SIZED FRAGE                                             | <b>L</b> .                 |                                       | _               |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             | s           | 4              | 5-5-4     | 9            | GW     | SANDY GRAVEL, COARSE TO FIN                                 | E, FEW TO 1 IN                                              | MAXIMUM, I                 | ROUNDED TO A                          | CULAR,          |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | -           |             |                | (11")     |              |        | 20-30% COARSE TO FINE SAND,<br>COAL AND IRON STAINING, MOI  | MOSTLY COARSE,<br>ST, GRAY AND BR                           | , 2-57 NONI<br>ROWN.       | PLASTIC FINE:                         | S, TRACE        |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             | 1              |           |              |        |                                                             |                                                             |                            |                                       | -               |  |  |
| 71           | 3.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 10   | 4           | S           | 5              | 4-11-6    | 17           | GP     | TOP 8 IN: SANDY GRAVEL, COA                                 | RSE TO FINE, SU                                             | BANGULAR :                 | TO ROUNDED, 1                         | 30-407          |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             |                | (18")     |              | GP     | COARSE TO FINE SAND, 5-82 N<br>BOTTOM 10 IN: SANDY GRAVEL.  | ONPLASTIC FINES<br>COARSE TO FINE                           | 5. TRACE CO<br>E. 1 IN MAX | DAL, GRAY.<br>(IMUM. SOME )           |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | -           |             | 1              | ł         |              |        | SHALE FRAGMENTS, 25-30% COA                                 | RSE TO FINE SAN                                             | 10, 5–7°2 NI               | NPLASTIC FI                           | NES, TRACE      |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | コ           |             | 1              |           | 1            |        | BLOWS/INCH: 1/2-1/2-1-1/3-1                                 | -2-2-2-1/1-1-1+                                             | 1-1-1                      |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 4           | S           | 6              | 7-12-11   | 23           | GP     | SANDY GRAVEL, COARSE TO FIN.<br>30-401 COARSE TO FINE SAND. | E GRAVEL SIZED<br>MOSTLY MEDIUM                             | SANDSTONE<br>TO FINE, 1    | FRAGMENTS TO<br>LESS THAN 5X          | NONPLASTIC      |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      | 1           |             |                |           |              |        | FINES, TRACE COAL AND IRON                                  | STAINING, GRAY                                              | AND BROWN                  |                                       |                 |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | 15   | -           |             |                |           |              |        | BLOR3/IRCH, //2-2-3-2-2-2/3                                 |                                                             |                            |                                       | -               |  |  |
|              | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | DATI | L           | (9. N       | IF AN          | SEA LEV   | FI           |        |                                                             |                                                             |                            |                                       |                 |  |  |
|              | 2.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ₽ G  | iou         | IND         | WATE           | RLEVEL    |              |        | US-SHELBY TUBE                                              | £                                                           | ORING                      | LOG                                   | _               |  |  |
| ~            | 3.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | BLOI | 73<br>. 54  | REQ         | UIRED<br>.E SP | TO DRIV   | E<br>A       |        | UQ-QSTERBERG                                                |                                                             |                            |                                       | _               |  |  |
| ΤE           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | DIST | ANC         | ES          | HOWN           | USING     |              |        |                                                             | BEAVER VAL                                                  | LEY POV                    | VER STATI                             | on unit-2       |  |  |
| NO           | 4.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | ()   | NC          | HES         | OF             | SAMPLE    | •            |        |                                                             | DUQUI                                                       | ESNE LI                    | GHT COMP                              | PANY            |  |  |
| -            | 5,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | RECO | PE          | RY.<br>INET | RATI           | ON RESIS  | TANCE        | :      |                                                             | SHIPPU                                                      | NGPORT,                    | PENNSYL                               | VANIA           |  |  |
|              | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | BLOV | IS/         | FT.         | L CI           | ASSIFICA  |              |        |                                                             |                                                             |                            |                                       |                 |  |  |
| GE           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | SYST | ÊM          |             |                |           |              |        |                                                             | SKETC                                                       | C. WEBST                   | EN ENG. CO<br>41-GSK-243A             | RP.             |  |  |
| Ē            | 7.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | S-SF | ·LE<br>·LI1 | TY<br>F BA  | RREL           | . SAMPLE  |              |        |                                                             | APPROVED                                                    | DATE                       | BORING NO.                            | SHEET           |  |  |
|              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |      |             |             |                |           |              |        |                                                             | 2200                                                        | 9/./8m                     | EOS-2                                 | 1 OF 3          |  |  |

|                         | BORING NO. 505-2                                                                                                                                                                                                                                                                                                                                              |                                                                         |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|-------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|----------|--------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|--|--|--|
|                         |                                                                                                                                                                                                                                                                                                                                                               | BEAVES                                                                  | V 61 1 1 | EN DURED 6                     | TAT10              | N 11N T             | T 2. SHIPPINGPORT. PA.                                                                                                                                                                                                                                            |  |  |  |  |  |  |  |
| SI                      | TE                                                                                                                                                                                                                                                                                                                                                            | BEAVER                                                                  | VALL     | EI POWER S                     |                    | N- UNI              | J.a. NB                                                                                                                                                                                                                                                           |  |  |  |  |  |  |  |
| ELEVATION<br>(FEET)(162 | DEPTH                                                                                                                                                                                                                                                                                                                                                         | SAMPLE<br>SAMPLE                                                        | SAMPLE   | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               |                                                                         |          |                                |                    |                     | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                             |  |  |  |  |  |  |  |
|                         | 15                                                                                                                                                                                                                                                                                                                                                            | - S                                                                     | 7        | 7-7-6<br>(9")                  | 13                 | GX<br>GP            | TOP 5 IN: <u>STLTY GRAVEL</u> , COARSE TO FINE, ANGULAR TO ROUNDED, 10-15Z COARSE<br>TO FINE SAND, MOSTLY FINE, 15-20Z NONPLASTIC FINES, BROWN.<br>BOTTOM 4 IN: <u>WEATHERED SANDSTONE FRACHENTS</u> , 1 <sup>1</sup> 5 IN MAXIMUM, 10-15Z COARSE<br>SAND, BROWN. |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               |                                                                         | 1        |                                |                    |                     | BLOWS/INCH: 2-1-1-1-1/1-1-1-1-2/1-1-1-1-1-1-                                                                                                                                                                                                                      |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               |                                                                         | 8        | 8-13-11<br>(8")                | 24                 | GP<br>GW            | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZE SANDSTONE FRACHENTS, SOME SHALE<br>TO 14 IN MAXIMUM, ANGULAR TO SUBROUNDED, 20-30% COARSE TO FINE SAND,<br>5-7% SLIGHTLY PLASTIC FINES, TRACE IRON STAINS, BROWN.<br>BLOWS/INCH: 1-1-1-1-2-2/2-2-3-2-2/2-1-2-2-2         |  |  |  |  |  |  |  |
| 703.9                   | 703.9 20 5 9 17-21-12 33 GW- <u>SANDY CRAVEL</u> , COARSE TO FINE, ROUNDED TO ANGULAE. SOME SANDSTONE AND<br>GP SHALE FRAGMENTS TO 1 IN MAXIMUM, LARGE SANDSTONE PRACMENT AT BOTTOM,<br>20-302 COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, LESS THAN 52<br>NONPLASTIC FINES, TRACE IRON STAINING, BROWN.<br>BLOWS/INCH: 3-3-2-3-3-3/4-4-4-3-3-3/3-2-3-2-1-1 |                                                                         |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|                         | •                                                                                                                                                                                                                                                                                                                                                             | -<br>-<br>-                                                             | 10       | 4-5-6<br>(13")                 | 11                 | SP                  | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY MEDIUM TO FINE, 2-6% COARSE<br>TO FINE ROUNDED GRAVEL, 2-5% NONPLASTIC FINES, MOIST, BROWN.                                                                                                                           |  |  |  |  |  |  |  |
|                         | 25                                                                                                                                                                                                                                                                                                                                                            |                                                                         | 11       | 4-6-7<br>(14")                 | 13                 | SP                  | SAND, SIMILAR TO ABOVE, MOSTLY COARSE TO MEDIUM.                                                                                                                                                                                                                  |  |  |  |  |  |  |  |
|                         | S    12    53/4"    53/4"    -    NO RECOVERY: BLOWS/INCH: 8-9-17-19      S    13    13-24-20    44    -    BROKEN, ROUNDED GRAVEL TO 15 IN (WASH?)      BLOWS/INCH: 2-2-2-2-3/5-5-3-3-4-4/3-3-4-3    -    -    -    -                                                                                                                                        | NO RECOVERY: BLOWS/INCH: 8-9-17-19                                      |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               | BROKEN, ROUNDED GRAVEL TO 15 IN (WASH?)                                 |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               |                                                                         |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
| 693.9                   | 30 -                                                                                                                                                                                                                                                                                                                                                          | SANDSTONE FRAGMENTS, 5-15% COARSE TO FINE GRAVEL, 10-15% COARSE TO FINE |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|                         | •                                                                                                                                                                                                                                                                                                                                                             |                                                                         | 1.5      | 16-19-26                       | 45                 | SP                  | TOP & THE SAND FIND, LESS THAN ST NOMPLASTIC PINDS REGIME                                                                                                                                                                                                         |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               | -                                                                       | <b>[</b> | (7")                           |                    | GH                  | BOTTOM 3 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS -                                                                                                                                                                              |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               | +                                                                       | 1        |                                |                    |                     | FINES, BROWN.<br>BLOWS/INCH: 2-2-3-4-3/3-3-2-3-4-4/4-5-4-4-5-4                                                                                                                                                                                                    |  |  |  |  |  |  |  |
|                         | 35 -                                                                                                                                                                                                                                                                                                                                                          |                                                                         | 16       | 14-16-27<br>(7")               | 43                 | GH                  | SILTY GRAVEL, COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRAGMENTS TO 14 IN, ANGULAR, 15-207 COARSE TO FINE SAND, 15-207 SLIGHTLY<br>TO MEDIUM PLASTIC FINES, BROWN, GRAY AND ORANGE.<br>BLOWS/INCH: 2-3-2-2-3-2/2-3-3-2-3-3/5-5-4-2-5-6        |  |  |  |  |  |  |  |
|                         | •                                                                                                                                                                                                                                                                                                                                                             | - 5                                                                     | 17       | 28-24-21<br>(11")              | 45                 | GW<br>GP            | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRAGMENTS TO 15 IN, ANGULAR, 20-25% COARSE TO FINE SAND, 5-10% SLIGHTLY<br>PLASTIC FINES, TRACE COAL AND IRON STAINS, BROWN AND GRAY.                                                  |  |  |  |  |  |  |  |
| (00.0                   |                                                                                                                                                                                                                                                                                                                                                               | -                                                                       |          |                                |                    |                     | pLUW3/1RUR: 4-3-8-4-4-3/9-3-4-4-3/9-4-3-3-3<br>                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
| 08319                   | 40 '                                                                                                                                                                                                                                                                                                                                                          | - s                                                                     | 18       | 11-11-10<br>(8")               | 21                 | 5P                  | SAND, POORLY GRADED, LESS THAN 5% COARSE TO FINE GRAVEL, COARSE TO FINE<br>SAND, MOSTLY COARSE TO MEDIUM, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-2-2-1-2-2/2-1-2-2-2-2-1-2                                                                        |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               |                                                                         |          |                                |                    |                     |                                                                                                                                                                                                                                                                   |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               | - s                                                                     | 19       | 9-11-14<br>(8")                | 25                 | SP                  | SAND, SIMILAR TO ABOVE, SOFT, BLACK, CARBONACEOUS SHALE PRAGMENT AT                                                                                                                                                                                               |  |  |  |  |  |  |  |
|                         |                                                                                                                                                                                                                                                                                                                                                               | ╀                                                                       | 4        |                                |                    |                     | BLOWS/INCH: 1-2-1-2-2-1/1-2-2-2-2-2/2-2-3-2-3                                                                                                                                                                                                                     |  |  |  |  |  |  |  |
|                         | foe f                                                                                                                                                                                                                                                                                                                                                         |                                                                         | <u> </u> |                                |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO SHEET                                                                                                                                                                                                             |  |  |  |  |  |  |  |
|                         | LEGEN                                                                                                                                                                                                                                                                                                                                                         | DINFO.                                                                  | SEE      | SHEET I.                       |                    | SKE                 | TCH No. 12241=GSK-245B 2004. 9/1/62. EOS-2 2 07 3                                                                                                                                                                                                                 |  |  |  |  |  |  |  |

|                        |                                                                                           |                       |                  |             |              |                  |                    |                     | BORING NO. EOS-2                                                                                                                                                                                                                                                                     |  |  |  |  |
|------------------------|-------------------------------------------------------------------------------------------|-----------------------|------------------|-------------|--------------|------------------|--------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
|                        |                                                                                           | BÉA                   | VER              | VALLI       | EY PO        | WER S            | TATIO              | N-UN1               | T 2, SHIPPINGPORT, PA. 10 10 12241.00                                                                                                                                                                                                                                                |  |  |  |  |
| 3                      | 11E -                                                                                     | -T                    | 3                |             | 1            |                  | 1                  |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
| ELEVATION<br>(FEET)(IE | DEPTH                                                                                     | (FEET)                | SAMPLE<br>TYPE ( | SAMPLE      | BLOWS (3)    | 0R<br>REC/ROD (4 | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                   |  |  |  |  |
|                        |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                        | 45                                                                                        |                       | 5                | 20          | 14-1<br>(7") | 08-55            | 163                | GP-<br>GN           | SANDY GRAVEL, WEATHERED SANDSTONE AND SHALE FRAGMENTS, 30-403 COARSE TO<br>FIME SAND, 5-103 SLIGHTLY PLASTIC FINES, TRACE COAL, BROWN, GRAY, ORANGE.<br>BLOWS/INCH: 2-1-2-3-3-3/ 5-5-27-38-19-14/16-12-8-7-7-5                                                                       |  |  |  |  |
|                        |                                                                                           | ┵                     | s                | 21          | 9-8-<br>(7") | 11               | 19                 | G₩-<br>GP           | SANDY GRAVEL, COARSE TO FINE GRAVEL, FEW FRAGMENTS TO 1 IN, ANGULAR TO<br>ROUNDED, 15-25% COARSE TO FINE SAND, 5-10% SLIGHTLY PLASTIC FINES, TRACE<br>COAL AND IRON STAINING, FEW WEATHERED SANDSTONE AND SHALE FRAGMENTS, BROWN.<br>BLOWS/INCH: 2-1-1-1-2-2/1-1-2-1-2-1/1-2-1-2-3-2 |  |  |  |  |
| 673.9                  | 673.9 50 5 22 12-14-28 42 GW-<br>(6") 42 GW-<br>GP 50 50 50 50 50 50 50 50 50 50 50 50 50 |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                        |                                                                                           |                       | s                | 23          | 14-1<br>(5") | 3-11             | 24                 | GW-<br>GP           | BLOWS/INCH: 2-2-2-2-2-2/3-3-2-2-2/2-6-4-5-5-6<br><u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRACMENTS, LARGE SANDSTONE FRACMENT AT TOP, ANCULAR TO SUBBOUNDED, 30-402                                                                        |  |  |  |  |
|                        | 55                                                                                        |                       |                  | 1           | 10-5         | 6_00             | 1.56               |                     | COARSE TO FINE SAND, 5-102 SLIGHTLY PLASTIC FINES, UNARGE AND GEAT.                                                                                                                                                                                                                  |  |  |  |  |
|                        |                                                                                           |                       | 5                | 24          | (11"         | )                | 1.11               |                     | WEATHERED SANDSTONE AND SIDES, SOFT, SORE SOFT CERTSTONE, THE HEAT, SHA                                                                                                                                                                                                              |  |  |  |  |
|                        |                                                                                           |                       | S                | 25          | 16-3<br>(15" | 0-70<br>)        | 100                | GP<br>GP            | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACMENTS<br>TO 14 IN, SOFT, 20-30Z COARSE TO FINE SAND, 10-15Z SLIGHTLY PLASTIC<br>FINES, TRACE IRON STAINING, GRAY.                                                                                                  |  |  |  |  |
| 663.9                  | 60                                                                                        |                       | <u>s</u>         | 26          | 100/         | 4 ''             | 100/4              | 14                  | <u>CLAYSTONE</u> , WEATHERED, SOFT, DARK GRAY.<br>BOTTOM OF BORING AT 60 FT 4 IN                                                                                                                                                                                                     |  |  |  |  |
|                        |                                                                                           | I<br>I<br>I<br>I<br>I |                  |             |              |                  |                    |                     | ELEVATION 663.6 FT                                                                                                                                                                                                                                                                   |  |  |  |  |
|                        |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                        |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
| ĺ                      |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                        |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
|                        |                                                                                           |                       |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
| {                      | {                                                                                         | ]                     |                  |             |              |                  |                    |                     |                                                                                                                                                                                                                                                                                      |  |  |  |  |
| NOTE :                 | For<br>Lege                                                                               | DORI                  | NG<br>FQ         | SUMM<br>SEE | LARY<br>SHEE | AND<br>TI.       |                    | STO<br>SKE          | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TCH No. 12241-GSK-243C DD A 9/./62 EOS-2. 3 OF 3                                                                                                                                                                           |  |  |  |  |

| 3<br>C<br>IN<br>D<br>3<br>S<br>D<br>M<br>S<br>C<br>C | ITE<br>OORDIN<br>ICLINAT<br>ATE : :<br>TATIC<br>EPTH<br>ETHOD<br>DR<br>SA<br>DR<br>PECIAL<br>COMMEN | BE<br>NATE<br>TION<br>STAF<br>GRO<br>TO<br>OS :<br>NILLI<br>RILLI<br>L TE | 33                                                        | NALLEY PO<br>NAUSO<br>ERTICAL<br>INISH<br>NATER (<br>ROCK<br>SOL<br>SOL<br>G OR IN    | 5/24/<br>5/24/<br>DEPT<br>(<br>3-1/8<br>2 IN (<br>KONE<br>ISTRI |                     | -UNIT 4                                                                                                                                                                                      |      |
|------------------------------------------------------|-----------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------|-----------------------------------------------------------|---------------------------------------------------------------------------------------|-----------------------------------------------------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------|
| ELEVATION<br>(FEET)(IC2)                             | DEPTH<br>(Feet)                                                                                     | SAMPLE<br>TYPE (7)                                                        | SAMPLE<br>NUMBER                                          | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                   | 3PT N<br>VALUE (5)                                              | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                           |      |
| 22.1                                                 | 0                                                                                                   | 5                                                                         | 1                                                         | 6-22-9                                                                                | 31                                                              | GP                  | TOP 5 IN: SANDY SLAG AND SANDSTONE FRACMENTS, GRAY.                                                                                                                                          | ┫    |
|                                                      | -                                                                                                   | <u> </u>                                                                  |                                                           | (9")                                                                                  |                                                                 | 59-<br>Sw           | BUTTON 4 IN: GRAVELLI SAND, 20-JOX COARSE TO FINE GRAVEL, PEN TO 1 IN,<br>ROUNDED TO SUBANGULAR, COARSE TO FINE SAND, 5-10% SLICHTLY PLASTIC FINES,<br>TRACE COAL AND IRON STAINING, BROWN.  | 4    |
|                                                      |                                                                                                     | 5                                                                         | 2                                                         | 9-16-13<br>(16")                                                                      | 29                                                              | SP<br>SW<br>GX      | TOP 4 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 12 IN: <u>SILTY GRAVEL</u> , COARSE TO FINE, ANGULAR TO ROUNDED, 10-20%<br>COARSE TO FINE SAND, 20-30% SLIGHTLY PLASTIC FINES, BROWN AND GRAY. |      |
|                                                      | 5                                                                                                   | \$                                                                        | 3                                                         | 4-7-7<br>(18")                                                                        | 14                                                              | ਰਮ<br>ਸੁ            | TOF 3 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 15 IN: <u>GRAVELLY SILT</u> , 20-25% COARSE TO FINE GRAVEL, ANGULAR TO<br>ROUNDED, 10-15% COARSE TO FINE SAND, ORANGE-BROWN.                   |      |
|                                                      |                                                                                                     | S                                                                         | 4                                                         | 7-4-4<br>(18")                                                                        | 8                                                               | ML.                 | GRAVELLY SILT, SLIGHTLY PLASTIC, 10-15% COARSE TO FINE GRAVEL, ROUNDED,<br>5-10% FINE SAND, TRACE COAL, BROWN AND ORANGE.                                                                    |      |
| 2.1                                                  | 10                                                                                                  | s                                                                         | 5                                                         | 3-4-10<br>(15")                                                                       | 14                                                              | ML                  | STMILAR TO ABOVE, 2 IN THICK COARSE SAND LAYER AT 8 IN FROM TOP OF SAMPLE.                                                                                                                   |      |
|                                                      | -                                                                                                   | s                                                                         | 6                                                         | 4-7-6<br>(18")                                                                        | 13                                                              | SM                  | SILTY SAND, UNIFORM, LESS THAN 5% FINE GRAVEL, ROUNDED, FINE SAND, 20-30% NONPLASTIC FINES, BROWN.                                                                                           |      |
| 1.<br>2.<br>3.<br>4.<br>5.                           | 15<br>DATUM<br>C GRO<br>BLOWS<br>2"O.D. 9<br>DISTAN<br>40b. H<br>( ) INC<br>RECOVE<br>STD. P        | IS NUND<br>REQUIND<br>CE 9<br>ALME<br>CHES<br>CHES<br>ERY.<br>ENET        | IEAN<br>WATE<br>UIRED<br>LE SPI<br>HOWN<br>IR FAI<br>OF S | SEA LEV<br>R LEVEL<br>TO DRIV<br>OON 6" O<br>USING<br>LLING 30'<br>BAMPLE<br>ON RESIS | EL<br>E<br>R<br>TANCE                                           |                     | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG<br>BEAVER VALLEY POWER STATION UNIT-<br>DUQUESNE LIGHT COMPANY<br>SHIPPINGPORT, PENNSYLVANIA                                           | Ň II |

- 6

|                          |           |                                                 |                    |                  |                                                                                                                                                                                                                                                                          |                    |                     | BORING NO. <u>E05-3</u><br>Sheet 2 of 3                                                                                                                                                                                                                                                                                                                                                               |
|--------------------------|-----------|-------------------------------------------------|--------------------|------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| s                        | ITE _     | BĒ.                                             | AVER               | VALL             | EY POWER S                                                                                                                                                                                                                                                               | TATIO              | N-UN1               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                                                                                                                              |
| ELEVATION<br>(FEET)(IE2) | DEPTH     | (FEET)                                          | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4)                                                                                                                                                                                                                                           | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                                                                                    |
|                          | <u> </u>  | <b>ب</b>                                        |                    | L                | L                                                                                                                                                                                                                                                                        |                    | ·                   |                                                                                                                                                                                                                                                                                                                                                                                                       |
|                          | 1.5       |                                                 | 5                  | 7                | 3-4-5<br>(18")                                                                                                                                                                                                                                                           | 9                  | SM                  | SILTY SAND, WIDELY GRADED, 20-25% COARSE TO FINE GRAVEL, ANGULAR TO<br>ROUNDED, COARSE TO FINE SAND, 15-20% NONPLASTIC FINES, TRACE ROOTS AND<br>IRON STAINS, DARK BROWN.                                                                                                                                                                                                                             |
|                          |           |                                                 | 9                  | в                | 3-17-20<br>(10")                                                                                                                                                                                                                                                         | 37                 | SM                  | SIMILAR TO ABOVE.<br>BLOWS/INCH: 3/2-1-1-3-5-5/4-4-4-2-4-2                                                                                                                                                                                                                                                                                                                                            |
| 702.1                    | 20        |                                                 | s                  | 9                | 3-3-3<br>(13")                                                                                                                                                                                                                                                           | 6                  | SP                  | SAND, FOORLY GRADED, LESS THAN 5% FINE GRAVEL, ROUNDED, COARSE TO FINE<br>SAND, MOSTLY MEDIUM TO FINE, LESS THAN 5% NONPLASTIC FINES, BROWN.                                                                                                                                                                                                                                                          |
|                          |           |                                                 | S                  | 10               | 2-3-6<br>(1B")                                                                                                                                                                                                                                                           | 9                  | SW                  | SAND, WELL GRADED, LESS THAN 57 FINE GRAVEL, ROUNDED, COARSE TO FINE SAND,<br>5-77 NONPLASTIC FINES, TRACE COAL, BROWN.                                                                                                                                                                                                                                                                               |
|                          | 25        | 5 5 11 8-7-10<br>(18")<br>5 12 6-11-13<br>(11") | S                  | 11               | 8-7-10<br>(18")                                                                                                                                                                                                                                                          | 17                 | SW<br>GP-<br>GW     | TOP 8 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, 1 IN MAXIMUM, ANGULAR<br>TO SUBROUNDED, 30-40% COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, LESS<br>THAN 5% NOMPLASTIC FINES, TRACE COAL AND IRON STAINS, BROWN.                                                                                                                                   |
|                          |           |                                                 | 24                 | GP-<br>GW        | BLOWS/INCH: 1-1-2-1-1-2/2-1-1-1-1-1/1-2-2-1-2-2<br>SANDY GRAVEL, COARSE TO FINE, FEW FRAGMENTS TO 1-1/2 IN, ANGULAR TO<br>ROUNDED, 15-25% COARSE TO FINE SAND, 10-15% NONPLASTIC FINES, TRACE COAL<br>AND IRON STAINING, BROWN.<br>BLOWS/INCH: 6/2-2-2-2-2-1/3-2-2-1-2-3 |                    |                     |                                                                                                                                                                                                                                                                                                                                                                                                       |
| 692.1                    | 30        |                                                 | S                  | 13               | 14-11-14<br>(8")                                                                                                                                                                                                                                                         | 25                 | GP<br>GW            | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS,<br>1-1/2 IN MAXIMUM, 15-20% COARSE TO FINE SAND, LESS THAN 5% NONPLASTIC<br>FINES, BROWN.<br>BLOWS/INCH: 2-2-3-3-2-2/1-2-2-2-2-2/2-3-2-3-2                                                                                                                                                                                   |
|                          |           | وللفريد                                         | s                  | 14               | 8-10-11<br>(10")                                                                                                                                                                                                                                                         | 21                 | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACHENTS<br>TO $1-1/2$ IN, 15-20% COARSE TO FINE SAND, 5-10% SLIGHTLY PLASTIC FINES,<br>TRACE IRON STAINING, RED, LIGHT GRAY AND BROWN, CONTAINED 1 IN THICK<br>COARSE TO FINE SAND SIZED COAL LENS AT 5 IN FROM TOP.<br>BLOWS/INCH: $2-2-1-1-1/1+1-2-2-2-2/2-2-2-1-2-2$                                                               |
|                          | 35        |                                                 | 5                  | 15               | 10-16-20<br>(9")                                                                                                                                                                                                                                                         | 36                 | SP<br>GP-<br>GV     | TOP 5 IN: <u>SAND</u> , POORLY GRADED, TRACE FINE GRAVEL, COARSE TO FINE SAND,<br>MOSTLY COARSE TO MEDIUM, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BOTTOM 4 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRACMENTS-<br>TO 1-1/2 IN, ANGULAR TO ROUNDED, 25-35% COARSE TO FINE SAND, 5-10%<br>SLIGHTLY PLASTIC FINES, TRACE COAL, BROWN.<br>BLONE (MUCH, $2-2-2-3-3-3/4-5-4/2-3-2$ |
|                          |           |                                                 | 5                  | 16               | 10 <b>-8</b> -7<br>(5")                                                                                                                                                                                                                                                  | 15                 | -<br>ମ୍ୟ            | TOP 3 IN: <u>GRAVEL</u> , COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1-1/2 IN,<br>WASH.<br>BOTTOM 2 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ANGULAR TO ROUNDED, 15-20%<br>COARSE TO FINE SAND, 5-7% SLIGHTLY PLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-2-1-2-1-2/2-1-2-1-1-1/2-1-1-1-1                                                                                                                 |
| 682.1                    | 40        |                                                 | S                  | 17               | 11-10-15<br>(2")                                                                                                                                                                                                                                                         | 25                 | GP<br>CH            | SANDY CRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, LARGE ANGULAR SANDSTONE<br>FRACMENT AT BOTTOM, 20-25% COARSE TO FINE SAND, 5-7% SLIGHTLY PLASTIC<br>FINES, TRACE COAL AND IRON STAINING, BROWN.<br>BLOWS/INCH: 2-2-1-2-2/1-2-2-1-2-2/2-3-3-3-2-2                                                                                                                                                    |
|                          |           |                                                 | S                  | 18               | 12-8-8<br>(10")                                                                                                                                                                                                                                                          | 16                 | GP-<br>GV<br>SV     | TOP 4 IN: SIMILAR TO ABOVE.<br>BOTTOM 6 IN: SAND, WELL GRADED, TRACE FINE GRAVEL, COARSE TO FINE SAND,<br>LESS THAN 5% NONPLASTIC FINES, TRACE COAL, BROWN.<br>BLOWS/INCH: 2-2-2-2-2-2-2-1-1-1-2/1-2-1-1-1-2                                                                                                                                                                                          |
| NOTE:                    | 45<br>FOR |                                                 | RING<br>NFD        | SUMM             | ARY AND                                                                                                                                                                                                                                                                  |                    | STO<br>SKE          | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TCH No. 12241-csk-244B DDA 9/./82 EDS-3 2 OF 3                                                                                                                                                                                                                                                                                              |

| SI                        | TE              | VER                | VALLE            | Y POWER S                      | TATIO              | N-UNI               | SHEET 3_OF      T 2, SHIPPINGPORT, PA.    J.O. NO. 12241.00                                                                                                                                                             |
|---------------------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET )(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>Number | BLOWS (3)<br>OR<br>REC/RGD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                      |
|                           | 45              | -                  | 10               | 10 15-11                       | 26                 |                     | CANNY CRAITEL COARSE TO FINE AWEIDAR TO ROTHINED. 20-102 COARSE TO FI                                                                                                                                                   |
|                           |                 |                    | 17               | (5")                           |                    | GW                  | SAND, LESS THAN 5% HOMPLASTIC FINES, TRACE COAL, BROWN.<br>BLOWS/INCH: 2-2-4-3-4-3/2-3-3-3-2-2/2-3-2-1-2-1                                                                                                              |
|                           | ┥               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               | s                  | 20               | 9-13-11<br>(9")                | 24                 | G?-<br>Gi           | <u>SIMILAR TO ABOVE</u> , FEW FRAGMENTS TO 1-1/2 IN.<br>BLOWS/INCH: 1-2-1-2-1-2/2-2-2-3-2/2-1-2-2-2-2                                                                                                                   |
| 672.1                     | 50 -            |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               | 5                  | 21               | 6-6-12<br>(5")                 | 18                 | ମ୍ମ<br>ମ୍ୟ          | SANDY GRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, 15-20% COARSE TO FIN<br>SAND, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 1-1-1-1-1-1/1-1-1-1-1-1/2-2-1-3-2-2                                                |
|                           | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               | 5                  | 22               | 12-13-14<br>(7")               | 27                 | GP                  | SANDY GRAVEL, BROKER, WEATHERED SANDSTORE AND SHALE FRACHERTS, COASE<br>FINE GRAVEL SIZED, TO 1-1/2 IN MAXIMUM, FEW COAL FRACHERTS, 15-20% COA<br>TO FINE SAND, 5-7% SLIGHTLY PLASTIC FINES, TRACE HICA, ORANGE, BROWH, |
|                           | Ē,              |                    | :                |                                |                    |                     | GRAY, BLACK.<br>BLOWS/INCH: 2-2-2-2-2-2/3-2-3-1-2-2/2-3-2-3-2-2                                                                                                                                                         |
| :                         |                 | s                  | 23               | 30-30-36<br>(14")              | 66                 | G₽                  | SANDY GRAVEL, BROKEN, WEATHERED SANDSTONE AND SHALE FRAGMENTS, COARSE<br>FINE GRAVEL SIZED TO 1-1/2 IN MAXIMUM, MOSTLY COARSE, 20-30% MEDIUM T<br>SAND, 5-10% SLIGHTLY PLASTIC FINES, TRACE COAL AND IRON STAINS, ORANG |
|                           | Ξ               |                    |                  |                                |                    |                     | RED, BROWN, GRAY.<br>BLOWS/INCH: 5-3-7-5-5/6-8-5-4-4-3/4-3-6-6-10-7                                                                                                                                                     |
|                           | -               | s                  | 24               | 25-13-15<br>(14")              | 28                 | G₽-<br>G₩           | SANDY GRAVEL, COARSE TO FINE, 1-1/2 IN MAXIMUM, ANGULAR TO ROUNDED, 20<br>COARSE TO FINE SAND, 5-7% NONPLASTIC FINES, TRACE IRON STAINS, ORANGE<br>BROWN.                                                               |
| 662 1                     |                 |                    |                  |                                |                    |                     | BLOWS/INCH: 4-6-5-5-2-3/2-3-2-2-2-2-2-2-2-3-3-3                                                                                                                                                                         |
|                           |                 | S                  | 25               | 18-30-80<br>(12")              | 110                | -                   | SHALE, COARSE TO FIME GRAVEL SIZED FRAGMENTS, SOFT, WEATHERED, 5-102 1<br>FINE SAND, 25-33% SLIGHTLY PLASTIC FINES, TRACE COAL, GRAY AND BROWN.<br>BLOWS/INCH: 3-4-3-2-3-3/3-4-4-4-7-8/15-20-13-13-10-9                 |
|                           | 4               | 5                  | 26               | 105/6"                         | 105/               |                     | <u>Shale</u> , Soft, Weathered, Gray.                                                                                                                                                                                   |
|                           |                 |                    | -                | (4")                           | 6"                 |                     |                                                                                                                                                                                                                         |
|                           | 4               |                    |                  |                                | ļ                  |                     | BOTTOM OF BORING AT 63.0 FT<br>Elevation 659.1 FT                                                                                                                                                                       |
|                           | 3               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | 3               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | 4               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                                                         |
|                           | -               |                    |                  | _                              |                    |                     |                                                                                                                                                                                                                         |

,

| SITEBEAVER VALLEY POWER STATION-UNIT 2  J.O. NO. 12241  BORING NO. EDS-4    COORDINATESN4164.41  E6101.98  GROUND ELEV. (1) 720.1 FT  SHEETOF3    INCLINATIONVERTICAL  BEARINGNA |                                                                     |                                                                                                                           |                                                                                            |                                                                                                                  |                        |                     |                                                                                                                                                                              |                                                               |                                                                                            |                                                                                             |                                   |  |  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------|------------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------|--------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|-----------------------------------|--|--|
| SF                                                                                                                                                                               |                                                                     | I TO<br>DOS :<br>DRILLI<br>BAMPL<br>DRILL<br>AL TE                                                                        | BEDI<br>NG !<br>.ING !<br>ING F<br>.ST IN                                                  | ROCK                                                                                                             | -1/8 I<br>IN 0.<br>DNE | N O.D.<br>D. SP     | (FI) TOTAL DEP<br>. ROLLER BIT, 3-1/4 IN ID CA<br>LIT SPOON AND 3 IN O.D. SHELI<br>TATION NONE                                                                               | TH DRILLEC                                                    |                                                                                            | 53.0                                                                                        |                                   |  |  |
| (FEET)(162)                                                                                                                                                                      | DEPTH<br>(FEET)                                                     | SAMPLE<br>TYPE (7)                                                                                                        | SAMPLE                                                                                     | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                                              | SPT N<br>VALUE (5)     | GROUP<br>SYMBOL (6) | SAMF                                                                                                                                                                         | PLE DESCR                                                     | RIPTION                                                                                    |                                                                                             |                                   |  |  |
| ).1                                                                                                                                                                              | 0                                                                   | <b>-</b> s                                                                                                                | 1                                                                                          | 3-6-11<br>(10")                                                                                                  | 17                     | GP-<br>GW           | SANDY GRAVEL, COARSE TO FIN<br>COARSE TO FINE SAND, 5-10%                                                                                                                    | E, 14 IN MAXI<br>NONPLASTIC FI                                | MUM, ANGULA<br>NES, TRACE                                                                  | R TO ROUNDED<br>ROOTS, IRON                                                                 | , 25-35%                          |  |  |
|                                                                                                                                                                                  | 5                                                                   | - S                                                                                                                       | 2                                                                                          | 11-17-12<br>(10")<br>10-13-10<br>(12")                                                                           | 2 29                   | GP-<br>GW<br>SW     | STAINING, BROWN.<br><u>SANDY GRAVEL</u> , SIMILAR TO ABA<br><u>5Z NONPLASTIC FINES</u> , DARK BU<br><u>GRAVELLY SAND</u> , WELL-CRADED,<br><u>SAND</u> 5.77 NONPLASTIC FINES | OVE, 30-40% C<br>ROWN.<br>20-30% COARS                        | COARSE TO FINE G                                                                           | NE SAND, LES<br>Ravel, coars                                                                | S THAN                            |  |  |
|                                                                                                                                                                                  |                                                                     |                                                                                                                           | 4                                                                                          | 9-7-7<br>(7")                                                                                                    | 14                     | SP                  | GRAVELLY SAND, POORLY GRADE<br>FINE SAND, 5-10% NONPLASTIC                                                                                                                   | 2-2-2-2/1-2-2<br>D, 20-307 COA<br>FINES, TRACE                | RSE TO FINE                                                                                | GRAVEL, MED<br>S, DARK BROW                                                                 |                                   |  |  |
| 0.1                                                                                                                                                                              | 10                                                                  | s<br>s                                                                                                                    | 5                                                                                          | 4-13-11<br>(12")                                                                                                 | 24                     | SP<br>GP            | TOP 7 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 5 IN: <u>SANDY BROKEN G</u><br>TRACE IRON STAINING.<br>BLOWS/INCH: 4/2-1-2-3-3-2/2                                             | RAY SANDSTONE                                                 | <u>,</u> 30-407 CO.                                                                        | ARSE TO MEDI                                                                                | UN SAND,                          |  |  |
|                                                                                                                                                                                  | 15                                                                  |                                                                                                                           | 6                                                                                          | 4-17-11<br>(12")                                                                                                 | 28                     | GP-<br>GW           | SANDY GRAVEL, COARSE TO FIN<br>ANGULAR TO ROUNDED, 25-35%<br>5-10% NONPLASTIC FINES, TRA<br>BLOWS/INCH: 4/1-4-3-3-3-3/3                                                      | E, FEW SANDST<br>COARSE TO FIN<br>CE IRON STAIN<br>-1-2-2-2-1 | ONE FRAGMEN<br>IE SAND, MOS<br>IING, GRAY A                                                | IS TO 15 IN 3<br>TLY MEDIUM T<br>ND DARK BROW                                               | MAKIMUM,<br>O FINE,<br>N.         |  |  |
| 1. (<br>2. 3. 6<br>4. (<br>5. 5<br>6. (<br>7. 5<br>5                                                                                                                             | DATUI<br>GR<br>CON<br>CON<br>CON<br>CON<br>CON<br>CON<br>CON<br>CON | M IS N<br>OUND<br>S REQ<br>SAMPI<br>NCE S<br>HAMME<br>NCHES<br>VERY.<br>PENET<br>S/FT.<br>ED SO<br>EM.<br>LE TY<br>LIT BA | MEAN<br>WATE<br>UIRED<br>LE SP<br>HOWN<br>IR FA<br>OF !<br>IRAT  <br>IL CL<br>PE:<br>ARREL | SEA LEV<br>R LEVEL<br>) TO DRIV<br>OON 6" O<br>I USING<br>LLING 30<br>SAMPLE<br>ON RESIS<br>ASSIFICA<br>. SAMPLE | EL<br>R<br>TANCE       | <u></u>             | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG                                                                                                                        |                                                               | BORING<br>ALLEY POL<br>UESNE LI<br>PINGPORT,<br>NE & WEBST<br>TCH No. 12<br>DATE<br>9/1/82 | LOG<br>WER STATI<br>GHT COM<br>PENNSYL<br>FER ENG. CC<br>241-GSK-245A<br>BORNG NO.<br>EOS-4 | ON UNIT-2<br>PANY<br>VANIA<br>RP. |  |  |

|                         |                                                                                                           |                   |          |                  |                                |                    |                     |                                                                                                                                                                        |                                                                                                                                                                                                                |                                             | BORING NO                                     | <u>508-4</u>    |  |  |  |
|-------------------------|-----------------------------------------------------------------------------------------------------------|-------------------|----------|------------------|--------------------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------|-----------------------------------------------|-----------------|--|--|--|
|                         |                                                                                                           | RFAI              | VER      | VALLE            | a power s                      | TATIO              | N-UNI'              | T 2. SHIPPINGPORT, PA.                                                                                                                                                 | 10.0                                                                                                                                                                                                           | 12241.0                                     | SHEET 2 (                                     | )F_ <u>3</u>    |  |  |  |
| ्ड<br>- रू              | ITE.                                                                                                      | T                 | 2        |                  |                                |                    |                     |                                                                                                                                                                        | J.O. M                                                                                                                                                                                                         | 0                                           |                                               |                 |  |  |  |
| ELEVATION<br>(FEET)(163 | DEPTH                                                                                                     | (FEET)<br>saura e | TYPE (   | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (S) | GROUP<br>SYMBOL (6) | SAMP                                                                                                                                                                   | E DESCR                                                                                                                                                                                                        | IPTION                                      |                                               |                 |  |  |  |
|                         |                                                                                                           |                   |          |                  |                                |                    |                     |                                                                                                                                                                        |                                                                                                                                                                                                                |                                             |                                               |                 |  |  |  |
|                         | 15                                                                                                        |                   | ω<br>    | 7                | 9-11-10<br>(14")               | 21                 | ST-<br>SV           | GRAVELLY SAND, 30-40% COARSE<br>TO ROUNDED, COARSE TO FINE S<br>NONPLASTIC FINES, TRACE IRON<br>BLOWS/INCE: 9/2-2-3-1-2-1/2-                                           | C TO FINE GRA<br>AND, MOSTLY<br>1, BROWN.<br>-2-2-1-1-2                                                                                                                                                        | VEL TO 14 1)<br>MEDIUM TO FI                | HAXTHUN, AN<br>INE, LESS THA                  | GULAR           |  |  |  |
| 1                       |                                                                                                           |                   | s        | 8                | 6-8-12<br>(8")                 | 20                 | GP-<br>GH           | SANDY GRAVEL, COARSE TO FINE<br>Sand, less than 52 homplast                                                                                                            | 2, ANGULAR TO<br>IC PINES, BRO                                                                                                                                                                                 | ROUNDED, 20<br>WN.                          | 328A00 20E-C                                  | TO FINE         |  |  |  |
| 700.1                   | 20                                                                                                        |                   | s        | 9                | 14-12-10<br>(13")              | 22                 | æ                   | SILTY GRAVEL, COARSE TO FINI<br>SANDSTONE AND SHALE FRAGMENT<br>PLASTIC FINES, TRACE COAL AN<br>IRON AND BLACK.                                                        | . HOSTLY NED<br>19, 10-15% co<br>10 MICA, TRAC                                                                                                                                                                 | IUM TO FINE<br>ARSE TO FINE<br>E IRON SYAII | GRAVEL SIZEI<br>E SAND, 15-20<br>NING, BROWN, | WEATHERED       |  |  |  |
|                         | ]                                                                                                         |                   | s        | 10               | 11-63-74<br>(14")              | 137                | SH<br>              | BLOWS/INCH: 2-2-2-3-2-3/2-3/<br>TOP 9 IN: <u>SILTY SAND</u> , POORLY<br>PLASTIC FIMES, TRACE FINE GI<br>BOTTOM 5 IN: <u>SLAG</u> , GRAY,<br>BLOWS/INCH: 1-1-1-3-4/7-13 | -2-2-2-1/1-2-<br>7 GRADED, MET<br>RAVEL, BROWN.<br>1-16-10-10-9/                                                                                                                                               | 2-1-2-2<br>NUM TO FINE<br>20-23-8-11-4      | SAND, 10-125<br>5-6                           |                 |  |  |  |
|                         | 25                                                                                                        |                   | s        | 11               | 10-11-17<br>(0")               | 28                 |                     | <u>NO RECOVERY</u><br>BLOWS/INCH: 1-2-1-2-2-2/2-1-2-2-2-2/2-3-3-4-3+2                                                                                                  |                                                                                                                                                                                                                |                                             |                                               |                 |  |  |  |
|                         |                                                                                                           |                   | s        | 12               | 10-9-6<br>(1")                 | 15                 | SP                  | GRAVELLY SAND, 20-30% COARS<br>TO FINE SAND, MOSTLY MEDIUM<br>BLOWS/INCH: 2-2-2-1-1-2/1-2-                                                                             | <u>SRAVELLY SAND</u> , 20-30% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, COARSE<br>TO FINE SAND, MOSTLY MEDIUM TO FINE, LESS THAN 5% NONPLASTIC FINES, BROM<br>BLOWS/INCH: 2-2-2-1-1-2/1-2-2-2-1-1/1-1-1-1-1-1 |                                             |                                               |                 |  |  |  |
| 690.1                   | 30                                                                                                        |                   | s        | 13               | 4-5-6<br>(11")                 | 11                 | CL.                 | SILTY CLAY, SLIGHTLY YO MODI<br>OCCASIONAL FINE GRAVEL TO<br>MOTTLED BROWN, GRAY BROWN N                                                                               | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, MEDIUM STIFF TO OCCASIONAL FINE GRAVEL TO $\frac{1}{3}$ in, rounded, 5-72 fine sand, mo motiled brown, gray brown with pockets of gray. $q_u$ (pp):2               |                                             |                                               |                 |  |  |  |
|                         |                                                                                                           | t t t t t t       | s        | 14               | 18-15-19<br>(13")              | 34                 | GP-<br>GH           | SANDT GRAVEL, COARSE TO FINI<br>FINE SAND, MOSTLY COARSE TO<br>TRACE IRON STAINING, GRAY.<br>BLOWS/INCH: 1-2-4-3-4-4/2-3-                                              | 2, FEW TO 1.5<br>MEDIUN, LESS<br>-3-2-2-3/3-3-                                                                                                                                                                 | IN MAXIMUN<br>Than 5% Noi<br>4-3-3-3        | , 30-40% COAL<br>Rplastic fini                | 25,             |  |  |  |
|                         | 35                                                                                                        | ╡                 | 5        | 15               | 3-4-4<br>(15")                 | 8                  | CL.                 | <u>SILTY CLAY</u> , SLIGHTLY TO MODI<br>Fine gravel, Moist, Gray.                                                                                                      | RATELY PLAST<br>q <sub>u</sub> (pp): 2.0                                                                                                                                                                       | TIC, MEDIUN :<br>MISF.                      | STIFF TO STI                                  |                 |  |  |  |
|                         |                                                                                                           | 1                 | US       | 1                | (23.5")                        |                    | ત્ત                 | SANDY CLAY, MODERATELY PLAST<br>COAL UP TO 3/8 IN, DARK GRAY                                                                                                           | ric, 10-157 )<br>rish brown.                                                                                                                                                                                   | EDIUM TO FI                                 | NE SAND, FEW                                  | PIECES -        |  |  |  |
|                         |                                                                                                           |                   | S        | 16               | 4-4-5<br>(12")                 | 9                  | CL.                 | SILTY CLAY, MEDIUM STIFF TO<br>FINE SAND, BROWN. Qu (PP)                                                                                                               | STIFF, MODEN<br>2.0TSF                                                                                                                                                                                         | ATELY PLAST                                 | IC, LESS THAT                                 | 4 5x            |  |  |  |
| 680.1                   | 40                                                                                                        |                   | V5       | 2                | (23'')                         |                    |                     | SIMILAR TO 516 (TUBE TRIMM                                                                                                                                             | INGS).                                                                                                                                                                                                         |                                             |                                               |                 |  |  |  |
|                         | S 17 4-4-4 8 CL <u>SILTY CLAY</u> , MEDIUM STIFF TO STIFF, SLI<br>THAN 52 PINE SAND, BROWN WITH GRAY MOTT |                   |          |                  |                                |                    |                     |                                                                                                                                                                        |                                                                                                                                                                                                                |                                             | ERATELY PLAS<br>pp); 2.25TSF                  | FIC, LESS       |  |  |  |
|                         |                                                                                                           |                   | US       | 3                | (0")                           |                    |                     | NO RECOVERY.                                                                                                                                                           |                                                                                                                                                                                                                |                                             |                                               | -               |  |  |  |
| NOTE:                   | FOR<br>LEGE                                                                                               | BORI<br>ND IN     | NG<br>FQ | SUMM<br>SEE      | ARY AND<br>Sheet 1.            |                    | STO                 | NE & WEBSTER ENG. CORP.<br>TCH No. 12241-65K-2458                                                                                                                      |                                                                                                                                                                                                                | DATE<br>9/-/82                              | BORING NO.<br>EOS- 4                          | SHEET<br>2 OF 3 |  |  |  |
|                       |                 | Ē              |                  |                                   |                    | 6                 | J.O, NU                                                                                           |
|-----------------------|-----------------|----------------|------------------|-----------------------------------|--------------------|-------------------|---------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(( | DEPTH<br>(Feet) | SAMPLE<br>TYPE | SAMPLE<br>NUMBER | BLOWS (3<br>OR<br>OR<br>REC/ROD ( | SPT N<br>VALUE (5) | GROUP<br>SYMBOL ( | SAMPLE DESCRIPTION                                                                                |
|                       | 45              | 5              | 18               | 0-4-4                             | 8                  | C1.               | SILTY CLAY, MODERATELY PLASTIC, MEDIUM STIFF TO STIFF, 102 VERY FINE                              |
|                       |                 | US             | 4                | (18")                             |                    |                   | SAND, BROWN. qu (pp): 1.75, 0.75, 1.23TSF                                                         |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
| 670.1                 | 50 -            | \$             | 19               | 4-6-5<br>(18")                    | 11                 | CL                | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC. SOFT TO MEDIUM STIFF, MOIST BROWN. $q_u$ (pp): 0.5TSF |
|                       |                 |                |                  | (011)                             |                    |                   |                                                                                                   |
|                       |                 | US             | 5                | (0")                              |                    | CL                | NO RECOVERS.<br>PUSHED SPLIT SPOON (S-20) - RECOVERED SILTY CLAY SIMILAR TO S-19.                 |
|                       |                 |                |                  |                                   |                    |                   | BOTTON OF BORING AT 53.0 PT                                                                       |
|                       |                 |                |                  | 1                                 |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       | 11              |                |                  |                                   |                    |                   |                                                                                                   |
| 1                     |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       | 1               |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       | 11              |                |                  |                                   |                    |                   |                                                                                                   |
|                       | l i l           |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |
|                       |                 |                |                  |                                   |                    |                   |                                                                                                   |

| SITE<br>COOR<br>INCLI<br>DATE<br>STATI<br>DEPT<br>METH<br>SPEC                                                                  | BEAN<br>DINATIO<br>: STA<br>IC GR<br>H TC<br>IODS :<br>DRILL<br>SAMF<br>ORILL<br>CALL 1<br>MENTS        | 722 VAL<br>TES<br>N<br>ART / I<br>ROUND<br>BED<br>LING<br>LING<br>ESTIN<br>3                      | LEY POWER<br>N4158.7<br>VERTICAL<br>FINISH<br>WATER I<br>ROCK<br>SOL 3-<br>SOL 2<br>ROCK NC<br>NG OR IN | 5/27/8<br>5/27/8<br>DEPTH<br>72.<br>7/8 IN<br>IN 0.2<br>DNE<br>STRUE | 7N-UN<br>BE<br>32<br>1 / D.<br>5<br>( ROL<br>0. SP | IT 2<br>E6103.3 GROUND EL<br>ARING IN<br>/ 5-28-82 CONTRACTO<br>NOT<br>ATERECORDELIFT: /NA C<br>[FI] TOTAL DEPT<br>LER BIT, 4 IN I.D. CASING, DR<br>PLIT SPOON AND 3 IN 0.D. OSTER<br>TATION NONE | LUING MUD                                                                                                 | BORING NO. <u>EOS-4A</u><br>SHEETOF _2OV<br>/JARVIS<br>COE 45<br>                                                     |
|---------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|----------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(16.2)<br>DEPTH                                                                                              | (FEET)<br>Sample                                                                                        | TYPE (7)<br>SAMPLE<br>NIMBER                                                                      | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                                     | SPT N<br>VALUE (5)                                                   | GROUP<br>SYMBOL (6)                                | SAMP                                                                                                                                                                                              | LE DESCRIPTION                                                                                            |                                                                                                                       |
| 35                                                                                                                              |                                                                                                         | 10 1<br>10 2<br>10 3                                                                              | (30.5")<br>5-4-5<br>(14")<br>(30")                                                                      | 9                                                                    | -<br>CL<br>CL                                      | NO SAMPLES<br><u>SILTY CLAY</u> , MODERATELY PLAST<br>ORGANICS, BROWN WITH SOME MO<br>qu (pp): 2.25, 2.0, 2.5TSP<br>NO RECOVERY.<br><u>SIMILAR TO S-1</u> (TRIMMINGS                              | TO 35.5 FT<br>IC, STIFP, LESS THAN 5<br>TILED CRAY.                                                       | Z FINE SAND, TRACE                                                                                                    |
| 45<br>1. DATI<br>2. ♀ G<br>3. ∂LO<br>2"QI<br>DIST<br>140R<br>4. ()<br>REC<br>5. STO<br>BLO<br>6. UNIF<br>SYST<br>7. SAM<br>S-SI | JM IS<br>ROUND<br>WS RE<br>D. SAM<br>ANCE<br>INCHE<br>OVERY<br>VS/FT<br>IED S<br>TEM.<br>PLE T<br>PLE T | MEAN<br>WATE<br>QUIREL<br>PLE SP<br>SHOWN<br>MER FA<br>S OF S<br>THATI<br>OIL CL<br>YPE<br>BARREL | SEA LEVI<br>R LEVEL<br>D TO DRIVI<br>DON 6" OI<br>I USING<br>SAMPLE<br>ON RESIST<br>ASSIFICAT           |                                                                      |                                                    | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG                                                                                                                                             | BORING<br>BEAVER VALLEY PO<br>DUQUESNE L<br>SHIPPINGPOR<br>STONE & WEBS<br>SKETCH No. 12<br>APPROVED DATE | BLOG<br>WER STATION UNIT-2<br>LIGHT COMPANY<br>F, PENNSYLVANIA<br>HTER ENG. CORP.<br>241-GSK-246A<br>BORING NG. SHEET |

•

|                         |                 |                   |                  |                                     |                    |                     | BORING NO                                                                                                                                                                 |
|-------------------------|-----------------|-------------------|------------------|-------------------------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                         |                 |                   |                  |                                     | T 1 1 1 0          |                     | SHEET 2_ OF                                                                                                                                                               |
| s<br>                   | TE              |                   | VALLI            | T POWER S                           |                    | T                   | J.O. NO. 12241.00                                                                                                                                                         |
| ELEVATION<br>(FEET)(162 | DEPTH<br>(feet) | SAMPLE<br>TYPE (1 | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4)      | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                        |
|                         | 45              | S                 | 2                | 3-3-4                               | 7                  | CL                  | SILTY CLAY, MODERATELY PLASTIC, MEDIUM STIFF, TRACE FINE SAND, MOIST,<br>BROWN WITH GRAY MOTTLING.<br>gu (pp): 1.25, 1.75TSF                                              |
|                         | -               | υο                | 4                | (29.8")                             |                    | CL                  | SIMILAR TO 5-2. (TRIMMINGS)                                                                                                                                               |
|                         | -               | s                 | 3                | 3-4-6<br>(18")                      | 10                 | CL                  | <u>SIMILAR TO S-2.</u> TRACE ORGANIC MATERIAL.<br>q <sub>u</sub> (pp): 1.25TSF                                                                                            |
| 670.4                   | 50              | υο                | 5                | (30")                               |                    | CL.                 | SIMILAR TO S-2. (TRIMMINGS)                                                                                                                                               |
|                         |                 | s                 | 4                | 3-4-5                               | 9                  | CI,                 | SANDY CLAY, MODERATELY PLASTIC, STIFF, 232 VERY FINE SAND, BROWN.                                                                                                         |
|                         | 55 -            | υo                | 6                | (30")                               |                    | CL                  | SIMILAR TO S-2. (TRINMINGS)                                                                                                                                               |
|                         |                 | s                 | 5                | 3-5-4<br>(18")                      | 9                  | с<br>с<br>с         | TOP 8 IN: <u>SIMILAR TO S-2.</u><br>BOTTOM 10 IN: <u>SILTY CLAY</u> , MODERATELY PLASTIC, SOFT, CONTAINS FINE SAND<br>LENSES LESS THAN 1 mm THICK, GRAY. q. (pp): 0.75TSF |
|                         | -               | vo                | 7                | (30.5")                             |                    |                     |                                                                                                                                                                           |
| 660.4                   | 60              | s                 | 6                | 2 <b>2-</b> 4<br>(18'')             | 6                  | GL.                 | SANDY CLAY, SLIGHTLY PLASTIC, 20-25% VERY FINE SAND, MEDIUM STIFF, SOME<br>VERY FINE SAND LENSES, 5 mm THICK, GRAY. qu (pp) 1.0, 0.75TSF                                  |
| 1                       | -               | υŌ                | 8                | (29.3")                             |                    |                     |                                                                                                                                                                           |
|                         | 65 -            | S                 | 7                | 3-3-6<br>(16'')                     | 9                  |                     | SANDY CLAY - SANDY SILT, SLIGHTLY PLASTIC, 15-207 VERY FINE SAND, CONTAINS<br>FINE SAND LENSES LESS THAN 1-2 mm THICK, NUMEROUS SMALL WHITE DEPOSITS,                     |
|                         | -               |                   |                  |                                     |                    |                     | I TER DIARETER, MOISI, DARK GRAT.                                                                                                                                         |
|                         |                 | S                 | В                | 29-28-19<br>(10")                   | 47 ·               | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS,<br>WEATHERED, MAXIMUM SIZE 1-1/2 IN.                                                             |
| 650.4                   | 70 -            |                   |                  |                                     |                    |                     |                                                                                                                                                                           |
|                         | -               | S                 | 9                | 3-15- <u>101</u><br>4 <sup>11</sup> | <u>116</u><br>10"  |                     | TOP 10 IN: <u>BROKEN SANDSTONE AND SHALE</u> , SOFT, WEATHERED.<br>BOTTOM 4 IN: <u>SHALE</u> , SOFT, CRAY.<br>BLOWS/IN: 3-2-3-1-3-1/3-2-2-2-3-3/6-10-40-45                |
|                         |                 |                   |                  |                                     |                    |                     | BOTTOM OF BORING AT 72 FT 10 IN<br>ELEVATION 647.6 FT                                                                                                                     |
| NOTE :                  | FOR BO          | RNG<br>NFQ        | SUMM<br>SEE      | ARY AND SHEET I.                    |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORNG NO. SHEET<br>TCH No. 12241-65K-246B DDH. 9/./8L EOS-4A 2 OF 2                                                                 |

| s        | ITEB             | EAVER        | R VALI      | .ey power                               | STATI          | :014 - I                                      | UNIT 2                                 | _ J.O. NO                              | BORING NO. BOS-5                      |  |  |  |  |  |  |
|----------|------------------|--------------|-------------|-----------------------------------------|----------------|-----------------------------------------------|----------------------------------------|----------------------------------------|---------------------------------------|--|--|--|--|--|--|
| С        | OORDIN           | ATE          | s _1        | 4 300                                   |                | E                                             | GROUND EL                              | EV. (I) _683.0                         |                                       |  |  |  |  |  |  |
| IÞ       |                  | ГЮN          |             | ERTICAL                                 |                | . BE                                          | ARING <u>NA</u> II                     | NSPECTOR                               | · · · · · · · · · · · · · · · · · · · |  |  |  |  |  |  |
| D        | ATE : :          | STAR         | IT/F        | INISH _                                 | 5/ <u>1/82</u> |                                               | / 6/2/82 CONTRACTO                     |                                        | GER/JARVIS                            |  |  |  |  |  |  |
| S        | TATIC            | GRO          | UND         | WATER I                                 | DEPT           | H / D/                                        | ATERECORDENITY /                       | DRILL RIG TYPE                         |                                       |  |  |  |  |  |  |
| D        | EPTH             | то           | BEDF        | коск —                                  | 51.            | 0                                             | (FT) TOTAL DEPT                        | TH DRILLED                             | 51.25 (FT)                            |  |  |  |  |  |  |
| M        | ETHOD            | S:           | NG (        | 9 <b>0</b> 11 <sup>7</sup>              | 1-1/B          | IN RO                                         | LLER BIT. 3-1/4 IN L.D. CASTN          | G. DRILLING MUD.                       |                                       |  |  |  |  |  |  |
|          | 54               |              | ING (       | sou _2                                  | 2 IN C         | .D. S                                         | PLIT SPOON, 3 IN O.D. SHELBY           | TUBE AND OSTERBERG.                    | · · · · · · · · · · · · · · · · · · · |  |  |  |  |  |  |
|          | DR               |              | NG F        | юск _                                   |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
| 9        | PECIAL           | . TE         | STIN        | IG OR IN                                | ISTR           | UMEN'                                         | TATION NONE                            |                                        |                                       |  |  |  |  |  |  |
|          |                  |              |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
| C        | OMMEN            | ITS .        |             |                                         |                |                                               |                                        | ···· · · · · · · · · · · · · · · · · · | · · · · · · · · · · · · · · · · · · · |  |  |  |  |  |  |
|          |                  |              |             |                                         |                |                                               |                                        | ······································ |                                       |  |  |  |  |  |  |
| ~        |                  |              |             |                                         |                |                                               |                                        |                                        | ·                                     |  |  |  |  |  |  |
| )(16.2   | ΞĤ               | а<br>        | u e         | (i) | 20             | 9                                             |                                        |                                        |                                       |  |  |  |  |  |  |
| EET      | JEP1<br>FEE      | AMP.         | AMP         | ND/C                                    | PT J           |                                               | SAMP                                   | LE DESCRIPTION                         | ON                                    |  |  |  |  |  |  |
| ະ        | -=               | ้จั          | 0 Ž         | HECO BIL                                | <b>"</b> \$    | S S                                           |                                        |                                        |                                       |  |  |  |  |  |  |
|          | L                | <u> </u>     | <b>L</b>    | <del>-</del>                            |                | <b>ل</b> ــــــــــــــــــــــــــــــــــــ | - <u></u>                              |                                        | · · · · · · · · · · · · · · · · · · · |  |  |  |  |  |  |
| 0        | 0_               | s            | 1           | 2-2-1                                   | 3              |                                               | NO RECOVERY                            |                                        |                                       |  |  |  |  |  |  |
|          | -                |              |             | (0)                                     |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | -                |              |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | -                | s            | 2           | 1-1-1                                   | 2              | HT.                                           | SILT, SLIGHTLY TO MODERATE             | LY PLASTIC, SOFT, T                    | RACE FINE SAND AND ROOTS,             |  |  |  |  |  |  |
|          | -                |              |             | (7")                                    |                |                                               | BROWN WITH ORANGE MOTTLING.            | •                                      |                                       |  |  |  |  |  |  |
|          |                  |              |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | ) <u> </u>       |              |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | -                | ð            | 1           | 1-1-1                                   | ľ              |                                               | MATERIAL.                              | IC, SUFI, 13-20¢ FI                    | C SARD, SOME ONGANIC                  |  |  |  |  |  |  |
|          | -                | 1            |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | -                |              | 1           |                                         |                |                                               |                                        |                                        | -                                     |  |  |  |  |  |  |
|          | -                | <sup>s</sup> | 1           | 1-1-2                                   | 3              |                                               | STRILAR TO ABOVE.                      |                                        |                                       |  |  |  |  |  |  |
| <u>,</u> | 10               | 1            |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
| .u.      | 10 -             | s            | 5           | 1-1-1                                   | 2              | м                                             | SANDY SILT, SLIGHTLY TO MOR            | DERATELY PLASTIC, 1                    | -20% FINE SAND, TRACE                 |  |  |  |  |  |  |
|          |                  | 1            |             | (7")                                    |                |                                               | ORGANIC MATERIAL, BROWN.               | -                                      |                                       |  |  |  |  |  |  |
|          | _                | υs           | 1           | (0")                                    |                |                                               | NO RECOVERY.                           |                                        |                                       |  |  |  |  |  |  |
|          | -                |              |             |                                         |                |                                               |                                        |                                        | -                                     |  |  |  |  |  |  |
|          |                  |              |             |                                         |                |                                               |                                        |                                        |                                       |  |  |  |  |  |  |
|          | .15              | s            | 6           | 1-2-2<br>(5")                           | 4              | ст.мц                                         | SANDY CLAY - SANDY SILT, MOI<br>BROWN. | DERATELY PLASTIC, S                    | OFT, 15-20% FINE SAND,                |  |  |  |  |  |  |
| ١.       | DATUM            | IS M         | IEAN        | SEA LEV                                 | ΈL             |                                               | UNDISTURBED SAMPLES                    | _                                      |                                       |  |  |  |  |  |  |
| 2.<br>3  |                  | UND<br>REO   | WATE        | R LEVEL                                 | C              |                                               | US-SHELBY TUBE<br>UO-OSTERBERG         | BOR                                    | ING LOG                               |  |  |  |  |  |  |
|          | 2"0.0. 5         |              | E SP        | OON 6" 0                                | R              |                                               |                                        |                                        | DOWER STATION HART-                   |  |  |  |  |  |  |
| A        | 1401b. H         |              | R FA        | LLING 30                                | <b>r</b> .     |                                               |                                        | DUQUESN                                | E LIGHT COMPANY                       |  |  |  |  |  |  |
| -        | RECOVE           | RY.          | UF :        |                                         | <b></b>        | -                                             |                                        | SHIPPINGP                              | ORT, PENNSYLVANIA                     |  |  |  |  |  |  |
| э.<br>-  | BLOWS            | ENET<br>/FT. | RATI        | UN RESIS                                | TANCE          | •                                             |                                        |                                        |                                       |  |  |  |  |  |  |
| 6.       | SYSTEM           | J 501        | IL CL       | ASSIFICA                                | TION           |                                               |                                        | STONE & W                              | EBSTER ENG. CORP.                     |  |  |  |  |  |  |
| 7.       | SAMPLI<br>S-SPLI | E TY<br>T BA | PE:<br>RREL | SAMPLE                                  |                |                                               |                                        | APPROVED DAT                           | TE BORING NO. SHEET                   |  |  |  |  |  |  |
|          | S-SPLI           | TBA          | RREL        | SAMPLE                                  |                |                                               |                                        | APPROVED DAT                           | TE BORNG NO. SHEET<br>2- EOS-5   OF 3 |  |  |  |  |  |  |

|                          |                                                                                                                                                                                                                                    |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                             | BORING NO                                       | EOS-5                        |  |  |
|--------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------|------------------|--------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------|------------------------------|--|--|
|                          |                                                                                                                                                                                                                                    |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                             | SHEET 2 (                                       | 0F_3                         |  |  |
| s                        |                                                                                                                                                                                                                                    | AVER               | VALLI            | EY POWER S                     | TATIO              | N-UNI               | T 2, SHIPPINCPORT, PA. J.O. NO. 12241.                                                                                                                                                                                                                      | 00                                              |                              |  |  |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(Feet)                                                                                                                                                                                                                    | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                          |                                                 |                              |  |  |
|                          |                                                                                                                                                                                                                                    |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                             |                                                 |                              |  |  |
|                          | 15                                                                                                                                                                                                                                 | S                  | 7                | 2 <b>-2-</b> 2<br>(15")        | 4                  | I./ML               | <u>SANDY CLAY-SANDY SILT, SLIGHTLY PLASTIC, SOFT, 20-2</u>                                                                                                                                                                                                  | 5% FINE SAND,                                   | BROWN.                       |  |  |
|                          |                                                                                                                                                                                                                                    | υο                 | 2                | (30")                          |                    | CL                  | SANDY CLAY, MODERATELY PLASTIC, 30-40% MEDIUM TO FI<br>FINE, MOTTLED LIGHT BROWN, GRAYISH BROWN AND YELLOW                                                                                                                                                  | NE SAND, MOST<br>BROWN.                         |                              |  |  |
| 663.0                    | 20-                                                                                                                                                                                                                                | s                  | 8                | 2-2-3<br>(18")                 | 5                  | CL.                 | SANDY CLAY, SLIGHTLY PLASTIC, 307 FINE SAND, BROWN<br>MOTTLING. qu(pp) = 0.75, 1.0 TSF                                                                                                                                                                      | AND GRAY WITH                                   | ORANGE                       |  |  |
|                          |                                                                                                                                                                                                                                    | υο                 | 3                | (28")                          |                    |                     |                                                                                                                                                                                                                                                             |                                                 |                              |  |  |
|                          | 25                                                                                                                                                                                                                                 | S                  | 9                | 3-2-2<br>(18")                 | 4                  | 년년-<br>19년          | TOP 13 IN: SIMILAR TO S-8.<br>Bottom 5 IN: Organic <u>clayey silt</u> , moderately to Hig<br>Fine Sand, Gray.                                                                                                                                               | HLY PLASTIC,                                    | TRACE                        |  |  |
|                          | 111                                                                                                                                                                                                                                | ບຣ<br>ຣ            | 10               | (27")                          | 13                 | SH-                 | SANDY CLAY, MODERATELY PLASTIC, 12-20X VERY FINE SA<br>(TUBE TRIMMINGS)<br>LAYERED SILTY SAND AND SANDY CLAY, LAYER THICKNESS                                                                                                                               | ND, GRAY.<br>1/4 IN TO 3/4                      | IF                           |  |  |
|                          |                                                                                                                                                                                                                                    |                    |                  | (18")                          |                    | CL ·                | SAND IS FINE, CLAY IS MODERATELY PLASTIC, SOFT, GRA                                                                                                                                                                                                         | Y.                                              | -                            |  |  |
| 633.0                    | 30111                                                                                                                                                                                                                              | S                  | 11               | 17-19-16<br>(6")               | 35                 | SP                  | GRAVELLY SAND, FINE TO COARSE GRAVEL TO 1 IN, COARS<br>FINE, 10-152 NOMPLASTIC FINES, CONTAINS SEVERAL PIE<br>SANDSTONE INDICATING SPOON SAMPLED COBBLE.<br>BLOWS/INCH: 2-3-3-3-3-3/4-3-3-4-2-3/2-3-3-3-2-3                                                 | E TO FINE SAN<br>CES OF FRACTU                  | D, MOSTLY                    |  |  |
|                          | <br>_                                                                                                                                                                                                                              | S                  | 12               | 10-20-14<br>(7")               | 34                 | GP                  | GRAVELLY SAND, 20-30% COARSE TO FINE GRAVEL, FEW SA<br>TO 1-1/2 IN, ANGULAR TO ROUNDED, COARSE TO FINE SAN<br>FINES, TRACE IRON STAINING, GRAY.<br>BLOWS/INCH: 2-2-2-2-2-2/1-3-3-3-5-5/3-4-2-2-1-2                                                          | NDSTONE FRACH<br>D, 5-10% NONP                  | ENTS<br>LASTIC               |  |  |
|                          | 35                                                                                                                                                                                                                                 | S                  | 13               | 19-18-12<br>(4")               | 30                 | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED, MOSTLY O<br>STONE AND SHALE, 14 IN MAXIMUM, ANGULAR (SOME ROUND<br>FINE SAND, LESS THAN 5% NONPLASTIC FINES, TRACE IRC<br>BLOWS/INCH: 4-4-2-4-2-3/4-3-2-3-3-3/2-2-2-2-2-2                                        | OARSE WEATHER<br>ED), 15-20% C<br>N STAINING, G | ED SAND-<br>OARSE TO<br>RAY. |  |  |
|                          | 1111                                                                                                                                                                                                                               | S                  | 14               | 21-10-6<br>(8")                | 16                 | GP                  | SANDY GRAVEL, COARSE TO FINE, ROUNDED, CONTAINS SOM<br>AND SHALE FRAGMENTS TO 1 IN MAXIMUM, 20-302 COABSE<br>NONPLASTIC FINES, GRAY.<br>BLOWS/INCH: 4-3-5-3-3-3/2-2-2-1-2-1/1-1-1-1-1-1                                                                     | E WEATHERED S<br>TO FINE SAND,                  | ANDSTONE                     |  |  |
| 643.0                    | 643.0 40 $\frac{1}{5}$ 15 9-11-9 20 GW $\frac{\text{GRAVEL}}{\text{ROUNDED}, 10-153}$ COARSE TO FINE, FEW FRACHENTS TO 14 IN, ANGULAR TO ROUNDED, 10-153 COARSE TO FINE SAND, GRAY.<br>BLOWS/INCH: 2-2-1-1-2-1/2-2-2-2/2-1-1-2-1-2 |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                             |                                                 |                              |  |  |
|                          |                                                                                                                                                                                                                                    | s                  | 16               | 25-10-9<br>(11")               | 19                 | GN<br>SP            | TOP 6 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ANGULAR, CO<br>FRACMENTS TO 14 IN MAXIMUM, 30-35% COARSE TO FINE S<br>BOTTOM 5 IN: <u>SAND</u> , FOORLY GRADED, 5-10% COARSE TO F<br>COARSE SAND, GRAY.<br>BLOWS(JNCH. 5-4-7-4-2-3/2-2-2-2-1-1/2-1-2-1-1-2) | NTAINS SANDST<br>AND, GRAY.<br>INE GRAVEL, R    | ONE -<br>OUNDED, -           |  |  |
| NOTE :                   | FOR BO                                                                                                                                                                                                                             | RING<br>NFD        | SUMM<br>See      | LARY AND<br>SHEET L            |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE<br>TCH No. 12241-GSK-2478                                                                                                                                                                                             | BORING NO.<br>EOS-5                             | SHEET<br>2 OF 3              |  |  |

.

.

|                          |                |                    |             |                                |                    |                     | BORING NO                                                                                                    | <u>eos-5</u> |
|--------------------------|----------------|--------------------|-------------|--------------------------------|--------------------|---------------------|--------------------------------------------------------------------------------------------------------------|--------------|
| s                        | ITE            | BEAVER             | VALL        | EY POWER                       | STATIC             | N-UNI               | IT 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                    | •            |
| ELEVATION<br>(FEET)(162) | DEPTH          | SAMPLE<br>TYPF (7) | SAMPLE      | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                           |              |
| ļ                        | 45             | l s                | 17          | 30-15-7                        | 22                 | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGME                                         | NTS,         |
|                          |                | -                  |             | (9")                           |                    |                     | FEW TO 1-1/2 IN MAXIMUM, WEATHERED, SOFT, 30-40% COARSE TO FINE SAND<br>TRACE IRON STAINING, BROWN AND GRAY. | ', _         |
|                          |                | -                  | 1           |                                |                    |                     | NO RECOVERY                                                                                                  | -            |
|                          |                |                    | 18          | $\frac{60}{1''}$               | 1"                 |                     | NO RECOVERT.                                                                                                 | -            |
|                          |                | s                  | 19          | 69-26-90                       | 116                | GP                  | SIMILAR TO S-17, DARK GRAY SHALE AT BOTTOM, SOFT.                                                            |              |
| 633.0                    | 50 -           |                    | -           | (14")                          |                    |                     |                                                                                                              |              |
|                          |                | <u></u>            | 20          | 100                            | <u>100</u>         |                     | SHALE, SOFT, DARK CRAY.                                                                                      | -            |
|                          | .              | -                  |             |                                |                    |                     | BOTTOM OF BORING AT 51 FT 3 IN                                                                               | _            |
|                          |                | -                  |             |                                |                    |                     | ELEVATION 631.75                                                                                             | -            |
|                          | ĺ              |                    |             |                                |                    |                     |                                                                                                              | -            |
|                          | .<br>          |                    |             |                                |                    |                     |                                                                                                              |              |
|                          |                | 4                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | 4                  |             |                                |                    |                     |                                                                                                              | _            |
|                          |                | 1                  |             |                                |                    |                     |                                                                                                              |              |
|                          |                | 4                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                |                    |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | 4                  |             |                                |                    |                     |                                                                                                              |              |
|                          | -              | -                  |             |                                |                    |                     | · · · · ·                                                                                                    | _            |
|                          |                |                    |             |                                |                    |                     |                                                                                                              |              |
|                          |                | 1                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | 1                  |             |                                |                    |                     |                                                                                                              | _            |
|                          |                | -                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                |                    |             |                                |                    |                     |                                                                                                              |              |
|                          |                | -                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | -                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | _                  |             |                                |                    |                     |                                                                                                              | -            |
|                          |                | 1                  |             |                                |                    |                     |                                                                                                              | -            |
|                          | .              | -                  |             |                                |                    |                     |                                                                                                              | _            |
|                          |                |                    |             |                                |                    |                     |                                                                                                              | -            |
| <u> </u>                 |                |                    |             |                                |                    |                     |                                                                                                              |              |
| NOTE                     | for e<br>Legen | oring<br>Dinfo.    | SUMM<br>SEE | IARY AND SHEET I.              |                    | STO                 | DNE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHE<br>TCH No. 12241-GSK-247C DDA7 4/9/8- EOS-5 3          | DF 3         |

| <b></b>      |                         |                       |                |                    |              |                | <u> </u>                                                     | ·····                          |                                  |                                       | 0 E05-6     |  |  |  |
|--------------|-------------------------|-----------------------|----------------|--------------------|--------------|----------------|--------------------------------------------------------------|--------------------------------|----------------------------------|---------------------------------------|-------------|--|--|--|
|              | SITE .                  | BEA                   | VER VA         | LLEY POW           | ER STAT      | ION-U          | NIT 2                                                        | J.O. NO. 12241 BURET L OF 3    |                                  |                                       |             |  |  |  |
|              | COOR                    | NAT                   | 'ES .          | DPBLN              |              |                | GROUND EL                                                    | EV. (I)                        |                                  |                                       |             |  |  |  |
|              | INCLIN                  |                       | N              | EINIGH             | 6/8/8        | . 86<br>12     | ARING                                                        | NSPECTOR _                     | EGER/                            | IARVIS                                |             |  |  |  |
|              | UAIE<br>STATI           | : 314<br>- AB         |                | WATED              | OFPT         | <u>-</u> ни    | NOT NAT FRECORDED(PT) / NA                                   |                                |                                  | 8-45                                  | <u> </u>    |  |  |  |
|              | OFPT1                   | . 90<br>1 TO          |                | ROCX               | 48.1         |                |                                                              | TH DRILLED                     | )                                | 48.1                                  | (FT)        |  |  |  |
|              | METH                    |                       |                |                    |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              | <u>-</u>                | DRILL                 | ING            | SOL                | 3-1/8<br>    | IN R<br>IG, WA | OLLER BIT TO ADVANCE HOLE, 3<br>TER.                         | IN O.D. SPL                    | IT SPOON TO                      | CLEAN OUT, 4                          | • IN I.D.   |  |  |  |
|              | ;                       | SAMP                  | LING           | SOL                | _ 2 IN       | 0.D.           | SPLIT SPOON                                                  | <u> </u>                       |                                  |                                       |             |  |  |  |
|              | :                       | DRILI                 | LING           | ROCK               |              |                | ·····                                                        |                                |                                  |                                       |             |  |  |  |
|              | SPECI                   | AL T                  | EST            | NG OR              | INSTR        | UMEN           | TATION 2 PT NORTON POROUS                                    | PIEZOMETER I                   | NSTALLED WI                      | TH TIP AT EL                          | 710.1       |  |  |  |
|              |                         |                       |                |                    |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              | COMM                    | ENTS                  | I              |                    |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              |                         |                       |                |                    |              |                | ······································                       |                                | ··· ···                          |                                       |             |  |  |  |
| _            |                         | Ī                     | ন              |                    | _ 1          | 1              | 1                                                            |                                |                                  | • • •                                 |             |  |  |  |
| 10 N<br>(162 | ΞF                      | - u                   | 티백성            | n D g              | <u>ع</u> الج | 9              |                                                              |                                |                                  |                                       |             |  |  |  |
| EET ST       | E F I                   | MP                    | ŽI P           |                    | 2 E E        | NBOU           | SAMP                                                         | LE DESCR                       | RIPTION                          |                                       |             |  |  |  |
|              | ·                       | 5                     | - G 3          |                    | 2   ×        | s,             |                                                              |                                |                                  |                                       |             |  |  |  |
|              | <u> </u>                |                       |                |                    |              | <u> </u>       |                                                              |                                |                                  | · · · · · · · · · · · · · · · · · · · |             |  |  |  |
| 745.1        | 0                       | _ s                   | 1              | 1-3-3              | 6            | - 1            | TOPSOIL, SILT, LESS THAN 5%                                  | FINE SAND, 1                   | .5 IN SANDS                      | TONE FRAGMEN                          | T AT TIP,   |  |  |  |
|              |                         | 4                     |                | (6")               |              |                | DARK BROWN.                                                  |                                |                                  |                                       | -           |  |  |  |
|              |                         | +                     | 1              |                    |              | 1              |                                                              |                                |                                  |                                       | -           |  |  |  |
|              |                         |                       | 2              | 4-4-6              | 10           | a              | SANDY CLAY, MODERATELY PLAST                                 | TIC, STIFF, 1                  | 2% COARSE T                      | O FINE GRAVE                          | L SIZED -   |  |  |  |
|              |                         | 1                     | _              | (18")              |              |                | SANDSTONE, SHALE AND COAL FE<br>BROWN, MOTTLED WITH YELLOW F | RAGMENTS, ANG<br>BROWN AND GRA | ULAR, 22% C<br>Y.                | DARSE TO FIN                          | E SAND,     |  |  |  |
| į            |                         |                       | ┥,             | 4-7-8              | 115          | ci.            | SIMILAR TO S-2.                                              |                                |                                  |                                       | -           |  |  |  |
|              | 5                       | ┥                     |                | (18")              |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              |                         | Ŧ                     | 7              |                    |              |                |                                                              |                                |                                  |                                       | -           |  |  |  |
|              |                         | d s                   | 4              | 6-8-8              | 16           | CL             | SIMILAR TO S-2.                                              |                                |                                  |                                       | -           |  |  |  |
|              |                         |                       | -              | (11)               |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              |                         |                       | 5              | 6-6-8              | 14           | പ              | SIMILAR TO S-2.                                              |                                |                                  |                                       | -           |  |  |  |
|              |                         | 4                     |                | (11")              |              |                |                                                              |                                |                                  |                                       | -           |  |  |  |
| 735.1        | 10                      | $\pm$                 | 7.             |                    |              |                |                                                              |                                |                                  |                                       |             |  |  |  |
|              | 1                       |                       | 6              | (18")              |              | CL             | FRAGMENTS, MOIST, BROWN.                                     | S, STIFF, UGC                  | ASIONAL COA                      | NGE SAMU AND                          |             |  |  |  |
|              |                         | <u>+</u> -            |                | +                  | +            | CL/<br>ML      | SILTY CLAY, SLIGHTLY PLASTIC                                 | C, 4% VERY FI                  | NE SAND, BR                      | own.                                  | -           |  |  |  |
|              |                         | <b>-</b> s            | 7              | 3-2-3              | 5            | <u>м.</u>      | TOP 4 IN: SANDY SILT. NONPLA                                 | ASTIC TO SLIG                  | HTLY PLAST                       | C. 15-20% FT                          | NE SAND.    |  |  |  |
|              | 1                       | 1_                    |                |                    | _/           | GP             | WET, BROWN.<br>MIDDLE 6 IN: SANDY GRAVET                     | COARSE TO FTN                  | E GRAVEL ST                      | ZED SANDSTON                          | E AND SHALE |  |  |  |
|              |                         | + <u>s</u>            | - 8            | 6-8-5              | 13           | <b>_</b>       | FRAGMENTS, 1 IN MAXIMUM, AN                                  | NGULAR TO ROU                  | MDED, 20-30                      | Z COARSE TO                           | FINE SAND,  |  |  |  |
|              | 15                      | 1                     |                | (14")              |              | SP             | BOTTOM 4 IN: SILTY SAND, UN                                  | FORM, FINE,                    | 10-157 NONP                      | LASTIC FINES                          | , BROWN.    |  |  |  |
|              | . DATU                  | M IS                  | MEAN           | SEA LE             | EVEL         |                | UNDISTURBED SAMPLES                                          |                                | BUDING                           | 1.06                                  |             |  |  |  |
| 3            | BLOW                    | IS RE                 | QUIRE          | D TO DR            | IIVE         |                | UO-OSTERBERG                                                 |                                | BURING                           |                                       | _           |  |  |  |
| E S          | Z'O.D.<br>DISTA         | NCE                   | PLE 3<br>Shows | POON 6"<br>N USING | Off.         |                |                                                              | BEAVER VA                      | ALLEY PON                        | VER STATI                             | ON UNIT-2   |  |  |  |
| Ş ₄          | <b>(40њ</b> .<br>. () Ц | HÂME<br>NCHES         | ier fi<br>S of | ALLING 3<br>SAMPLE | 50".         |                |                                                              | DUQ                            | UESNE LI                         | GHT COM                               | PANY        |  |  |  |
|              | RECO                    | VERY                  | TRAT           | ION RES            | STANCE       |                |                                                              | SHIPP                          | INGPORT,                         | PENNSYL                               | VANIA       |  |  |  |
|              | BLOW                    | 9/FT.                 |                | ACCIEIO            |              | -              |                                                              | <b></b>                        |                                  |                                       |             |  |  |  |
| <b>.</b>     | SYST                    | CD 3*<br>EM.<br>1 F T | VAE            |                    |              |                |                                                              | STON SKET                      | NE & WEBS1<br><u>I'CH_No.</u> 12 | FER ENG. CC<br>241-GSK-248A           | RP.         |  |  |  |
| "]"          | S-SP                    | LITE                  | ARRE!          |                    | .8           |                |                                                              | APPROVED                       | DATE                             | BORING NO.                            | SHEET       |  |  |  |
| 1            |                         |                       |                |                    |              |                |                                                              | אפרע                           | 9/1/82                           | 205-6                                 | OF 3        |  |  |  |

.

|                                   |                 |                   |                  |                                |                    |                     |                                                                                                                                                                                                                                                                                                          |                                            |                                            | BORING NO                              | <u>EOS6</u>       |  |
|-----------------------------------|-----------------|-------------------|------------------|--------------------------------|--------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------|--------------------------------------------|----------------------------------------|-------------------|--|
|                                   |                 | AVER              | VALLI            | TY POWER S                     | τάτιο              | N-UNI               | T 2, SHIPPINGPORT, PA.                                                                                                                                                                                                                                                                                   |                                            | 12241.0                                    | 0                                      | /* <u></u> _      |  |
| S<br>S                            | не <u>—</u>     |                   | 1                |                                |                    |                     |                                                                                                                                                                                                                                                                                                          | 0.0. 1                                     |                                            |                                        |                   |  |
| ELEVATION<br>(FEET)(IE2           | DEPTH<br>(feet) | SAMPLE<br>TYPE (7 | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPI                                                                                                                                                                                                                                                                                                    | LE DESCR                                   |                                            |                                        |                   |  |
|                                   |                 |                   |                  |                                |                    |                     |                                                                                                                                                                                                                                                                                                          |                                            |                                            |                                        |                   |  |
|                                   | 15 -            | s                 | 8                |                                |                    |                     |                                                                                                                                                                                                                                                                                                          |                                            |                                            |                                        |                   |  |
|                                   |                 | S                 | 9                | 3-4-5<br>(17")                 | 9                  | SP<br>SP            | TOP 6 IN: <u>SIMILAR TO S-8</u> , BO<br>BOTTOM 11 IN: <u>SAND</u> , COARSE T<br>FINES, BROWN, CONTAINS OCCAS<br>PLASTIC, BROWN.                                                                                                                                                                          | TTOM 4 IN.<br>O FINE, MOST<br>IONAL POCKET | LY MEDIUM TO<br>OF SILTY CL                | FINE, 7-87 1<br>AY, MODERATE           | NONPLASTIC -      |  |
|                                   | -               | s                 | 10               | 5+6-7<br>(15")                 | 13                 | SP                  | SAND, POORLY GRADED, 5-7% FI<br>7-12 % NONPLASTIC FINES, MOIS                                                                                                                                                                                                                                            | NE GRAVEL, R<br>T, BROWN.                  | OUNDED, MEDI                               | UM TO FINE S.                          | AND,              |  |
| 725.1                             | 20              | s                 | 11               | 4-3-5<br>(15")                 | 8                  | SP<br>ML            | TOP 2 IN: <u>SIMILAR TO S-10</u> .<br>BOTTOM 13 IN: <u>SILT</u> , NONPLAST                                                                                                                                                                                                                               | IC, TRACE FI                               | NE GRAVEL SI                               | ZED SANDSTON                           | E AND             |  |
|                                   |                 |                   |                  |                                |                    |                     | COAL, SOME LENSES OF SANDY S                                                                                                                                                                                                                                                                             | ILT, MOIST,                                | BROWN .                                    |                                        |                   |  |
|                                   | -               | s                 | 12               | 4-6-9<br>(18")                 | 15                 | ML                  | <u>SILT</u> , NONPLASTIC, TRACE FINE                                                                                                                                                                                                                                                                     | SAND, WET,                                 | BROWN.                                     |                                        | ļīļ               |  |
|                                   | 25              | s.                | 13               | 4-3-4<br>(18'')                | 7                  | ML-<br>Sm           | LAYERED SILT AND SILTY FINE S<br>NONPLASTIC FINES, WET, BROWN                                                                                                                                                                                                                                            | SAND, TRACE                                | FINE GRAVEL                                | SIZED ROCK F                           | RAGMENTS,         |  |
|                                   |                 | s                 | 14               | 2-3-4<br>(18")                 | 7                  | CL                  | <u>SILTY CLAY-CLAYEY SILT,</u> SLIG<br>SAND, BROWN.                                                                                                                                                                                                                                                      | HTLY TO MODE                               | RATELY PLAST                               | TC, 1% VERY                            | FINE              |  |
|                                   | -               | s                 | 15               | 2-2-2<br>(10")                 | 4                  | SH                  | <u>SILTY SAND</u> , UNIFORMLY GRADED,<br>TO SLIGHTLY PLASTIC FINES, WE                                                                                                                                                                                                                                   | ), 20-25% NON                              | PLASTIC                                    |                                        |                   |  |
| 715.1                             | 30 -            | 30 5 16           | 16               | 3-7-5                          | 12                 | SP<br>GP            | P 11 IN: <u>SAND</u> , UNIFORMLY GRADED, FINE, 5-7% NONPLASTIC FINES, WET, BROWN."<br>TITOM 7 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ANGULAR TO ROUNDED, FEW<br>UNDSTONE FRADMENTS TO 1.5 IN, SOME COAL, 20-30% COARSE TO FINE SAND, 5% -<br>ICHTY PLASTIC FINES, TRACE IRON STAINING, BROWN, ORANGE. |                                            |                                            |                                        |                   |  |
|                                   | _               |                   |                  | (18'')                         |                    | SM                  | TOP 10 IN: SILTY SAND, 5-10%                                                                                                                                                                                                                                                                             | COARSE TO FI                               | INE GRAVEL S                               | IZED COAL FRA                          | GMENTS            |  |
|                                   | -               | s                 | 17               | 5-4+4<br>(18")                 | 8                  | GP -<br>GW          | TO 1 IN, FINE SAND, 15-202 N<br>BOTTOM 8 IN: <u>SANDY GRAVEL</u> , CO<br>SANDSTONE FRAGMENTS TO 1 IN,<br>NONPLASTIC FINES, BROWN.                                                                                                                                                                        | DARSE TO FIN<br>15-26% COARS               | NES.<br>E GRAVEL, AN<br>SE TO FINE S.      | SULAR TO ROUN<br>AND, LESS THA         | IDED, FEW -       |  |
|                                   |                 | s                 | 18               | 4-5-5                          | 10                 | SP-<br>SM           | SILTY SAND, FINE, TRACE FINE<br>FINES, SANDSTONE FRAGMENTS AT                                                                                                                                                                                                                                            | GRAVEL AND (<br>BOTTOM.                    | COAL FRAGMEN                               | rs, 10–15% אמ                          | ONPLASTIC         |  |
|                                   | 35-             |                   | ļ                | (18")                          | /                  | SP-<br>SM<br>GP     | TOP 13 IN:SIMILAR TO S-18.<br>BOTTOM 5 IN: <u>SANDY GRAVEL</u> , CO                                                                                                                                                                                                                                      | OARSE TO FIN                               | E GRAVEL SIZ                               | ED SANDSTONE                           | AND SHALE -       |  |
|                                   | -               | s                 | 19               | 7-8-11<br>(18")                | 19                 |                     | FRAGMENTS TO I IN MAXIMUM, AN<br>SAND, TRACE IRON STAINING, B                                                                                                                                                                                                                                            | NGULAR TO ROUROWN, GRAY,                   | UNDED, 15-20<br>BLACK.                     | Z COARSE TO I                          | TINE              |  |
|                                   | -               |                   |                  |                                | Ľ.                 | SP                  | GRAVELLY SAND, 25-35% COARSE<br>COAL, ANGULAR TO ROUNDED, ME<br>TRACE IRON, BROWN, GRAY.                                                                                                                                                                                                                 | TO FINE GRA<br>DIUM TO FINE                | VEL SIZED SA<br>SAND, 5-10%                | NONPLASTIC                             | TNES,             |  |
|                                   | -               | S                 | 20               | 49- <u>81</u><br>2"            | <u>81</u><br>2"    | GP                  | TOP 13 IN: SANDY GRAVEL, COA                                                                                                                                                                                                                                                                             | RSE TO FINE,                               | ROUNDED, SO                                | ME BROKEN SAL                          | NDSTONE           |  |
| 705.1 40 - S 21 26-34-17 51 (18") |                 |                   |                  |                                |                    |                     | COAL, BROWN, GRAY, ORANGE BR<br>BUTTOM 5 IN: <u>GRAVEL</u> , BROWN S.<br>GRAY.                                                                                                                                                                                                                           | ANDSTONE PRA                               | CMENTS TO 1.                               | 5 1N,SAMPLED                           | COBBLE,           |  |
|                                   | -               | s                 | 22               | 20 <b>-16</b> -103<br>(18")    | 119                | GM                  | TOP 12 IN: <u>SILTY GRAVEL</u> , COAR<br>ANGULAR, 25-30% COARSE TO FI<br>BROWN.                                                                                                                                                                                                                          | SE TO FINE G<br>NE SAND, 15-               | RAVEL, MOSTL<br>20% SLIGHTLY               | Y COARSE TO<br>PLASTIC FIN             | I IN,<br>ES, WET, |  |
|                                   | -               |                   |                  |                                |                    | GP                  | BOTTOM 6 IN: <u>SANDSTONE FRAGM</u><br>BLOWS/INCH: 3-3-2-4-4-4/2-2-                                                                                                                                                                                                                                      | <u>ents</u> , sample<br>1-2-3-6/5-4-       | 30-34-18-12                                |                                        |                   |  |
|                                   | 45              | s                 | 23               | 33-107-33<br>(11")             | 140                | GP                  | SANDY GRAVEL, COARSE TO FINE<br>1.5 IN MAXIMUM, 20-25% COARS<br>FINES, TRACE MICA, TRACE IRO                                                                                                                                                                                                             | GRAVEL SIZE<br>E TO MEDIUM<br>N STAINS, BR | D SANDSTONE<br>SAND, 5-10%<br>OWN, GRAY, C | FRAGMENTS, S<br>SLIGHTLY PLA<br>RANGE. | OME COAL,         |  |
| NOTE :                            | FOR BO          | RING<br>NFO.      | SLIMIN<br>SEE    | ARY AND SHEET I.               |                    | STO                 | NE & WEBSTER ENG. CORP.<br>TCH No. 12241-GSK-2483                                                                                                                                                                                                                                                        | APPROVED                                   | DATE<br>9/./82                             | BORING NO.<br>EOS-6                    | SHEET<br>2 OF 3   |  |

|                          |                 |                    |                  |                |                      |                    |                     |                                                                                                                                          | SHEET 3 | OF_ |
|--------------------------|-----------------|--------------------|------------------|----------------|----------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------|---------|-----|
| SI                       | TE              | AVER               | VALLE            | ey poi         | VER S                | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241                                                                                                    | .00     |     |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(Feet) | SAMPLE<br>TYPE (7) | SAMPLE<br>Number | (E) SMOTE      | 0R<br>REC/RQD (4)    | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                       |         |     |
| I                        | 45              | S                  | 24               | 36-2<br>(13")  | 3-41<br>)            | 69                 | æ                   | SIMILAR TO S-22, TOP.                                                                                                                    |         |     |
|                          |                 | S                  | 25               | 21-7:<br>(11") | L+ <u>103</u><br>15" | 174<br>75"         | GP                  | TOP 5 IN: <u>SIMILAR TO S-22</u> , TOP.<br>MIDDLE 2 IN: <u>SANDSTONE FRAGMENTS</u> , SOPT, GRAY.<br>BOTTOM 4 IN: <u>COAL FRAGMENTS</u> . |         |     |
|                          |                 |                    |                  |                |                      |                    |                     | BOTTOM OF BORING AT 48.1 FT<br>Elevation 697.0 FT                                                                                        |         |     |
|                          | 1111            |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          |                 |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          | 111             |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          | .               |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
| i<br>t                   | بانيب           |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          |                 |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          | LLLL            |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          | TITI            |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          | 1 1 1 1 1       |                    |                  |                |                      |                    |                     |                                                                                                                                          |         |     |
|                          |                 |                    |                  | -              |                      |                    | L                   |                                                                                                                                          |         | 1   |

|          | 9                                                                          | ITE              | BEAVE         | R VAL                                                                                            | LEY POWER          | STATI    | (ON-U)     | IIT 2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | . J.O. NO.                            | 12241                         | BORING N      | <b>b.</b> <u>EOS-7</u> |  |  |  |  |
|----------|----------------------------------------------------------------------------|------------------|---------------|--------------------------------------------------------------------------------------------------|--------------------|----------|------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------|-------------------------------|---------------|------------------------|--|--|--|--|
|          | c                                                                          | OORDI            | NATE          | s _                                                                                              | N3812              |          |            | GROUND EL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | .EV. (I)                              | 759.9                         | SHEET         | OF _2                  |  |  |  |  |
|          | IN                                                                         | ICLINA           | TION          |                                                                                                  | VERTICAL           |          | . 88       | ARING NA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | SPECTOR                               | JW McCOY                      |               |                        |  |  |  |  |
|          | D,                                                                         | ATE:             | STAF          | RT / F                                                                                           | INISH _            | 6/3/8    | 2          | / _6/3/82 CONTRACTO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | R / DRILLI                            | ER _EGER/JAE                  | RVIS          |                        |  |  |  |  |
|          | S                                                                          | TATIC            | GRO           | UND                                                                                              | WATER              | DEPT     | H/D        | ATE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | DRILL RIG                             | TYPE                          | E 45          |                        |  |  |  |  |
|          | D                                                                          | ЕРТН             | то            | BEDF                                                                                             | ROCK               | 44       | . 5        | (FT) TOTAL DEPT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | H DRILLE                              | D45.                          | .0            | (FT)                   |  |  |  |  |
|          | N                                                                          | ETHO             | ) <b>s</b> :  |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            | DF               |               | NG :                                                                                             | SO4L _3            | 1/8      | IN RO      | LLER BIT TO ADVANCE HOLE, 3 I                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | N O.D. SPLI                           | T SPOON TO CL                 | EAN OUT       |                        |  |  |  |  |
|          |                                                                            | S                | AMPL          | ING                                                                                              | <b>SOL</b> _2      | IN O     | .D. S      | PLIT SPOON                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                       |                               |               | <del></del>            |  |  |  |  |
|          | s                                                                          | DI<br>PECIAI     | RILLI<br>L TE | ING P<br>Stin                                                                                    | ROCK<br>IG OR IN   | ISTRI    | JMEN       | TATION _2 FT POROUS STONE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | PIEZOMETER,                           | INSTALLED WI                  | TH TIP AT EL  | , 716.9                |  |  |  |  |
| ľ        | COMMENTS BORING ADVANCED WITHOUT WATER. DID NOT ENCOUNTER ANY GROUNDWATER. |                  |               |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          | COMMENTS BORING ADVANCED WITHOUT WATER. DID NOT ENCOUNTER ANY GROUNDWATER. |                  |               |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            |                  |               |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            |                  |               |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
| F        | <u>N</u>                                                                   |                  | Ē             | 1                                                                                                | ~ ~                | T        | 6          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
| <u>8</u> | 1)(6                                                                       | •ТН<br>Е Т)      |               | j<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L<br>L |                    | Z        | چ<br>بر او |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       | <b></b>                       |               |                        |  |  |  |  |
| ¥        | FEE                                                                        | DEF              | A M           | AMP                                                                                              |                    | SP 1     | NBO N      | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                       |                               |               |                        |  |  |  |  |
| 13       | Ţ                                                                          |                  | 5             | o z                                                                                              | REC B              | ≍        | ۍ ا        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
| ⊢        |                                                                            |                  |               |                                                                                                  | 1                  |          | <b></b>    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
| 750      | 0                                                                          | 0                | T             | 1                                                                                                | r                  |          | 1          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ··· · · · · · · · · · · · · · · · · · |                               |               |                        |  |  |  |  |
| , ,,,    | .,                                                                         | •                | s             | 1                                                                                                | 4-7-9<br>(5")      | 16       | -          | FILL, SLAG AND SILTY GRAVE<br>GRAY.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | L, COARSE T                           | O FINE, TRACE                 | ROOTS AND I   | RON STAINS,            |  |  |  |  |
|          |                                                                            | -                |               |                                                                                                  | ·- /               |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               | -                      |  |  |  |  |
|          |                                                                            |                  | s             | 2                                                                                                | 4-7-6              | 13       | н          | GRAVELLY SILT, SLIGHTLY PLA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | STIC, 10-15                           | Z COARSE TO P                 | INE GRAVEL S  | IZED -                 |  |  |  |  |
|          |                                                                            | -                |               |                                                                                                  | (11")              |          |            | WEATHERED SANDSTONE AND SHA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | LE, ROUNDED                           | TO SUBANGULA                  | R, 15-207 CC  | ARSE -                 |  |  |  |  |
|          |                                                                            | -                |               | ]                                                                                                |                    |          |            | TOP 8 IN: GRAVELLY SILT-GRA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | VELLY CLAY.                           | SLIGHTLY TO                   | MODERATELY F  | LASTIC, -              |  |  |  |  |
|          |                                                                            | 5 -              | s             | 3                                                                                                | 6-5- <b>6</b>      | 11       |            | 20-30% COARSE TO FINE GRAVE<br>BOTTOM 10: COAL AND SHALE F                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | L, SOME WOO<br>RAGMENTS, W            | D FRACMENTS,<br>IDELY GRADED. | GRAY AND BRO  | ₩N. +<br>'INE          |  |  |  |  |
|          |                                                                            | -                |               | 4                                                                                                | (18")              |          |            | GRAVEL AND SAND SIZED FRAGMENTS, TRACE IRON STAINING.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            | -                |               | 1                                                                                                |                    | ł        |            | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, STIFF, CONTAINS FEW LAYER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            | -                | s             | 4                                                                                                | 7-6-5<br>(16")     | 11       | CL         | OF COAL FRAGMENTS AND SANDS<br>SHALE FRAGEMENTS, 7-10% COA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | TONE FRAGME<br>RSE TO FINE            | NTS TO 1.5 IN<br>SAND, VERY S | N MAXIMUM, FE | W RED<br>T. BROWN.     |  |  |  |  |
|          |                                                                            |                  |               | 1                                                                                                |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            | •                | s             | 5                                                                                                | 4-7-6              | 13       | CL         | <u>SIMILAR TO 5-4</u> , MOTTLED BRO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | WN AND ORAN                           | GE.                           |               | -                      |  |  |  |  |
|          |                                                                            |                  | 1             |                                                                                                  | (10)               |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               | -                      |  |  |  |  |
| 749.     | 9                                                                          | 10 —             | -             |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               |                        |  |  |  |  |
|          |                                                                            |                  | s             | 6                                                                                                | 3-5-8<br>(13")     | 13       | CL         | SIMILAR TO S-4, CONTAINED 1<br>FRACTION, MOTTLED GRAY AND                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | IN THICK L<br>BROWN.                  | AYER OF SILTY                 | CLAY WITHOU   | T COARSE               |  |  |  |  |
|          |                                                                            | -                | <u> </u>      |                                                                                                  |                    |          | 1          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               |               | -                      |  |  |  |  |
|          |                                                                            | _                |               | ],                                                                                               | 7_9_9              | 14       |            | SANDY CLAY, SLIGHTLY PLASTI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | C, STIFF, O                           | CCASIONAL FIN                 | E GRAVEL SIZ  | ED _                   |  |  |  |  |
|          |                                                                            | -                |               | ′                                                                                                | (16")              |          |            | STAINING, SLIGHTLY MOIST, B                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ROWN.                                 | THE ONEO, OUT                 | - ALMOR LRUP  | -                      |  |  |  |  |
|          |                                                                            | -                |               | ]                                                                                                |                    |          | _          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       |                               | 10 169        | CDAILE                 |  |  |  |  |
|          |                                                                            | 15               | s             | 8                                                                                                | 4-7-7<br>(13")     | 14       | CL         | <u>SANDY CLAY</u> , SLIGHTLY TO MDI<br>TO 3/4 IN MAXIMUM, ANGULAR,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ERATELY PLA<br>15-20% COA             | RSE TO FINE S                 | SAND, BROWN.  | GRAVEL _               |  |  |  |  |
|          | 1.<br>2.                                                                   |                  | IS NUND       | IEAN<br>WATE                                                                                     | SEA LEV<br>R LEVEL | EL       |            | UNDISTURBED SAMPLES<br>US-SHELBY TUBE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | _                                     | BORING                        | LOG           | _                      |  |  |  |  |
| ø        | з.                                                                         | 2°0.0. 9         |               | UIREO<br>.E SP                                                                                   | 000 6 0            | ne<br>NR | A          | SAMPLE CONTAINS PIFCES OF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                       |                               |               |                        |  |  |  |  |
| TE       |                                                                            | DISTAN           | CE SI         | HOWN<br>R FA                                                                                     | USING<br>Lling 30  | <b>#</b> | а.         | SANDSTONE 1.5 IN DIAMETER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | BEAVER                                | ALLEY PO                      | VER STATI     | ON UNIT-2              |  |  |  |  |
| N        | 4.                                                                         | ( ) IN(          | HES           | OF                                                                                               | SAMPLE             | •        |            | AND 1/8 IN THICK, INDICATING<br>SAMPLER PENETRATED COBBLE<br>SUPERINGER SUPERINGER SUPER SUPERINGER SUPERINGER SUPERINGER SUPERINGER SUPERINGER SUPERINGER SUPER SUPERINGER SUPERINGER SUPER SUPER SUPERINGER SUPERINGER SUPER |                                       |                               |               |                        |  |  |  |  |
|          | 5,                                                                         | STD. P           | ENET          | RATI                                                                                             | ON RESIS           | TANCE    | :          | OR BOULDER. TYPICAL OF<br>THIS MATERIAL.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                       | PINGPORT,                     | PENNSYL       | VANIA                  |  |  |  |  |
| ND       | 6.                                                                         | UNIFIE           | /FT.<br>D SQI |                                                                                                  | ASSIFICA           | TION     |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | A STO                                 | WE & WERGI                    | ER ENG CO     | RP.                    |  |  |  |  |
| GE       | 7.                                                                         | SYSTER<br>SAMPL: | А.<br>Е ТҮ    | PE:                                                                                              |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SKI                                   | TCH No. 1224                  | 1-GSK-249A    | ,                      |  |  |  |  |
| Ľ        | -                                                                          | S-SPL            | TBA           | RREL                                                                                             | SAMPLE             | :        |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | APPROVED                              | DATE                          | BORNG NO.     |                        |  |  |  |  |
|          |                                                                            |                  |               |                                                                                                  |                    |          |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                       | 1 41/1 -                      | 1 203-7       | 1 ' <del>-</del> ' 4   |  |  |  |  |

| BORING NO.               |                 |                    |                                       |                                |                    |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                  |  |  |  |  |
|--------------------------|-----------------|--------------------|---------------------------------------|--------------------------------|--------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| ł                        |                 |                    |                                       |                                |                    |                                                                                                                                                                                                       | SHEET 2 OF 2                                                                                                                                                                                                                                                                                     |  |  |  |  |
| s                        |                 | BEAVER             | VALL                                  | EY POWER S                     | TATIO              | N-UNI                                                                                                                                                                                                 | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                         |  |  |  |  |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(FFFT) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER                      | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6)                                                                                                                                                                                   | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                               |  |  |  |  |
|                          | <u>.</u>        |                    | · · · · · · · · · · · · · · · · · · · |                                | <u> </u>           |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                  |  |  |  |  |
|                          | 15              | 5                  | 8                                     |                                | ]                  |                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                  |  |  |  |  |
|                          |                 | s                  | ,                                     | 6-6-6<br>(14")                 | 12                 | CL.                                                                                                                                                                                                   | TOP 10 IN: <u>SILTY CLAY</u> , SLIGHTLY TO MODERATELY PLASTIC, STIFF, 15-20Z<br>COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACMENTS TO 1.5 IN<br>MAXIMUM, FEW COAL FRAGMENTS, ORANGE, BROWN AND GRAY.                                                                                      |  |  |  |  |
|                          |                 |                    | 10                                    | 19-17-12                       |                    | <u>е</u>                                                                                                                                                                                              | BOTTOM 4 IN: <u>SILTY CLAY</u> , MEDIUM STIFF, MODERATELY PLASTIC, TRACE FINE<br>SAND, MOIST, BROWN. (SINILAR TO ABOVE BUT WITHOUT COARSE FRACTION).                                                                                                                                             |  |  |  |  |
| 739.9                    | 20              |                    |                                       | (7")                           |                    | GP                                                                                                                                                                                                    | SANDI LRAVEL, FOORLI GRADED, COARSE GRAVEL SIZED SANDSTONE FRAGMENTS,<br>MOSTLY 1.5 IN, 20-25% COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM,<br>5% NONPLASTIC FINES, BROWN. (CONTAINED LAYER OF SOFT CLAYEY SILT AT<br>TOP OF SAMPLE). (SEE NOTE 8).                                             |  |  |  |  |
|                          |                 | s                  | 11                                    | 4-15-17<br>(14")               | 32                 | SP                                                                                                                                                                                                    | TOP 10 IN: <u>SILTY SAND</u> , SLIGHTLY PLASTIC, 15-207 COARSE TO FINE GRAVEL<br>SIZED SANDSTONE AND SHALE FRAGMENTS, FEW TO 1 IN MAXIMUM, ANGULAR,<br>15-202 SLIGHTLY PLASTIC FINES TRACE COAL BEAM                                                                                             |  |  |  |  |
|                          | -               | s                  | 12                                    | 99-10-8                        | 18                 | GP<br>SP                                                                                                                                                                                              | BOTTOM 4: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS, .<br>30-40% COARSE TO FINE SAND, 5% NONPLASTIC FINES, LIGHT GRAY.<br>TOP 5 IN: SIMILAR TO S-11, TOP.                                                                                                            |  |  |  |  |
|                          |                 |                    |                                       | (10")                          |                    | GP                                                                                                                                                                                                    | BOTTOM 5 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE<br>FRAGMENTS TO 1 IN MAXIMUM, 20-30% COARSE TO FINE SAND, 5% NONPLASTIC                                                                                                                                                 |  |  |  |  |
|                          | 25 -            | s<br>-             | 13                                    | 9-59-26<br>(10")               | 85                 | GP                                                                                                                                                                                                    | LAYERED SANDY GRAVEL AND SANDY CLAYEY SILT, SANDY GRAVEL, COARSE TO FINE<br>GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN, ANGULAR, 30-40% COARSE TO<br>FINE SAND, LESS THAN 5% NONPLASTIC FINES, TAN (SAMPLED COBBLE).                                                                             |  |  |  |  |
| Ì                        |                 | s                  | 14                                    | 6-7-10<br>(14")                | 17                 | ML.<br>GP                                                                                                                                                                                             | CLAYEY SILT, SLIGHTLY TO MODERATELY FLASTIC, SOFT, 10-15% COARSE TO FINE<br>GRAVEL, ANGULAR, BROWN.<br>TOP 6 IN: <u>SANDY GRAVEL</u> , COARSE GRAVEL SIZED SANDSTONE FRACMENTS TO 1.5<br>IN MAXIMUM. 20-25% COARSE TO FINE SAND. LIGHT GRAV.                                                     |  |  |  |  |
|                          | -               |                    | 15                                    | 9-9-11                         | 20                 | M                                                                                                                                                                                                     | BOTTOM 8 IN: <u>GRAVELLY SILT</u> , SLIGHTLY PLASTIC, 12-15% COARSE TO FINE<br>GRAVEL SIZED SANDSTONE, SHALE AND COAL FRAGMENTS, 1.5 IN FRAGMENT AT TIP,<br>LESS THAN 5% FINE SAND, TRACE IRON STAINS, BROWN.<br>SANDY GRAVEL, WIDTLY GRADED, COARSE TO FINE GRAVEL, MOSTLY COARSE TO 1 IN.      |  |  |  |  |
| 729.9                    | 10 -            |                    |                                       | (12")                          |                    |                                                                                                                                                                                                       | ANGULAR TO SUBROUNDED SANDSTONE, 202 COARSE TO FINE SAND, 15-20%<br>NONSLIGHTLY PLASTIC FINES, BROWN.                                                                                                                                                                                            |  |  |  |  |
|                          |                 | s<br>              | 16                                    | 8-9-8<br>(12")                 | 17                 | GP                                                                                                                                                                                                    | <u>SIMILAR TO S-15</u> .                                                                                                                                                                                                                                                                         |  |  |  |  |
|                          | -               | s                  | 17                                    | 7-14-14<br>(12")               | 28                 | GP                                                                                                                                                                                                    | SIMILAR TO S-15, SAMPLED COBBLE, SOME FRAGMENTS ROUGHLY THE DIAMETER OF<br>SAMPLER.<br>BLOWS/INCH: 1-1-1-1-2/1-1-2-1-4-5/4-1-3-2-2-2                                                                                                                                                             |  |  |  |  |
| -                        | 35 -            | s                  | 18                                    | 12-14-11<br>(14")              | 25                 | GP                                                                                                                                                                                                    | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN<br>MAXIMUM, 15-20% COARSE TO FINE SAND, 7% NONPLASTIC FINES, GRAY. SOME<br>POCKETS OF SILTY FINE SAND, BROWN.<br>BLOWS/INCH: 2-1-2-3-2-2/1-3-3-3-2-2/2-2-2-2-1                                                           |  |  |  |  |
|                          |                 | <br>               | 19                                    | 10-9-13<br>(12'')              | 22                 | SP                                                                                                                                                                                                    | <u>GRAVELLY SAND</u> , 15-25Z COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, FEW<br>SANDSTONE FRAGMENTS TO 1.5 IN, FINE SAND, LESS THAN 5% NONPLASTIC FINES,<br>BROWN.                                                                                                                               |  |  |  |  |
|                          | -               | s                  | 20                                    | 4-7-25<br>(14")                | 32                 | SP<br>SP<br>CP                                                                                                                                                                                        | BLOWS/INCH: 2-1-2-1-2-2/1-2-1-2-2-1/2-1-2-4-2-2<br>TOP 3 IN: <u>SAND</u> , UNIFORM, FINE, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>MIDDLE 10 IN: <u>SAND</u> , UNIFORM, MEDIUM TO FINE, TRACE FINE GRAVEL, COAL<br>FRAGMENTS, IRON STAINING, BROWN.<br>BOTTOM I IN: <u>SANDSTONE FRAGMENTS</u> . |  |  |  |  |
| 719.9                    | 40 -            | s                  | 21                                    | 32-24-24<br>(14")              | 46                 | GP                                                                                                                                                                                                    | SANDY CRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS<br>TO 1.5 IN MAXIMUM, 20-25% COARSE TO FINE SAND, LESS THAN 5% NONPLASTIC<br>FINES, TRACE COAL AND IRON STAINING, LIGHT GRAY AND BROWN. SANDSTONE<br>FRAGMENT AT TIP. BLOWS/INCH: 3-7-3-5-4-10/6-3-4-5-3-3/3-4-3-4-5-5   |  |  |  |  |
|                          | -               |                    | 22                                    | 100/5"                         | 100<br>5"          | SP <u>GRAVELLY SAND</u> , POORLY GRADED, 15-202 COARSE TO FINE GRAVEL TO 1 IN<br>MAXIMUM, ANGULAR, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 10-122<br>NON TO SLIGHTLY PLASTIC FINES, MOIST, BROWN. |                                                                                                                                                                                                                                                                                                  |  |  |  |  |
|                          |                 |                    | 2/                                    | (17")                          | 52                 |                                                                                                                                                                                                       | SANDY GRAVEL, SIMILAR TO S-15, AT 7 IN. FROM TOP - 2 IN. THICK SEAM OF<br>FINE SAND, 15-20% NONPLASTIC FINES, MOIST, BROWN.                                                                                                                                                                      |  |  |  |  |
|                          | 45              | <u> </u>           |                                       | (0")                           | <u></u>            |                                                                                                                                                                                                       | END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                                                |  |  |  |  |
| NOTE :                   | FOR B<br>LEGEN  | ORING<br>NFQ.      | SUMM                                  | SHEET I.                       |                    | SKE                                                                                                                                                                                                   | TCH NO. 12241-GSK-249R                                                                                                                                                                                                                                                                           |  |  |  |  |

|      |          |                     |                  | VAT f         | EV POURS                                                                                      | STATI        | 0N-11N  | <br>[† 2                         |                                |                               | 2241                                  |                               | 0. <u>EOS-7A</u>                             |  |  |
|------|----------|---------------------|------------------|---------------|-----------------------------------------------------------------------------------------------|--------------|---------|----------------------------------|--------------------------------|-------------------------------|---------------------------------------|-------------------------------|----------------------------------------------|--|--|
|      | 3        |                     |                  | e i           | N3814.6                                                                                       |              | E       | 6136.2                           |                                | FLEV (1) 759.6 FT. SHEETOF _2 |                                       |                               |                                              |  |  |
|      |          |                     |                  | 3 _<br>\      | ERTICAL                                                                                       |              | <br>8F  |                                  |                                |                               | J.W. McCOY                            |                               |                                              |  |  |
|      | D        | ATE :               | STAR             |               | INISH _                                                                                       | 6/3/8        | 2       | / <u>6/3/82</u>                  | CONTRACTO                      | R / DRILLE                    |                                       | RVIS                          |                                              |  |  |
| ļ    | s        | TATIC               | GRO              | UND           | WATER                                                                                         | DEPT         | H / D.  | ATE NA (FT)                      | /                              | DRILL RIG 1                   |                                       | 5 45                          |                                              |  |  |
|      | D        | ЕРТН                | то               | BEDF          |                                                                                               | A.           |         | (FT)                             | TOTAL DEPT                     | H DRILLED                     | 24.                                   | 5                             | <u>(FT)</u>                                  |  |  |
|      | M        | ЕТНО                | DS :             |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               |                                              |  |  |
| ŀ    |          | O                   | RILLI            | NG            | sol _3                                                                                        | 1/8          | IN 0.1  | D. ROLLER BIT TO                 | ADVANCE HOLE                   | <u>3 IN O.D.</u>              | SPLIT SPOON                           | TO CLEAN OUT                  |                                              |  |  |
|      |          | S                   |                  | ING           | <b>SOL</b> _2                                                                                 | IN O         | .D. 51  | PLIT SPOON AND 3                 | IN O.D. SHEL                   | BY_TUBE                       |                                       |                               |                                              |  |  |
| ĺ    |          | 0                   | RILLI            | NG F          | юск —                                                                                         |              |         |                                  |                                |                               |                                       |                               |                                              |  |  |
|      | S        | PECIA               | L TE             | STIN          | ig or in                                                                                      | ISTRI        | JMEN    | TATION _2 FT                     | NORTON POROUS                  | PIEZOMETER                    | INSTALLED W                           | ITH TIP AT E                  | <u>L. 738.1</u>                              |  |  |
|      |          | -                   |                  | DR            |                                                                                               |              | HVPST   | 0F F0S-7                         |                                | ····                          |                                       |                               |                                              |  |  |
|      | C        | OMME                | NIS.             | 51            |                                                                                               |              | 114 00  | 01 000                           |                                |                               | · · · · · · · · · · · · · · · · · · · |                               |                                              |  |  |
| ĺ    |          | -                   |                  |               | •                                                                                             |              |         |                                  |                                |                               |                                       |                               |                                              |  |  |
| ⊨    | ~        |                     | 1 ~              | 1             |                                                                                               | <u> </u>     |         |                                  |                                |                               |                                       |                               |                                              |  |  |
| B    | )(16.2   | F₽                  |                  | u e           | (a)<br>(a)<br>(b)<br>(b)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c)<br>(c | <b>]</b> 🧕 🖸 | (9)     |                                  |                                |                               |                                       |                               |                                              |  |  |
| ١.   | EET      | 143 (<br>143 -      | JAP 1            | NAPI<br>NBR   | OWS<br>VD/O                                                                                   |              | ROU     |                                  | SAMP                           | LE DESCR                      | RIPTION                               |                               |                                              |  |  |
| ELE  | 5        |                     | 5                | 13 Z          | EC A BL                                                                                       | ¶¢ ≹         | 9 I S   |                                  |                                |                               |                                       |                               |                                              |  |  |
|      |          |                     |                  | <b>I</b>      | <u> </u>                                                                                      |              |         |                                  | =                              | <u></u>                       |                                       |                               | <u>.                                    </u> |  |  |
| 759  | 6        | 0                   | <b>—</b>         |               |                                                                                               | 1            | <b></b> | ·                                |                                | <u> </u>                      |                                       |                               |                                              |  |  |
|      |          | •                   | 7                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          |                     | 1                | ľ             | l                                                                                             |              |         | NO SAMPLES T                     | 07 FT.                         |                               |                                       |                               | -                                            |  |  |
|      |          | -                   | -                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | _                                            |  |  |
|      |          |                     | 1                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          |                     | -                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          | 5 -                 | 7                |               | }                                                                                             |              |         |                                  |                                |                               |                                       |                               |                                              |  |  |
|      |          |                     | 1                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          |                     | 4                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          | -                   |                  | ],            | (168)                                                                                         |              |         |                                  |                                |                               |                                       |                               | _                                            |  |  |
|      |          |                     | <u>ן</u> ״       |               | (15*)                                                                                         |              |         |                                  |                                |                               |                                       |                               | -                                            |  |  |
|      |          |                     | ╉──              |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               | _                                            |  |  |
| 749  | .6       | 10 -                |                  | 2             | (25.5%)                                                                                       |              |         | 8                                |                                |                               |                                       |                               | _                                            |  |  |
|      |          |                     | -                | -             | (=3.13 )                                                                                      |              |         | -                                |                                |                               |                                       |                               | -                                            |  |  |
|      |          |                     | - s              | ι,            | 10-7-6                                                                                        | 13           | CL/     | SANDY CLAY-SAN                   | DY SILT, SLIC                  | HTLY PLASTIC                  | . STIFF. 20-                          | 25% COARSE "                  | TO FINE                                      |  |  |
|      |          | _                   | 1                |               | (13'')                                                                                        |              | ML.     | SAND, 10% FINE                   | GRAVEL TO 1/                   | 4 IN, MOIST,                  | BROWN.                                | CON CONTRE                    |                                              |  |  |
|      |          |                     | +                |               |                                                                                               |              |         |                                  |                                |                               |                                       |                               |                                              |  |  |
|      |          |                     | <b>1</b> °       | 2             | (10")                                                                                         | 14           | CL      | SANDY CLAY, SL<br>OCCASIONAL COA | IGHTLY TO MOD<br>RSE GRAVEL TO | ERATELY PLAS                  | TIC, STIFF.<br>CARSE TO FIN           | 10% FINE GR.<br>HE SAND, MOI: | AVEL,                                        |  |  |
|      |          | 15                  |                  |               |                                                                                               |              |         | BROWN.                           |                                |                               |                                       |                               |                                              |  |  |
|      | Į.       | DATUN               | I IS M           | EAN           | SEA LEV                                                                                       | EL           |         | UNDISTURBED                      | SAMPLES                        |                               |                                       |                               |                                              |  |  |
|      | 2.<br>3. |                     | DUNO 1<br>9 REGI | WATE<br>JIREO | TO DRIV                                                                                       | E            |         | US-SHELBY<br>UO-OSTERB           | TUBE<br>Erg                    |                               | BORING                                | LOG                           | -                                            |  |  |
| ŝ    |          | Z <sup>H</sup> O.D. |                  | E SP          | 00N 6" 0                                                                                      | R            |         |                                  |                                |                               |                                       | VED GTATI                     |                                              |  |  |
| FO   | A        | 140Ib.              | HAMME            | RFA           | LLING 30                                                                                      | н,<br>,      |         |                                  |                                |                               | UESNE II                              | GHT COM                       |                                              |  |  |
| Z    | -        | RECOV               | ERY.             |               | DAMPLE                                                                                        |              |         |                                  |                                | SHIPF                         | PINGPORT.                             | PENNSYL                       |                                              |  |  |
| 0    | э.       | STD.  <br>BLOWS     | PENET<br>B/FT.   | RATI          | UN RESIS                                                                                      | TANCE        |         |                                  |                                |                               |                                       |                               |                                              |  |  |
| SE N | 6.       | UNIFI               | ED SOI<br>M.     | L CL          | ASSIFICA                                                                                      | TION         |         |                                  |                                | STON                          | E E WEBST                             | ER ENG. CO                    | RP.                                          |  |  |
| Ē    | 7.       | SAMPL<br>S-SPL      | .E TYI<br>.IT BA | PÉ:<br>RRÉL   | SAMPLE                                                                                        |              |         |                                  |                                | APPROVED                      | DATE                                  | BORING NO.                    | SHEET                                        |  |  |
| -    |          | -                   |                  |               |                                                                                               |              |         |                                  |                                |                               | 9/./0-                                | EOS-7A                        | 1 OF 2                                       |  |  |

| SI                       | TE              | AVER               | VALLE            | Y POWER S                      | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                |
|--------------------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                      |
|                          |                 |                    |                  |                                | 1                  |                     |                                                                                                                                                         |
|                          | 15              | 5                  | 3                | 5-6-9<br>(13")                 | 15                 | CL                  | SIMILAR TO 5-2, 20-30% COARSE TO FINE GRAVEL TO 1 IN.                                                                                                   |
|                          |                 | s                  | 4                | 4-6-6                          | 12                 | сL                  | TOP 13 IN: SILTY CLAY, MODERATELY PLASTIC, MEDIUM STIFF, MOTTLED (                                                                                      |
|                          |                 |                    |                  | (16")                          |                    | ML.                 | AND BROWN.<br>BOTTOM 3 IN: <u>SILT</u> , LOOSE, TRACE FINE SAND, WET, BROWN.                                                                            |
| 730 6                    | 20 _            | s                  | 5                | 11-15-14                       | 29                 | GP                  | SANDY GRAVEL, WEATHERED SANDSTONE FRAGMENTS TO 1 IN MAXIMUM, 25-30                                                                                      |
| ,,,,,,,                  | -               |                    |                  | (13 )                          |                    |                     | COAL AND IRON STAINING, BROWN AND GRAY.                                                                                                                 |
|                          | -               | S                  | 6                | 20-20-8<br>(18'')              | 28                 | GP                  | GRAVEL, COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN MAXIMUM<br>LIGHT GRAY. CONTAINS POCKETS OF SANDY SILT, 10-15% FINE SAND, VEI<br>MOIST, BROWN. |
|                          | -               | 5                  | 7                | 8-11-18                        | 29                 | GP                  | SIMILAR TO S-6.                                                                                                                                         |
|                          | -               |                    |                  | (13")                          |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     | ELEVATION 735.1 FT                                                                                                                                      |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          |                 |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
|                          | -               |                    |                  |                                |                    |                     |                                                                                                                                                         |
| IOTE: I                  | FOR BO          | RING               | SUMM             | ARY AND                        |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. S                                                                                                      |

| C                          | OORDINATES                                                                                                                                            | N3944                                                                                            |                         | · _                       | E6185 GROUND EL                                                                                                                                                                                                                            | EV. (1) 732.7                                                                              | SHEETOF _3                                                |  |  |
|----------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|-------------------------|---------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------|-----------------------------------------------------------|--|--|
| IN<br>D<br>S<br>D<br>M     | ICLINATION .<br>ATE:START<br>TATIC GROUI<br>EPTH TO BU<br>ETHODS:<br>DRILLING                                                                         | VERTICAL<br>/ FINISH 5<br>NDWATER 0<br>EDROCK<br>3 SOIL 3                                        | -19+8;<br>DEPTI<br>52.0 | BE<br>2<br>H / D<br>IN RO | ARING NA INSPECTOR J.W. MCCOY   / 5/20/82 CONTRACTOR / DRILLER EGER DRILLING/JARVIS   NOT ATERECORDED(FT) / DRILL RIG TYPE CME 45   (FI) TOTAL DEPTH DRILLED 52.0 (FT)   VLLER BIT, 3-1/4 IN I.D. CASING, DRILLING MUD Statement Statement |                                                                                            |                                                           |  |  |
| s                          | SAMPLIN<br>DRILLIN<br>PECIAL TES<br>OMMENTS                                                                                                           | IG SOIL 2<br>3 ROCK                                                                              | .0 IN<br>STRU<br>G FLUI |                           | SPLIT SPOON     TATION   NONE     35.0 AND 40.0 FT                                                                                                                                                                                         |                                                                                            |                                                           |  |  |
| ELEVATION<br>(FEET)(IE2)   | DEPTH<br>(FEET)<br>Sample<br>TYPE (7)                                                                                                                 | NUMBER<br>BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                    | SPT N<br>VALUE (5)      | GROUP<br>SYMBOL (6)       | SAMP                                                                                                                                                                                                                                       | PLE DESCRIPTION                                                                            |                                                           |  |  |
| 732.7                      | ° - S                                                                                                                                                 | 1 2-5-3<br>(14")                                                                                 | 8                       | ML/<br>SM                 | SANDY SILT, DENSE, SLIGHTLY<br>GRADING TO SILTY SAND, TRAC                                                                                                                                                                                 | MOIST, FEW SANDSTONE FF<br>E FINE GRAVEL, 30-402 NG                                        | AGMENTS AND ROOTS,<br>NNPLASTIC FINES, BROWN.             |  |  |
|                            | -<br>-<br>-<br>-<br>-<br>-<br>-                                                                                                                       | 2 4-4-6 (18")                                                                                    | 10                      | 쩐                         | <u>SILT</u> , NONPLASTIC TO SLIGHT<br>FEW SMALL SAND SEAMS, WET, 7                                                                                                                                                                         | LY PLASTIC, 0-5% FINE S<br>BROWN.                                                          | ND, TRACE ORGANICS,                                       |  |  |
| -                          | 5 <u>-</u> S                                                                                                                                          | 3 (16")                                                                                          | 9                       | ML<br>SP                  | TOP 13 IN: <u>SIMILAR TO ABOVE</u><br>BOTTOM 3 IN: <u>SAND</u> , FINE, FE<br>FRAGMENTS TO 0.5 IN, 0-5% No                                                                                                                                  | W FINE GRAVEL AND WEATHI<br>ONFLASTIC FINES, BROWN.                                        | ERED SANDSTONE                                            |  |  |
|                            | 5                                                                                                                                                     | 4 4-4-4<br>(16")                                                                                 | 8                       | SP<br>ML                  | TOP 13 IN: <u>SAND</u> , COARSE TO<br>GRAVEL, 0-5% NOMPLASTIC FIN<br>BOTTOM 3 IN: <u>SILT</u> , NOMPLAST<br>BLDWS/INCH: 1-1-1/2-1/2//1-                                                                                                    | FINE, MOSTLY COARSE TO P<br>ES, BROWN.<br>IC TO SLIGHTLY PLASTIC,<br>1/2-1/2-1/1/2-1/2-1-1 | EDIUM, 2-5% FINE<br>BROWN.                                |  |  |
| 722.7                      |                                                                                                                                                       | 5 3-3-4<br>(14")                                                                                 | 7                       | SP<br>SP                  | TOP 4 IN: <u>SILTY SAND</u> , FINE,<br>FINES, MOIST, BROMN.<br>BOTTOM 10 IN: SAND, COARSE<br>GRAVEL, 5% NONPLASTIC FINES<br>BLOWS/INCH: 1-1/2-1/3//1/2-                                                                                    | TRACE COARSE-MEDIUM SAN<br>TO FINE, MOSTLY COARSE 1<br>, MOIST, BROWN.<br>1/2-1/2/11-1/2-1 | ND, 15-20% NONPLASTIC                                     |  |  |
|                            |                                                                                                                                                       | 5 6-4-3<br>(18")                                                                                 | 7                       | ଗ୍ୟ                       | CRAVELLY SAND, WELL CRADED<br>TO FINE, SUBANGULAR TO ROUN.<br>FINES, TRACE COAL, BROWN.<br>BLOWS/INCH: 1-1-1-1-1//4/                                                                                                                       | . 20-302 COARSE TO FINE<br>Ded, coarse to fine sant<br>/1/2-1/2-1/2                        | GRAVEL, MOSTLY MEDIUM<br>D, TRACE NONPLASTIC              |  |  |
| I.<br>2.<br>3.<br>4.<br>5. | DATUM IS MEA<br>GROUND WA<br>BLOWS REQUIN<br>2"Q.D. SAMPLE<br>DISTANCE SHO<br>14010. HAMMER<br>() INCHES OI<br>RECOVERY.<br>STD. PENETRA<br>BLOWS/FT. | AN SEA LEVI<br>AN SEA LEVI<br>RED TO DRIVI<br>SPOON 6" OI<br>WWN USING<br>FALLING 30<br>F SAMPLE | I J<br>EL<br>R<br>TANCE |                           | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG                                                                                                                                                                                      | BORING<br>BEAVER VALLEY PO<br>DUQUESNE L<br>SHIPPINGPORT                                   | LOG<br>WER STATION UNIT<br>IGHT COMPANY<br>, PENNSYLVANIA |  |  |

|                          |                                                                                                            |           |             |              |                   |                  |                                                                                                                                                                                                                                   | BORING NO. EOS-9                                                                                                                                                                                                                                                                                                     |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|--------------------------|------------------------------------------------------------------------------------------------------------|-----------|-------------|--------------|-------------------|------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| l                        |                                                                                                            |           |             | 11.11.1      | ev paura a        | - <b>.</b> - • • | N_11517                                                                                                                                                                                                                           | T 2 SHIPPINGPORT PA                                                                                                                                                                                                                                                                                                  |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| SI                       | TE.                                                                                                        | BE        | AVER        | VALL         | LI POWER S        | 14110            | N-UN1                                                                                                                                                                                                                             | J.O. NO                                                                                                                                                                                                                                                                                                              |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| ELEVATION<br>(FEET)(162) | ELEVATION<br>(FEET)(16.2)<br>DEPTH<br>(FEET)<br>SAMPLE<br>TYPE<br>NUMBER<br>BLOWS (3)<br>OR<br>REC/ROD (4) |           |             |              |                   |                  | GROUP<br>SYMBOL (6)                                                                                                                                                                                                               | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                   |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| <u> </u>                 |                                                                                                            |           |             |              |                   |                  |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                      |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          | 15                                                                                                         | 1.1.1     | s           | 7            | 4-3-3<br>(18")    | 6                | SM<br>SP                                                                                                                                                                                                                          | TOP 9 IN: SILTY SAND, 10-15% COARSE TO FINE GRAVEL, SUBANGULAR TO ROUNDED,<br>COARSE TO FINE SAND, MOSTLY FINE, 10-15% NONPLASTIC FINES.<br>BOTTOM 9 IN: <u>SAND</u> , FINE, 2-6% FINE GRAVEL, 0-5% NONPLASTIC FINES, TRACE<br>COARSE SAND SIZED COAL FRACHENTS, BROWN.<br>BLOWS/INCH: 1-1/2-1/2-1/2/1/2-1/2-1/2-1/2 |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            |           | s           | 8            | 9-11~10<br>(1B")  | 21               | CW<br>V                                                                                                                                                                                                                           | SANDY GRAVEL, WIDELY GRADED, SUBANGULAR TO ANGULAR WEATHERED SANDSTONE<br>FRAGMENTS TO 1 IN MAXIMUM, 25-35% COARSE TO FINE SAND, 5-10% NONPLASTIC<br>FINES, TRACE COAL, FEW IRON STAINS, BROWN.                                                                                                                      |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| 712.7                    | 20                                                                                                         | CLILI     | S           | 9            | 4-3-2<br>(16")    | 5                | SP                                                                                                                                                                                                                                | GRAVELLY SAND, 20-30% COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS,<br>MAXIMUM SIZE 1 IN, ANGULAR TO ROUNDED, COARSE TO FINE SAND, MOSTLY FINE,<br>5-10% SLIGHTLY PLASTIC FINES, TRACE GOAL FRAGMENTS, IRON STAINS AT BOTTON,<br>WET AT BOTTOM, BROWN.                                                            |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            | TITI      | s           | 10           | 9-10-13<br>(16")  | 23               | GW                                                                                                                                                                                                                                | SANDY GRAVEL, WEATHERED SANDSTONE AND SHALE FRAGMENTS TO 1 IN MAXIMUM,<br>ANGULAR, 15-25% COARSE TO FINE SAND, 2-5% NONPLASTIC FINES, IRON STAINS,<br>MOIST, BROWN AND GRAY.                                                                                                                                         |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          | 25                                                                                                         |           | S           | 11           | 27-17-13<br>(12") | 30               | CP                                                                                                                                                                                                                                | GRAVEL, WEATHERED SANDSTONE AND LIMESTONE(?) FRAGMENTS TO 1-1/2 IN,<br>ANGULAR TO SUBANGULAR, SOME IRON STAINING, 5-10% NONPLASTIC FINES, TRACE<br>SHALE FRAGMENTS, DRY, BROWN.                                                                                                                                      |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            |           | S           | 12           | 5-9-19<br>(13")   | 28               | SP<br>SP                                                                                                                                                                                                                          | TOP 3 IN: <u>SAND</u> , UNIFORM, FINE, TRACE FINE GRAVEL, TRACE NONPLASTIC FINES,<br>BROWN.<br>MIDDLE 1 IN: <u>SAME AS ABOVE</u> , DARK BROWN.                                                                                                                                                                       |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| 702.7                    | 30                                                                                                         |           | s           | 13           | 11-15-15<br>(13") | 30               | SM                                                                                                                                                                                                                                | BOTTOM, JIGHT BORNN.<br>BOTTOM, JIGHT BORNN.<br>BLOWS/INCH: 1-2-1/2-1/2//1-1-2-2-2/0-4-3-3-3-3<br>TOP 4 IN: <u>SILTY SAND</u> , 10-152 FINE GRAVEL TO 1/2 IN, ANGULAR TO SUBROUNDED, -<br>COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 25-302 NORPLASTIC FINES, DRY.                                                  |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            |           |             | ;            | 1111              | s                | 14                                                                                                                                                                                                                                | 9-7-14                                                                                                                                                                                                                                                                                                               | 22 | SW<br>SM                                                                                                                                                                                                                     | BOTTOM 9 IN: SAND, WELL GRADED, COARSE TO FINE, 0-5% FINE GRAVEL, 0-5%<br>NONPLASTIC FINES, PEM SANDSTONE FRACMENTS TO 1 IN MAXIMUM, IRON STAINING,<br>BROWN. BLOWS/INCH: 2-2-2-1-2-2/3-2-3-3-2-2/3-1-3-3-3-2<br>TOP 15 IN: SIMILAR TO S-13, TOP 4 IN. |  |  |
|                          | 35                                                                                                         | 1111      |             |              | (18")             | 27               | SP<br>SM                                                                                                                                                                                                                          | BOTTOM 3 IN: <u>SAND</u> , COARSE TO FINE, MOSTLY COARSE TO MEDIUM, TRACE FINE<br>GRAVEL, MOIST, BROWN.<br>BLOWS/INCH: 1-2-2-1-2-1/1-1-1-1-1-2/1-2-2-3-3-3                                                                                                                                                           |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            |           | •-          |              |                   |                  | (18")                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                      | GP | BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, 1 IN MAXIMUM, ANGULAR<br>TO ROUNDED, 15-202 COARSE TO FINE SAND, MOSTLY FINE, 10-152 NONPLASTIC<br>FINES, BROWN.<br>BLOWS/INCH: 2-2-3-2-2/2-2-3-2-2-2/3-2-3-3-2-2 |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            | 1.1.1     | S           | 16           | 16-20-25          | 45               | GM                                                                                                                                                                                                                                | SILTY GRAVEL, COARSE TO FINE GRAVEL, FEW TO 1 IN MAXIMUM, ANGULAR TO<br>ROUNDED, 1 IN SANDSTONE FRAGMENTS AT BOTTOM, 10-15% COARSE TO FINE SAND,<br>MOSTLY FINE, 15-20% NONPLASTIC FINES, DRY, BROWN (SIMILAR TO S-13, TOP<br>4 IN). BLOWS/INCH: 2-2-3-3-3-3/4-3-2-3-3-5/4-3-2-2-7-7                                 |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| 692.7                    | 40                                                                                                         |           | S           | 17           | 5-5-8<br>(8")     | 13               | GW                                                                                                                                                                                                                                | SANDY GRAVEL, COARSE TO FINE GRAVEL, FEW TO 1 IN MAXIMUM, ANGULAR TO<br>ROUNDED, 15-20% COARSE TO FINE SAND, 0-5% SLIGHTLY PLASTIC FINES, MOIST,<br>BROWN.<br>BLOWS/INCH: 1/2-1-1-1-1//1-1-1/2-1-1//1-1-2-2-1-1                                                                                                      |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
|                          |                                                                                                            |           | ş           | 18           | 12-19-22<br>(16") | 41               | GP SANDY GRAVEL, MOSTLY LARGE, WEATHERED SANDSTONE AND SHALE FRAGMENTS TO<br>1-1/2 IN, SOME SHALE FRAGMENTS, 15-20% COARSE TO FINE SAND, 2-5%<br>NONPLASTIC FINES, MOIST, BROWN.<br>BLOWS/INCH: 3-2-2-2-2-1/2-2-2-4-5/7-3-3-4-3-2 |                                                                                                                                                                                                                                                                                                                      |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |
| NOTE:                    | FOR                                                                                                        | 80<br>240 | RING<br>NFQ | SUMAN<br>SEE | LARY AND SHEET I. |                  | STO<br>SKE                                                                                                                                                                                                                        | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TCH No. 12241-65K-251B DD H 9/1/82 EOS-9 2 OF 3                                                                                                                                                                                                            |    |                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                        |  |  |

| ĺ                        |                                                                                                              |                  |                      |                                                                                     |                                              |            | BORING NO. EOS-9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |  |  |  |
|--------------------------|--------------------------------------------------------------------------------------------------------------|------------------|----------------------|-------------------------------------------------------------------------------------|----------------------------------------------|------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|                          |                                                                                                              |                  |                      |                                                                                     |                                              |            | SHEET 3 OF 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
| SI                       | SITE BEAVER VALLEY POWER STATION-UNIT 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                 |                  |                      |                                                                                     |                                              |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| ELEVATION<br>(FEET)(162) | ELEVATION<br>(FEET)((E.2)<br>DEPTH<br>(FEET)<br>SAMPLE<br>SAMPLE<br>NUMBER<br>BLOWS (3)<br>OR<br>REC/RQD (4) |                  |                      |                                                                                     |                                              |            | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |  |
|                          |                                                                                                              |                  |                      |                                                                                     |                                              |            |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |  |  |  |
| 682.7                    |                                                                                                              | 5<br>5<br>5<br>5 | 19<br>20<br>21<br>23 | 13-9-19<br>(11")<br>45/6"<br>(5")<br>30-23-20<br>(10")<br>15-98/3"<br>(6")<br>30/0" | 28<br>45/6 <sup>°</sup><br>98/3 <sup>°</sup> |            | TOP 4 IN: <u>GRAVEL</u> , SANDSTORE PRACHEMENTS TO 1-1/2 IN, SCHE SHALE PRACHENTS,<br>5-107 COARSE TO FINE SAND, 5-107 SLIGHTLY PLASTIC FINES, WIT, BRACH.<br>MIDDLE 3 IN: <u>SAND CARVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE<br>PRACHEMENS, 5-107 SLIGHTLY PLASTIC FINES, WIT, BROCH.<br>BLONSTINGT. 4-2-2-11/2-2-2-1-2/2-12/2-3-2-5<br>SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACHENTS,<br>MOUNDED TO ANULAR, LARGE SANDSTONE RRACHENT ROTTON, LOI-51 COARSE<br>TIME SAND, 2-53 SLIGHTLY PLASTIC FINES, NOIST, GRANGE, BLACK AND BROWN.<br>BLONS/INGT: 8-4-2-7-1-1-7<br>SANDY GRAVEL, SINILAR TO ABOVE, MAXIMUM PARTICLE SIZE 1-1/2 IN, 5-73<br>SLIGHTLY PLASTIC FINES, BROWN.<br>BLONS/INGT: 9-4-4-7-7-5/(53-4-2-2/4-3-2-3-3-5)<br>YFATTMEED SHALE, 10-155 FINE SAND, 10-153 SLIGHTLY PLASTIC TO MEDIUM<br>PLASTIC FINES, ORANGE, BLACK, GRAY BROWN.<br>REFUSAL<br>BOTTOM OF BORING AT 52.0 FT<br>ELEVATION 680.7 FT |  |  |  |
| NOTE:                    | FOR BO                                                                                                       | RING             | SUMM                 | ARY AND SHEET L                                                                     |                                              | STO<br>SKE | NE & WEBSTER ENG. CORP. APPROVED DATE BORNS NO. SHEET<br>TCH NO. 12241-658-251C DDA 9/1/82 E0S-9 3 OF 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |

|                | SITE BEAVER VALLEY POWER STATIO<br>COORDINATES N4097.3<br>INCLINATION VERTICAL<br>DATE : START / FINISH 6/10/8:<br>STATIC GROUNDWATER DEPTI<br>DEPTH TO BEDROCK NA<br>METHODS :<br>DRILLING SOL 3-1/8<br>SAMPLING SOL 2 IN 0<br>DRILLING ROCK NONE<br>SPECIAL TESTING OR INSTRU |                                                                                                                         |                                                                                                               |                                                                                                         |                                                                                              |                                                      |                     | T 2 J.O. NO. 2   6137.4 GROUND ELEV. (I)72   ARING INSPECTOR   / INSPECTOR   / 6/11/82 CONTRACTOR / DRILLE   ATE RECORDENT) / DRILL RIG I    INTOTAL DEPTH DRILLE   .D. ROLLER BIT, 4 IN I.D. CASING AND DRIL   SPLIT SPOON AND 3 IN O.D. SHELBY TUBE   TATION NONE | 12241<br>J. W. MCC<br>REGER/JARV<br>TYPE<br>66.5<br>.LINC_MUD                                    | BORING NO. 205-10<br>SHEETOF<br>207<br>7IS<br>55<br>(FT)<br>(FT)                                                          |
|----------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|------------------------------------------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------|
| ELEVATION      | (FEET)(162)                                                                                                                                                                                                                                                                     | DEPTH<br>(FEE T)                                                                                                        | SAMPLE                                                                                                        | TYPE (7)<br>SAMPLE                                                                                      | NUMBER<br>BLOWS (3)<br>AND/OR                                                                | RECOVERY (4)<br>SPT N<br>VALUE (M)                   | GROUP<br>SYMBOL (6) | SAMPLE DESCI                                                                                                                                                                                                                                                        | RIPTION                                                                                          |                                                                                                                           |
| 72             | 0.7                                                                                                                                                                                                                                                                             | 0                                                                                                                       | - 5                                                                                                           | 1                                                                                                       | 11-21-2<br>(13")                                                                             | 21 42                                                | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE TO 1 IN MAX<br>5-10% SLIGHTLY PLASTIC FINES, BROWN, GRA                                                                                                                                                                                | IMUM, 20-30Z<br>Y AND ORANGE.                                                                    | COARSE TO FINE SAND,                                                                                                      |
|                |                                                                                                                                                                                                                                                                                 | 5                                                                                                                       |                                                                                                               | 2                                                                                                       | 6-5-3<br>(11")                                                                               | 8                                                    | SP-<br>Sw           | <u>GRAVELLY SAND, 20-30% COARSE TO FINE GRA</u><br>COARSE TO FINE, MOSTLY MEDIUM TO FINE, 5                                                                                                                                                                         | VEL, FEW FRAG<br>-10% SLIGHTLY                                                                   | MENTS TO 1.5 IN,<br>PLASTIC FINES, BROWN                                                                                  |
| 71             | 0.7                                                                                                                                                                                                                                                                             | 10                                                                                                                      |                                                                                                               | 3                                                                                                       | 2-1-1<br>(10")                                                                               | 2                                                    | SP-<br>SV           | <u>Gravelly Sand,</u> 30-35% coarse to fine gra<br>Fine sand, Mostly medium to fine, 5-10%                                                                                                                                                                          | VEL, ANGULAR<br>SLIGHTLY PLAS                                                                    | TO ROUNDED, COARSE TO<br>TIC FINES, GRAY.                                                                                 |
|                |                                                                                                                                                                                                                                                                                 | 15                                                                                                                      |                                                                                                               | 4                                                                                                       | 5-5-5<br>(15")                                                                               | 10                                                   | SP∽<br>SW           | GRAVELLY SAND, 15-25% COARSE TO FINE GRA<br>ROUNDED, COARSE TO FINE SAND, MOSTLY MED<br>FINES, GRAY.                                                                                                                                                                | VEL, 1 IN MAX<br>IUM TO FINE,                                                                    | IHUM, ANGULAR TO<br>5-10% NONPLASTIC                                                                                      |
| LEGEND / NOTES | 1.<br>2.<br>3.<br>4.<br>5.<br>6.<br>7.                                                                                                                                                                                                                                          | DATUI<br>GR<br>BLOW<br>2"Q.D.<br>DISTA<br>4016.<br>( ) JI<br>RECO'<br>STD.<br>BLOW<br>UNIFIL<br>SYSTE<br>SAMPI<br>S-SPL | N IS<br>OUNC<br>S RE<br>SAMI<br>NCE<br>HAMM<br>NCHES<br>VERY<br>PENE<br>S/FT.<br>ED S<br>EM.<br>LE T<br>LIT E | MEAN<br>WATI<br>QUIRE<br>PLE SI<br>SHOWI<br>MER F/<br>SOF<br>TRAT<br>DIL CI<br>DIL CI<br>YPE:<br>DARREI | SEA LI<br>ER LEVE<br>D TO DA<br>POON 6<br>N USING<br>ALLING<br>SAMPLE<br>ION RES<br>LASSIFIC | EVEL<br>IL<br>IVE<br>30",<br>ISTANCE<br>CATION<br>LE |                     | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG<br>BEAVER VA<br>DUQ<br>SHIPF<br>SKE<br>APPROVED                                                                                                                                                               | BORING<br>ALLEY POW<br>UESNE LIG<br>PLNGPORT,<br>NE & WEBSTE<br>ICH NO. 12241<br>DATE I<br>9//82 | LOG<br>ER STATION UNIT-2<br>HT COMPANY<br>PENNSYLVANIA<br>ER ENG. CORP.<br>L-GSK-252A<br>BORNG NO. SHEET<br>EOS-10 I OF 3 |

|                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             | BORING NO                                                            | EOS-10            |  |
|------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------|------|------------------|----------|----|--------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------|---------------------------------------------|----------------------------------------------------------------------|-------------------|--|
|                                                                              | _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | <b></b>     |      |                  |          |    |                                                                                                                                                              | T 2 CHIDDINGDODT DA                                                                                                              |                                               | 17261 (                                     | SHEET 2                                                              | of                |  |
| SITE BEAVER VALLEY POWER STATION-UNIT 2, SHIPPINGPORT, PA. J.O. NO. 12241.00 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
| ELEVATION<br>(FEET)(162                                                      | ELE VATION<br>(FEET )(162)<br>C FEET )(162)<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPLE<br>SAMPL |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
|                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
|                                                                              | 15                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |             |      |                  |          |    |                                                                                                                                                              | ·                                                                                                                                |                                               |                                             |                                                                      | -                 |  |
|                                                                              | <u>-</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      | 4                 |  |
|                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | s           | 5    | 4-2-3<br>(9")    |          | 5  | SP                                                                                                                                                           | S <u>AND</u> , TRACE FINE GRAVEL, MOS<br>NONPLASTIC FINES, BROWN.                                                                | TLY MEDIUM T                                  | O FINE SAND,                                | , LESS THAN 5                                                        |                   |  |
| 700.7                                                                        | 20 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |             |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |             |      |                  |          |    | :                                                                                                                                                            |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | s           | 6    | 10-14-<br>(12'') | -26      | 40 | CP<br>G₩                                                                                                                                                     | SANDY GRAVEL, COARSE TO FINE<br>COARSE TO FINE SAND, MOSTLY<br>BOTTOM 3 IN: BROKEN LIGHT GR                                      | GRAVEL TO 1<br>COARSE TO ME                   | .5 IN, ANGUL<br>DIUM, TRACE<br>FRACMENTS 1  | AR TO ROUNDE<br>IRON STAININ<br>TO 1.5 IN.                           | D, 25-35 <b>%</b> |  |
|                                                                              | 25 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |             |      |                  |          |    |                                                                                                                                                              | BLOWS/INCH: 2-1-1-2-2-2/1-1-                                                                                                     | -1-2-5-4/4-3-                                 | 6-4-3-6                                     |                                                                      |                   |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | s           | 7    | 18-23-           | -36      | 59 | HEL.                                                                                                                                                         | TOP 7 IN: GRAVELLY SILT, 15-                                                                                                     | -20% COARSE T                                 | O FINE GRAVE                                | SL, MOSTLY ME                                                        |                   |  |
| 690.7                                                                        | 30                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |             |      | (17")            | 1        |    | G?-<br>GV                                                                                                                                                    | FINE, ANGULAR TO SUBANGULAR,<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> ,<br>BROKEN SANDSTONE, 25-357 COA<br>AND IRON STAINING, BROWN. | 5-102 FINE<br>COARSE TO FI<br>RSE TO FINE     | SAND, VERY I<br>NE GRAVEL, J<br>Sand, TRACE | DRY, BROWN.<br>L.5 IN, ANGUI<br>NONPLASTIC F                         | AR, SOME          |  |
|                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |             |      |                  |          |    |                                                                                                                                                              | 550#3/1NGH: 1-4-3-3-4-3/4-3-                                                                                                     |                                               | u-y- <b>-</b> -y                            |                                                                      | 11                |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | s           | 8    | 2-4-5<br>(15")   |          | 9  | 9 SM TOP 5 IN: <u>SILTY SAND</u> , 10-15% COARSE TO FINE<br>SAND, SOME MEDIUM AND COARSE, 10-15% NONPLAS<br>CL. BOTTOM 10 IN: SILTY CLAY, SLIGHTLY TO MODERA |                                                                                                                                  |                                               | INE GRAVEL,<br>PLASTIC FINE<br>DERATELY PLA | AVEL, SUBANGULAR, FINE<br>C FINES, BROWN.<br>LY PLASTIC, STIFF, 5-72 |                   |  |
|                                                                              | -<br>35 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | <b> </b>    | 1    | -                |          |    |                                                                                                                                                              | COARSE TO FINE GRAVEL, SOME<br>GRAVISH BROWN.                                                                                    | ROOTS, POCKE                                  | TS OF COAL I                                | FRAGMENTS, MO                                                        | UST, DARK         |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | S           | 9    | 3-5-6<br>(17")   |          | 11 | HCL                                                                                                                                                          | CLAYEY SILT, SLIGHTLY TO MOU<br>SANDSTONE AND COAL FRAGMENTS<br>TRACE ROOTS, GRAY. q <sub>u</sub> (pp):                          | PERATELY PLAS<br>, FEW SANDST<br>1.25, 1.75TS | TIC, TRACE I<br>ONE FRAGMENT<br>F           | FINE GRAVEL S<br>IS TO 1 IN NE                                       | IZED<br>AR TOP,   |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | S           | 10   | 3-4-6<br>(13")   |          | 10 | CL<br>CL                                                                                                                                                     | TOP 4 IN: <u>CLAYEY SILT-SILTY</u><br>BOTTOM 9 IN: <u>SILTY CLAY</u> , SLI<br>HOIST, GRAY BROWN. qu (pp):                        | CLAY, SIMILA<br>GRTLY TO MOD<br>1.5, 1.75TSF  | R TO 5-9.<br>Erately play                   | STIC, MEDIUM                                                         | STIFF,            |  |
| 680.7                                                                        | 40 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | υs          | I    | (23.5'           | ')       |    | CL                                                                                                                                                           | SILTY CLAY, SLIGHTLY TO MODE<br>MOIST, BROWN. (TUBE TRIMMINGS                                                                    | RATELY PLAST                                  | TC, OCCASION                                | NAL GRAVEL TO                                                        | 1 IN, -           |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | s           | 11   | 4-3-3<br>(18")   |          | 6  | CL.                                                                                                                                                          | SILTY CLAY, SLIGHTLY TO MODE<br>FINE GRAVEL TO 1/2 IN, SOME<br>q <sub>u</sub> (pp): 1.75, 2.00 TSF                               | RATELY PLAST<br>FINE SAND, M                  | IC, MEDIUM S<br>OIST, BROWN                 | STIFF, OCCASI                                                        |                   |  |
|                                                                              | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | US          | 2    | (23")            |          |    |                                                                                                                                                              | <u>SIMILAR TO S-11.</u> (TUBE TRIM                                                                                               | (INGS)                                        |                                             |                                                                      |                   |  |
|                                                                              | 45                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | <u> </u>    |      |                  |          |    |                                                                                                                                                              |                                                                                                                                  |                                               |                                             |                                                                      |                   |  |
| NOTE:                                                                        | FOR BO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | NRNG<br>NFQ | SUMM | ARY A            | ND<br>I. |    | STO                                                                                                                                                          | NE & WEBSTER ENG. CORP.<br>TCH No.12241-CSK-252B                                                                                 | APPROVED                                      | DATE<br>9/182                               | BORING NO.                                                           | SHEET             |  |

|                          | _                                                                            |                              |             |                                |                    |                     |                                                                                                                                                              | BORING NO                   | EOS-10      |  |  |
|--------------------------|------------------------------------------------------------------------------|------------------------------|-------------|--------------------------------|--------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------|-------------|--|--|
|                          |                                                                              |                              |             |                                |                    |                     |                                                                                                                                                              | SHEET 3 (                   | 0F          |  |  |
| s                        | SITE BEAVER VALLEY POWER STATION-UNIT 2, SHIPPINGPORT, PA. J.O. NO. 12241.00 |                              |             |                                |                    |                     |                                                                                                                                                              |                             |             |  |  |
| ELEVATION<br>(FEET)(162) | DEPTH                                                                        | (FEET)<br>Sample<br>Type (7) | SAMPLE      | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                           |                             |             |  |  |
|                          | - 7 e                                                                        |                              |             | T                              |                    |                     |                                                                                                                                                              |                             |             |  |  |
|                          | 43                                                                           |                              | 12          | 4-9-11<br>(6")                 | 20                 | SP<br>SW            | GRAVELLY SAND, 15-20% COARSE TO FINE GRAVEL, ROUNDED<br>TO FINE SAND, 3-5% NONPLASTIC FINES, BROWN                                                           | ) TO SUBANGULA              | R, COARSE - |  |  |
|                          |                                                                              | - s                          | 13          | 8-7-9<br>(8")                  | 16                 | SP-<br>SW           | <u>GRAVELLY SAND</u> , 10-15% COARSE TO FINE GRAVEL, ANGULAN<br>TO FINE SAND, 5-7% NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-1-2-1-1-1/1-1-1-2-1/2-2-1-2-1-1 | to ROUNDED,                 | COARSE      |  |  |
| 670.7                    | 50                                                                           |                              | 14          | 5-5-6<br>(8")                  | 11                 | GP                  | SANDY GRAVEL, MEDIUM TO FINE, SUBANGULAR TO ROUNDED,<br>SAND, MOSTLY COARSE TO MEDIUM.                                                                       | . 25-30% COARS              | SE TO FINE  |  |  |
|                          |                                                                              | - s                          | 15          | 12-9-8<br>(8")                 | 17                 | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE, 1.5 1N MAXIMUM, MOSTLY<br>FRAGMENTS, ANGULAR TO ROUNDED, 15-207 COARSE TO FINE<br>NONPLASTIC FINES AND COAL, BROWN             | BROKEN SANDS<br>SAND, TRACE | STONE       |  |  |
|                          |                                                                              |                              |             |                                |                    |                     | BLOWS/INCH: 2-2-3-2-2-1/2-1-2-1-1-2/2-1-1-2-1                                                                                                                |                             | 111         |  |  |
|                          |                                                                              | 5                            | 16          | 10-14-8<br>(9")                | 22                 | GP-<br>GW           | <u>SIMILAR TO 5-15</u> , 5-7% NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-1-1-2-1-3/3-3-2-2-2-2/1-1-2-1-1-2                                                    |                             |             |  |  |
|                          |                                                                              |                              | 17          | 9-7-7                          | 14                 | GP-                 | <u>SIMILAR TO S-15, 7-10% NOMPLASTIC FINES, BROWN.</u>                                                                                                       |                             | 111         |  |  |
| 660.7                    | 60                                                                           |                              |             |                                |                    | G                   |                                                                                                                                                              |                             | .           |  |  |
|                          |                                                                              | ط<br>با<br>با                |             |                                |                    |                     |                                                                                                                                                              |                             | .           |  |  |
|                          | 45                                                                           | -                            |             |                                |                    |                     |                                                                                                                                                              |                             | 11          |  |  |
|                          | 02                                                                           | - s                          | 18          | 41-42-34<br>(14")              | 76                 | GP                  | <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED GRAY SHALE<br>SANDSTONE FRACMENTS, TRACE SLIGHTLY PLASTIC FINES, C<br>STAINING.                            | AND ORANGE E                | ROWN        |  |  |
|                          |                                                                              |                              |             |                                |                    |                     |                                                                                                                                                              |                             |             |  |  |
|                          |                                                                              | 1                            |             |                                |                    |                     | BOTTOM OF BORING AT 66.5 FT<br>Elevation 654.2 FT                                                                                                            |                             | -           |  |  |
|                          |                                                                              | -                            | [           |                                |                    |                     |                                                                                                                                                              |                             |             |  |  |
|                          |                                                                              | ]                            |             |                                |                    |                     |                                                                                                                                                              |                             |             |  |  |
|                          |                                                                              | E                            |             |                                |                    |                     |                                                                                                                                                              |                             | -           |  |  |
|                          |                                                                              | -                            |             |                                |                    |                     |                                                                                                                                                              |                             | 1           |  |  |
|                          |                                                                              | 1                            |             |                                |                    |                     |                                                                                                                                                              |                             | -           |  |  |
| NOTE: I                  | For<br>Legei                                                                 | BORING                       | SUMN<br>SEE | LARY AND SHEET I.              |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE<br>TCH No12241-654-2520 DH 91.162                                                                                      | BORING NO.                  | SHEET       |  |  |

-

## APPENDIX 2.5C

RELATIVE DENSITY PLOTS FOR VERIFICATION BORINGS

TERRA PROBE DENSIFICATION MAIN INTAKE STRUCTURE Tables for Appendix 2.5C

Table 2.5C-1

## TERRA PROBE DENSIFICATION AT MAIN INTAKE STRUCTURE VERIFICATION BORINGS

| Description                                                | Boring Number                                                     |
|------------------------------------------------------------|-------------------------------------------------------------------|
| Test Panel 1                                               | TH-1 through TH-6                                                 |
| Summary Plot - Terra Probe<br>before initial densification | 537T through 548T                                                 |
| Summary Plot - Terra Probe<br>after densification          | 549T, 550T, 553T, 554T,<br>565T, 566T, 567T, 570T<br>through 577T |
| Borings performed before<br>initial densification          | 537T through 548T                                                 |
| Borings performed after<br>initial densification           | 549T through 558T<br>562T through 564T<br>568T and 569T           |
| Test Panel 2                                               | 559T through 561T                                                 |
| Borings performed after<br>redensification offshore        | 565T through 567T<br>570T and 571T                                |
| Borings performed after<br>redensification onshore         | 572T through 577T                                                 |


































































































APPENDIX 2.5D

LABORATORY TEST DATA IN SITU SOILS

BEAVER VALLEY POWER STATION
# APPENDIX 2.5D

# TABLE OF CONTENTS

| <u>Section</u>                               | Title                                                                                                  | <u>Page</u>                          |
|----------------------------------------------|--------------------------------------------------------------------------------------------------------|--------------------------------------|
| 2.5D.1                                       | INTRODUCTION                                                                                           | 2.5D-1                               |
| 2.5D.2                                       | INDEX TESTS                                                                                            | 2.5D-1                               |
| 2.5D.2.1<br>2.5D.2.2<br>2.5D.2.3<br>2.5D.2.4 | Grain Size Analyses<br>Specific Gravity<br>Atterberg Limits and Natural Water Contents<br>Unit Weights | 2.5D-1<br>2.5D-1<br>2.5D-1<br>2.5D-2 |
| 2.5D.3                                       | CONSTANT RATE OF STRAIN CONSOLIDATION TESTS                                                            | 2.5D-2                               |
| 2.5D.3.1<br>2.5D.3.2                         | Procedure<br>Results                                                                                   | 2.5D-2<br>2.5D-2                     |
| 2.5D.4                                       | INCREMENTAL CONSOLIDATION TESTS                                                                        | 2.5D-3                               |
| 2.5D.4.1<br>2.5D.4.2                         | Procedure<br>Results                                                                                   | 2.5D-3<br>2.5D-3                     |
| 2.5D.5                                       | UNCONFINED COMPRESSION TESTS                                                                           | 2.5D-3                               |
| 2.5D.6                                       | UNCONSOLIDATED UNDRAIN TRIAXIAL<br>COMPRESSION TESTS                                                   | 2.5D-4                               |
| 2.5D.6.1<br>2.5D.6.2                         | Procedure<br>Results                                                                                   | 2.5D-4<br>2.5D-4                     |
| 2.5D.7                                       | CONSOLIDATED ISOTROPICALLY UNDRAINED<br>TRIAXIAL COMPRESSION TESTS                                     | 2.5D-4                               |
| 2.5D.7.1<br>2.5D.7.2                         | Procedure<br>Results                                                                                   | 2.5D-4<br>2.5D-4                     |
| 2.5D.8                                       | CONSOLIDATED DRAINED DIRECT SHEAR TESTS                                                                | 2.5D-5                               |
| 2.5D.8.1<br>2.5D.8.2                         | Procedure<br>Results                                                                                   | 2.5D-5<br>2.5D-5                     |
| 2.5D.9                                       | REFERENCES FOR APPENDIX 2.5D                                                                           | 2.5D-5                               |

# LIST OF TABLES

| Table<br><u>Number</u> | Title                                                                        |
|------------------------|------------------------------------------------------------------------------|
| 2.5D-1                 | Summary of Specific Gravity Determinations                                   |
| 2.5D-2                 | Atterberg Limits and Natural Water Contents                                  |
| 2.5D-3                 | Summary of In-Place Density Tests at Reactor<br>Containment Foundation Grade |
| 2.5D-4                 | Summary of Constant Rate of Strain Consolidation<br>Tests                    |
| 2.5D-5                 | Summary of Incremental Consolidation Tests                                   |
| 2.5D-6                 | Summary of Unconfined Compression Tests                                      |
| 2.5D-7                 | Summary of Unconsolidated Undrained Triaxial<br>Compression Tests            |
| 2.5D-8                 | Summary of Consolidated Undrained Triaxial<br>Compression Tests              |
| 2.5D-9                 | Summary of Consolidated Undrained Triaxial<br>Compression Tests by Others    |

LIST OF FIGURES

| Figure<br><u>Number</u>                                                                                                                                                                                                               | Title                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                                                                                                                                                                                                                                       | Gradation Curves                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
| 2.5D-1<br>2.5D-2<br>2.5D-3<br>2.5D-4<br>2.5D-5<br>2.5D-6<br>2.5D-7<br>2.5D-8<br>2.5D-9<br>2.5D-10<br>2.5D-10<br>2.5D-11<br>2.5D-12<br>2.5D-13<br>2.5D-14<br>2.5D-15<br>2.5D-16<br>2.5D-17<br>2.5D-18<br>2.5D-19<br>2.5D-20<br>2.5D-21 | Boring 802, Sample 6, 8, 9 Combined<br>Boring 802, Samples 12, 13<br>Boring 1005, Samples 19, 20<br>Boring 1008, Samples 18, 20<br>Boring 1009, Samples 18, 20<br>Boring 1012, Samples 15, 16<br>Boring 1012, Samples 11, 12<br>Boring 1014, Samples 20, 21<br>Bag Samples A-2, A-4<br>Bag Samples A-2, A-4<br>Bag Samples B-1, B-2, B-3<br>Bag Samples B-4, B-5<br>Bag Samples B-4, B-5<br>Bag Samples C-2, C-3, C-4<br>Bag Samples C-1, C-5<br>Bag Samples D-1, D-2, D-3, D-5<br>Bag Sample D-3b<br>Bag Sample D-4<br>Bag Sample E-2<br>Bag Samples E-3, E-4<br>In-Place Density Tests - Reactor Containment<br>Plasticity Chart |
|                                                                                                                                                                                                                                       | Constant Rate of Strain Consolidation Tests                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
| 2.5D-22<br>2.5D-23<br>2.5D-24<br>2.5D-25<br>2.5D-26<br>2.5D-27<br>2.5D-28<br>2.5D-29                                                                                                                                                  | Boring AB6, Sample 7D<br>Boring AB6, Sample 9F<br>Boring of6, Sample 13F<br>Boring of9, Sample 1F<br>Boring of9, Sample 2F<br>Bag 1<br>Bag 2<br>Block I, Sample 1F                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |
|                                                                                                                                                                                                                                       | Incremental Consolidation Tests                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |
| 2.5D-30<br>2.5D-31<br>2.5D-32<br>2.5D-33<br>2.5D-34<br>2.5D-35<br>2.5D-36<br>2.5D-37<br>2.5D-38                                                                                                                                       | Consolidation Test Report - Boring OF7, Sample 1F<br>Displacement-Log Time Plot - Boring OF7, Sample 1F<br>Consolidation Test Report - Boring OF7, Sample 4B<br>Displacement-Log Time Plot - Boring OF7, Sample 4B<br>Displacement-Log Time Plot - Boring OF7, Sample 4B<br>Displacement-Log Time Plot - Boring OF7, Sample 4B                                                                                                                                   |

LIST OF FIGURES (Cont)

| Figure<br><u>Number</u>                                                                                                                                                                                                                                                                                                                                      | Title                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |
|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 2.5D-39<br>2.5D-40<br>2.5D-41<br>2.5D-42<br>2.5D-43<br>2.5D-44<br>2.5D-45<br>2.5D-46<br>2.5D-46<br>2.5D-47<br>2.5D-50<br>2.5D-50<br>2.5D-51<br>2.5D-52<br>2.5D-53<br>2.5D-54<br>2.5D-55<br>2.5D-55<br>2.5D-57<br>2.5D-58<br>2.5D-59<br>2.5D-61<br>2.5D-61<br>2.5D-61<br>2.5D-62<br>2.5D-63<br>2.5D-64<br>2.5D-65<br>2.5D-64<br>2.5D-67<br>2.5D-67<br>2.5D-68 | Consolidation Test Report - Boring PL 1, Sample 1B2<br>Displacement-Log Time Plot - Boring PL1, Sample 1B2<br>Displacement-Log Time Plot - Boring PL1, Sample 1B2<br>Displacement-Log Time Plot - Boring PL1, Sample 1B2<br>Consolidation Test Report - Boring PL2, Sample 2B1<br>Displacement-Log Time Plot - Boring PL3, Sample 5F<br>Displacement-Log Time Plot - Block Sample 1<br>Displacement-Log Time Plot - Block Sample 1<br>Displacement- Time Plot - Block Sample 1<br>Displacement- Time Plot - Block Sample 1<br>Displacement- Time Plot - Block Sample 1<br>Displacement - Time Plot - Block Sample 1 |
| 2.5D-69<br>2.5D-70<br>2.5D-71<br>2.5D-72<br>2.5D-73<br>2.5D-74<br>2.5D-75                                                                                                                                                                                                                                                                                    | Unconfined Compression Tests<br>Boring AB1, Sample 13F<br>Boring AB1, Sample 15E<br>Boring AB2, Sample 15E<br>Boring AB5, Sample 12E<br>Boring AB6, Sample 7E<br>Boring AB6, Sample 9E<br>Boring AB10, Sample 10E<br>Unconsolidated Undrained Triaxial Compression Tests                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |
| 2.5D-76                                                                                                                                                                                                                                                                                                                                                      | Boring PL3, Samples 1F, 3F<br>Boring PL3, Sample 5F                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |

2.5D-77Boring PL3, Sample 5E2.5D-78Block Sample 1, Sample 1A

LIST OF FIGURES (Cont)

| Figure<br><u>Number</u>                                                                                                                                                                            |                                                                                                                                                                                                                                                                                                                                                                                                                         | <u>Title</u>                                                                                |                                                                                                                                                         |  |  |  |  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|--|
| 2.5D-79<br>2.5D-80<br>2.5D-81                                                                                                                                                                      | Block Sample 1, Sample<br>Block Sample 1, Sample<br>Triaxial Test Strength                                                                                                                                                                                                                                                                                                                                              | 1B<br>1C<br>Summary,                                                                        | Block Sample 1                                                                                                                                          |  |  |  |  |
|                                                                                                                                                                                                    | Consolidated Isotropica<br>Compression Tests                                                                                                                                                                                                                                                                                                                                                                            | ally Undra                                                                                  | ained (CIUC) Triaxial                                                                                                                                   |  |  |  |  |
| 2.5D-82<br>2.5D-83<br>2.5D-84<br>2.5D-85<br>2.5D-86<br>2.5D-87<br>2.5D-88<br>2.5D-89<br>2.5D-90<br>2.5D-91<br>2.5D-91<br>2.5D-92<br>2.5D-93<br>2.5D-94<br>2.5D-95<br>2.5D-95<br>2.5D-96<br>2.5D-97 | Boring AB1, Sample 15F<br>Triaxial Test Strength<br>Boring AB5, Sample 12D<br>Triaxial Test Strength<br>Boring AB6, Sample 7f<br>Triaxial Test Strength<br>Boring AB10, Sample 100<br>Triaxial Test Strength<br>Boring OF6, Sample 13E<br>Triaxial Test Strength<br>Boring OF7, sample 1E<br>Triaxial Test Strength<br>Boring OF9, Sample 1B<br>Boring OF9, Sample 1C<br>Boring OF9, Sample 1D<br>Boring OF9, Sample 1E | Summary,<br>Summary,<br>Summary,<br>Summary,<br>Summary,<br>Summary,                        | Boring AB1, Sample 15F<br>Boring AB5, Sample 12D<br>Boring AB6, Sample 7F<br>Boring AB10, Sample 10D<br>Boring OF6, Sample 13E<br>Boring OF7, Sample 1E |  |  |  |  |
| 2.5D-98<br>2.5D-99<br>2.5D-100<br>2.5D-101<br>2.5D-102                                                                                                                                             | Triaxial Test Strength<br>1C, 1D, 1E<br>Boring OF9, Sample 4D<br>Triaxial Test Summary,<br>Block Sample 1, Sample<br>Effective Stress Path,<br>(Drafted)                                                                                                                                                                                                                                                                | Summary, Boring OF9, Samples 1B,<br>Boring OF9, Sample 4D<br>1E<br>Block Sample 1,Sample 1E |                                                                                                                                                         |  |  |  |  |
|                                                                                                                                                                                                    | Dire                                                                                                                                                                                                                                                                                                                                                                                                                    | ect Shear                                                                                   | Test                                                                                                                                                    |  |  |  |  |

2.5D-103 Direct Shear Test Report Boring 906, Sample 1 2.5D-104 Direct Shear Test Summary Boring 906, Sample 1

# 2.5D.1 INTRODUCTION

The purpose of this report is to summarize and present the data obtained from tests performed to evaluate the index and engineering properties of the in situ soils at the Beaver Valley Power Station (BVPS) site.

Attempts to obtain undisturbed samples of the in situ sands and gravels were unsuccessful. Consequently, laboratory testing of undisturbed samples was only performed on the intermediate and lower terrace silts and clays. Grain size analyses on the terrace sands and gravels are discussed in Section 2.5D.2.1, and in situ density tests performed at the foundation elevation of the reactor containment are given in Section 2.5D.2.4.

2.5D.2 Index Tests

## 2.5D.2.1 Grain Size Analyses

Sixty-two grain size analyses were performed on soil samples obtained from exploratory borings for Beaver Valley Power Station - Unit 1 (BVPS-1). Grain size analyses, predominantly of the upper terrace sands and gravels, performed on samples from the 100 series borings can be found in Appendix 2F of BVPS-1 FSAR (Duquesne Light Company (DLC) 1972a). Grain size analyses performed on the 300 series borings can be found in Appendix 2H of BVPS-1 FSAR (DLC 1972b).

Fourteen additional grain size analyses were performed on samples obtained from exploratory borings for Beaver Valley Power Station - Unit 2 (BVPS-2). They are presented on Figures 2.5D-1, 2.5D-2, 2.5D-3, 2.5D-4, 2.5D-5, 2.5D-6, 2.5D-7 and 2.5D-8.

To document the soils at the foundation grade of the BVPS-2 reactor containment, in-place density tests were performed at the locations shown on Figure 2.5D-9. At each of the test locations, a bag sample was obtained and a grain size analysis was performed. The in-place density tests are described in Section 2.5D.2.4. The grain size analyses performed on the bag samples are shown on Figures 2.5D-10, 2.5D-11, 2.5D-12, 2.5D-13, 2.5D-14, 2.5D-15, 2.5D-16, 2.5D-17, 2.5D-18, 2.5D-19 and 2.5D-20.

2.5D.2.2 Specific Gravity

Four specific gravity determinations were made on samples of the in situ soils from the BVPS site in accordance with the procedures given in Appendix IV (Department of the Army 1965). The results are summarized in Table 2.5D-1.

2.5D.2.3 Atterberg Limits and Natural Water Contents

Atterberg limits and natural water contents were performed on samples of the clays and silts of the intermediate and lower terraces as an index of their variability across the site. Natural water contents were performed in accordance with ASTM D2216 (Standard Methods of Laboratory Determination of Moisture Content of Soil). Atterberg limits were determined in accordance with the methods presented in Appendix III (Department of the Army 1965). The grooving tool was as specified in ASTM D423 (Liquid Limit of Soils).

Table 2.5D-2 summarizes the Atterberg limit and water content data. Additional natural water contents can be found by referring to the various test summary tables presented herein. Data from the 300 series borings were prepared by others and are included for completeness (DLC 1972a). The plasticity chart shown on Figure 2.5D-21 indicates that the in situ silts and clays are in general slightly to moderately plastic. The points plot roughly parallel to the A line indicating a similar mineralogy across the site.

2.5D.2.4 Unit Weights

Dry unit weights were determined for each sample tested. These data are presented in the various test summary tables. At the completion of the reactor containment excavation to approximately el 679 feet, a series of the in-place density tests were performed in accordance with ASTM D1556 (Test for Density of Soil in Place by the Sand Cone Method) at the locations shown on Figure 2.5D-9. A summary of the tests is presented in Table 2.5D-3. Bag samples were recovered and grain size analyses were performed. The grain size analyses are shown on Figures 2.5D-10, 2.5D-11, 2.5D-12, 2.5D-13, 2.5D-14, 2.5D-15, 2.5D-16, 2.5D-17, 2.5D-18, 2.5D-19 and 2.5D-20.

As a result of this program, a lens of stiff silty clay was discovered beneath the northern portion of the excavation, which was later removed end replaced with compacted fill. A more detailed discussion is given in Section 2.5.4.5.

2.5D.3 Constant Rate of Strain Consolidation Tests

2.5D.3.1 Procedure

Four constant rate of strain consolidation (CRSC) tests were performed on 2.5-inch diameter specimens of in situ clay soils trimmed from undisturbed samples. Specimen preparation was in accordance with Appendix VIII (Department of the Army 1965). Tests were performed according to the procedures described in Wissa and Heilburg (1969). The maximum past consolidation pressure was calculated by the Schmertmann method (Ladd 1971).

2.5D.3.2 Results

Table 2.5D-4 summarizes the results of all CRSC tests performed. Individual test results are presented on Figures 2.5D-22, 2.5D-23, 2.5D-24, 2.5D-25, 2.5D-26, 2.5D-27, 2.5D-28 and 2.5D-29. Three tests were performed on undisturbed block and bag samples recovered from a stiff silty clay layer that was discovered beneath the northern portion of the reactor containment excavation. The clay was removed from beneath the containment foundation. The clay layer extends beneath the northern portion of the safeguards area and refueling water storage tank.

Classification tests indicate that the clay has a liquid limit of 50, a plastic limit of 23, and a natural water content of about 23 percent. A natural water content equal to the plastic limit indicates that the clay has been overconsolidated. The presence of small fissures and discoloration along the fissure surfaces suggests that the precompression may be due to desiccation. The maximum past pressure ranged between about 13 and 18 ksf. The estimated overburden pressure prior to excavation for the containment foundation was about 7.5 ksf, resulting in an overconsolidation ratio (OCR) of between 1.7 and 2.4.

Incremental consolidation tests, unconsolidated undrained (UU) triaxial compression tests, and consolidated isotropically undrained (CIU) triaxial compression tests performed on this material are described in subsequent sections.

2.5D.4 Incremental Consolidation Tests

2.5D.4.1 Procedure

Four incremental consolidation tests were performed: three on clay samples from the lower terrace and one on the stiff silty clay recovered from the reactor containment excavation. Tests were performed on 2.5-inch diameter specimens in accordance with the method given in Appendix VIII (Department of the Army 1965). The maximum past consolidation pressure was determined by the Schmertmann method (Ladd 1971).

2.5D.4.2 Results

Table 2.5D-5 summarizes the results of the tests performed. Individual test results are presented on Figures 2.5D-30, 2.5D-31, 2.5D-32, 2.5D-33, 2.5D-34, 2.5D-35, 2.5D-36, 2.5D-37, 2.5D-38, 2.5D-39, 2.5D-40, 2.5D-41, 2.5D-42, 2.5D-43, 2.5D-44, 2.5D-45, 2.5D-46, 2.5D-47, 2.5D-48, 2.5D-49, 2.5D-50, 2.5D-51, 2.5D-52, 2.5D-53, 2.5D-54, 2.5D-55, 2.5D-56, 2.5D-57, 2.5D-58, 2.5D-59, 2.5D-60, 2.5D-61, 2.5D-62, 2.5D-63, 2.5D-64, 2.5D-65, 2.5D-66, 2.5D-67 and 2.5D-68.

2.5D.5 Unconfined Compression Tests

Forty-one unconfined compression tests were performed on undisturbed and remolded specimens for the site investigations of BVPS-1 (DLC 1972a, 1972b, 1979). A summary of the test results is presented in Table 2.5D-6. Individual test plots are presented for the AB series borings on Figures 2.5D-69, 2.5D-70, 2.5D-71, 2.5D-72, 2.5D-73, 2.5D-74 and 2.5D-75. The remaining test figures can be found in the references indicated. 2.5D.6 Unconsolidated Undrained Triaxial Compression Tests

2.5D.6.1 Procedure

Six unconsolidated undrained triaxial compression tests were performed on undisturbed tube and block samples in accordance with the procedures given in Appendix X (Department of the Army 1965). No membrane correction was considered necessary.

# 2.5D.6.2 Results

Curves of deviator stress versus axial strain are presented for each test on Figures 2.5D-76, 2.5D-77, 2.5D-78, 2.5D-79, 2.5D-80 and 2.5D-81. A summary of the test data is presented in Table 2.5D-7.

The block samples were recovered from a stiff silty clay layer beneath the reactor containment excavation as described in DLC's Report on Soil Densification (DLC 1976). The undrained shear strength of this material is about 4.3 ksf. Constant rate of strain and incremental consolidation tests were performed on this material and are described in Sections 2.5D.3 and 2.5D.4, respectively. Consolidated undrained isotropically triaxial compression tests on this material are presented in Section 2.5D.7.

2.5D.7 Consolidated Isotropically Undrained Triaxial Compression Tests

## 2.5D.7.1 Procedure

Two consolidated isotropically undrained (CIU) triaxial compression tests were performed during the site investigation for BVPS-2; one from boring OF6 and one from a block sample recovered from the stiff silty clay layer beneath the reactor containment excavation. Four additional tests were performed on undisturbed samples from the AB series borings for BVPS-1 (DLC 1979). These tests were performed in accordance with Department of the Army (1965) Appendix X.

Fifteen additional CIU tests were performed by Goldberg-Zoino and Associates, Inc. (DLC 1972a) on samples from the 300 series borings.

# 2.5D.7.2 Results

Table 2.5D-8 summarizes the CIU test results for all of the CIU tests performed by Stone & Webster Engineering Corporation (SWEC). Table 2.5D-9 summarizes the test results for the 300 series borings. Individual test results for the tests given in Table 2.5D-8 are shown on Figures 2.5D-82, 2.5D-83, 2.5D-84, 2.5D-85, 2.5D-86, 2.5D-87, 2.5D-88, 2.5D-89, 2.5D-90, 2.5D-91, 2.5D-92, 2.5D-93, 2.5D-94, 2.5D-95, 2.5D-96, 2.5D-97, 2.5D-98, 2.5D-99, 2.5D-100, 2.5D-101 and 2.5D-102. Individual test results from the 300 series borings are given in the BVPS-1 FSAR (DLC 1972a).

2.5D.8 Consolidated Drained Direct Shear Tests

# 2.5D.8.1 Procedure

Five drained direct shear tests were performed on specimens of sandy clay from sample 1 of boring 906 in accordance with the procedures given in Appendix IX (Department of the Army 1965). Two different rates of displacement were used: 1.5 mm/hr and 40 mm/hr.

# 2.5D.8.2 Results

The test results are shown on Figure 2.5D-103 and are summarized on Figure 2.5D-104. The drained friction angle is approximately 29.5 degrees with a cohesion intercept of about 0.15 tsf. The rate of strain has little effect on the test results for the materials tested.

2.5D.9 References for Appendix 2.5D

Department of the Army 1965. Engineer Manual 1110-2-1906, Laboratory Soil Testing: Appendix III, Liquid and Plastic Limits; Appendix IV, Specific Gravity; Appendix V, Grain Size Analysis; Appendix VIII, Consolidation Test; Appendix IX, Drained Direct Shear Test; Appendix X, Triaxial Compression Test. Office of the Chief Engineers.

Duquesne Light Company (DLC) 1972a. Appendix 2F, Final Safety Analysis Report - Beaver Valley Power Station - Unit 1. Prepared by Stone & Webster Engineering Corporation, Boston, Mass.

Duquesne Light Company 1972b. Appendix 2H, Final Safety Analysis Report - Beaver Valley Power Station - Unit 1. Prepared by Stone & Webster Engineering Corporation, Boston, Mass.

Duquesne Light Company 1976. Report on Soil Densification Program - Beaver Valley Power Station - Unit 2. Prepared by Stone & Webster Engineering Corporation, Boston, Mass.

Duquesne Light Company 1979. Soil Analysis of Turbine Building and Northern Yard Area, Beaver Valley Power Station - Unit 1. Prepared by Stone & Webster Engineering Corporation, Boston, Mass.

Ladd, C.C. 1971. Strength Parameters and Stress-Strain Behavior of Clays. Prepared by Massachusetts Institute of Technology, Department of Civil Engineering, Cambridge, Mass.

Wissa, A. and Heilberg, S. 1969. Analysis of Turbine Building and A New One Dimensional Consolidation Test. Beaver Valley Power Station - Unit 1. Prepared by Massachusetts Institute of Technology, Department of Civil Engineering, Cambridge, Mass. Tables for Appendix 2.5D

# Table 2.5D-1

# SUMMARY OF SPECIFIC GRAVITY DETERMINATIONS

| <u>Boring</u> | Sample<br>and<br><u>Section</u> | Depth<br>(ft) | Elev<br>(ft)    | Specific<br>Gravity<br>(G) | <u>Material</u> |
|---------------|---------------------------------|---------------|-----------------|----------------------------|-----------------|
| 802           | 6, 8, 9                         | 20-31.5       | 715.0-<br>703.5 | 2.65                       | Gravelly sand   |
| PL1           | 1B2                             | 14.0          | 666.0           | 2.67                       | Sandy silt      |
| PL2           | 2B1                             | 16.5          | 664.4           | 2.67                       | Clayey silt     |
| PL3           | 5G                              | 23.0          | 659.5           | 2.74                       | Sandy clay      |

# 1 of 1

# BVPS-2 UFSAR TABLE 2.5D-2

# ATTERBERG LIMITS AND NATURAL WATER CONTENTS

| Boring<br><u>No.</u> | Sample<br>and<br><u>Section</u>                                            | Depth<br>_(ft)                                                                                     | Elevation<br>(ft)                                                                                                    | Natural<br>Water<br><u>Content</u>                           | Liquid<br>Limit<br><u>(%)</u>                                | Plastic<br>Limit<br><u>(%)</u>                               | Plasticity<br>Index<br>(%)                                   |
|----------------------|----------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------|--------------------------------------------------------------|--------------------------------------------------------------|--------------------------------------------------------------|
| AB2                  | SS17                                                                       | 37.5-39.0                                                                                          | 667.7-666.2                                                                                                          | 25.7<br>23.4                                                 | 34.4<br>24.8                                                 | 20.2<br>18.5                                                 | 14.2<br>6.3                                                  |
|                      | SS18<br>SS19                                                               | 40.0-41.5<br>42.5-44.0                                                                             | 665.2-663.7<br>662.7-661.2                                                                                           | 31.9<br>26.0                                                 | 26.2<br>23.4                                                 | 19.2<br>18.8                                                 | 7.0<br>4.6                                                   |
| AB10                 | SS16                                                                       | 35.0-36.5                                                                                          | 670.8-669.3                                                                                                          | 24.1                                                         | 28.4                                                         | 19.1                                                         | 9.3                                                          |
| C30                  | SS3                                                                        | 14.0-15.5                                                                                          | 686.0-684.5                                                                                                          | 27.1                                                         | 47.1                                                         | 24.7                                                         | 22.4                                                         |
| OF7                  | US1G                                                                       | 49.3-49.5                                                                                          | 671.7-671.5                                                                                                          | 45.8                                                         | 67.5                                                         | 37.7                                                         | 29.8                                                         |
|                      | US4B                                                                       | 60.2-60.4                                                                                          | 660.8-660.6                                                                                                          | 24.0                                                         | 30.1                                                         | 18.1                                                         | 12.0                                                         |
| OF8                  | SS11                                                                       | 55-56.5                                                                                            | 666.0-664.5                                                                                                          | 44.2<br>39.8                                                 | 58.6<br>34.3                                                 | 31.7<br>26.9                                                 | 26.9<br>7.4                                                  |
|                      | SS12                                                                       | 60-61.5                                                                                            | 661.0-659.5                                                                                                          | 42.1                                                         | 50.4                                                         | 28.0                                                         | 22.4                                                         |
| OF9                  | US1F<br>US2G<br>US4A<br>US4G                                               | 48.0-48.2<br>53.4-54.0<br>59.5-59.7<br>60.9-61.0                                                   | 673.0-672.8<br>667.6-667.0<br>661.5-661.3<br>660.1-660.0                                                             | 43.2<br>35.9<br>30.5<br>36.7                                 | 55.6<br>56.8<br>30.7<br>38.6                                 | 31.6<br>29.2<br>19.7<br>22.0                                 | 24.0<br>27.6<br>11.0<br>16.6                                 |
| PL-1<br>PL-2<br>PL-3 | ST1/1B2<br>ST2/2B1<br>ST1G<br>ST2G<br>ST3G<br>ST4G<br>ST4G<br>ST5F<br>ST6G | 14.0-14.3<br>16.5-16.6<br>7.5-7.7<br>11.4-11.5<br>13.8-14.0<br>17.3-17.5<br>23.2-23.4<br>28.2-28.4 | 666.0-665.7<br>664.4-664.3<br>675.0-674.8<br>671.1-671.0<br>668.7-668.5<br>665.2-665.0<br>659.3-659.1<br>654.3-654.1 | 46.6<br>49.3<br>26.2<br>24.6<br>26.7<br>30.1<br>30.3<br>39.0 | 55.0<br>60.5<br>44.2<br>41.2<br>45.0<br>43.7<br>38.6<br>45.3 | 30.0<br>32.8<br>21.7<br>21.7<br>21.9<br>24.8<br>23.4<br>27.8 | 25.0<br>27.7<br>22.5<br>19.5<br>23.1<br>18.9<br>15.2<br>17.5 |
| 301<br>305           | ST3<br>ST3                                                                 | 11.8-12.5<br>5.2-5.5                                                                               | 668.1-668.1<br>666.0-665.7                                                                                           | 23.1<br>47.5                                                 | 43<br>51                                                     | 24<br>39                                                     | 19<br>12                                                     |
| 306<br>308           | ST5<br>ST4                                                                 | 5.5-5.9<br>9.4-9.6<br>6.8-6.9                                                                      | 665.4-665.2<br>668.1-668.0                                                                                           | 47.9<br>73.0<br>69.0                                         | 46<br>83<br>76                                               | 38<br>44<br>42                                               | 8<br>39<br>34                                                |
| 310<br>906<br>919    | ST13<br>ST1<br>SS18                                                        | 7.2-7.5<br>25.3-25.7<br>5.0-7.0<br>35-36.5                                                         | 667.7-667.4<br>654.2-653.8<br>684.4-682.2<br>681.0-679.5                                                             | 62.8<br>25.2<br>16.7<br>21.9                                 | 57<br>28<br>24.3<br>46.5                                     | 34<br>18<br>16.8<br>23.6                                     | 23<br>10<br>7.5<br>22.9                                      |

# TABLE 2.5D-2 (Cont)

| Boring<br><u>No.</u> | Sample<br>and<br><u>Section</u> | Depth<br>_(ft)     | Elevation<br>(ft)          | Natural<br>Water<br><u>Content</u> | Liquid<br>Limit<br><u>(%)</u> | Plastic<br>Limit<br><u>(%)</u> | Plasticity<br>Index<br>(%) |
|----------------------|---------------------------------|--------------------|----------------------------|------------------------------------|-------------------------------|--------------------------------|----------------------------|
| 920                  | SS3<br>SS19                     | 24-25.5<br>56-57.5 | 712.0-710.5<br>680.0-678.5 | 21.3<br>24.8                       | 36.5<br>43.7                  | 19.9<br>22.6                   | 16.6<br>21.1               |
|                      | Bag 1*<br>Bag 2*<br>Bag 3*      |                    | 679.0<br>679.0<br>679.0    | 23.6<br>22.8<br>22.5               | 50.1<br>47.4<br>46.3          | 23.0<br>23.2<br>23.3           | 27.1<br>24.2<br>23.0       |

# NOTE:

\*Recovered from stiff silty clay lens beneath reactor containment excavation.

# Table 2.5D-3

# SUMMARY OF IN-PLACE DENSITY TESTS AT REACTOR CONTAINMENT FOUNDATION GRADE

| Test<br><u>No.</u>          | Test Location*                                                                                      | Elevation<br>(ft)                         | Dry<br>Unit<br>Weight<br>(pcf)**          | Moisture<br>Content<br>(%)      | <u>Field</u><br>Description                                                                 |
|-----------------------------|-----------------------------------------------------------------------------------------------------|-------------------------------------------|-------------------------------------------|---------------------------------|---------------------------------------------------------------------------------------------|
| A2                          | 1'E, 1'S of A2                                                                                      | 679.5                                     | 132.2                                     | 8.6                             | Layered sandy<br>clay<br>and sand                                                           |
| A3<br>A3A                   | 1.5'E, 2'S of A3<br>1'E, 1'S of A3                                                                  | 679.5<br>673.5                            | 101.8<br>127.8                            | 20.2<br>9.7                     | Clay<br>Sand and gravel<br>with clay                                                        |
| A4<br>A4A<br>A5             | 1'E, 1'S of A4<br>2.5'W, 1'N of A4<br>4'W, 0.5'N of A5                                              | 679.5<br>673.7<br>674.6                   | 109.8<br>110.3<br>103.4                   | 18.4<br>12.8<br>22.1            | Clay<br>Sand and clay<br>Clay                                                               |
| B1                          | 3'E, 1'S of B1                                                                                      | 679.5                                     | 126.1                                     | 7.1                             | Sand and gravel with clay                                                                   |
| B2                          | 2.5'E, 1'S of B2                                                                                    | 679.5                                     | 132.0                                     | 5.7                             | Sand and gravel<br>and clay                                                                 |
| B3                          | 2.5'E, 3.5'S of<br>B3                                                                               | 679.5                                     | 134.2                                     | 9.9                             | Sand and gravel with clay                                                                   |
| B4<br>B5                    | 2.5'E, 3'S of B4<br>1.5'W, 3'S of B5                                                                | 679.5<br>679.5                            | 131.4<br>113.9                            | 8.4<br>8.0                      | Clay and sand<br>Sandy clay                                                                 |
| C1                          | 2'E, 2'N of Cl                                                                                      | 679.5                                     | 129.8                                     | 5.8                             | Sand and gravel with clay                                                                   |
| C2<br>C3<br>C4<br>C5        | 1'E, 1'S of C2<br>3'W, 2'N of C3<br>2.5'E, 1'N of C4<br>2.5'W, 3'S of C5                            | 679.5<br>679.5<br>679.5<br>679.5          | 140.4<br>125.8<br>136.0<br>127.4          | 5.5<br>9.5<br>6.3<br>6.9        | Sand and gravel<br>Sand and gravel<br>Sand and gravel<br>Sand and gravel                    |
| D1<br>D2<br>D3<br>D3B<br>D4 | 2.5'E, 2'N of D1<br>2'W, 1.5'S of D2<br>4'W, 5'N of D3<br>1'E, 1.5'S of D3<br>1.5'E, 1.5'S of<br>D4 | 679.5<br>679.5<br>679.0<br>679.5<br>679.5 | 129.8<br>125.4<br>128.0<br>127.5<br>135.7 | 5.1<br>4.5<br>5.8<br>5.1<br>5.0 | Sand and gravel<br>Sand and gravel<br>Sand and gravel<br>Sand and gravel<br>Sand and gravel |
| D5<br>E2                    | 1'W, 2'N of D5<br>1.5'E, 1.5'N of<br>E2                                                             | 679.0<br>679.5                            | 128.1<br>134.6                            | 5.9<br>4.0                      | Sand and gravel<br>Sand and gravel                                                          |
| E3                          | 1.5'E, 2.5'N of<br>E3                                                                               | 679.0                                     | 129.2                                     | 4.9                             | Sand and gravel                                                                             |
| E4                          | 2'E, 1'N of E4                                                                                      | 679.0                                     | 114.6                                     | 6.7                             | Sand and gravel                                                                             |

# NOTES:

\*Location plan shown on Figure 2.5D-21. \*\*Grain size analyses at test locations given on Figures 2.5D-9 through 2.5D-19.

# SUMMARY OF CONSTANT RATE OF STRAIN (CRSC) CONSOLIDATION TESTS

\*\*\*

| Boring<br><u>No.</u>      | Sample<br><u>No.</u>                                 | Depth<br>(ft)                  | Elevation<br>(ft)                                     | Speci<br>Diameter<br>(in)                     | men<br>Height<br>_(in)_                      | Initial<br>Water<br>Content<br>(%)           | Liquid<br>Limit<br><u>(%)</u> | Plastic<br>Limit<br>(%)                | Dry<br>Unit<br>Weight<br>(pcf)                  | Initial<br>Void<br><u>Ratio</u>              | Rate of<br>Strain<br><u>(%/min)</u>                | Maximum<br>Past<br>Pressure<br>(ksf)      | Compr<br>Ra<br>**<br><u>Lab</u>                    | ression<br>itio<br>***<br>Field                | Recom-<br>pression<br><u>Ratio**</u>               | Material<br>Description                                                                                  |
|---------------------------|------------------------------------------------------|--------------------------------|-------------------------------------------------------|-----------------------------------------------|----------------------------------------------|----------------------------------------------|-------------------------------|----------------------------------------|-------------------------------------------------|----------------------------------------------|----------------------------------------------------|-------------------------------------------|----------------------------------------------------|------------------------------------------------|----------------------------------------------------|----------------------------------------------------------------------------------------------------------|
| AB6                       | US7D<br>US9F                                         | 16.0<br>21.7                   | 673.7                                                 | 2.5<br>2.5                                    | 1.00<br>1.00                                 | 24.2<br>28.1                                 | -                             | -                                      | 97.6<br>94.8                                    | 0.727<br>0.777                               | 0.080<br>0.060                                     | 3.8<br>4.0                                | 0.114<br>0.117                                     | 0.122<br>0.125                                 | 0.010<br>0.011                                     | Silty clay (CL)<br>Sandy clay<br>(CL)                                                                    |
| OF6<br>OF9<br>-<br>-<br>- | US13F<br>US1F<br>US2F<br>Bag 1<br>Bag 2<br>Block 1-F | 54.4<br>48.0<br>53.2<br>-<br>- | 666.6<br>673.0<br>667.8<br>678.0*<br>678.0*<br>679.0* | 2.5<br>2.5<br>2.5<br>2.5<br>2.5<br>2.5<br>2.5 | 0.75<br>0.75<br>0.75<br>1.00<br>0.75<br>1.00 | 35.0<br>43.2<br>44.9<br>23.6<br>22.8<br>22.0 | 29.6<br>55.6<br>50.1<br>47.4  | 18.9<br>31.6<br>-<br>23.0<br>23.2<br>- | 84.3<br>74.6<br>74.8<br>102.0<br>102.6<br>105.0 | 1.00<br>1.26<br>1.27<br>0.64<br>0.63<br>0.60 | 0.033<br>0.044<br>0.027<br>0.039<br>0.040<br>0.029 | 6.7<br>6.0<br>6.5<br>13.0<br>18.0<br>18.0 | 0.124<br>0.165<br>0.178<br>0.126<br>0.126<br>0.118 | 0.172<br>0.208<br>0.212<br>0.140<br>-<br>0.160 | 0.020<br>0.019<br>0.020<br>0.022<br>0.018<br>0.021 | Silty clay (CL)<br>Silt (MH)<br>Silty clay (CL)<br>Silty clay (CH)<br>Silty clay (CL)<br>Silty clay (CL) |

NOTES:

\*Recovered from silty clay lens at base of containment excavation.
\*\*From laboratory curve.
\*\*\*From Schmertman Construction except for test on Bag 2 which used Casegrande Construction.

# SUMMARY OF INCREMENTAL CONSOLIDATION TESTS

|                 | m-                                      |
|-----------------|-----------------------------------------|
| Boring<br>No.   | on Material<br><u> o** Description</u>  |
| OF7             | Sandy silt (MH)<br>Sandy clay<br>(CL)   |
| PL2             | Sandy silt (MH)<br>Clayey silt<br>(MH)  |
| PL3<br>-        | Sandy clay<br>(CL)<br>Silty clay (CL)   |
| PL2<br>PL3<br>- | Sar<br>Cla<br>(MH<br>Sar<br>(CL<br>Silt |

# NOTES:

\*Recovered from silty clay lens at base of containment excavation. \*\*From laboratory curve. \*\*\*From Schmertman Construction.

# SUMMARY OF UNCONFINED COMPRESSION TESTS

| Boring<br><u>No.</u> | Sample<br><u>No.</u> | Test<br><u>No.</u> | Depth<br>_(ft)_ | Elevation<br>(ft) | <u>Specir</u><br>Diameter<br><u>(in)</u> | <u>nen</u><br>Height<br>_(in)_ | Average<br>Water<br>Content<br>_(%) | Rate of<br>Strain<br><u>(%/min)</u> | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(ksf) | Axial Strain<br>at<br>(σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br><u>(%)</u> |
|----------------------|----------------------|--------------------|-----------------|-------------------|------------------------------------------|--------------------------------|-------------------------------------|-------------------------------------|------------------------------------------------------------|---------------------------------------------------------------------------------------|
| 109                  | ST3                  | 109-3N             | 7               | 683.6             | 1.4                                      | 2.8                            | 19.8                                | 7.1                                 | 2.0                                                        | 5.0                                                                                   |
|                      | ST6                  | 109-6N             | 13-15           | 677.6-675.6       | 2.8                                      | 5.6                            | 23.3                                | 1.4                                 | 5.0                                                        | 6.0                                                                                   |
|                      |                      | 109-6R             |                 |                   | 2.8                                      | 5.6                            | 22.6                                | 1.6                                 | 2.8                                                        | 11.5                                                                                  |
|                      | ST7                  | 109-7N             | 18-20           | 672.6-670.6       | 2.8                                      | 5.6                            | 26.4                                | 1.68                                | 1.1                                                        | 16.0                                                                                  |
|                      |                      | 109-7R             |                 |                   | 1.4                                      | 2.8                            | 24.8                                | 2.99                                | 1.8                                                        | 72.8                                                                                  |
|                      | ST9                  | 109-9N             | 22-24           | 668.6-666.6       | 2.8                                      | 5.6                            | 23.5                                | 1.79                                | 0.7                                                        | 8.0                                                                                   |
| 110                  | ST2                  | 110-2N             | 7-9             | 682.1-680.1       | 2.8                                      | 5.6                            | 19.1                                | 1.96                                | 3.9                                                        | 3.2                                                                                   |
|                      | STA                  | 110-2R<br>110 GN   | 15 17           | 674 1 672 1       | 2.0                                      | 5.0                            | -                                   | 1.40                                | 2.0                                                        | 4.0                                                                                   |
|                      | 510                  | 110-0N<br>110 6P   | 10-17           | 074.1-072.1       | 2.0                                      | 5.0                            | 21.7                                | 2.00                                | 4.5                                                        | 16.0                                                                                  |
|                      | <u>от</u> 2          | 110-0IX<br>110 QNI | 21.22           | 668 1 666 1       | 2.0                                      | 5.6                            | 22. <del>4</del><br>23.8            | 2.09                                | J.U<br>1 3                                                 | 5.0                                                                                   |
|                      | 515                  | 110-9N<br>110-9P   | 21-22           | 000.1-000.1       | 2.0                                      | 5.6                            | 25.0                                | 2.00                                | 0.5                                                        | 14.0                                                                                  |
|                      | ST11                 | 110-11N            | 28-28.5         | 661.1-660.6       | 2.8                                      | 5.6                            | 23.0                                | 1.83                                | 1.3                                                        | 4.5                                                                                   |
| 111                  | ST1                  | 111-1N             | 7               | 683.0             | 2.8                                      | 5.6                            | 23.5                                | 1.73                                | 1.5                                                        | 13.0                                                                                  |
|                      |                      | 111-1R             |                 |                   | 2.8                                      | 5.6                            | 22.2                                | 1.68                                | 1.7                                                        | 16.0                                                                                  |
|                      | ST2                  | 111-2N             | 14              | 676.0             | 2.8                                      | 5.6                            | 24.4                                | 1.77                                | 3.5                                                        | 12.0                                                                                  |
|                      |                      | 111-2R             |                 |                   | 2.8                                      | 5.6                            | 23.6                                | 1.77                                | 2.3                                                        | 17.5                                                                                  |
|                      | ST2A                 | 111-2AN            | 15              | 675.0             | 1.4                                      | 2.8                            | 23.0                                | 8.21                                | 5.1                                                        | 15.0                                                                                  |
|                      |                      | 111-2AR            |                 |                   | 1.4                                      | 2.8                            | 22.5                                | 8.57                                | 4.0                                                        | 21.0                                                                                  |
| 117                  | ST2                  | 117-2N             | 11.5            | 680.5             | 2.8                                      | 5.6                            | 22.3                                | 3.74                                | 5.4                                                        | 13.0                                                                                  |
|                      | ST5                  | 117-5N             | 17.5            | 674.6             | 2.8                                      | 5.6                            | 26.0                                | 4.05                                | 2.4                                                        | 5.0                                                                                   |
|                      | ST10                 | 117-10N            | 28              | 664.1             | 2.8                                      | 5.6                            | 33.4                                | 3.12                                | 2.7                                                        | 14.0                                                                                  |

# Rev. 0

| Material Description                               | Test*<br><u>Reference</u> |
|----------------------------------------------------|---------------------------|
| Brown silty clay<br>Brown silty clay               | 5<br>5                    |
| Brown silty clay                                   | 5                         |
| Brown silty fine sand                              | 5                         |
| Brown clayey silt                                  | 5                         |
| Brown silty clay                                   | 5                         |
| Brown sandy clayey silt                            | 5                         |
| Brown silty sand                                   | 5                         |
| Brown silty clay                                   | 5                         |
| Brown silty clay                                   | 5                         |
| Brown silty clay                                   | 5                         |
| Brown silty clay<br>Brown silty clay<br>Brown clay | 5<br>5<br>5               |

# TABLE 2.5D-6 (Cont)

| Boring<br>No. | Sample<br><u>No.</u> | Test<br><u>No.</u> | Depth<br>_(ft)_ | Elevation<br>(ft)          | <u>Specir</u><br>Diameter<br><u>(in)</u> | nen<br>Height<br><u>(in)</u> | Average<br>Water<br>Content<br>(%) | Rate of<br>Strain<br><u>(%/min)</u> | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(ksf) | Axial Strain<br>at<br>(σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(%) | Material Description                    | Test*<br><u>Reference</u> |
|---------------|----------------------|--------------------|-----------------|----------------------------|------------------------------------------|------------------------------|------------------------------------|-------------------------------------|------------------------------------------------------------|--------------------------------------------------------------------------------|-----------------------------------------|---------------------------|
| 301           | ST2                  | 301-2N             | 9-11            | 691.6-689.5                | 2.8                                      | 5.6                          | 23.4                               | 2.03                                | 2.5                                                        | 6.0                                                                            | Brown silty clay,<br>some sand lenses   | 6                         |
|               |                      | 301-2R             |                 |                            | 2.8                                      | 5.6                          | 18.2                               | 2.42                                | 3.0                                                        | 13.0                                                                           |                                         |                           |
|               | ST5                  | 301-5N             | 15-17           | 685.6-683.6                | 2.8                                      | 5.6                          | 26.7                               | 2.21                                | 2.8                                                        | 14.0                                                                           | Gray, clayey organic silt               | 6                         |
|               |                      | 301-5R             |                 |                            | 2.8                                      | 5.6                          | 28.2                               | 2.35                                | 2.1                                                        | 16.0                                                                           |                                         |                           |
| 302           | ST3                  | 302-3N             | 15-17           | 688.2-686.2                | 2.8                                      | 5.6                          | 14.7                               | 1.71                                | 0.6                                                        | 7.0                                                                            | Brown silty sand                        |                           |
|               | ST5                  | 302-5N             | 19-21           | 684.2-682.2                | 2.8                                      | 5.6                          | 15.4                               | 1.79                                | 0.5                                                        | 8.0                                                                            | Brown silty sand                        | 6                         |
| 303           | ST5                  | 303-5N             | 8-10            | 688.0-686.0                | 2.8                                      | 5.6                          | 16.7                               | 1.58                                | 0.7                                                        | 8.0                                                                            | Brown silty sand                        | 6                         |
|               | ST12                 | 303-<br>12N        | 22-24           | 674.0-672.0                | 2.8                                      | 5.6                          | 26.3                               | 1.87                                | 1.79                                                       | 16.0                                                                           | Brown silty clay                        | 6                         |
| 305           | ST5                  | 305-5N             | 8-10            | 663.2-661.2                | 2.8                                      | 5.6                          | 43.2                               | 2.32                                | 0.5                                                        | 6.0                                                                            | Organic sandy silt,<br>trace clay       | 6                         |
| 306           | ST2                  | 306-2N             | 2-4             | 672.8-670.8                | 2.8                                      | 5.6                          | 62.0                               | 1.67                                | 0.4                                                        | 5.0                                                                            | Brown silty sand<br>w/organics          | 6                         |
| 307           | ST3<br>ST7           | 307-3N<br>307-7N   | 4-6<br>12-14    | 671.0-669.0<br>663.0-661.0 | 2.8<br>2.8                               | 5.6<br>5.6                   | 78.6<br>37.6                       | 1.33<br>2.17                        | 0.9<br>0.3                                                 | 2.5<br>4.8                                                                     | Brown clayey silt<br>Organic silty sand | 6<br>6                    |
| AB1           | 13F                  | _                  | 29.5            | 675.5                      | 2.8                                      | 6.5                          | 24.8                               | 0.23                                | 3.1                                                        | 9.0                                                                            | Silty clay                              | 7                         |
|               | 15E                  | -                  | 32.2            | 672.8                      | 2.8                                      | 6.5                          | 24.9                               | 0.23                                | 2.75                                                       | 5.3                                                                            | Silty clay                              | 7                         |
| AB2           | 15E                  | -                  | 33.8            | 671.4                      | 2.9                                      | 6.5                          | 27.3                               | 0.25                                | 1.0                                                        | 11.8                                                                           | Sandy clay                              | 7                         |
| AB5           | 12E                  | -                  | 24.1            | 681.3                      | 2.9                                      | 6.5                          | 23.7                               | 0.28                                | 5.2                                                        | 5.0                                                                            | Silty clay                              | 7                         |
| AB6           | 7E                   | -                  | 16.2            | 673.5                      | 2.9                                      | 6.5                          | 25.2                               | 0.31                                | 0.6                                                        | 1.5                                                                            | Silty clay                              | 7                         |
|               | 9E                   | -                  | 21.1            | 668.6                      | 2.9                                      | 6.5                          | 26.8                               | 0.31                                | 0.3                                                        | 1.8                                                                            | Sandy clay                              | 7                         |
| AB10          | 10E                  | -                  | 24.1            | 681.7                      | 2.9                                      | 6.5                          | 23.4                               | 0.28                                | 4.2                                                        | 5.2                                                                            | Silty clay                              | 7                         |

# NOTES:

\*Refers to reference (Section 2.5D.9) in which test results can be found.

# SUMMARY OF UNCONSOLIDATED UNDRAINED (UU) TRIAXIAL COMPRESSION TESTS

|                      |                                 |                       |                   | Specim           | en               |                                |                                    |                                     |                                |                                                            | Axial Strain                                 |  |
|----------------------|---------------------------------|-----------------------|-------------------|------------------|------------------|--------------------------------|------------------------------------|-------------------------------------|--------------------------------|------------------------------------------------------------|----------------------------------------------|--|
| Boring<br><u>No.</u> | Sample<br>and<br><u>Section</u> | Depth*<br><u>(ft)</u> | Elevation<br>(ft) | Diameter<br>(in) | Height<br>_(in)_ | Water<br>Content<br><u>(%)</u> | Dry Unit<br>Weight<br><u>(pcf)</u> | Rate of<br>Strain<br><u>(%/min)</u> | Confining<br>Pressure<br>(ksf) | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(ksf) | at<br>(σ <sub>1</sub> - σ <sub>3</sub> )<br> |  |
| PL3                  | 1F                              | 6.9                   | 675.6             | 2.87             | 7.15             | 24.3                           | 100.3                              | 0.28                                | 1.00                           | 4.4**                                                      | 15                                           |  |
|                      | 3F                              | 13.2                  | 669.3             | 2.90             | 7.06             | 23.5                           | 100.9                              | 0.28                                | 2.00                           | 4.4**                                                      | 15                                           |  |
|                      | 5E                              | 22.6                  | 659.9             | 2.89             | 7.08             | 21.2                           | 93.3                               | 0.28                                | 2.50                           | 3.4**                                                      | 15                                           |  |
| Block I***           | IC                              | -                     | 679               | 2.59             | 6.05             | 22.5                           | 100.8                              | 0.33                                | 28.8                           | 8.3**                                                      | 19                                           |  |
|                      | IA                              | -                     | 679               | 2.58             | 5.93             | 22.1                           | 101.3                              | 0.33                                | 14.4                           | 8.7**                                                      | 20.5                                         |  |
|                      | IB                              | -                     | 679               | 2.57             | 5.42             | 22.3                           | 101.0                              | 0.31                                | 7.2                            | 8.6**                                                      | 20                                           |  |

# NOTES:

\*Depth to top of section cut for testing. \*\*No defined peak observed in stress-strain curve. \*\*\*Recovered from stiff silty clay lens at bottom of containment foundation.

Rev. 0

Material **Description** 

Silty clay Silty clay Silty clay

Silty clay Silty clay Silty clay

# SUMMARY OF CONSOLIDATED ISOTROPICALLY - UNDRAINED ( $\overline{CIUC}$ ) TRIAXIAL COMPRESSION TESTS

|                      |                              |                                      |                                           |                                   |                                 | Specimen Properties<br>Initial After Consolidation |                                      |                                           |                                      |                                      |                                           |                                             |                                  |
|----------------------|------------------------------|--------------------------------------|-------------------------------------------|-----------------------------------|---------------------------------|----------------------------------------------------|--------------------------------------|-------------------------------------------|--------------------------------------|--------------------------------------|-------------------------------------------|---------------------------------------------|----------------------------------|
| Boring<br><u>No.</u> | and<br>Section<br><u>No.</u> | Depth<br>_(ft)                       | Elevation<br>(ft)                         | <u>Specin</u><br>Diameter<br>(in) | nen<br>Height<br>_(in)_         | Water<br>Content<br>(%)                            | Dry Unit<br>Weight<br>(pcf)          | Void<br><u>Ratio</u>                      | Water<br>Content<br>(%)              | Dry Unit<br>Weight<br>(pcf)          | Void<br><u>Ratio</u>                      | Effective<br>Confining<br>Pressure<br>(ksf) | Back<br>Pressure<br>(ksf)        |
| AB1                  | 15F                          | 32.7                                 | 672.3                                     | 2.8                               | 6.5                             | 23.2                                               | 104.2                                | 0.617                                     | 24.1                                 | 106.7                                | 0.579                                     | 3.0                                         | 6.5                              |
| AB5                  | 12D                          | 27.0                                 | 678.4                                     | 2.8                               | 6.5                             | 22.4                                               | 104.4                                | 0.614                                     | 23.1                                 | 106.1                                | 0.589                                     | 2.5                                         | 10.0                             |
| AB6                  | 7F                           | 16.0                                 | 673.7                                     | 2.8                               | 6.5                             | 28.5                                               | 95.4                                 | 0.766                                     | 28.7                                 | 96.4                                 | 0.748                                     | 1.0                                         | 6.5                              |
| AB10                 | 10D                          | 23.6                                 | 682.2                                     | 2.8                               | 6.5                             | 22.4                                               | 105.9                                | 0.592                                     | 22.8                                 | 107.6                                | 0.566                                     | 3.0                                         | 9.4                              |
| OF6                  | 13E                          | 54.0                                 | 667.0                                     | 1.4                               | 3.5                             | 33.2                                               | 83.2                                 | 1.019                                     | 31.9                                 | 88.2                                 | 0.905                                     | 7.6                                         | 6.9                              |
| OF7                  | 1E                           | 48.7                                 | 672.3                                     | 1.4                               | 3.4                             | 49.8                                               | 68.3                                 | 1.467                                     | 48.0                                 | 72.1                                 | 1.338                                     | 6.0                                         | 10.1                             |
| OF9                  | 1B<br>1C<br>1D<br>1E<br>4D   | 46.7<br>47.0<br>47.3<br>47.7<br>60.3 | 674.3<br>674.0<br>673.7<br>673.3<br>660.7 | 1.4<br>1.4<br>1.4<br>1.4<br>1.4   | 3.4<br>3.1<br>3.3<br>3.4<br>3.3 | 49.0<br>31.0<br>46.2<br>49.3<br>25.6               | 66.5<br>73.7<br>69.3<br>68.6<br>93.2 | 1.524<br>1.278<br>1.424<br>1.449<br>0.803 | 41.3<br>38.4<br>43.8<br>49.1<br>26.7 | 75.8<br>76.6<br>74.0<br>70.9<br>96.9 | 1.215<br>1.192<br>1.268<br>1.369<br>0.732 | 12.0<br>9.0<br>6.0<br>2.6<br>8.4            | 5.0<br>11.4<br>5.8<br>8.6<br>8.6 |
| Block I*             | IE                           |                                      | 679 <sup>±</sup>                          | 2.7                               | 5.8                             | 22.2                                               | 93.4                                 | 0.805                                     | 20.6                                 | 96.4                                 | 0.748                                     | 7.2                                         | 8.7                              |

# NOTE:

\*Recovered from stiff silty clay lens beneath reactor containment excavation.

Rev. 0

| (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(ksf) | Axial Strain<br>at<br>(σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(%) | Material<br>Description |
|------------------------------------------------------------|--------------------------------------------------------------------------------|-------------------------|
| 5.9                                                        | 13.5                                                                           | Silty clay              |
| 7.2                                                        | 13.0                                                                           | Silty clay              |
| 1.8                                                        | 13.7                                                                           | Silty clay              |
| 6.7                                                        | 12.4                                                                           | Silty clay              |
| 6.1                                                        | 11.1                                                                           | Clay                    |
|                                                            |                                                                                |                         |

| 6.7        | 8.0  | Sandy silt |
|------------|------|------------|
| 7.3        | 5.3  | Sandy silt |
| 0.0<br>5.4 | 4.0  | Sandy slit |
| 0.4<br>3.0 | 5.1  | Sandy silt |
| 9.9        | 7.3  | Sandy sin  |
|            |      |            |
| 8.8        | 15.1 | Silty clay |
|            |      |            |

# SUMMARY OF CONSOLIDATED UNDRAINED TRIAXIAL COMPRESSION TESTS\*,\*\* **BY OTHERS**

|                      |                        |               |                   | Initial                 |                      | Effective<br>Confining Pressure | Back Pressure           |                                                            | Axial Strain<br>at                                       |                                         |
|----------------------|------------------------|---------------|-------------------|-------------------------|----------------------|---------------------------------|-------------------------|------------------------------------------------------------|----------------------------------------------------------|-----------------------------------------|
| Boring<br><u>No.</u> | Sample<br><u>No.**</u> | Depth<br>(ft) | Elevation<br>(ft) | Water Content***<br>(%) | Unit Weight<br>(pcf) | σ <sub>c</sub><br>(psi)         | μ <sub>6</sub><br>(psi) | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(psi) | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(%) | Material Description                    |
| 305                  | 3                      | 4.0-4.5       | 667.2-666.7       | 71.4                    | 83                   | 7.6                             | 77.4                    | 12.0                                                       | 8.0                                                      | Gray to black organic silty sand        |
|                      |                        | 5.2-5.5       | 666.0-665.7       | 47.4                    | 94                   | 14.3                            | 92.6                    | 14.5                                                       | 5.5                                                      | Gray silty sand and organic clayey silt |
|                      |                        | 5.5-5.9       | 665.7-665.3       | 49.8                    | 99                   | 43.5                            | 82.4                    | 41.0                                                       | 10.0                                                     | Gray sandy clayey silt                  |
| 306                  | 5                      | 8.0-8.4       | 666.8-666.4       | 67.0                    | 88                   | 7.2                             | 66.5                    | 11.8                                                       | 6.0                                                      | Mottled brown silty clay                |
|                      |                        | 8.4-8.7       | 666.4-666.1       | 76.6                    | 88                   | 14.2                            | 72.8                    | 15.0                                                       | 5.0                                                      | Mottled brown silty clay                |
|                      |                        | 9.0-9.4       | 665.8-665.4       | 69.2                    | 94                   | 42.2                            | 63.3                    | 29.5                                                       | 5.0                                                      | Mottled brown silty clay                |
| 301                  | 3                      | 11.8-12.1     | 668.8-668.5       | 23.5                    | 126                  | 7.2                             | 61.7                    | 28.0                                                       | 14.0                                                     | Mottled brown sandy, clayey silt        |
|                      |                        | 11.8-12.1     | 668.8-668.5       | 23.6                    | 126                  | 13.6                            | 73.2                    | 48.0                                                       | 16.0                                                     | Mottled brown sandy, clayey silt        |
|                      |                        | 12.1-12.5     | 668.5-668.0       | 22.3                    | 128                  | 41.5                            | 62.6                    | 62.0                                                       | 12.0                                                     | Mottled brown sandy, clayey silt        |
| 308                  | 4                      | 6.8-7.2       | 668.1-667.7       | 74.5                    | 89                   | 7.0                             | 64.7                    | 11.8                                                       | 6.0                                                      | Mottled brown silty clay                |
|                      |                        | 7.5-7.9       | 667.4-667.0       | 77.5                    | 100                  | 14.2                            | 72.8                    | 17.9                                                       | 10.0                                                     | Mottled brown silty clay                |
|                      |                        | 7.5-7.9       | 667.4-667.0       | 79.3                    | 98                   | 43.1                            | 91.8                    | 35.0                                                       | 6.0                                                      | Mottled Brown silty clay                |
| 310                  |                        | 24.3-24.7     | 655.2-654.8       | 26.9                    | 130                  | 6.7                             | 57.0                    | 40                                                         | 720                                                      | Brown clayey sand                       |
|                      |                        | 24.7-25.0     | 654.8-654.5       | 25.1                    | 128                  | 14.2                            | 62.8                    | 43.0                                                       | 13.0                                                     | Brown clayey sand                       |
|                      |                        | 25.0-25.3     | 654.5-654.2       | 24.6                    | 128                  | 42.2                            | 61.2                    | 61                                                         | 720                                                      | Brown clayey sand                       |

# NOTES:

\*Test procedure found in Appendix 2H of BVPS-1 FSAR, Figure 2H-39. \*\*Soil test specimens approximately 1.4 in diameter and 3.5 in high. \*\*\*Atterberg limits of sections of tube sample given in Figure 2H-37 and 2H-38 of above reference.





FIGURE 2.5D-1



I GURE N ٠ σ

Ö



# GRADATION CURVES



FIGURE 2.5D-3

.



STONE & WEBSTER ENGINEERING CORPORATION

GRADATION CURVES



GRADATION

CURVE

S



CLIENT

FIGURE N 5 D - 5



GRADATION CURVES



FIGRUE 2.5D-6





П

I GURE

N

ភ ០ -

~



GRADATION CURVES



FIGURE 2.5D-8





STONE 9 WEBSTER ENGINEERING



# GRADATION CURVES





\* Based on sand and gravel fractions only

FIGURE 2.5D-1

N

STONE & WEBSTER ENGINEERING CORPORATION

GRADATION

CURVES



# GRADATION CURVES





GRADATION CURVES




# GRADATION CURVES





П

STONE & WEBSTER ENGINEERING CORPORATION

RADATION CURVE

S

ດ









IGURE 2.5D-18

П







GRADATION CURVES



T

I GURE

N

5 D -

20

60 SLIGHTLY PLASTIC MODERATELY PLASTIC HIGHLY PLASTIC 50 сн 40 PLASTICITY INDEX UTHNE 30 Ð . O. 20 CL МΗ 10 • CL - ML ML 0 10 20 30 70 0 60 80 90 100 40 50 LIQUID LIMIT LEGEND FIGURE 2.5D-21 O BAG SAMPLES-STIFF SILTY CLAY PLASTICITY CHART FROM REACTOR CONTAINMENT EXCAVATION. BEAVER VALLEY POWER STATION - UNIT 2

)

.

)

FINAL SAFETY ANALYSIS REPORT







# CONSOLIDATION TEST REPORT







# CONSOLIDATION TEST REPORT





# CONSOLIDATION TEST REPORT



| CLIENT<br>DUQUESNE LIGHT COMPAN<br>SITE<br>BEAVER VALLEY - UNIT | Y<br>2                                |                                        | J.O. NUMBE<br>12859.01<br>DATE<br>1 APR 77 | 2   | BORING NUM<br>OF7<br>SAMPLE NUR<br>1F | BER                               |
|-----------------------------------------------------------------|---------------------------------------|----------------------------------------|--------------------------------------------|-----|---------------------------------------|-----------------------------------|
| DISPLACEMENT vs. LOG TI                                         | ME PLOT                               |                                        |                                            | •   | 49.1 FT                               | •                                 |
| BEAVER VALLEY - UNIT<br>DISPLACEMENT vs. LOG TI                 |                                       |                                        |                                            |     | 1F<br><b>DEPTH</b><br><u>49.1</u> FT  | 1 10 100 1000 1000 1000 1000 1000 |
|                                                                 | · · · · · · · · · · · · · · · · · · · | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | 9                                          | 6   | œ                                     | 0.1                               |
| 10.                                                             |                                       | TICAL DIAL REA                         | P<br>P                                     | 10. | 10.                                   | -                                 |



# CONSOLIDATION TEST REPORT



FIGURE

2.50-

32

# CONSOLIDATION TEST REPORT

| UTTE  DATE  DATE  DATE    EEAVER VALLEY UNIT 2  4 APR 77  IF    DISPLACEMENT vs. LOG TIME PLOT  001  001                                                                                                                                                  | CLIENT<br>DUQUESNE LIGHT COMPANY              | J.O. NUMBER<br>12859.01 | BORING NUMBER                                                                   |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------|-------------------------|---------------------------------------------------------------------------------|
| DISPLACEMENT VE. LOG TIME FLOT                                                                                                                                                                                                                            | BEAVER VALLEY UNIT 2                          | APR 77                  | SAMPLE NUMBER                                                                   |
| - 7,7 x 10 <sup>-3</sup> am <sup>2</sup> /sec<br>= 7,7 x 10 <sup>-3</sup> am <sup>2</sup> /sec<br>= 7,7 x 10 <sup>-3</sup> am <sup>2</sup> /sec<br>1 u = 100 §<br>u = 100 §<br>u = 7,0 x 10 <sup>-3</sup><br>0 0<br>0 0<br>1 1 0 000<br>E.LAPSO THG = min | DISPLACEMENT vs. LOG TIME PLOT                | ł                       | <b>Дертн</b><br>49.1 FT.                                                        |
|                                                                                                                                                                                                                                                           | DISPLACEMENT VS. LOG TIME PLOT                |                         | DEPTH<br>49.1 FT.<br>00001<br>0001<br>0001<br>0001<br>0001<br>0001<br>0001<br>0 |
|                                                                                                                                                                                                                                                           | $c_{v} = 7.7 \times 10^{-3} \text{ cm}^{2/s}$ |                         |                                                                                 |
|                                                                                                                                                                                                                                                           |                                               | 12.                     | 12.7                                                                            |

# CONSOLIDATION TEST REPORT





FIGURE

2.5D-35

STONE & WEBSTER ENGINEERING CORPORATION

# CONSOLIDATION TEST REPORT





# CONSOLIDATION TEST REPORT



| CLIENT<br>DUQUESNE LIGE<br>SITE<br>BEAVER VALLEY | IT COMPA               | NY                     |              |             |        | J.O. NU<br>1<br>DATE<br>29 | MBER<br>2241<br>MAR 77 | BORING NUM<br>PL1<br>SAMPLE NUT<br>1B2<br>DEPTH | MBER                            |
|--------------------------------------------------|------------------------|------------------------|--------------|-------------|--------|----------------------------|------------------------|-------------------------------------------------|---------------------------------|
|                                                  | WATER CONTENT (%) 45.9 | Z VOID RATIO, Co 1.385 |              |             |        |                            | WAW 242                | 1 B2<br>DEPTH<br>14.0                           | APPLIED STRESS - tons per sq ft |
|                                                  |                        | το                     | juəo<br>tuəo | Preq - VIAF | Na JAI | XA                         | SAND'                  | 57<br>50<br>50                                  | 0.1                             |

# CONSOLIDATION TEST REPORT



| DUQUESNE LIGHT COMPA | NY<br>2                                               | J.O. NUMBER<br>12241<br>DATE<br>8 MAR 77 | BORING NUMBER<br>PL1<br>SAMPLE NUMBER<br>1B2 |
|----------------------|-------------------------------------------------------|------------------------------------------|----------------------------------------------|
| DISPLACEMENT vs. LOG | TIME PLOT                                             | •                                        | <b>рертн</b><br>14.0 FT                      |
|                      |                                                       |                                          | 10000                                        |
| 1 .0 TSF             | $J_{\rm V} = 9.1 \text{ x } 10^{-3} \text{ cm}^2/s^6$ |                                          | 1000                                         |
|                      |                                                       |                                          | 100<br>100                                   |
|                      |                                                       |                                          |                                              |
|                      | 2.8 × 10 <sup>-3</sup>                                |                                          |                                              |
| 10.5                 |                                                       | 11.0                                     | 11.1<br>11.2<br>0.1                          |

# CONSOLIDATION TEST REPORT



# CONSOLIDATION TEST REPORT

| DUQUESNE LI                                                                                 | GHT COMPANY |           | ,         |          | J.O. NI<br>1224 | Imber<br>1 | BORING NU                             | MBER     |
|---------------------------------------------------------------------------------------------|-------------|-----------|-----------|----------|-----------------|------------|---------------------------------------|----------|
| HEAVER VALL                                                                                 | EY - UNIT 2 |           |           |          | DATE<br>8 MA    | R 77       | SAMPLE NU                             | MBER     |
| ISPLACEMENI                                                                                 | vs. LOG TIM | E PLOT    |           |          | L               | •          | <b>DEPTH</b><br>14.0 FT               |          |
|                                                                                             |             |           |           |          |                 |            |                                       |          |
|                                                                                             |             |           |           |          |                 |            |                                       |          |
|                                                                                             |             |           |           |          |                 | - 0        | · · · · · · · · · · · · · · · · · · · |          |
|                                                                                             |             |           |           | <u> </u> |                 |            |                                       | -        |
|                                                                                             |             |           | <u></u>   | <u> </u> |                 | G          |                                       | 4        |
|                                                                                             |             |           |           |          |                 | -13        | 9                                     |          |
|                                                                                             | ₿           |           |           |          |                 |            |                                       | No.      |
|                                                                                             |             |           |           |          |                 | _ 6        | -/                                    |          |
|                                                                                             |             |           |           |          |                 | _ ^        |                                       | -        |
|                                                                                             |             |           |           |          |                 |            |                                       | ]        |
|                                                                                             |             |           |           |          |                 | d          |                                       |          |
|                                                                                             |             |           |           |          |                 |            |                                       |          |
| · · · · · · · · · · · · · · · · · · ·                                                       |             |           |           | ļ        |                 |            |                                       | 8        |
|                                                                                             |             |           |           |          |                 | /          |                                       |          |
|                                                                                             |             |           |           | <u> </u> | /               |            |                                       |          |
|                                                                                             |             |           |           |          | 9               |            |                                       | - I<br>回 |
|                                                                                             |             |           |           |          | 7               |            |                                       | L L      |
|                                                                                             |             |           |           |          | د               |            |                                       |          |
|                                                                                             |             |           | /         |          |                 |            |                                       | APS      |
|                                                                                             |             |           |           |          |                 |            |                                       | 특강 턱     |
| ······                                                                                      | ~           |           |           | <u> </u> |                 |            |                                       | -        |
|                                                                                             |             |           |           |          |                 |            |                                       | -        |
|                                                                                             |             |           |           |          |                 |            |                                       | -        |
|                                                                                             |             | /         |           |          |                 |            |                                       | 4        |
| I                                                                                           |             | ŝ         |           |          |                 |            |                                       |          |
|                                                                                             |             |           |           |          |                 |            |                                       | -        |
| /                                                                                           |             | <b>x</b>  |           | +        |                 |            |                                       | 4        |
|                                                                                             |             | ••        |           |          |                 |            |                                       | 1        |
| /                                                                                           |             |           |           |          |                 |            |                                       | -        |
| •                                                                                           | <u> </u>    |           |           | <u> </u> |                 |            |                                       | 4        |
|                                                                                             |             |           |           |          |                 |            |                                       |          |
|                                                                                             |             |           | L         | <u>↓</u> |                 |            |                                       | 0.1      |
| ,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>,<br>, | 5 5 5 T     | с<br>С    | ×.        | 2.1      | и<br>С          |            |                                       | N        |
|                                                                                             | · ~         | unu - DNI | DIAL READ | TADI TRI | w               |            |                                       |          |

# CONSOLIDATION TEST REPORT



| CLIENT<br>DUQUESNE LIG<br>SITE<br>BEAVER VALLE | HT COMPANY<br>Y UNIT 2 |      | J.O. NUM<br>1222<br>DATE<br>18'MA | BER<br>1<br>.R 77     |                           | BOR<br>PI<br>SAM<br>21<br>DEP<br>10 | ING NUMB<br>L2<br>PLE NUME<br>B1<br>TH<br>6.5 FT | ER                           |
|------------------------------------------------|------------------------|------|-----------------------------------|-----------------------|---------------------------|-------------------------------------|--------------------------------------------------|------------------------------|
|                                                | ine sand, brown        | 2.67 |                                   | 6                     | 4                         | 385                                 | ]                                                |                              |
|                                                | CLAYEY SILT, 5-10%     | GS   |                                   | MATER CONTENT (%) 49. | PRY UNIT WEIGHT (PCF) 70. |                                     |                                                  | LIED STRESS - tons per sq ft |
|                                                |                        | 0    |                                   |                       |                           | 54                                  |                                                  | 0.1 1.0 APPI                 |

# CONSOLIDATION TEST REPORT



# CONSOLIDATION TEST REPORT

| DUQUESNE LIGHT COMPANY<br>TE<br>BEAVER VALLEY UNIT 2 | J.O. NUMBER<br>12241<br>DATE<br>18 MAR 77 | BORING NUMBER<br>PL2<br>SAMPLE NUMBER<br>2R1 |
|------------------------------------------------------|-------------------------------------------|----------------------------------------------|
| ISPLACEMENT vs. LOG TIME PLOT                        |                                           | DEPTH<br>16.5 FT                             |
|                                                      |                                           | 10000                                        |
|                                                      | $\infty = 3.3 \times 10^{-3}$             | 1000                                         |
|                                                      |                                           | D TIME - min                                 |
| $\Pi = \frac{1}{2}$                                  |                                           |                                              |
|                                                      |                                           |                                              |
|                                                      | 11.2                                      | 11.4                                         |

### CONSOLIDATION TEST REPORT





# CONSOLIDATION TEST REPORT


| CLIENT<br>DUQUESNE LIGHT COMPANY    | J.O. NUMBER<br>12241                                             | BORING NUMBER<br>PL3            |
|-------------------------------------|------------------------------------------------------------------|---------------------------------|
| BEAVER VALLEY UNIT 2                | 28 MAR 77                                                        | SAMPLE NUMBER<br>5F             |
|                                     |                                                                  | <b>рертн</b><br>23.2 FT         |
| SANDY CLAY, 20–30% Fine sand, brown | MATER CONTENT (%) 33.1<br>MATER CONTENT (%) 33.1<br>B9.6<br>B9.6 | APPLIED STRESS - tons per sq ft |
| Jneored - MIAATS LAIX               |                                                                  | 14                              |

## CONSOLIDATION TEST REPORT



| CLIENT<br>DUQUESNE LIGHT COMPANY<br>SITE<br>BEAVER VALLEY UNIT 2<br>DISPLACEMENT vs. LOG TIME PLOT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | J.O. NUMBER<br>12241<br>DATE<br>28 MAR 77 | BORING NUMBER<br>PL3<br>BAMPLE NUMBER<br>5F<br>DEPTH<br>23-2 FT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |
|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| 10.6 $10.7$ $10.8$ $10.8$ $10.8$ $10.8$ $10.8$ $10.9$ $0.9$ $0.6$ $10.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0.9$ $0$ |                                           | 11.3<br>11.3<br>11.3<br>11.3<br>0.1<br>1.00<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.0000<br>1.00000<br>1.0000<br>1.0000<br>1.00000<br>1.00000 |
| ICAL DIAL READING - mm                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | VERT                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |

## CONSOLIDATION TEST REPORT



| DUQUESNE LIGHT COMPAN  | Y        | J.O. NUMBER<br>12241<br>DATE<br>28 MAR 77 | BORING NUMBER<br>PL3<br>SAMPLE NUMBER |
|------------------------|----------|-------------------------------------------|---------------------------------------|
| DISPLACEMENT vs. LOG T | IME PLOT |                                           | DEPTH                                 |
|                        |          |                                           |                                       |
| ×. 8. C                |          |                                           | 0.1                                   |

## CONSOLIDATION TEST REPORT

| CLIENT<br>DUQUESNE LIGHT COMPANY | 12241     | BORING NUMBER                                 |
|----------------------------------|-----------|-----------------------------------------------|
| BEAVER VALLEY UNIT 2             | 12 MAY 76 | BLOCK SAMPLE I                                |
| INCREMENTAL CONSOLIDATION TEST   | <b>.</b>  | DEPTH                                         |
|                                  | ······    | §                                             |
|                                  |           |                                               |
| - SILTY CLAY, very stiff, brown  | A         |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  | /         |                                               |
|                                  |           | Š                                             |
|                                  |           | Led Det Let Let Let Let Let Let Let Let Let L |
|                                  |           | iips                                          |
|                                  |           | I I I I I I I I I I I I I I I I I I I         |
|                                  | /         | LESS                                          |
| 633                              |           | LLS 3                                         |
| 0   0   3                        | 7         |                                               |
|                                  |           | E C C                                         |
|                                  |           | ·····································         |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
|                                  |           |                                               |
| ∞ <del>v</del> v o               | 10        | ·0                                            |
| STRAIN - percent                 |           | -                                             |

FIGURE 2.5D-56

•









,

| TIME CURVE: 64.0 KSF STRESS INCREMENT | BORING NUMBER<br>SAMPLE NUMBER<br>BLOCK_SAMPLE<br>DEPTH | J.O. NUMBER<br>12241<br>DATE<br>4 JUN 76 | Y | IGHT COMPA                       | CLIENT<br>DUQUESNE L<br>SITE<br>BEAVER VAL |
|---------------------------------------|---------------------------------------------------------|------------------------------------------|---|----------------------------------|--------------------------------------------|
|                                       |                                                         |                                          |   | $\frac{64.0 \text{ KSI}}{2^{a}}$ |                                            |



## CONSOLIDATION TEST REPORT

| SITE     | DUQU                                 | TESNE    | LIG  | HT CO | MPA | NY |     |              |          |       |       | J.O. N<br>12<br>DATE | UMBE<br>241 | 2<br>    |     | BORING   |        |       |  |  |  |
|----------|--------------------------------------|----------|------|-------|-----|----|-----|--------------|----------|-------|-------|----------------------|-------------|----------|-----|----------|--------|-------|--|--|--|
| <u> </u> | TIME CURVE: 2.0 KSF STRESS INCREMENT |          |      |       |     |    |     |              |          |       |       | <u></u>              | 0.01        | 10       |     | DEPTH    |        |       |  |  |  |
|          | <u> </u>                             | 1        | r in | 1     |     |    |     |              |          |       |       | 1                    |             |          |     | <u>_</u> |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     | _        |        | 40    |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       | +   |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             | $\vdash$ |     |          |        | 30    |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      | -     |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      | <u> </u> |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          | 1   | 1        |        | а<br> |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       | ļ                    |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    | 2/2 |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    | {   | 3            |          |       |       |                      |             |          | 1   |          |        | 20    |  |  |  |
|          |                                      |          |      | _     |     |    |     |              |          |       |       | ļ                    |             | II       |     | _        |        |       |  |  |  |
|          |                                      |          |      |       |     |    | ,   | <            |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       | -   |    | c   | <u>,</u>     |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    | i   | د<br>ا       |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     | <b>A</b>     |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      | +           |          | +   |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        | 0     |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      | $]$         |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              | <u> </u> |       |       | <u> </u>             |             |          | +   |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      |       |     |    |     |              |          |       |       |                      |             |          |     |          |        |       |  |  |  |
|          |                                      |          |      | 1     |     |    |     |              |          | _     |       | -                    |             |          |     |          |        | ŀ     |  |  |  |
| -        | <b>_</b>                             |          |      | -     |     |    |     |              | L        | L     |       | <u> </u>             |             |          |     |          | 1      | 0     |  |  |  |
|          | <b>1.</b> 0                          |          |      |       |     |    | -   | <b>1 • 1</b> |          |       |       |                      |             |          |     |          | ۲<br>- | ¥ • • |  |  |  |
|          |                                      |          |      |       |     |    | VE  | ⊣<br>RTIC#   | L DI     | AL RE | ADING | - mn                 | n           | 1        | FIG | URE 2    | .5D-6  |       |  |  |  |

•

| BEAVER VALLEY UNIT 2 DATE SAMPLE NUMBER<br>BEAVER VALLEY UNIT 2 29 JUN 76 BLOCK SAMPLE | E I  |
|----------------------------------------------------------------------------------------|------|
| TIME CURVE: 4.0 KSF STRESS INCREMENT -                                                 |      |
|                                                                                        |      |
| 9                                                                                      |      |
|                                                                                        | ÷    |
|                                                                                        |      |
|                                                                                        | !    |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        | ( se |
|                                                                                        | nute |
|                                                                                        | (mi  |
|                                                                                        | E-   |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
| │                                                                                      |      |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        | i    |
|                                                                                        | :    |
|                                                                                        |      |
|                                                                                        |      |
|                                                                                        |      |
| VERTICAL DIAL READING _ mm EIGURE 2.50-                                                | -64  |

| SITE   | DUQUI<br>BEAVI<br>TIME | ESNE I<br>ER VAI<br>CURVI | LIGHT | COMP<br>UNIT<br>O KSF | 2<br>STR | ESS 1 | INCREM | ENT     |   |                   | DAT       | 12241<br>12241<br>29 JT | UN 76    |   | SAMP<br>BLOC | -<br>PLE NL<br>CK SAI | MBER<br>MPLE I |
|--------|------------------------|---------------------------|-------|-----------------------|----------|-------|--------|---------|---|-------------------|-----------|-------------------------|----------|---|--------------|-----------------------|----------------|
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       | 40             |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   | 1            |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         | - |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       | Q              |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       | (C)            |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           | ····· |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   | cm <sup>2</sup> / | ` <u></u> |                         |          |   |              |                       | (ສ             |
|        |                        |                           |       |                       |          |       |        |         |   | 0-2               |           |                         |          |   |              |                       | 20<br>Inute    |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          | 4 |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       | 5              |
|        |                        |                           |       |                       |          |       |        |         |   | "<br>ט<br>-       | >         |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         |          | • |              |                       |                |
|        |                        |                           |       |                       |          |       |        | <u></u> |   |                   |           |                         |          | 1 |              |                       | 10             |
|        |                        |                           |       |                       |          |       |        |         |   |                   |           |                         | /        |   |              |                       |                |
|        |                        |                           | ``    |                       |          |       |        |         |   |                   |           |                         | $\vdash$ |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         |   |                   | _         |                         |          |   |              |                       |                |
|        |                        |                           |       |                       |          |       |        |         | - |                   | $\geq$    |                         | -        |   |              |                       |                |
|        |                        |                           |       |                       | -        | •     |        |         |   |                   |           |                         |          |   |              |                       | 0              |
| [<br>[ | • •                    |                           |       | ·                     | 1        | 7•77  |        | ·       |   | 11.5              |           |                         |          |   | 4<br>11      |                       |                |



| DUQUE<br>BEAVE<br>TIME | ESNE L<br>ER VAL<br>CURVI | LEY |         | MPAN<br>T 2<br>KSF | Y<br>STR | ESS | INCI   | REMEN  | T    |                |           | 12<br>DATE<br>20  | 2241<br>9 JUN | <b>.R</b><br>76  |            |                          | BLOC | -<br>E NUM<br>K SAM                   | DER<br>PLE |
|------------------------|---------------------------|-----|---------|--------------------|----------|-----|--------|--------|------|----------------|-----------|-------------------|---------------|------------------|------------|--------------------------|------|---------------------------------------|------------|
| <u></u>                |                           |     |         | T                  | 1        |     |        |        |      | 1              | 1         |                   |               |                  | <u> </u>   |                          |      | I                                     | 1          |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       | 0          |
| Γ                      |                           |     |         |                    | $\top$   |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       | 4          |
| -                      |                           |     |         |                    | +        |     |        |        |      | <u> </u>       |           |                   |               |                  | _          |                          |      | <u> </u>                              |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               | •                |            |                          |      |                                       |            |
| -                      |                           |     |         |                    | +        | _   |        |        |      |                | 1         |                   |               |                  | $\uparrow$ |                          |      | 1                                     |            |
|                        |                           |     |         | <u> </u>           |          |     |        |        |      | - <b> </b>     |           |                   |               |                  |            |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       |            |
| F                      |                           |     |         | +                  |          |     |        |        |      | +              |           |                   |               | $\left  \right $ |            |                          |      |                                       |            |
|                        |                           |     |         | <u> </u>           | _        |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       | g          |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       |            |
| ┢                      |                           |     |         | +                  | +-       |     | ······ |        |      | - <del> </del> |           |                   |               | ┝┼╴              | +          |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      | /sec           |           |                   |               |                  |            |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      | ر<br>۲         |           |                   |               |                  |            |                          |      |                                       |            |
| -                      |                           |     |         |                    |          |     |        |        |      | س              |           |                   |               | ┝╌┠╌             |            |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      | 10             |           |                   |               |                  |            |                          |      |                                       |            |
|                        |                           |     |         | 1                  |          |     |        |        |      | ×              |           |                   |               | Π                |            |                          |      |                                       |            |
| -                      |                           |     |         | -                  | _        |     |        |        |      |                |           |                   |               | ļ                |            |                          |      |                                       | 8          |
|                        |                           |     |         |                    |          |     |        |        |      | 11             |           |                   |               | •                |            |                          |      |                                       |            |
| F                      |                           |     |         | 1                  |          |     |        |        |      | می<br>ا        | 1         |                   | <u> </u>      | #                |            |                          | 1    | <u>}</u>                              | 1          |
|                        |                           |     |         |                    |          |     |        |        |      | _              |           |                   | ļ             | ∥                |            |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               | V                |            |                          |      |                                       |            |
| -                      |                           |     |         | +                  |          |     |        |        |      |                | +         |                   |               |                  | -+-        |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       |            |
| -                      |                           |     |         | -                  |          |     |        |        |      |                |           |                   | -/-           |                  |            |                          |      |                                       | 10         |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   | /             |                  |            |                          |      |                                       | ł          |
| Γ                      |                           |     |         |                    |          |     |        |        |      |                |           |                   | V             |                  |            |                          |      |                                       | 1          |
|                        |                           |     | x       |                    | _        |     |        |        |      |                | <b>_</b>  |                   | 1             |                  |            | · · · · <u>· · · · ·</u> |      |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               | ł                |            |                          |      |                                       |            |
| -                      |                           |     |         | 1                  | -        |     |        |        |      |                |           |                   |               |                  | +          |                          |      |                                       | 1          |
|                        |                           |     |         | <b>_</b>           |          |     |        |        |      |                | $\models$ | $\square$         |               |                  |            |                          | L    |                                       |            |
|                        |                           |     |         |                    |          |     |        |        |      |                |           |                   |               |                  |            |                          |      |                                       |            |
| L<br>6                 |                           | -•  | <b></b> |                    | -        |     |        | L<br>≀ | L    | <br>M          | Ļ         | i<br><del>1</del> | L,            | L                |            |                          | L    | ـــــــــــــــــــــــــــــــــــــ | 0          |
| 11.                    |                           | 5   | •       |                    | 12.      |     | 0      | 2      | •    | 12             | 2         | 2                 | 0             | 2                |            | 2                        |      |                                       | 2          |
|                        |                           |     |         |                    | VE       | RTI | CAL    | DIAL   | REAL | ING -          | mm        |                   | -             | •                |            |                          |      | •                                     |            |

| DUG    | r<br>Quesni  | ELIG  | HT CC    | MPAN     | Y        |       |       |        |       |          | J.O.   | NUMB<br>2241 | ER.          |          | BORIN       | G NUM    | SER.        |
|--------|--------------|-------|----------|----------|----------|-------|-------|--------|-------|----------|--------|--------------|--------------|----------|-------------|----------|-------------|
| BEA    | VER          | VALLE | Y UNI    | IT 2     |          |       |       |        |       |          | 30     | ) JUN        | 76           |          | BLOC        | K SAM    | PLE I       |
| TIM    | IE CUI       | RVE:  | 64.0     | KSF a    | STRES    | S INC | REMEN | T      |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          | 40          |
|        |              |       |          |          |          |       |       |        |       |          |        |              | <b>├●</b> ── |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          | <u> </u> |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       | <u> </u> |          |          |       |       |        |       |          |        |              |              |          |             |          | 30          |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          | ļ        |       |       |        |       | q        | 2      |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          | :<br>! |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       | -2       |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          | ;<br>; | <u> </u>     |              |          |             |          | (es)        |
|        |              |       |          |          |          |       |       |        |       | v        | ;      |              |              |          |             |          | 20<br>Linut |
|        |              |       |          |          |          |       |       |        |       | "        | ▶      | ┝─┥          |              |          |             |          | E<br>L      |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       | ~      |       |          |        | 1            |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          | <br>     |       |       |        |       |          |        | /            |              |          |             |          | 10          |
|        |              |       |          |          |          |       |       |        |       |          |        | /            |              |          |             |          |             |
|        |              |       | ×        |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       |          |        |              |              |          |             |          |             |
|        |              |       |          |          |          |       |       |        |       | <b>*</b> |        |              |              |          |             |          |             |
|        |              |       |          | <u> </u> |          |       | -     | $\leq$ | -     |          |        |              |              |          |             |          |             |
| v<br>۲ | L            |       | 0        | 5        |          | 0     | 0     | L      |       |          | )      |              | └<br>┥       | <u> </u> | <u> </u>    | <u>م</u> |             |
| 2      | • <b>7</b> 1 | ç     | 14.      | ç        | 14       | ŝ     | 14.   |        |       | 2        |        | (<br>-       | 13.          |          | <u>1</u> .  | د<br>د   |             |
|        |              |       |          |          |          | V     | ERTIC | AL DI  | AL RE | ADINO    | } – m  | m            | F            | IGU      | <b>RE 2</b> | .5D-0    | 58          |

UNCONFINED COMPRESSION







UNCONFINED COMPRESSION



UNCONFINED COMPRESSION



UNCONFINED COMPRESSION TEST REPORT



UNCONFINED COMPRESSION



UNDRAINED COMPRESSION



### UNDRAINED COMPRESSION



### UNDRAINED COMPRESSION TEST REPORT



 $\overline{}$ 

#### UNDRAINED COMPRESSION TEST REPORT



#### UNDRAINED COMPRESSION TEST REPORT



### TRIAXIAL TEST STRENGTH SUMMARY





FIGURE 2.5D-82

### TRIAXIAL TEST STRENGTH SUMMARY





### TRIAXIAL TEST STRENGTH SUMMARY




# TRIAXIAL TEST STRENGTH SUMMARY





# TRIAXIAL TEST STRENGTH SUMMARY





# TRIAXIAL TEST STRENGTH SUMMARY



TRIAXIAL TEST REPORT



# TRIAXIAL TEST STRENGTH SUMMARY









# TRIAXIAL TEST REPORT



TRIAXIAL TEST STRENGTH SUMMARY



# TRIAXIAL TEST REPORT



# TRIAXIAL TEST STRENGTH SUMMARY



# TRIAXIAL TEST REPORT



 $SIN\phi = TAN \propto$ 

FIGURE 2.5D-102

EFFECTIVE STRESS PATH CIUTEST BLOCK SAMPLE I, SAMPLE IE

BEAVER VALLEY POWER STATION - UNIT 2 FINAL SAFETY ANALYSIS REPORT





# DIRECT SHEAR TEST REPORT



# DIRECT SHEAR TEST SUMMARY

| DUQUESNE LIGHT COMPANY                                          |                                     |                                    |                                    |        |                    | J.O. NUMBER<br>11700 |                           | EXPLORATION TYPE AND NUMBER<br>BORING 906 |                |          |       |       |               |       |       |  |
|-----------------------------------------------------------------|-------------------------------------|------------------------------------|------------------------------------|--------|--------------------|----------------------|---------------------------|-------------------------------------------|----------------|----------|-------|-------|---------------|-------|-------|--|
| BEAVER VALLEY                                                   |                                     |                                    |                                    |        |                    | 21 JUN 74            |                           |                                           | SAMPLE NUMBERS |          |       |       |               |       |       |  |
| TYPE OF TEST : DRAINED SI                                       |                                     |                                    |                                    |        | PECIMEN SIZE : 2.5 |                      | IN. DIAMETER BY 1.0 IN. I |                                           |                |          | HIGH  |       |               |       |       |  |
| TYPE OF SPECIMEN : UNDISTURBED RATE OF DISPLACEMENT : SEE BELOW |                                     |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
| SAMPLE NUMBER                                                   |                                     |                                    |                                    | 101    |                    | 102                  |                           | 1 D                                       |                | 1 E      |       | 1 F   |               |       |       |  |
|                                                                 | WATER CONTENT (%)                   |                                    |                                    |        |                    | 17.9                 |                           | 18.0                                      |                | 18.2     |       | 19.7  |               | 18.8  |       |  |
| ITIA                                                            | DRY UNIT WEIGHT, $\gamma_{d}$ (PCF) |                                    |                                    |        | 105                | 5.9                  | 108.                      | 1                                         | 10             | 7.8      | 10    | 07.6  | 10            | 9.2   |       |  |
| Z                                                               | VOID RATIO, e <sub>o</sub>          |                                    |                                    |        |                    | 0.592                |                           | 0.559                                     |                | 0.565    |       | 0.567 |               | 0•543 |       |  |
| ER<br>ER                                                        | NO NC                               | NORMAL STRESS, G                   |                                    |        |                    | 2.02                 |                           | 1.09                                      |                | 2.02     |       | 0.30  |               | 0.30  |       |  |
| Υ.F.                                                            | VOID RATIO, ec                      |                                    |                                    |        | 0.350              |                      | 0.441                     |                                           | 0.430          |          | 0.559 |       | 0 <b>.540</b> |       |       |  |
| ц.                                                              | SHEAR STRESS, $	au_{f}$ (ISF)       |                                    |                                    |        | SF)                | 1.20                 |                           | 0 <b>.8</b> 48                            |                | 1.27     |       | 0•341 |               | 0.268 |       |  |
| VILUI                                                           | STR                                 | LESS RATIO, $\tau_{f}/\bar{o}_{n}$ |                                    |        |                    |                      | 0.596                     |                                           | 0.778          |          | 0.628 |       | 1.137         |       | 0.893 |  |
| T FZ                                                            | ¢'=                                 | ARC                                | TAN $	au_{f}/ar{\sigma}_{n}$ (deg) |        |                    |                      | 30.8                      |                                           | 37.9           |          | 32.1  |       | 48.7          |       | 41.8  |  |
| 4                                                               | VOID RATIO, e <sub>f</sub>          |                                    |                                    |        | 0.350              |                      | 0.43                      | 0.430                                     |                | 0.414    |       | 0.525 |               | 0•533 |       |  |
| RATE OF DISPLACEMENT                                            |                                     |                                    | 1.5mm/hr 40mm/hr                   |        | 'nr                | 40mm/hr              |                           | 40mm/hr                                   |                | 1.5mm/hr |       |       |               |       |       |  |
|                                                                 | i                                   | 2.0                                |                                    |        | <b></b>            |                      |                           |                                           |                |          |       | 1     |               | -     |       |  |
|                                                                 |                                     |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
|                                                                 |                                     |                                    |                                    |        |                    |                      |                           |                                           | <u>u</u>       |          |       |       |               |       |       |  |
|                                                                 |                                     |                                    |                                    |        | •                  |                      | то ш<br>1 б л             | mm /br                                    |                | _        | _     |       |               |       |       |  |
|                                                                 |                                     |                                    |                                    |        |                    |                      |                           | um/111                                    |                |          |       | 0     |               |       |       |  |
|                                                                 | ب<br>ب                              |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
|                                                                 | HL U                                |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
|                                                                 | Ž<br>Ш<br>Х                         |                                    |                                    |        |                    |                      |                           | •                                         |                | 1        |       |       |               |       |       |  |
|                                                                 | S_1                                 |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
|                                                                 | E A R                               |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
|                                                                 | т<br>v                              |                                    |                                    | ٥<br>۵ |                    |                      |                           |                                           |                |          |       | ļ     | _             |       |       |  |
|                                                                 |                                     | ~                                  |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |
| 0 1.0 2.0                                                       |                                     |                                    |                                    |        |                    |                      | 3.0                       |                                           |                |          |       |       |               |       |       |  |
| NORMAL STRESS, Jn                                               |                                     |                                    |                                    |        |                    |                      |                           |                                           |                |          |       |       |               |       |       |  |

APPENDIX 2.5E

TECHNICAL REPORT

STABILITY OF SLOPES

AT THE

EMERGENCY OUTFALL STRUCTURE

# STABILITY OF SLOPES AT THE EMERGENCY OUTFALL STRUCTURE

# Prepared for BEAVER VALLEY POWER STATION - UNIT 2

DUQUESNE LIGHT COMPANY PITTSBURGH, PENNSYLVANIA

> by DONALD D. HUNT

> > **JUNE 1983**

Approved by gineer

Division Head

Engineering Management

Copyright 1983 STONE & WEBSTER ENGINEERING CORPORATION BOSTON, MASSACHUSETTS 02107

# TABLE OF CONTENTS

| <u>Section</u>                             | n <u>Title</u>                                                                                                                                                                                     | <u>Page</u>                            |
|--------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------|
| 1                                          | INTRODUCTION                                                                                                                                                                                       | 1-1                                    |
| 2                                          | SUBSURFACE INVESTIGATION                                                                                                                                                                           | 2-1                                    |
| 3                                          | SUBSURFACE CONDITIONS                                                                                                                                                                              | 3-1                                    |
| 4                                          | GROUNDWATER                                                                                                                                                                                        | 4-1                                    |
| 5<br>5.1<br>5.2<br>5.3<br>5.4<br>5.5       | PROPERTIES OF SUBSURFACE MATERIALS<br>COARSE COLLUVIUM<br>FINE COLLUVIUM<br>ALLUVIAL SOILS<br>UNCONTROLLED FILL<br>COMPACTED GRANULAR FILL.                                                        | 5-1<br>5-1<br>5-2<br>5-4<br>5-4        |
| 6<br>6.1<br>6.1.1<br>6.2<br>6.2.1<br>6.2.2 | SLOPE STABILITY.<br>COLLUVIAL SLOPE.<br>Colluvial Slope - Static Case.<br>Colluvial Slope - Dynamic Case.<br>RIVERWARD SLOPE.<br>Riverward Slope - Static Case.<br>Riverward Slope - Dynamic Case. | 6-1<br>6-2<br>6-2<br>6-4<br>6-4<br>6-4 |
| 7                                          | CONCLUSIONS                                                                                                                                                                                        | 7-1                                    |
| 8                                          | REFERENCES                                                                                                                                                                                         | 8-1                                    |

APPENDIXES

A BORING LOGS AND TEST PIT LOGSB LABORATORY TESTING

# LIST OF FIGURES

| Title                                                             |
|-------------------------------------------------------------------|
| Boring Location Plan                                              |
| Subsurface Profile - Section 1-1                                  |
| Subsurface Profile - Section 1-1, Detail                          |
| Subsurface Profile - Section 2-2                                  |
| Subsurface Profile - Section 3-3                                  |
| Subsurface Profile - Section 4-4                                  |
| Colluvial Slope Stability - Section 1-1A Static Case -<br>Bishop  |
| Colluvial Slope Stability - Section 1-1A Morgenstern -<br>Price   |
| Colluvial Slope Stability - Section 1-1A Dynamic Case -<br>Bishop |
| Riverward Slope Stability - Section 4-4 Static Case               |
| Riverward Slope Stability - Section 4-4 Dynamic Case              |
|                                                                   |

ii

## INTRODUCTION

This report presents the results of an evaluation of the stability of slopes in the vicinity of the Beaver Valley Power Station - Unit 2 (BVPS-2) emergency outfall structure (EOS). The work performed to prepare this report comprised subsurface investigation, laboratory testing and slope stability analyses.

The EOS is a Quality Assurance Category I structure which is to be constructed at the far western end of the site (Figure 1-1). Its intended purpose is to provide missile protection for the emergency discharge point of the service water system and to raise the discharge point above the elevation of the probable maximum flood (el 730 feet). Service water normally flows through the circulating water system to the cooling tower. In the event that this route is blocked, the service water will be rerouted through the EOS to the Ohio River. Piping leading from the EOS to the Ohio River is not classified as Category I.

#### SUBSURFACE INVESTIGATION

Eleven borings (EOS series) were drilled in the study area during the months of May and June 1982 by Eger Drilling, Inc. of Bridgeville, Pa. under the supervision of the Stone & Webster Engineering Corporation (SWEC). The locations of these borings and a number of others performed for previous investigations are shown in Figure 1-1. The PL series of borings were performed by others in conjunction with the construction of a sludge transport pipeline for the Bruce Manfield Power Plant (GAI 1976). Logs of borings performed under the supervision of SWEC are contained in Appendix A.

Four piezometers were installed, one each in borings EOS-1, -6, -7 and -7A. Piezometer EOS-6, apparently damaged during installation, is considered inoperable. Installation records for each piezometer can be found in Appendix A.

Along the plant access road, at the base of the valley wall, eight test pits were excavated at the locations shown in Figure 1-1. The test pit logs are given in Appendix A.

## SUBSURFACE CONDITIONS

The Beaver Valley Power Station is founded on a glacial outwash terrace deposited by higher stages of the Ohio River during the Pleistocene age. The study area is at the extreme western end of the terrace where it begins to pinch out against the steep bedrock valley wall. Soil profiles developed from the subsurface investigation data are presented in Figures 3-1 through 3-5. The locations of the sections are shown in Figure 1-1. Due to the complex character of the soil deposits, it was generally not possible to develop a soil profile showing specific continuous soil types between adjacent borings. Instead, at least with the soil data obtained from the EOS borings, soils were categorized according to mode of deposition and roughly by gradation, e.g., coarse and fine colluvium.

Generally, the borings indicate that, on the steep valley walls, the bedrock surface is overlain by what is termed coarse colluvium, derived from the weathering of the parent sandstone bedrock at higher elevations. From the split spoon samples, it can be described as a sandy gravel, largely composed of weathered and decomposed, angular sandstone fragments contained within a matrix of more severely weathered sandstone. The coarse colluvium is, in turn, overlain by fine colluvium derived from the weathering of shales, claystones, and limestones. It is a heterogeneous sandy clay containing fragments of the parent rock. The colluvial soils diminished in thickness with increasing elevation on the valley wall and were found to be absent above el 850 feet (GAI 1976). At the base of the valley and extending northward to the river, there exists an interfingering of the colluvial soils with the outwash and alluvial soils deposited by the Ohio River. Figure 3-2 illustrates in greater detail the complexity of the soil conditions in the vicinity of borings EOS-1 and EOS-6. To the north of boring EOS-10 (Figure 3-1), the terrace has been eroded and portions of the original granular outwash deposits have been replaced with more recent river deposits of silt and clay. This layer is discussed in greater detail in Section 5.3.

### GROUNDWATER

Several piezometers were installed within the granular terrace soils of the main plant area and groundwater levels have been recorded on a regular basis since 1977. Groundwater levels in the terrace have been found to follow quite closely the levels of the Ohio River with little observed time lag (SWEC, 1980). Based on the soil profiles in the study area, there should be good groundwater communication between the granular soils of the terrace and the Ohio River. Consequently, groundwater levels within the terrace can be expected to closely follow the various flood stages of the Ohio River which are as follows:

Normal water level: el 665 feet Twenty-five year flood: el 690 feet Probable maximum flood: el 730 feet

Four piezometers were installed within the soils of the colluvial slope, one each in borings EOS-1, -6, -7, and -7A. The piezometer installed in boring EOS-6 was damaged and is considered inoperable. Installation details can be found in Appendix A. The piezometer in boring EOS-7, installed near the top of rock, did not indicate the presence of groundwater, nor did any of other piezometers, even after a heavy rain. The relatively impermeable surface soils and the steepness of the valley walls limit the percolation of runoff into the underlying coarse colluvium and groundwater flow through the bedrock is small; observed groundwater flow from bedrock wells averaged 2 to 4 gpm and surface bedrock seeps along joints were generally less than 1 to 2 gpm (DLC 1983).

Field descriptions indicated that the layered silt/silty sand found in borings EOS1 and EOS6 (Figure 3-2) was somewhat wetter than the surrounding, more coarse grained soil and Hendron (1975) noted what appeared to be a spring at what is approximately the location of boring EOS-1. As mentioned previously, a piezometer was installed in boring EOS-1, but it did not indicate the presence of groundwater. Eight test pits were excavated along the plant access road as shown in Figure 1-1. They should have penetrated this apparent wet zone if it existed beneath the access road; groundwater was not encountered in any of the test pits. It is felt that the condition found in borings EOS-1 and EOS-6 is localized and not extensive.

#### PROPERTIES OF SUBSURFACE MATERIALS

A laboratory testing program was conducted to evaluate the properties of the fine grained soils in the study area. Details of the testing program and the results contained in Appendix B are discussed in this section. Properties of the coarse grained soils, for which undisturbed sampling was not possible, were evaluated from correlations with standard penetration test (SPT) blow counts and sample descriptions, such as Terzaghi and Peck (1967) and Department of the Navy (1971).

#### 5.1 COARSE COLLUVIUM

The effective friction angle of the coarse colluvium will be high due to the nature of the material and, for this analysis, a value of 40 degrees was selected. The total unit weight was assumed to be 135 pcf.

#### 5.2 FINE COLLUVIUM

Shales, claystones and limestones have weathered to form the fine colluvium consisting of a heterogeneous sandy clay containing numerous rock fragments. As the material creeps downslope during the weathering process, slickensides (presheared, polished surfaces) may develop which represent inherent planes of weakness along which residual strength properties are considered to apply.

Two direct shear tests were performed on remolded split spoon samples of fine colluvium to evaluate its residual friction angle, details of which are presented in Appendix B. One test was performed on a sandy clay (Appendix B, Figure B-28) from which the coarse material was removed by washing on a No. 40 sieve and the other test was performed on a sample of silty clay (Appendix B, Figure B-25). The measured residual friction angles were 22 and 28 degrees, respectively.

Testing specimens of the fine colluvium which included the coarse grained materials would have resulted in measured residual friction angles which were too high, since there would have been a high probability of including a rock fragment across the relatively small shear plane of the direct shear device. In situ, the residual friction angle will be larger due to the presence of the rock fragments, gravel, and sand within the fine colluvium. Therefore, sufficient conservatism is included in the chosen residual friction angle of 28 degrees for the stability analysis. Also, the heavy vegetation on the slope will act to reinforce the slope and in effect increase the friction angle of the soil cover. The total unit weight of the fine colluvium was assumed to be 125 pcf.

Two additional direct shear tests on remolded split spoon samples of fine colluvium recovered from the PL series of borings were reported by Hendron (1975). They were performed on what was described in the boring logs as sandy clayey silt (liquid limit = 29 percent; plasticity index = 7.4 percent). The residual friction angles measured were 32 degrees and 33 degrees.

5.3 ALLUVIAL SOILS

Sand, Silty Sand, Sandy Gravel, Gravelly Sand

The derivation of the engineering soil properties of the terrace sands and gravels at the Beaver Valley site, fully described in Section 2.5.4.2 of the BVPS-2 Final Safety Analysis Report (DLC 1983), are summarized below:

| Total ur | nit | weight:   |        |         |  |
|----------|-----|-----------|--------|---------|--|
| above    | gr  | oundwater | table: | 125 pcf |  |
| below    | gr  | oundwater | table: | 136 pcf |  |

Effective friction angle: 30 degrees

For the purpose of simplifying the computer model, a value of 125 pcf was used for this material, since potential failure surfaces critical to the integrity of the EOS did not pass through the terrace sands and gravels below the groundwater table.

#### Silty Clay

The properties of the silty clay layer of the riverward slope were evaluated from the results of the laboratory testing program described in Appendix B.

The top surface of the clay (Figure 3-1) is at about el 690 feet. From borings EOS-4 and EOS-4A, the upper 25 feet is described as moderately plastic and medium stiff to stiff, with standard penetration test N values in the range of 8 to 10 blows per foot. As an index test, unconfined compressive strengths measured in the field with a pocket penetrometer were 1 to 2 tons per square foot (tsf). A consolidation test (Appendix B, Figure B-7) performed on an undisturbed specimen of the upper clay indicates it to be slightly overconsolidated to normally unconsolidated under the weight of the recently added uncontrolled fill.

At about el 665 feet there is a color change from brown to gray. The consistency of the lower clay is described as soft to medium stiff with standard penetration test N values of 6 to 9 blows per foot. Field unconfined compressive strengths were 0.75 to 1.0 tsf. Atterberg limits of the upper and lower clays are similar indicating consistent minerology.

The primary differences between the upper and the lower clay appear mainly to be color and consistency, probably as a direct result of a lowering of the groundwater table. This hypothesis is supported by the change in color and consistency at about el 665 feet, corresponding to the present normal water level of the Ohio River.

Two series of consolidated isotropically undrained (CIUC) triaxial compression tests were performed on undisturbed samples from boring

EOS-4A: sample UO-4 from the upper clay and sample UO-7 from the lower clay. The effective friction angle measured for the upper clay was 33.7 degrees (Appendix B, Figure B-14). The effective friction angles measured for the samples of the lower clay were 31 and 33.9 degrees (Appendix B, Figure B-18). Due to some sample disturbance upon extrusion from the sampling tube, it was necessary to trim the test specimens of the lower clay to a smaller diameter. Consequently, the measured effective friction angle may be low as a result of some sample disturbance. Closer to the river, as indicated by samples recovered from boring EOS-5, the soils are softer in consistency and more silty. The upper 12 feet (el 683 feet to el 671 feet) of boring EOS-5 is described as brown sandy silt/silt with standard penetration test N values of 2 to 3 blows per foot. Between el 671 feet and 655 feet, the soil is described as sandy clay/sandy silt with standard penetration test N values of 4 to 5 blows per foot. A series of three CIUC triaxial compression tests was performed on an undisturbed sample of the sandy clay from boring EOS-5 (Appendix B, Figure B-21). An effective friction angle of approximately 35 degrees was measured.

Based on the triaxial test results, for the analysis of the riverward slope for static or long-term conditions, the recent river deposits were considered to be a single layer with an effective friction angle of 32 degrees and a total unit weight of 125 pcf. Failure surfaces critical to the emergency outfall structure were considered to be through the deeper soils and, for this reason, the generalized soil profiles (Figures 6-5 and 6-6) do not include the upper silts found in boring EOS-5.

Undrained strength parameters of the clay were used for the dynamic analysis of the slope. The entire clay layer was assumed to be normally consolidated and was divided into several sublayers. The undrained strength of each sublayer was determined from the ratio of undrained shear strength to effective confining pressure,  $s_u/\overline{\sigma}_c$ ; an average value of 0.4 was determined from the triaxial test results. The friction angle was set equal to zero for the undrained case.

### Silt, Sandy Silt, Layered Silt/Silty Sand

As shown in the soil profile detail in Figure 3-2, in the vicinity of borings EOS-1 and EOS-6, there is an upper layer of dense or stiff, somewhat clayey, silt which is underlain by a layered silt/silty fine sand.

A series of three CIUC triaxial compression tests was performed on an undisturbed specimen of the upper material from boring EOS-1A. It was described as a slightly plastic, silty clay/clayey silt and the measured effective friction angle was 34.2 degrees (Appendix B, Figure B-8). Pore pressure response measured during the tests indicate that the soil is dilative during shear; i.e., the pore pressures increased during the early stages of the test and then decreased.

A single CIUC triaxial compression test was performed on the layered silt/silty fine sand (Figure B-12). The effective friction angle was 35.5 degrees, and this test specimen was also dilative during shear.

Soil conditions are complex at the base of the valley where it intersects the terrace. To account for the types of soils shown in Figure 3-2 in the stability analysis, the generalized soil profile discussed in Section 6 includes a trapezoidal zone of soil at the bottom of the valley with an effective friction angle of 30 degrees and a total unit weight of 120 pcf.

#### 5.4 UNCONTROLLED FILL

Uncontrolled fill placed in this area was material removed from onsite excavations. Boring EOS-4 shows the fill material to be sandy gravel and gravelly sand with standard penetration test N values of 14 to 29 blows per foot. The material from boring EOS-10 is similar but blow counts are lower, ranging between 2 and 10 blows per foot. Since the fill was placed without control and with a minimum of compactive effort, in-place densities can be expected to vary widely. Based on the granular materials encountered in the borings, the uncontrolled fill was assumed to have an effective friction angle of 30 degrees and a total unit weight of 120 pcf. In terms of the overall stability analysis of the riverward slope, the soil properties assumed for the uncontrolled fill are not critical since its effect is mostly to add weight to a given slice.

#### 5.5 COMPACTED GRANULAR FILL

Some regrading and filling of the area around the EOS will be required. The evaluation of engineering soil properties for the compacted granular fill used at the site was discussed in Section 2.5.4.5 of the BVPS-2 Final Safety Analysis Report (DLC 1983). The following properties were used in this analysis:

| Effective friction and | gle: 36 degrees |
|------------------------|-----------------|
| Total unit weight:     | 136 pcf         |

## SLOPE STABILITY

Both static and dynamic stability of the slopes in the vicinity of the EOS were examined. The steep valley wall to the south of the EOS is termed the colluvial slope; the terrace to the north of the EOS is termed the riverward slope. The stability analysis of each slope will be discussed under a separate heading.

Two methods of analysis were employed: the simplified Bishop method and the Morgenstern-Price method. The simplified Bishop method assumes a circular arc failure surface and the Morgenstern-Price method allows for an arbitrary shaped failure mass, which, in this analysis, was assumed to be a sliding wedge with straight line boundaries. The stability analyses were performed using the computer program Lease II (SWEC 1980).

The dynamic stability analyses included the effect of earthquake accelerations and the resulting inertial forces applied to the potential sliding mass in the event of the safe shutdown earthquake (SSE). Lease II uses a pseudo-static approach in which a constant force is computed as the weight of a given slice multiplied by a seismic coefficient.

This type of analysis is considered conservative since the applied inertial forces are constant and are related to the peak acceleration of the SSE. No consideration is given to the time variation of acceleration during an actual earthquake event nor is consideration given to the cyclic nature of the direction of acceleration and the resulting seismic forces.

The horizontal ground surface acceleration for the SSE at the site has been determined to be 0.125g and the vertical acceleration is taken as two thirds of the horizontal or 0.08g (DLC 1983). Therefore, the following seismic coefficients were used for both the riverward slope and colluvial slope stability analyses:

horizontal:  $\alpha_h$  = 0.125 vertical:  $\alpha_v$  = 0.08

The pseudo-static forces were applied horizontally away from the slope, i.e., downslope, and vertically down.

## 6.1 COLLUVIAL SLOPE

Profiles representative of soil conditions within the colluvial slope in the immediate vicinity of the EOS are presented in Figures 3-1 and 3-3. As discussed in Section 3, the soil conditions within the slope consist of fine colluvium overlying coarse colluvium which, in turn, overlies bedrock. At the base of the slope, there has been a complex and somewhat unpredictable interfingering of the colluvial soils with alluvial sands and silts deposited by the Ohio River.

#### BVPS-2 UFSAR

On a large scale, the soil and conditions depicted in Figures 3-1 and 3-3 are similar. Since the data obtained from the EOS series of borings are the most recent (1982) and since the engineering soil properties used were developed from soil samples obtained from these borings, more emphasis was placed on the EOS borings when developing the computer model to use in the stability analysis. Section 1-1A (Figure 6-1) depicts the generalized soil profile used in the analysis of the colluvial slope. It was developed from Figure 3-1, but simplified somewhat to facilitate the analysis.

The EOS is shown on the generalized soil profiles at its approximate distance from the toe of the slope so that conclusions can be drawn regarding the impact of any potential sliding of the slope on the structure.

Groundwater levels within the lower portion of the slope were assumed to coincide with the normal level of the Ohio River at el 665 feet for the static case and at the level of the 25-year flood, el 690 feet, for the dynamic case.

#### 6.1.1 Colluvial Slope - Static Case

Results of the analysis indicate that the overall stability of the colluvial slope under static or long-term loading conditions is acceptable. Several typical circular arc failure surfaces examined using the simplified Bishop method of analysis are shown in Figure 6-1. Failure surfaces analyzed using the Morgenstern-Price method are shown in Figure 6-2.

The analysis shows that the fine colluvium is generally stable and that movement of the entire mass of soil downslope is not likely to occur. Safety factors of potential failure surfaces which include the majority of the fine colluvium are greater than 1.5. However, since the residual friction angle of the fine colluvium chosen for the analysis is approximately equal to the angle of the slope above el 780 feet, the analysis did indicate that minor surface sloughing of the upper slope (above el 780 feet) was possible; i.e., shallow circular arcs exhibited safety factors of 1.0. Deeper circular arcs through the upper slope, such as the typical one shown in Figure 6-1, had safety factors of about 1.3 and were considered marginally stable. The distance from the EOS to the toe of this circular arc is about 160 feet and is sufficient to preclude structural damage should movement along the surface actually occur.

#### 6.1.2 Colluvial Slope - Dynamic Case

Results of the analysis, including the effect of pseudo-static forces determined for the peak accelerations of the SSE, indicate that the overall stability of the colluvial slope is acceptable. Failure surfaces examined using the Morgenstern-Price method are shown in Figure 6-2. Several typical circular arcs analyzed using the simplified Bishop method are shown in Figure 6-3.

The analysis shows that movement of the entire mass of fine colluvium downslope is not likely. Safety factors of potential failure

#### BVPS-2 UFSAR

surfaces which include the majority of the fine colluvium are 1.1 or greater. However, the minimum factor of safety for circular arcs within the fine colluvium above the mid-slope region was 0.8. A typical circle is shown in Figure 6-3. A factor of safety less than 1.0 indicates that the fine colluvium in this region may not possess sufficient residual shear strength to resist the additional forces developed by the horizontal and vertical accelerations applied to the soil by the SSE.

Neglecting the fact that the LEASE II pseudo-static analysis is conservative and assuming that some movement of the slope will actually occur in the event of the SSE, the amount of movement that may occur along this typical failure surface was estimated by an approach first suggested by Newmark (Newmark 1965) using the computer program SIDES (SWEC 1980). This analysis is based upon the following:

- 1. An earthquake acceleration-time history record may be input, normalized to any peak acceleration.
- 2. No motion will occur within the slope until the strength of the soil is exceeded, i.e., the acceleration is greater than the limiting acceleration producing a safety factor of 1.0. For the typical circle shown in Figure 6-3, the limiting horizontal and vertical accelerations were 0.045g and 0.030g, respectively.
- 3. When the soil moves, it slides as a rigid mass downslope; movement upslope is conservatively disallowed.
- 4. Displacements occurring each time the soil strength is exceeded are cumulative throughout the duration of the earthquake.

The time histories from the El Centro 1940 earthquake northsouth component and the 1952 Kern County earthquake (S69E component of the Taft record) were used. The El Centro record was chosen because it is representative of the strongest motions available from deep soil sites, whereas Taft was chosen because of its wide frequency range and strong motion characteristics. For these acceleration-time histories, the cumulative displacement of the slope was less than 1 in. This magnitude of movement is considered to be small and, given the distance of the EOS from the potential sliding mass, no damage to the EOS will occur.

Safety factors increase as the circles become larger. A typical circular arc with a safety factor of 1.0 is shown in Figure 6-3 and is considered to be marginally stable. At the limit is the large radius circular arc which includes almost all of the fine colluvium within the sliding mass, its safety factor is 1.1. Sliding wedge stability analyses were performed using the Morgenstern-Price method for failure surfaces which included most of the fine colluvium. The results shown in Figure 6-2 indicate factors of safety of 1.1 to 1.2 which are similar to those obtained using the Bishop method (Figure 6-3) for the large radius circle.

#### BVPS-2 UFSAR

In summary, the overall stability of the colluvial slope for dynamic conditions is acceptable. Failure surfaces which could affect the structure should movement actually occur along them exhibited safety factors which were adequate. Safety factors for extensive failure surfaces encompassing most of the fine colluvial material were about 1.1 for both the simplified Bishop and the Morgenstern-Price methods of analysis. The fine colluvium above about the mid-slope region appears to be unstable. However, failure surfaces within this area will involve limited amounts of material and are of a sufficient distance away from the EOS that any potential movement along the failure surfaces will not affect the structure itself.

### 6.2 RIVERWARD SLOPE

Generalized soil profiles showing the soil properties used in the stability analysis of the riverward slope are given in Figures 6-4 and 6-5. The soil profiles, based on Section 3-3 (Figure 3-5), are simplified somewhat to facilitate analysis. An amount of compacted granular fill was added to the soil profile in the immediate vicinity of the structure to account for raising the existing grade to el 730 feet. No differentiation was made between granular colluvial and alluvial soils for the purpose of assigning soil properties to the computer model. The soil model used in the dynamic analysis (Figure 6-6) is the same as that for the static case except that undrained soil properties were substituted for the clay layer.

6.2.1 Riverward Slope - Static Case

Results of the analysis (Figure 6-4) indicate that the riverward slope is stable under static loading conditions. The minimum factor of safety was found to be 1.6 for a very shallow circular arc. Potential failure surfaces through the silty clay layer were considered to be the most critical to the structure; two typical circular arcs are shown which exhibit adequate safety factors.

## 6.2.2 Riverward Slope - Dynamic Case

Results of the analysis indicate that the riverward slope is stable under dynamic loading conditions. The analysis was performed using both the simplified Bishop method and the Morgenstern-Price method. A number of typical failure surfaces is shown in Figure 6-5.

Since the most critical failure surfaces were considered to be through the clay layer, the groundwater level was taken at el 665 feet, the normal Ohio River level. Generally, the dynamic analyses, including the effect of SSE loading, is performed for groundwater levels appropriate to the 25-year flood. As stated in Section 4, the groundwater level within the granular soils of the riverward slope will closely follow the level of the Ohio River. However, given its lower permeability, it is unlikely that groundwater levels in the clay would change substantially during the relatively short duration of the 25year flood. It was therefore considered acceptable to evaluate failure surfaces through the clay layer with the groundwater level taken at el 665 feet. A deep circular arc, the majority of which passed through granular soils, was analyzed for both normal water and the 25-year flood as shown in Figure 6-5. The computed safety factors were acceptable.

## CONCLUSIONS

The results of the stability analyses indicate that there is a potential for movement of the upper portions of the colluvial slope; however, these movements will not affect the EOS, should they actually occur. The lower portion of the colluvial slope is stable for both static and dynamic loading conditions.

The riverward slope is stable for both static and dynamic loading conditions.
### SECTION 8

### REFERENCES

Department of the Navy. Design Manual, Soil Mechanics, Foundations and Earth Structures, NAVFAC DM-7, p 7-13-17. 1971.

Duquesne Light Company (DLC). Preliminary Safety Analysis Report, Beaver Valley Power Station-Unit 2. 1972.

Duquesne Light Company (DLC). Final Safety Analysis Report, Beaver Valley Power Station-Unit 2. 1983.

General Analytics, Inc. (GAI). Subsurface Investigation for Routing of Sludge Transport Pipes Around Beaver Valley Power Station. Little Blue Run Development Area. Prepared for Dravo Corporation. 1974.

Hendron, A.J. Letter report to Gibbs & Hill, New York concerning GAI (1974). Dated May 2, 1975.

Newmark, N.M. Effects of Earthquakes on Dams and Embankments. Geotechnique Vol. 15, No. 2. June 1965.

Stone & Webster Engineering Corporation (SWEC). Seismically Induced Displacements of Embankments and Slopes (SIDES). GT-009. 1979.

Stone & Webster Engineering Corporation (SWEC). Slope Stability Analysis (LEASE II), GT-108. 1980.

Stone & Webster Engineering Corporation (SWEC). Report on Settlement Monitoring Program. Beaver Valley Power Station-Unit 2. Prepared for Duquesne Light Company, Pittsburgh, Pa. 1981.

Terzaghi, K. and Peck, R., Soil Mechanics in Engineering Practice. p. 341. John Wiley and Sons, N.Y. 1967.

8-1

# APPENDIX A

# BORING LOGS AND TEST PIT LOGS

| 5<br>C             | TE BEAVER                                                        | VALLE          | N3843          | STATIC        | נאט-אכ                                                                  | <u>T 2</u> J.O. NÖ<br><u>E6223</u> GROUNO ELEV. (1) _74                               | 12241                                  |                                        | F           |  |  |  |  |
|--------------------|------------------------------------------------------------------|----------------|----------------|---------------|-------------------------------------------------------------------------|---------------------------------------------------------------------------------------|----------------------------------------|----------------------------------------|-------------|--|--|--|--|
| - IA               | CLINATION                                                        | VER            | TICAL          |               | . 86                                                                    | ARING NA INSPECTOR                                                                    | J.W. MCCC                              | Y                                      |             |  |  |  |  |
| D                  | ATE : STAR                                                       | T/FI           | INISH .        | 6-4-8         | 32                                                                      | / CONTRACTOR / DRILLE                                                                 | REGER/J                                | ARVIS                                  |             |  |  |  |  |
| 5                  | TATIC GRO                                                        | UNDW           | ATER           | DEPT          | H / D                                                                   | TERECONDED(FT) / DRILL RIG 1                                                          | TYPE                                   | E 45                                   | ·           |  |  |  |  |
| D                  | ЕРТН ТО                                                          | BEDR           | оск _          | 52            | 2.0                                                                     | (PT) TOTAL DEPTH DRILLE                                                               | <b>9</b> 52                            | .0                                     | <u>(FT)</u> |  |  |  |  |
| M                  | ETHODS :                                                         |                |                |               |                                                                         |                                                                                       |                                        | i.                                     |             |  |  |  |  |
|                    | DRILLI                                                           | NG S           | OL_            | 3-1/8         | IN RC                                                                   | LLER BIT, 3-1/4 IN I.D. CASING, WATER                                                 |                                        |                                        |             |  |  |  |  |
|                    | SAMPL                                                            | ING S          | SOIL _         | 2 IN C        | ).D. 5                                                                  | PLIT SPOON                                                                            |                                        |                                        |             |  |  |  |  |
|                    | ORILLI                                                           | NG R           | оск 🕘          | NONE          |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
| . S                | PECIAL TE                                                        | STIN           | g or in        | STR           | STRUMENTATION A IT FOROUS STORE PLEZOMETER INSTALLED WITH TIP AT EL 718 |                                                                                       |                                        |                                        |             |  |  |  |  |
| c                  | OMMENTS .                                                        | NOI            | NE             |               |                                                                         |                                                                                       |                                        | ······································ |             |  |  |  |  |
|                    |                                                                  |                |                | 1             |                                                                         |                                                                                       | ······································ |                                        |             |  |  |  |  |
| NON CON            | 1 E E 19                                                         | <u>ات ا</u>    | (n) 5<br>      | 2 8           | ر م<br>ا                                                                |                                                                                       | · .                                    |                                        |             |  |  |  |  |
| FEET               | PEP<br>AMP                                                       |                | ND/C           | 1             |                                                                         | SAMPLE DESCI                                                                          | RIPTION                                |                                        |             |  |  |  |  |
| 3                  | vī -                                                             | v <b>≭</b>     | ∎ ≺ X          | <sup>‴≯</sup> | 5                                                                       |                                                                                       |                                        |                                        |             |  |  |  |  |
|                    |                                                                  | ┺┈┈╍┻          |                |               |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
| 741.0              | 0 <b>_</b> S                                                     |                | 1-3-5          | 8             | ML ·                                                                    | TOP 6 IN: SANDY SILT, DENSE, 102 FINE GE                                              | AVEL TO 3/8                            | IN, ANGULAR, 1                         | 15-207      |  |  |  |  |
|                    | 4                                                                |                | (12")          | 1             |                                                                         | COARSE TO FINE SAND, CONTAINS ROOTS AND<br>MOIST, DARK BROWN AND BLACK.               | ORGANIC MA                             | TTER, VERY SLIC                        | GHTLY       |  |  |  |  |
|                    |                                                                  |                |                |               | CL .                                                                    | BOTTOM 6 IN: SANDY CLAY, SLIGHTLY PLAST                                               | IC, STIFF, O                           | CCASIONAL FINE                         | GRAVEL,     |  |  |  |  |
|                    | , - I s                                                          | 2              | 4-13-9         | 22            | cı.                                                                     | SIMILAR TO S-1. BOTTOM 6 IN.                                                          | G SLIGHILI :                           | MUISI, LIGHI DI                        | KUWN .      |  |  |  |  |
|                    |                                                                  |                | (12")          | <b>_</b>      |                                                                         | <u></u>                                                                               |                                        |                                        | _           |  |  |  |  |
|                    |                                                                  |                | 1.5.6          |               | _                                                                       |                                                                                       |                                        |                                        | -           |  |  |  |  |
|                    | 5 –                                                              |                | (12")          | -             |                                                                         | <u>Berren o In</u> , Berren o In, eldt Brown                                          |                                        |                                        | -           |  |  |  |  |
|                    | s                                                                | 4              | 5-6-8          | 14            | CL                                                                      | SILTY CLAY, MODERATELY PLASTIC, STIFF, 2                                              | TINE SAND                              | , SLIGHTLY MOIS                        | T, BROWN    |  |  |  |  |
|                    | 4                                                                |                | (18")          |               |                                                                         | MOTTLED WITH YELLOW AND SOME GRAY, SMALL<br>POCKETS OF SANDY CLAY WITH SOME COARSE AN | . POCKETS OF<br>ND MEDIUM SA           | LIGNITE, CONTA<br>ND, TRACE SUBAN      | NGULAR -    |  |  |  |  |
| Î                  | -                                                                | .              |                |               |                                                                         | GRAVEL TO 0.5 IN MAXIMUM.                                                             |                                        |                                        | ~~ ~        |  |  |  |  |
|                    | - °                                                              | , '            | (16")          |               | ML                                                                      | SILIT CLATER SILI, SCIONILI FLASTIC                                                   | , ALDION SI                            | IFF, NUISI, BRO                        |             |  |  |  |  |
|                    | <b>-</b> s                                                       | 6              | 6-6-5          | 11            | HL.                                                                     | SILT, NONPLASTIC TO SLIGHTLY PLASTIC, 5%                                              | VERY FINE                              | SAND, MOIST, BE                        | ROWN.       |  |  |  |  |
|                    |                                                                  |                | (18")          |               |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
|                    | s                                                                | 7              | 2-3-6<br>(13") | 9             | ML.                                                                     | SIMILAR TO S-6.                                                                       |                                        |                                        | -           |  |  |  |  |
|                    | 1                                                                |                | /              |               |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
|                    | - 5                                                              | 8              | 4-5-5<br>(17") | 10            | HL.                                                                     | SIMILAR TO S-6, CONTAINS OCCASIONAL 5mm                                               | FINE SAND                              | LENS.                                  | -           |  |  |  |  |
|                    |                                                                  |                |                |               |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
|                    | _ s                                                              | 9              | 6-7-9<br>(15") | 16            | ML                                                                      | TOP 7 IN: <u>SANDY SILT</u> , NONPLASTIC TO SLIC<br>MOIST, BROWN.                     | HTLY PLASTI                            | C, 30-40% FINE                         | SAND.       |  |  |  |  |
| [                  | . 1.                                                             |                |                |               | SM                                                                      | BOTTOM 8 IN: <u>SILTY SAND</u> , UNIFORM, FINE,                                       | 10-15% NONP                            | LASTIC FINES, E                        | BROWN.      |  |  |  |  |
| - <del></del>      |                                                                  |                |                | 1             | L                                                                       |                                                                                       |                                        |                                        |             |  |  |  |  |
| 2.                 |                                                                  | NATER          | LEVEL          | -             |                                                                         | US-SHELBY TUBE                                                                        | BORING                                 | LOG                                    |             |  |  |  |  |
| » <sup>3.</sup>    | 2"0.0. SAMPL                                                     | LIKEO<br>E SPO |                | R             |                                                                         | UU-VUIENDERU                                                                          |                                        |                                        |             |  |  |  |  |
| 4                  | DISTANCE SH<br>14015, HAMMEI                                     | 10WN<br>R FAL  | USING 30       |               |                                                                         | BEAVER W                                                                              | ALLEY PO                               | NER STATION                            | N UNIT-2    |  |  |  |  |
|                    | 14015 HAMMER FALLING 3                                           |                | AMPLE          |               |                                                                         | DUQ                                                                                   | UESNE LI                               | GHT COMPA                              | NY          |  |  |  |  |
| 2 4.               | 4. ( ) INCHES OF SAMPLE<br>Recovery.<br>5 STD Denetration Design |                |                |               |                                                                         |                                                                                       |                                        |                                        |             |  |  |  |  |
| 92 4.<br>5.        | RECOVERY.                                                        | RATIO          | N RESIS        | TANCE         |                                                                         | SHIP                                                                                  | INGPORT,                               | PENNSTLVA                              | ANIA        |  |  |  |  |
| 0 4.<br>5.<br>1 6. | RECOVERY.<br>STO. PENETI<br>BLOWS/FT.<br>UNIFIED SOIL            | RATIO          | N RESIS        | TANCE         |                                                                         | SHIP                                                                                  | NE & WEBSI                             | PENNSTLV                               |             |  |  |  |  |

|                          |              | RFANCE       | VATT     | EY POWER S                     | TATIO              | N-1)NI              | T 2. SHIPPINCPORT, PA. 12241.00                                                                                                                                                                  |
|--------------------------|--------------|--------------|----------|--------------------------------|--------------------|---------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(IE2) | TE HLADO     | SAMPLE       | SAMPLE   | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                               |
|                          | ·            |              | <u> </u> | <u> </u>                       | · · ·              | I                   |                                                                                                                                                                                                  |
|                          | 15           | - 5          | 10       | 6-5-7                          | 12                 | HL-                 | LAYERED SILT AND SILTY FINE SAND, SLIGHTLY PLASTIC FINES, CLAYEY SILT<br>CONTAINING COARSE TO FINE GRAVEL SIZED ROCK FRAGMENTS AT BOTTOM.                                                        |
|                          |              | s            | 11       | 5-5-7<br>(14")                 | 12                 | SM<br>ML            | TOP 2 IN: <u>Silty Sand</u> , Fine, Few Fine Gravel.<br>Bottom 12 IN: <u>Silt</u> , Nonplastic to very slightly plastic, moist, brown.                                                           |
| i                        |              | 5            | 12       | 3-5-5                          | 10                 | ML                  | SIMILAR TO S-11, BOTTOM 12 IN, CONTAINS FINE SAND LENSES ABOUT 2 mm THICK.                                                                                                                       |
|                          |              | <b>-</b>     | -        | 4-2-3                          | 5                  | SM                  | TOP 10 IN: <u>Silty Sand</u> , Fine, 10-15% Nonplastic Fines, Brown.                                                                                                                             |
| 721.0                    | 20 -         | ┫            |          | (12")                          |                    | ML                  | BOTTOM 2 IN: <u>Silt</u> , Slightly Plastic, Brown.                                                                                                                                              |
|                          |              | d s          | 14       | 3-1-6<br>(17")                 | 7                  | SM                  | TOP 6 IN AND BOTTOM 1 IN: <u>SILTY SAND</u> , FINE, 10-15% NONPLASTIC FINES, WET<br>ORANGE-BROWN.                                                                                                |
|                          |              | 5            | 15       | 3-4-4                          | 8                  | ML-                 | TOP 8 IN: <u>Sili</u> , slowit flasht, uki-brown.<br>Top 8 in: <u>layered sandy sili and siliy fine sand</u> , nonplastic fines, brown                                                           |
|                          |              |              | 16       | (1/)<br>3-2-3                  | 5                  | SP                  | SAND, UNIFORM, FINE, 5-10% NONPLASTIC FINES, BROWN.                                                                                                                                              |
|                          | 25           |              | ┦        |                                |                    |                     |                                                                                                                                                                                                  |
|                          |              | <b>-</b> ] ° | 17       | (15")                          | 4                  | SP                  | SINILAR TO S-10.                                                                                                                                                                                 |
|                          |              |              | 18       | (18")                          | 11                 | SM<br>GP            | TOP 12 IN: <u>SIMILAR TO S-16</u> .<br>MIDDLE 2 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED WEATHERED SHALE<br>FRAGMENTS, ANGULAR.                                                     |
|                          |              | ┥╴           | 19       | 3-4-3                          | 7                  | SP<br>SP            | BOTTOM 4 IN: <u>Sand</u> , Uniform, fine, Hoist, Brown.<br>Top 6 In: <u>Sand</u> , fine, trace silt, Brown.                                                                                      |
| 711 0                    | 10           |              | 20       | (13")                          | Ļ                  | GP-<br>GW           | BOTTOM 7 IN: <u>SANDY CRAVEL</u> , COARSE TO FINE, I IN MAXIMUM, ANGULAR TO ROUM<br>20-302 Coarse to Fine Sand, Brown.<br>Top 5 in: Stity Sand, Fine 10-152 Coarse to Fine Gravel, Rounded, 5-72 |
|                          | <b>J</b> U - |              |          | (9")                           | Ĺ                  | GP                  | NONPLASTIC FINES.<br>BOTTOM 4 IN: GRAVEL COARSE TO FINE, 1 IN MAXIMUM, ANGULAR TO ROUNDED,<br>TRACE SAND LET CEAY AND BROAD OFFANC OILY SNELLAND FEEL                                            |
|                          |              | s            | 21       | 5-3-3<br>(5")                  | 6                  | GP-<br>GV           | SANDY CRAVEL, COARSE TO FINE, 1.5 IN MAXIMUM, ANGULAR TO ROUNDED, 15-202<br>COARSE TO FINE SAND, 5-82 NONFLASTIC FINES, TRACE IRON STAINING, BROWN,                                              |
| 1                        |              | - <u>s</u>   | 22       | 4-3-5                          | 8                  | GP-                 | GRAY ORANGE.<br><u>Similar to 5–21</u> .                                                                                                                                                         |
|                          |              | <b>-</b> s   | 23       | 5-5-5                          | 10                 | GP-                 | SIMILAR TO S-21.                                                                                                                                                                                 |
|                          | 35 •         | 5            | 24       | 7-4-5                          | ,                  | GP                  | TOP 5 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS                                                                                                                  |
|                          |              | ┫            | 1        | (13")                          |                    | GP-                 | 11.5 IN NAXIMUM, 10–157 COARSE TO FINE SAND, LESS THAN 57 NONPLASTIC FINE<br>Gray.<br>Bottom & In: <u>Sandy Gravel</u> , coarse to fine, Rounded, 20–307 coarse to fin                           |
|                          |              | - 5          | 25       | 9-8-13                         | 21                 | GW                  | SAND, LESS THAN 5% NONPLASTIC FINE5, TRACE IRON STAINS, BROWN.<br><u>No recovery</u> .                                                                                                           |
| i                        |              | 5            | 26       | 8-9-8                          | 17                 | GP                  | SANDY GRAVEL. COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN,<br>INCLUE AS SOME ROUNDED CRAVEL 15-202 COARSE TO FINE SAND TRACE NOMPLAST                                              |
| 701.0                    | 40 -         | ┫            | -        |                                |                    |                     | FINES, IRON STAINS AND COAL, GRAY.                                                                                                                                                               |
|                          |              | s            | 27       | 13-19-22<br>(14")              | 41                 | GP                  | <u>SIMILAR TO 5-26</u> .<br>BLOWS/INCH: 2-2-3-2-2/3-3-3-3-4-3/4-3-4-4-3-4                                                                                                                        |
|                          | •            |              | 1        |                                |                    |                     |                                                                                                                                                                                                  |
|                          |              | <b>_</b>     | 28       | 9-11-20<br>(13")               | 31                 | SP                  | <u>BAND</u> , POORLY GRADED, MEDIUM TO FINE, 5-10% COARSE TO FINE GRAVEL,<br>BUBANGULAR TO ROUNDED, 1.5 IN SANDSTONE FRAGMENT AT TOP, TRACE NONPLASTIC<br>BINES BOOM                             |

|                          |         |                    |        |                   | ( <b>*</b> ****    | NT 1/117            | BORING NO. EOS-<br>SHEET 3 OF 3<br>SHEET 3 OF 3                                                                                                                                                                                                                                             |
|--------------------------|---------|--------------------|--------|-------------------|--------------------|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(JE2) | DEPTH T | SAMPLE<br>TYPE (7) | SAMPLE | BLOWS (3)         | SPT N<br>VALUE (S) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                          |
|                          | 45      | 5                  | 29     | 12-25-3           | 1 1 56             |                     | SANDY GRAVEL, BROKEN COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE                                                                                                                                                                                                                        |
|                          | -       | <u> </u>           |        | (16")             | 1                  |                     | FRACHERTS TO 1.5 IN MAXIMUM, ANGULAR, FEW ROUNDED, 10-15% COARSE TO FIN<br>SAND, LESS THAM 5% NONPLASTIC FINES, TRACE COAL, BROWN AND GRAY.<br>BLOWS/INCH: 2-1-2-2-3/4-5-3-5-4-4/5-5-7-5-3-6                                                                                                |
|                          |         | 5                  | 30     | 23-34-11<br>(12") | .1 145             | GP                  | SANDY CRAVEL, BROKEN COARSE TO FINE GRAVEL SIZED SANDSTONE TRAGMENTS TO<br>1.5 IN MAXIMUM, ANGULAR, 30-402 COARSE TO FINE SAND, 10-13% NONPLASTIC<br>FINES, TRACE COAL AND IRON STAINING, BROWN.<br>BLOWS/INCH: 2-2-3-4-5-7/5-5-8-6-5/13-20-18-18-17-25                                     |
| 691.0                    |         | S                  | 31     | 47-50-1<br>(18")  | 13 163             | SP<br>GP            | TOP 2 IN: <u>SAND</u> , FINE, TRACE FINE GRAVEL, 5-102 NONPLASTIC FINES, ORANGE<br>BROWN.<br>BOTTOM 16 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE AND<br>BOTTOM 16 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE AND<br>BULLE FROM NOT DO 15 STICKTY |
|                          |         | s                  | 32     | 37- <u>105</u>    | 105                |                     | PLASTIC FINES, TRACE COAL, BROWN, GRAY, ORANGE-BROWN.<br>BLOWS/INCH: 3-4-6-4-5-25/18-14-5-5-3-5/20-27-15-17-13-21                                                                                                                                                                           |
|                          |         | S                  | 33     | 50<br>57          | 50<br>12"          | a.                  | SILTY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS<br>TO 1.0 IN MAXIMUM, ANGULAR, 5-10% FINE SAND, 5-20% SLIGHTLY PLASTIC<br>FINES, TRACE COAL, IRON STAINS, ORANGE, BROWN, GRAY.                                                                                      |
|                          | -       | ļ                  |        |                   |                    |                     | REFUSAL                                                                                                                                                                                                                                                                                     |
| · .                      | -       |                    |        |                   |                    |                     | ELEVATION 688.96 FT                                                                                                                                                                                                                                                                         |
|                          | -       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | -       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | ļ       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | -       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
| ŕ                        |         |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | -       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | -       |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          |         |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          |         |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          |         |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          |         |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | 11      |                    |        |                   |                    |                     |                                                                                                                                                                                                                                                                                             |
|                          | -       |                    |        | ļ                 |                    |                     |                                                                                                                                                                                                                                                                                             |

ł

FORM G-G-O PIEZOMETER INSTALLATION REPORT PIEZOMETER NO. P-EOS-1 STONE & WEBSTER ENGINEERING CORP. J.Q. NO. SITE 12241 Beaver Valley Power Station-Unit 2 INSPECTOR \_\_ J. W. McCoy DATE \_\_\_\_\_6-7-82 ORILLER Eger/Jarvis COORDINATES N3843 E6223 741.0 ft \_ GROUND ELEV. INSTALLED IN BORING EOS-1 743.9. ft ELEY. TOP OF LEADS. RIG & CREW TIME \_\_\_\_\_ 3 hours DETAILED INSTALLATION DESCRIPTION : Hole cleaned to 52.0 ft. Filled with sand to 27 ft. Bentonite seal from 27 to 25 ft. 35" Sand placed from 25 to 22 ft 9 in. Porous stone - SOILTEST piezometer mmm 7771 with centering spider 10 ft from piezometer tip installed. 1 ft 9 in sand placed above piezometer. 3.5 ft Bentonite seal placed. Sand placed 2 ft from ground surface. Guard pipe grouted into place. Sand DESCRIPTION OF PIEZOMETER TIP AND STAND PIPE ASSEMBLY 15'6" 2 ft section of SOILTEST porous 521 Bentohite stone piezometer. Approximately 24 ft 3/4 in I.D. PVC 19' riser pipe. 20'9" 2 DESCRIPTION OF SOIL AT TIP ELEVATION : 22'9" Silt - wet with sandy silt lenses. Sand 251 251 Bentonite 27' Sand 52' NOTE : SKETCH IN ALL COMPONENTS PERTINENT TO THE INSTALLATION WITH APPLICABLE, DIMENSIONS EG : FILTER SAND, SEALS, GROUT, CASING, ETC.

#### 2.5E-31

| 51<br>C1<br>IN<br>D1<br>51<br>N1<br>S1<br>C1 | ITE<br>OORDIN<br>ICLINA<br>ATE :<br>FATIC<br>EPTH<br>ETHOU<br>DF<br>SA<br>DF<br>PECIAL<br>OMMEN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ALLIN<br>STAR<br>GROU<br>TO I<br>SS:<br>NILLIN<br>NMPLI                                                              | VALL<br>3<br>VE<br>VE<br>VF/F/<br>JNDW<br>9E DR<br>1G 3<br>NG 3<br>NG 3<br>NG 8<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10 | EY POMER<br>6.5 FT ::<br>RTICAL<br>INISH<br>NATER (<br>OCK                                          | STATION-<br>SOUTH OF<br>B<br>5/7/82<br>DEPTH / I<br>NA<br>-1/8 IN 0.<br>IELBY TUBE<br>INE<br>STRUME | NIT 2         GROUND           EARING         NA           / 6/7/82         CONTRAC           DATE         (PT) /           (PT)         TOTAL DE           D. AUGER TO ADVANCE HOLE, 3 | J.O. NO. <u>12241</u><br>ELEV. (I)741.<br>INSPECTOR <u>J.V. K</u><br>TOR / DRILLER<br>DRILL RIG TYPE<br>PTH DRILLED<br>IN O.D. SPLIT SPOON | BORING NO. 205-1<br>SHEET_LOF_2<br>EGER/JARVIS<br>CHE 45<br>22.0 FT (FT)<br>USED TO CLEAN OUT HOLE.                                                        |
|----------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEWION<br>(FEET)(N2)                        | DEPTH<br>(FEE T)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | SAMPLE<br>TYPE (7)                                                                                                   | SAMPLE<br>NUMBER                                                                                                                            | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                                 | SPT N<br>VALUE (S)<br>GROUP                                                                         | SAN                                                                                                                                                                                     | APLE DESCRIPTI                                                                                                                             | ION                                                                                                                                                        |
| 41.0                                         | 0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                      |                                                                                                                                             |                                                                                                     |                                                                                                     | NO SAMPLES T                                                                                                                                                                            |                                                                                                                                            |                                                                                                                                                            |
| 31.0                                         | 5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | US                                                                                                                   | 1 2                                                                                                                                         | (28")<br>(0")                                                                                       |                                                                                                     | <u>SILTY CLAY - CLAYEY SILT,</u><br>(SCHEWHAT DILATIVE ON HANT                                                                                                                          | SLICHTLY PLASTIC, 42<br>DLING).                                                                                                            | Y FINE SAND. LIGHT BROWN,                                                                                                                                  |
|                                              | DATUM<br>GROU<br>GROU<br>GROU<br>GROU<br>GROU<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS<br>CONS | IS ME<br>JND W<br>REQU<br>AMPLE<br>CE SH<br>AMMER<br>HES (<br>RY.<br>ENETR<br>'FT.<br>) SOIL<br>I.<br>T TPP<br>T BAF | AT IO                                                                                                                                       | SEA LEVIEL<br>LEVEL<br>TO DRIVI<br>DON 6" OU<br>USING<br>LING 30"<br>AMPLE<br>N RESIST<br>ASSIFICAT |                                                                                                     | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG                                                                                                                                   | BOR<br>BEAVER VALLEN<br>DUQUESN<br>SHIPPINGP<br>STONE & V<br>SKETCH NA<br>APPROVED DA                                                      | RING LOG<br>Y POWER STATION UNIT<br>NE LIGHT COMPANY<br>PORT, PENNSYLVANIA<br>WEBSTER ENG. CORP.<br>0. 12241-CSK-242A<br>TE BORNON ISHEET<br>FOR NO. SHEET |



| <br>  <br>  <br>  <br>  <br> | NCLINA<br>DATE :<br>ITATIC<br>DEPTH<br>IETHOU<br>DF<br>SI<br>OI<br>IPECIA                | TION<br>STAF<br>GRC<br>TO<br>DS :<br>RILLI<br>AMPL<br>RILLI<br>L TE | RT/F<br>DUND<br>BED<br>NG<br>NG<br>NG<br>NG<br>STIN     | VERTICAL<br>INISH<br>NATER D<br>ROCK<br>SOL<br>ROCK<br>IG OR INI                | 5/21/<br>EPT<br>60.0<br>-1/8<br>IN 0<br>ONE<br>STRU | 82<br>H / D.<br>IN RC<br>D.D. S | EARING NA INSPECTOR J. W. HCCOY<br>/ _5/24/82 CONTRACTOR / DRILLER ECER/JARVIS<br>NATE 40'10" (PT) / _5/27/82 DRLL RIG TYPE<br>(PT) TOTAL DEPTH DRILLED 60.3 (PT)<br>OLLER BIT, 3-1/4 IN I.D. CASING DRILLING MUD<br>SPLIT SPOON                                                                                                    |
|------------------------------|------------------------------------------------------------------------------------------|---------------------------------------------------------------------|---------------------------------------------------------|---------------------------------------------------------------------------------|-----------------------------------------------------|---------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| (                            |                                                                                          | NTS                                                                 |                                                         | ······································                                          |                                                     |                                 |                                                                                                                                                                                                                                                                                                                                     |
| ELEVATION<br>(FEET)(62)      | DEPTH<br>(FEE T)                                                                         | SAMPLE<br>TYPE (7)                                                  | SAMPLE                                                  | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                             | SPT N<br>VALUE (5)                                  | GROUP<br>SYMBOL (6)             | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                  |
|                              |                                                                                          | ·                                                                   | ·                                                       |                                                                                 |                                                     | ·                               |                                                                                                                                                                                                                                                                                                                                     |
| 723.9                        |                                                                                          |                                                                     | 1                                                       | 4-5-3                                                                           | 8                                                   | GP                              | SLAG, COARSE TO FINE GRAVEL SIZED, 10-20% COARSE TO FINE SAND, BROWN,<br>(FILL).                                                                                                                                                                                                                                                    |
|                              |                                                                                          | s                                                                   | 2                                                       | 2-3-2<br>(13")                                                                  | 5                                                   | SW                              | SAND, WELL GRADED, COARSE TO FINE, 10-15% COARSE TO FINE GRAVEL,<br>ROUNDED, LESS THAN 5% NONPLASTIC FINES, BROWN.                                                                                                                                                                                                                  |
|                              | 5                                                                                        | S                                                                   | .3                                                      | 2-2-3<br>(14")                                                                  | 5                                                   | SP<br>SP<br>GP                  | TOP 4 IN: <u>SAND</u> , UNIFORM, FINE, 2-5% NONPLASTIC FINES, TRACE GRAVEL, BRUMIDDLE 5 IN: <u>SAND</u> , FINE, 5-7% FINE GRAVEL, LESS THAN 5% NONPLASTIC FINE<br>VERY MOIST, LIGHT BROWN.<br>BOTTOM 5 IN: <u>COAL</u> , FINE GRAVEL SIZED FRACMENTS.                                                                               |
|                              |                                                                                          | s                                                                   | 4                                                       | 5-5-4<br>(11")                                                                  | 9                                                   | ଔ                               | SANDY GRAVEL, COARSE TO FINE, FEW TO 1 IN MAXIMUM, ROUNDED TO ANGULAR,<br>20-302 COARSE TO FINE SAND, MOSTLY COARSE, 2-52 NONPLASTIC FINES, TRACI<br>COAL AND IRON STAINING, MOIST, GRAY AND BROWN.                                                                                                                                 |
| 713.9                        | 10                                                                                       | s                                                                   | S.                                                      | 4-11-6<br>(18")                                                                 | 17                                                  | CP<br>GP                        | TOP & IN: <u>SANDY CRAVEL</u> , COARSE TO FINE, SUBANGULAR TO ROUNDED, 30-40%<br>COARSE TO FINE SAND, 5-8% NONPLASTIC FINES, TRACE COAL, GRAY.<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, 1 IN MAXIMUM, SOME WEATHER<br>SHALE FRACHENTS, 25-30% COARSE TO FINE SAND, 5-7% NONPLASTIC FINES, TR<br>IRON STAINING, BROWN. |
|                              |                                                                                          | s                                                                   | 6                                                       | 7-12-11<br>(9")                                                                 | 23                                                  | GP                              | BLOWS/INCH: 1/2-1/2-1-1/3-1-2-2-2-1/1-1-1-1-1<br><u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRACMENTS TO 14 IN<br>30-40% COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, LESS THAN 5% NONPLAY<br>FINES, TRACE COAL AND IRON STAINING, GRAY AND BROWN.<br>BLOWS/INCH: 7/2-2-3-2-2-1/3-2-1-2                             |
| / NOTES                      | DATUM<br>C GRO<br>BLOWS<br>2"Q.D. 3<br>DISTAN<br>1401b. H<br>( ) INC<br>RECOVE<br>STD. P | IS M<br>NO REQUIND<br>CE SI<br>AMME<br>CHES<br>ERY.<br>ENET         | LAN<br>WATE:<br>UIRED<br>E SPH<br>HOWN<br>R FAI<br>OF S | SEA LEVEL<br>TO DRIVE<br>DON 5" OF<br>USING<br>LLING 30".<br>AMPLE<br>DN RESIST | ANCE                                                |                                 | I<br>UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG<br>BEAVER VALLEY POWER STATION UN<br>DUQUESNE LIGHT COMPANY<br>SHIPPINGPORT, PENNSYLVANIA                                                                                                                                                                                |

| SI                      | TE    | BEAVER                | VALL     | EY POWER S                     | TAT10              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                                             |
|-------------------------|-------|-----------------------|----------|--------------------------------|--------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET )062 | DEPTH | SAMPLE                | SAMPLE   | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                   |
|                         |       | <b>T</b> -2           | τ.       |                                |                    |                     | TOR 5 TH. STITY CRAVEL COARSE TO FINE ANGINAR TO REPORT. 10-152 COARS                                                                                                                                                                                                                                                |
| 10 E E MIN              |       |                       |          | (9")                           |                    | GP                  | TO FINE SAND, MOSTLY FINE, 15-201 NONPLASTIC FINES, BROWN.<br>BOTTOM 4 IN: WEATHERED SANDSTONE FRAGMENTS, 14 IN MAXIMUM, 10-151 COARSE<br>SAND, BROWN.<br>BLOWS/INCH: 2-1-1-1-1/1-1-1-1-2/1-1-1-1-1                                                                                                                  |
|                         |       | - s                   |          | 8-13-11<br>(8")                | 24                 | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZE SANDSTONE FRACMENTS, SOME SHALE<br>TO 14 IN MAXIMUM, ANGULAR TO SUBROUNDED, 20-30% COARSE TO FINE SAND.<br>5-7% SLIGHTLY PLASTIC FINES, TRACE IRON STAINS, BROWN.<br>BLOWS/INCH: 1-1-1-1-2-2/2-2-3-2-2/2-1-2-2-2-2                                                          |
| 703.9                   | 20 -  |                       | ,        | 17-21-12<br>(11")              | 33                 | GW-<br>GP           | SANDY GRAVEL, COARSE TO FINE, ROUNDED TO ANGULAR. SOME SANDSTONE AND<br>SHALE FRAGMENTS TO 1 IN MAXIMUM, LARCE SANDSTOME FRAGMENT AT BOTTOM,<br>20-30% COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, LESS THAN 5%<br>NONPLASTIC FINES, TRACE IRON STAINING, BROWN,<br>BLOWS/INCH: 3-3-2-3-3/4-4-4-3-3-3/3-2-3-2-1-1  |
|                         | -     | - 5                   | 10       | 4-5-6<br>(13")                 | 11                 | SP                  | SAND, POORLY GRADED, COARSE TO FINE, MOSTLY MEDIUM TO FINE, 2-62 COARSE<br>TO FINE ROUNDED GRAVEL, 2-52 NONPLASTIC FINES, MOIST, BROWN.                                                                                                                                                                              |
|                         | 25    | - S                   | 11       | 4-6-7<br>(14")                 | 13                 | SP                  | <u>SAND, SIMILAR TO ABOVE, MOSTLY COARSE TO MEDIUM.</u>                                                                                                                                                                                                                                                              |
|                         | -     | -<br>-<br>-<br>-<br>- | 12<br>13 | 53/4"<br>13-24-20<br>(1")      | 53/4"<br>44        | -                   | <u>NO RECOVERY</u> : BLOWS/INCH: 8-9-17-19<br>BROKEN, ROUNDED GRAVEL TO 1½ IN (WASH?)<br>BLOWS/INCH: 2-2-2-2-3/5-5-3-3-4-4/3-3-4-3-4-3                                                                                                                                                                               |
| 693.9                   | 30 -  | 5                     | -14      | 10-10-13<br>(5")               | 23                 | -                   | <u>SANDSTONE FRAGMENTS</u> , 5-15% COARSE TO FINE GRAVEL, 10-15% COARSE TO FINE<br>SAND.                                                                                                                                                                                                                             |
|                         | -     |                       | 15       | 16-19-26<br>(7")               | 45                 | SP<br>GW            | TOP 4 IN: <u>SAND</u> , FINE, LESS THAN 52 NONPLASTIC FINES, BROWN.<br>Bottom 3 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENT<br>TO 14" MAXIMUM, 10-202 COARSE TO FINE SAND, LESS THAN 52 NONPLASTIC<br>FINES, BROWN.                                                                     |
|                         | 35 -  |                       | 16       | 14-16-27<br>(7")               | 43                 | ଦ୍ୟ                 | BLONS/INCH: 2-2-2-3-4-3/3-3-2-3-4-4/4-5-4-4-5-4<br><u>SILTY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRAMENTS TO 14 IN, ANGULAR, 15-202 COARSE TO FINE SAND, 15-202 SLIGHTLY<br>TO MEDIUM PLASTIC FINES. BROWN, GRAY AND ORANGE.<br>BLOWS/INCH: 2-3-2-2-3-2/2-3-3-2-3-3/5-5-4-2-5-6 |
|                         | -     |                       | 17       | 28-24-21<br>(11")              | 45                 | GW-<br>GP           | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRAGMENTS TO 14 IN, ANGULAR, 20-25% COARSE TO FINE SAND, 5-10% SLIGHTLY<br>PLASTIC FINES, TRACE COAL AND IRON STAINS, BROWN AND GRAY.<br>BLOWS/INCH: 4-5-8-4-4-3/4-5-4-4-3/4-4-3-4-3-3                                                    |
| 683.9                   | 40 -  | s                     | 18       | 11-11-10<br>(8")               | 21                 | SP                  | SAND, POORLY GRADED, LESS THAN 52 COARSE TO FINE GRAVEL, COARSE TO FINE<br>SAND, MOSTLY COARSE TO MEDIUM, LESS THAN 53 NOMPLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-2-2-1-2-2/2-1-2-2-2/1-2-2-2-1-2                                                                                                                     |
|                         |       | -<br>-<br>-<br>-      | 19       | 9-11-14<br>(8")                | 25                 | SP                  | SAND, SIMILAR TO ABOVE, SOFT, BLACK, CARBONACEOUS SHALE FRAGMENT AT<br>BOTTOM.<br>BLOWS/INCH: 1-2-1-2-2-1/1-2-2-2-2-2/2-2-3-2-3                                                                                                                                                                                      |

|              |       |                        |         |                   |        |            | BORING NO. EDS-2                                                                                                                                                                                                                                                                  | - |
|--------------|-------|------------------------|---------|-------------------|--------|------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|
|              |       |                        |         |                   |        |            | SHEET 3 OF 3                                                                                                                                                                                                                                                                      | - |
| S            | ITE _ | BEAVER                 | VALL    | EY POWER S        | TATIO  | N-UNI      | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                          |   |
| <b>1</b>     |       | Ē                      |         | în 🕄              | 6      | 9          |                                                                                                                                                                                                                                                                                   | ٦ |
| ET )(        | Ē     | L L                    |         | NS NG             | E H    | 22         | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                |   |
| ELEV<br>(FE) |       | SAI S                  | SAL SAL | BLO               | N N    | STE G      | · · ·                                                                                                                                                                                                                                                                             |   |
|              |       |                        |         | L                 | I      |            |                                                                                                                                                                                                                                                                                   | 4 |
|              | 45    | Т.                     | 1       |                   | 1      | -          | ALLEY CALLET LEATURED CLUTCTANE AND CULLE EDICAENTE 10-407 COADEE TO                                                                                                                                                                                                              | - |
|              |       |                        | 20      | (7")              | 103    | GW<br>GW   | SAND GAVEL, WEATHERED SANDYINE AND SALE FAMILENTS, SO-NO CONSE TO<br>FINE SAND, S-10% SLIGHTLY PLASTIC FINES, TRACE COAL, BROWN, GRAY, ORANGE.<br>BLOWS/INCH: 2-1-2-3-3-3/ 5-5-27-38-19-14/16-12-8-7-7-5                                                                          |   |
|              | .     | ┛                      |         |                   |        | <i>a</i> u | SANDY CRAVET COARSE TO FINE CRAVET FEW FRACMENTS TO 1 IN. ANGULAR TO                                                                                                                                                                                                              | - |
|              |       |                        |         | (7")              | 19     | GP<br>GP   | SAUL ONATEL, COARSE TO FINE SANGE, FUN TANDERY PLASTIC FINES, TRACE<br>ROUNDED, 15-252 COARSE TO FINE SAND, 5-102 SLIGHTLY PLASTIC FINES, TRACE<br>COAL AND IRON STAINING, FEW WEATHERED SANDSTONE AND SHALE FRACMENTS, BROWN,<br>BLOWS/INCH: 2-1-1-1-2-2/1-1-2-1-2-1/1-2-1-2-3-2 |   |
| 673.9        | 50 .  | s                      | 22      | 12-14-28          | 42     | GW-        | TOP 3 IN: SIMILAR TO ABOVE.                                                                                                                                                                                                                                                       |   |
|              |       | F                      |         | (6")              |        | GP         | BOTTOM 3 IN: <u>SAND</u> , POORLY GRADED, COARSE TO FINE, MOSTLY COARSE TO MEDIUM.<br>LESS THAN 52 NONPLASTIC FINES, BROWN.                                                                                                                                                       | - |
|              |       | -                      | 1       |                   | l.     |            | BLOWS/INCH: 2-2-2-2-2/3-3-2-2-2/2-6-4-5-5-6                                                                                                                                                                                                                                       | - |
|              |       | <b>-</b> <sup>\$</sup> | 23      | 14-13-11<br>(5")  | 24     | GW-<br>GP  | SANDY CRAVEL, COARSE TO FINE GRAVEL SIZED WEATHERED SANDSTONE AND SHALE<br>FRACMENTS, LARGE SANDSTONE FRACMENT AT TOP, ANGULAR TO SUBROUNDED, 30-402<br>COARSE TO FINE SAND, 5-102 SLIGHTLY PLASTIC FINES, ORANGE AND GRAY.                                                       |   |
|              |       | -                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   |   |
|              | 25    | <b>-</b> s             | 24      | 19-56-99          | 155    | -          | WEATHERED SANDSTONE AND SHALE, SOFT, SOME SOFT CLAYSTONE, TACE MICA, CRAY                                                                                                                                                                                                         | - |
|              |       | 7-                     | 4       |                   |        |            |                                                                                                                                                                                                                                                                                   | - |
|              |       | 1_                     | 4       |                   |        |            |                                                                                                                                                                                                                                                                                   | 7 |
|              |       | - S                    | 25      | 16-30-70<br>(15") | 100    | GW<br>GP   | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS<br>TO 14 IN, SOFT, 20-30% COARSE TO FINE SAND, 10-15% SLIGHTLY PLASTIC                                                                                                                                    | 7 |
|              |       | 1                      | 1.      |                   |        |            | FIRES, TABLE FROM STALKING, ORAL.                                                                                                                                                                                                                                                 | 7 |
| 663.9        | 60.   | -                      | 26      | 100/4 "           | 100/4  |            | CLAYSTONE, WEATHERED, SOFT, DARK GRAY.                                                                                                                                                                                                                                            | 1 |
|              |       |                        |         |                   | ·<br>• |            | BOTIOM OF BORING AT 60 FT 4 IN<br>Elevation 663.6 FT                                                                                                                                                                                                                              |   |
|              | .     | _                      |         |                   |        | 1          |                                                                                                                                                                                                                                                                                   |   |
|              |       | -                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 1 |
|              |       | 7                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | Е |
|              | -     | 7                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 7 |
|              |       | 1                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 7 |
|              |       | -                      | ľ       |                   |        | ŀ          |                                                                                                                                                                                                                                                                                   | 1 |
|              |       | -                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 4 |
|              |       | 1                      | 1       |                   |        |            |                                                                                                                                                                                                                                                                                   | 7 |
|              |       | 1                      |         |                   | 1      |            |                                                                                                                                                                                                                                                                                   | コ |
|              |       | F                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 1 |
|              | 1     | 7                      |         | 1                 |        | 1          |                                                                                                                                                                                                                                                                                   | H |
|              | .     | 7                      | ŀ       |                   |        | <b>.</b>   |                                                                                                                                                                                                                                                                                   | - |
|              |       | 1                      |         | [                 | [      |            |                                                                                                                                                                                                                                                                                   | 7 |
|              |       | 4                      |         |                   |        |            |                                                                                                                                                                                                                                                                                   | 1 |
|              |       | -                      | 1       | r                 | 4      |            |                                                                                                                                                                                                                                                                                   |   |

Rev. 0

| c          | OORDI                                          | NATE          | :s _                  | N4050       |          |               | E6147 GROUND ELEV. (1) 722.1 FT SHEETOF                                                                                   |
|------------|------------------------------------------------|---------------|-----------------------|-------------|----------|---------------|---------------------------------------------------------------------------------------------------------------------------|
| If         | ICLINA                                         | TION          | V                     | ERTICAL     |          | . BE          | EARING NA INSPECTOR J.W. MCCOY                                                                                            |
| D          | ATE :                                          | STAF          | RT / F                | FINISH _    | 5/24     | /82           | / 5/25/82 CONTRACTOR / DRILLER                                                                                            |
| 5          | TATIC                                          | GRO           | UND                   | WATER       | DEPT     | H/D           | ATERECORDEDIFT) / DRILL RIG TYPE CHE 45                                                                                   |
| D          | EPTH                                           | то            | BEDI                  | ROCK        |          | 5.5           | (FT) TOTAL DEPTH DRILLED63(FT)                                                                                            |
| Ň          | ETHO                                           | ) <b>s</b> :  |                       |             |          |               |                                                                                                                           |
|            | DI                                             | NLLI          | NG                    | soil _      | 3-1/8    | IN RO         | OLLER BIT, 3-1/4 IN I.D. CASING, DRILLING MUD                                                                             |
|            | S                                              | <b>Imp</b> L  | ING.                  | SOL -       | 2 IN (   | <u>, p. s</u> | SPLIT SPOON                                                                                                               |
|            | D                                              | RILLI         | ING F                 | HOCK _      | NONE     |               |                                                                                                                           |
| 5          | PECIA                                          | L TE          | STIN                  | ig or in    | ISTR     | UMEN          |                                                                                                                           |
| -          | <u></u>                                        |               |                       |             |          |               |                                                                                                                           |
| c          | OMME                                           | NTS .         |                       |             | ·        |               |                                                                                                                           |
|            | -                                              | <u> </u>      |                       |             |          | <u>.</u>      |                                                                                                                           |
|            |                                                |               |                       |             |          |               |                                                                                                                           |
| N          |                                                | E             |                       | jn €        |          | 9             |                                                                                                                           |
|            |                                                |               |                       | S S S       | N N      | Şq            |                                                                                                                           |
|            | 10 J                                           | SAN           | I S B                 |             |          | <b>B</b>      |                                                                                                                           |
| <u></u>    |                                                |               | L                     | <u> </u>    |          |               | <u> </u>                                                                                                                  |
| _          |                                                |               |                       |             |          |               |                                                                                                                           |
| 722.1      | 0.                                             | s             | 1                     | 6-22-9      | 31       | GP            | TOP 5 IN: SANDY SLAG AND SANDSTONE FRAGMENTS, GRAY.                                                                       |
|            | •                                              |               |                       | 0.0         |          | SW-           | ROUNDED TO SUBANGULAR, COARSE TO FINE SAND, 5-10% SLIGHTLY PLASTIC FINE                                                   |
|            |                                                | 1             | ]                     | 1           |          |               | TRACE COAL AND IRON STAINING, BROWN.                                                                                      |
|            | -                                              | 5             | 2                     | 9-16-13     | 29       | SP-           | TOP 4 IN: STHILAR TO ABOVE.                                                                                               |
|            |                                                | ]             |                       | (16")       |          | SW<br>GN      | BOTTOM 12 IN: SILTY CRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, 10-202                                                    |
|            | •                                              | ┥─            | 1                     | 1           |          |               | COARSE TO FINE SAND, 20-30% SLIGHTLY PLASTIC FINES, BROWN AND GRAY.                                                       |
|            | 5 _                                            | 5             | 1.                    | 4-7-7       | 14       | Gr            | TOP 3 IN: SINTLAR TO ABOUT                                                                                                |
|            |                                                | <u>ן</u>      | [                     | (18")       |          | ML            | BOTTOM 15 IN: GRAVELY SILT, 20-25% COARSE TO FINE GRAVEL, ANGULAR TO                                                      |
|            |                                                | <b>}</b>      | ł                     | [           | [        |               | AUGHDEN, 194134 LUARSE 10 FINE SAND, URANGE-BROWN.                                                                        |
|            |                                                |               | 1                     | 1           |          | 1             |                                                                                                                           |
|            |                                                | s             | 4                     | 7-4-4       | 8        | ML            | GRAVELLY SILT, SLIGHTLY PLASTIC, 10-15% COARSE TO FINE GRAVEL, ROUNDED,<br>5-10% FINE SAND, TRACE COAL, BROWN AND OBLAGE. |
| ·          |                                                | <u> </u>      |                       |             |          |               |                                                                                                                           |
|            | •                                              | 1             |                       |             | 1        |               |                                                                                                                           |
| 12.1       | 10 -                                           | 5             | 5                     | 3-4-10      | 14       | м             |                                                                                                                           |
| Ì          | -                                              | ľ             | 1                     | (15")       | <b>[</b> |               | SAMPLE.                                                                                                                   |
|            |                                                | <u> </u>      |                       | {           | 1        |               |                                                                                                                           |
|            | -                                              | ┣—            |                       |             |          |               |                                                                                                                           |
|            |                                                | s             | 6                     | 4-7-6 (18") | 13       | SM            | SILTY SAND, UNIFORM, LESS THAN 5% FINE GRAVEL, ROUNDED, FINE SAND,<br>20-30% NONPLASTIC FINES, BROWN.                     |
|            | -                                              |               |                       |             | [        |               |                                                                                                                           |
|            | 15                                             |               |                       |             |          |               |                                                                                                                           |
| ١.         | DATUM                                          | 15 M          | IEAN                  | SEA LEV     | EL,      |               | UNDISTURBED SAMPLES                                                                                                       |
| 2.         |                                                | UND           | WATE                  | R LEVEL     | F        |               | US-SHELBY TUBE BORING LOG                                                                                                 |
|            | 2 0.0. 5                                       |               | E SP                  | 00N 6" 0    | R        |               |                                                                                                                           |
| n .        | DISTAN<br>14015. H                             | ge si<br>Anne | hown<br>Ir fa         | LLING 30    |          |               | BEAVER VALLEY POWER STATION UN                                                                                            |
|            | ( ) INC                                        | HES           | OFS                   | SAMPLE      |          |               | DUQUESNE LIGHT COMPANY                                                                                                    |
| NOLES<br>4 | RECOV                                          |               |                       |             | TANCE    |               | SHIPPINGPORT, PENNSYLVANIA                                                                                                |
| 5. 10N -   | RECOVI                                         | ENET          | RATI                  | ON RESIS    | MILOL    |               |                                                                                                                           |
| 5 4 5 6    | RECOVI<br>STD. P<br>BLOWS<br>UNIFIEI           | ENET          | RATI                  | ASSIFICAT   | TION     |               | A STANE & WERSTER ENG CORD                                                                                                |
|            | RECOVI<br>STD. P<br>BLOWS<br>UNIFIEI<br>SYSTEM | ENET          | RAT ((<br>L CL<br>PE: | ASSIFICA    | TION     |               | STONE & WEBSTER ENG. CORP.<br>SKETCH No. 12241-GSK-244A                                                                   |

|              | TE       | BEA    | VER                | VALLE            | Y POWER S                      | TATIO              | n-unt               | BORING NO. 205-3<br>SHEET 2 OF 3                                                                                                                                                                                                                                                   |
|--------------|----------|--------|--------------------|------------------|--------------------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| (FEET ) 062) | DEPTH    | (FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                 |
| •            | <u> </u> |        |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                                                    |
|              | 15       |        | s                  | 7                | 3-4-5<br>(18")                 | 9                  | SM .                | SILTY SAND, WIDELY GRADED, 20-25% COARSE TO FINE GRAVEL, ANGULAR TO<br>ROUNDED, COARSE TO FINE SAND, 15-20% NONPLASTIC FINES, TRACE ROOTS AND<br>IRON STAINS, DARK BROWN.                                                                                                          |
|              |          |        | s ·                | 8                | 3-17-20<br>(10")               | 37                 | SM                  | <u>SIMILAR TO ABOVE</u> .<br>BLOWS/INCH: 3/2-1-1-3-5-5/4+4-4-2-4-2                                                                                                                                                                                                                 |
|              |          | ]      |                    |                  | (,                             |                    |                     |                                                                                                                                                                                                                                                                                    |
| 02.1         | 20       | ╉      | S                  | 9                | 3-3-3<br>(13")                 | 6                  | SP                  | SAND, POORLY CRADED, LESS THAN 5% FINE GRAVEL, ROUNDED, COARSE TO FINE<br>SAND, MOSTLY MEDIUM TO FINE, LESS THAN 5% NONFLASTIC FINES, BROWN.                                                                                                                                       |
|              |          | 4      |                    |                  |                                |                    |                     |                                                                                                                                                                                                                                                                                    |
|              |          |        | s                  | 10               | 2-3-6<br>(18")                 | 9                  | SW                  | SAND, WELL GRADED, LESS THAN 5% FINE GRAVEL, ROUNDED, COARSE TO FINE SAND,<br>3-7% Nonplastic fines, trace coal, brown.                                                                                                                                                            |
|              | 25       |        | S                  | 11               | 8-7-10<br>(18")                | . 17               | SW<br>GP-           | TOP 8 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, 1 IN MAXIMUM, ANGULAR                                                                                                                                                            |
|              |          | }      |                    |                  |                                |                    | GW                  | TO SUBROUNDED, JU-402 COARSE TO FINE SAND, HOSTEF COARSE TO ADDICH, LEDS<br>THAN 5% NONPLASTIC FINES, TRACE COAL AND IRON STAINS, BROWN.                                                                                                                                           |
| i            |          | ╡      | 5                  | 12               | 6-11-13<br>(11")               | 24                 | GP-<br>Gw           | SANDY GRAVEL, COARSE TO FINE, FEW FRAGMENTS TO 1-1/2 IN, ANGULAR TO<br>ROUNDED, 15-25% COARSE TO FINE SAND, 10-15% NONPLASTIC FINES, TRACE COAL<br>AND IRON STAINING, BROWN.                                                                                                       |
| 692.1        | 30       |        | 5                  | 13               | 14-11-14                       | 25                 | 62-<br>GW           | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS.                                                                                                                                                                                                           |
|              |          |        |                    | !                |                                |                    |                     | FINES, BROWN.<br>BLOWS/INCH: 2-2-3-3-2-2/1-2-2-2-2-2/2-2-3-2-3-2                                                                                                                                                                                                                   |
|              |          | Ì      | S                  | 14               | 8-10-11<br>(10")               | 21                 | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACTENTS<br>TO 1-1/2 IN, 15-20X COARSE TO FINE SAND, 5-10X SLIGHTLY PLASTIC FINES,<br>TRACE IRON STAINING, RED, LIGHT GRAY AND BOG'N, CONTAINED 1 IN THICK<br>COARSE TO FINE SAND SIZED COAL LENS AT 5 IN FROM TOP. |
|              | 35       | 4      | 5                  | 14               | 10-16-20                       | 16                 | SP                  | BLOWS/INCH: 2-2-1-1-1-1/1-1-2-2-2-2/2-2-2-1-2-2<br>TOP 5 IN: SAND, POORLY GRADED, TRACE FINE GRAVEL, COARSE TO FINE SAND,                                                                                                                                                          |
|              |          |        |                    |                  | (9")                           |                    | GP-<br>GW           | NOSTLY COARSE TO MEDIUM, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BOTTOM 4 IN: SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS-<br>TO 1-1/2 IN, ANGULAR TO ROUNDED, 23-35% COARSE TO FINE SAND, 5-10%<br>SLIGHTLY PLASTIC FINES, TRACE COAL, BROWN.                  |
|              |          | ╉      | s                  | 16               | 10-8-7<br>(5")                 | 15                 | -                   | BLOWS/INCH: 2-2-2-2-1-1/1-2-2-3-5-3/4-5-4-2-3-2<br>TOP 3 IN: <u>GRAVEL</u> , COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1-1/2 IN,<br>WASH.<br>BOTTOM 2 IN: SANDY GRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, 15-203                                                                |
|              | ļ        | +      |                    |                  |                                |                    | "                   | COARSE TO FINE SAND, 5-72 SLIGHTLY PLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-2-1-2-1-2/2-1-2-1-1/2-1-1-1-1                                                                                                                                                                            |
| 682.1        | 40       | + +    | \$                 | 17               | 11-10-15<br>(2")               | 25                 | CP<br>GW            | SANDY CRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, LARCE ANGULAR SANDSTONE<br>FRACHENT AT BOTTOM, 20-252 COARSE TO FINE SAND, 5-72 SLIGHTLY PLASTIC<br>FINES, TRACE COAL AND IRON STAINING, BROWN,<br>BLOWS/INCH: 2-2-1-2-2/1-2-2-1-2-2/2-3-3-3-2-2                                 |
|              |          |        | S                  | 18               | 12 <b>-8-8</b><br>(10")        | 16                 | GP-<br>GW<br>SW     | TOP 4 IN: <u>SIMILAR TO ABOVE</u> .<br>BOTTOM 6 IN: <u>SAND</u> , WELL GRADED, TRACE FINE GRAVEL, COARSE TO FINE SAND,<br>LESS THAN 52 NONPLASTIC FINES, TRACE COAL, BROWN.                                                                                                        |
| 1            |          | -      |                    |                  |                                |                    |                     | BLOWS/INCH: 2-2-2-2-2/1-2-1-1-2/1-2-1-1-2                                                                                                                                                                                                                                          |
| OTE :        | FOR      | BOR    | ING                | SUM              | ARY AND                        |                    | STO                 | INE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET                                                                                                                                                                                                                            |

.

| •                         |           |        |                    |             |            | 412B C               | 74710              | N_1181              | BORING NO. <u>EOS-3</u><br>SHEET <u>3</u> OF <u>3</u>                                                                                                                                                                                                                      |
|---------------------------|-----------|--------|--------------------|-------------|------------|----------------------|--------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET )(162) | DEPTH     | (FEET) | SAMPLE<br>TYPE (7) | SAMPLE      | BLOWS (3)  | REC/ROD (4)          | SPT N<br>VALUE (5) | GROUP<br>SYNBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                         |
|                           | 45        | -      | s                  | 19          | 18-<br>(5" | 15-11<br>)           | 26                 | GP-<br>GW           | SANDY GRAVEL, COARSE TO FINE, ANGULAR TO ROUNDED, 20-30% COARSE TO FINE<br>SAND, LESS THAN 5% NONPLASTIC FINES, TRACE COAL, BROWN.<br>BLOWS/INCH: 2-2-4-3-4-3/2-3-3-3-2-2/2-3-2-1-2-1                                                                                      |
|                           |           |        | 5                  | 20          | 9-1<br>(9" | 3-11<br>)            | 24                 | GP-<br>GW           | SIMILAR TO ABOVE, FEW FRACMENTS TO 1-1/2 IN.<br>BLOWS/INCH: 1-2-1-2-1-2/2-2-2-3-2/2-1-2-2-2-2                                                                                                                                                                              |
| 72.1                      | 50        |        | S                  | 21          | 6-6<br>(5" | -12                  | 18                 | GP-<br>GW           | SANDY GRAVEL. COARSE TO FINE, ANGULAR TO ROUNDED, 15-202 COARSE TO FINE<br>SAND, LESS THAN 52 NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 1-1-1-1-1-1/1-1-1-1-1/2-2-1-3-2-2                                                                                                    |
|                           |           |        | S                  | 22          | 12-<br>(7* | 13-14<br>)           | 27                 | GP                  | SANDY GRAVEL, BROKEN, WEATHERED SANDSTONE AND SHALE FRAGMENTS, COARSE TO<br>FINE GRAVEL SIZED, TO 1-1/2 IN MAXIMUM, FEW COAL FRAGMENTS, 15-20% COARSE<br>TO FINE SAND, 5-7% SLIGHTLY PLASTIC FINES, TRACE MICA, ORANGE, BROWN,<br>CRAY, BLACK.                             |
|                           | 55        |        | 5                  | 23          | 30-<br>(14 | 30-36<br>")          | <b>66</b>          | GP.                 | SANDY CRAVEL, BROKEN, WEATHERED SANDSTONE AND SHALE FRAGMENTS, COARSE TO<br>FINE GRAVEL SIZED TO 1-1/2 IN MAXIMUM, MOSTLY COARSE, 20-30% MEDIUM TO FINE<br>SAND, 5-10% SLIGHTLY PLASTIC FINES, TRACE COAL AND IRON STAINS, ORANGE,<br>RED, BROWN, GRAY.                    |
|                           |           |        | S                  | 24          | 25-<br>(14 | 13-15<br>")          | 28                 | GP-<br>GW           | BLOWS/INCH: 3-3-7-3-5-3/6-8-5-4-4-3/4-3-6-6-10-7<br><u>SANDY GRAVEL</u> , COARSE TO FINE, 1-1/2 IN MAXIMUM, ANGULAR TO ROUNDED, 20-302<br>COARSE TO FINE SAND, 5-72 NONPLASTIC FINES, TRACE IRON STAINS, ORANGE<br>BROWN.<br>BLOWS/INCH: 4-6-5-5-2-3/2-3-2-2-2/2-2-3-2-3-3 |
| 2.1                       | 60        |        | s                  | 25          | 18-<br>(12 | - <b>30-80</b><br>") | 110                | -                   | SHALE, COARSE TO FINE GRAVEL SIZED FRAGMENTS, SOFT, WEATHERED, 5-10Z FINE<br>FINE SAND, 25-35Z SLIGHTLY PLASTIC FINES, TRACE COAL, GRAY AND BROWN.<br>BLOWS/INCH: 3-4-3-2-3-3/3-4-4-4-7-8/15-20-13-13-10-9                                                                 |
|                           |           |        | 5                  | 26          | 105<br>(4" | /6"<br>')            | 105/<br>6"         |                     | SHALE, SOFT, WEATHERED, GRAY.                                                                                                                                                                                                                                              |
|                           |           |        |                    |             |            |                      |                    |                     | ELEVATION 659.1 FT                                                                                                                                                                                                                                                         |
|                           |           |        |                    |             |            |                      |                    |                     |                                                                                                                                                                                                                                                                            |
|                           |           |        |                    |             |            |                      |                    |                     |                                                                                                                                                                                                                                                                            |
|                           |           |        | I                  |             |            |                      |                    |                     |                                                                                                                                                                                                                                                                            |
| )TE: P                    | OR<br>EGF |        | ting<br>NFQ.       | SUMM<br>SEE | ARY<br>SHE | AND<br>ET 1          | 4                  | STO<br>SKE          | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TGH NG. 12241-GSK-244C DDA 9/.62 EOS-3 3 OF 3                                                                                                                                                                    |

|                | S<br>C<br>D<br>S<br>S<br>C             | ITE<br>CORING<br>NCLIN<br>ATE<br>TATIC<br>EPTI<br>IETH<br>PECI                                            | BEAU<br>DINAT<br>IATIO<br>STA<br>C GR<br>H TO<br>ODS:<br>CORILL<br>SAMP<br>DRILL<br>T<br>DRILL<br>T<br>ENTS         | ER VAL<br>ES<br>N<br>RT/<br>OUND<br>BED<br>LING<br>ESTII                | LLEY POWER<br>N4164.41<br>VERTICAL<br>FINISH                                                                 | 5/2/<br>5/2/<br>DEPT<br>-1/5 J<br>IN 0.<br>DNE<br>STR | IN 0. SI            | NIT 2<br>E6101.98 GROUND E<br>EARING NA<br>/ 5/26/82 CONTRACT<br>ATERECORDECTOPY / NA<br>(FT) TOTAL DEP<br>D. ROLLER BIT. 3-1/4 IN ID CA<br>PLIT SPOON AND 3 IN 0.D. SHEL<br>ITATION NONE | _ J.O. NO. <u>12241</u><br>LEV. (I) <u>720.1 FT</u><br>INSPECTOR <u>J.W. MCC</u><br>OR / DRILLER <u>EGER/</u><br>DRILL RIG TYPE<br>TH DRILLED<br>SING, DRILLING MUD<br>BY TUBE | SHEETOF                                                                                                                  |
|----------------|----------------------------------------|-----------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------|-------------------------------------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|
| ELEWION        | (FEET)(16.2)                           | DEPTH                                                                                                     | SAMPLE                                                                                                              | SAMPLE                                                                  | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                                          | SPT N<br>VALUE (5)                                    | GROUP<br>SYMBOL (6) | SAMF                                                                                                                                                                                      | PLE DESCRIPTION                                                                                                                                                                | • <u>•</u> ••••••••••••••••••••••••••••••••••                                                                            |
| 720            | .1                                     | 0                                                                                                         | - s                                                                                                                 | 1                                                                       | 3-6-11<br>(10")                                                                                              | 17                                                    | GP-<br>GW           | SANDY GRAVEL, COARSE TO FIN<br>COARSE TO FINE SAND, 5-102<br>STAINING, BROWN.                                                                                                             | E, 1¼ IN MAXIMUM, ANGU<br>NONPLASTIC FINES, TRAC                                                                                                                               | TLAR TO ROUNDED, 25-35%<br>E ROOTS, IRON                                                                                 |
|                |                                        |                                                                                                           | - 5                                                                                                                 | 2                                                                       | 11-17-12<br>(10")                                                                                            | 29                                                    | GP-<br>GW           | <u>SANDY GRAVEL</u> , SIMILAR TO AB<br>57 NONPLASTIC FINES, DARK B                                                                                                                        | OVE, 30-407 COARSE TO<br>ROWN.                                                                                                                                                 | FINE SAND, LESS THAN                                                                                                     |
|                |                                        | 5                                                                                                         | - 5                                                                                                                 | 3                                                                       | 10-13-10<br>(12")                                                                                            | 23                                                    | SW                  | GRAVELLY SAND, WELL-GRADED,<br>SAND, 5-7Z NONPLASTIC FINES<br>BLOWS/INCH 1-1-2-2-2-2/3-2-                                                                                                 | 20-30% COARSE TO FINE<br>, TRACE COAL AND IRON<br>2-2-2-2/1-2-2-2-2-1                                                                                                          | GRAVEL, COARSE TO FINE<br>STAINING, DARK BROWN.                                                                          |
|                |                                        |                                                                                                           |                                                                                                                     | 4                                                                       | ,<br>9-7-7<br>(7")                                                                                           | 14                                                    | SP                  | <u>GRAVELLY SAND</u> , POORLY GRADE<br>FINE SAND, 5-107 NONPLASTIC                                                                                                                        | D, 20-30% COARSE TO FI<br>FINES, TRACE IRON STA                                                                                                                                | NE GRAVEL, MEDIUM TO<br>INS, DARK BROWN.                                                                                 |
| 710            | 0.1                                    | 10                                                                                                        |                                                                                                                     | 5                                                                       | 4-13-11<br>(12")                                                                                             | 24                                                    | SP<br>GP            | TOP 7 IN: <u>Similar to above</u> .<br>Bottom 5 IN: <u>Sandy Broken G</u><br>Trace iron Statning.<br>BLOWS/INCH: 4/2-1-2-3-3-2/2                                                          | RAY SANDSTONE, 30-40%                                                                                                                                                          | COARSE TO MEDIUM SAND,                                                                                                   |
|                |                                        |                                                                                                           |                                                                                                                     | 6                                                                       | 4-17-11<br>(12")                                                                                             | 28                                                    | GP-<br>GW           | SANDY GRAVEL, COARSE TO FIN<br>ANGULAR TO ROUNDED, 25-35%<br>5-10% NONPLASTIC FINES, TRA<br>BLOWS/INCH: 4/1-4-3-3-3-3/3                                                                   | E, FEW SANDSTONE FRAGM<br>COARSE TO FINE SAND, M<br>DE IRON STAINING, GRAY<br>-1-2-2-2-1                                                                                       | IENTS TO 14 IN MAXIMUM,<br>IOSTLY MEDIUM TO FINE,<br>AND DARK BROWN.                                                     |
| LEGEND / NOTES | 1.<br>2.<br>3.<br>4.<br>5.<br>6.<br>7. | DATUI<br>GR<br>BLOW<br>2"0.D.<br>DISTA<br>40b.<br>( ) II<br>RECO<br>STD.<br>BLOW<br>UNIFI<br>SAMP<br>S-SP | I IS<br>OUND<br>S REG<br>SAMP<br>NCE I<br>HAMM<br>NCHES<br>VERY.<br>PENE<br>S/FT.<br>ED SC<br>EM.<br>LE T'<br>LIT B | L.<br>WEAN<br>WATE<br>DUREL<br>LE SP<br>BHOWN<br>ER FA<br>OF :<br>TRATI | SEA LEVI<br>R LEVEL<br>D TO DRIVI<br>DOON 6" OI<br>I USING<br>ILLING 30"<br>SAMPLE<br>ON RESIST<br>ASSIFICAT | EL<br>E<br>TANCE                                      | • • •               | UNDISTURBED SAMPLES<br>US-SHELBY TUBE<br>UO-OSTERBERG                                                                                                                                     | BORIN<br>BEAVER VALLEY P<br>DUQUESNE<br>SHIPPINGPOR<br>SHIPPINGPOR<br>SKETCH NO.<br>APPROVED DATE                                                                              | G LOG<br>OWER STATION UNIT<br>LIGHT COMPANY<br>T, PENNSYLVANIA<br>ISTER ENG. CORP.<br>12241-GSK-245A<br>BORING NO. SHEET |

|                          |       | <u> </u>           |        |                                |                    |                     | BORING NO                                                                                                                                                                                                                                            |
|--------------------------|-------|--------------------|--------|--------------------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SI                       | TE    | BEAVER             | VALL   | EY POWER S                     | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                             |
| ELEVATION<br>(FEET)(162) | DEPTH | SAMPLE<br>TVDE (1) | SAMPLE | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                   |
|                          | 15    |                    | , '    | 9-11-10<br>(14")               | 21                 | SP-<br>SW           | GRAVELLY SAND, 30-402 COARSE TO FINE GRAVEL TO 14 IN MAXIMUM, ANGULAR<br>TO ROUNDED, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, LESS THAN 52<br>NONPLASTIC FINES, TRACE IRON, BROWN.<br>BLOWS/INCH: 9/2-2-3-1-2-1/2-2-2-1-1-2                       |
|                          | -     |                    | 8      | 6-8-12<br>(8")                 | 20                 | ଟ-<br>ଟ-<br>ମ       | SANDY GRAVEL. COARSE TO FINE, ANGULAR TO ROUNDED, 20-302 COARSE TO FINE<br>SAND, LESS THAN 52 NONPLASTIC FINES, BROWN.                                                                                                                               |
| 700.1                    | 20 -  |                    | ],     | 14-12-10<br>(13")              | 22                 | G                   | SILTY GRAVEL, COARSE TO FINE, MOSTLY MEDIUM TO FINE GRAVEL SIZED WEATHERED<br>SANDSTONE AND SHALE FRAGMENTS, 10-15% COARSE TO FINE SAND, 15-20% SLIGHTLY<br>PLASTIC FINES, TRACE COAL AND MICA, TRACE IRON STAINING, BROWN, GRAY.<br>IRON AND BLACK. |
|                          | -     |                    | 10     | 11 <b>-63</b> -74<br>(14")     | 137                | 5M<br>              | TOP 9 IN: <u>SILTY SAND</u> , POORLY GRADED, MEDIUM TO FINE SAND, 10-12Z NON-<br>PLASTIC FINES, TRACE FINE GRAVEL, BROWN.<br>BOTTOM 5 IN: <u>SLAG</u> , GRAY.<br>BLOWS/INCH: 1-1-1-1-3-4/7-11-16-10-10-9/20-23-8-11-6-6                              |
|                          | 25 .  | - S                | 111    | 10-11-17<br>(0")               | 28                 | -                   | NO RECOVERY<br>BLOWS/INCH: 1-2-1-2-2-2/2-1-2-2-2/2-3-3-4-3-2                                                                                                                                                                                         |
|                          | -     |                    | 12     | 10-9-6<br>(1")                 | 15                 | SP                  | CRAVELLY SAND, 20-30% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, COARSE<br>TO FINE SAND, MOSTLY MEDIUM TO FINE, LESS THAN 5% NONPLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-2-2-1-1-2/1-2-2-2-1-1/1-1-1-1-1                                               |
| <b>690.</b> 1            | 30 -  |                    | 13     | 4-5-6<br>(11")                 | 11                 | c.                  | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, MEDIUM STIFF TO STIFF,<br>OCCASIONAL FINE GRAVEL TO $\frac{1}{2}$ IN, ROUNDED, 5-72 FINE SAND, MOIST,<br>MOTTLED BROWN, GRAY BROWN WITH POCKETS OF GRAY. $q_u$ (pp):2.5TSF                               |
|                          |       |                    | 14     | 18-15-19<br>(13")              | 34                 | 68-<br>64           | SANDY GRAVEL, COARSE TO FINE, FEW TO 1.5 IN MAXIMUM, 30-402 COARSE TO<br>FINE SAND, MOSTLY COARSE TO MEDIUM, LESS THAN 52 NONPLASTIC FINES,<br>TRACE IRON STAINING, GRAY.<br>BLOWS/INCH: 1-2-4-3-4-4/2-3-3-2-2-3/3-3-4-3-3-3                         |
|                          | 35 -  |                    | 15     | 3-4-4<br>(15")                 | 8                  | c.                  | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, MEDIUM STIFF TO STIFF, TRACE FINE GRAVEL, MOIST, GRAY. $q_u$ (pp): 2.0TSF.                                                                                                                               |
|                          |       | US                 | 1      | (23.5*)                        |                    | ĊĻ.                 | SANDY CLAY, MODERATELY PLASTIC, 10-152 MEDIUM TO FINE SAND, FEW PIECES<br>COAL UP TO 3/8 IN, DARK GRAYISH BROWN.                                                                                                                                     |
|                          | ,     | - 5                | 16     | 4-4-5<br>(12")                 | , °                | CL                  | SILTY CLAY, MEDIUM STIFF TO STIFF, MODERATELY PLASTIC, LESS THAN 52<br>FINE SAND, BROWN. qu (pp): 2.0TSF                                                                                                                                             |
| 980.1                    | 40 *  | us<br>S            | 2      | (23")<br>4-4-4<br>(12")        | 8                  | a                   | SINILAR TO S16 (TUBE TRIMMINGS).<br><u>SILTY CLAY</u> , MEDIUM STIFF TO STIFF, SLIGHTLY TO MODERATELY PLASTIC, LESS<br>THAN 52 FINE SAND, BROWN WITH GRAY MOTTLING. qu (pp): 2.2515F                                                                 |
|                          |       |                    | 3      | (0")                           |                    |                     | NO RECOVERY.                                                                                                                                                                                                                                         |
| OTE : 1                  | FOR B | loring<br>D NFQ    | SUMA   | HARY AND SHEET I.              | 4                  | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TCH No. 12241-GSK-2438 DN 9/1/82 EOS- 4 2 OF 3                                                                                                                                             |

..

| )((62) |                | 3     | w e:  | 6            | (*) (        | , 3            | 9               |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
|--------|----------------|-------|-------|--------------|--------------|----------------|-----------------|------------------|---------------------|-------------|-----------------|------------------|------------------|----------------------|----------|-------|----------|-------|-------|---|
| (FEET) | DEPTI<br>(FEET | SAMPL | SAMPL | BLOWS        | OR<br>REC/RO | SPT N<br>VALUE | GROUF<br>SYMBOL |                  |                     |             |                 | SAMF             |                  | DESCR                | IPTIO    | N .   |          |       |       |   |
|        | 45             |       |       |              |              | 1              | r .             |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
|        |                | s     | 18    | 0-4-<br>(18" | -4<br>')     | 8              | CL              | SAND,            | CLAY,<br>BROWN      | MODI        | RATEL<br>Iu (pp | Y PLAS<br>): 1.7 | TIC, M<br>5, 0.7 | EDIUM S1<br>5, 1.251 | IFF TO   | STIFF | 7, 10% V | ERY F | INE   | - |
|        | ·              | US .  | 4     | (16"         | )            |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | • |
|        | -              | s     | 19    | 4-6-<br>(18' | -5<br>')     | 11             | CL.             | SILTY<br>BROWN.  | , <u>CLAY</u><br>qu | SLI(<br>(PP | GHTLY<br>): 0.5 | TO MOD           | ERATEL           | Y PLAST              | LC, SOFT | TON   | IEDIUM S | TIFF, | MOIST |   |
| •••    | - 00           |       |       | а.           |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              | US    | 5     | (0")         |              |                | CL              | NO REC<br>PUSHED | OVERY<br>SPLI       | T SP        | DON (S          | -2Ó) -           | RECOV            | ERED ST              | LTY CLAY | SIMI  | LAR TO   | s-19. | ,     | - |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  |                     | Ē           | OTTOM           | OF BO            | RING AT          | r 53.0 A             | T        |       |          |       |       |   |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  | •                   |             |                 |                  |                  |                      |          |       |          |       |       |   |
|        | _              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | _              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              | ļ              |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
|        |                |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | • |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
|        |                |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | • |
|        | -              |       | ŀ     |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              | ļ              |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       |   |
| 1      | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        |                |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | • |
|        |                |       |       |              |              |                |                 | į .              |                     |             |                 |                  |                  |                      |          |       |          |       |       | - |
|        | -              |       |       |              |              |                |                 |                  |                     |             |                 |                  |                  |                      |          |       |          |       |       | • |

| C<br>IN<br>D<br>S<br>D<br>M                  | CORDINATE<br>ICLINATION<br>ATE: STAI<br>TATIC GRO<br>EPTH TO<br>ETHODS:<br>DRILLI<br>SAMPI                                                                                          | ES<br>RT/F<br>DUNDY<br>BEDR                                            | N4158.7<br>ERTICAL<br>INISH<br>WATER C<br>NOCK<br>SOL 3-<br>SOL 2              | 5/27/82<br>EPTH /<br>72.5<br>7/8 IN 1<br>IN 0.D. | E6105.3<br>BEARINGN<br>                                              | GROUND EL                                       | EV.(I)<br>ISPECTOR _<br>IR / DRILLEI<br>DRILL RIG T<br>IN DRILLED<br>ILLING MUD<br>IBERG | 0.4<br>J.W. MCCO<br>EGER/.<br>YPE<br>72.8                               | SHEET_L                                               | <u>OF 2</u>                            |
|----------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------|--------------------------------------------------------------------------------|--------------------------------------------------|----------------------------------------------------------------------|-------------------------------------------------|------------------------------------------------------------------------------------------|-------------------------------------------------------------------------|-------------------------------------------------------|----------------------------------------|
| S                                            | ORILL<br>PECIAL TE                                                                                                                                                                  | ING R<br>Estin                                                         | OCK <u>NO</u><br>G OR IN                                                       | STRUM                                            | ENTATION NONE                                                        |                                                 |                                                                                          |                                                                         | · · · · · · · · · · · · · · · · · · ·                 |                                        |
| ELEVATION<br>(FEET)(162)                     | DEPTH<br>(FEET)<br>SAMPLE                                                                                                                                                           | SAMPLE<br>NUMBER                                                       | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                            | 9PT N<br>VALUE (5)<br>GROUP                      | SYMBOL (6)                                                           | SAMP                                            | LE DESCR                                                                                 |                                                                         |                                                       |                                        |
|                                              |                                                                                                                                                                                     |                                                                        |                                                                                |                                                  |                                                                      | NO SAMPLES                                      | TO 35.5 FT                                                                               | <del></del>                                                             |                                                       | -                                      |
|                                              | 35 - UO                                                                                                                                                                             | 1                                                                      | (30.5")                                                                        |                                                  |                                                                      |                                                 |                                                                                          |                                                                         |                                                       |                                        |
| 680.4                                        | 40 110                                                                                                                                                                              | 1                                                                      | 5-4-5<br>(14")                                                                 | 9 (                                              | L SILTY CLAY, MO<br>ORGANICS, BROW<br>Qu (PP): 2.25,<br>NO RECOVERY. | DERATELY PLAST<br>N WITH SOME MO<br>2.0, 2.5TSF | IC, STIFF. L<br>Ttled gray.                                                              | ERS THAN 5%                                                             | FINE SAND, 1                                          | TRACE                                  |
|                                              |                                                                                                                                                                                     | 3                                                                      | (30")                                                                          | - (                                              | L SIMILAR TO 5-1                                                     | (TRIMMINGS                                      | )                                                                                        |                                                                         |                                                       | -                                      |
| 1.<br>2.<br>3.<br>3.<br>4.<br>5.<br>6.<br>7. | 45 5<br>DATUM IS &<br>Q. GROUND<br>BLOWS REQ<br>20.0. SAMPI<br>015TANCE 3<br>401b. HAMMA<br>() INCHES<br>RECOVERY.<br>510. PENET<br>9LOWS/FT.<br>JAIFIED SO<br>SYSTEM.<br>SAMPLE TY | LE AN S<br>WATER<br>UIRED<br>LE SPO<br>HOWN<br>IR FAL<br>OF S<br>RATIO | SEA LEVEL<br>TO DRIVE<br>DON 6" OF<br>USING<br>LING 30".<br>AMPLE<br>IN RESIST | ANCE                                             | UNDISTURBED<br>US-SHELBY<br>UO-OSTER                                 | SAMPLES<br>Tube<br>Berg                         | BEAVER VA<br>DUQI<br>SHIPP                                                               | BORING<br>LLEY POW<br>JESNE LIN<br>INGPORT,<br>E & WEBST<br>CH No. 1224 | LOG<br>YER STATH<br>GHT COMP<br>PENNSYL<br>ER ENG. CO | -<br>DN UNIT-2<br>PANY<br>VANIA<br>RP. |

| SI           | ITE   | BEAVER                | VALL   | EY POWER S                         | TATIO              | N-UN I              | I 2. SHIPPINGPORT. PA. J.O. NO. 12241.                                                                                                                            | BORING NO                      | E05-44<br>DF        |
|--------------|-------|-----------------------|--------|------------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------|---------------------|
| (FEET )(162) | DEPTH | SAMPLE<br>TYPE (7)    | SAMPLE | BLOWS (3)<br>OR<br>REC/RQD (4)     | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                |                                |                     |
|              | 45    | s                     | 2      | 3-3-4                              | 7                  | CL                  | SILTY CLAY, MODERATELY PLASTIC, MEDIUM STIFF, TRACE                                                                                                               | FINE SAND, M                   | DIST.               |
| •            |       | ╡                     | ŀ      |                                    |                    |                     | BROWN WITH GRAY MOTILING.<br>qu (pp): 1.25, 1.75TSF                                                                                                               |                                |                     |
|              | -     | 1 00                  | 4      | (29.8")                            |                    | сı                  | SIMILAR TO S-2. (TRINMINCS)                                                                                                                                       |                                | -                   |
| • .          |       | -<br>-<br>-<br>-<br>- | 3      | 3-4-6<br>(18")                     | 10                 | cL                  | SIMILAR TO S-2. TRACE ORGANIC MATERIAL.<br>qu (pp): 1.25T5F                                                                                                       |                                |                     |
| 10,4         | 50-   | ╡──                   | 4      |                                    |                    |                     |                                                                                                                                                                   |                                | -                   |
|              |       |                       | 5      | (30")                              |                    | CL                  | <u>SIMILAR TO S-Z.</u> (TRIMMINGS)                                                                                                                                |                                | -                   |
|              | -     | s s                   | 4      | 3-4-5                              | 9                  | CL                  | SANDY CLAY, MODERATELY PLASTIC, STIFF, 23% VERY FIN                                                                                                               | E SAND, BROWN                  |                     |
|              |       | ]                     |        |                                    | ·                  |                     |                                                                                                                                                                   | · ·                            | -                   |
|              | 55    | -                     | ]      |                                    |                    |                     |                                                                                                                                                                   |                                | -                   |
|              |       | <b>-</b> "            | 6      | (30")                              |                    | CL                  | SIMILAR TO 5-2. (TRIMMINGS)                                                                                                                                       |                                |                     |
|              | -     | 5                     | . 5    | 3-5-4<br>(18")                     | 9                  | 다                   | TOP 8 IN: <u>SIMILAR TO S-2.</u><br>BOTTOM 10 IN: <u>SILTY CLAY</u> , MODERATELY PLASTIC, SOFT,<br>LENSES LESS THAN 1 mm THICK, GRAY. q <sub>u</sub> (pp): 0.75TS | CONTAINS FIN<br>F              | E SAND              |
|              |       | - vo                  | ,      | (30.5")                            |                    |                     |                                                                                                                                                                   |                                | -                   |
|              |       | ]                     |        | · ,                                |                    |                     |                                                                                                                                                                   |                                | -                   |
| 50.4         | 60 -  |                       | .6     | 2-2-4<br>(18")                     | 6                  | cī.                 | <u>SANDY CLAY</u> , SLIGHTLY PLASTIC, 20-25% VERY FINE SAND<br>VERY FINE SAND LENSES, 5 mm THICK, GRAY. qu (PP)                                                   | , MEDIUM STIF<br>1.0, 0.75TSF  | F, SOME             |
| -            | _     | - 00                  | •      | (29.3")                            |                    |                     |                                                                                                                                                                   |                                | -                   |
|              | ·     | 7                     |        |                                    |                    |                     |                                                                                                                                                                   |                                | -                   |
|              | 65 -  |                       | ,      | 3-3-6<br>(16")                     | 9                  | a,                  | SANDY CLAY - SANDY SILI, SLIGHTLY PLASTIC, 15-202 VE<br>FINE SAND LENSES LESS THAN 1-2 mmn THICK, NUMEROUS S<br>1 mmn DIAMETER, HOIST, DARK GRAY.                 | RY FINE SAND,<br>Mall white de | CONTAINS<br>POSITS, |
| ĺ            |       | 1                     | 1      | 1                                  |                    |                     |                                                                                                                                                                   |                                | -                   |
|              | -     | s                     | 8      | 29-28-19<br>(10")                  | 47                 | CP                  | <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE<br>WEATHERED, MAXIMUM SIZE 1-1/2 IN.                                                                  | AND SHALE FR                   | AGMENTS ,           |
| ~            |       |                       | ]      |                                    |                    |                     |                                                                                                                                                                   | •                              | -                   |
| 0.4          | 70 -  | 1                     |        |                                    |                    |                     |                                                                                                                                                                   |                                | -                   |
|              |       | 4                     |        |                                    |                    |                     |                                                                                                                                                                   |                                |                     |
|              | -     |                       | l°     | <sup>13-15-<u>101</u><br/>4"</sup> | 116<br>10"         |                     | TOP 10 IN: <u>BROKEN SANDSTONE AND SHALE</u> , SOFT, WEATH<br>BOTTOM 4 IN: <u>SHALE</u> , SOFT, GRAY.<br>BLOWS/IN: 3-2-3-1-3-1/3-2-2-2-3-3/6-10-40-45.            | IRED.                          | -                   |
|              | 1     |                       |        |                                    |                    |                     | BOTTOM OF BORING AT 72 FT 10 IN<br>Elevation 647.6 Ft                                                                                                             | · · ·                          |                     |
|              | FOR 8 |                       | 31.00  |                                    |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE                                                                                                                             | BORING NO.                     | SHEET               |

-

| 3<br>0<br>0<br>5<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0<br>0 | COORDINATE<br>COORDINATE<br>NCLINATION<br>DATE : STAN<br>ITATIC GRC<br>DEPTH TO<br>IETHODS :<br>DRILLI<br>SAMPL<br>DRILLI<br>DRILLI<br>DRILLI<br>COMMENTS                           | R VALL<br>ES J<br>RT/F<br>BEDJ<br>NG S<br>LING F<br>SSTIN                            | LEY POWER<br>N4300<br>Vertical<br>Finish 3<br>Water 1<br>Rock<br>Sol<br>Sol<br>Rock<br>Ig or in       | 51.3-1/8<br>51.3-1/8<br>51.0<br>51.0<br>51.0<br>51.0<br>51.0<br>51.0<br>51.0<br>51.0 | <u>E</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u><br><u>B</u> | UNIT 2<br>GOS7 GROLAD E<br>GARING NA<br>/ 6/2/82 CONTRACT<br>ATERECORDELIGTY /<br>(FT) TOTAL DEF<br>LLER BIT, 3-1/4 IN I.D. CASY<br>PLIT SPOON, 3 IN O.D. SHELBY<br>TATION NONE | J.O. NO<br>:LEV (1) 683.0<br>INSPECTOR<br>INSPECTOR<br>OR / DRILLER<br>DRILL RIG TYPE<br>DRILLED<br>SI<br>ING. DRILLING MUD.<br>Y TUBE AND OSTERBERG. | BORING NO. E05-<br>SHEETOF3                                                               |
|------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(IL2)                                                     | DEPTH<br>(FEET)<br>SAMPLE<br>TVBF (7)                                                                                                                                               | SAMPLE                                                                               | BLOWS (3)<br>AND/OR<br>RECOVERY (4)                                                                   | SPT N<br>VALUE (3)                                                                   | GROUP<br>SYMBOL (6)                                                                                                              | SAM                                                                                                                                                                             | PLE DESCRIPTION                                                                                                                                       |                                                                                           |
| 683.0                                                                        | 0 5                                                                                                                                                                                 | 1                                                                                    | 2-2-1<br>(0")                                                                                         | 3                                                                                    |                                                                                                                                  | NO RECOVERY                                                                                                                                                                     |                                                                                                                                                       |                                                                                           |
|                                                                              |                                                                                                                                                                                     | 2                                                                                    | 1-1-1<br>(7")                                                                                         | 2                                                                                    | ML                                                                                                                               | <u>Silt</u> , slightly to moderate<br>Brown with orange mottling                                                                                                                | ELY PLASTIC, SOFT, TRACE 1<br>;.                                                                                                                      | FINE SAND AND ROOTS,                                                                      |
|                                                                              | 5 \$                                                                                                                                                                                | 3                                                                                    | 1-1-1                                                                                                 | 2                                                                                    | મા                                                                                                                               | <u>Sandy Silt</u> , slightly plas?<br>Material.                                                                                                                                 | TIC, SOFT, 15-20% FINE SAU                                                                                                                            | ND, SOME ORGANIC                                                                          |
|                                                                              | e<br>e<br>s                                                                                                                                                                         | 4                                                                                    | 1-1-2                                                                                                 | 3                                                                                    | ML                                                                                                                               | SINILAR TO ABOVE.                                                                                                                                                               | · 7                                                                                                                                                   |                                                                                           |
| 673.Q                                                                        | 10 <b>-</b> 5                                                                                                                                                                       | 5                                                                                    | 1-1-1<br>(7")                                                                                         | 2                                                                                    | ML                                                                                                                               | <u>SANDY SILT</u> , SLIGHTLY TO MC<br>ORGANIC MATERIAL, BROWN.                                                                                                                  | DDERATELY PLASTIC, 15-202                                                                                                                             | FINE SAND, TRACE                                                                          |
|                                                                              | - us                                                                                                                                                                                | 1                                                                                    | (0")                                                                                                  |                                                                                      |                                                                                                                                  | NO RECOVERY.                                                                                                                                                                    |                                                                                                                                                       |                                                                                           |
|                                                                              | _15 S                                                                                                                                                                               | 6                                                                                    | 1-2-2<br>(5")                                                                                         | 4                                                                                    | CL/ML                                                                                                                            | SANDY CLAY - SANDY SILT, MC<br>BROWN.                                                                                                                                           | DERATELY PLASTIC, SOFT, 1                                                                                                                             | LS-20% FINE SAND,                                                                         |
| I. 2. 3.<br>3. 4. 5. 6. 7.<br>7.                                             | DATUM IS M<br>GROUND SECONS<br>2"GL DANFL<br>10ISTANCE SI<br>1401b. HAMME<br>() INCHES<br>SECOVERY.<br>STD. PENET<br>BLOWS/FT.<br>UNIFIED SUSSIFT.<br>SAMPLE TYI<br>SAMPLE TYI<br>T | IEAN<br>WATEI<br>LIRED<br>E STHOWN<br>R FAL<br>OF S<br>RATIC<br>L CLI<br>PEI<br>RPFI | SEA LEVI<br>R LEVEL<br>TO DRIVI<br>DON S" OL<br>USING<br>LING 30".<br>AMPLE<br>DN RESIST<br>ASSIFICAT | EL<br>R<br>FANCE                                                                     |                                                                                                                                  | UNDISTURBED SAMPLES<br>US-Shel,by Tuge<br>UO-OSTERBERG                                                                                                                          | BORING<br>BEAVER VALLEY POW<br>DUQUESNE LI<br>SHIPPINGPORT,<br>STONE & WEBST<br>SKETCH NO. 12<br>APPROVED L DATE                                      | LOG<br>VER STATION UNIT<br>GHT COMPANY<br>PENNSYLVANIA<br>ER ENG. CORP.<br>1241-GSK- 247A |

|                             |             |                   |                    |        | CY POURS C                    | TATIO              | N+LINT              | BORING NO. 205-5<br>SHEET 2 OF 3                                                                                                                                                                                                                                              |
|-----------------------------|-------------|-------------------|--------------------|--------|-------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162) CA | HLd30       | (FEET)            | SAMPLE<br>TYPE (7) | SAMPLE | REC/ROD (4)                   | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                            |
|                             | 15          |                   |                    | r      | · .                           |                    |                     |                                                                                                                                                                                                                                                                               |
|                             |             | 111               | s                  | ,      | 2-2-2<br>(15")                | 4                  | a.m.                | SANDY CLAY-SANDY SILT, SLIGHTLY PLASTIC, SOFT, 20-25% FINE SAND, BROWN.                                                                                                                                                                                                       |
|                             |             |                   | UÒ                 | 2      | ~(30")                        |                    | CL                  | SANDY CLAY, MODERATELY PLASTIC, 30-402 MEDIUM TO FINE SAND, MOSTLY<br>FINE, MOTTLED LIGHT BROWN, CRAYISH BROWN AND YELLOW BROWN.                                                                                                                                              |
| 563.0                       | 20          | ыТы               | s                  | 8      | 2-2-3<br>(18")                | 5                  | ci.                 | SANDY CLAY, SLIGHTLY PLASTIC, 30Z FINE SAND, BROWN AND GRAY WITH ORANGE<br>MOTTLING. qu(pp) = 0.75, 1.0 TSF                                                                                                                                                                   |
|                             |             | 111               | υō                 | 3      | (28")                         | · .                |                     |                                                                                                                                                                                                                                                                               |
|                             | - 25        | 1111              | 5                  | 9      | 3-2-2<br>(18'')               | 4                  | CL<br>ML-<br>MH     | TOP 13 IN: SIMILAR TO S-8.<br>BOTTOM 5 IN: ORGANIC <u>CLAYEY SILT</u> , MODERATELY TO HIGHLY PLASTIC, TRACE                                                                                                                                                                   |
|                             |             | 1111              | US                 | 4      | (27")                         |                    |                     | SANDY CLAY, MODERATELY PLASTIC, 12-202 VERY FINE SAND, GRAY.<br>(TUBE TRIMMINGS)                                                                                                                                                                                              |
|                             |             | LLL               | S                  | 10     | 2-2-11<br>(18")               | 13                 | SM-<br>CL           | LAYERED SILTY SAND AND SANDY CLAY, LAYER THICKNESS 1/4 IN TO 3/4 IN,<br>SAND IS FINE, CLAY IS MODERATELY PLASTIC, SOFT, GRAY.                                                                                                                                                 |
| \$53.0                      | 30          | LILL              | 5                  | 11     | 17-19-16<br>(6")              | 35                 | SP                  | <u>GRAVELLY SAND</u> , FINE TO COARSE GRAVEL TO 1 IN, COARSE TO FINE SAND. MOSTLY<br>FINE, 10-137 NONPLASTIC FINES, CONTAINS SEVERAL PIECES OF FRACTURED<br>SANDSTONE INDICATING SPOON SAMPLED COBBLE.<br>BLOWS/INCH: 2-3-3-3-3-3/4-3-3-4-2-3/2-3-3-3-2-3                     |
|                             |             | بلفين             | S                  | 12     | 10-20-14<br>(7")              | 34                 | GP                  | GRAVELLY SAND, 20-30% COARSE TO FINE GRAVEL, FEW SANDSTONE FRAGMENTS<br>TO 1-1/2 IN, ANGULAR TO ROUNDED, COARSE TO FINE SAND, 5-10% NONPLASTIC<br>FINES, TRACE IRON STAINING, GRAY.<br>BLOWS/INCH: 2-2-2-2-2-2/1-3-3-3-5-5/3-4-2-2-1-2                                        |
|                             | 35          | L. L. L. L. L. L. | S                  | 13     | 19-18-12<br>(4")              | 30                 | CP                  | SANDY CRAVEL, COARSE TO FINE GRAVEL SIZED, MOSTLY COARSE WEATHERED SAND-<br>STONE AND SHALE, 14 IN MAXIMUM, ANGULAR (SOME ROUNDED), 15-20% COARSE TO<br>FINE SAND, LESS THAN 5% NOMPLASTIC FINES, TRACE IRON STAINING, GRAY.<br>BLOWS/INCH: 4-4-2-4-2-3/4-3-2-3-3-3/2-2-2-2-2 |
|                             |             | LILL              | 5                  | 14     | 21-10-6<br>(8")               | 16                 | GP                  | SANDY GRAVEL, COARSE TO FINE, ROUNDED, CONTAINS SOME WEATHERED SANDSTONE<br>AND SHALE FRAGMENTS TO 1 IN MAXIMUM, 20-30% COARSE TO FINE SAND, TRACE<br>NOMPLASTIC FINES, GRAY.<br>BLOWS/INCH: 4-3-5-3-3-3/2-2-2-1-2-1/1-1-1-1-1                                                |
| i43.0                       | 40          | 1111              | S                  | 13     | 9-11-9<br>(9")                | 20                 | GW                  | GRAVEL, WELL GRADED, COARSE TO FINE, FEW FRAGMENTS TO 15 IN, ANGULAR TO<br>ROUNDED, 10-15% COARSE TO FINE SAND, GRAY.<br>BLOWS/INCH: 2-2-1-1-2-1/2-2-1-2-2-2/2-1-1-2-1-2                                                                                                      |
|                             |             |                   | S                  | 16     | 25-10 <del>-</del> 9<br>(11") | 19                 | GW<br>SP            | TOP 6 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ANGULAR, CONTAINS SANDSTONE<br>FRAGMENTS TO 15 IN MAXIMUM, 30-35% COARSE TO FINE SAND, GRAY.<br>BOTTOM 5 IN: <u>SAND</u> , POORLY GRADED, 5-10% COARSE TO FINE GRAVEL, ROUNDED,<br>COARSE SAND, GRAY.                         |
| OTE :                       | 1 4:<br>FOR | 80                | RING               | SUMA   | LARY AND                      |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET                                                                                                                                                                                                                        |

|                                           | AVER VAL                     | TEX POUER S                              | TATION             |                     | BORING NO. <u>E05-5</u><br>SHEET <u>3 OF 3</u><br>12241.00                                                                                                                                |
|-------------------------------------------|------------------------------|------------------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162)<br>DEPTH<br>I TI | SAMPLE<br>TYPE (7)<br>SAMPLE | NUMBER<br>BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                        |
| 45                                        | S 1                          | 7 30-15-7<br>(9")                        | 22                 | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS,<br>FEW TO 1-1/2 IN MAXIMUM, WEATHERED, SOFT, 30-402 COARSE TO FINE SAND,<br>TRACE IRON STAINING, BROWN AND GRAY. |
|                                           |                              | 8 <u>60</u><br>1"                        | <u>60</u><br>1"    |                     | NO RECOVERY.                                                                                                                                                                              |
| 13.0 50 -                                 | S 1                          | 9 69-26-90<br>(14")                      | 116                | GP                  | SIMILAR TO S-17, DARK GRAY SHALE AT BOTTOM, SOFT.                                                                                                                                         |
| 1                                         | 3 2'                         | 0 <u>100</u><br>3"                       | <u>100</u><br>3''  |                     | SHALE, SOFT, DARK GRAY.<br>BOTTOM OF BORING AT 51 FT 3 IN                                                                                                                                 |
|                                           |                              |                                          |                    |                     | ELEVATION 631.75                                                                                                                                                                          |
| 1.1.1                                     |                              |                                          |                    |                     |                                                                                                                                                                                           |
| 1                                         |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
|                                           |                              |                                          |                    |                     |                                                                                                                                                                                           |
| TE: FOR BO                                | RING SLA                     | MARY AND                                 |                    | STO                 | NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET                                                                                                                                    |

| S               | TE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | EAVER                    | VALLET         | Y POWER                                                                                                                  | STAT           | ION-U         | NIT 2                                |                          | _ J.O. NO     | 12241                                              | BORING N                       | 0. <u>10</u>  |
|-----------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------|----------------|--------------------------------------------------------------------------------------------------------------------------|----------------|---------------|--------------------------------------|--------------------------|---------------|----------------------------------------------------|--------------------------------|---------------|
| C               | OORDIN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ATES                     | <u></u> N:     | 3848                                                                                                                     |                |               | E6173                                | GROUND EL                | EV. (1) 745.  | 1                                                  | SHEET                          | <u>.0r _3</u> |
| IN              | CLINAT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 'ION                     | V              | ERTICAL                                                                                                                  |                | 86            | ARING                                |                          | NSPECTOR _    | J.W. MCCC                                          | )Y                             |               |
| D               | ATE : S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | TART                     | /Fin           | ISH                                                                                                                      | 6/8/8          | 2             | <u> </u>                             | CONTRACTO                | DR / DRILLE   | R                                                  | ARVIS                          | ·····         |
| 5               | STATIC GROUNDWATER DI         DEPTH TO BEDROCK         METHODS:         DRHLLING SOL         SAMPLING SOL         ORHLLING ROCK         SPECIAL TESTING OR INS         COMMENTS         COMMENTS         S         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         I         II         III         IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | JEFINI UNIE UNIEL NU ITE |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
|                 | EFTH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ТО 191<br>п.,            | EDRO           | UK                                                                                                                       | <u></u>        |               |                                      | IVIAL DEP                | IN UNILLEI    | ·                                                  |                                | <u></u>       |
| -               | DP:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 9.<br>9. 1 1 1 1 1       | 6 80           |                                                                                                                          | 3-1/8<br>CASIN | IN R<br>G, WA | OLLER BIT TO ADV<br>TER.             | ANCE HOLE, 3             | IN O.D. SPL   | IT SPOON TO                                        | CLEAN OUT, 4                   | IN I.D        |
|                 | SA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | MPLIN                    | NG SO          | х <u>–</u>                                                                                                               | 2 IN           | 0.D.          | SPLIT SPOON                          |                          |               |                                                    |                                |               |
|                 | OR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | ILLIN                    | G ROC          | ск                                                                                                                       |                |               |                                      |                          | ·             |                                                    |                                |               |
| \$              | STATIC GROUNDWATER D         DEPTH TO BEDROCK                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                          | STR            | JMEN                                                                                                                     | TATION 2 FT NO | RTON POROUS   | PIEZOMETER I                         | NSTALLED WIT             | TH TIP AT EL  | 710.1                                              |                                |               |
| ~               | STATIC GROUNDWATER D         DEPTH TO BEDROCK45         METHODS:         DRILLING SOL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                          | <u> </u>       |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
| v               | DATE: START / FINISH       .64         STATIC GROUNDWATER DE       DEPTH TO BEDROCK       .48         METHODS:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
| ନ୍ଦ             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ন                        |                |                                                                                                                          | 1              |               |                                      |                          |               | ·····                                              |                                |               |
| 100 H           | ŦĒ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                          |                | 8 8 8<br>8 8 8<br>8 8 8                                                                                                  | z S            | <u>ار تو</u>  |                                      |                          |               |                                                    |                                |               |
| FEE             | DEF<br>(FE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                          |                | AND/                                                                                                                     |                | S S           |                                      | SAMP                     | LE DESCA      | RIPTION                                            |                                |               |
| <u>สั</u>       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          |                | REC                                                                                                                      |                | ι<br>Ο        |                                      |                          | - <u> </u>    |                                                    |                                |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
| 745.1           | ٩                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | s                        | 1 1.           | -3-3<br>6")                                                                                                              | 6              | -             | TOPSOIL, SILT, I<br>DARK BROWN.      | ESS THAN 52              | FINE SAND, 1  | 5 IN SANDS                                         | TONE FRACMENT                  | AT TIP        |
|                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          |                |                                                                                                                          |                |               |                                      |                          |               | <b>A# A2 · · · · · · · · · · · · · · · · · · ·</b> |                                |               |
|                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | s                        | 2 4            | -4-6<br>18'')                                                                                                            | 10             | CL            | SANDY CLAY, MODE<br>SANDSTONE, SHALE | AND COAL FI              | AGMENTS, ANG  | ULAR, 22% CO                                       | ) FINE GRAVEI<br>DARSE TO FINE | SIZED         |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          | ,              |                                                                                                                          |                |               | BROWN, MOTTLED V                     | ITH YELLOW I             | BROWN AND GRA | Ι,                                                 |                                |               |
|                 | ۲_,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | s                        | 3 4-           | -7-8<br>18")                                                                                                             | 15             | CL.           | SIMILAR TO S-2.                      |                          |               |                                                    |                                |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
|                 | - 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | s                        | 4 6            | -8-8                                                                                                                     | 16             | CL            | SIMILAR TO S-2.                      |                          |               |                                                    |                                |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          | Ċ              | 18")                                                                                                                     |                |               |                                      |                          |               |                                                    |                                |               |
|                 | 4                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | -                        |                | _4_9                                                                                                                     |                |               | CTWT1 40 40 4-5                      |                          |               |                                                    | -                              |               |
|                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                          |                | - <b></b><br>11")                                                                                                        | 14             | <u>с</u> ь,   |                                      |                          |               |                                                    | •                              | •             |
| 735.1           | 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | s                        | 6 4<br>()      | -5-5<br>18")                                                                                                             | 10             | CL.           | SILTY CLAY, SLIC<br>FRACMENTS, MOIST | HTLY PLASTIC<br>, BROWN. | C, STIFF, OCC | ASIONAL COA                                        | RSE SAND AND                   | COAL          |
|                 | 7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | -+                       | _ <u> </u>     | -                                                                                                                        | $\mathbb{H}$   | CL/           | SILTY CLAY, SLIC                     | HTLY PLASTIC             | . 4% VERY FI  | NE SAND. BR                                        | DWN.                           |               |
|                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | s                        | 7 3.           | -2-3<br>14")                                                                                                             | 5              | MT.           | TOP 4 IN: SANDY                      | SILT. NONPLA             | STIC TO SLIG  | HTLY PLASTI                                        | C. 15-20% FI                   | NE SAND       |
|                 | <u>_</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                          |                | /                                                                                                                        | V              | (TP           | WET, BROWN.                          | DV CRAVET                |               | IF CRAVEL ST                                       | 7FD SANDSTON                   | ר אות כו      |
|                 | Ŧ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 5                        | 8 6            | -8-5                                                                                                                     | 13             | ,             | FRACMENTS, 1 II                      | MAXIMUM, AN              | GULAR TO ROU  | NDED, 20-30                                        | COARSE TO                      | FINE SAL      |
| ┯┛              | 15                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                          | Ċ              | 14")                                                                                                                     |                | SP            | BOTTOM 4 IN: SI                      | TY SAND, UNI             | FORM, FINE,   | 10-15% NONP                                        | LASTIC FINES                   | BROWN         |
| 1.1             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | IS ME/                   | AN SE          |                                                                                                                          | EL             |               | UNDISTURBED                          | AMPLES                   |               | ROPING                                             | 106                            |               |
| 3.              | LOWS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | REQUI                    | RED TI         | O DRIVE                                                                                                                  | E              |               | UO-OSTERSE                           | RG                       | . <del></del> | 500100                                             | <u> </u>                       | -             |
|                 | E Q.D. SA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | E SHO                    | SPOOI          | n 6 Of<br>Sing                                                                                                           | त<br>।         |               |                                      |                          | BEAVER V      | ALLEY PON                                          | VER STATI                      | ON UN         |
| 5 4             | 401b. HA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | HES O                    | FALLI<br>F SAN | ING 30".<br>IPLE                                                                                                         |                |               |                                      | · ·                      | ουα           | UESNE LI                                           | GHT COM                        | PANY          |
|                 | RECOVE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | RY.<br>Netr/             | ATION          | RESIST                                                                                                                   | ANCE           |               |                                      |                          | SHIPF         | NGPORT,                                            | PENNSYL                        | VANIA         |
| < <b> </b> 5. ∮ |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                          |                |                                                                                                                          |                |               |                                      |                          |               |                                                    |                                |               |
| - 5.<br>GN 6.   | LOWS/                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | FT.<br>SOIL              | CLAS           | SIFICAT                                                                                                                  | ION            |               |                                      |                          | A STOP        | IF S WEDOT                                         | FD FNG CO                      | 80            |
| 5.<br>GN 39.    | DATE : START / FINISHS/E<br>STATIC GROUNDWATER DEP<br>DEPTH TO BEDROCK<br>METHODS :<br>DRILLING SOL<br>SAMPLING SOL<br>ORILLING ROCK<br>SPECIAL TESTING OR INST<br>COMMENTS<br>COMMENTS<br>COMMENTS<br>COMMENTS<br>COMMENTS<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>D<br>D<br>S<br>S<br>S<br>S<br>D<br>D<br>D<br>D<br>D<br>D<br>S<br>S<br>S<br>D<br>S<br>D<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D<br>D _ | ION                      |                | алана<br>1917 - Алана Ал |                | STON<br>SKE   | NE & WEBSI                           | TER ENG. CO              | RP.           |                                                    |                                |               |

| SI                        | TE 31           | LAVER              | VALL              | EY POWER S                     | TATIO              | n-un i                 | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                                                  |
|---------------------------|-----------------|--------------------|-------------------|--------------------------------|--------------------|------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET )(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPL.E<br>NUMBER | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6)    | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                        |
|                           | - 18            |                    |                   | r                              |                    |                        |                                                                                                                                                                                                                                                                                                                           |
|                           |                 | 5                  | 9                 | 3-4-5<br>(17")                 | 9                  | SP<br>SP               | TOP 6 IN: <u>SIMILAR TO S-8</u> , BOTTOM 4 IN.<br>BOTTOM 11 IN: <u>SAND</u> , COARSE TO FINE, MOSTLY MEDIUM TO FINE, 7-8% NONPLASTIC<br>FINES, BROWN, CONTAINS OCCASIONAL POCKET OF SILTY CLAY, MODERATELY<br>PLASTIC, BROWN.                                                                                             |
|                           | -               | s                  | 10                | 5-6-7<br>(15")                 | 13                 | SP                     | SAND, POORLY GRADED, 5-72 FINE GRAVEL, ROUNDED, MEDIUM TO FINE SAND,<br>7-12 % NONPLASTIC FINES, MOIST, BROWN.                                                                                                                                                                                                            |
| 725.1                     | 20              | s                  | 11                | 4-3-5<br>(15")                 | 8                  | SP<br>ML               | TOP 2 IN: <u>SIMILAR TO S-10</u> .<br>BOTTON 13 IN: <u>SILT</u> , NONFLASTIC, TRACE FINE GRAVEL SIZED SANDSTONE AND<br>COAL, SOME LENSES OF SANDY SILT, MOIST, BROWN.                                                                                                                                                     |
|                           |                 | Ś                  | 12                | 4-6-9<br>(18")                 | 15                 | ML                     | SILT, NONPLASTIC, TRACE FINE SAND, WET, BROWN.                                                                                                                                                                                                                                                                            |
|                           | 25 -            | 5                  | 13                | 4-3-4<br>(18'')                | 7                  | ML-<br>SM              | LAYERED SILT AND SILTY FINE SAND, TRACE FINE GRAVEL SIZED ROCK FRAGMENTS,<br>NONPLASTIC FINES, WET, BROWN.                                                                                                                                                                                                                |
|                           |                 | s                  | 14                | 2-3-4<br>(18")                 | 7                  | CL ·                   | <u>SILTY CLAY-CLAYEY SILT</u> , SLIGHTLY TO MODERATELY PLASTIC, 12 VERY FINE<br>SAND, BROWN.                                                                                                                                                                                                                              |
|                           | -               | 5                  | 15                | 2-2-2<br>(10")                 | 4                  | SM                     | SILTY SAND, UNIFORMLY GRADED, FINE, TRACE COARSE SAND, 20-25% NONPLASTIC<br>TO SLIGHTLY PLASTIC FINES, WET, BROWN.                                                                                                                                                                                                        |
| 715.1                     | 30              |                    | 16                | 3-7-5<br>(18")                 | 12                 | SP<br>GP               | TOP 11 IN: <u>SAND</u> , UNIFORMLY GRADED, FINE, 5-77 NORPLASTIC FINES, WET, BROWN.<br>BOTTOM 7 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE, ANGULAR TO ROUNDED, FEW<br>SANDSIONE FRAGMENTS TO 1.5 IN, SOME COAL, 20-302 COARSE TO FINE SAND, 52<br>SLIGHTLY PLASTIC FINES, TRACE IRON STAINING, BROWN, ORANGE.              |
|                           | -               | s                  | 17                | 5-4-4<br>(18")                 | 8                  | sh<br>GP-<br>GW        | TOP 10 IN: <u>SILTY SAND</u> , 5-103 COARSE TO FINE GRAVEL SIZED COAL FRAGMENTS<br>TO 1 IN, FINE SAND, 15-202 NONFLASTIC FINES.<br>BOTTOM 8 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, FEW<br>SANDSTONE FRAGMENTS TO 1 IN, 15-2G% COARSE TO FINE SAND, LESS THAN 5%<br>NONFLASTIC FINES, BROWN. |
|                           | -<br>35 -       | s                  | 18                | 4-5-5<br>(18")                 | 10                 | SP-<br>SM<br>SP-<br>SM | SILTY SAND, FINE, TRACE FINE GRAVEL AND COAL FRAGMENTS, 10-15% NONPLASTIC<br>FINES, SANDSTONE FRAGMENTS AT BOTTOM.<br>TOP 13 IN:SIMILAR TO S-18.                                                                                                                                                                          |
|                           |                 | 5                  | 19                | 7-8-11<br>(18")                | 19                 | GP                     | BOTTON 5 IN: <u>SANDY CRAVEL</u> , CURRE IN FIRE GRAVEL SIZED SANDSTONE AND SHALL FRACMENTS TO I IN MAXIMUM, ANGULAR TO ROUNDED, 15-20% COARSE TO FINE SAND, TRACE IRON STAINING, BROWN, GRAY, BLACK.                                                                                                                     |
|                           | -               | 5                  | 20                | 49- <u>81</u><br>2"            | 81<br>2 "          | Sr<br>GP               | COAL, ANGULAR TO ROUNDED, MEDIUM TO FINE SAND, S-102 NONPLASTIC FINES.<br>TRACE IRON, BROWN, GRAY.<br>TOP 13 IN: SANDY GRAVEL, COARSE TO FINE, ROUNDED, SOME BROKEN SANDSTONE                                                                                                                                             |
| 705.1                     | 40-             | s                  | 21                | 26-34-17<br>(18")              | 51                 | GP                     | AND SHALE, 20-302 COARSE TO FINE SAND, LESS THAN 52 NONPLASTIC FINES, TRACE<br>COAL, BROWN, GRAY, ORANGE BROWN.<br>BOTTOM 5 IN: <u>GRAVEL</u> , BROWN SANDSTONE FRAGMENTS TO 1.5 IN, SANPLED COBBLE,<br>GRAY.                                                                                                             |
|                           | -               | s                  | 22                | 20-16-103<br>(18")             | 119                | GH<br>GP               | TOP 12 IN: <u>SILTY GRAVEL</u> , COARSE TO FINE GRAVEL, NOSTLY COARSE TO 1 IN,<br>ANGULAR, 25-302 COARSE TO FINE SAND, 15-202 SLIGHTLY PLASTIC FINES, WET,<br>BROWN.<br>BROWN & IN: SANDSTONE FRAGMENTS, SAMPLED COBBLE.                                                                                                  |
|                           | -               | s                  | 23                | 33-107-33<br>(11")             | 140                | GP                     | BLOWS/INCH: 3-3-2-4-4-4/2-2-1-2-3-6/5-4-30-34-18-12<br>SANDY CRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS, SOME COAL,<br>1.5 IN MAXIMUM, 20-252 COARSE TO MEDIUM SAND, 5-103 SLIGHTLY PLASTIC                                                                                                                  |
|                           | 45<br>FOR 80    | RNG                |                   | ARY AND<br>SHEET L             | 4                  | STO                    | FINES, TRACE HICA, TRACE IRON STAINS, BROWN, GRAT, ORANGE.<br>NE & WEBSTER ENG. CORP. APPROVED DATE BORING NO. SHEET<br>TCH No. 12241-CSK-2483 DDA 9/102 EOS-6 2 OF 3                                                                                                                                                     |

|                          |                                       |        |                                   |                    |                     |                                                                                                                                          | BORING NO. | EOS-4 |
|--------------------------|---------------------------------------|--------|-----------------------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------|------------|-------|
| SI                       | TE_BEAVER                             | VALU   | EY POWER S                        | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.                                                                                                   | 00         |       |
| ELEVATION<br>(FEET)(IE2) | DEPTH<br>(FEET)<br>Sample<br>Type (7) | SAMPLE | BLOWS (3)<br>OR<br>REC/RGD (4)    | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                       |            | -     |
| a                        | ·                                     |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          | 45 <b>-</b> S                         | 24     | 36-28-41<br>(13")                 | 69                 | CP.                 | SIMILAR TO S-22, TOP.                                                                                                                    |            |       |
|                          |                                       | 25     | 21-71- <u>103</u><br>15"<br>(11") | <u>174</u><br>75"  | GP                  | TOP 5 IN: <u>SIMILAR TO S-22</u> , TOP.<br>MIDDLE 2 IN: <u>SANDSTONE FRAGMENTS</u> , SOFT, GRAY.<br>BOTTOM 4 IN: <u>COAL FRAGMENTS</u> . |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          | [ ייין                                |        |                                   |                    |                     | BOTTOM OF BORING AT 48.1 FT<br>Elevation 697.0 FT                                                                                        |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          | -                                     |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
| · .                      |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    | ļ                   |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          | -                                     |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
| Í                        |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          | ]                                     |        |                                   |                    |                     |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    | ŀ                   |                                                                                                                                          |            |       |
|                          |                                       |        |                                   |                    | ·                   |                                                                                                                                          |            |       |
|                          |                                       | L      | <u> </u>                          |                    |                     |                                                                                                                                          |            |       |

| SITE                                                                  | · · · · · · · · · · · · · · · · · · · | J.O. NO.    |
|-----------------------------------------------------------------------|---------------------------------------|-------------|
| Beaver Valley Power Station-Unit 2                                    | <u></u>                               | 12241       |
| DATE DRILLER Eger/Jarvis                                              |                                       | J. W. McCoy |
| COORDINATES N3848 E6173 GR                                            | OUND ELEV. 74                         | 5.1 Ft      |
| INSTALLED IN BORING EOS-6 ELEV. TOP                                   | OF LEADS746                           | .9 ft       |
| RIG & CREW TIME 5 hours                                               |                                       |             |
| •                                                                     |                                       |             |
| DETAILED INSTALLATION<br>DESCRIPTION :                                | *                                     |             |
| Hole cleaned to 48.2 ft.                                              |                                       |             |
| (Bottom of drill casing).                                             | ╶║ ┌──┐──╟╴                           |             |
| Install Norton porous tube piezometer 28                              |                                       | - 22"       |
| centering spider approximately                                        | 7                                     | mmm         |
| Sand placed to 30 ft-3 in.                                            |                                       |             |
| Bentonite pellets placed to 29 ft-4 in.                               |                                       |             |
| Difficulty when pulling casing - had to hold piezometer down since it | Sand                                  |             |
| tended to pull out with casing.                                       | Janu                                  |             |
| Sand to ground surface.                                               |                                       |             |
| Guard pipe grouted in place.                                          |                                       |             |
| •                                                                     |                                       |             |
|                                                                       |                                       |             |
| DESCRIPTION OF PIEZOMETER TIP                                         |                                       | 147         |
| 2 ft Norton porous tube.                                              | Bentonite                             |             |
| Approximately 35 ft-3/4 in I.D.                                       | 30                                    | 48.<br>'3'' |
| PVC riser pipe with centering                                         |                                       |             |
| spider.                                                               |                                       |             |
|                                                                       |                                       |             |
|                                                                       |                                       |             |
|                                                                       |                                       | 21          |
| ELEVATION :                                                           |                                       | 1           |
| Sand and gravel                                                       | V                                     |             |
| NOTE: Piezometer appears clogged with silt.                           |                                       | 13.2'       |
| Probe to 31.75 ft from top of casing.                                 | Sand                                  |             |
| Abandon piezometer. 8/11/82                                           |                                       |             |
|                                                                       |                                       |             |
|                                                                       |                                       |             |
|                                                                       |                                       |             |

## 2.5E-51

| s         | TE BEAVER V                  | ALLEY POW                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | ER STAT    | ION-UI      | NIT 2                                                             | J.O. NO. 12241                                |                                                   |
|-----------|------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------|-------------|-------------------------------------------------------------------|-----------------------------------------------|---------------------------------------------------|
| . C       | OORDINATES                   | <u>N3812</u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |            |             | GROUND E                                                          | LEV. (I)                                      | SHEET_LOF_2                                       |
| H         | ICLINATION _                 | VERTICAL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |            | . BE        | ARING                                                             | NSPECTOR Med                                  | DY                                                |
| D         | ATE : START                  | / FINISH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | _6/3/8     |             | NOT RECORDED                                                      | DR / DRILLER                                  |                                                   |
| S         | TATIC GROUP                  | WWATER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | DEPT       | "H7U<br>4 5 | AIE/                                                              | DHALL RIG ITPE .                              | 45.0                                              |
|           | EPTH 10 BE                   | DROCK .                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |            | •           | IVIAL DEP                                                         | IN UKILLED                                    | (F1                                               |
| N         | 0011 11003 .<br>0011 1 1100  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 3 1/A      | TN RO       | LLER BIT TO ADVANCE HOLE, 3 1                                     | IN G.D. SPLIT SPOON 1                         | O CLEAN OUT                                       |
|           | CAMPIN                       | G SOL -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2 IN C     | ).D. S      | PLIT SPOON                                                        | ·                                             |                                                   |
|           | DRILLING                     | ROCK                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |            |             |                                                                   |                                               |                                                   |
| . 3       | PECIAL TEST                  | FING OR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | INSTR      | UMEN        | TATION 2 FT POROUS STONE                                          | PIEZOMETER, INSTALLE                          | D WITH TIP AT EL. 716.9                           |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             | · · · · · · · · · · · · · · · · · · ·                             | <u>.</u>                                      |                                                   |
| c         | OMMENTS                      | ORING ADV                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | NICED W    | I THOU      | T WATER. DID NOT ENCOUNTER AN                                     | Y GROUNDWATER.                                |                                                   |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   | ·                                             |                                                   |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   |                                               | · · · · · · · · · · · · · · · · · · ·             |
| র         | িম                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | = I        |             |                                                                   |                                               |                                                   |
| N01       | EFUJU                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            | <u> </u>    | · · ·                                                             |                                               |                                                   |
| MT<br>EE1 |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             | SAMF                                                              | PLE DESCRIPTIC                                | DN .                                              |
|           | 0 =  3   3                   | 포금국                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 8          | <u>ع</u> ا  |                                                                   |                                               |                                                   |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | - 1        |             | · · · · · · · · · · · · · · · · · · ·                             |                                               |                                                   |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            | <b>1</b>    |                                                                   |                                               |                                                   |
| 759.9     | 0 <b>s</b> :                 | 1 4-7-9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 16         | -           | FILL, SLAG AND SILTY GRAVE                                        | EL, COARSE TO FINE, 1                         | TRACE ROOTS AND IRON ST.                          |
|           |                              | (3")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |            |             | GRAY.                                                             |                                               |                                                   |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   |                                               |                                                   |
|           |                              | 4-7-6                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 113        | ML          | GRAVELLY SILT, SLIGHTLY PLA<br>WEATHERED SANDSTONE AND SHA        | NSTIC, 10-15Z COARSE<br>NLE, ROUNDED TO SUBAN | TO FINE GRAVEL SIZED<br>NGULAR, 15-20% COARSE     |
|           | ┥ ┥                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            | m/          | TO FINE SAND, SOME ROOTS SI                                       | IGHTLY MOIST, GRAY.                           | ·                                                 |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 1          | <b>™</b> ∎  | TOP 8 IN: <u>GRAVELLY SILT-GRA</u><br>20-30% COARSE TO FINE GRAVE | VELLY CLAY, SLIGHTLY<br>L, SOME WOOD FRAGMEN  | TO MODERATELY PLASTIC<br>TTS, GRAY AND BROWN.     |
|           | 5                            | (16")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 1"         | -           | BOTTOM 10: COAL AND SHALE I<br>GRAVEL AND SAND SIZED FRACE        | FRAGMENTS, WIDELY GRATENTS, TRACE IRON STA    | DED, COARSE TO FINE                               |
|           | $\overline{+}$               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   |                                               |                                                   |
|           | s s                          | 7-6-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 11         | CL          | SILTY CLAY, SLIGHTLY TO MOD<br>OF COAL FRAGMENTS AND SANDS        | STONE FRAGMENTS TO 1.                         | IFF, CONTAINS FEW LAYER:<br>5 IN MAXIMUM, FEW RED |
|           |                              | (16")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |            |             | SHALE FRAGEMENTS, 7-102 COA                                       | ARSE TO FINE SAND, VE                         | ERY SLIGHTLY MOIST, BROW                          |
|           | ſ ŢŢ                         | 4-7-6                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | <b>.</b> , | ] _ [       |                                                                   | WIN AND OBANGE                                |                                                   |
|           |                              | (16")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |            |             | STATEMA TO S-4, MULLED BRG                                        |                                               |                                                   |
| 40 -      |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   |                                               |                                                   |
| /47.9     | - s (                        | 5 3-5-8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 13         | CL.         | SIMILAR TO S-4. CONTAINED                                         | L IN THICK LAYER OF S                         | SILTY CLAY WITHOUT COAR                           |
|           | 1                            | (13")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |            |             | FRACTION, MOTTLED GRAY AND                                        | BROWN.                                        |                                                   |
|           |                              | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |            |             |                                                                   | C. STIFF OCCASTONAL                           | . FINE CRAVEL STREN                               |
|           |                              | 7-8-8                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 16         | CL          | SANDSTONE PARTICLE, 15-202                                        | COARSE TO FINE SAND,                          | , SOME MINOR IRON                                 |
|           |                              | (16")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |            |             | STAINING, SLIGHTLY MOIST, E                                       | KUWN .                                        |                                                   |
| ſ         |                              | 4-7-7                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 14         | CL.         | SANDY CLAY, SLIGHTLY TO MOL                                       | ERATELY PLASTIC, STI                          | IFF, 10-15% FINE GRAVEL                           |
|           | 15 8 8                       | ' (13")                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |            | Ĺ           | TO 3/4 IN MAXIMUM, ANGULAR,                                       | 15-20% COARSE TO FI                           | INE SAND, BROWN.                                  |
| 1.        | DATUM IS MEA                 | N SEA LE<br>Ter Levei                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | VEL        |             | UNDISTURBED SAMPLES<br>US-SHELBY TUBE                             | BORI                                          | NG LOG                                            |
| 3.        | BLOWS REQUIR                 | SOL 3<br>G SOL 2<br>FOCK -<br>FING OR IN<br>ORING ADVAN<br>C S SOL 2<br>FOCK -<br>FING OR IN<br>SOL 7<br>C S SOL 2<br>FOCK -<br>FOCK -<br>S SOL 2<br>FOCK -<br>S SOL 2<br>S SOL 2<br>FOCK -<br>S SOL 2<br>S SOL | IVE<br>OR  |             | UO-OSTERBERG                                                      |                                               |                                                   |
| ű l       | DISTANCE SHOW                | VN USING                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | <b>~</b> " | 8.          | SAMPLE CONTAINS PIECES OF<br>SANDSTONE 1.5 IN DIAMETER            | BEAVER VALLEY                                 | POWER STATION UN                                  |
| 2 4.      | () INCHES OF                 | SAMPLE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ν.         |             | AND 1/8 IN THICK, INDICATING<br>SAMPLER PENETRATED COBBLE         | DUQUESNE                                      | LIGHT COMPANY                                     |
| 5         | RECOVERY.<br>Std. Penetra    | TION RESI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | STANCE     | ſ           | OR BOULDER. TYPICAL OF                                            | SHIPPINGPO                                    | RT, PENNSYLVANIA                                  |
| ę [       | BLOWS/FT.                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ATION      | -           | INIS MALERIAL.                                                    |                                               | · · ·                                             |
| ຍີ່ ອີ    | SYSTEM.                      | venagir ili                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |            |             |                                                                   |                                               | ISTER ENG. CORP.                                  |
| ۳         | SAMPLE TYPE:<br>S-SPLIT BARR | EL SAMPL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | E          |             |                                                                   | APPROVED DATI                                 | E BORNG NO. SHEET                                 |
|           |                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |            |             |                                                                   |                                               |                                                   |

| State         State <th< th=""><th>S</th><th>ITE_</th><th>BEAVE</th><th>R VAL</th><th>EY POWER S</th><th>TATIO</th><th>N-UN I</th><th>BORING NO. <u>E05-7</u><br/>SHEET <u>2</u> OF <u>2</u><br/>T 2, SHIPPINGPORT, PA. J.O. NO. <u>12241.00</u></th><th>-</th></th<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | S                         | ITE_  | BEAVE            | R VAL  | EY POWER S                     | TATIO                  | N-UN I              | BORING NO. <u>E05-7</u><br>SHEET <u>2</u> OF <u>2</u><br>T 2, SHIPPINGPORT, PA. J.O. NO. <u>12241.00</u>                                                                                                                                                                                                                                        | -     |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------|-------|------------------|--------|--------------------------------|------------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------|
| 15         5         6         6         12         CL         TOP 10 IN: SILTY CLAY. SILCHILY TO MOREATELY FLASTIC, STIFT, 15-202           16         5         6         64-6         12         CL         COARSE TO FINE GAMEL SIZED SANDSTORE AND SHALF FRAGERISTS TO 1.5 IN           17         5         10         18-17-122         29         GF         CANNEN, FEV COARSE, GAMALE AND SHALF, HODEARCELY FLASTIC TALE THE           18.9         10         18-17-122         29         GF         SATEY CANVEL, POOLY CRABED, COARSE CANVEL SUBE SANDSTORE FRAGERIS, FEV TO 110         SIEL SATEY CANVEL, POOLY CRABED, COARSE CANVEL SUBE SANDSTORE FRAGERIS, FEV TO 110           19.0         5         11         4-13-17         32         SF         SATEY CANVEL, POOLY CRABED, COARSE CANVEL SUBE SANDSTORE FRAGERIS, SIEL SAND                                                                                                                                                                                                                                                                                                                                                   | ELEVATION<br>(FEET )(162) | DEPTH | (FEET)<br>SAMPLE | SAMPLE | BLOWS (3)<br>OR<br>REC/RQD (4) | SPT N<br>VALUE (5)     | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                              |       |
| 5         9         6-6-6         12         CL         TOP 10 IN: SILT CLAY. SLICHTLY TO MODERATELY PLASTIC. STIFF, 15-201           5         10         19-17-12         29         CR         COMPLY CLAY. SLICE SAMBING ADD SALE FACTORY TO 1.5 IN           5         10         19-17-12         29         CR         Somp. Soilst, BRON. GENHLAR TO ADDRE DAVID UP CLAY. SLICE SAMBING TO ADDRE FACTORY           5         10         19-17-12         29         CR         Somp. Soilst, BRON. GENHLAR TO ADDRE SLICY HOUT COUNSE FACTORY           5         10         19-17-12         29         CR         Somp. Soilst, BRON. GENHLAR TO ADDRE SAMD, SOIL Y CALE TO ADDRES ADDR. SOIL Y TAKET ADDRES TO FINE GAVEL           5         11         4-13-17         32         ST         SOMPLASTIC FINES, BRON. CONTAINED LAYR SLICE SAMDSTONE FACTORY           5         12         99-10-8         18         ST         TOP TO IN SIGNAL FACTORY         ST         TOP TO THE CALE FACTORY           5         12         99-10-8         18         ST         TOP TO IN SIGNAL FACTORY         ST         TOP TO THE CALE FACTORY           6         13         5-55-26         83         CALE TANDY CALE FACTORY         CLASS TO THE SAMD, ST MORTAST TO THE SAMD, ST MORTAST TO THE CALE TAND TO THE SAMD, ST MORTAST TO THE SAMD, ST MORTAST TO THE SAMD, ST MORTAST TO THE SAMD, TANDY CALE TA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | <u></u>                   | 15    | 5                | - 8    | 1                              |                        |                     |                                                                                                                                                                                                                                                                                                                                                 | ┥     |
| 39.9       20       5       10       19-17-12       29       G7         39.9       20       5       11       4-13-17       22       G7       SAND, MOLT, SLED COASE TO THE SAND, NOSTLI COASE TO ATHORNEY, SLED SANDEY COASE TO ATHORNEY, SLED SANDEY COASE TO ATHORNEY, SLED SANDEY, COASE TO THE SAND, NOSTLI COASE TO ATHORNEY, SLED SANDEY, COASE TO THE SAND, NOSTLI COASE TO THE SAND, NOSTLI COASE TO THE SAND, NOSTLI COASE TO THE SAND, SLED SANDEY, SLED SAND                                                                                                                    |                           |       |                  | 9      | 6-6-6<br>(14")                 | 12                     | er<br>L             | TOP 10 IN: <u>SILTY CLAY</u> , SLIGHTLY TO MODERATELY PLASTIC, STIFF, 15-202<br>COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRACKENTS TO 1.5 IN<br>MAXINUM, FEW COAL FRAGMENTS, ORANGE, BROWN AND GRAY.<br>BOTTOM 4 IN: SILTY CLAY, MEDIUM SITEF, MODERATELY PLASTIC, TRACE FINE                                                            |       |
| 39.9       20       5       11       4-13-17       32       SF       TOP 10 THE SILTY SAMP, SIGTLY PLASTIC, 13-201 COARSE TO FINE GAUKE, MAXIMUM, ANGULAR, 13-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 13-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 13-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 13-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 15-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 15-201 SIGNET V PLASTIC TINES, TRACE COL, BROWN, ANGULAR, 150-400 COARSE TO FINE GAUKEL SIZED SANDSTOME FRAGMENTS, TO S-13, TOP SIGNET V PLASTIC PLASTIC FINES, TANK CORRECT OF THE GAUKEL SIZED SANDSTOME FRAGMENTS TO 1.5         5       13       9-59-26       85       OT         5       14       6-7-10       DT       DT       DT         6       14       6-7-10       DT       DT       DT       DT         7       14       6-7-10       DT       DT       DT       DT       DT         7       14       6-7-10       DT       DT       DT       DT       DT       DT         7       14       6-7-10       DT<                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                           |       |                  | 10     | 19-17-12<br>(7")               | 29                     | GP                  | SAND, HOIST, BROWN. (SIMILAR TO ABOVE BUT WITHOUT COARSE FRACTION).<br><u>SANDY CRAVEL</u> , POORLY GRADED, COARSE GRAVEL SIZED SANDSTONE FRAGMENTS,<br>MOSTLY 1.5 IN, 20-23% COARSE TO FINE SAND, MOSTLY COARSE TO NEDIUM,<br>5% NORPLASTIC FINES, BROWN. (CONTAINED LAYER OF SOFT CLAYEY SILL AT                                              |       |
| S         12         99-10-6         6         SAMEY CAN'EL, COASE TO FINE GANUE, SIZED SAMESTORE FLAGMENTS, -<br>10-402 COASE TO FINE SAME, STORE SAME, STORE CAN'E           S         12         99-10-6         18         5         10-402 COASE TO FINE SAME, STORE SAME, STORE TO FINE GANUE, SIZED SAMESTORE<br>FLAGMENTS TO 1.TN MAILENT, DODOI COASE TO FINE SAME, STORE<br>FLAGMENTS TO 1.TN MAILENT, DODOI COASE TO FINE SAME, SAMESTORE<br>FLAGMENTS TO 1.TN MAILENT, SAME CANVEL, COASE TO FINE SAME, SAMESTORE FLAGMENTS TO 1.S IN, SAMETLE COASE TO FINE<br>CANVEL SIZED SAMESTORE FLAGMENTS TO 1.S IN, SAMETLE COASE TO FINE<br>CANVEL SIZED SAMESTORE FLAGMENTS TO 1.S IN, SAMETLE COASE TO FINE<br>CANVEL SIZED SAMESTORE FLAGMENTS TO 1.S IN, SAMETLE COASE TO FINE<br>CANVEL SIZED SAMESTORE FLAGMENTS TO 1.S IN FLAGMENTS TO 1.S<br>IN MAILENT, SLICHTLY FLASTIC, SOFT, 10-135 COASE TO FINE<br>CANVEL SIZED SAMESTORE, SAME AND, CLAR FLAGMENTS TO 1.S<br>IN MAILENT, 70-235 COASE TO FINE SAME AND COASE TO FINE SAME<br>CANVEL SIZED SAMESTORE, SAME AND CAN FLAGMENTS IN SIZE SAMESTORE<br>IN MAILENT, 70-235 COASE TO FINE SAME AND COASE TO FINE SAME<br>SAMETLES SAMESTORE, SAME AND COASE TO FINE SAME, SAME CANVEL COASE TO FINE SAME<br>SAMETLES SAMESTORE, 200 COASE TO FINE SAME, SAME AND CANSE TO FINE SAME, SAME<br>SAMETLES SAMESTORE, 200 COASE TO FINE SAME, SAME STORE, SOME<br>FLAGMENTS TO 1.S IN MAILENT STOLES THAN STAMESTORE FLAGMENTS TO 1.5 IN<br>MAILENT SIZE SAMESTORE, 200 COASE TO FINE SAME, SAME STORE, SOME<br>FLAGMENTS TO 1.S IN, SAME STORE TO FINE SAME, SAME STORE SAME<br>SAMETLES SAMESTORE FLAGMENTS TO 1.S IN, FINE SAME, SAME STORE FLAGMENTS TO 1.S IN<br>MAILENT SIZE SAMESTORE TO FINE GANYEL, COASE TO FINE SAME, SAME STORE FLAGMENTS TO 1.S IN<br>MAILENT SIZE SAMESTORE TO FINE GANYEL, COASE TO FINE SAME STORE FLAGMENTS TO 1.S IN<br>MAILENT SIZE SAMESTORE TO FINE SAME STORE FLAGMENTS TO 1.S IN<br>MAILENT SAME STORE TO FINE SAME STORE FLAGMENTS TO 1.S IN<br>MAILENT SAME STORE F | 39.9                      | 20    | - 5              | 11     | 4-15-17<br>(14")               | 32                     | SP                  | TOP OF SAMPLE). (SEE NOTE 8).<br>TOP 10 IN: <u>SILTY SAND</u> , SLIGHTLY PLASTIC, 15-20% COARSE TO FINE GRAVEL<br>SIZED SANDSTONE AND SHALE FRAGMENTS, FEW TO 1 IN MAXIMUM, ANGULAR,<br>15-20% SLIGHTLY PLASTIC FINES, TRACE COAL, BROWN.                                                                                                       |       |
| 25 5 13 9-59-26 85 GP TINES, TAN. (SAMPLED COBBLE?).<br>LAYER D SANDS GARL AND SAMP CLATY SILT, SANDY GRAVEL, COARSE TO FINE CARVEL (10")<br>S 14 6-7-10 17 GP TINE SAND, LESS THAN 52 NOMENASTIC FINES, TAN (SAMPLED COBBLE).<br>CLAYER SILT, SILGHILY TO MODERATELY PLASTIC, SOFT, 10-15% COARSE TO FINE CARVEL, MOULAR BROWN.<br>TO P 6 IN: SANDY GRAVEL, COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5<br>IN MACHINEW, 20-25% COARSE TO FINE SAND, LIGHT GRAV.<br>BOTTOM E TH: GRAVELLY SILT, SILGHILY TASTIC, 12-15% COARSE TO FINE<br>GRAVEL, SIZED SANDSTONE, SALLE AND COAL FRAGMENTS, 1.5 IN FRAGMENT AT IFP.<br>GRAVEL SIZED SANDSTONE, SALLE AND COAL FRAGMENTS, 1.5 IN FRAGMENT AT IFP.<br>GRAVEL SIZED SANDSTONE, SALLE AND COAL FRAGMENTS, 1.5 IN FRAGMENT AT IFP.<br>GRAVEL SIZED SANDSTONE, SALLE AND COAL FRAGMENTS, 1.5 IN FRAGMENT AT IFP.<br>GRAVEL SIZED SANDSTONE, SALLE AND CARSE TO FINE GRAVEL, MOSTLY COARSE TO 1 IN<br>ANCTLAR TO SUBROUNDED SANDSTONE, 2.01 COARSE TO FINE SAND, 15-202<br>NONSLIGHTLY FLASTIC FINES, BROWN. 29.9 30 5 16 8-9-8 17 GP SIMILAR TO S-15. 310 5 17 7-14-14 28 GP SIMILAR TO S-15. 3111 12-14-11 25 GP SIMILAR TO S-15. 312 7-14-14 28 GP SIMILAR TO S-15. 313 18 12-14-11 25 GP SIMILAR TO S-15. 314 12-14-11 25 GP SIMILAR TO S-15. 315 18 12-14-11 25 GP SIMILAR TO S-15. 315 18 12-14-11 25 GP SIMILAR TO S-15. 316 19 10-9-13 12 22 SP SIMILAR TO S-15. 317 7-14-14 28 GP SIMILAR TO S-15. 318 19 10-9-13 12 22 SP SIMILAR TO S-10.5 IN, FINE SAND, TRAMENTAR TO RAMENTAR TO SAMPLER TO SAMPLER. 32 4-7-23 32 SP SIMILAR TO S-15. 32 41-7-23 32 SP SIMILAR TO S-15. N. FINE SAND, BORDEN, BUOMSTONE FRAGMENTS TO 1.5 IN, FINE SAND, LESS THAN 52 NONPLASTIC FINES, BONN. 319 10-9-13 12 22 100/5' 100 SP FAMOLENTS TO 1.5 IN, FINE SAND, MORELASTIC FINES, BONN. 32 20 4-7-23 32 SP SIMILAR TO SAMPLENCE TO FINE GRAVEL, COAL FRAGMENTS TO 1.5 IN, FINE                                |                           |       |                  | 12     | 99-10-8<br>(10")               | 18                     | GP<br>SP<br>GP      | BOTTOM 4: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS,<br>30-402 COARSE TO FINE SAND, 5% NONPLASTIC FINES, LIGHT GRAV.<br>TOP 5 IN: <u>SIMILAR TO 5-11</u> , TOP.<br>BOTTOM 5 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE<br>FRAGMENTS TO 1 IN MAXIMUM, 20-302 COARSE TO FINE SAND, 5% NONPLASTIC |       |
| $\begin{array}{c} \begin{array}{c} & & & & & & & \\ & & & & & & \\ & & & & $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                           | 25    | s<br>s           | -      | 9-59-26<br>(10")               | 85                     | GP                  | FINES, TAN. (SAMPLED COBBLE?).<br>LAYERED SANDY GRAVEL AND SANDY CLAYEY SILT, SANDY GRAVEL, COARSE TO FINE<br>GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN, ANOULAR, 30-40% COARSE TO<br>FINE SAND, LESS THAN 5% NONPLASTIC FINES, TAN (SAMPLED COBBLE).<br>CLAYER SULT SILENTLY TO MODERATELY PLASTIC SOFT. 10-15% COARSE TO FINE                | 1-1-1 |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                           |       | -<br>-<br>-      | 14     | 6-7-10<br>(14")                | 17                     | GP<br>ML            | GRAVEL, ANGULAR, BROWN.<br>TOP 6 IN: <u>SANDY GRAVEL</u> , COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5<br>IN MAXIMM, 20-253 COARSE TO FINE SAND, LIGHT GRAY.<br>BOTTOM 8 IN: <u>GRAVELLY SILT</u> , SLIGHTLY PLASTIC, 12-152 COARSE TO FINE                                                                                                  | 1 1 1 |
| $19.9 \begin{array}{c c c c c c c c c c c c c c c c c c c $                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                           |       |                  | 11     | 9-9-11<br>(12")                | 20                     | GP                  | GRAVEL SIZED SANDSTONE, SHALE AND COAL FRACHENTS, I.S IN FRAGMENT AT TIP,<br>LESS THAN 52 FINE SAND, TRACE IRON STAINS, BROWN.<br><u>SANDY GRAVEL</u> , WIDELY GRADED, COARSE TO FINE GRAVEL, MOSTLY COARSE TO 1 IN.<br>ANGULAR TO SUBROUNDED SANDSTONE, 202 COARSE TO FINE SAND, 15-202<br>NONSLIGHTLY PLASTIC FINES, BROWN.                   |       |
| 19.9 40 $19.9 40 $ $19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (12")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40 $ $17 7-14-14 (14")  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40  19.9 40 40  19.9 40 4$                                                                              | 729.9                     | 30    |                  | 16     | 8-9-8<br>(12")                 | 17                     | G₽                  | SIMILAR TO S-15.                                                                                                                                                                                                                                                                                                                                |       |
| 351812-14-11<br>(14")25GPSANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN<br>MAXIMUM, 15-20% COARSE TO FINE SAND, 7% NOMPLASTIC FINES, GRAY. SOME<br>POCKETS OF SITY FINE SAND, BROWN.<br>BLOWS/INCH: 2-1-2-3-2-2/1-3-3-3-2-2/2-2-2-2-1351910-9-13<br>(12")22SPGRAVELLY SAND, 15-25% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, FEW<br>SANDSTONE FRACMENTS TO 1.5 IN, FINE SAND, LESS THAN 5% NONPLASTIC FINES.<br>BROWN.<br>BLOWS/INCH: 2-1-2-1-2-2/1-2-1-2-2-1/2-1-2-4-2-25204-7-25<br>(14")32SP5204-7-25<br>(14")32SP62132-24-24<br>(14")6P31SANDSTONE FRACMENTS.<br>SANDSTONE FRAGMENTS.19.94052132-24-24<br>(14")6P652132-24-24<br>(14")6P6P522100/5"100<br>(17")SPGRAVELLY SAND, CARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS<br>FRAGMENTS.19.94052132-24-24<br>(14")6P652132-24-24<br>(14")6P522100/5"100<br>(17")SP522100/5"100<br>(17")52347-22-30<br>(17")SP52427-135135<br>13552427-135135<br>13552427-135135<br>13552427-135135<br>13552427-135135<br>13552427-135135<br>135 <td< td=""><td></td><td></td><td></td><td>1,</td><td>7-14-14<br/>(12")</td><td>28</td><td>GP</td><td>SIMILAR TO S-15, SAMPLED COBBLE, SOME FRAGMENTS ROUGHLY THE DIAMETER OF<br/>SAMPLER.<br/>Blows/Inch: 1-1-1-1-1-2/1-1-2-1-4-5/4-1-3-2-2-2</td><td></td></td<>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                           |       |                  | 1,     | 7-14-14<br>(12")               | 28                     | GP                  | SIMILAR TO S-15, SAMPLED COBBLE, SOME FRAGMENTS ROUGHLY THE DIAMETER OF<br>SAMPLER.<br>Blows/Inch: 1-1-1-1-1-2/1-1-2-1-4-5/4-1-3-2-2-2                                                                                                                                                                                                          |       |
| 19.94051910-9-13<br>(12")22SPGRAVELLY SAND, 15-25% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, FEW<br>SANDSTONE FRAGMENTS TO 1.5 IN, FINE SAND, LESS THAN 52 NONPLASTIC FINES.<br>BROWN.<br>BLOWS/INCH: 2-1-2-1-2-2/1-2-1-2-4-2-219.9405204-7-25<br>(14")32SPSPSAND, UNIFORM, FINE, LESS THAN 53 NONPLASTIC FINES, BROWN.<br>MIDDLE 10 IN: SAND, UNIFORM, MEDIUM TO FINE, TRACE FINE GRAVEL, COAL<br>FRAGMENTS, IRON STAINING, BROWN.<br>BOTTOM 1 IN: SANDSTONE FRAGMENTS.19.94052132-24-24<br>(14")6P6P505050506P522100/5"100SP50GRAVELLY SAND, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS<br>FRAGMENT AT TIP.<br>BLOWS/INCH: 3-7-3-5-4-10/6-3-4-5-3-3/3-4-3-4-5-5522100/5"100SPGRAVELLY SAMD, POORLY GRADED, 15-20% COARSE TO FINE GRAVEL TO 1 IN<br>MAXIMUM, ANCULAR, COARSE TO FINE SAND.<br>MAXIMUM, ANCULAR TO SHAD. MOSTLY MEDIUM TO FINE, 10-12%<br>NON TO SLICATULY PLASTIC FINES, MOIST, BROWN.<br>SANDSTLY MEDIUM TO FINE, 10-12%<br>NON TO SLICATULY PLASTIC FINES, MOIST, BROWN.52427-13513513552427-135135REFUSAL/NO RECOVERY.<br>END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                           | 35    | - s              | 18     | 12-14-11<br>(14")              | 25                     | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN<br>MAXIMUM, 15-202 COARSE TO FINE SAND, 72 NONPLASTIC FINES, GRAY. SOME<br>POCKETS OF SILTY FINE SAND, BROWN.<br>BLOWS/INCH: 2-1-2-3-2-2/1-3-3-2-2/2-2-2-2-2-1                                                                                                          |       |
| 19.9       40       5       20       4-7-25       32       SP       TOF 3 IN: SAND, UNIFORM, FINE, LESS THAN 52 NONPLASTIC FINES, BROWN.<br>MIDDLE 10 IN: SAND, UNIFORM, MEDIUM TO FINE, TRACE FINE GRAVEL, COAL<br>FRAGMENTS, IRON STAINING, BROWN.<br>BOTTOM 1 IN: SANDSTOME FRAGMENTS.         19.9       40       5       21       32-24-24       48       GP                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                           |       | - 5              | 19     | 10-9-13<br>(12")               | 22                     | SP                  | GRAVELLY SAND, 15-25% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, FEW<br>SANDSTONE FRACMENTS TO 1.5 IN, FINE SAND, LESS THAN 5% NONPLASTIC FINES,<br>BROWN.<br>BLOWS/INCH: 2-1-2-1-2-2/1-2-1-2-2-1/2-1-2-4-2-2                                                                                                                                   |       |
| 40       5       21       32-24-24       48       GP       SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS         19.9       40       5       21       32-24-24       48       GP       SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS         5       21       32-24-24       48       GP       SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS         5       22       100/5"       100       SP       GRAVELLY SAND, POORLY GRADED, 13-203       COARSE TO FINE GRAVEL TO 1 IN         5       22       100/5"       100       SP       GRAVELLY SAND, POORLY GRADED, 15-202       COARSE TO FINE GRAVEL TO 1 IN         6       50       23       47-22-30       52       SANDY GRAVEL, SIMILAR, COARSE TO FINES, MOIST, BROWN.         5       24       27-135       135       SANDY GRAVEL, SIMILAR TO S-15, AT 7 IN. FROM TOP - 2 IN. THICK SEAM OF         6       5       24       27-135       135         7       135       135       REFUSAL/NO RECOVERY.       END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                           |       |                  | 20     | 4-7-25<br>(14")                | 32                     | SP<br>SP<br>GP      | TOF 3 IN: <u>SAND</u> , UNIFORM, FINE, LESS THAN 52 NONPLASTIC FINES, BROWN.<br>MIDDLE 10 IN: <u>SAND</u> , UNIFORM, MEDIUM TO FINE, TRACE FINE GRAVEL, COAL<br>FRAGMENTS, IRON STAINING, BROWN.<br>BOTTOM 1 IN: <u>SANDSIONE FRAGMENTS</u> .                                                                                                   |       |
| 5       22       100/5"       100       SP       GRAVELLY SAND, POORLY GRADED, 15-202 COARSE TO FINE GRAVEL TO 1 IN         5       22       100/5"       5"       SP       GRAVELLY SAND, POORLY GRADED, 15-202 COARSE TO FINE GRAVEL TO 1 IN         5       23       47-22-30       52       SANDY GRAVEL, SIMILAR TO S-15, AT 7 IN. FROM TOP - 2 IN. THICK SEAM OF         5       24       27-135       135       REFUSAL/NO RECOVERY.       END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 719.9                     | 40    | 5                | 21     | 32-24-24<br>(14")              | 48                     | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS .<br>TO 1.5 IN MAXIMUM, 20-25% COARSE TO FINE SAND, LESS THAN 5% NONPLASTIC<br>FINES, TRACE COAL AND IRON STAINING, LIGHT GRAY AND BROWN. SANDSTONE<br>FRAGMENT AT TIP. BLOWS/INCH: 3-7-3-5-4-10/6-3-4-5-3-3/3-4-3-4-5-5                                                | 1.1.1 |
| FINE SAND, 15-202 NONPLASTIC FINES, MOIST, BROWN.<br>S 24 27-135 135 REFUSAL/NO RECOVERY. END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                           |       |                  | 22     | 100/5"<br>47-22-30             | <u>100</u><br>5"<br>52 | SP                  | GRAVELLY SAND, POORLY GRADED, 15-202 COARSE TO FINE GRAVEL TO 1 IN<br>MAXIMUM, ANGULAR, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 10-122<br>NON TO SLIGHTLY PLASTIC FINES, MOIST, BROWN.<br>SANDY GRAVEL, SIMILAR TO S-15, AT 7 IN. FROM TOP - 2 IN. THICK SEAM OF                                                                            |       |
| 145 F 1 1(0") 16" F E                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ·                         | 45    |                  | 24     | 27-135<br>(0")                 | 135<br>6"              |                     | FINE SAND, 15-202 NONPLASTIC FINES, MOIST, BROWN.<br>REFUSAL/NO RECOVERY. END OF BORING AT 45 FT. EL. 714.9                                                                                                                                                                                                                                     |       |

| STONE G WEBSTER EN<br>STE<br>Beaver Valley Power Stat | ion-Unit 2        | ·F.            |                                       |       | J.O. NO.<br>12241 |
|-------------------------------------------------------|-------------------|----------------|---------------------------------------|-------|-------------------|
| DATE6-9-82                                            | ORILLER Eger/     | Jarvis         |                                       | OR _J | . W. Mc           |
| COORDINATES N3812                                     | E6140             |                |                                       | 759.9 | Ft                |
| INSTALLED IN BORING EOS                               | 5-7 E             | LEV. TOP OF    | LEADS.                                | 761.  | 65 ft             |
| RIG & CREW TIME4 h                                    | ours              |                |                                       |       |                   |
|                                                       |                   |                |                                       | · · · |                   |
| DETAILED INSTALLATION DESCRIPTION :                   |                   |                |                                       |       |                   |
| Hole cleaned to 45.0 ft.                              | •                 | <u></u> П      | · · · · · · · · · · · · · · · · · · · |       | •<br>•<br>•       |
| Porous stone SOILTEST pi                              | lezometer         | 3'             |                                       |       | 21'               |
| 4 ft above tip.                                       | 1061              |                |                                       |       | mm                |
| Bentonite seal between 3                              | 9.0               |                |                                       |       |                   |
| and 36 ft-11 in.<br>Sand to approximately gr          | ound              |                |                                       |       |                   |
| surface.                                              | ace               |                |                                       |       |                   |
| data hihe Broared in hi                               |                   |                |                                       |       |                   |
|                                                       |                   | S              | and                                   |       | · · ·             |
|                                                       |                   |                |                                       |       |                   |
| AND STAND PIPE ASSEMB                                 | ER TIP            |                |                                       |       | '11"              |
| 2 ft section Norton porc                              | ous tube.         | Be             | ntonite                               | 20    | i .               |
| Approximately 43 ft-3/4 riser pipe with centerin      | in PVC<br>spider. |                |                                       | "     |                   |
|                                                       | -0 -r ·           |                |                                       |       |                   |
| · ·                                                   |                   |                | -=-                                   |       | Ŧ                 |
| 05000000000 AT                                        | -                 | · [            | F                                     |       | 21                |
| ELEVATION OF SOIL AT                                  |                   |                | E                                     |       | Ī.                |
| Sandy gravel - rock at 4                              | 45.0 ft.          | s              | and                                   |       | Ţ                 |
|                                                       | · .               |                |                                       |       | Í                 |
|                                                       |                   | <b>ن</b> ـــــ |                                       |       |                   |
|                                                       |                   |                |                                       |       |                   |
|                                                       |                   |                |                                       |       |                   |

## 2.5E-54

|                     |                             | EAVER        | <u>VAL</u>  | LEY POWER          | STAT           | (ON-UN      | IT 2                                  | J.O. NO. 12241                        | BORING NO.                            |
|---------------------|-----------------------------|--------------|-------------|--------------------|----------------|-------------|---------------------------------------|---------------------------------------|---------------------------------------|
| 1                   | COORDI                      | NATE         | :s .        | N3814.6            |                |             | GROUND E                              | LEV. (1)                              |                                       |
|                     | INCLINA                     | TION         |             | VERTICAL           |                | . B(        | EARINGI                               | NSPECTOR Med                          | 07                                    |
|                     | DATE :                      | STAF         | RT / I      | FINISH             | 6/3/0          | 32          | / CONTRACT                            | OR / DRILLEREGER/                     | JARVIS                                |
| *                   | STATIC                      | GRO          | UND         | WATER              | DEPT           | HZD         | ATE /                                 | DRILL RIG TYPE                        | ME 45                                 |
| { ·                 | DEPTH                       | TO           | BED         | ROCK _             | NA             |             | (FT) TOTAL DEP                        | TH DRILLED2                           | 4.5 (P                                |
|                     | METHO                       | )\$:<br>     |             | - 01               | 3.1/9          | TN 0        |                                       | A TH O B CRITE CROOM                  |                                       |
|                     | D                           |              | NG          | 301L .             | 2 IN C         | .D. S       | PLIT SPOON AND 3 IN O.D. SHET         | RY TURE                               | N TO CLEAN OUT                        |
| <b>.</b>            | э.<br>М                     |              | ing i       | BOCK               |                |             |                                       |                                       |                                       |
|                     | SPECIA                      | LTE          | STI         | NG OR I            | NSTR           | UMEN        | TATION 2 FT NORTON POROU              | S PIEZOMETER INSTALLED                | WITH TIP AT EL. 738                   |
|                     |                             |              |             |                    |                |             | · · · · · · · · · · · · · · · · · · · |                                       |                                       |
|                     | COMME                       | NTS .        | DR          | ILLED 5            | FT NOR         | THWES       | OF EOS-7                              |                                       | <u></u>                               |
|                     | -                           |              |             |                    |                |             |                                       | ·····                                 |                                       |
|                     |                             |              |             |                    |                |             |                                       |                                       | <u></u>                               |
| N N                 |                             | <u> </u>     |             | Î Î                | È la           | (9)         |                                       |                                       |                                       |
| ATIC<br>ET)         | EPT)<br>EE1                 | I H          | Ę           | SM SM              |                | 12 S        | SAME                                  | LE DESCRIPTION                        |                                       |
| 5                   | ŌĿ                          | N. F         | 8           |                    | 5   <b>b</b> 3 | 12 2        |                                       |                                       |                                       |
|                     | <u> </u>                    | <u> </u>     | <u> </u>    | <u> </u>           |                | <u> </u>    | L                                     | · · · · · · · · · · · · · · · · · · · |                                       |
| 760 (               | <u> </u>                    | <b>T</b>     | <u> </u>    |                    | <b></b>        | <del></del> |                                       |                                       |                                       |
| / 39.0              |                             | 1            |             |                    |                |             |                                       |                                       |                                       |
|                     | 1 :                         | İ            | 1           |                    |                |             | NO SAMPLES TO 7 FT.                   |                                       |                                       |
|                     |                             | ]            |             |                    |                |             |                                       |                                       |                                       |
| а.<br>С             |                             | 1            |             |                    |                |             |                                       |                                       |                                       |
|                     | -                           |              |             |                    |                | ۲ I         |                                       |                                       |                                       |
|                     | 5 -                         |              |             |                    |                |             |                                       |                                       |                                       |
|                     |                             | 1            | İ           |                    |                |             |                                       |                                       |                                       |
|                     | {                           |              |             |                    | 1              | ·           |                                       |                                       |                                       |
| 5                   |                             |              | Ι.          | (15%)              |                |             |                                       |                                       |                                       |
|                     | :                           | 1            | <b>.</b>    |                    |                |             |                                       |                                       |                                       |
|                     | ·                           | <u></u>      | 1           | · .                | 1              |             |                                       | 1                                     |                                       |
| 749.6               | 10 -                        | US           | 2           | (25.5")            |                |             |                                       |                                       |                                       |
|                     | :                           | 1            |             |                    |                |             |                                       |                                       |                                       |
|                     |                             | s            | 1           | 10-7-6             | 13             | CL/         | SANDY CLAY-SANDY SILT, SLIC           | CHTLY PLASTIC, STIFF, 2               | 0-25% COARSE TO FINE                  |
|                     | -                           | <b> </b>     |             |                    |                |             | SAND, 10% FINE GRAVEL TO 1            | 4 IN, MOIST, BROWN.                   |                                       |
|                     |                             | s            | 2           | 5-7-7              | 14             | CL          | SANDY CLAY, SLIGHTLY TO MOT           | ERATELY PLACTIC STITE                 | 107 FINE CRAVET                       |
|                     | -                           | 1            |             | (10")              |                |             | OCCASIONAL COARSE GRAVEL TO<br>BROWN. | D 1 IN, 20% COARSE TO F               | INE SAND, MOIST,                      |
|                     | 15                          |              |             | L                  |                | ليل         |                                       | r                                     |                                       |
| 1                   |                             | IS M         | EAN<br>WATE | SEA LEVEL          | VEL            |             | UNDISTURBED SAMPLES<br>US-SHELBY TUBE | BORING                                | LOG                                   |
| 3.                  | BLOWS                       | REQU         |             |                    | VE             |             | UO-OSTERBERG                          |                                       |                                       |
| ۳Ì                  | DISTAN                      | CE SI        | IOWN        | USING              | 477<br>4       |             |                                       | BEAVER VALLEY PO                      | WER STATION UN                        |
| 2 4                 | 14018. H                    | HES          | OF 1        | LLING 30<br>SAMPLE | <b>.</b>       |             |                                       | DUQUESNE L                            | IGHT COMPANY                          |
| . [                 | RECOVE<br>STD. P            | RY.<br>Éneti | RATI        | ON RESIS           | TANCE          | :           |                                       | SHIPPINGPORT                          | r, PENNSYLVANIA                       |
| <b>)</b> 5.         | BI OWO                      | FT.          |             |                    |                |             |                                       |                                       | · · · · · · · · · · · · · · · · · · · |
|                     | UNIFIE                      | SO           | L CL        | ASSIFICA           | TION           |             |                                       | I A STONE & WEDG                      | TFB FNC COBD                          |
| 5.<br>DN 6.<br>9017 | UNIFIEL<br>SYSTEN<br>SAMPLI | D SOI        | L CL        | ASSIFICA           | TION           |             |                                       | SKETCH No. 12                         | TER ENG. CORP.<br>241-GSK-250A        |

| SI           | TE_85           | AVER               | VALLE            | Y POWER S                      | TATIO              | N-UNI               | BORING NO. <u>E05-7A</u><br>SHEET <u>2</u> OF <u>2</u><br>1 2, SHIPPINGPORT, PA. J.O. NO. <u>12241.00</u>                                                                                |
|--------------|-----------------|--------------------|------------------|--------------------------------|--------------------|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| (FEET )(162) | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE<br>NUMBER | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                       |
|              | 15              | s                  | 3                | 5-6-9                          |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  | (13")                          | 15                 | CL                  | <u>SIMILAR TO S-2</u> , 20-30% COARSE TO FINE GRAVEL TO 1 IN.                                                                                                                            |
|              |                 | s                  | 4                | 4-6-6<br>(16")                 | .12                | CL<br>ML            | TOP 13 IN: <u>SILTY CLAY</u> , NODERATELY PLASTIC, MEDIUM STIFF, MOTTLED GRAY<br>AND BROWN.<br>BOTTOM 3 IN: <u>SILT</u> , LOOSE, TRACE FINE SAND, WET, BROWN.                            |
| .6           | 20              | s                  | 5                | 11-15-14<br>(13")              | 29                 | GP                  | SANDY GRAVEL, WEATHERED SANDSTONE FRAGMENTS TO 1 IN MAXIMUM, 25-302<br>COARSE TO FINE SAND, MOSTLY MEDIUM FINE, 5-102 NONFLASTIC FINES, TRACE<br>COAL AND IRON STAINING, BROIN AND GRAY. |
| •            |                 | s                  | 6                | 20-20-8<br>(18")               | 28                 | GP_                 | <u>GRAVEL</u> , COARSE GRAVEL SIZED SANDSTONE FRAGMENTS TO 1.5 IN MAXIMUM,<br>LIGHT GRAY. CONTAINS POCKETS OF SANDY SILT, 10-15% FINE SAND, VERY<br>MOIST, BROWN.                        |
| . 1          | -               | s                  | 7                | 8-11-18<br>(13")               | -29                | GP                  | <u>SIMILAR TO 5-6</u> .                                                                                                                                                                  |
|              |                 |                    |                  |                                |                    |                     | BOTTOM OF BORING AT 24.5 FT<br>Elevation 735.1 Ft                                                                                                                                        |
|              | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     | -                                                                                                                                                                                        |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  | · .                            |                    |                     |                                                                                                                                                                                          |
|              | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              | -               |                    |                  |                                |                    |                     |                                                                                                                                                                                          |
|              |                 | 1                  |                  |                                |                    |                     |                                                                                                                                                                                          |
|              | _               |                    |                  |                                |                    |                     | ·                                                                                                                                                                                        |
| 'E :         | FOR BO          | RNG                | SUMM<br>SEE      | ARY AND SHEET L                |                    | STO                 | TCH NG. 12241-GSK-2508 DATE BORING NO. SHEET                                                                                                                                             |



.

| :            | SITE BEAVE              | R VALL          | EY POWER         | STATIC        | N-UNI      | <u>T 2</u>                                                         | J.O. NO. <u>12241</u>                          | BORING NO. EOS                      |
|--------------|-------------------------|-----------------|------------------|---------------|------------|--------------------------------------------------------------------|------------------------------------------------|-------------------------------------|
| I            | COORDINAT               | TES .           | N3944            |               |            | E6185 GROUND E                                                     | LEV. (1) 732.7                                 | SHEETOF                             |
| I            | INCLINATIO              | M               | ERTICAL          |               | . 88       | ARINGNA                                                            | INSPECTOR                                      | HCCOY                               |
| I            | DATE : ST               | ART /           | FINISH           | <u>5-19-8</u> | 32         | / 5/20/82 CONTRACT                                                 | OR / DRILLER _EGE                              | R DRILLING/JARVIS                   |
|              | STATIC GF               | ROUND           | WATER            | DEPT          | 'H / D     | AT ERECORDED (FT) /                                                | DRILL RIG TYPE                                 | CME 45                              |
| ł            | <b>DEPTH TO</b>         | ) BED           | ROCK _           | 52.(          | )          | (FT) TOTAL DEP                                                     | TH DRILLED                                     | 52.0 (PT)                           |
| ļ            | METHODS :               |                 |                  |               |            |                                                                    | · · · · · · · · · · · · · · · · · · ·          |                                     |
|              | ORILI                   | LING            | SOIL             | 2.0 11        | IN RO      | SPLIT SPOON                                                        | NG, DRILLING MUD                               | <u> </u>                            |
|              | SAMF                    | PLING           | SOL              |               |            |                                                                    |                                                |                                     |
| ,            | SPECIAL 1               | LING            | NG OR            | INSTR         |            | TATION NONE                                                        |                                                |                                     |
|              |                         |                 |                  |               | -          |                                                                    |                                                |                                     |
|              | COMMENTS                | 5LO             | ST DRILL         | ING FLU       | ID AT      | 35.0 AND 40.0 FT                                                   |                                                |                                     |
|              |                         |                 |                  |               |            |                                                                    | · · · · · · · · · · · · · · · · · · ·          |                                     |
| -            |                         |                 |                  |               |            |                                                                    |                                                | · .                                 |
| - 9          |                         | ខ               |                  | e .           | 6          |                                                                    |                                                |                                     |
|              | E                       |                 |                  |               | 57         |                                                                    |                                                |                                     |
| FEE          | NAME OF                 |                 | AND              |               | 83         | SAM                                                                | PLE DESCRIPTION                                | JN                                  |
| ŭ            |                         |                 |                  | ¥ [           | <b>_</b> o | · · · · · · · · · · · · · · · · · · ·                              |                                                |                                     |
|              |                         |                 |                  |               |            |                                                                    |                                                | · · ·                               |
| 732.7        | 0_5                     | 1               | 2-5-3            | 8             | ML/        | SANDY SILT, DENSE, SLIGHTLY                                        | MOIST, FEW SANDSTON                            | E FRAGMENTS AND ROOTS,              |
|              |                         |                 | (14*)            |               |            | GRADING TO SILIT SAND, TRAC                                        | CE FINE GRAVEL, 30-40                          | IZ NONPLASTIC FINES, BROW           |
|              |                         |                 |                  |               |            |                                                                    |                                                |                                     |
|              | s                       | 2               | 4-4-6            | 10            | MIL.       | SILT, NONPLASTIC TO SLIGHT                                         | TLY PLASTIC, 0-5% FIN                          | E SAND, TRACE ORGANICS,             |
|              |                         |                 | (18")            |               |            | FEW SMALL SAND SEAMS, WET,                                         | BROWN.                                         |                                     |
|              |                         |                 |                  |               |            |                                                                    |                                                |                                     |
|              | 5 _ 5                   | 3               | 3-4-5            | 9             | ML.        | TOP 13 IN: SIMILAR TO ABOVE                                        | <u>E</u> .                                     |                                     |
|              |                         |                 | (16")            |               | SP         | BOTTOM 3 IN: <u>SAND</u> , FINE, FE<br>FRAGMENTS TO 0.5 IN, 0-52 N | EW FINE GRAVEL AND WE<br>Nonplastic fines, bro | LATHERED SANDSTONE                  |
|              |                         |                 |                  |               |            |                                                                    |                                                |                                     |
|              |                         | - 4             | 4-4-4            | 8             | SP         | TOP 13 IN: SAND, COARSE TO                                         | FINE, MOSTLY COARSE                            | TO MEDIUM, 2-5% FINE                |
|              |                         |                 | (16")            |               | м          | GRAVEL, 0-5% NONPLASTIC FIN<br>BOTTOM 3 IN: SILT, NONPLAST         | NES, BROWN.<br>FIC TO SLIGHTLY PLAST           | IC, BROWN.                          |
|              |                         | 1               |                  | 1             |            | BLOWS/INCH: 1-1-1/2-1/2//1-                                        | -1/2-1/2-1//1/2-1/2-1                          | -1                                  |
| 722.7        | 10-5                    | - 5             | 3-3-4            | 7             | SP         | TOP 4 IN: SILTY SAND, FINE.                                        | , TRACE COARSE-MEDIUM                          | SAND, 15-202 NONPLASTIC             |
|              | 1 7                     |                 | (14")            |               | SP         | FINES, MOIST, BROWN.<br>BOTTOM 10 IN: SAND, COARSE                 | TO FINE, MOSTLY COAR                           | SE TO MEDIUM, TRACE FINE            |
|              | ±                       | -               | ·                |               |            | GRAVEL, 5% NONPLASTIC FINES<br>BLOWS/INCH: 1-1/2-1/3//1/2-         | S, MOIST, BROWN.<br>-1/2-1/2//1-1/2-1/2-1      | ,                                   |
|              | │ – .                   |                 | 6-4-1            | ,             | ~          | GRAVELLY SAND DELL CRADET                                          | 0. 20-302 COAPCE TO E                          | INE GRAVEL MOSTLY MENTE             |
|              | ו ז'                    | ľ               | (18")            |               | <b> </b> " | TO FINE, SUBANGULAR TO ROUN                                        | NDED, COARSE TO FINE                           | SAND, TRACE NONPLASTIC              |
|              | +-                      | -               |                  |               |            | BLOWS/INCH: 1-1-1-1-1//4/                                          | /1/2-1/2-1/2                                   |                                     |
| . <b>.</b> . | 15                      |                 | I                |               |            |                                                                    | · · ·                                          | ·····                               |
| 1.           | DATUM IS                | MEAN            | SEA LE           | VEL           |            | UNDISTURBED SAMPLES                                                | 809                                            |                                     |
| 3            | BLOWS RE                | OUIRE           | TODR             | IVE           |            | UO-OSTERBERG                                                       |                                                |                                     |
| ES           | 2"O.D. SAMI<br>DISTANCE | PLE SF<br>SHOWN | POON 6"<br>USING | OR .          |            |                                                                    | BEAVER VALLEY                                  | POWER STATION UNI                   |
| 54           | 4015. HAM               | NER FÅ          | LLING 3          | 0".           |            |                                                                    | DUQUESNE                                       | LIGHT COMPANY                       |
| <u>׀</u> ר   | RECOVERY                | TRAT            |                  | STANCE        | •          |                                                                    | SHIPPINGPO                                     | RT, PENNSYLVANIA                    |
| <u>ْ او</u>  | BLOWS/FT                |                 | A00151A          | 47104         | •          |                                                                    |                                                |                                     |
| Ш<br>Ш<br>С  | SYSTEM                  |                 | ASSIFIC          | A I IUN       |            |                                                                    | STONE & WI                                     | EBSTER ENG. CORP.<br>12241-GSK-251A |
| <u>، ا</u> ۳ | SAMPLE T<br>S-SPLIT &   | YPE:<br>BARREL  | SAMPL            | ε             |            | •                                                                  | APPROVED DAT                                   | E BORNS NO. SHEET                   |
|              | •                       |                 |                  |               |            |                                                                    | DDH 9/1/0                                      | - 1 POS-9   1 OF                    |

| s                       | ITE.  | BEAVI            | R VALL    | EY POWER S                     | TATIO                                                                                     | N-UNI                                                                                                                                                                                                                                                           | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                                                     |
|-------------------------|-------|------------------|-----------|--------------------------------|-------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162 | DEPTH | (FEET)<br>SAMPLE | SAMPLE    | BLOWS (3)<br>OR<br>REC/ROD (4) | SPT N<br>VALUE (5)                                                                        | GROUP<br>SYMBOL (6)                                                                                                                                                                                                                                             | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                           |
|                         |       |                  |           |                                |                                                                                           |                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                              |
|                         | 15    |                  | 3   7<br> | 4-3-3<br>(18")                 | 6.                                                                                        | SM<br>SP                                                                                                                                                                                                                                                        | TOP 9 IN; <u>SILTY SAND</u> , 10-152 COARSE TO FINE GRAVEL, SUBANGULAR TO ROUNDED,<br>COARSE TO FINE SAND, MOSTLY FINE, 10-152 NONPLASTIC FINES.<br>BOTTOM 9 IN: <u>SAND</u> , FINE, 2-63 FINE GRAVEL, 0-52 NONPLASTIC FINES, TRACE<br>COARSE SAND SIZED COAL FRAGMENTS, BROWN.<br>BLOWS/INCH: 1-1/2-1/2-1/2/1/2/1/2-1/2-1/2 |
|                         |       |                  | 5 8       | 9-11-10<br>(18")               | 21                                                                                        | ଙ୍କ                                                                                                                                                                                                                                                             | SANDY GRAVEL, WIDELY GRADED, SUBANGULAR TO ANGULAR WEATHERED SANDSTONE<br>FRAGMENTS TO 1 IN MAXIMUM, 25-35% COARSE TO FINE SAND, 5-10% NONPLASTIC<br>FINES, TRACE COAL, FEW IRON STAINS, BROWN.                                                                                                                              |
| 712.7                   | 20    |                  | 5 9       | 4-3-2<br>(16")                 | 5                                                                                         | SP                                                                                                                                                                                                                                                              | <u>GRAVELLY SAND</u> , 20-307 COARSE TO FINE GRAVEL SIZED SANDSTONE FRAGMENTS,<br>MAXIMUM SIZE 1 IN, ANGULAR TO ROUNDED, COARSE TO FINE SAND, MOSTLY FINE,<br>5-107 SLIGHTLY PLASTIC FINES, TRACE COAL FRAGMENTS, IRON STAINS AT BOTTOM,<br>WET AT BOTTOM, BROWN.                                                            |
|                         |       |                  | 5 10      | 9-10-13<br>(16")               | 23                                                                                        | GW                                                                                                                                                                                                                                                              | SANDY CRAVEL, WEATHERED SANDSTONE AND SHALE FRAGMENTS TO 1 IN MAXIMUM,<br>ANGULAR, 15-252 COARSE TO FINE SAND, 2-52 NONPLASTIC FINES, IRON STAINS,<br>MOIST, BROWN AND GRAY.<br>BLOWS/INCH: 2-2-1-2-1-1/1-1-2-2-2-2/3-2-2-2-2                                                                                                |
|                         | 25    |                  | 5 11      | 27-17-13<br>(12")              | 30                                                                                        | GP                                                                                                                                                                                                                                                              | <u>GRAVEL</u> , WEATHERED SANDSTONE AND LIMESTONE (?) FRACMENTS TO 1-1/2 IN,<br>ANGULAR TO SUBANGULAR, SOME IRON STAINING, 5-10% NONPLASTIC FINES. TRACE<br>SHALE FRAGMENTS, DRY, BROWN.<br>BLOWS/INCH: 2-3-5-4-5-3/3-3-1-2-2/3/2-2-2-2-1                                                                                    |
|                         |       | -                | 12        | 5-9-19<br>.(13")               | 28                                                                                        | SP                                                                                                                                                                                                                                                              | TOP 3 IN: SAND, UNIFORM, FINE, TRACE FINE GRAVEL, TRACE NONPLASTIC FINES,<br>BROWN.                                                                                                                                                                                                                                          |
|                         |       |                  | +         |                                | K                                                                                         | SP                                                                                                                                                                                                                                                              | RIDULE 1 IN: <u>SAME AN ADDEE</u> , DANA BROWN.<br>BOTTOM 9 IN: <u>SAME</u> UNIFORM, FIME, TRACE FINE GRAVEL, ROCK FRAGMENT AT<br>BOTTOM, LIGHT BROWN.                                                                                                                                                                       |
| 702.7                   | 30    |                  | 13        | 11-15-15<br>(13")              | 30                                                                                        | SM<br>SW                                                                                                                                                                                                                                                        | TOP 4 IN: <u>SILTY SAND</u> , 10-15% FINE GRAVEL TO 1/2 IN, ANGULAR TO SUBROUNDED.<br>COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 25-30% NONPLASTIC FINES, DRY,<br>BROWN, DENSE AND HARD IN NATURAL STATE, PARTICLES APPEAR WATER-BORKE.<br>BOTTOM 9 IN: <u>SAND</u> , WELL GRADED, COARSE TO FINE, 0-5% FINE GRAVEL, 0-5%   |
|                         |       |                  | 14        | 9-7-14<br>(18")                | -14 22 SM TOP 15 IN: SIMILAR TO 5-13, TOP 4 I<br>"") SP BOTTOM 3 IN: SAND, COARSE TO FINE | NONFLASTIC FIRES, FEW SANDSTONE FRAGMENTS TO 1 IN MAXIMUM, IRON STAINING,<br>BROWN. BLOWS/INCH: 2-2-2-1-2-2/3-2-3-3-2-2/3-1-3-3-3-2<br>TOP 15 IN: <u>SIMILAR TO S-13</u> , TOP 4 IN.<br>BOTTOM 3: IN: SAND. COARSE TO FINE. MOSTLY COARSE TO MEDIUM. TRACE FINE |                                                                                                                                                                                                                                                                                                                              |
|                         |       | +                |           |                                | $\vdash$                                                                                  |                                                                                                                                                                                                                                                                 | GRAVEL, MOIST, BRCWN.<br>BLOWS/INCH: 1-2-2-1-2-1/1-1-1-1-2/1-2-2-3-3-3                                                                                                                                                                                                                                                       |
|                         | 35    |                  | - 15      | 11-12-15<br>(18")              | 27                                                                                        | SM<br>GP                                                                                                                                                                                                                                                        | TOP 8 IN: <u>SIMILAR TO S-13</u> , TOP 4 IN<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, 1 IN MAXIMUM, ANGULAR<br>TO ROUNDED, 15-202 COARSE TO FINE SAND, MOSTLY FINE, 10-15% NONPLASTIC<br>FINES, BROWN.<br>BIOSIGNUM, 2-2-2-2-2/2-2-2-2/2-2-3-2-2-2                                                       |
|                         |       |                  | 16        | 16-20-25                       | 45                                                                                        | GH                                                                                                                                                                                                                                                              | SILTY CRAVEL, COARSE TO FINE GRAVEL, FEW TO 1 IN MAXIMUM, ANGULAR TO<br>ROUNDED, 1 IN SANDSTONE FRACMENTS AT BOTTOM, 10-15% COARSE TO FINE SAND,<br>NOSTLY FINE, 15-20% NONPLASTIC FINES, DRY, BROWN (SIMILAR TO S-13, TOP<br>4 (N), BLOWS/INCH: 2-2-3-3-3/4-3-2-3-3-5/4-3-2-2-7                                             |
| 692.7                   | 40    |                  | 5 17      | 5-5-8<br>(8")                  | 13                                                                                        | GW                                                                                                                                                                                                                                                              | SANDY GRAVEL, COARSE TO FINE GRAVEL, FEW TO 1 IN MAXIMUM, ANGULAR TO<br>ROUNDED, 15-20% COARSE TO FINE SAND, 0-5% SLIGHTLY PLASTIC FINES, MOIST,<br>BROWN.<br>BLOWS/INCH: 1/2-1-1-1-1//1-1-1/2-1-1//1-1-2-2-1-1                                                                                                              |
|                         |       |                  | 18        | 12-19-22<br>(16")              | 41                                                                                        | GP                                                                                                                                                                                                                                                              | SANDY GRAVEL, MOSTLY LARGE, WEATHERED SANDSTONE AND SHALE FRAGMENTS TO<br>1-1/2 IN, SOME SHALE FRAGMENTS, 15-202 COARSE TO FINE SAND, 2-52<br>NONPLASTIC FINES, MOIST, BROWN.<br>NICHS/INVE: 3.2-2.2-2.2-1/2.2-2-4.4-5/7-3-3-4-3-2                                                                                           |

|                           |            |       |           |                    |        |              |            |                   | •          |           |                     | BORING NO. CUSTY                                                                                                                                                                                                                                                                                      |      |
|---------------------------|------------|-------|-----------|--------------------|--------|--------------|------------|-------------------|------------|-----------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------|
|                           |            |       |           |                    |        |              |            |                   | CT 61      |           | - 11 <b>N T</b>     | T 2 SHIPPINGPORT, PA                                                                                                                                                                                                                                                                                  |      |
|                           | SIT        | Ε_    | BE.       | AVER               | VAL    |              |            |                   | 314        |           |                     | J.O. NO                                                                                                                                                                                                                                                                                               |      |
| ELEVATION<br>(FEET )(162) |            | DEPTH | (FEET)    | SAMPLE<br>TVDE (7) | SAMPLE | NUMBER       | BLOWS (3)  | OR<br>REC/ROD (4) | N 1.95     | VALUE (S) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                    |      |
|                           | 4          | 5     |           | s                  | 19     | Т            | 13-        | 9-19              | T          | 28        | GP                  | TOP 4 IN: CRAVEL, SANDSTONE FRACMENTS TO 1-1/2 IN, SOME SHALE FRACMENTS                                                                                                                                                                                                                               | • -  |
|                           |            |       | 4         |                    |        | ┦            | (11        | .")               | ╀          |           | SP<br>CP            | S-104 COARSE TO FIRE SARD, S-104 SLIGHTLIF FLASTIC FIRES, URL AND DEALS<br>MIDDLE SIN <u>SAND</u> , UNIFORM, FIRE, O-52 NORFLASTIC FIRES, WET, BROWN.<br>BOTTOM 4 IN: <u>SANDY GRAVEL</u> , COARSE TO FIRE GRAVEL SIZED SANDSTONE AND SHA<br>DEALEMENTS $\leq 100$ SITCHITY DIASTIC FIRES LIFT. BROWN | Le - |
|                           |            |       | _         | s                  | 20     |              | 45/        | 6"                | 45         | 16        | GP                  | FRAMERIS J-IUA SLIVATILI ELASILE FINES, WEI, BROWN.<br>BLOWS TINCH: 4-2-2-2-1-2-2-1-1-2/2-5-2-5-2<br>S <u>ANDY GRAVEL</u> , COARSE TO FINE GRAVEL SIZED SANDSTONE AND SHALE FRAGMENTS.                                                                                                                | -    |
|                           |            |       | -         | •                  | 1,     | ╀            | (5"        | ')<br>.27-2       | +          |           |                     | ROUNDED TO ANGULAR, LARGE SANDSTONE FRAGMENT AT BOTTOM, 10-15% COARSE TO<br>FINE SAND, 2-5% SLIGHTLY PLASTIC FINES, MOIST, ORANGE, BLACK AND BROWN.                                                                                                                                                   | -    |
|                           |            |       | -         |                    | -      | $\downarrow$ | (10        | ")                | 1          | _         | GP                  | SANDY CRAVEL, SIMILAR TO ABOVE, MAXIMUM PARTICLE SIZE 1-1/2 IN, 5-7%                                                                                                                                                                                                                                  | -    |
| 682.                      | ۲ <b> </b> | 90    | Η         | · 5                | 22     |              | 15-<br>(6" | 98/3<br>')        | " 98       | /3        |                     | BLOWS/INCH: 3-4-4-7-7-5/6-3-5-4-3-2/4-3-2-3-3-5                                                                                                                                                                                                                                                       |      |
|                           |            |       |           |                    | 23     | T            | 30/        | 0"                | $\uparrow$ | $\neg$    | -                   | WEATHERED SHALE, LU-13% FIRE SAND, 10-13% SLIGHTLE PLASITE TO HEDRON<br>PLASTIC FINES, ORANGE, BLACK, GRAY BROWN.<br>REFUSAL                                                                                                                                                                          |      |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     | BOTTOM OF BORING AT 52.0 FT                                                                                                                                                                                                                                                                           |      |
|                           |            |       |           |                    |        | ł            |            |                   |            |           |                     | ELEVATION 080.7 FI                                                                                                                                                                                                                                                                                    | _    |
|                           |            |       |           |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       |      |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           | ł          |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | _         |                    |        | 1            |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       |           |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       |      |
|                           |            |       | _         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | i         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       |           |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | 1         |                    | :      |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       |      |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | 1         |                    |        |              |            | •                 |            |           |                     |                                                                                                                                                                                                                                                                                                       | _    |
|                           |            |       | 7         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | _    |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       |      |
|                           |            |       | -         |                    |        |              |            |                   | 1          |           | ar i                |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | -         |                    | 1      |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | -         |                    |        |              |            |                   |            |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
|                           |            |       | 1         |                    |        |              |            |                   | ╞          |           |                     |                                                                                                                                                                                                                                                                                                       | -    |
| OTE                       | : FO       | R     | 801<br>10 | RING<br>NFO        | SUM    | MAI<br>S     | RY<br>Hef  | AND               | 4          | 4         | STO                 | NE & WEBSTER ENG. CORP. APPROVED OATE BORING NO. SHEET<br>TCH NO. 12241-058-2510 DDA 9/1/8- E0S-9 3 OF 3                                                                                                                                                                                              |      |

|                        | OOKD                               | INATE                         |                | N4097                     | 7.3        |              | . <u> </u>       | (6137.4 GROUND ELEV. (1) 720.7 STEET                                                  |
|------------------------|------------------------------------|-------------------------------|----------------|---------------------------|------------|--------------|------------------|---------------------------------------------------------------------------------------|
| INCLINATION VERTICAL B |                                    |                               |                |                           |            |              | . <b>BE</b><br>2 | ARING NA INSPECTOR J. W. MCCOY                                                        |
| 0                      | ATE :                              | STA                           | RT / I         |                           |            | 10/0         | <u> </u>         | ATTRECORDENTLY CONTRACTOR / DRILLER                                                   |
| 3                      | TATIC<br>Poth                      |                               |                | WATE:                     |            | EF I<br>Na   | n / P            | (FT) TOTAL DEPTH OPHLED 66.5 (FT)                                                     |
|                        | STHO                               | 08.                           |                |                           |            |              |                  |                                                                                       |
|                        | 0                                  | RILL                          | NG             | SOIL.                     | 3          | -1/8         | IN O             | .D. ROLLER BIT, 4 IN I.D. CASING AND DRILLING MUD                                     |
|                        | 9                                  | AMPL                          | ING            | SOL.                      | 2          | IN           | 0.D.             | SPLIT SPOON AND 3 IN G.D. SHELBY TUBE                                                 |
|                        | 0                                  | RILL                          | ING I          | ROCK                      | N          | ONE          |                  |                                                                                       |
| 5                      | PECIA                              | IL TI                         | ESTN           | IG OR                     | HN S       | TRI          | MEN              | TATION NONE                                                                           |
|                        | -                                  |                               |                |                           |            |              |                  |                                                                                       |
| C                      | OMME                               | INTS                          |                |                           |            |              |                  |                                                                                       |
|                        | -                                  |                               |                |                           |            |              |                  |                                                                                       |
| রা                     | -                                  |                               | -              |                           | -          |              |                  |                                                                                       |
| T)()6                  | E H                                | <b>_</b>                      | <u>ן</u><br>קר | 1.0                       | ت<br>چ     | - 10<br>     | د و              |                                                                                       |
| FEE                    | DEP                                | ANA                           |                |                           | N.         | 14<br>1<br>1 | S S              | SAMPLE DESCRIPTION                                                                    |
| -1                     |                                    | <b>"</b>                      |                |                           | REC        | >            | <b>5</b>         |                                                                                       |
|                        |                                    |                               |                |                           |            |              |                  | · · · · · · · · · · · · · · · · · · ·                                                 |
| ). 7                   | 0                                  | <b>_</b> \$                   | 1              | 11-21-                    | -21        | 42           | GP-              | SANDY GRAVEL, COARSE TO FINE TO 1 IN MAXIMUM, 20-302 COARSE TO FINE SAND,             |
|                        |                                    | 1                             | 1              | (13")                     | '.         |              | ଔ                | 3-104 SLAURILT PLASTIC FINES, BROWN, GRAY AND ORANGE.                                 |
|                        |                                    | ┥                             |                | 1                         |            |              |                  | · · · ·                                                                               |
|                        | •                                  | 1                             | 1              | 1                         |            |              |                  |                                                                                       |
|                        |                                    | <b>-</b> s                    | 2              | 6-5-3                     |            | 8            | SP-              | GRAVELLY SAND, 20-30% COARSE TO FINE GRAVEL, FEW FRAGMENTS TO 1.5 IN,                 |
|                        |                                    | ┨                             | -              | <b>[</b> <sup>(1)</sup> , |            |              |                  | where to take, morter neuton to time, J-104 Scientis fixedit fines. BROWN             |
|                        | 5 •                                | 1                             |                | <u> </u>                  |            |              |                  |                                                                                       |
|                        |                                    | -                             |                |                           |            |              |                  | · · · · ·                                                                             |
|                        |                                    | 7                             | 1              |                           |            |              |                  |                                                                                       |
|                        |                                    | 1                             |                |                           |            |              |                  |                                                                                       |
|                        |                                    | s                             | 3              | 2-1-1                     |            | 2            | SP-              | GRAVELLY SAND, 30-352 COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, COARSE TO            |
|                        |                                    | 1                             | 1              | (10")                     |            |              | SW               | FINE SAND, MOSTLY MEDIUM TO FINE, 5-102 SLIGHTLY PLASTIC FINES, GRAY.                 |
| ).7                    | 10 -                               |                               |                |                           |            |              |                  |                                                                                       |
|                        |                                    | -                             |                | l                         | •          |              |                  |                                                                                       |
|                        |                                    |                               | 1              | 1                         |            |              |                  |                                                                                       |
|                        | -                                  | 1                             | 1              |                           |            |              |                  |                                                                                       |
|                        |                                    | 5                             | 4              | 5-5-5                     |            | 10           | SP-              | GRAVELLY SAND, 15-252 COARSE TO FINE GRAVEL, 1 IN MAXIMUM, ANGULAR TO                 |
|                        |                                    | -                             | 1              | (15")                     |            |              | SW               | ROUNDED, COARSE TO FINE SAND, MOSTLY MEDIUM TO FINE, 5-102 NONPLASTIC<br>FINES, GRAY. |
|                        | 15                                 |                               | 1_             |                           |            | _            |                  | ·····                                                                                 |
| i.<br>2                | DATUN<br>17 GP                     | is i<br>Sume                  | AEAN.<br>WATE  | SEA L                     | .EVE<br>El | L            |                  | UNDISTURGED SAMPLES<br>US-SHELBY TUBE RODING LOG                                      |
| 3.                     | BLOWS                              | REC                           | UIRE           | TO D                      | RIVE       |              |                  | UO-OSTERBERG                                                                          |
|                        | 2°0.D.<br>DISTAI                   | SAMPI<br>VCE S                | le sf<br>Hown  | USING                     | i or       |              |                  | BEAVER VALLEY POWER STATION UNIT-                                                     |
|                        | 14016.                             | HAMM                          | OF             |                           | 30".<br>E  |              |                  | DUQUESNE LIGHT COMPANY                                                                |
|                        | RECOV                              | ERY.                          | ••••••         |                           | _          |              |                  | SHIPPINGPORT, PENNSYLVANIA                                                            |
| <b>4</b> .             |                                    |                               |                |                           |            |              |                  |                                                                                       |
| <b>4</b> .<br>5.       | STD.                               | PENET<br>S/FT.                | RATI           | ON RE                     | 51517      |              |                  |                                                                                       |
| 4.<br>5,<br>6.         | STD. I<br>BLOWS<br>UNIFIE<br>SYSTE | PENET<br>S/FT,<br>ED SO<br>M. | RATI           | ASSIF                     | ICATI      | ON           |                  | STONE & WEBSTER ENG. CORP.                                                            |
|                          |                                            |                    |        |                                  |                    |                     | SHEET 2 OF 3                                                                                                                                                                                                                                                                                                |
|--------------------------|--------------------------------------------|--------------------|--------|----------------------------------|--------------------|---------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| SI                       | <b>TF</b> 88                               | AVER               | VALLI  | EY POWER S                       | TATIO              | N-UNI               | T 2, SHIPPINGPORT, PA. J.O. NO. 12241.00                                                                                                                                                                                                                                                                    |
| ELEVATION<br>(FEET)(162) | DEPTH<br>(FEET)                            | SAMPLE<br>TYPE (7) | SAMPLE | BLLOWIS (3)<br>OR<br>REC/NOD (4) | SPT N<br>VALUE (5) | GROUP<br>SYNBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                          |
|                          |                                            |                    |        | F                                | 1                  |                     |                                                                                                                                                                                                                                                                                                             |
|                          | 1                                          |                    |        |                                  |                    |                     |                                                                                                                                                                                                                                                                                                             |
|                          |                                            | S                  | 5      | 4-2-3                            |                    | SP                  | SAND, TRACE FINE GRAVEL, MOSTLY MEDIUM TO FINE SAND, LESS THAN 5% NONFLASTIC FINES, BROWN.                                                                                                                                                                                                                  |
| 700.7                    | 20                                         |                    |        |                                  |                    |                     |                                                                                                                                                                                                                                                                                                             |
|                          |                                            |                    |        | -<br>-                           |                    |                     |                                                                                                                                                                                                                                                                                                             |
|                          | 4                                          |                    |        |                                  |                    |                     |                                                                                                                                                                                                                                                                                                             |
|                          | 2                                          | S                  |        | 10-14-26<br>(12")                | 40                 | GP-<br>GW           | SANDY CRAVEL, COASE TO FINE GRAVEL TO 1.3 IN, ANGULAR TO ROUGHD, 23-<br>COARSE TO FINE SAND, MOSTLY COARSE TO MEDIUM, TRACE IRON STATNING, BRC<br>BOTTOM 3 IN: BROKEN LIGHT GRAY <u>SANDSTONE</u> FRACMENTS TO 1.5 IN.<br>BLONS/INCH: 2-1-1-2-2-2/1-1-2-5-4/4-3-5-4-3-6                                     |
|                          | -                                          |                    |        |                                  |                    |                     |                                                                                                                                                                                                                                                                                                             |
|                          |                                            |                    |        |                                  |                    |                     |                                                                                                                                                                                                                                                                                                             |
| (00.7                    |                                            | S                  | 7      | 18-23-36<br>(17")                | 59                 | ਸL<br>GP-<br>GH     | TOP 7 IN: <u>GRAVELLY SILT</u> , 15-20% COARSE TO FINE GRAVEL, MOSTLY MEDIUM T<br>FINE, ANGULAR TO SUBANGULAR, 5-10% FINE SAND, VERY DRY, BROWN.<br>BOTTOM 10 IN: <u>SANDY GRAVEL</u> , COARSE TO FINE GRAVEL, 1.5 IN, ANGULAR, SG<br>BROKEN SANDSTONE, 25-35% COARSE TO FINE SAND, TRACE NONFLASTIC FINES, |
| 690.7                    | ,<br>, , , , , , , , , , , , , , , , , , , |                    |        |                                  |                    |                     | AND IRUW SIALAING, BROWN,<br>BLOWS/INCH: 1-4-3-3-4-3/4-3-4-5-3-4/9-7-6-5-4-5                                                                                                                                                                                                                                |
|                          | , I                                        | s                  | 8      | 2-4-5                            | ,                  | 511                 | TOP 5 IN: SILTY SAND, 10-15% COARSE TO FINE GRAVEL, SUBANGULAR, FINE                                                                                                                                                                                                                                        |
|                          | 35                                         |                    |        | (15")                            |                    | CL.                 | BOTTOM 10 IN: <u>SLUTY CLAY</u> , SLIGHTLY TO MODERATELY PLASTIC, STIFF, 5-7Z<br>COARSE TO FINE GRAVEL, SOME ROOTS, POCKETS OF COAL FRAGMENTS, MOIST, 1<br>GRAYISH BROWN.                                                                                                                                   |
|                          |                                            | S                  | 9      | 3-5-6<br>(17")                   | 11                 | нL                  | CLAYEY SILT, SLIGHTLY TO MODERATELY PLASTIC, TRACE FINE GRAVEL SIZED SANDSTONE AND COAL FRAGMENTS, FEW SANDSTONE FRAGMENTS TO 1 IN NEAR TO TRACE ROOTS, GRAY, $q_u$ (pp): 1.25, 1.75TSF                                                                                                                     |
|                          |                                            | 5                  | 10     | 3-4-6<br>(13")                   | 10                 | с <b>л</b> а<br>са  | TOP 4 IN: <u>CLAYEY SILT-SILTY CLAY</u> , SIMILAR TO S-9.<br>BOTTOM 9 IN: <u>Silty Clay</u> , slightly to moderately plastic, medium stiff<br>moist, gray brown. q <sub>u</sub> (pp): 1.5, 1.75TSF                                                                                                          |
| 680.7                    | 40                                         | US                 | 1      | (23.5")                          |                    | æ                   | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, OCCASIONAL GRAVEL TO 1 IN<br>MOIST, BROWN. (TUBE TRIMMINGS)                                                                                                                                                                                                     |
|                          |                                            | S                  | 11     | 4-3-3<br>(18")                   | 6                  | сī.                 | SILTY CLAY, SLIGHTLY TO MODERATELY PLASTIC, MEDIUM STIFF, OCCASIONAL FINE GRAVEL TO $1/2$ IN, SOME FINE SAND, MOIST, BROWN. $q_u$ (pp): 1.75, 2.00 TSP                                                                                                                                                      |
|                          |                                            | US                 | 2      | (23")                            |                    |                     | SIMILAR TO S-11. (TUBE TRIMMINGS)                                                                                                                                                                                                                                                                           |
|                          | 45 -                                       |                    |        | l ·                              |                    |                     |                                                                                                                                                                                                                                                                                                             |

|                                   | TE BI           | EAVER              | VALL   | EY POWER                       | STATIO             | N-UNI               | T 2, SHIPPINGPORT, PA.                                                                                                                                               |
|-----------------------------------|-----------------|--------------------|--------|--------------------------------|--------------------|---------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET)(162) <u>c</u> | DEPTH<br>(FEET) | SAMPLE<br>TYPE (7) | SAMPLE | BLOWS (3)<br>OR<br>REC/POD (4) | SPT N<br>VALUE (S) | GROUP<br>SYMBOL (6) | SAMPLE DESCRIPTION                                                                                                                                                   |
|                                   | 45              | 5                  | 12     | 4-9-11<br>(6")                 | 20                 | SP<br>SW            | GRAVELLY SAND, 15-202 COARSE TO FINE GRAVEL, ROUNDED TO SUBANGULAR, COARSE<br>TO FINE SAND, 3-5% NONPLASTIC FINES, BROWN                                             |
|                                   |                 | s                  | 13     | 8-7-9<br>(8")                  | 16                 | SP<br>SW            | GRAVELLY SAND, 10-15% COARSE TO FINE GRAVEL, ANGULAR TO ROUNDED, COARSE TO FINE SAND, 3-7% NONFLASTIC FINES, BROWN.<br>BLOWS/INCH: 2-1-2-1-1/1-1-1-1-2-1/2-2-1-2-1-1 |
| (70.7                             | -               | 5                  | 14     | 5-5-6<br>(8")                  | 11                 | GP                  | SANDY GRAVEL, MEDIUM TO FINE, SUBANGULAR TO ROUNDED, 25-30% COARSE TO FIN<br>SAND, MOSTLY COARSE TO MEDIUM.                                                          |
|                                   | -               |                    | 1.5    | 12-9-8                         | 17                 | GP-                 | SANNY GRAVEL, COARSE TO FINE, 1.5 TH MAXIMUM MOSTLY REGIVEN SANDSTONE                                                                                                |
|                                   | -               |                    |        | (8")                           |                    | GN                  | TRAGENTS, ANGUASE TO FARE, 15-20% COARSE TO FINE SAND, TRACE<br>NONFLASTIC FINES AND COAL, BROWN.<br>BLONS/INCH: 2-2-3-2-2-1/2-1-2-1-1-2/2-1-1-1-2-1                 |
|                                   |                 |                    | 16     | 10-14-8                        | 22                 | GP-                 | SIMILAR TO S-15, 5-77 NONPLASTIC FINES, BROWN.                                                                                                                       |
|                                   | -               |                    |        | (9")                           |                    | GW                  | BLOWS/INCH: 2-1-1-2-1-3/3-3-2-2-2/1-1-2-1-1-2                                                                                                                        |
|                                   | -               | s                  | 17     | 9-7-7                          | 14                 | GP-                 | <u>SIMILAR TO 5-13</u> , 7-102 NONPLASTIC FINES, BROWN.                                                                                                              |
| 660.7                             | 60              |                    |        |                                |                    |                     |                                                                                                                                                                      |
|                                   |                 |                    |        |                                |                    |                     |                                                                                                                                                                      |
|                                   | 65              |                    |        |                                |                    |                     |                                                                                                                                                                      |
|                                   | -               | s                  | 18     | 41-4 <b>2-3</b> 4<br>(14")     | 76                 | GP                  | SANDY GRAVEL, COARSE TO FINE GRAVEL SIZED GRAY SHALE AND ORANGE BROWN<br>SANDSTONE FRACHENTS, TRACE SLIGHTLY PLASTIC FINES, COAL AND IRON<br>STAINING.               |
|                                   | -               |                    |        |                                |                    |                     | BOTTOM OF BORING AI 66.5 FT<br>Elevation 654.2 Ft                                                                                                                    |
|                                   | -               |                    |        |                                |                    |                     |                                                                                                                                                                      |
|                                   | -               |                    |        |                                |                    |                     |                                                                                                                                                                      |
| - A                               | -               |                    |        |                                |                    |                     |                                                                                                                                                                      |

| 0                           |                                                                                                                                                                                                                                                                                | e. · N                                                                                                     | 3737                                                                                                   |                                           | 1 67                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 114.6 SHEET I OF                                                                                                                                                                                                                                                 |
|-----------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------|-------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ING                         |                                                                                                                                                                                                                                                                                |                                                                                                            |                                                                                                        |                                           | BEAI                                                                                                                                                                                                                   | RING LOGGED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | BYD. HACNETLL                                                                                                                                                                                                                                                    |
| DA                          | TE : STAR                                                                                                                                                                                                                                                                      | T/FIN                                                                                                      | IISH 8-10                                                                                              | -77                                       | _/                                                                                                                                                                                                                     | 8-11-77 CONTRACTOR / DRL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | ER RATHOND/KODITEK                                                                                                                                                                                                                                               |
| ST                          | ATIC GROU                                                                                                                                                                                                                                                                      | JNDWA'                                                                                                     | TER DE                                                                                                 | Р <b>Т</b> Н /                            | DAT                                                                                                                                                                                                                    | E / DRILL RIG                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | ; TYPE                                                                                                                                                                                                                                                           |
| DE                          | PTH TO I                                                                                                                                                                                                                                                                       | EDRO                                                                                                       | ск                                                                                                     | 7                                         | 77.0                                                                                                                                                                                                                   | TOTAL DEPTH DRILL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ED                                                                                                                                                                                                                                                               |
| ME                          | THODS :                                                                                                                                                                                                                                                                        |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             | DRILLIN                                                                                                                                                                                                                                                                        | IG 501                                                                                                     | L _*                                                                                                   | CASING                                    | 1, 3                                                                                                                                                                                                                   | 7/8" ROLLER BIT, AW RODS, MED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | · · · ·                                                                                                                                                                                                                                                          |
|                             | SAMPLI                                                                                                                                                                                                                                                                         | NG SO                                                                                                      | ж. <u>2</u> *                                                                                          | 0.D. 9                                    | FLIT                                                                                                                                                                                                                   | SPOOR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                  |
|                             | DRILLI                                                                                                                                                                                                                                                                         | IG ROO                                                                                                     | CK                                                                                                     |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
| SP                          | ECIAL TES                                                                                                                                                                                                                                                                      | STING                                                                                                      | OR INST                                                                                                | <b>FRUME</b>                              | ENTA                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
| 05                          |                                                                                                                                                                                                                                                                                | ·····                                                                                                      |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                            |
| RE                          | MARKS                                                                                                                                                                                                                                                                          |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             |                                                                                                                                                                                                                                                                                |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
| ₹                           | =                                                                                                                                                                                                                                                                              | <u> </u>                                                                                                   | (2)<br>D.S.<br>DM(3)                                                                                   | LLES<br>EC(4                              | 90                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | FIELD AND LA                                                                                                                                                                                                                                                     |
|                             |                                                                                                                                                                                                                                                                                | - La                                                                                                       | OWS<br>L                                                                                               | N S                                       | S.                                                                                                                                                                                                                     | SAMPLE DESCRIPTION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | TEST RESULTS                                                                                                                                                                                                                                                     |
|                             | 55 35                                                                                                                                                                                                                                                                          |                                                                                                            | 1 1 5 U                                                                                                |                                           | ng:                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | COMMENTS                                                                                                                                                                                                                                                         |
|                             |                                                                                                                                                                                                                                                                                |                                                                                                            | <u> </u>                                                                                               | 1.0                                       | 0                                                                                                                                                                                                                      | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                  |
| — T                         |                                                                                                                                                                                                                                                                                | тт                                                                                                         |                                                                                                        | П                                         |                                                                                                                                                                                                                        | ······································                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                  |
|                             | - 1                                                                                                                                                                                                                                                                            |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ·                                                                                                                                                                                                                                                                |
|                             | j                                                                                                                                                                                                                                                                              |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             |                                                                                                                                                                                                                                                                                |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             |                                                                                                                                                                                                                                                                                |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             |                                                                                                                                                                                                                                                                                | -                                                                                                          |                                                                                                        |                                           |                                                                                                                                                                                                                        | SILTY SAND, UNIFORM, FINE, 35-455 NO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | NFLASTIC FINES, LIGHT BROWN, 1                                                                                                                                                                                                                                   |
| <i>1</i> 30                 | . ]].                                                                                                                                                                                                                                                                          |                                                                                                            | 5-5-7                                                                                                  | 12                                        | SM                                                                                                                                                                                                                     | LATER OF UNIFORM FINE SAND, LESS THAT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | n 5% Fines.                                                                                                                                                                                                                                                      |
|                             | ´ヿ <b>Ľ</b>                                                                                                                                                                                                                                                                    | -  *                                                                                                       |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             | ∃⊢                                                                                                                                                                                                                                                                             | ┥                                                                                                          | 4-5-6                                                                                                  | n                                         | зм                                                                                                                                                                                                                     | SILTY SAND, SINGLAR TO ABOVE.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                  |
|                             | · -   s                                                                                                                                                                                                                                                                        | 2                                                                                                          |                                                                                                        |                                           |                                                                                                                                                                                                                        | FIRST FIRST ACCOUNTS ON UNLOSS                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | • .                                                                                                                                                                                                                                                              |
|                             | ᅴ는                                                                                                                                                                                                                                                                             | -                                                                                                          |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
| -                           |                                                                                                                                                                                                                                                                                |                                                                                                            |                                                                                                        |                                           |                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                                                                                                                                                  |
|                             | -1.                                                                                                                                                                                                                                                                            | ٦.,                                                                                                        | 7-8-8                                                                                                  | 16                                        | SK                                                                                                                                                                                                                     | SILTY SAND, WIDELY GRADED. 20-25% SI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | BROUNDED GRAVEL TO 0.7 INCH MAI                                                                                                                                                                                                                                  |
| 725                         | ] s                                                                                                                                                                                                                                                                            | 3                                                                                                          | 7-8-8                                                                                                  | 16                                        | SM                                                                                                                                                                                                                     | SILTY SAMD, WIDELY GRADED, 20-25% SU<br>HOSTLY FINE SAND, 15-20% NONFLASTIC                                                                                                                                                                                                                                                                                                                                                                                                                                                                | BROUNDED GRAVEL TO 0.7 INCH MAL<br>FINES, BROWN.                                                                                                                                                                                                                 |
| 725                         |                                                                                                                                                                                                                                                                                | 3                                                                                                          | 788                                                                                                    | 16                                        | SM                                                                                                                                                                                                                     | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, BOOSTLY GRADED, 25-200                                                                                                                                                                                                                                                                                                                                                                                                                        | BROUNDED GRAVEL TO 0.7 INCH MAN<br>FINES, BROWN.                                                                                                                                                                                                                 |
| 725                         | -   s<br>  s<br>  s                                                                                                                                                                                                                                                            | 3                                                                                                          | 7 <b>-8-8</b><br>5-7-8                                                                                 | 16<br>15                                  | SM<br>SP<br>SM                                                                                                                                                                                                         | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-305<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B                                                                                                                                                                                                                                                                                                                                                                                 | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.                                                                                                                                                                |
| 725                         | s<br>                                                                                                                                                                                                                                                                          | 3                                                                                                          | 7 <b>-8-8</b><br>5-7-8                                                                                 | 16<br>15                                  | SM<br>SP<br>SM                                                                                                                                                                                                         | SILTY SAND, WIDELT GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELLY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B                                                                                                                                                                                                                                                                                                                                                                                | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.                                                                                                                                                                |
| 725                         |                                                                                                                                                                                                                                                                                | 3                                                                                                          | 7-8-8<br>5-7-8<br>15-20-20                                                                             | 16<br>15<br>40                            | SM<br>SP-<br>SM<br>GP-                                                                                                                                                                                                 | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-307<br>MOSTLY FINE SAND, 5-10% NONFLASTIC F                                                                                                                                                                                                                                                                                                                                                                                 | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI                                                                                                                              |
| 725                         |                                                                                                                                                                                                                                                                                | 3                                                                                                          | 7 <b>-8-8</b><br>5-7-8<br>15-20-20                                                                     | 16<br>15<br>40                            | SM<br>SP-<br>SM<br>GP-<br>GM                                                                                                                                                                                           | SILTY SAND, WIDELY GRADED, 20-25% SI<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-305<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, POORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL                                                                                                                                                                                                                                                                                                 | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.                                                                                                       |
| 725                         | , , , , , , , , , , , , , , , , , , ,                                                                                                                                                                                                                                          | 3                                                                                                          | 7 <b>-8-8</b><br>5-7-8<br>15-20-20                                                                     | 16<br>15<br>40                            | SH<br>SP-<br>SM<br>GP-<br>GR-                                                                                                                                                                                          | SILTY SAND, WIDELY GRADED, 20-255 SU<br>MOSTLY FINE SAND, 15-205 NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-305<br>MOSTLY FINE SAND, 5-105 NONFLASTIC B<br>SANDY GRAVEL, POORLY GRADED, ANGULAE<br>30-355 MOSTLY FINE SAND, 5-105 NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                                                                                              | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.                                                                                                       |
| 725                         |                                                                                                                                                                                                                                                                                | 3 4 4 5 1 6                                                                                                | 7- <b>8-8</b><br>5-7-8<br>15-20-20<br>13-16-12                                                         | 16<br>15<br>40<br>28                      | SH<br>SP-<br>SM<br>GP-<br>GM<br>GP-<br>GM                                                                                                                                                                              | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC F<br>SANDY GRAVEL, POORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                                                                                              | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAIL<br>ASTIC FINES, BROWN.                                                                                                       |
| 20                          | - S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>- S | 3<br>                                                                                                      | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>IMPLE                                                        | 16<br>15<br>40<br>28                      | SM<br>SP-<br>SM<br>GP-<br>GM<br>GP-<br>GN                                                                                                                                                                              | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-305<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, FOORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                                                                                              | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.                                                                                                       |
| 725                         | SPLIT BAR                                                                                                                                                                                                                                                                      | 3<br>4<br>5<br>REL SA<br>SAMPL                                                                             | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>IMPLE<br>ES (U.G.S.)                                         | 16<br>15<br>40<br>28                      | SK<br>SP-<br>SX<br>GP-<br>GX<br>GR-<br>GN<br>UN                                                                                                                                                                        | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, POORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                                                                                              | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG                                                                                         |
| 725<br>20<br>5<br>UN        | SPLIT BAR<br>SPLIT BAR<br>DISTURBED<br>US_SMELE<br>UF_FIXED                                                                                                                                                                                                                    | 3<br>4<br>5<br>6<br>REL SA<br>SAMPL<br>Y TUB<br>PISTOI                                                     | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>Las (0.3.)<br>KE                                             | 16<br>15<br>40<br>28                      | SH<br>SP-<br>SH<br>GP-<br>GH<br>GR<br>DAT<br>UNI<br>UNI<br>2 °C                                                                                                                                                        | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELLY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, ROORLY GRADED, ANGULAF<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                                                                                             | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAIL<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLEY FOWER STATION UNI:                                                      |
| 725                         | SPLIT BAR<br>SPLIT BAR<br>DISTURBED<br>US_ SHELE<br>UF_ FIXED<br>UO_ OSTER                                                                                                                                                                                                     | A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A                | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>MPLE<br>ES (0.0.3.)<br>KE<br>N                               | 16<br>15<br>40<br>28                      | SH<br>SP-<br>SH<br>GP-<br>GH<br>CH<br>L BL<br>C<br>OIS                                                                                                                                                                 | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC I<br>SANDY GRAVIE, POORLY GRADED, ANGULAF<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVIE, SIMILAR TO ABOVE.<br>UM IS MEAN SEA LEVEL<br>LESS OTHERWISE ROCATED<br>DWS REDURRED TO DRIVE<br>D. SAMPLE SPOON 6" OR<br>TANCE SHOWN USING 1401A                                                                                                                               | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLEY FOWER STATION UNIT<br>UQUEBNE LIGHT COMPANY                             |
| 725                         | SPLIT BAR<br>SPLIT BAR<br>DISTURBED<br>US_ SHELB<br>US_ FIXED<br>UO_ OSTER<br>UD_ DEMISS<br>UP_ PITCH                                                                                                                                                                          | A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A                | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>MMPLE<br>.ES (U.O.S.)<br>KE<br>N                             | 16<br>15<br>40<br>28                      | SH<br>SP-<br>SN<br>GP-<br>GN<br>GR-<br>GN<br>SL<br>C<br>SL<br>C<br>SL<br>C<br>SL<br>C<br>SL<br>SL<br>C<br>SL<br>SL<br>C<br>SL<br>SL<br>C<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL<br>SL | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC F<br>SANDY GRAVEL, POORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SINILAR TO ABOVE.<br>UM IS MEAN SEA LEVEL<br>LESS OTHERWISE NONCATED<br>DWS REQURRED TO DRIVE<br>10. SAMPLE SPOON 6" OR E<br>TANCE SHOWN USING HOILAID<br>NONCATES USE OF 300 Ib                                                                                               | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLEY FOVER STATION UNIT<br>UQUESNE LIGHT COMPANY<br>ITTSBURGH, PENNSYLVANIA  |
| 725                         | SPLIT BAR<br>DISTURBED<br>US_SHELE<br>US_SHELE<br>US_SHELE<br>UD_DENISI<br>UD_DENISI<br>UD_DENISI<br>UD_DENISI<br>STD_PENE<br>RESISTANCI                                                                                                                                       | A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A                | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>13-16-12<br>NMPLE<br>.ES (U.D.3.)<br>N<br>N<br>S/FT          | 16<br>15<br>40<br>28                      | SH<br>SP-<br>SX<br>GP-<br>GM<br>GP-<br>GM<br>UNI<br>UNI<br>UNI<br>COS<br>HAI<br>HAI<br>HAI                                                                                                                             | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>ORAVELY SAND, POORLY GRADED, 25-305<br>MOSTLY FINE SAND, 5-10% NONFLASTIC F<br>SANDY GRAVEL, FOORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, FOORLY GRADED, ANGULAS<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SINILAR TO ABOVE.<br>UM IS MEAN SEA LEVEL<br>LESS OTHERWISE MOKATED<br>DUS SAMPLE SPOON 6" OR<br>TANCE SHOWN USING 1401h<br>MMER FALLING 30"<br>MORATES USE OF 300 IN<br>MMER NCHES OF PENETR. | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLEY FOWER STATION UNIT<br>UQUESNE LIGHT COMPANY<br>TITISBURGH, PENNSYLVANIA |
| 725<br>20<br>8.<br>UN<br>20 | SPLIT BAR<br>SPLIT BAR<br>DISTURBED<br>US_SHELE<br>UF_FIXED<br>UD_OSTER<br>UD_OENISI<br>UD_DENISI<br>UD_DENISI<br>RESISTANCI                                                                                                                                                   | A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A<br>A                                         | 7-8-8<br>5-7-8<br>15-20-20<br>13-16-12<br>13-16-12<br>MPLE<br>ES (U.0.5)<br>KE<br>N<br>N<br>S/FT       | 16<br>15<br>40<br>28<br>1                 | SH<br>SP-<br>SH<br>GP-<br>GH<br>GR<br>GR-<br>GR<br>CH<br>SH<br>CH<br>SH<br>CH<br>SH<br>CH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH<br>SH                                                | SILTY SAND, WIDELY GRADED, 20-25% ST<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, POORLY GRADED, ANGULAR<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SINILAR TO ABOVE.<br>UN IS MEAN SEA LEVEL<br>LESS OTHERWISE NONCATED<br>DWS REDURED TO DRIVE<br>10. SAMPLE SPOON 6" OR<br>TANCE SHOWN USING MOIN<br>MMER FALLING 30"<br>NDICATES USE OF 300 IB<br>MMER<br>NOTES INCHES OF PENETR.<br>UNDISTURGED SAMPLER                       | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLET FOWER STATION UNIT<br>UQUEENE LIGHT COMPANY<br>ITTSBURGE, PENNSYLVANIA  |
| 20<br>8.<br>UN<br>VN        | SPLIT BAR<br>SPLIT BAR<br>DISTURBED<br>US_SHELS<br>US_SHELS<br>UF_FIXED<br>UO_OSTER<br>UD_DENISH<br>UP_PITCH<br>STD_PENE<br>RESISTANCI<br>GROUNDWAY                                                                                                                            | 3<br>4<br>5<br>6<br>1<br>1<br>6<br>1<br>1<br>6<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1<br>1 | 7-8-5<br>5-7-8<br>15-20-20<br>13-16-12<br>I3-16-12<br>I3-16-12<br>ES (U.O.3.)<br>N<br>N<br>N<br>N<br>N | 16<br>15<br>40<br>28<br>1.<br>2<br>3<br>4 | SH<br>SP-<br>SH<br>GP-<br>GH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH<br>CH                                                                                                             | SILTY SAND, WIDELY GRADED, 20-25% SU<br>MOSTLY FINE SAND, 15-20% NONFLASTIC<br>GRAVELLY SAND, POORLY GRADED, 25-30%<br>MOSTLY FINE SAND, 5-10% NONFLASTIC B<br>SANDY GRAVEL, POORLY GRADED, ANGULAF<br>30-35% MOSTLY FINE SAND, 5-10% NONFL<br>SANDY GRAVEL, SIMILAR TO ABOVE.<br>'UM IS MEAN SEA LEVEL<br>LESS OTHERWASE MOLCATED<br>DUS REQUIRED TO DRIVE<br>D.D. SAMPLE SPOON 6" OR<br>TANCE SHOWN USING MOLTS<br>MAKER FALLING 30"<br>WORCATES USE OF 300 IB<br>MMER FALLING 30"<br>NOTES INCHES OF PENETR.<br>UNDISTUMBED SAMPLE      | BROUNDED GRAVEL TO 0.7 INCH MAI<br>FINES, BROWN.<br>SUBROUNDED GRAVEL TO 0.6 INCH<br>INES, BROWN.<br>TO SUBANGULAR TO 1.0 INCH MAXI<br>ASTIC FINES, BROWN.<br>BORING LOG<br>EAVER VALLEY FOWER STATION UNIT<br>SUGUESNE LIGHT COMPANY<br>ITTSBURCH, PENNSYLVANIA |

|                     |       |     |       |                  |                                       |             |                 | BORING SUS-L                                                                                                                                                                      |
|---------------------|-------|-----|-------|------------------|---------------------------------------|-------------|-----------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
|                     |       |     |       |                  |                                       |             |                 | SHEET_2_ OF _4                                                                                                                                                                    |
| S                   | ITE / | LO  | CAT   | ION              | BEAVER                                | VALLI       | a 70            | MER STATION UNIT 2 J.O. NO. 12241                                                                                                                                                 |
| ELEVATION<br>(FEET) | DEPTH |     | TYPE  | SAMPLE<br>NUMBER | BLOWS (2)<br>OR UDS<br>PENETRATION(3) | SPT N WULES | BROLP SMIROL    | FIELD AND LAB.<br>SAMPLE DESCRIPTION TEST RESULTS /<br>COMMENTS                                                                                                                   |
|                     |       |     | 5     | 7                | 6-7-13                                | 20          | SP-<br>SM       | SAND, FOORLY GRADED, 3-85 GRAVEL TO 0.6 INCH MAXIMUM, MOSTLY MEDIUM TO<br>FIRE SAND, 5-85 NOHFLASTIC FINES, BROWN.                                                                |
|                     |       |     | s     | 8                | 11-15-16                              | 31          | SP-             | GRAVELLY SAND, FOORLY GRADED, 25-35% SUBROUNDED TO SUBANGULAR GRAVEL TO -<br>0.7 INCH MALINUM, MOSTLI NEDIUM TO FINE SAND, 5-10% NONFLASTIC FINES, -<br>BROWN.                    |
| 15                  | 20    |     | s     | 9                | 9- <del>9-1</del> 2                   | 21          | SP<br>Sm        | SAND, UNIFORM, FINE, 5-8% NONPLASTIC FINES, LIGHT BROWN.                                                                                                                          |
|                     |       |     | 5     | 10               | 8-13-19                               | 32          | SP-<br>SM<br>GP | TOP 3 INCHES: SAND, SIMILAR TO ABOVE.<br>BOTTOM: SANDI GRAVEL, FOORLY GRADED, MOSTLY COARSE ANGULAR TO 1.0 INCH<br>MAXDRM, 20-30% COARSE TO FINE SAND, LESS THAN 5% FINES, BROWN. |
| 710                 | 25    |     | s     | ļ                | 9-27-20                               | 47          | GP              | SANDI GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                   |
| •                   |       |     | 5     | 12               | 8-12-15                               | 27          | gp              | SANDY GRAVEL, SIMILAR TO ABOVE ENCEPT ONLY 20-25% SAND.                                                                                                                           |
| 705                 |       |     | 8     | 13               | 10-22-20                              | 42          | GP              | SANDI GRAVEL, SIMILAR TO # 12.                                                                                                                                                    |
|                     | 30    |     | s     | <b>1 1</b>       | 13-16-20                              | 36          | 07-<br>GN       | SANDI GRAVIL, FOORLI GRADED, SUBANGULAR TO U.S INCH MAXIMUM, 30-35%<br>NOSTLY MEDIUM TO FINE SAND, 5-6% NONPLASTIC FINES, DARK BROWN.                                             |
|                     |       |     | s     | 15               | 18-21-20                              | 42          | ср.<br>Сн       | SANDI GRAVEL, SINCLAR TO ABOVE.                                                                                                                                                   |
| 00                  | 35 -  |     | s     | 16               | 16-24-20                              | u           | SP-<br>SK       | TOF SAND, FOORLY GRADED, COARSE TO FINE, MOSTLY FINE, 5-85 NONFLASTIC FINE<br>ENDIN.<br>BOTTOM 3 INCERS: GRAVEL.                                                                  |
|                     |       |     | 3     | 17               | 9-11-10                               | 21          | sp-<br>Sm       | SANDY CRAVEL, FOORLY GRADED, SUBANGULAR TO 1.2 INCH MAXIMUM, 25-35% MOSTLY<br>FINE SAND, 5-8% NONFLASTIC FINES, BROWN.                                                            |
| 695                 |       |     | s     | 18               | 7-5-5                                 | 10          | GM              | SILTI GRAVEL, WIDELY GRADED, ANGULAR TO SUBANGULAR TO 0.6 INCH MAIIMOM,<br>25-35% COARSE TO FINE SAND, 12-15% NONPLASTIC FINES, BROWN AND GRAY.                                   |
|                     | #     |     | 5     | 19               | 8-8-7                                 | 15          | SM              | SILTT SAND, WIDELY GRADED, 35-40% SUBROUNDED TO 0.7 INCH MAXIMUM, MOSTLY<br>FINE SAND, 12-15% MONFLASTIC FINES, BROWN.'                                                           |
|                     |       |     | 8     | 20               | 7-6-7                                 | 13          | SM              | SILTY SAND, SIMILAR TO ABOVE.                                                                                                                                                     |
| 69U                 | 45    | 1   | s     | 21               | 7-5-7                                 | u           | GH              | SILTY GRAVEL, WIDELY GRADED, SUBANGULAR TO 0.8 INCH MAXIMUM, 30-35% MOSTLA<br>MEDIUM TO FINE SAND, 12-15% NONPLASTIC FINES, BROWN                                                 |
| ÓTE :               | FOR I | D M | VG SU |                  | EET 1                                 | <b>A</b> s  | TON             | E & WEBSTER ENG. CORP ISSUED BY DATE AI BORING NO. SHEET                                                                                                                          |

|                |               |      |        |                  |                                        |                              |                            | BORING                                                                                                                                        | - |
|----------------|---------------|------|--------|------------------|----------------------------------------|------------------------------|----------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------|---|
| •              | <b>TE</b> /   | 1.04 | - AT 1 |                  | BZAVER V                               | ALLEY                        | FON                        | R STATION UNIT 2 J.O. NO. 12241                                                                                                               | - |
| (FEET)         | DEPTH         |      | TYPE   | SAMPLE<br>NUMBER | BLOWS (2)<br>OR UDS.<br>PENETRATION(3) | SPT N VALUES<br>OR UDS REC[4 | anoue smead <sup>(5)</sup> | FIELD AND LAB.<br>SAMPLE DESCRIPTION TEST RESULTS /<br>COMMENTS                                                                               | _ |
|                |               | ╉    | F      |                  |                                        |                              |                            |                                                                                                                                               |   |
|                |               |      | s      | 22               | 7-5-6                                  | Ľ                            | QP                         | SARDY GRATEL, FOORLY GRADED, MOSTLY COARSE SUBROUNDED TO 1.1 INCH MAXIMUR<br>25-30% MOSTLY COARSE AND MEDIUM SAND, LESS THAN 5% FINES, BROWN. |   |
|                |               |      | 8      | 23               | 4-8-15                                 | 23                           |                            | NO RECOVERT                                                                                                                                   |   |
| 685            | 50 _          |      |        |                  | _ / /                                  |                              |                            |                                                                                                                                               |   |
|                |               |      | 8      | 24               | 7-0-0                                  |                              |                            | BU RECUVERI                                                                                                                                   |   |
|                | -             |      | 8      | 25               | 11-8-7                                 | 15                           | SP-                        | SAND, UNIFORM, FINE, 5-605 NONPLASTIC FINES, BROWN.                                                                                           |   |
| 680            | 55 . •        |      | s      | 26               | 7-6-7                                  | ט                            | ۵P                         | SANDI GRAVEL, POORLY GRADED, NOSILY COARSE SUBROUNDED TO 1.2 INCH MAXIMUM<br>25-33% MOSILY COARSE AND MEDIUM SAND, LESS THAN 5% FINES, BROWN. |   |
|                |               |      | s      | 27               | 11-13-10                               | 23                           | GP-                        | SANDY GRAVEL, FOORLY GRADED, SUBROUNDED TO 1.1 INCH MAXIMUM, 30-35% MOSTI<br>MEDIUM TO FINE SAND, 5-LO% NONFLASTIC FINES, BROWN.              |   |
|                | -             |      |        |                  |                                        |                              | GR                         |                                                                                                                                               |   |
| 675            | <u> 6</u> 0 - | ]    | s      | 28               | 10-10-16                               | 26                           |                            | NOT ENOUTH SAMPLE FOR ACCURATE CLASSIFICATION (APPEARS TO BE SIMILAR TO                                                                       |   |
|                |               |      | s      | 29               | 6-8-8                                  | 16                           | GP                         | SANDI GRATEL, FOORLI GRADED, MOSTLY COARSE, SUBHOUNDED TO 1.0 12CH WALLAN<br>25-305 COARSE TO FINE SAND, 8-125 NONPLASTIC FINES, BROWN.       |   |
|                |               |      | s      | 30               | 8-9-8                                  | 17                           | GP<br>GN                   | SANDY GRAVEL SIMILAR TO ABOVE.                                                                                                                |   |
| 670            | 65            |      | 3      | 31               |                                        |                              |                            | DRILLER OVER DRILLED & FOOT AND WENT TO NEXT SAMPLE                                                                                           |   |
|                |               |      |        |                  |                                        |                              |                            |                                                                                                                                               |   |
|                | -             |      | 8      | 32               | 17-16-17                               | 33                           |                            | NO RECOVERY - 2 ATTEMPTS                                                                                                                      |   |
| 665            |               |      | s      | 33               | 8-9-8                                  | 17                           | 62-<br>GH                  | COARSE SAND, 5-10% NONPLASTIC FINES, BROWN.                                                                                                   |   |
| •              | 70            |      | 5      | 34               | 10-9-12                                | 21                           | GP-<br>GH                  | SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                               |   |
| :              | -             |      | s      | 35               | 9 <b>-</b> 72- <u>100</u>              | 1.72+                        | SP-                        | SAND, UNIFORM, FIR, 5-6% NONPLASTIC FINES, BROWN.                                                                                             |   |
| <del>6</del> 0 | 75            |      | -      | 36               | 3#<br>13-52-46                         | 98                           | SM<br>GP-                  | TOP SANDY GRAVEL, FOORLY GRADED, SUBANGULAR TO 0.8 INCH MAXIMUM, 35-40% MOSTLY FINE SAND, 8-125 NONFLASTIC FAMES, BROWN, BOTTOM 7 INCHES:     |   |
|                | 75            | -    | s      | 36               | 13-52-46                               | 98                           | ICP-                       | HOSTLY FINE SAND, 8-125 NONFLASTIC FINES, BROWN, BOTTOM 7 INCHES :                                                                            |   |



BVPS-2 UFSAR

Rev. 0

|    | SITE                   | 1              | ,OCA1           |               |                |       | يو بلساية | <u>1 101</u>               | 4200 0                                                                                                | . 3 .      |
|----|------------------------|----------------|-----------------|---------------|----------------|-------|-----------|----------------------------|-------------------------------------------------------------------------------------------------------|------------|
|    | COOF                   | RDIN/          | ATES            | <u> </u>      | 500.0          |       |           | <u><u> </u></u>            | GEOLIU GROUND ELEV. (I) 767-7 [STEET TOP_                                                             |            |
|    | INCL                   | NAT            | ION .           |               |                | 80    | 77        | BEA                        | ARING LOGGED BY                                                                                       | _          |
|    | DATE                   |                | IARI            | 77);          | NISH           |       |           | ′                          |                                                                                                       |            |
|    | STAT                   | 16 9<br>14 7   |                 | NUW/          | AT ER          | UEP   |           | 62.0                       |                                                                                                       | <b>—</b> . |
|    | METI                   |                | о в<br>-        | LONG          | /un _          |       |           |                            |                                                                                                       | <u> </u>   |
|    |                        | DRI            | P.<br>L. L. IMG | 2 90          |                | 4" (  | CASI      | NG, 3                      | 3 7/8" ROLLER BIT, AW RODS, MUD                                                                       |            |
|    |                        | SAI            |                 | ig si         | OL.            |       | 2* (      | o.p.                       | SFLIT SPOON                                                                                           | _          |
|    |                        | DRI            |                 | S RO          | CK .           |       |           |                            |                                                                                                       |            |
|    | SPE                    |                | TES             | TING          | ORI            | NST   | RUM       | ENT                        | (ATION                                                                                                | _          |
|    |                        |                |                 |               |                |       |           |                            |                                                                                                       |            |
|    | REM/                   | ARKS           | 3               |               |                |       |           |                            |                                                                                                       |            |
|    |                        | _              |                 | :             |                |       |           |                            |                                                                                                       | ·          |
|    |                        |                |                 |               |                |       |           |                            |                                                                                                       |            |
|    | Τ                      |                |                 |               |                | 0     | 83        | <u>B</u>                   |                                                                                                       |            |
| Ē  | Ĩ                      | E              | 22              | L H           | NS (3          | e     | <b>J</b>  | P.                         | FIELD AND LAS                                                                                         |            |
| Ĩ  | - W                    | <u>ا</u> و     | 8 X X<br>7 X    | N N N         | l S s          | E TR  | NG        | 5                          | SAMPLE DESCRIPTION TEST RESULTS                                                                       | ,<br>,     |
| -  |                        |                |                 |               |                | N.    | 58        | 1 g                        | COMMENTS                                                                                              |            |
|    |                        |                |                 |               |                |       |           |                            |                                                                                                       |            |
|    | Т                      | ᠴ              |                 | T             | T              |       |           | Γ                          |                                                                                                       |            |
|    |                        | · •            |                 |               |                |       |           |                            | 0-6' FLACED FILL TO LEVEL AREA FOR DRILLING.                                                          |            |
|    | 1                      | 7              |                 |               | j              |       |           |                            |                                                                                                       |            |
|    |                        | -              |                 |               |                |       |           |                            |                                                                                                       | -          |
|    |                        | 7              |                 |               | 1              |       |           |                            |                                                                                                       |            |
|    |                        |                | 1               |               |                |       |           |                            |                                                                                                       | · •        |
|    | 5                      | -              |                 |               |                |       |           |                            |                                                                                                       | .—         |
|    |                        | 1              | ·               | ļ             |                |       |           |                            | THE A TACHERS. SANTA CLAY STICHTY PLACETC 15 34 MONTY PLACET                                          |            |
|    |                        | 4              | s               | 1             | 3-3-4          |       | 7         | CL                         | BOWN.                                                                                                 | •          |
|    |                        | 7              | <b> </b>        | ł             | ·              |       |           |                            | DITON: STATE CLAR, SECONDER FLASTIC, BROWN,                                                           |            |
|    |                        | 1              |                 |               | 2-2-4          |       | 4         | SP-                        |                                                                                                       | -          |
|    | 1                      | -              | ) S             | 2             | ~~~~           | 1     | Q         | SM                         | NONPLASTIC FINES, BROWN.                                                                              |            |
| 20 | 10                     | 4              |                 | 1             | į.             | İ     |           |                            |                                                                                                       | _          |
|    |                        | _              |                 |               | 2-3-3          |       | 6         | SP-                        | SAND, POURLI GRADED, 8-12% GRAVEL TO 0.5 INCH MAXIMUM, MOSTLY FINE<br>5-10% NONFLASTIC FINES, BROWN.  | SAND,_     |
|    | 1                      | 7              | F               | ļĺ            |                |       |           | SM                         |                                                                                                       | -          |
|    |                        | Ţ              |                 | 1             |                |       | • •       |                            | TOP 4 INCHES : SAND, UNIFORM, FINE, 5-10% NONPLASTIC FINES, BROWN.                                    |            |
|    |                        | -              | S               | 4             | 4-2-7          |       | 12        | 30                         | DUTIONI CLAINE SAND, UNIFORM, FINE, 20-25% SLIGHTLY PLASTIC FINES, B                                  | NOWN,      |
|    |                        | 1              |                 | 1             |                |       | .         |                            | THE THEY, STITUT OLLY STICHT V DELETTS BELIEF BORNING VILLE                                           |            |
| 15 | 15                     | 4              | S               | 5             | 3-5-5          |       | 10        | SN                         | SAND, UNIFORM, FINE, 12-15% NONPLASTIC FINES, BRUM. FOLLOWED BY 3                                     | INCH -     |
| Γ  | 5_ S                   | PLIT           | BARR            | EL S          | AMPLE          |       |           | . DA1                      | NTUM IS MEÁN SEA LEVEL                                                                                |            |
| 1  | UNDIS                  |                | BED S           |               | LES (U.)<br>Ne | D.S.) |           | UN<br>2 84                 | NLESS OTHERWISE INDICATED BORING LOG                                                                  |            |
|    | _ UI                   | F_ FI          | XED             | PISTO         | )N             |       |           | 2"(                        | O.D. SAMPLE SPOON 6" OR BEAVER VALLEY FOWER STATION UNIT :                                            | 1          |
|    |                        | )_ 09<br>)_ 09 | STERB<br>ENISON | ERG<br>I      |                |       |           | DIS                        | STANCE SHOWN USING 140 Ib. DUQUESNE LIGHT COMPANY                                                     |            |
|    |                        |                | TCHE            | R             |                |       |           | *                          | INDICATES USE OF 300 Ib PITTSBURGH, PENNSYLVANIA                                                      |            |
|    | UF                     |                | ENET            |               | Nec -          |       |           |                            |                                                                                                       |            |
|    | UF<br>N_S'<br>RE       | TD. P<br>SIST  | ANCE            | BLOY          | N<br>VS/FT     |       | :         | HA<br>3. DE                | AMMER<br>ENOTES INCHES OF PENETR                                                                      |            |
|    | N_S<br>N_S<br>RE<br>GI | TD P<br>SIST   | DWATI           | RATIC<br>BLOV | ON<br>VS/FT    |       | :         | HA<br>3. DE<br>0F<br>4. DE | AMMER<br>ENOTES INCHES OF PENETR.<br>F UNDISTURBED SAMPLER<br>ENOTES INCHES OF STONE & WERSTER ENG CO |            |

|                     |       |        |                |                  |                                       |             |                           | BORING SIS-2                                                                                                                                                                                                      |           |
|---------------------|-------|--------|----------------|------------------|---------------------------------------|-------------|---------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------|
| S                   | TE    | / LO   | CATI           | ON               | BEAVE                                 | LIAV A      | .ET PC                    | SHEET 2         OF 3           DHER STATION UNIT 2         J.O. NO. 12241                                                                                                                                         | _         |
| ELEVATION<br>(FEET) | DEPTH | (FEET) | SAMPLE<br>TYPE | SAMPLE<br>NUMBER | BLOWS (2)<br>OR UDA<br>PEDETRATION(3) | SPT N WULES | ande smart <sup>(5)</sup> | FIELD AND LAB.<br>SAMPLE DESCRIPTION TEST RESULTS /<br>COMMENTS                                                                                                                                                   |           |
|                     |       |        |                |                  |                                       |             |                           | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                             |           |
|                     |       |        | 5              | 6                | 676                                   | 13          | SX                        | TO 0.7 INCE MAINTER, COARGE TO FINE SAND, 25-25% SLIGHTLY PLASTIC FINES<br>BROWN.<br>SILTE SAND, WIDELI GRADED, 35-45% SUBROUNDED GRAVEL TO 0.7 INCH MAIDNON<br>MOSTLY FINE SAND, 12-15% NONPLASTIC FINES, BROWN. | (, .<br>- |
|                     |       |        | 5              | 7                | 8-11-15                               | 26          | GP<br>GH                  | SANDT GRAVEL, FOORLY GRADED, SUBANGULAR TO 1.0 INCH MAXIMUM, 30-40% MOS<br>COARSE AND NEDIUM SAND, 8-12% NONPLASTIC FINES, BROWN.                                                                                 | TL:       |
| 710                 | 20    |        | 5              | 8                | 13-22-24                              | 46          | GP<br>GM                  | SANDY GRAVEL, FOORLY GRADED, ANGULAR TO 1.0 INCH MAXIMUM, 25-30% MOSTLY<br>MEDIUM TO FINE SAND, 5-10% NOUPLASTIC FINES, DARK BROWN.                                                                               | -         |
|                     | 1     |        | s              | 9                | 26-30-35                              | 65          | GP<br>C24                 | SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                   |           |
| 705                 | 25    |        | s              | 10               | 56-100/4                              | 100         |                           | NO RECOVERY                                                                                                                                                                                                       | -         |
|                     |       |        | 8              | 11               | 7-7-11                                | 18          | SP                        | SAND, UNIFORM, MEDIUM TO FINE, MOSTLY MEDIUM LESS THAN 5% FINES, BROWN.                                                                                                                                           |           |
|                     |       |        | 8              | 12               | 6-7-10                                | 17          | SP-<br>Sm                 | SAND, UNIFORM, MEDIUM TO FINE, TRACE OF GRAVEL TO 5-LOS NONPLASTIC FINE<br>BROWN.                                                                                                                                 | s,        |
| 700                 | 30    | T      | 3              | 13               | 28-23-21                              | 44          |                           | NO RECOVERY                                                                                                                                                                                                       |           |
|                     |       | I      | s              | 14               | 9 <b>-8-</b> 14                       | 22          | GP                        | SANDY GRAVEL, FOORLY GRADED, ANGULAR (FRESHLY FRACTURED) TO 0.9 INCH<br>MAXDRIM, 20-30% MOSTLI FINE SAND, LESS THAN 5% FINES, BROWN.                                                                              | -         |
| 695                 | 35    |        | s              | 15               | 14-16-15                              | 31          | GP                        | SANDY GRAVEL, SIMILAR TO ABOVE.                                                                                                                                                                                   | -         |
|                     |       |        | s              | 16               | 21-31-27                              | 58          | 67-<br>GH                 | SANDY GRAVEL, FOORLY GRADED, ANGULAR TO 1.2 INCH MAXIMUM, 30-35% MOSTL<br>Fine Sand, 5-10% Nonplastic Fines, Brown.                                                                                               | r j       |
|                     |       |        | S              | 17               | 14-10-12                              | 22          | GP<br>GH                  | SANDT GRAVEL, FOORLY GRADED, ANGULAR TO SUBANGULAR TO 1.1 INCR MAXIMUM, 25-305 MUSTLI COARSE AND MEDIUM SAND, 5-85 NONPLASTIC FINES, DARK BROWN                                                                   | •         |
| <del>69</del> 0     | 40    |        | S              | 18               | 10-14-21                              | 35          | GP<br>GN                  | SANDT GRAVEL, SIDULAR TO ABOVE.                                                                                                                                                                                   | -         |
|                     |       |        | s              | 19               | 23-26-24                              | 50          | ан                        | <u>SILTI GRAVEL</u> , WIDELI GRADED, SUBANGULAR TO 1.1 INCH MAXIMUM, 25-30% MOS<br>FINE SAND, 12-15% NONPLASTIC FINES, BRJER.                                                                                     | πJ        |
| 685                 | 45    | -      | s              | ່<br>20          | 14-14-13                              | 27          | <b>C2H</b>                | SILTY GRAVEL, WIDELY GRADED, ANGULAR TO SUBANGULAR TO 1.0 INCH MAXIMUM<br>30-355 MOSTLY FINE SAND, 12-155 NONPLASTIC FINES, BROWN.                                                                                | ,         |

| e                   | TE / 1          | 100    | <b>AT</b> 14 | ON     | BEAVER                    | VALLEY      | PUW                        | R STATION UNIT                                   | 2                                                             | . J.O. NO                             | SHEET                                    |                             |
|---------------------|-----------------|--------|--------------|--------|---------------------------|-------------|----------------------------|--------------------------------------------------|---------------------------------------------------------------|---------------------------------------|------------------------------------------|-----------------------------|
| ELEVATION<br>(FEET) | DEPTH<br>(FEET) | SAMPLE | TYPE         | SAMPLE | BLOWS (2)<br>OR LDS       | SPT N WLIES | arour smeal <sup>(5)</sup> | SAMPLE                                           | DESCRIPTION                                                   |                                       | FIELD AN<br>TEST RES<br>COMMENT          | D LAB.<br>IULTS /<br>S      |
|                     |                 |        | 5            | 21     | y-11-22                   | 33          | сн                         | SILTY CRAVEL,                                    | SDALLAR TO ABOVE.                                             | •                                     |                                          |                             |
| ć <b>a</b> .        |                 |        | s            | 22     | 13-12-9                   | 21          | GP                         | SANDI GRAVEL,<br>15-20% COARSE                   | FOORLY GRADED, SU<br>TO FINE SAND, LES                        | BANGULAR, MOS<br>IS THAN 55 NOT       | STLI COARSE TO 1.2<br>NPLASTIC FIRES, BI | INCH MAXIMON                |
|                     | - v             |        | s            | 23     | 9-10-11                   | 21          | GH.                        | SILTY GRAVEL,<br>20-30% COARSE                   | WIDELY GRADED, AN<br>TO FINE SAND, 12-                        | IGULAR TO SUBL<br>-15% NONPLAST       | ANGULAR TO 1.1 INC<br>IC FINES, BROWN.   | H MAXIMUM,                  |
|                     |                 |        | S            | 24     | 11-8-9                    | 17          | GH                         | SILTI GRAVEL,                                    | SINILAR TO ABOVE.                                             | •                                     |                                          | -                           |
| 675                 | 55 -            |        | s            | 25     | 13-16-1                   | 7 33        |                            | NO RECOVERY                                      |                                                               |                                       |                                          | •<br>•<br>•                 |
|                     |                 |        | s            | 26     | 10-7-8                    | 15          | GP-<br>GM                  | SANDY GRAVEL,<br>COARSE AND MO<br>UNIFORM FINE S | FOORLY GRADED, SU<br>EDIUM SAND, 5-8% N<br>BAND, NEAR BOTTUM. | IBROUNDED TO (<br>IONPLASTIC FIR      | 0.9 INCH MAXIMUN,<br>NES, BROWN. 1 INCH  | 30-35% MOSTLI<br>I LAYER OF |
| ~                   |                 |        | s            | 27     | 8-16-38                   | 54          | GP-<br>GH                  | <u>SANDY GRAVEL</u> ,<br>MUSTLY FINE SA          | FOORLY GRADED, MO<br>IND, 5-10% MONFLAS                       | STLY ANGULAR                          | TO 1.0 INCE MAXIN<br>ROWN.               | ICH, 25-35%                 |
| 670                 | 6υ -            |        | s            | 28     | 2 <del>9-41</del> -1<br>4 | od ret      | СМ                         | SILTY GRAVEL,<br>30-35% COARSE                   | WIDELY GRADED, A:<br>TO FINE SAND, 12-                        | CULAR TO 1.2<br>185 NUNPLASTI         | INCH MAXIMUM (CAU<br>IC FINES, BROWN.    | GHT IN SHOE).               |
|                     |                 |        | S            | 29     | 141/4"                    | 140         | Ļ                          | SHALE, SEVERE                                    | Y WEATHERED, DECO                                             | MPOSED,                               |                                          |                             |
| 665                 | 65 _            |        |              |        |                           |             |                            | -                                                | TOP OF R                                                      | DCK AT 62.0'<br>DRING AT 62.4         | <b>T</b> .                               |                             |
|                     | -               |        |              |        |                           |             |                            |                                                  |                                                               | · · · · · · · · · · · · · · · · · · · | · · ·                                    | -                           |
|                     | -<br>-<br>-     |        |              |        |                           |             |                            |                                                  |                                                               |                                       |                                          | -                           |
|                     |                 |        |              |        |                           |             |                            |                                                  |                                                               |                                       |                                          |                             |
|                     |                 |        |              |        | × 400 J                   |             | L.                         |                                                  | 185                                                           |                                       |                                          | SHEET                       |

BVPS-2 UFSAR

| 6           |                                | TES 8         | 4095.0         |            | <u>z 6</u>        | 245.0 GROUND E                                           | LEV. (I)                                                                                    |
|-------------|--------------------------------|---------------|----------------|------------|-------------------|----------------------------------------------------------|---------------------------------------------------------------------------------------------|
| 1           | NGLINATIC                      | N             |                |            | BEA               | RING                                                     | LOGGED BY D. MACNETLL                                                                       |
| D           | ATE : ST                       | ART / F       | ÍNISH 🛛        | 27/77      | /                 | 7/29/77 CONTRACT                                         | OR / DRILLER                                                                                |
| S           | TATIC G                        | OUND          | WATER D        | EPTH       | / DA1             | re /                                                     | DRILL RIG TYPE                                                                              |
|             | EPTH TO                        | BEDI          | ROCK           |            | 58.               | TOTAL DEP                                                | TH DRILLED                                                                                  |
| N           | ETHODS                         | :             |                |            |                   |                                                          |                                                                                             |
|             | DRIL                           | LING          | 501L <u>4</u>  | " CASI     | NG, 3             | 7/8" ROLLER BIT, AN RODS,                                | MUD                                                                                         |
|             | SAM                            | PLING         | SOL            | 2" O.D.    | , SPL             | IT SPOON                                                 |                                                                                             |
|             | DRIL                           | LING F        | ROCK           |            |                   |                                                          |                                                                                             |
| S           | PECIAL                         | TESTIN        | IG OR INS      | STRUM      | ENT               | ATION                                                    |                                                                                             |
|             |                                |               |                |            |                   |                                                          |                                                                                             |
| R           | EMAKKS                         |               |                |            |                   | · · · · · · · · · · · · · · · · · · ·                    |                                                                                             |
|             |                                |               | -              |            |                   |                                                          | · · · · · · · · · · · · · · · · · · ·                                                       |
|             | 7                              |               |                |            | 12                |                                                          |                                                                                             |
| NOC         | EF                             | : u   u       | 8 (S) (E)      |            |                   | . · · ·                                                  | FIELD AND LAB.                                                                              |
| WT<br>EET   |                                |               |                | 2 2        | 3                 | SAMPLE DESCRIP                                           | TION TEST RESULTS /                                                                         |
| 13<br>13    |                                | 5   19        | ž o Š          |            | Į Ž               |                                                          | CCMMENTS                                                                                    |
|             | <u> </u>                       |               | 1 0            | 10.9       |                   | 1                                                        |                                                                                             |
|             | <u> </u>                       | r - r-        | · · · · ·      |            | 1                 | · · · · · · · · · · · · · · · · · · ·                    |                                                                                             |
|             | ]                              |               |                |            |                   |                                                          |                                                                                             |
|             | l j                            |               |                |            |                   | ``                                                       |                                                                                             |
|             |                                |               |                |            |                   |                                                          |                                                                                             |
|             |                                |               |                |            |                   | · · ·                                                    |                                                                                             |
|             | -                              |               |                |            |                   |                                                          |                                                                                             |
|             | 5                              | s             | 6-6-7          | 13         |                   | NO RECOVERY                                              |                                                                                             |
|             |                                |               |                |            | İ.                |                                                          |                                                                                             |
| 700         |                                |               | - 3-3-/        |            |                   | TOP 4 INCHES: SILTI SAND,<br>COARSE TO FINE, MOSTLY FIL  | WIDELY GRADED, 25-305 GRAVEL TO 0.6 INCH MA<br>NE SAND, 12-155 NONPLASTIC FINES, BROWNISH G |
|             | -                              | S             | 2              | <b> </b> ′ |                   | BOTTOM: SITLY CLAY, SLIGH                                | T TO MODERATELY PLASTIC, 8-125 FINE SAUD, BR                                                |
|             |                                |               |                |            |                   |                                                          |                                                                                             |
|             |                                | s             | 3 4-5-5        | 10         | GL                | CLAYEY SAND, WIDELY GRADE                                | D, 12-18% GRAVEL TO 0.6 INCH MAXIMUM, MOSTLY                                                |
|             | <u> </u>                       |               |                | ľ          | 1                 | SAND, 35-40% SLIGHT TO MOI<br>FOCKETS OF MODERATELY FLAS | DERATELY PLASTIC FINES, BROWN, WITH À FEW SMI<br>Stic <u>Silty Clay</u> , Throughout.       |
|             | <sup></sup>                    |               | 2-1-1          | 4          | _ ا               | SAUDY CLAY OF TOUR TO LODE                               |                                                                                             |
| 695         |                                | <b>.</b> .    | •              | °          | <b>1</b>          | WHILL WERE, SLIGHT TO HULE                               | MULICI FLADILU, LANCOS FINE SAND, BROWN, MOI                                                |
|             | -                              | $\square$     |                |            |                   |                                                          |                                                                                             |
|             |                                | s :           | 5 1-3-2        | 6          | CL                | SILTY CLAY, MODERATELY PLA                               | STIC, ONE PLECE OF SUBRUUNDED 0.5 INCH GRAVI                                                |
|             |                                |               |                |            |                   | 5-105 FIRE SAND, BROWN.                                  |                                                                                             |
|             | 15 -                           | s i           | 2-3-3          | 6          | CL                | SITLY CLAY, MODERATELY PLA                               | STIC, 8-125 FINE GRAVEL TO 0.5 INCH MAXIMUM,                                                |
|             |                                |               |                |            |                   | D-LUD FINE SAND, BRUWN.                                  | 1                                                                                           |
| u           | NDISTURB                       | ED SAN        | APLES (U.D.S   | 5.)        | L DA'<br>UN       | LESS OTHERWISE NOICATED                                  | BORING LOG                                                                                  |
|             | US_ SH                         | ELOY T        | UBE            |            | 2. BL             | OWS REQUIRED TO DRIVE                                    |                                                                                             |
| Ĩ           | U0_ 051                        | TERBERG       | i Uni<br>S     |            | DIS               | U.D. SAMPLE SPOON 6 ON<br>TANCE SHOWN USING 1401b.       | BEAVER VALLEY FOWER STATION UNIT 2                                                          |
| <u>'-</u> [ | UD_ DE                         | VISON<br>CHER |                |            | HA                | NMER FALLING 30"                                         | DUQUESHE LIGHT COMPANY                                                                      |
| 2 I         | UP Pri                         |               |                |            | <del></del>       |                                                          |                                                                                             |
| 2           | UP_ PIT                        | NETRA         | FION           |            | HA                |                                                          | PITTSBURGH, PENNSYLVANIA                                                                    |
| N N N       | UP_ PIT<br>i_STD PE<br>RESISTA | NETRAI        | TION<br>OWS/FT |            | HA<br>3. DE<br>OF | NAMER<br>NOTES INCHES OF PENETR.<br>UNDISTURBED SAMPLER  | PITTSBURGH, PENNSYLVANIA                                                                    |

|                     |                 |         |      |                  |                                        |             |                            | BORING _SIS-1                                                                                                                                                                        |
|---------------------|-----------------|---------|------|------------------|----------------------------------------|-------------|----------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| g                   | ITE /           | LOC     | AT   | ON               | BEAVER                                 | VALLI       | <u>a</u> 70'               | VER STATION UNIT 2 J.O. NO. 12241                                                                                                                                                    |
| ELEVATION<br>(FEET) | DEPTH<br>(FEET) | SA MPLE | TYPE | SAMPLE<br>NUMBER | BLOWS (2)<br>OR UDS.<br>PENETRATION(3) | SPT N WLUES | orour smeat <sup>(3)</sup> | FIELD AND LAB.<br>SAMPLE DESCRIPTION TEST RESULTS /<br>COMMENTS                                                                                                                      |
|                     |                 |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
| •••                 |                 |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
| <b>69</b> 0         |                 |         | s    | 7                | 1-2-3                                  | 5           | CL                         | STITY CLAY, SDULAR TO ABOVE.                                                                                                                                                         |
|                     |                 |         | s    | 8                | 2-4-5                                  | 9           | СН                         | SILTI CLAY, MODERTELY TO HIGHLY PLASTIC, LESS THAN 5% MOSTLY FINE SAND,<br>BROWN, MOIST, STIFF, HARD DRY STRENGTH.                                                                   |
| • •                 | 20              |         | s    | 9                | 3-3-5                                  |             | CL                         | TOP 4 INCRESS: SANDY CLAY, SLIGHT TO NODERATELY PLASTIC, 10-15% SUBROUND<br>CRAVEL TO 0.4 INCH MAXIMUM, 12-15% NOSTLY FINE SAND, GRAY, PIECE OF WOO<br>NELE BOTTOM.                  |
| 685                 |                 | ]       |      |                  |                                        |             |                            | TRACE OF ORGANIC MATERIAL THROUGHOUT.                                                                                                                                                |
|                     |                 |         | 5    | 10               | 2-2-7                                  | 1           |                            | DILIT WHAT, FUNDATION FUNDILG, GALAD GRAVEL (BLACK CUAL) LESS THAN 5% FINE SAND, GRAT.                                                                                               |
|                     | 25 -            |         | 3    | u                | 2-3-6                                  | 9           | CL                         | SILTI CLAY, MODERATELY PLASTIC, OCCASSIONAL GRAVEL TO 0.5 INCH MAXIMUM,<br>LESS THAN 5% FINE SAND, BROWNISH GRAY.<br>BOTTOM 7 INCHER: SILTI CLAY, MODERATELY FLASTIC, GRAVISH BROWN. |
| 6 <b>8</b> u        |                 | ]       | 5    | 12               | 3-4-7                                  | <u>n</u>    | CL                         | SILTY CLAY, MODERATIELY TO HIGHLY PLASTIC, BROWN.                                                                                                                                    |
|                     | -               | ]       |      | •-               | <b>-</b>                               | -           |                            |                                                                                                                                                                                      |
|                     |                 | ╡╽      | 5    | 13               | 5-7-10                                 | 17          | CL                         | SILTY GLAY, SIMILAR TO ABLVE.                                                                                                                                                        |
|                     | - 15            |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
|                     |                 |         | S    | 14               | 6-7-7                                  | 14          | CL                         | SILTI CLAY, SIMILAR TO ABOVE, EXCEPT OCCARSIONAL GRAVEL TO U.S INCH HAXIMUM.                                                                                                         |
| 675                 |                 |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
| 4.                  |                 | -       | S    | -15              | 5-7-8                                  | 15          | CH                         | SILT CLAY, HIGHTLY PLASTIC, TRACE OF SAND, BROWN, STIFF, HIGH DRI<br>STRENGTH.                                                                                                       |
|                     |                 | ] [     | 8    | 16               | 2-3-4                                  | 7           | CH                         | SILTY CLAY, HIGHLY FLASTIC TRACE OF FINE SAND. SOFT. BROWN.                                                                                                                          |
|                     | 35 -            |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
| 670                 |                 | 11      | s    | 17               | 3-5-6                                  | 11          | CH                         | SILTY CLAY, SIMILAR TO ABOVE EDEPT FIRM TO STIFF.                                                                                                                                    |
|                     | -               | 11      | -    | •1               |                                        |             |                            |                                                                                                                                                                                      |
|                     |                 |         | S    | 18               | 2-5-6                                  | l n         | CH                         | SILTY CLAY, HIGHLY PLASTIC, OCCASSIONAL GRAVEL TO 0.5 INCH MAXIMUM, LES<br>THAN 55 FIRE SAND, BROWN, STIFF, HIGH DRY STRENGTH.                                                       |
|                     | <sup>4</sup> "  |         | 5    | 19               | 3-4-7                                  | <u>п</u>    | CL                         | SILTY CLAY, MODERATELY TO HIGHLY PLASTIC, 5-10% GRAVEL TO 0.5 INCH MAXIMUM, 3-8% FINE SAND, BROWN.                                                                                   |
| 065                 |                 |         | S    | <b>2</b> U       | 2-3-5                                  | 8           | CI.                        | SANDY CLAY, HODERATELY FLASTIC, 12-18% FINE SAND, BROWN, FIRM.                                                                                                                       |
|                     |                 |         |      |                  |                                        |             |                            |                                                                                                                                                                                      |
| _                   | 45              | 1       | S    | 21               | 2-4-4                                  | L°          | L <sub>cr</sub>            | SANDI ULAI, SIMILAR TU ABUVE,                                                                                                                                                        |



### 2.5E-73

|                                                       |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           | 10                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                   |                                                                                 | 6919 6 -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 600 a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | SHEET I OF                                                                                                                                                                                                                                                                                      | 3                                                                                                   |
|-------------------------------------------------------|---------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------|---------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------|
| C                                                     | OORDIN                                                                                      | ATES                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | N.                                                                                                                                                                                                                                                                                                                        | 4240.0                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                   | <u> </u>                                                                        | GROUND EL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | LEV. (I)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | MAGNETLL.                                                                                                                                                                                                                                                                                       |                                                                                                     |
| 11                                                    | ICLINAT                                                                                     | 10N -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | 4.1                               | BEA                                                                             | RING                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | LOGGED BY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                 | _                                                                                                   |
| Ð                                                     | ATE : S                                                                                     | TART                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | / F16                                                                                                                                                                                                                                                                                                                     | IISH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                   | _′                                                                              | CONTRACTO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | DR / DRILLER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                 | _ '                                                                                                 |
| S                                                     | TATIC                                                                                       | GROUI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | NDW/                                                                                                                                                                                                                                                                                                                      | TER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | DEPTH                             | /DAT                                                                            | TE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | ORILL RIG TYPE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
| D                                                     | EPTH                                                                                        | TO BE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | DRO                                                                                                                                                                                                                                                                                                                       | ск _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                   | <u>71.3'</u>                                                                    | TOTAL DEP                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | TH DRILLED                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | <u>'1.5'</u>                                                                                                                                                                                                                                                                                    |                                                                                                     |
| M                                                     | ETHOD                                                                                       | <b>S</b> .:                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
|                                                       | DR                                                                                          | ILLING                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 50                                                                                                                                                                                                                                                                                                                        | WL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 4" CASI                           | NG, 3                                                                           | 7/8" HOLLER BIT, AN HODS, I                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | MUD                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | ·                                                                                                                                                                                                                                                                                               | <b>-</b> '                                                                                          |
|                                                       | SA                                                                                          | MPLIN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | G \$(                                                                                                                                                                                                                                                                                                                     | XL.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 2" 0.                             | D. Sr.                                                                          | LIT SPOON                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                                           |                                                                                                     |
|                                                       | DR                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | RO                                                                                                                                                                                                                                                                                                                        | CK _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | ·····                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                 | <del>.</del> .                                                                                      |
| S                                                     | PECIAL                                                                                      | TES                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | ring                                                                                                                                                                                                                                                                                                                      | ORI                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | NSTRUN                            | MENT!                                                                           | ATION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 | -                                                                                                   |
| _                                                     |                                                                                             | -                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | • <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                 | -                                                                                                   |
| R                                                     | EMARK                                                                                       | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 | -                                                                                                   |
|                                                       |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 | -                                                                                                   |
|                                                       |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
| N                                                     |                                                                                             | ы                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | wœ                                                                                                                                                                                                                                                                                                                        | ର ଜ                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | N SI SI                           | E,                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
| ET )                                                  | E TH                                                                                        | PE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 148                                                                                                                                                                                                                                                                                                                       | is a                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                   |                                                                                 | SAMPLE DESCRIPT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | TION                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | TEST PESLITE                                                                                                                                                                                                                                                                                    | ,                                                                                                   |
| E K                                                   |                                                                                             | SAK<br>TY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | S.                                                                                                                                                                                                                                                                                                                        | <u>2</u> ×                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                   | 5                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | COMMENTE                                                                                                                                                                                                                                                                                        |                                                                                                     |
| 5                                                     |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | \$ 8 B                            |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
|                                                       |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
|                                                       |                                                                                             | T ?                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |                                                                                                                                                                                                                                                                                                                           | <u> </u>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     |                                   | T                                                                               | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
|                                                       | -                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | ]                                                                                                                                                                                                                                                                                                                         | j                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | ÷                                                                                                                                                                                                                                                                                               |                                                                                                     |
|                                                       | ] ]                                                                                         |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1                                                                                                                                                                                                                                                                                               |                                                                                                     |
|                                                       |                                                                                             | •                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 | 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                 |                                                                                                     |
| :                                                     |                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                   |                                                                                 | SANDY CLAY, MODERATELY PLA                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | STIC, 8-125 GRAVEL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | TO 0.6 INCH MAXIMUM, 1                                                                                                                                                                                                                                                                          | -25%                                                                                                |
| 695                                                   |                                                                                             | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | . 4                               | CL                                                                              | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | STIC, 8-125 GRAVEL<br>SOFT, SATURATED TR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | TO 0.6 INCH MAILNUM, 1<br>ACES OF ORGALIC MATERIA                                                                                                                                                                                                                                               | _25 <b>%</b><br>L                                                                                   |
| 695                                                   |                                                                                             | s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | 4                                 | CL                                                                              | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROTOHOUT.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | TO 0.6 INCH MAXIMUM, 1<br>MGES OF ORGANIC MATERIA                                                                                                                                                                                                                                               | -25 <b>%</b><br>L                                                                                   |
| 695                                                   |                                                                                             | S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | . 2                               | CL                                                                              | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY_CLAY, SINCLAR TO ABO                                                                                                                                                                                                                                                                                                                                                                                                                                                         | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | TO 0.6 INCH MAXIMUM, 1<br>AGES OF ORGANIC MATERIA                                                                                                                                                                                                                                               | -25 <b>%</b><br>L                                                                                   |
| 695                                                   |                                                                                             | s<br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | - 2                               | CL<br>CL                                                                        | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO                                                                                                                                                                                                                                                                                                                                                                                                                                                         | NSTIC, 8-12% GRAVEL<br>, SOFT, SATURATED TR<br>NVE.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | TO 0.6 INCH MAXIMIM, 1<br>AGES OF ORGANIC MATERIA                                                                                                                                                                                                                                               | -25 <b>%</b><br>L                                                                                   |
| 695                                                   | ~<br>~                                                                                      | s<br>s<br>s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1                                                                                                                                                                                                                                                                                                                         | 2-2-3<br>1-1-1<br>2-3-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | . 2                               | CL                                                                              | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDCLAR TO ABO<br>NO RECOVERY                                                                                                                                                                                                                                                                                                                                                                                                                                          | STIC, 8-12% GRAVEL<br>SOFT, SATURATED TR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | TO 0.6 INCH MAXIMUM, 1<br>AGES OF ORGANIC MATERIA                                                                                                                                                                                                                                               | -25 <b>%</b><br>L                                                                                   |
| 695                                                   |                                                                                             | 3<br>  S<br>  S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 1                                                                                                                                                                                                                                                                                                                         | 2-2-3<br>1-1-1<br>2-3-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | 2                                 | CL<br>CL                                                                        | SANDY GLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDGLAR TO ABO<br>NO RECOVERY                                                                                                                                                                                                                                                                                                                                                                                                                                          | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | TO 0.6 INCH MAXIMUM, 15<br>AGES OF ORGANIC MATERIA                                                                                                                                                                                                                                              | -25 <b>%</b><br>L                                                                                   |
| 695<br>iSu                                            | 11 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                      | 3<br> <br> <br> <br> <br> <br>                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 1<br>2<br>3                                                                                                                                                                                                                                                                                                               | 2-2-2<br>1-1-1<br>2-3-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | . 2                               | CL<br>CL<br>BM                                                                  | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDULAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTY SAND,<br>U.7 INCH MAXIMUM, COARSE                                                                                                                                                                                                                                                                                                                                                                                 | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR<br>WIDELT GRADED, 20-4<br>TO FINE SAND, 35-447                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | TO 0.6 INCH MAXIMUM, 1:<br>ACES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NOH TO SLIGHTLY PLAST                                                                                                                                                                                       | -25 <b>%</b><br>L                                                                                   |
| 695<br>19u                                            | 2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2<br>2 | s<br>s<br>s                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 4<br>2<br>8<br>9                  | CL<br>CL<br>SM<br>ML<br>CL                                                      | SANDY CLAY, MODERATELY PLA<br>COARSE TO FILE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTY SAND,<br>U.7 INCH MALINOR, COARSE<br>FINES, BROWN, FOLLOWED BT<br>BROWN, BOTTOM 3 INCHESS: S                                                                                                                                                                                                                                                                                                                      | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR<br>NVE.<br>NVE.<br>10 FINE SAND, 35-403<br>13 INCH LATE OF SIL<br>ANDI SILTE NO FLAST                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | TO 0.6 INCH MAXIMUM, 1:<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NOW TO SLIGHTLY PLAS<br>LTY CLAY, SLIGHTLY PLAS<br>LTY CLAY, SLIGHTLY PLAS                                                                                                                                  | -25%<br>L                                                                                           |
| 695<br>iQu                                            | 12 12 12 12 12 12 12 12 12 12 12 12 12 1                                                    | S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | . 2<br>. 8                        | CL<br>CL                                                                        | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTY SAND,<br>U.7 INCH MAXIMUM, COARSE<br>FINES, BROWN, COLOMED BY<br>FINES, BROWN, COLOMED BY<br>SINCE, SOUTOM 3 INCHES: S<br>DARK BROWN.                                                                                                                                                                                                                                                                             | NUTE.<br>WIDELY GRADED, 20-4<br>TO FINE SAND, 35-4<br>MANDY SILT.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | TO 0.6 INCH MAXIMUM, 1:<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NON TO SLIGHTLY PLAST<br>IC, 20-30% UNIFORM, FIN                                                                                                                                                            | -25%<br>L<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>-<br>- |
| 695<br>iQu                                            | 14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1                                                      | S S S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | 1                                                                                                                                                                                                                                                                                                                         | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 2<br>8<br>16                      | CL<br>CL<br>CL                                                                  | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDCLAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILIT SAND,<br>U.7 INCH MAXIMUM, COARSE<br>FINES, BROWN, FOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARK BROWN.<br>SANDY CLAY, SLIGHT TO MODE                                                                                                                                                                                                                                                                          | NSTIC, 8-12% GRAVEL<br>, SOFT, SATURATED TR<br>WE.<br>WE.<br>TO FINE SAND, 35-447<br>: 3 INCE LAYER OF <u>\$1</u><br>AND <u>T SILT.</u> NO. FLAST.<br>ERATELY PLASTIC. 10-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | TO 0.6 INCH MAILMUM, 14<br>AGES OF ORGALIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>NOH TO SLIGHTLY PLAST<br>LTY CLAY, SLIGHTLY PLAST<br>1C, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAUL                                                                                                        | -25%<br>L<br>C<br>IC<br>TIC<br>E<br>SAX                                                             |
| 695<br>IQU                                            | 5 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2                                                     | S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1 3 4 5                                                                                                                                                                                                                                                                                                                   | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5<br>4-4-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2<br>8<br>16<br>7                 | CL<br>CL<br>SM<br>CL<br>CL                                                      | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDGLAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILIT SAND,<br>U.7 INCH MAXIMUM, COARSE<br>FINIS, BROWN, FOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARK BROWN.                                                                                                                                                                                                                                                                                                        | NSTIC, 8-125 GRAVEL<br>SOFT, SATURATED TR<br>WIDELI GRADED, 20-4<br>TO FINE SAND, 35-40<br>INCH LAYER OF SLI<br>ANDY SLIT. WONPLAST<br>RATELY PLASTIC, 10-1<br>TED, TRACE OF URGANIN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | TO 0.6 INCH MAXIMUM, 15<br>AGES OF ORGATIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NOH TO SLIGHTLY PLAS<br>1TT CLAY, SLIGHTLY PLAS<br>1C, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAUL<br>C MATERIAL.                                                                                        | -25%<br>L<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I |
| 695<br>IPU                                            | 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5                                                       | S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1 2 3 4 5                                                                                                                                                                                                                                                                                                                 | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5<br>4-4-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 2<br>8<br>16<br>7                 | CL<br>CL<br>CL<br>CL                                                            | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: STLIT SAND,<br>U.? INCH MAILMUN, COARSE<br>FINES, BROWN, FOLLOWED BY<br>BROWN, BOTTON 3 INCHES: S<br>DARK BROWN, SOFT, SATURAT                                                                                                                                                                                                                                                                                          | NSTIC, S-125 GRAVEL<br>, SOFT, SATURATED TR<br>, MIDELY GRADED, 20<br>TO FINE SAND, 35-407<br>: 3 INCH LAYER OF SI<br><u>ANDY SILT.</u> NO. FLAST<br>CRATELY FLASTIC, 10<br>TED, TRACE OF URGANIC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | TO 0.6 INCH MAXIMUM, 1:<br>ACES OF ORGATIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NOH TO SLIGHTLY PLAST<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SATT<br>C MATERIAL.                                                                                                                  | -25%                                                                                                |
| 695<br>990                                            | 5 21 22 22 22 22 22 22 22 22 22 22 22 22                                                    | S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1 2 3 4 5 6                                                                                                                                                                                                                                                                                                               | 2-2-2<br>1-1-1<br>2-3-3<br>8-7-5<br>4-4-3<br>3-3-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2<br>8<br>16<br>7<br>6            | CL<br>CL<br>CL<br>CL                                                            | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTY SAND,<br>U.7 INCH MALINGH, COARSE<br>FINES, BROWN, FOLLOWED BY<br>BROWN, SOTH 3 INCHES: S<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO                                                                                                                                                                                                                                                              | NETIC, 8-125 GRAVEL<br>, SOFT, SATURATED TR<br>NVE.<br>, MIDELI GRADED, 20-1<br>TO FINE SAND, 35-40<br>I JINCH LAYER OF SI<br>ANDI SILT. NONPLAST.<br>SRATELY PLASTIC, 10-1<br>TED, TRACE OF URGANIC<br>NVE.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | TO 0.6 INCH MAXIMUM, 1<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>% NUM TO SLIGHTLY PLASS<br>INT CLAY, SLIGHTLY PLASS<br>IC, 21-30% UNIFORM, FIN<br>15% COARSE TO FINE SAUL<br>C MATERIAL.                                                                                        | -25%                                                                                                |
| 695<br>190                                            | 10                                                                                          | S<br>S<br>S<br>S<br>BARR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | 1<br>2<br>3<br>4<br>5<br>6<br>EL 3                                                                                                                                                                                                                                                                                        | 2-2-4<br>1-1-1<br>2-3-3<br>8-7-9<br>4-4-3<br>3-3-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2<br>8<br>16<br>7<br>6            | CL<br>CL<br>CL<br>CL                                                            | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHESI SILTY SAND,<br>U.7 INCH MAXIMUM, COARSE<br>FINES, BROWN, SOLLOWED BY<br>FINES, BROWN, SOLLOWED BY<br>DARK BROWN, SOTT, SINCE<br>DARK BROWN, SOTT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO                                                                                                                                                                                                                                 | STIC, 8-125 GRAVEL<br>, SOFT, SATURATED TR<br>WVE.<br>, WIDELY GRADED, 20-4<br>TO FINE SAND, 35-40<br>(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)<br>(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NON TO SLIGHTLY PLAS<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAND<br>C MATERIAL.                                                                                                                   | -25%                                                                                                |
| 695<br>990<br>6855<br>9<br>0                          | 10                                                                                          | S<br>S<br>S<br>S<br>BARR<br>BED S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | 1<br>2<br>3<br>4<br>5<br>6<br>EL 3                                                                                                                                                                                                                                                                                        | 2-2-2<br>1-1-1-1<br>2-3-5<br>8-7-5<br>4-4-3<br>3-3-3<br>3-3-3<br>3-3-3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | 2<br>8<br>16<br>7<br>6            | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDCLAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTT SAND,<br>U.7 INCH MAXIMUM, COARSE<br>FINES, BOWN, TOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARE BROWN, SOFT, SATURAT<br>SANDY CLAY, SIDCLAR TO ABO<br>NUM IS MEAN SEA LEVEL<br>NLESS OTHERWISE NDCATED<br>OWS REQUIRED TO DRIVE                                                                                                                                                                                | STIC, 8-125 GRAVEL<br>, SOFT, SATURATED TR<br>WE.<br>WE.<br>VWE.<br>IN FINE SAND, 35-440<br>IN SILT. NO. FLAST<br>CANENT FLASTIC, 10-3<br>TED, TRACE OF URGANIC<br>WE.<br>BOR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGANIC MATERIA<br>25% SUBFOUNDED GRAVEL 1<br>NOH TO SLIGHTLY PLAS<br>10, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAUL<br>C MATERIAL.                                                                                                                      | -25%                                                                                                |
| 695<br>90<br>685<br>9                                 | 10                                                                                          | S<br>S<br>S<br>S<br>BARR<br>BARRED<br>S<br>HELBY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | 1<br>2<br>3<br>4<br>5<br>6<br>EL 3<br>5<br>1<br>0<br>1<br>1<br>1<br>9                                                                                                                                                                                                                                                     | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5<br>4-4-3<br>3-3-3<br>3-3-3<br>3-3-3<br>3<br>AMPLE<br>S(U.)<br>SE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | 2<br>8<br>16<br>7<br>6<br>D.S.)   | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIDCLAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILIT SAND,<br>U. 7 INCH MAILMUM, COARSE<br>FINES, BROWN, SOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIGHT TO MODE<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIGHTAR TO ABO<br>UTUM IS MEAN SEA LEVEL<br>NUESS OTHERWISE MODERATED<br>OWS REQUIRED TO DRIVE<br>O.D. SAMPLE SPOON 6" OR                                                                                      | STIC, 8-125 GRAVEL<br>, SOFT, SATURATED TR<br>WE.<br>WE.<br>TO FINE SAND, 35-447<br>: 3 INCE LAYER OF SI<br>ANDI SILT. NONPLAST<br>ERATELY PLASTIC, 10-3<br>TED, TRACE OF ORGANIC<br>WE.<br>BOR<br>BEAVER VAL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>NOH TO SLIGHTLY PLAST<br>LTY CLAY, SLIGHTLY PLAST<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAND<br>C MATERIAL.                                                                                         | -25%                                                                                                |
| 695<br>90<br>685<br>9                                 | 10                                                                                          | S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1<br>2<br>3<br>4<br>5<br>EL 3<br>SAMP<br>FUI<br>PISTO                                                                                                                                                                                                                                                                     | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5<br>4-4-3<br>3-3-3<br>3-3-3<br>3-3-3<br>3-3-3<br>5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>9-7-7<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-7<br>8-7-5<br>8-7-5<br>8-7-5<br>8-7-7<br>8-7-7<br>8-7-7<br>8-7-7<br>8-7-7<br>8-7-7<br>8-7 | 2<br>8<br>16<br>7<br>6<br>D.S.)   | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, MODERATELY PLA<br>COARSE TO FILE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILIT SAND,<br>U.7 INCH MAILMUM, COARSE<br>FINDS, BROWN, FOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARE BROWN, SOFT, SATURAT<br>SANDY CLAY, SINGLAR TO ABO<br>TOM IS MEAN SEA LEVEL<br>NUESS OTHERWISE NOCATED<br>OWS REQUIRED TO DRIVE<br>O.D. SAMPLE SPOON & CR<br>STANCE SHOWN USING 14016                                                                                                                         | STIC, 8-125 GRAVEL<br>SOFT, SATURATED TR<br>WIDELI GRADED, 20<br>TO FINE SAND, 35-407<br>INCE LAYER OF SI<br>ANDI SILT. 100:FLAST<br>ERATELY PLASTIC, 10<br>TED, TRACE OF ORCAND<br>VE.<br>BOR<br>BEAVER VAL<br>DUQUESNE 1.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | TO 0.6 INCH MATINUM, 14<br>AGES OF ORGALIC MATERIA<br>S NOH TO SLIGHTLY PLAST<br>LTY CLAY, SLIGHTLY PLAST<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAUL<br>C MATERIAL.                                                                                                                  | -25%                                                                                                |
| 695<br>90<br>685<br>9<br>1<br>1<br>100<br>1           | 10<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15                              | S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1<br>2<br>3<br>4<br>5<br>5<br>5<br>AMP<br>FUI<br>PISTO<br>ERG<br>R                                                                                                                                                                                                                                                        | 2-2-2<br>1-1-1<br>2-3-5<br>8-7-5<br>4-4-3<br>3-3-3<br>3-3-3<br>3-3-3<br>3<br>AMPLE<br>ES (U.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 2<br>8<br>16<br>7<br>6<br>0.5.)   | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES 1 STLIT SAND,<br>U.? INCH MAINMON, COARSE<br>FINES, BROWN, FOLLOWED BY<br>BROWN, BOTTON 3 INCHES: S<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO<br>TUM IS MEAN SEA LEVEL<br>NUESS OTHERWISE NONCATED<br>OWS REQUIRED TO DRIVE<br>O.D. SAMPLE SPOON & OR<br>STANCE SHOWN USING HOLD<br>NUMMER FALLING 30"<br>INDICATES USE OF 300 ID                                                                       | STIC, S-125 GRAVEL<br>SOFT, SATURATED TR<br>WIDELY GRADED, 20<br>TO FINE SAND, 35-407<br>INCH LAYER OF SI<br>ANDI SILT. 100:FLAST<br>ERATELY PLASTIC, 10<br>TED, TRACE OF ORCANIC<br>WE.<br>BOR<br>BEAVER VAL<br>DUQUESNE L                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | TO 0.6 INCH MAXIMUM, 15<br>AGES OF ORGATIC MATERIA<br>SOURCE DE ORGATIC MATERIA<br>SNUH TO SLIGHTLY PLASS<br>IC, 20-305 UNIFORM, FIN<br>15% COARSE TO FINE SATT<br>C MATERIAL.<br>ING LOG<br>LET FOWER STATION UNIT<br>LIGHT COMPANY                                                            | -25%                                                                                                |
| 695<br>990                                            | 5                                                                                           | S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1<br>2<br>3<br>4<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5                                                                                                                                                                                                                | 2-2-2<br>1-1-1-1<br>2-3-3<br>8-7-5<br>4-4-3<br>3-3-3<br>AMPLE<br>LES (U.<br>SE<br>IN<br>N<br>SS/FT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           | 2<br>8<br>16<br>7<br>6<br>D.S.)   | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, HODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: STLTT SAND,<br>U.7 INCH MAILTOR, COARSE<br>FINES, BROWN, FOLLOWED BY<br>BROWN, BOTHN J INCHES: S<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO<br>UTUM IS MEAN SEA LEVEL<br>NLESS OTHERWISE NONCATED<br>O.D. SAMPLE SPOON 6" OR<br>STANCE SHOWN USING HOIL<br>NAMMER FALLING 30"<br>HOUCATES USE OF 300 IB<br>NAMMER                            | STIC, 8-125 GRAVEL,<br>SOFT, SATURATED TR<br>WE.<br>WE.<br>NVE.<br>SATURATED TR<br>NUELI GRADED, 20-1<br>TO FINE SAND, 35-403<br>I JICH LAYER OF SI<br>ANDI SILT, WOWFLAST.<br>SRATELY PLASTIC, 10-1<br>TED, TRACE OF ORGANIC<br>VE.<br>BOR<br>BEAVER VAL<br>DEQUESNE L<br>PITTSBURCH                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGALLIC MATERIA<br>SOUTH OF ORGALLIC MATERIA<br>NUM TO SLIGHTLY PLASS<br>INT CLAY, SLIGHTLY PLASS<br>IC, 2J-305 UNIFORM, FIN<br>15% COARSE TO FINE SAUL<br>C MATERIAL.<br>ING LOG<br>LEY FOWER STATION UNIT<br>IGHT COMPANY<br>I, FEMISYLVANIA              | -255<br>L<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>I |
| 685 5 5 U M L                                         | 5                                                                                           | S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | 1<br>2<br>3<br>4<br>5<br>6<br>EL S<br>5<br>6<br>EL S<br>5<br>6<br>EL S<br>5<br>6<br>EL S<br>6<br>EL S<br>7<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5<br>5 | 2-2-2<br>1-1-1-1<br>2-3-3<br>8-7-5<br>44-3<br>3-3-3<br>3-3-3<br>3-3-3<br>3-3-3<br>5<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>8-7-5<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10<br>10                                                                                                                                                                                                                                                                                                                                                                                                                         | 2<br>8<br>9 16<br>7<br>6<br>0.5.) | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDY CLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDY CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHES: SILTY SAND,<br>U.7 INCH MAXIMUR, COARSE<br>FINIS, BROWN, SOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO<br>DARK BROWN, SOFT, SATURAT<br>SANDY CLAY, SIMILAR TO ABO<br>TO SAMPLE SPOON & OR<br>STANCE SHOWN USING HOIR<br>STANCE SHOWN USING HOIR<br>STANCE SHOWN USING HOIR<br>INDICATES USE OF 300 IB<br>NUMMER<br>INCHES OF PENETR.<br>UNDER INCHES OF PENETR.   | STIC, 8-125 GRAVEL,<br>SOFT, SATURATED TR<br>WE.<br>WE.<br>STO FINE SAND, 35-405<br>STO FINE SAND, 35-4 | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NOH TO SLIGHTLY PLAS<br>INT CLAY, SLIGHTLY PLAS<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SANT<br>C MATERIAL.<br>ING LOG<br>LET FOWER STATION UNIT<br>LIGHT COMPANY<br>1, FERIESTLVANIA              | -255<br>L<br>I<br>I<br>I<br>I<br>I<br>I<br>I<br>E<br>S<br>A<br>X                                    |
| 695 00 / NOTES 000 / 10 / 10 / 10 / 10 / 10 / 10 / 10 | 10<br>10<br>10<br>10<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15<br>15            | S<br>S<br>S<br>S<br>S<br>BARR<br>R8ED<br>S<br>TERB<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DENISO<br>DEN | 1<br>2<br>3<br>4<br>5<br>5<br>6<br>FLL S<br>5<br>5<br>AMP<br>PISTO<br>ERG<br>R<br>R<br>ATIO<br>BLOW<br>ER                                                                                                                                                                                                                 | 2-2-2<br>1-1-1-1<br>2-3-5<br>8-7-5<br>4-4-3<br>3-3-3<br>3-3-3<br>AMPL E<br>LES (U.<br>N<br>N<br>N<br>N<br>N<br>N<br>N<br>N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 2<br>8<br>16<br>7<br>6            | CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>CL<br>C | SANDT GLAY, MODERATELY PLA<br>COARSE TO FINE SAND, GRAY,<br>THROUGHOUT.<br>SANDT CLAY, SIMILAR TO ABO<br>NO RECOVERY<br>TOP 4 INCHESI SILTT SAND,<br>U.7 INCH MAXIMUR, COARSE<br>FINIS, BROWN, SOLLOWED BY<br>BROWN, BOTTOM 3 INCHES: S<br>DARK BROWN, SOTT, SATURAT<br>SANDT CLAY, SIMILAR TO ABO<br>DARK BROWN, SOTT, SATURAT<br>SANDT CLAY, SIMILAR TO ABO<br>TOWN REQUIRED TO DRIVE<br>O.D. SAMPLE SPOON 6" OR<br>STANCE SHOWN USING HOID<br>INDICATES USE OF 300 ID<br>INDICATES USE OF 300 ID<br>INMER<br>HOTES INCHES OF PENETR.<br>UNDISTURBED SAMPLE | STIC, 8-125 GRAVEL<br>SOFT, SATURATED TR<br>WIDELY GRADED, 20-4<br>TO FINE SAND, 35-40<br>ANDY SILT. NO. PLAST<br>GRATELY PLASTIC, 10-1<br>FRATELY PLASTIC, 10-1<br>FRATELY PLASTIC, 10-1<br>NVE.<br>BOR<br>BEAVER VAL<br>DUQUESNE L<br>PITTSBURCH<br>STONE &                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            | TO 0.6 INCH MAXIMUM, 14<br>AGES OF ORGANIC MATERIA<br>25% SUBROUNDED GRAVEL 1<br>\$ NUH TO SLIGHTLY PLAST<br>IC, 20-30% UNIFORM, FIN<br>15% COARSE TO FINE SAND<br>C MATERIAL.<br>ING LOG<br>LET FOWER STATION UNIT<br>ING LOG<br>LET FOWER STATION UNIT<br>ING FERISYLVANIA<br>WEBSTER ENG. CO | -25%                                                                                                |

12241 - GA(B) - 162A

|                     |                 |                |                   |                                       |             |           |                                                   | BORING SUS-4                                 |
|---------------------|-----------------|----------------|-------------------|---------------------------------------|-------------|-----------|---------------------------------------------------|----------------------------------------------|
|                     |                 |                |                   |                                       |             |           |                                                   | SHEET OF _                                   |
| S                   | ITE / L         | OCAT           | ION               | BEAVER                                | VALLE       | I FO      | WER STATION UNIT 2 J.O. NO. 12241                 |                                              |
| ELEVATION<br>(FEET) | DEPTH<br>(FEET) | 3AMPLE<br>TYPE | SAMPLE<br>NUMBER  | BLOWS (2)<br>OR UDS<br>PENETRATION(3) | SPT N WLIES | BRUP SMED | SAMPLE DESCRIPTION                                | FIELD AND LAB.<br>TEST RESULTS /<br>COMMENTS |
|                     |                 | E              | 5                 |                                       | ſ           |           |                                                   |                                              |
|                     |                 | s              | 7                 | 3-3-5                                 | 8           | ᇿ         | SILTY CLAY, NODERATELY PLASTIC, 8-125 FINE SAND,  | MEDIUM BROWN.                                |
|                     |                 | s              | 8                 | 2-2-3                                 | 5           | CL        | SILTY CLAY, SDULAR TO ABOVE.                      |                                              |
| 6 <b>8</b> J        | 20 -            |                |                   | 2-4-4                                 | 8           | CL        | SILTY CLAY, SIMILAR TO ABOVE.                     |                                              |
|                     |                 |                |                   |                                       |             |           |                                                   |                                              |
|                     |                 | S              | 10                | 2-3-4                                 | 7           | CL        | SILTY CLAY, STRULAR TO ABOVE.                     |                                              |
| 615                 | 25              | s              | <b>1</b> <u>1</u> | 2-2-3                                 | 5           | 대         | SILTY CLAY. SIMILAR TO ABOVE.                     |                                              |
|                     |                 | s              | -                 | 1-2-2                                 | 4           | CL.       | SILTI CLAY. SIMILAR TO ABOVE.                     |                                              |
|                     |                 |                |                   |                                       |             |           |                                                   | • *<br>•<br>•                                |
| 670                 |                 |                | 13                | (->-<                                 |             |           | SILI ULA, SILLAR IV ABOVE.                        |                                              |
|                     |                 | S              | 14                | 2-3-3                                 | 6           | CL        | SILTY CLAY, SDOLAR TO ABOVE.                      |                                              |
|                     |                 |                | 15                | 7-5-5                                 | 4           | GL        | SILTY CLAY, SIMILAR TO ABOVE.                     |                                              |
|                     |                 |                |                   | 1-2-2                                 | 4           | CL.       | SILTY CLAY, SINILAR TO ABOVE. BOTTOM & INCHES: (  | ORGAULC BROWL.                               |
| 665                 | 35 -            |                |                   |                                       |             |           |                                                   |                                              |
|                     |                 | S              | 17                | 1/18*                                 | 1           | CT.       | SILTY CLAY, MODERATELY PLASTIC, 3-65 FINE SAND,   | GRAY.                                        |
|                     |                 | s              | 18                | u/18"                                 | U           | a         | SILTY CLAY, SIMILAR TO ABOVE, EXCEPT LESS THAN S  | SAND.                                        |
| <del>66</del> 0     | 4               | s              | 19                | 1-1-2                                 | 3           | С         | SILTY CLAY, SIMILAR TO # 18.                      |                                              |
|                     |                 |                |                   | 1.2 6                                 |             |           |                                                   |                                              |
|                     |                 | S              | 20                | 5-5-5                                 | đ           |           | 2 THIN LAYERS.                                    | IDAN 79 FINE SAND, GR                        |
| 655                 | 45              | S              | 21                | 2-2-3                                 | 5           | ᇿ         | BANDI CLAY, SLIGHTLI TO MODERATELY PLASTIC, 15-20 | S FIRE SAND, GRAY.                           |

|                 |              |          |             |                                |            |                       |                                                                    |                                                                                      |                                      |                                        | BORING 2                                          |                                          |
|-----------------|--------------|----------|-------------|--------------------------------|------------|-----------------------|--------------------------------------------------------------------|--------------------------------------------------------------------------------------|--------------------------------------|----------------------------------------|---------------------------------------------------|------------------------------------------|
|                 |              | 0.04     |             | BEAVE                          | R VAL      | LEY F                 | UWER STATION UN                                                    | 17 2                                                                                 |                                      | 0 12241                                | SHEET_3                                           | . OF <u>3</u>                            |
| VATION<br>EET ) | EPTH<br>EETH | WPLE     |             | OWS (2)<br>R UDS<br>TRATION(3) | N VALUES   | P Smec <sup>(5)</sup> | SAMPLE                                                             | DESCRIPTION                                                                          |                                      | ·                                      | FIELD AND<br>TEST RESU                            | LAB.<br>.TS /                            |
| ELE<br>F        |              | <i>.</i> | 25          | Pin o Pin                      | SPT<br>T J | <b>BPOL</b>           |                                                                    |                                                                                      |                                      |                                        | COMMENTS                                          |                                          |
|                 | T            | · .      | - <u></u> - |                                | <b>r</b>   |                       | ,,                                                                 |                                                                                      |                                      |                                        |                                                   |                                          |
|                 |              | s        | 22          | 3-15-11                        | 24         | CL .                  | TOP 5 INCHES:<br>LAYER OF <u>SILT</u><br>BOTTUM 1 INCH:            | SANDY CLAY, SLIG<br>CLAY, MODERATEL<br>SANDY GRAVEL, 1.                              | HTLY PLAS<br>Y TO HIGH<br>2 INCH G   | TIC, 30-35<br>AY PLASTIC<br>RAVEL CAUG | S FINE SAND, G<br>, GRAY.<br>HT IN SHOE.          | RAY, 1 1:08                              |
|                 |              | s        | 23          | 7-9-10                         | 19         | 60                    | CLAYEY GRAVEL,<br>SAND, 35-40% S<br>SILTY CLAY, AN                 | WIDELY GRADED, S<br>LIGHTLY PLASTIC D<br>D A THIN LAYER OF                           | SUBROUNDE<br>FINES, GR<br>F BLACKISI | TO O.S II<br>NY, 1 I CH<br>I BROWN: OR | NCH MAXIMUM, 2<br>LAYER OF HIGH<br>GAPIC MATERIAL | 0-255 FINE<br>LY PLASTIC                 |
| 650             | 50           | s        | 24          | 6-10-12                        | 22         | GЖ                    | SILTY GRAVEL,<br>25-30% COARSE                                     | WIDELY GRADED, SU<br>TO FINE SAND, 12                                                | ubrounded<br>-185 NonPi              | TO SUBALG                              | ULAR TO 1.1 IN<br>ES, GRAY.                       | CH MAXIMUN,                              |
|                 |              | s        | 25          | 14-7-11                        | 18         | GM                    | SILTY GRAVEL,<br>25-30% COARSE                                     | WIDELY GRADED, AN<br>TO FINE SAND, 12-                                               | NGULAR (FI<br>-155 NON FI            | ACTURED)                               | TO U.S INCH MA                                    | XIMUM, -                                 |
| 645             | 55           | s        | 26          | 20-18-23                       | 41         | См                    | SILTY GRAVEL,                                                      | SIMILAR TO ABOVE.                                                                    | •                                    |                                        |                                                   | 111                                      |
|                 |              | s        | 27          | 9 <b>-8-</b> 7                 | 15         | Сан                   | SILTY GRAVEL,                                                      | SIMILAR TO ABOVE.                                                                    | •                                    |                                        |                                                   |                                          |
|                 |              | s        | 28          | 8-6-6                          | 12         | GP                    | SANDY GRAVEL,<br>25-35% MUSTLY                                     | FOORLY GRADED, SU<br>COARSE SAND, LES                                                | UBROUNDED<br>SS THAN 59              | to subang<br>Fines, di                 | ULAR TO 1.0 IN<br>ROWN.                           | CH MAXIMIM,                              |
| <b>64</b> 0     | 60           |          | s 29        | 30-16-18                       | 34         | Сэн                   | TOP 3 INCHES:<br>BOTTOM 1 INCH:<br>MUSTLY FINE SA                  | SAUDY CRAVEL, SIL<br>SILTY GRAVEL, SILTY GRAVEL, SILTY GRAVEL, SILTY GRAVEL, SICH PL | MILAR TO A                           | BOVE.<br>1 TO 1.0 I<br>25, BROWN.      | CH MAXIMUM, 2                                     |                                          |
| -               |              |          | s 30        | 16-20-19                       | 39         | GM                    | <u>SILTI GRAVEL</u> ,                                              | SIMILAR TO ABOVE                                                                     |                                      |                                        |                                                   | - T- T- T- T- T- T- T- T- T- T- T- T- T- |
| 635             | 65 -         | s        | 31          | 27-23-40                       | 63         | GP-<br>GM             | SANDY GRAVEL,<br>30-35% MOSTLY                                     | POORLY GRADED, S<br>MEDIUM SAND, 5-1                                                 | UBROUNDED<br>OS HONPLA               | TO SUBANG<br>STIC FINES                | ULAR TO 1.0 I<br>, BROWN.                         | NCH MAXIMUM                              |
|                 |              |          | s 32        | 22-23- <u>10</u><br>5"         | 132+       | GP-<br>GM             | SANDY GRAVEL.<br>MOSTLY FINE SA                                    | FOCRLY GRADED, S<br>ND, 5-8% NONPLAS                                                 | UBANGULAR<br>TIC FINES               | TO 1.2 IN<br>, BROWN.                  | CH MAXIMUN,                                       | 25-30 <b>≴ -</b><br>-                    |
|                 |              |          | s 33        | 39-56- <u>100</u><br>3"        | 156+       | SM                    | TOP 4 INCHES:<br>BOTTOM: SILTY<br>MAXIMUM MOSTLY<br>OF SANDY SILT, | <u>GRAVEL</u> ,<br><u>SAND</u> , WIDELY GRA<br>FINE SAND, 25-3<br>THROUGHOUT.        | DED, 25-3<br>5% NONPLA               | os Subroun<br>Stic Fines               | DED GRAVEL TO<br>, BROWN. SMAL                    | 0.7 INCH                                 |
| 630             | - 0          |          | s 34        | 50- <u>100</u><br>5"           | 100+       | GM                    | TOP 8 INCHES:<br>35-40% COARSE<br>BOTTOM 2 INCHE                   | SILTY GRAVEL, WI<br>TO FINE SAND, 15<br>S: GRAVEL, SEVER                             | DELY GRAD<br>                        | ED, SUBA<br>TO SLIGHTI<br>RED DECOMP   | GULAR TO 0.8<br>Y FLASTIC FIN<br>OSED, GRAY, S    | INCH MAXIMUM<br>ES, BROWN.<br>HALE.      |
|                 |              |          |             |                                |            |                       |                                                                    | TOP OF                                                                               | PROCK AT                             | 71.3'<br>T 71.5'                       |                                                   |                                          |
| OTF -           | FOP P0       |          | SUBMIAS     | AND I                          |            |                       |                                                                    |                                                                                      | UED BY                               | DATE N                                 | BORING NO.                                        | SHEET                                    |

|           |           |                    | N 43     | 30.0                                                                            |        | <b>z</b> 6     | 180.0 200 00 5                        | EV (1) 682.0                          | SHEET I OF              |
|-----------|-----------|--------------------|----------|---------------------------------------------------------------------------------|--------|----------------|---------------------------------------|---------------------------------------|-------------------------|
|           |           | 18123<br>100       |          |                                                                                 | _      |                |                                       | LOGOED BY G. ZAWAD                    | <b>A</b> ::             |
|           |           | - NN -             | / 61     |                                                                                 | 13-77  | OCA            | 6-13-77 CONTRACT                      | DE / DELLES MAINON                    | D/KODITEK               |
|           | DALE      | CRÓU               |          | tian 11                                                                         |        | /              |                                       |                                       |                         |
|           | SIALIC    | GRUU               |          |                                                                                 |        | 7 UA:<br>56.01 | TOTAL OF                              |                                       | 56.41                   |
|           | DEPTH     | TOB                | EDRO     | CK                                                                              |        |                | IOTAL DEP                             |                                       |                         |
|           | METHOU    | S :                |          |                                                                                 |        |                | 7/85 BOLLER BIT. AV BODS              | MITTO                                 |                         |
|           | DR        |                    | i 30     | <b>.</b>                                                                        | 28.0.0 | 97             | TT SHON                               |                                       |                         |
|           | SA        |                    | IG \$40  | ж. <u> </u>                                                                     |        |                |                                       | <u> </u>                              | <u> </u>                |
|           | DR        |                    | s RO     | CK                                                                              |        |                |                                       | · · · ·                               |                         |
|           | SPECIAL   | . 123              |          | URINS                                                                           | 1 KUM  | EN 1/          | A I IVII                              |                                       |                         |
|           |           |                    |          |                                                                                 |        |                | · · · · · · · · · · · · · · · · · · · | · .                                   | ·····                   |
| I         | REMARK    | s                  |          |                                                                                 |        |                |                                       |                                       | ·                       |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       | ······                  |
|           |           |                    | <u> </u> |                                                                                 |        |                |                                       |                                       |                         |
| R         |           | ц. ·               |          | ()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>()<br>( | S S    | <b>B</b> y     |                                       |                                       |                         |
| NIO<br>11 | PTH       | 13 d               | 14       | ) SM                                                                            | 138    | Ę              |                                       | - ION                                 | TEST RESULTS /          |
|           | DE<br>(FE | T Y                | SAN      | l 2 g E                                                                         | 28     | 5              |                                       |                                       | COMMENTS                |
| μ.        |           |                    | LĪ       |                                                                                 | 8 8    | ŝ              | · ·                                   |                                       |                         |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       |                         |
|           |           |                    | Ĺ.       | <u> </u>                                                                        |        |                |                                       |                                       | ······                  |
|           | -         |                    |          |                                                                                 |        |                |                                       |                                       | -                       |
| 400       |           |                    |          |                                                                                 |        |                |                                       |                                       | •                       |
|           | -         |                    |          |                                                                                 |        |                |                                       |                                       | _                       |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       |                         |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       |                         |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       |                         |
|           | ? ]       |                    |          | Į                                                                               |        |                |                                       |                                       |                         |
|           | -         |                    |          |                                                                                 |        | 1              |                                       |                                       |                         |
| 675       |           |                    | 1        |                                                                                 |        | 1              |                                       |                                       |                         |
| •         |           |                    |          | Į.                                                                              |        |                | ,                                     | н.<br>Г                               | · · · · · ·             |
|           |           |                    |          |                                                                                 |        |                |                                       |                                       |                         |
|           |           | ł                  |          |                                                                                 |        |                |                                       | and the second second                 | -                       |
|           | 10 _      |                    |          | · ·                                                                             |        |                |                                       |                                       | · · · · ·               |
|           |           |                    | Ι,       | 1-1-1                                                                           | 2      | HL             | SANDY SILT, HODERATELY FLA            | STIC, 10-20% VERY FINE S.             | IND, BROWN              |
|           |           | ١Ľ                 | Į-       | · ·                                                                             | 1      | 1              |                                       |                                       |                         |
| 670       | -         |                    | ł        |                                                                                 |        | 1              |                                       |                                       | -                       |
|           |           | s                  | 2        | 1/18"                                                                           | 1      | HE             | SANDY SILT, SIMILAR TO AB             | DVT.                                  |                         |
|           |           |                    | ł        | 1                                                                               |        |                |                                       |                                       |                         |
|           |           |                    | 1.       | 1.7.84                                                                          | ,      |                |                                       | 6-166 WEEK MANN MANN                  |                         |
| - 1       |           | s                  | 13       | -/ -0                                                                           | 1      | <u> </u>       | AUD GRAY                              | J-LJP VARI FINE SAND. T               | WILL OF UNDERICS, BROWN |
| 4         | S. SPLIT  |                    |          |                                                                                 |        | 1. DA'         | TUM IS MEAN SEA LEVEL                 | BABINA                                | 100                     |
|           | US_       | SHELBY             | TU       | 523 (0.0.3<br>BE                                                                |        | 2. BL          | OWS REDURED TO DRIVE                  |                                       |                         |
| 5         |           | FIXED              | PISTO    | )N -                                                                            |        | 2"             | O.D. SAMPLE SPOON S" OR               |                                       |                         |
| 5         | U0_ (     | JS TERO<br>DENISON | eng<br>i |                                                                                 |        | HA             | MMER FALLING 30"                      | DUCUESNE LIGHT                        | COMPANY                 |
| ž         | UP_1      | TCHE               | R        |                                                                                 |        | ¥              | NDICATES USE OF 300 Ib                | PITTSBURGH. PEN                       | NSTLVANIA               |
|           | RESIS     | TANCE              | BLOW     | /s<br>/s/ft                                                                     |        | на<br>3. de    | NOTES INCHES OF PENETR.               | · · · · · · · · · · · · · · · · · · · |                         |
| 2 -       | ⊻         |                    | C D .    |                                                                                 |        | OF             | UNDISTURBED SAMPLER                   |                                       |                         |
|           |           |                    |          |                                                                                 |        | - 06           | NULLA INUNEA UT                       | I AN STONE & WEE                      | STER ENG. CORP.         |
| B         | - GROU    |                    |          |                                                                                 |        | UN             | DISTURBED SAMPLE                      |                                       |                         |

|                     |                 |        |      |                  |                    |            |                 |                                                                                                                                                               | BORING                                                               |
|---------------------|-----------------|--------|------|------------------|--------------------|------------|-----------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|
|                     |                 |        |      |                  |                    |            |                 |                                                                                                                                                               | SHEET _2 OF _3_                                                      |
| S                   | ITE / L         | .004   | 110  | ON               | BEAVE              | R VALL     |                 | LER STATION UNIT 2 J.O. NO                                                                                                                                    | 122/1                                                                |
| ELEVATION<br>(FEET) | DEPTH<br>(FEET) | SAMPLE | TYPE | SAMPLE<br>NUMBER | PENETRATION(3)     | SPT N WLLE | aroup small     | SAMPLE DESCRIPTION                                                                                                                                            | FIELD AND LAB.<br>TEST RESULTS /<br>COMMENTS                         |
|                     | <b></b>         | гŦ     | - 1  | ,                |                    | <b>.</b>   |                 | ······································                                                                                                                        |                                                                      |
| 605                 |                 |        | 5    | 4.               | 1/18"              | 1          | ML              | SILT, MODERATELY PLASTIC, LESS THAN 5% VERY 1                                                                                                                 | TINE SAND, GRAY.                                                     |
|                     |                 |        | s    | 5                | 2/2/2              | 4          | HL.             | SANDY SILT, MODERATELY PLASTIC, 15-25% VERY 1                                                                                                                 | TINE SAND, GRAY.                                                     |
| 660                 | 20              |        | s    | 6                | 2/1/2              | 3          |                 | NO RECOVERS                                                                                                                                                   |                                                                      |
|                     |                 |        | s    | 7                | 1/18"              | 1          | ML              | SILTI SAND, UNIFORM, VERT FINE, 30-40% SLIGHT                                                                                                                 | LY PLASTIC FILES, GRAI.                                              |
|                     | 25              |        | s.   | 8                | 1/5/3              | 8          | CL              | CLAVEY SAND, UNIFORM, VERY FINE, 30-405 NODES<br>SAND, TOP 3 INCHES: SIMILAR TO SE, REMAINING                                                                 | - GRADING TO SAND, FOORLY                                            |
| 977                 | -               |        | 5    | 9                | 4/ 44/ 44          | ľ          | SM              | GRADED, FINE TO COARDE, MUSILI FINE & MEDIUM C<br>& GRAY.                                                                                                     | HAVEL, TRACES OF WOOJ, BROWN                                         |
|                     |                 |        | 5.   | 10               | 8-11-19            | 30         |                 | NO RECOVERY                                                                                                                                                   |                                                                      |
| 650                 | -               |        | 5    | n                | 11-13-11           | 1 24       |                 | NO RELOVERT                                                                                                                                                   |                                                                      |
|                     |                 |        | S    | 12               | 7-10-15            | 25         | SC              | <u>CLAYEY SAND</u> , WELL GRADED, PIECE OF GRAVEL CA                                                                                                          | UGET IN SHOE.                                                        |
|                     | 35              |        | s    | 13               | 10-9-5             | 14         | SW              | <u>GRAVELLY SAND</u> , WELL GRADED, 20-30% GRAVEL TO<br>COARSE SAND, LESS THAN 5% NONFLASTIC FINES, O                                                         | 1.2 INCH MAXIMUM, FINE TO<br>RAY.                                    |
| 646                 |                 |        | 5    | 14               | 9-10-9             | 19         |                 | NO RECOVERY                                                                                                                                                   |                                                                      |
|                     | -<br>40 -       |        | s    | 15               | 3-3-5              | 8          |                 | NO RECOVERY                                                                                                                                                   |                                                                      |
| 640                 | -               |        | S    | 16               | 9 <del>-9-</del> 7 | 16         | SM              | SILT SAND, FINE TO COARSE, LESS THAN 5% GRAV                                                                                                                  | EL.                                                                  |
|                     |                 | s<br>- |      | 17               | 17-20-19           | 39         | SW<br>SM<br>SW- | PLASTIC FINES, 30-40% GRAVEL TO 14 INCH, YELL<br>BOTTOM 3 INCHES: LESS TEAN 5% FINES, 30-40%<br>PIECES, GRAY.<br>GRAVELY SAND, WELL GRADED, FINE TO COARSE, 5 | OW-BROWN.<br>GRAVEL FRACTURED, 3/4 INCH<br>-10% NON-SLIGHTLY PLASTIC |
|                     | 45              |        | S    | 18               | 18-13-11           | 24         | SM              | FINES, 30-40% GRAVEL TO 1+ INCH, GRAY AND BRO                                                                                                                 |                                                                      |

12241 - GA(B) - 1638

|                     |                            |            |          |                |             | -               | SHEET 3 OF 3                                                                                                                                                                                       |
|---------------------|----------------------------|------------|----------|----------------|-------------|-----------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| ELEVATION<br>(FEET) | DEPTH<br>(FEET)            |            | SAMPLE & | PENETRATION(3) | SPT N WLUES | arcue small (3) | JO. NO<br>FIELD AND LAB.<br>SAMPLE DESCRIPTION TEST RESULTS /<br>COMMENTS                                                                                                                          |
|                     |                            |            |          | · · ·          | т-          | <u> </u>        | 1                                                                                                                                                                                                  |
| 635                 | 111                        | <b>S</b> . | 19       | 11-11-9        | 20          | GW              | SANDY GRAVEL, FIRE TO COARSE, MOSTLY FIRE & MEDIUM (TO 1. INCH FRACTURED<br>LESS THAN 105 MON-SUIGHTLY PLASTIC FIRES, SOME LENSES OF SLIGHTLY PLASTIC<br>FIRES, 30-405 FIRE TO COARSE SAND, BROWN. |
|                     |                            | s          | 20       | 12-11-1        | 1 22        |                 | NO RECOVERT                                                                                                                                                                                        |
|                     | 2<br>7<br>7<br>7<br>7<br>7 | \$         | 21       | 12-10-2        | 0 30        | sv              | GRAVELLY SAND, FINE TO COARSE, WELL GRADED, 10-15% HON-SLIGHTLY PLASTIC<br>FINES, 20-30% GRAVEL FRAMENTS TO 14 LHCE, BROWN.                                                                        |
| (                   | TLI                        | s          | 22       | 39-40-4        | 0 80        | SP-<br>Sm       | SILTY SAND, FOORLY GRADED, FINE TO COARSE, MOSTLY FINE & MEDIUM, 105<br>NON-SILICHTLY FLASTIC FD:ES, 10-155 GRAVEL TO 1 INCH, BROWN.                                                               |
| 000                 | -<br>55 -<br>-             | S          | 23       | 50-75          | 754         | SP-<br>SM       | <u>SILTY SAND</u> , SAME AS ABOVE.                                                                                                                                                                 |
|                     |                            | 8          | 24       | 100/5*         | 1004        |                 | VEATHERED SHALE.                                                                                                                                                                                   |
| -                   |                            |            |          |                |             |                 | TOP OF NOCK AT 56.0'<br>END OF BORING AT 56.4'                                                                                                                                                     |
|                     |                            |            | -        |                |             |                 |                                                                                                                                                                                                    |
|                     |                            |            |          |                |             |                 |                                                                                                                                                                                                    |
|                     |                            |            |          |                | -           |                 |                                                                                                                                                                                                    |
|                     | OR BO                      | <u>ا</u>   |          | -              |             | 1               |                                                                                                                                                                                                    |

Rev. 0

| NTE.                                              | DRILL                                                                                                                                                                                          | E VALLEY PORE AT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          | TIDE<br>OCATIC                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | - <u>DNIT</u><br>DN <u>- SNI</u><br>DR                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | <u>10. 1</u><br>201101<br>N.LEO                         | 99.1. | 40. No. 11700 SOR ING No. 101                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |
|---------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------|-------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| LEV.<br>887                                       | SEPTS<br>FEET                                                                                                                                                                                  | 87867A<br>86883:071cm                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | -lef                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | ::                                                      |       | LASSALTON OF SECONDER                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |
|                                                   | 080                                                                                                                                                                                            | ND 8L. 200.61                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 10 1                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | BORING                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | 10.                                                     |       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                                                   |                                                                                                                                                                                                | DARK BROWN SANDY<br>CLAY WITH SOME<br>GRAVEL, COMPACT,<br>HOIST<br>DARK BROWN SANDY<br>CLAY, STIFF<br>TRACE OF YCLAY,<br>TRACE OF YCLAY,<br>TRACE OF YCLAY,<br>DARK GRAY CLAY<br>WIT<br>DARK GRAY CLAY,<br>WIT<br>DARK GRAY SANDY<br>SOME SAID, WET<br>DARK BROWN SAID<br>SOME SILT, VET                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | Non Link         78           NTL         78           NTL         78           NTL         23           NTL         24           NTT         200           NTL         100           NTL |                                                         | . 보   | SILT - SANDY, FIRM, BROWN<br>SAME AS STI<br>CLAY - FIRM, LEAN, SILTY, TRACE OF SANDSTONE<br>FRACHENTE, BROWN<br>SAME AS STI<br>SAME AS STI<br>SAME AS STI<br>CLAY - FIRM, LEAN, ORGANIC, SILTY, UNIFORM, GRAY<br>SAME AS STI<br>SAME AS |  |
|                                                   |                                                                                                                                                                                                | AATO, SOME MEDICT<br>GRAVEL AND SILT,<br>WET                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | 12 43<br>11 56                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                         |       | SAND - MEDIUM AND COARSE, SILTY, SOME GRAVEL,<br>UNIFORM, ROUNDED, DARK GRAY<br>SAME AS SSIL<br>SAME AS SSIL                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |
|                                                   |                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                         |       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
|                                                   |                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |                                                         |       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |
| L FIG<br>10 10 10 10 10 10 10 10 10 10 10 10 10 1 | UNIES 00           BLOURS 0           DRIVE 0           DRIVE 0           PRANEE 0           NOVER 1           NOVER 1           NOVER 1           NOVER 1           NOVER 1           NOVER 1 | But on Accovery only only $\gamma = 10 \cdot 0$ backets $\gamma$ is a 10 \cdot 0 backet $\gamma$ or backet some $\gamma$ for the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second secon | UN OF NO<br>LLLING<br>LLING<br>U 12<br>SPORALTI<br>UPP PER<br>LEF<br>THE<br>NO<br>UPP PER<br>LEF<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>SOL<br>ST<br>ST<br>SOL<br>ST<br>ST<br>SOL<br>ST<br>ST<br>ST<br>SOL<br>ST<br>ST<br>ST<br>ST<br>ST<br>ST<br>ST<br>ST<br>ST<br>ST | TE THE I<br>IC " ALE<br>OH THE<br>AND THE<br>AND THE<br>AND THE<br>AND THE<br>AND THE<br>AND THE<br>AND THE<br>AND THE                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | NARCH<br>WIES<br>MES<br>MES<br>MES<br>MES<br>MES<br>MES |       | BORING LOG 301<br>BEAVER VALLEY POWER STATION - UNIT NG<br>GNIPPINGPORT, PENESYLVANIA<br>DOQUESKE LIGHT COMPARY<br>STONE & WESSTER GUNGHERENNG COMPORATION<br>LIJOO-EME-19                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |

| 8 WEBSTER               | ENGINEERING CO                                                                                                                             | RP.                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | TP-1                                                                                                                                                                                                                                                                                                                      |  |
|-------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| AVER VALLEY I           | POWER STATION - UNI                                                                                                                        | T 2                                                                                                                                                                                                                               | J.O. NO.<br>12241.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   | 00                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | SHEET<br>1 OF 1                                                                                                                                                                                                                                                                                                           |  |
| T<br>CKHOE – CASE       | 780B                                                                                                                                       | LOCATION<br>N 3885.7, E6204.3                                                                                                                                                                                                     | 3                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | GROUND ELEV.<br>733.5                                                                                                                                                                                                                                                                                                     |  |
| AVATED<br>GUST 11, 1982 | 2                                                                                                                                          | CONTRACTOR<br>DICK CORPORATION                                                                                                                                                                                                    | · · · · · · · · · · · ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              | LOGGED BY<br>J. W. MCCOY                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         |                                                                                                                                                                                                                                                                                                                           |  |
| DEPTH<br>(FEET)         |                                                                                                                                            | SAMP                                                                                                                                                                                                                              | LE DESCRIPTIO                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        | )N                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                           |  |
| _                       | CLAYEY SILT/SILT,                                                                                                                          | SLIGHTLY PLASTIC, MOIST,                                                                                                                                                                                                          | LIGHT BROWN WIT                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | TH GRAY MOTTLING. (C                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                             | L/ML)                                                                                                                                                                                                                                                                                                                     |  |
| -                       |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
| 5                       |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  | -                                                                                                                                                                                                                                                                                                                         |  |
|                         |                                                                                                                                            |                                                                                                                                                                                                                                   | ·                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
| . –                     | SILTY SAND/SAND,                                                                                                                           | UNIFORM, FINE, BROWN (SM-                                                                                                                                                                                                         | SP).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
| 10                      | SILT, NON TO SLIC                                                                                                                          | GHTLY PLASTIC, VERY MOIST,                                                                                                                                                                                                        | SOME LAYERING,                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | LIGHT BROWN (ML).                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |                                                                                                                                                                                                                                                                                                                           |  |
| -                       | SAND, TRACE FINE                                                                                                                           | GRAVEL, MEDIUM-FINE SAND,                                                                                                                                                                                                         | FEW NODULES OF                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       | SILT, MOIST, BROWN                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               | (SP)                                                                                                                                                                                                                                                                                                                      |  |
| -                       |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
| 15                      |                                                                                                                                            | BOTTOM OF TES<br>GROUNDWATER N                                                                                                                                                                                                    | T PIT: 14 FT.<br>OT ENCOUNTERED.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                     | N.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                           |  |
| -                       |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
|                         |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      | an an an an Araba<br>An Araba an Araba<br>An Araba                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                           |  |
|                         |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
| -                       |                                                                                                                                            |                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |                                                                                                                                                                                                                                                                                                                           |  |
|                         | A         WEBSTER           AVER VALLEY I           CXHOE - CASE           AVATED           GUST 11, 198           DEPTH           (FEET ) | B WEBSTER ENGINEERING CO<br>AVER VALLEY POWER STATION - UNI<br>CKHOE - CASE 780B<br>AVATED<br>GUST 11, 1982<br>DEPTH<br>(FEET)<br>- CLAYEY SILT/SILT,<br>- SILTY SAND/SAND,<br>- 10 - SILT, NON TO SLIC<br>- SAND, TRACE FINE<br> | B WEBSTER ENGINEERING CORP.<br>AVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY POWER STATION - UNIT 2<br>INVER VALLEY SILT/SILT, SLIGHTLY PLASTIC, MOIST,<br>INVER VALLEY SILT/SILT, SLIGHTLY PLASTIC, MOIST,<br>INVER VALLEY SAND/SAND, UNIFORM, FINE, BROWN (SM-<br>INVER VALLEY SAND, TRACE FINE GRAVEL, MEDIUM-FINE SAND,<br>IS<br>IS<br>IS<br>IS<br>IS<br>IS<br>IS<br>IS<br>IS<br>IS | A WEBSTER ENGINEERING CORP.<br>VVER VALLEY POWER STATION - UNIT 2<br>I.O. NO.<br>12241.<br>I.O. NO.<br>12241.<br>I.O. NO.<br>12241.<br>I.O. NO.<br>12241.<br>I.O. NO.<br>12241.<br>I.O. NO.<br>12241.<br>N. 3885.7, E6204.3<br>AWATED<br>DICK CORPORATION<br>DEPTH<br>OFFIT<br>OFFIT<br>I.O. NO.<br>I.O. NO.<br>12241.<br>NO. NO.<br>I.O. NO.<br>12241.<br>NO. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I.O. NO.<br>I. | A WEBSTER ENGINEERING CORP.<br>IVER VALLEY FOWER STATION - UNIT 2<br>LOCATION N 3885.7, E6204.3<br>ANATED<br>CONTRACTOR DICK CORPORATION J. W. MCCOY<br>DEPTH<br>(FEET)<br>CLAYEY SILT/SILT, SLIGHTLY PLASTIC, MOIST, LIGHT BROWN WITH GRAY MOTTLING. (C<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S<br>S |  |

| BITE    | BEAVER VALL           | EY POWER STATION - I                                                                                                                 | UNIT 2                                                                         |                                                                                | J.O. NO.<br>12241                                                                  | .00                               | SHEET                 | 1 0F 1                            |
|---------|-----------------------|--------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------|--------------------------------------------------------------------------------|------------------------------------------------------------------------------------|-----------------------------------|-----------------------|-----------------------------------|
| QUIPMEI | NT<br>BACKHOE – C     | ASE 780B                                                                                                                             | LOCATION<br>N39                                                                | 949.6, E6167.                                                                  | .7                                                                                 |                                   | GROUND ELEV.<br>733.5 |                                   |
| ITE EX  | CAVATED<br>AUGUST 11, | 1982                                                                                                                                 | CONTRACTOR<br>DIC                                                              | CK CORPORATIO                                                                  | ON                                                                                 | LOGGED BY                         | . W. MCCOY            |                                   |
| LEV.    | DEPTH<br>(FEET)       |                                                                                                                                      |                                                                                | SAMPL                                                                          | E DESCRIPTIO                                                                       | ON                                |                       |                                   |
| 13.5    |                       | SANDY SILT, BROWN                                                                                                                    | N                                                                              |                                                                                |                                                                                    |                                   |                       |                                   |
|         |                       | <u>SILT</u> , NONPLASTIC                                                                                                             | TO SLIGHTLY PLA                                                                | ASTIC, TRACE                                                                   | FINE SAND, VER                                                                     | Y MOIST, LIGHT                    | BROWN. (ML).          | -                                 |
|         |                       | -                                                                                                                                    |                                                                                |                                                                                |                                                                                    |                                   |                       |                                   |
|         | 5                     |                                                                                                                                      |                                                                                |                                                                                |                                                                                    |                                   |                       |                                   |
|         |                       |                                                                                                                                      |                                                                                |                                                                                |                                                                                    |                                   |                       | _                                 |
|         | ]                     | CANDY STIT                                                                                                                           |                                                                                |                                                                                |                                                                                    |                                   |                       | -                                 |
|         |                       | - SANDY SILT                                                                                                                         | -<br>                                                                          |                                                                                |                                                                                    | ·                                 |                       |                                   |
|         |                       | - <u>SANDY SILT</u><br>- <u>SAND</u> , TRACE FINI                                                                                    | E GRAVEL, MEDIUN                                                               | 4 TO FINE SAN                                                                  | ND, BROWN. (SP)                                                                    | •                                 |                       |                                   |
|         | 10                    | <u>SANDY SILT</u><br><u>SAND</u> , TRACE FINI                                                                                        | E GRAVEL, MEDIUM                                                               | 1 TO FINE SAN                                                                  | ND, BROWN. (SP)                                                                    | •                                 |                       |                                   |
|         | 10                    | - <u>SANDY SILT</u><br><u>SAND</u> , TRACE FINI<br>- <u>SILT</u> , NONPLASTIC                                                        | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA                                            | 4 TO FINE SAN<br>ASTIC, VERY N                                                 | ND, BROWN. (SP)<br>MOIST, MODERATE                                                 | Ly Stiff to Sof                   | T, LIGHT BROW         |                                   |
|         | 10                    | - <u>SANDY SILT</u><br>SAND, TRACE FINI<br>- <u>SILT</u> , NONPLASTIC<br><u>SAND</u> , 10-15% GRAY<br>GRAYISH BROWN. (S              | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).                 | 4 TO FINE SAN<br>ASTIC, VERY M<br>SUBANGULAR,                                  | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE                               | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROWN        |                                   |
|         | 10                    | <u>SANDY SILT</u><br><u>SAND, TRACE FINI</u><br><u>SILT, NONPLASTIC</u><br><u>SAND, 10-15% GRAY</u><br><u>GRAYISH BROWN. (S</u>      | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).                 | 4 TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,                                  | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE                               | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROW         |                                   |
|         | 10                    | - <u>SANDY SILT</u><br><u>SAND</u> , TRACE FINI<br><u>SILT</u> , NONPLASTIC<br><u>SAND</u> , 10-15% GRAY<br>GRAYISH BROWN. (S        | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT         | 4 TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H                 | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"                 | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROW         | N. (ML).                          |
|         | 10                    | <u>SANDY SILT</u><br><u>SAND, TRACE FINI</u><br><u>SILT, NONPLASTIC</u><br><u>SAND, 10-15% GRAY<br/>GRAYISH BROWN. (S</u>            | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT<br>GROU | A TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H<br>JNDWATER NOT | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"<br>ENCOUNTERED. | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROW         |                                   |
|         | 10                    | - <u>SANDY SILT</u><br><u>SAND</u> , TRACE FINI<br><u>SILT</u> , NONPLASTIC<br><u>SAND</u> , 10-15% GRAY<br><u>GRAYISH BROWN.</u> (S | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT<br>GROU | 4 TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H<br>JNDWATER NOT | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"<br>ENCOUNTERED. | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROW         | N. (ML).                          |
|         | 10                    | <u>SANDY SILT</u><br><u>SAND, TRACE FINI</u><br><u>SILT, NONPLASTIC</u><br><u>SAND, 10-15% GRAY</u><br><u>GRAYISH BROWN. (S</u>      | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT<br>GROU | A TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H<br>JNDWATER NOT | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"<br>ENCOUNTERED. | LY STIFF TO SOF                   | T, LIGHT BROW         | -<br>N. (ML).<br>-<br>-<br>-<br>- |
|         | 10                    | <u>SANDY SILT</u><br><u>SAND, TRACE FINI</u><br><u>SILT, NONPLASTIC</u><br><u>SAND, 10-15% GRAV</u><br><u>GRAVISH BROWN. (S</u>      | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT<br>GROU | 4 TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H<br>JNDWATER NOT | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"<br>ENCOUNTERED. | LY STIFF TO SOF<br>SAND, TRACE CO | T, LIGHT BROW         |                                   |
|         | 10                    | SANDY SILT<br>SAND, TRACE FINI<br>SILT, NONPLASTIC<br>SAND, 10-15% GRAY<br>GRAYISH BROWN. (S                                         | E GRAVEL, MEDIUM<br>TO SLIGHTLY PLA<br>VEL, ROUNDED TO<br>SP).<br>BOTT<br>GROU | A TO FINE SAN<br>ASTIC, VERY N<br>SUBANGULAR,<br>TOM OF TEST H<br>JNDWATER NOT | ND, BROWN. (SP)<br>MOIST, MODERATE<br>COARSE TO FINE<br>PIT: 13'9"<br>ENCOUNTERED. | LY STIFF TO SOF                   | T, LIGHT BROW         |                                   |

| STONE          | 8 WEBSTE               | R ENGINEERING C                       | ORP.                                                |                                        |                                       | TP-3                                                                                                             |  |  |  |
|----------------|------------------------|---------------------------------------|-----------------------------------------------------|----------------------------------------|---------------------------------------|------------------------------------------------------------------------------------------------------------------|--|--|--|
| SITE           | AVER VALLE             | Y POWER STATION - UN                  | NIT 2                                               | J.O. NO.                               | •1                                    | SHEET<br>1 OF 1                                                                                                  |  |  |  |
| EQUIPMEN<br>B/ | T<br>CKHOE – CAS       | SE 780B                               | LOCATION N3949.9, E6112.2                           |                                        |                                       | GROUND ELEV. 733.5                                                                                               |  |  |  |
| DATE EXC       | AVATED<br>IGUST 11, 19 | 982                                   | CONTRACTOR<br>DICK CORPORATIO                       | )N                                     | LOGGED BY J. W. N                     | . MCCOY                                                                                                          |  |  |  |
| ELEV.          | DEPTH<br>(FEET)        |                                       | SAMPLE DESCRIPTION                                  |                                        |                                       |                                                                                                                  |  |  |  |
| 733.5          |                        | SAND, SILT, GRAVE                     | L, SLAG GRAY.                                       | ······································ | · · · · · · · · · · · · · · · · · · · |                                                                                                                  |  |  |  |
|                | -                      | SILT, 5-10% FINE<br>FEW LARGE ROUNDED | SAND, SLIGHTLY PLASTIC, MO<br>GRAVEL TO 6 IN. (ML). | DERATELY STIFF :                       | IO SOFT, VERY MOIS                    | T, LIGHT BROWN,                                                                                                  |  |  |  |
|                | - 5 -                  | - · · ·                               |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                | •                      | -                                     |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                | -                      | <u>SAND</u> , 5-7% COARSE             | TO FINE GRAVEL, MEDIUN T                            | 0 FINE SAND, MOI                       | ST, BROWN. (SP).                      |                                                                                                                  |  |  |  |
| . '            |                        | -                                     |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                | 10 -                   | -                                     |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        | <u>SILT</u> , NON TO SLIG             | HTLY PLASTIC, MODERATELY                            | STIFF, VERY MOIS                       | T. (ML)                               |                                                                                                                  |  |  |  |
|                |                        | <u>SAND</u> , 5-7% FINE G             | RAVEL, COARSE TO FINE SAN                           | D, TRACE NONPLAS                       | TIC FINES, VERY M                     | DIST. BROWN. (SP-SW)                                                                                             |  |  |  |
| · · ·          | 15 -                   |                                       | BOTTOM OF TEST P<br>Groundwater not                 | IT: 13.5 FT. (CA<br>ENCOUNTERED.       | VING)                                 |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        |                                       |                                                     |                                        |                                       |                                                                                                                  |  |  |  |
|                |                        | 1                                     | · · ·                                               |                                        |                                       |                                                                                                                  |  |  |  |
|                | and the second         |                                       |                                                     |                                        |                                       | and the second second second second second second second second second second second second second second second |  |  |  |
|                |                        |                                       |                                                     |                                        | •                                     | · · · · · · · · · · · · · · · · · · ·                                                                            |  |  |  |

|               | & WEBSTER               |                                            | ORP.                                                  | J.O. NO.                                      | <u></u>                               | TP-4                  |  |  |  |  |  |
|---------------|-------------------------|--------------------------------------------|-------------------------------------------------------|-----------------------------------------------|---------------------------------------|-----------------------|--|--|--|--|--|
| BE            | EAVER VALLEY            | POWER STATION - UN                         | NIT 2                                                 | 1                                             | 2241.00                               | 1 OF 1                |  |  |  |  |  |
| QUIPMEN<br>BA | IT<br>ACKHOE – CASE     | 780B                                       | LOCATION N3895, E6240                                 | , <u>, , , , , , , , , , , , , , , , , , </u> |                                       | GROUND ELEV.<br>733.5 |  |  |  |  |  |
| ATE EXC       | CAVATED<br>EPTEMBER 21, | 1982                                       | CONTRACTOR DICK CORPORA                               | TION                                          | LOGGED BY                             | ICCOY/D. HUNT         |  |  |  |  |  |
|               | DEPTH<br>(FEET)         | SAMPLE DESCRIPTION                         |                                                       |                                               |                                       |                       |  |  |  |  |  |
| 33.5          | _                       | FILL, SLAG, CINI                           | DERS, GRAY.                                           |                                               |                                       |                       |  |  |  |  |  |
| Ì             |                         | SANDY CLAY/SILTY<br>MOTTLED BROWN AN       | CLAY, STIFF, MOIST, COA<br>ND GRAY. (CL).             | RSE TO FINE GRAV                              | EL SIZED WEATHERE                     | D SHALE FRAGMENTS,    |  |  |  |  |  |
|               | 5                       | CLAYEY SILT/SILT<br>SIZED WEATHERED        | TY CLAY, SLIGHTLY TO MODE<br>SHALE, 15-20% FINE SAND, | RATELY PLASTIC,<br>BROWN. (CL).               | STIFF, MOIST, SOM                     | E FINE GRAVEL         |  |  |  |  |  |
|               | _                       | SIMILAR TO ABOVI                           | E, GRAY.                                              |                                               | · · · · · · · · · · · · · · · · · · · |                       |  |  |  |  |  |
|               | -                       | <u>SAND</u> , 10-15% FIN<br>Estimate water | NE GRAVEL, ROUNDED, MDEIL<br>CONTENT AT 7-8% (SP).    | M TO FINE SAND,                               | 5% NONPLASTIC FIN                     | ES, DAMP, BROWN,      |  |  |  |  |  |
|               | 10                      |                                            |                                                       |                                               |                                       |                       |  |  |  |  |  |
|               | . 1                     |                                            |                                                       |                                               |                                       |                       |  |  |  |  |  |
|               | -                       |                                            |                                                       |                                               |                                       | •                     |  |  |  |  |  |
|               |                         |                                            | BOTTOM OF TEST<br>Groundwater No                      | PIT: 15 FT.<br>T ENCOUNTERED                  |                                       |                       |  |  |  |  |  |
|               |                         |                                            |                                                       |                                               | · ·                                   |                       |  |  |  |  |  |
|               | · · · · ·               |                                            |                                                       |                                               |                                       |                       |  |  |  |  |  |
| н             |                         |                                            | •<br>•                                                |                                               |                                       |                       |  |  |  |  |  |
|               | -                       |                                            |                                                       |                                               |                                       |                       |  |  |  |  |  |
| 1             |                         |                                            |                                                       |                                               |                                       |                       |  |  |  |  |  |

| eld<br>Ione | TEST PIT I<br>& WEBSTE | LOG<br>R ENGI | NEERING CO  | RP.          |                                         |                                       |                                            | TEST PIT NO.<br>TP-5 |  |  |
|-------------|------------------------|---------------|-------------|--------------|-----------------------------------------|---------------------------------------|--------------------------------------------|----------------------|--|--|
| TE B        | EAVER VALLES           | POWER         | STATION - U | NIT 2        |                                         | J.O. NO. 1224                         | 41                                         | SHEET<br>1 OF 1      |  |  |
| UIPMEN      | IT                     |               |             | LOCATION     |                                         |                                       |                                            | GROUND ELEV.         |  |  |
| B           | ACKHOE - CA            | SE 780        | 8           |              | N3825, E6320                            |                                       |                                            | 731.6                |  |  |
|             | CAVATED                |               |             | CONTRACTOR   | ITRACTOR J. W. MCCOY/D. HUNT            |                                       |                                            |                      |  |  |
| LEV.        | DEPTH<br>(FEET)        |               | ·           |              | SAMPI                                   | LE DESCRIPTIO                         | DN.                                        |                      |  |  |
| 1.6         | -                      | SANDY         | SILT, VERY  | DENSE, OCCAS | IONAL GRAVEL T                          | 0 2 IN., ROUNDED                      | , MOTTLED GRAY AND                         | BROWN. (ML).         |  |  |
|             |                        | SILTY         | CLAY/CLAYEY | SILT, STIFF  | TO VERY STIFF                           | , MODERATELY PL                       | ASTIC, BROWN. (CL)                         | •                    |  |  |
|             | - 5                    | SAND,         | FINE, 7-10% | NONPLASTIC   | FINES, DAMP, B                          | ROWN.                                 |                                            |                      |  |  |
|             | · · –                  | SILTY         | SAND, FINE, | DENSE, WET,  | GRAY BROWN. (1                          | DILATIVE) (SM).                       |                                            |                      |  |  |
| `           |                        |               | · · ·       |              |                                         | · · · · · · · · · · · · · · · · · · · |                                            |                      |  |  |
|             | _                      |               |             |              |                                         |                                       |                                            |                      |  |  |
|             | 10 -                   |               |             |              |                                         | . *                                   | ter en en en en en en en en en en en en en | -                    |  |  |
| :           | · · · -                | SAND,         | 10-15% FINE | GRAVEL, ROU  | NDED, MEDIUM T                          | 0 FINE SAND, 5%                       | NONPLASTIC FINES,                          | MOIST, BROWN, (SP).  |  |  |
|             | -                      |               |             | <u> </u>     | BOTTOM OF<br>GROUNDWATE                 | TEST PIT: 13 FT<br>R NOT ENCOUNTER    | ED.                                        |                      |  |  |
|             | 15 -                   | 1             |             |              |                                         |                                       |                                            |                      |  |  |
|             |                        |               |             |              |                                         |                                       |                                            |                      |  |  |
|             |                        |               |             | · .          | . · · · · · · · · · · · · · · · · · · · | * <sup>1</sup>                        |                                            |                      |  |  |
|             | -                      |               |             | · .          |                                         |                                       | 2                                          |                      |  |  |
|             | _                      | 1             |             |              |                                         |                                       |                                            |                      |  |  |
|             | -                      | 4             |             |              |                                         |                                       |                                            |                      |  |  |
| 1. A. A.    | -                      | 4             |             |              |                                         |                                       |                                            |                      |  |  |
|             | -                      | 4             |             |              |                                         |                                       |                                            |                      |  |  |
|             | ant e traine t         | 1             |             |              |                                         |                                       |                                            |                      |  |  |
|             | l                      | <u> </u>      |             |              |                                         |                                       |                                            |                      |  |  |

| ELD                                   | TEST PIT                                                                                                                                                                                                                                             | LOG               |                             |                  |                    | TEST         | PIT NO.                               |          |
|---------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------|-----------------------------|------------------|--------------------|--------------|---------------------------------------|----------|
| TONE                                  | & WEBSTE                                                                                                                                                                                                                                             | ER ENGINEERING    | CORP.                       |                  |                    |              | TP-6                                  |          |
| TE                                    |                                                                                                                                                                                                                                                      |                   |                             | J.O. NO.         |                    | SHEET        |                                       |          |
| BE                                    | AVER VALLE                                                                                                                                                                                                                                           | Y POWER STATION - | UNIT 2                      |                  | 12241              |              | 1 OF 1                                |          |
| UIPMEN                                | τ                                                                                                                                                                                                                                                    |                   | LOCATION                    |                  |                    | GROU         | ID ELEV.                              |          |
| BA                                    | CKHOE - CA                                                                                                                                                                                                                                           | SE 780B           | N3775, E6                   | 400              |                    |              | 733.4                                 |          |
| TE EXC                                | AVATED                                                                                                                                                                                                                                               |                   | CONTRACTOR                  |                  | LOGGED BY          |              |                                       |          |
|                                       | PTEMBER 21                                                                                                                                                                                                                                           | , 1982            | DICK CORP                   | ORATION          |                    | J. W. MCCOY  |                                       |          |
|                                       | DEPTH                                                                                                                                                                                                                                                |                   | · · · · ·                   | SAMPLE DESC      | RIPTION            |              |                                       |          |
| LEV.                                  | (FEET)                                                                                                                                                                                                                                               | ,,,,,,,,,         |                             |                  |                    |              | · · · · · · · · · · · · · · · · · · · |          |
| , , , , , , , , , , , , , , , , , , , |                                                                                                                                                                                                                                                      | FILL, SLAG, CIND  | ERS                         |                  |                    |              | · · · · · · · · · · · · · · · · · · · |          |
|                                       |                                                                                                                                                                                                                                                      | -                 |                             |                  |                    | · ·          |                                       | -        |
|                                       |                                                                                                                                                                                                                                                      | SILTY SAND, TRAC  | E COARSE TO FINE GRAV       | EL, OCCASIONAL L | ARGE ROUNDED COBBI | E, FINE SAND | , 30-40%                              | -        |
|                                       |                                                                                                                                                                                                                                                      | THICK DAMP BRO    | , CUNTAINS LAYERED ZU<br>MN | NES OF CLAYEY SI | LT AND FINE SAND,  | 174 TO I IN. |                                       | -        |
| 1                                     | 5 -                                                                                                                                                                                                                                                  |                   |                             |                  |                    | •            | a.                                    | -        |
|                                       | -                                                                                                                                                                                                                                                    |                   |                             | •                | · · · ·            |              |                                       | -        |
|                                       |                                                                                                                                                                                                                                                      | 4                 |                             |                  |                    |              |                                       |          |
|                                       |                                                                                                                                                                                                                                                      | <b>_</b>          |                             |                  |                    |              | •                                     | -        |
|                                       |                                                                                                                                                                                                                                                      | _                 |                             |                  |                    |              |                                       | -        |
|                                       | 10 -                                                                                                                                                                                                                                                 |                   |                             |                  |                    |              |                                       | _        |
|                                       |                                                                                                                                                                                                                                                      | <b></b>           |                             |                  |                    |              |                                       |          |
| ŀ                                     |                                                                                                                                                                                                                                                      | 4                 |                             |                  |                    |              |                                       | -        |
|                                       |                                                                                                                                                                                                                                                      | _                 |                             |                  | 2 <sup>- 1</sup>   |              |                                       | -        |
|                                       |                                                                                                                                                                                                                                                      | · ·               | •                           |                  |                    |              |                                       | _        |
|                                       | 15 -                                                                                                                                                                                                                                                 |                   | ·                           |                  |                    |              |                                       |          |
|                                       | 1.5                                                                                                                                                                                                                                                  |                   | BOTTOM                      | OF TEST PIT: 15  | FT.                |              |                                       | _        |
| l                                     |                                                                                                                                                                                                                                                      | ]                 | GROUNDW                     | ATER NOT ENCOUNT | ERED.              |              |                                       | _        |
|                                       |                                                                                                                                                                                                                                                      |                   |                             |                  |                    |              |                                       | _        |
| 1.1                                   |                                                                                                                                                                                                                                                      |                   |                             |                  |                    |              |                                       |          |
|                                       |                                                                                                                                                                                                                                                      | 7                 |                             |                  |                    |              |                                       |          |
|                                       | ан на станата.<br>При на станата на станата на станата на станата на станата на станата на станата на станата на станата на стана<br>При на станата на станата на станата на станата на станата на станата на станата на станата на станата на стана | 7                 |                             |                  |                    |              |                                       | _        |
|                                       |                                                                                                                                                                                                                                                      | 1                 |                             |                  |                    |              |                                       |          |
|                                       |                                                                                                                                                                                                                                                      | 1                 |                             | ·                |                    |              |                                       | -        |
| 1. I                                  | e de la composition de la composition de la composition de la composition de la composition de la composition de                                                                                                                                     | 1                 |                             |                  |                    |              |                                       | -        |
|                                       |                                                                                                                                                                                                                                                      | 4                 |                             |                  |                    |              |                                       |          |
| ł                                     |                                                                                                                                                                                                                                                      | - <b>-</b>        |                             |                  |                    |              |                                       | <u> </u> |

FORM 6-13-0 FIELD TEST PIT LOG TEST PIT NO. TP-7 STONE & WEBSTER ENGINEERING CORP. SITE J.O. NO. SHEET 12241 BEAVER VALLEY POWER STATION - UNIT 2 1 of 1 EQUIPMENT BACKHOE - CASE 780B GROUND ELEV. 734.2 LOCATION N3745, E6375 DATE EXCAVATED CONTRACTOR LOGGED BY SEPTEMBER 21, 1982 DICK CORPORATION J. W. MCCOY/D. HUNT DEPTH SAMPLE DESCRIPTION ELEV. (FEET) 734.2 CLAYEY SILT, DENSE, 15-20% FINE GRAVEL SIZED SHALE FRAGMENTS, BROWN. SILTY SAND, FINE, 20-40% NONPLASTIC FINES, OCCASIONAL SILT LAYERS, DAMP, DENSE, BROWN. 5 10 SILTY SAND, FINE, 15-20% NONPLASTIC FINES, OCCASIONAL COBBLE, ROUNDED, LIGHT GRAY, GRADING TO SAND, FINE, TRACE COARSE TO FINE GRAVEL AND OCCASIONAL LARGE COBBLE, DAMP (W=6%), MEDIUM DENSE TO DENSE. BOTTOM OF TEST PIT: 14 FT. 15 GROUNDWATER NOT ENCOUNTERED.

ł

| TONE          | TEST PIT L<br>8 WEBSTEI | LOG<br>R ENGINEERING CO | ORP.                            |                                           |                                  | TEST PIT NO.<br>TP-8                   |  |
|---------------|-------------------------|-------------------------|---------------------------------|-------------------------------------------|----------------------------------|----------------------------------------|--|
| ITE<br>BI     | EAVER VALLEY            | POWER STATION - UN      | NIT 2                           | J.O. NO. 12                               | 241                              | SHEET<br>1 OF 1                        |  |
| QUIPMEN<br>B/ | ACKHOE - CASI           | E 780B                  | LOCATION                        |                                           |                                  | GROUND ELEV.                           |  |
| TE EX         | CAVATED<br>EPTEMBER 21, | 1982                    | CONTRACTOR<br>DICK COR          | RPORATION                                 | LOGGED BY<br>J. W. MCCOY/D. HUNT |                                        |  |
| LEV.          | DEPTH<br>(FEET)         |                         | Si                              | AMPLE DESCRIPT                            | ION                              |                                        |  |
|               | -                       | STRUCTURAL              | <u>. FILL</u> , FOR SWS PIPELIN | NE.                                       |                                  |                                        |  |
|               |                         |                         |                                 |                                           |                                  |                                        |  |
|               | -                       | SILTY CLAY              | Y, SLIGHTLY TO MODERATI         | ELY PLASTIC, STIFF,                       | MOIST, MOTTLED GRAY              | – BROWN. –                             |  |
|               | 10 -                    | CLAYEY SII              | LT, TRACE COARSE TO FIN         | NE GRAVEL, 20-30% F                       | INE SAND, BROWN. (DR             | HER, LESS PLASTIC , -                  |  |
|               | -                       | GRAVELLY S              | SAND, 20-25% COARSE TO          | FINE GRAVEL, 5-107                        | NONPLASTIC FINES, F              | ROWN.                                  |  |
|               | - 15 -                  |                         |                                 |                                           |                                  | •••                                    |  |
| 5. I          | -                       |                         | BOTT<br>GROU                    | DM OF TEST PIT: 15<br>NDWATER NOT ENCOUNI | FT.<br>TERED.                    | •••••••••••••••••••••••••••••••••••••• |  |
|               | -                       |                         | en.<br>Notes                    |                                           | •                                |                                        |  |
| ·             | -                       | 4                       |                                 |                                           |                                  |                                        |  |
|               | ľ · _                   | ]                       |                                 |                                           |                                  |                                        |  |

# APPENDIX B LABORATORY TESTING

# APPENDIX B

# TABLE OF CONTENTS

| <u>Sectio</u> | n <u>Title</u>                              | <u>Page</u> |
|---------------|---------------------------------------------|-------------|
| B1            | INTRODUCTION                                | B-1         |
| B2            | INDEX TESTS                                 | B-1         |
| B2.1          | GRAIN SIZE ANALYSES                         | B-1         |
| B2.2          | ATTERBURG LIMITS AND NATURAL WATER CONTENTS | B-1         |
| B3            | CONSTANT RATE OF STRAIN CONSOLIDATION TESTS | B-1         |
| B4            | TRIAXIAL COMPRESSION TESTS                  | B-1         |
| B5            | DIRECT SHEAR TESTS                          | B-2         |
| B6            | REFERENCES                                  | B-2         |

B-i

# APPENDIX B

## LIST OF TABLES

# <u>Table</u>

# <u>Title</u>

- B-1 Atterburg Limits and Natural Water Contents
- B-2 Summary of Consolidated Isotropically Undrained Compression Tests
- B-3 Summary of Direct Shear Tests

### APPENDIX B

## LIST OF FIGURES

| <u>Figure</u> | Title                                                                         |
|---------------|-------------------------------------------------------------------------------|
| B-1           | Gradation Curves: Boring EOS-1, Sample S-7                                    |
| B-2           | Gradation Curves: Boring EOS-1, Sample S-12                                   |
| B-3           | Gradation Curves: Boring EOS-1A, Sample US4E                                  |
| B-4           | Gradation Curves: Boring EOS-6, Samples S-2 and S-14                          |
| B-5           | Gradation Curves: Boring EOS-6, Samples S-12 and S-15                         |
| B-6           | Plasticity Chart                                                              |
| B-7           | Consolidation Test Report: Boring EOS-4, Sample US1B5                         |
| B-8           | Triaxial Test Strength Summary: Boring EOS-1A, Samples US1D, US1E, and US1F   |
| B-9           | Triaxial Test Report: Boring EOS-1A, Sample US1D                              |
| B-10          | Triaxial Test Report: Boring EOS-1A, Sample US1E                              |
| B-11          | Triaxial Test Report: Boring EOS-1A, Sample US1F                              |
| B-12          | Triaxial Test Strength Summary: Boring EOS-1A, Sample<br>US4E                 |
| B-13          | Triaxial Test Report: Boring EOS-1A, Sample US4E                              |
| B-14          | Triaxial Test Strength Summary: Boring EOS-4A, Samples U04D, U04E, and U04F   |
| B-15          | Triaxial Test Report: Boring EOS-4A, Sample U04D                              |
| B-16          | Triaxial Test Report: Boring EOS-4A, Sample U04E                              |
| B-17          | Triaxial Test Report: Boring EOS-4A, Sample U04F                              |
| B-18          | Triaxial Test Strength Summary: Boring EOS-4A, Samples U07B and U07C          |
| B-19          | Triaxial Test Report: Boring EOS-4A, Sample U07B                              |
| B-20          | Triaxial Test Report: Boring EOS-4A, Sample U07C                              |
| B-21          | Triaxial Test Strength Summary: Boring EOS-5, Samples<br>U02D, U02E, and U02F |
| B-22          | Triaxial Test Report: Boring EOS-5, Sample U02D                               |

B-iii

## 2.5E-92

LIST OF FIGURES (Cont)

| <u>Figure</u> | Title                                                |
|---------------|------------------------------------------------------|
| B-23          | Triaxial Test Report: Boring EOS-5, Sample U02E      |
| B-24          | Triaxial Test Report: Boring EOS-5, Sample U02F      |
| B-25          | Direct Shear Test Summary: Boring EOS-7A, Sample S-4 |
| B-26          | Direct Shear Test Report: Boring EOS-7A, Sample S-4  |
| B-27          | Direct Shear Test Report: Boring EOS-7A, Sample S-4  |
| B-28          | Direct Shear Test Summary: Boring EOS-6, Sample S-4  |
| B-29          | Direct Shear Test Report: Boring EOS-6, Sample S-4   |
| B-30          | Direct Shear Test Report: Boring EOS-6, Sample S-4   |

B-iv

#### B1 INTRODUCTION

The purpose of the laboratory testing program described herein was to evaluate the index and engineering properties of the soil samples recovered from EOS series of borings that are pertinent to the study of the stability of slopes in the vicinity of the emergency outfall structure. The scope of the testing program consisted of the following:

Atterberg Limits and Grain Size Analyses Consolidation Test Triaxial Compression Tests Direct Shear Tests

B2 INDEX TESTS

B2.1 Grain Size Analyses

Eight grain size analyses were performed on split spoon samples, the results of which are shown in Figures B-1, B-2, B-3, B-4, and B-5. Tests were performed in accordance with Appendix V of WES (1970).

B2.2 Atterberg Limits and Natural Water Contents

Atterberg limits and natural water contents were performed on selected split spoon and undisturbed samples as summarized in Table B-1 and Figure B-6. Natural water content determinations were made in accordance with ASTM D2216. Atterberg limits were determined in accordance with the methods presented in Appendix III of WES (1970); however, the grooving tool used was as specified in ASTM D423.

B3 CONSTANT RATE OF STRAIN CONSOLIDATION TESTS

A single constant rate of strain consolidation (CRSC) test was performed on a 2.5-inch diameter by 1.0-inch high specimen of sandy clay trimmed from an undisturbed sample from boring EOS-4 on the riverward slope. Specimen preparation was in accordance with Appendix VIII of WES (1970). Testing was performed according to the procedures described by Wissa and Heilberg (1969).

The results of the test (Figure B-7) indicate that the clay is only slightly overconsolidated with a maximum past pressure of about 5.6 ksf, compared to an in situ vertical effective stress of about 4.6 ksf.

#### B4 TRIAXIAL COMPRESSION TESTS

Twelve consolidated isotropically undrained triaxial compression tests were performed on undisturbed samples of the alluvial soils in accordance with the methods described in Appendix X of WES (1970). Table B-2 summarizes the results shown in Figures B-8, B-9, B-10, B-11, B-12, B-13, B-14, B-15, B-16, B-17, B-18, B-19, B-20, B-21, B-22, B-23, and B-24.

B-1

#### B5 DIRECT SHEAR TESTS

Two direct shear tests were performed on 2.5-inch diameter, remolded specimens of the fine colluvial material, the results of which are presented in Figures B-25, B-26, B-27, B-28, B-29, and B-30 and summarized in Table B-3. The tests were performed in accordance with Appendix IX of WES (1970); sample preparation is described in this section.

Since the residual friction angle was desired, tests were performed on remolded split spoon samples, using only the finer fraction of the fine colluvium. Test 3 was performed on a sample of silty clay which did not contain the coarser fractions, and Test 4 was performed on the minus No. 40 sieve fraction of the sample. The remolded samples were tamped into the direct shear box to obtain an initial specimen height of one inch. The specimens were initially consolidated to approximately twice the in situ vertical effective stress. The normal load on the test specimens was then reduced and the specimens reconsolidated to approximately the in situ vertical effective stress. Tests were terminated after about one inch of cumulative horizontal displacement since material was slaking from between the halves of the direct shear box.

The measured residual friction angles were 22 and 28.4 degrees.

#### B6 REFERENCES

U.S. Army Engineers Waterways Experiment Station (WES). Laboratory Soils Testing. Engineer Manual 1110-2-1906. Department of the Army. 1970.

Wissa, A. and Heilberg, S. New One Dimensional Consolidation Test. Research Report 69-9. Soils Publication No. 229. Prepared by Massachusetts Institute of Technology, Department of Civil Engineering, Cambridge, Massachusetts. 1969.

American Society for Testing Materials. Standard Test Method for Liquid Limit of Soils. ASTM D423-66 (Reapproved 1972).

American Society for Testing Materials. Laboratory Determination of Moisture Content of Soil. ASTM D2216-71.

B-2

# TABLE B-1

#### ATTERBURG LIMITS AND NATURAL WATER CONTENTS

| Boring<br><u>No.</u> | Sample<br><u>No.</u> | Depth<br>(ft)       | Elevation<br>(ft) | Natural<br>Water<br>Content<br>(%) | Liquid<br>Limit<br><u>(%)</u> | Plastic<br>Limit<br><u>(%)</u> | Plasticity<br>Index<br><u>(%)</u> | Finer<br>than<br>No. 200<br>(%) | Finer<br>than<br>No. 200<br>(%) |
|----------------------|----------------------|---------------------|-------------------|------------------------------------|-------------------------------|--------------------------------|-----------------------------------|---------------------------------|---------------------------------|
| EOS-1                | S4                   | 5.5-7.0             | 735.5-734.0       | 22.1                               | 43.7                          | 22.0                           | 21.7                              | 98                              | CL                              |
|                      | S5                   | 7.0-8.5             | 734.0-732.5       | 25.8                               | 31.8                          | 19.2                           | 12.6                              |                                 | CL                              |
|                      | S6<br>S7             | 8.5-10.0<br>10-11.5 | 731.0-729.5       | 26.7<br>15.9                       | 28.5<br>23.8                  | 17.8                           | 6.8                               | 95<br>42                        | ML-CL                           |
|                      | S8                   | 11.5-13.0           | 729.5-728.0       | 19.8                               | 24.6                          | 19.8                           | 4.6                               | 94                              | ML-CL                           |
|                      | S10                  | 14.5-16.0           | 726.5-725.0       | 19.7                               | 21.1                          | 13.7                           | 7.4                               |                                 | ML-CL                           |
|                      | S12                  | 17.5-19.0           | 723.5-722.0       | 26.8                               | 24.8                          | 22.8                           | 2.0                               | 96                              | ML                              |
|                      | S13(T)               | 19.0-20.5           | 722.0-720.5       | 22.4                               |                               |                                |                                   |                                 | SM                              |
|                      | S14                  | 20.5-22.0           | 720.5-719.0       | 29.5                               |                               |                                |                                   |                                 | SM                              |
|                      | S15 (T)              | 22.0-23.5           | 719.0-717.5       | 28.4                               |                               |                                |                                   |                                 | ML                              |
|                      | S15(B)               | 22.0-23.5           | 719.0-717.5       | 13.1                               |                               |                                |                                   |                                 | SM                              |
|                      | S16                  | 23.5-25.0           | 717.5-716.0       | 7.7                                |                               |                                |                                   |                                 | SP                              |
|                      | S17                  | 25.0-26.5           | 716.0-714.5       | 29.6                               |                               |                                |                                   |                                 | SP                              |
| EOS-1A               | US1E                 | 11.1-11.6           | 729.9-729.4       | 27.6                               | 26.0                          | 21.2                           | 4.8                               | 96                              | ML-CL                           |
| EOS-4                | US185                | 36.9-37.1           | 683.2-683.0       | 28.5                               | 42.0                          | 22.5                           | 19.5                              | 88                              | CL                              |
|                      | S16                  | 38.0-39.5           | 682.1-680.6       | 26.2                               | 35.0                          | 22.5                           | 12.5                              | 87                              | CL                              |
|                      | S18                  | 45.0-46.5           | 675.1-673.6       | 27.0                               | 39.9                          | 23.7                           | 16.2                              | 90                              | CL                              |
| EOS-4A               | U04E                 | 47.2-47.8           | 673.2-672.6       | 26.5                               | 34.5                          | 19.5                           | 15.0                              |                                 | CL                              |
|                      | S4                   | 52.5-54.0           | 667.9-666.4       | 31.0                               | 34.5                          | 21.6                           | 12.9                              | 77                              | CL                              |
|                      | U07C                 | 58.8-59.3           | 661.6-661.1       | 25.7                               | 30.2                          | 17.1                           | 13.1                              |                                 | CL                              |
|                      | U07F                 | 59.9-60.4           | 660.6-660.0       | 25.6                               | 30.7                          | 17.5                           | 13.2                              |                                 | CL                              |
|                      | S6                   | 60.5-62.0           | 659.9-658.4       | 28.6                               | 28.7                          | 20.2                           | 8.5                               | 68                              | CL                              |
| EOS-5                | S3                   | 5.0-6.5             | 678.0-676.5       | 37.8                               | 52.9                          | 30.5                           | 22.4                              | 82                              | MH                              |
|                      | S6                   | 13.5-15.0           | 669.5-668.0       | 29.2                               | 33.0                          | 22.7                           | 10.3                              | 71                              | CL                              |
|                      | U02E                 | 18.7-19.3           | 664.3-663.7       | 26.7                               | 31.4                          | 19.1                           | 12.3                              |                                 | CL                              |
|                      | S8                   | 20.0-21.5           | 663.0-661.5       | 24.0                               | 28.4                          | 16.2                           | 12.2                              | 70                              | CL                              |
|                      | S9                   | 24.0-25.5           | 659.0-657.5       | 28.9                               | 31.9                          | 17.8                           | 14.1                              | 76                              | CL                              |
| EOS-6                | S2                   | 2.0-3.5             | 743.1-741.6       | 19.1                               | 36.3                          | 23.6                           | 12.7                              | 66                              | CL                              |
|                      | S4                   | 6.0-7.5             | 739.1-737.6       | 19.2                               | 42.0                          | 19.5                           | 22.5                              |                                 | CL                              |
|                      | S6                   | 10.0-11.5           | 735.1-733.6       | 22.3                               | 35.3                          | 19.2                           | 16.1                              |                                 | CL                              |
|                      | S7                   | 12.0-13.5           | 733.1-731.6       | 27.8                               | 26.2                          | 19.3                           | 6.9                               | 96                              | CL-ML                           |

1 of 2

2.5E-96

# TABLE B-1 (Cont)

| Boring<br>No. | Sample<br><u>No.</u> | Depth<br>(ft) | Elevation<br>(ft)        | Natural<br>Water<br>Content<br>(%) | Liquid<br>Limit<br><u>(%)</u> | Plastic<br>Limit<br><u>(%)</u> | Plasticity<br>Index<br><u>(%)</u> | Finer<br>than<br>No. 200<br>(%) | Finer<br>than<br>No. 200<br>(%) |
|---------------|----------------------|---------------|--------------------------|------------------------------------|-------------------------------|--------------------------------|-----------------------------------|---------------------------------|---------------------------------|
| FORG          | 011                  | 20 0 21 E     | 70E 1 700 C              |                                    | 20.2                          | 24 1                           | E 1                               |                                 | МТ                              |
| EOS-6         | SII<br>C10           | 20.0-21.5     | 725.1 - 725.0            | 25.0                               | 29.2                          | 24.I<br>20 E                   | 5.1                               |                                 |                                 |
| (COIL)        | 51Z<br>012           |               | 723.1-721.6              | 25.4                               | 27.0<br>Nom                   | 20.5                           | 0.5                               |                                 |                                 |
|               | 513                  | 24.0-25.5     | /21.5-/19.6              | 27.6                               | NOII-                         |                                |                                   |                                 |                                 |
|               | S14                  | 26 0-27 5     | 719 1-717 6              | 30 7                               | 30 5                          | 197                            | 10 8                              | 99                              | CT                              |
|               | S15                  | 28 0-29 5     | 717 1 - 715 6            |                                    | Non-                          |                                |                                   |                                 |                                 |
|               | 515                  | 20.0 29.0     | , 1, 1, 1, 1, 1, 0, 1, 0 |                                    | plastic                       |                                |                                   |                                 |                                 |
| EOS-7         | S4                   | 6.0-7.5       | 753.9-752.4              | 17.0                               | 34.0                          | 23.4                           | 10.6                              | 54                              | CL                              |
| EOS-7A        | S4                   | 17.0-18.5     | 742.6-741.1              | 22.8                               | 42.9                          | 18.4                           | 24.5                              |                                 | CL                              |
| EOS-10        | S9                   | 35.5-37.0     | 685.2-683.7              | 21.5                               | 36.1                          | 19.8                           | 16.3                              | 75                              | CL                              |
| Test Pit 1    |                      | 0.0-7.0       | 733.5-726.5              |                                    |                               |                                |                                   | 98                              | ML                              |
|               |                      | 9.5-10.5      | 724.0-723.0              |                                    |                               |                                |                                   | 99                              | ML                              |
| Test Pit 2    |                      | 1.1-7.5       | 732.4-726.0              |                                    |                               |                                |                                   | 92                              | ML                              |

2 of 2
## TABLE B-2 SUMMARY OF CONSOLIDATED ISOTROPICALLY - UNDRAINED (CIUC) TRIAXIAL COMPRESSION TESTS

|                      |                                 |               |                   |                  |                  |                         | INITIAL                        | SPECIME              | EN PROPER<br>AFTER CO   | TIES<br>DNSOLIDAT              | ION                  |                               |                           |                                             |                                 |                                                            |                                                                                   |                                     |
|----------------------|---------------------------------|---------------|-------------------|------------------|------------------|-------------------------|--------------------------------|----------------------|-------------------------|--------------------------------|----------------------|-------------------------------|---------------------------|---------------------------------------------|---------------------------------|------------------------------------------------------------|-----------------------------------------------------------------------------------|-------------------------------------|
| Boring<br><u>No.</u> | Sample<br>and<br><u>Section</u> | Depth<br>(ft) | Elevation<br>(ft) | Diameter<br>(in) | Height<br>_(in)_ | Water<br>Content<br>(%) | Dry<br>Unit<br>Weight<br>(pct) | Void<br><u>Ratio</u> | Water<br>Content<br>(%) | Dry<br>Unit<br>Weight<br>(pct) | Void<br><u>Ratio</u> | Liquid<br>Limit<br><u>(%)</u> | Plastic<br>Limit<br>_(%)_ | Effective<br>Confining<br>Pressure<br>(ksf) | Back<br>Pressure<br>uo<br>(ksf) | (σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(ksf) | Vertical<br>strain at<br>(σ <sub>1</sub> - σ <sub>3</sub> ) <sub>max</sub><br>(%) | Soil Description                    |
| EOS-1A               | US1D                            | 10.5          | 730.5             | 2.9              | 6.5              | 28.2                    | 95.8                           | 0.747                | 26.3                    | 98.6                           | 0.697                |                               |                           | 6.0                                         | 6.5                             | 7.2                                                        | 14.4                                                                              | Silty Clay-Clayey                   |
|                      | US1E                            | 11.1          | 729.9             | 2.9              | 6.5              | 27.6                    | 96.2                           | 0.740                | 26.2                    | 98.3                           | 0.702                | 26.0                          | 21.2                      | 3.0                                         | 6.5                             | 6.2                                                        | 14.8                                                                              | Silty Clay-Clayey                   |
|                      | US1F                            | 11.6          | 729.4             | 2.9              | 6.0              | 28.1                    | 97.2                           | 0.721                | 27.5                    | 98.2                           | 0.703                |                               |                           | 1.5                                         | 5.8                             | 5.8                                                        | 15.9                                                                              | Silty Clay-Clayey<br>silt           |
| EOS-1A               | US4E                            | 20.8          | 720.2             | 2.9              | 6.0              | 24.3                    | 91.4                           | 0.829                | 29.5                    | 92.2                           | 0.814                |                               |                           | 3.0                                         | 13.0                            | 6.3                                                        | 13.9                                                                              | Layered Silt and<br>Silty Fine Sand |
| EOS-41               | UO4D                            | 46.6          | 673.8             | 2.9              | 7.1              | 26.5                    | 98.5                           | 0.699                | 24.1                    | 102.9                          | 0.626                |                               |                           | 5.5                                         | 5.8                             | 5.3                                                        | 14.9                                                                              | Sandy Clay                          |
|                      | UO4E                            | 47.2          | 673.2             | 2.9              | 7.0              | 26.5                    | 98.3                           | 0.702                | 23.4                    | 103.8                          | 0.612                | 34.5                          | 19.5                      | 7.0                                         | 6.5                             | 5.9                                                        | 15.0                                                                              | Sandy Clay                          |
|                      | UO4F                            | 47.8          | 672.6             | 2.9              | 7.0              | 27.8                    | 96.4                           | 0.735                | 23.1                    | 104.0                          | 0.608                |                               |                           | 10.0                                        | 6.5                             | 7.9                                                        | 15.6                                                                              | Sandy Clay                          |
| EOS-4A               | UO7B                            | 58.2          | 662.2             | 2.5              | 6.2              | 26.0                    | 99.5                           | 0.681                | 21.2                    | 107.9                          | 0.551                |                               |                           | 14.0                                        | 9.4                             | 10.7                                                       | 14.0                                                                              | Sandy Clay                          |
|                      | UO7C                            | 58.8          | 661.6             | 1.4              | 3.5              | 25.7                    | 103.3                          | 0.619                | 22.3                    | 109.6                          | 0.526                | 30.2                          | 17.1                      | 7.0                                         | 10.1                            | 6.1                                                        | 8.8                                                                               | Sandy Clay                          |
| EOS-5                | UO2D                            | 18.1          | 664.9             | 2.9              | 7.0              | 27.9                    | 97.4                           | 0.718                | 26.1                    | 99.3                           | 0.685                |                               |                           | 2.00                                        | 7.2                             | 2.9                                                        | 14.4                                                                              | Sandy Clay                          |
|                      | UO2E                            | 18.7          | 664.3             | 2.9              | 7.0              | 26.7                    | 97.5                           | 0.716                | 25.0                    | 100.1                          | 0.671                | 31.4                          | 19.1                      | 4.0                                         | 7.2                             | 4.5                                                        | 12.9                                                                              | Sandy Clay                          |
|                      | UO2F                            | 19.3          | 663.7             | 2.9              | 7.0              | 28.5                    | 94.8                           | 0.764                | 25.7                    | 98.8                           | 0.694                |                               |                           | 6.0                                         | 9.4                             | 5.3                                                        | 14.8                                                                              | Sandy Clay                          |

## TABLE B-3

## SUMMARY OF DIRECT SHEAR TESTS

| Boring<br><u>No.</u> | Sample<br>No.*       | Depth<br>_(ft) | Elevation<br>(ft) | Water<br>Content<br>(%) | Liquid<br>Limit<br>_(%) | Plastic<br>Limit<br>_(%) | Residual<br>Friction<br>Angle<br>(Degrees) | <u>Symbol</u> |  |
|----------------------|----------------------|----------------|-------------------|-------------------------|-------------------------|--------------------------|--------------------------------------------|---------------|--|
| EOS – 7A<br>EOS – 7  | S4 (90%)<br>S9 (10%) | 17.0-18.5      | 742.6-741.1       | 22.8                    | 42.9                    | 18.4                     | 28.2                                       | CL            |  |
| EOS-6                | S4 (90%)<br>S3 (10%) | 6.0-7.5        | 739.1-737.6       | 19.2                    | 42.0                    | 19.5                     | 22.0                                       | CL            |  |

## NOTE:

\* Remolded specimen consists of material from split spoon samples in the percentages indicated. Additional data provided only for major constituent of test specimen. Specimen dimensions: 2.5-inch diameter by 1.0-inch height.

1 of 1